Topic: Land Use Planning; Zoning; Comprehensive

Planning

Resource Type:RegulationsState:LouisianaJurisdiction Type:Municipal

Municipality: City of Mandeville

Year (adopted, written, etc.): 1993-2002

Community Type – applicable to: Urban; Suburban

Title: City of Mandeville Comprehensive Land Use

Regulation Ordinance

Document Last Updated in Database: March 2, 2016

Abstract

Adopted in 1993 and last updated in June 2002, this very extensive comprehensive plan serves as the zoning ordinance for the city.

Resource

COMPREHENSIVE LAND USE REGULATION ORDINANCE

ORDINANCE #92-14 Adopted June 10, 1993 Latest Revision June 27, 2002 Table of Contents

DIVISION I - GENERAL AND ADMINISTRATIVE PROVISIONS

ARTICLE 1 - TITLE AND BASIC PROVISIONS

1.1	<u>TITLE</u>	. 1
1.2	SCOPE AND PURPOSE	. 1
	1.2.1 Scope of Enactment	. 1
	1.2.2 Content Description	. 1
	1.2.3 Purpose	. 2
	1.2.4 Character of Community	. 2
1.3	BASIC INTENT PROVISIONS	. 2
	1.3.1 Consistency with the Comprehensive Plan	. 2
	1.3.2 Applicability	. 3
	1.3.3 Conflicting Provisions	. 3
	1.3.4 Relief from other Provisions	
	1.3.5 Captions & Headings	
	1.3.6 Severability of Provisions	. 4
1.4	AMENDMENTS TO THE LAND USE REGULATIONS	. 4

1.5 <u>RESERVED</u>	4
1.6 RESERVED	4
1.7 <u>RESERVED</u>	4
1.8 <u>RESERVED</u>	4
1.9 ENFORCEMENT, VIOLATIONS AND PENALTY PROVISIONS	4
1.9.1 Administration of Enforcement	
1.9.2 Violations & Enforcement Proceedings	5
01.9.3 Responsibility for Violations	
1.9.4 Compliance Information Required	
1.9.5 Penalty	
ADMICLE O CENTED AT A DAMINICAD AMINE DE ONICIONO	
ARTICLE 2 - GENERAL ADMINISTRATIVE PROVISIONS	_
2.1 MANDEVILLE PLANNING COMMISSION PROVISIONS	
2.1.1 Mandeville Planning Commission Reauthorized	
2.1.2 Number and Terms of Members, Appointments, Qualification & F	
2.1.3 Election and Term of Chairman	
2.1.4 Rules and Records of Proceedings, Meetings and Quorum	
2.1.5 Compensation	
2.1.6 Expenditures	
2.1.7 Personnel and Duties of Secretary	
2.1.8 Powers and Duties of the Planning Commission	
2.2 MANDEVILLE ZONING COMMISSION PROVISIONS	
2.2.1 Mandeville Zoning Commission Reauthorized	
2.2.2 Numbers and Terms of Members, Appointments, Qualification	ons and
Removal	0
2.2.3 Election and Term of Chairman	
2.2.4 Rules and Records of Proceedings, Meetings and Quorum	
2.2.5 Compensation and Expenditures	
2.2.6 Personnel and Duties of Secretary	
2.2.7 Powers and Duties of the Zoning Commission	
2.2.8 Additional Powers and Duties of the Zoning Commission	
2.3 RESERVED	
2.4 COMMUNITY APPEARANCE COMMISSION PROVISIONS	
2.4.1 Community Appearance Commission Reauthorized	
2.4.2 Number and Terms of Members, Appointments, Qualifica	
Removal	
2.4.3 Flection and Term of Chairman	14

2.4.4 Rules and Records of Proceedings, Meetings and Quorum	11
2.4.5 Compensation	
2.4.6 Expenditures	
2.4.7 Appointment of a Secretary	
2.4.8 Powers and Duties of the CAC	
2.4.9 Appeals of CAC Decisions	
2.5 RESPONSIBILITIES OF CITY DEPARTMENTS AND STAFF	
2.5.1 Responsibilities of Department of Planning and Development in	10
Administering the CLURO	16
2.5.2 Qualifications and Duties of the City Planner	
2.5.3 Qualifications and Duties of the Building Inspector	
2.5.4 Responsibilities of Public Works Director	
2.5.1 Responsibilities of 1 ubite works bifector	1
ARTICLE 3 - GENERAL DEFINITIONS	
3.1 GENERAL PROVISIONS	20
3.2 GENERAL INTENT	
3.3 GENERAL DEFINITIONS OF THE LAND USE REGULATIONS	
3.5 GENERAL DEL INTEGNO OF THE BRIND COL REGUERITIONS	20
DIVISION II - BUILDING AND ZONING REGULATIONS	
DIVIDION II DOIDDING INVO ZONING NEGOEITIONS	
ARTICLE 4 - GENERAL AND NON-CONFORMING PROVISIONS AND	
ADMINISTRATIVE PROCEDURES	
4.1 GENERAL BUILDING AND ZONING PROVISIONS	51
4.1.1 Zoning Affects Every Structure & Use	
4.1.2 Reduction in Lot Area	
4.1.3 Required Yard Areas	
4.1.4 Change in Zoning District Boundary	
4.2 NON-CONFORMING PROVISIONS	
4.2.1 General Non-Conforming Provisions	
4.2.2 Provisions for Legally Non-Conforming Uses	
4.2.3 Provisions for Legally Non-Conforming Development Sites	
4.2.4 Provisions for Non-Conforming Structures/Buildings	
4.2.5 Provisions for Legally Non-Conforming Lots-of-Record	
4.3 ADMINISTRATIVE PROCEDURES	50
4.3.1 Procedures & Fees for Amendments to the Land Use Regulations	59
4.3.2 Procedures & Fees for Site Plan Review for	
Administrative & Zoning Permits	62
4.3.3 Procedures & Fees for Conditional Use Permits &	02
now i recauted a record deliamental ober crimes a	

Planned District Zoning	67
4.3.4 Procedures & Fees for Filing Appeals & Requests	
for Variances to the Zoning Board	75
4.3.5 Procedures & Fees for Issuance of a Home Occupation Permit	
ARTICLE 5 - BUILDING CODES AND PERMITTING REQUIREMENTS	
5.1 GENERAL PROVISIONS	83
5.1.1 Title	83
5.1.2 Building Inspector as Administrator	83
5.1.3 Code Remedial	
5.1.4 Scope	
5.1.5 Rules & Procedures of Building Inspector	83
5.1.6 Permit Required	84
5.1.7 Permit Fees, General	84
5.1.8 Failure to Obtain a Permit	84
5.1.9 Double Permit Fees as Penalty	84
5.1.10 Right of Entry	84
5.1.11 Posting Permit & Inspection Cards	84
5.1.12 Stop Work Orders	
5.1.13 Revocation of Permits	85
5.1.14 Issuance of Permit Not a Waiver	85
5.1.15 Unsafe Buildings	85
5.1.16 Floor Loads	86
5.1.17 Floor Load Signs Required	86
5.1.18 Loads in Excess of Posted Capacity	
5.1.19 Establishing Guidelines For Potential Damage to City Streets,	Rights-
Of-Way,	
and Drainage During New Construction Or	Major
Renovation	87
5.2 <u>CODES ADOPTED</u>	87
5.2.1 Codes of Southern Building Code Congress International, Inc	87
5.2.2 National Electric Code	
5.2.3 Grading Supplement	
5.2.4 Plumbing Supplement	
5.2.5 Sewer & Water Supplement	
5.2.6 Storm Drainage Supplement	
5.3 GENERAL PERMITTING AND INSPECTION REQUIREMENTS	94
5.3.1 When a Building Permit is Required	

	FOOG TIME IN THE TOTAL TO THE	o 4
	5.3.2 Special Approvals Required Prior to Permits	
	5.3.3 General Procedure For All Permit Applications	
	5.3.4 General Requirements for Development Permit Applications	
	5.3.5 Waiving of Requirements for Submittal of Plans	
	5.3.6 Amendment of Permit Documents	
	5.3.7 Conditions Attached to a Building Permit	
	5.3.8 Abandonment & Extension of Building Permits	
	5.3.9 Adequate Waste Disposal for Each Building Site During Construction.	
	5.3.10 Action on Application for a Building Permit	
	5.3.11 Inspection - General	
	5.3.12 Inspections Required	98
	5.3.13 Minimum Inspections Required	
	5.3.14 Inspection & Approval	
	5.3.15 Prohibition of Concealing in any Manner	
	5.3.16 Approval to Apply Plaster	
5.4	PERMITTING REQUIREMENTS BY PERMIT TYPE	99
	5.4.1 Development Permits	99
	5.4.2 Permit Application Requirements for Single Family	
	Residences & Two-Family Residences1	00
	5.4.3 Development Permit Application Requirements for	
	Non-Residential & Multi-Family Development1	01
	5.4.4 Application Requirements for Development Permits for	_
	Residential Renovations & Accessory Uses 1	02
	5.4.5 Requirements for Public Improvement Permits1	
55	FEES FOR PERMIT APPLICATIONS	
0.0	5.5.1 General Structure of Fees for Development Permits for Buildings1	
	5.5.2 Fees for Development Permits1	
	5.5.3 Required Building Permit Fees for Minor Remodeling & Renovation 1	
	5.5.4 Required Fees for Sign Permits1	
	5.5.5 Required Fees for Landscape and/or Tree & Shrub Removal Perm	
	5.5.6 Required Fee for Clearing Permit1	
		UJ
	5.5.7 Required Fees for Single &	ΛΓ
	Two-Family Residential Accessory Use Permits	
	5.5.8 Required Fees for Plumbing Permits & Inspections	
	5.5.9 Required Fees for Electrical Permits & Inspections	
	5.5.10 Required Fees for Mechanical Equipment Permits & Inspections1	
	5.5.11 Reinspection Fees	
	5.5.12 Fees for Public Improvement Permits1	07

5.6 CERTIFICATES OF OCCUPANCY PROVISIONS	107
5.6.1 Requirement for Certificate of Occupancy	
5.6.2 Certificate of Occupancy for New Construction	
5.6.3 Temporary Certificate of Occupancy	
5.6.4 Certificate of Occupancy for Existing Buildings	
5.6.5 Records to be Maintained by Building Inspector	
5.7 CERTIFICATES OF COMPLETION	
5.8 REQUIRED LICENSING AND BONDING OF CONTRACTORS	108
5.8.1 Bonding of Contractors Required	
5.8.2 Licensing of Contractors	
ARTICLE 6 - LAND USE CLASSIFICATIONS	
6.1 PURPOSE AND GENERAL PROVISIONS OF USE CLASSIFICATIONS	112
6.2 RESIDENTIAL USE CLASSIFICATIONS	
6.3 CIVIC USE CLASSIFICATION	
6.4 COMMERCIAL USE CLASSIFICATIONS	
6.5 INDUSTRIAL USE CLASSIFICATIONS	
6.6 AGRICULTURAL USE CLASSIFICATIONS	
6.7 COMBINED USE CLASSIFICATIONS	
6.8 ACCESSORY USE CLASSIFICATIONS	139
ARTICLE 7 - ZONING DISTRICT REGULATIONS	
7.1 TITLE AND PURPOSE	1/1
7.2 ADOPTION, AMENDMENT & INTERPRETATION OF ZONING DISTRICTS	
7.2.1 Creation of Zoning Districts	
7.2.2 Application of Base & Overlay Districts	
7.2.3 Ordinance Linkage with Official Zoning Map	
7.2.4 Amendments to Official Zoning Map	
7.2.5 Determination of District Boundaries	
7.3 DEFINITIONS APPLICABLE TO ZONING DISTRICT REGULATIONS	
7.4 GENERAL PROVISIONS OF THE ZONING DISTRICT REGULATIONS	
7.5 BASE DISTRICT REGULATIONS BY ZONING DISTRICTS	
7.5.1 R-1 Single-Family Residential District	
7.5.2 R-1X Single-Family Existing Residential District	
7.5.3 R-2 Two-Family Residential District	
7.5.4 R-3 Multi-Family Residential District	
7.5.5 MH Mobile Homes District	
7.5.6 I-Institutional District	
7.5.7 O-Open Space/Recreational District	
· · · · · · · · · · · · · · · · · · ·	

	7.5.8 B-1 Neighborhood Business District	158
	7.5.9 B-2 Highway Business District	162
	7.5.10 B-3 Old Mandeville Business District	166
	7.5.11 B-4 Major Crossroads Business District	171
	7.5.12 O/R - Office/Residential District	
	7.5.13 PM-1 Planned Marina District - Waterfront Lots	180
	7.5.14 PM-2 Planned Marina District - Non-Waterfront Lots	185
	7.5.15 PD Planned District	189
	7.5.16 M-1 Light Manufacturing District	192
	7.5.17 M-2 General Manufacturing District	
	7.5.18 TC Town Center District	
7.6	OVERLAY ZONING DISTRICTS	197
	7.6.1 D-0 Drainage Overlay District	197
	7.6.2 G-O Gateway Overlay District	199
	7.6.3 L-O Lakefront Overlay District	
	7.6.4 TC Town Center Overlay District	
7.7 <u>T</u>	ABLE OF PERMITTED USES BY ZONING DISTRICT	
	7.7.1 Table of Permitted Uses By Zoning District	
	7.7.2 Use of Symbols in Table of Permitted Uses	
	7.7.3 Zoning Board Jurisdiction over Administrative Use Permits	212
ARTICLE 8 -	SUPPLEMENTAL DEVELOPMENT AND USE REGULATIONS	
	UPPLEMENTAL REGULATIONS - GENERAL	223
0.1 <u>5</u>	8.1.1 Supplemental Lot, Yard & Open Space Regulations	
	8.1.2 Supplemental Use Regulations	
	8.1.3 Supplemental Fence & Wall Regulations	
	8.1.4 Supplemental Regulations for Temporary Buildings	
	8.1.5 Supplemental Regulation of Accessory Buildings & Structures	
	8.1.6 Supplemental Regulations for Manufactured Modular Buildings	
	8.1.7 Supplemental Regulations for Recreational Vehicles	
	8.1.8 Standards for Emergency, Construction or Repair	201
	Residences on Construction Sites	232
	8.1.9 Construction of Buildings on Improved Streets Only	
	8.1.10 Supplemental Regulations for Outdoor Lighting	
8.2 S	PECIAL USES CRITERIA	
5.2 <u>5</u>	8.2.1 Special Residential Use Criteria	
	8.2.2 Special Civic Uses Criteria	
	8.2.3 Special Commercial Uses Criteria	

8.2.4 Special Industrial Uses Criteria	259
8.2.5 Special Combined Uses Criteria	
8.2.6 Special Accessory Uses Criteria	
8.3 FLOOD DAMAGE PREVENTION REGULATIONS	
8.3.1 Authorization, Findings of Fact, Purpose & Methods	
8.3.2 Definitions of Flood Damage Prevention Regulations	
8.3.3 General Provisions	
8.3.4 Administration	271
8.3.5 Provisions for Flood Hazard Reduction	274
8.4 TRAFFIC IMPACT ANALYSIS PROVISIONS	
8.4.1 Title & Purpose	279
8.4.2 Traffic Impact analysis	279
8.4.3 Threshold Conditions Requiring a Traffic Impact Analysis	279
8.4.4 Responsibility for Required Traffic Impact Analysis	
8.4.5 Scope of Traffic Impact Analysis & Study Area	
8.4.6 Traffic Data Requirements	
8.4.7 Application of Requirements	284
8.4.8 Actions Based on the Results of a Traffic Analysis	284
ARTICLE 9 - PARKING, LANDSCAPING, AND OPEN SPACE REGULATIONS	
9.1 PARKING & LOADING REGULATIONS	286
9.1.1 General Provisions	
9.1.2 Construction Design Standards for Parking & Loading	
9.1.3 Location & Maintenance	293
9.1.4 Minimum Off-Street Parking Requirements by Use	
9.1.5 Off-Street Loading & Unloading Areas	
9.1.6 Parking Plan Approval	
9.1.7 Inspection of Parking Prior to Construction	
9.2 LANDSCAPE REGULATIONS	
9.2.1 Short Title	
9.2.2 Purpose & Intent	
9.2.3 Definitions of Landscape Regulation Terminology	
9.2.4 Provisions for Landscaping on Public Property	310
9.2.5 Provisions for Landscaping on Private Property	
9.2.5.1 Obstructions to Vision Prohibited	314
9.2.5.2 Vegetation Protection Zones	314
9.2.5.3 Vegetation Protection Zone Barriers	314
9.2.5.4 Landscape Requirements in Low-Density Residential District	
9.2.5.5 Landscape Requirements in Districts Other than	

Low-Density Residential	315
9.2.5.5.1 Periphery Landscape (Greenbelt) Requirements	
9.2.5.5.2 Site Interior Planting Regulations	316
9.2.5.5.3 Buffer Zone Requirements	
9.2.5.6 Landscape Maintenance & Replacement Provisions	319
9.2.5.7 Live Oak Protection Requirements	
9.2.5.8 Screening of Trash & Garbage Cans	
9.2.5.9 Supplemental Planting & Plan Preparation Requirements	321
9.2.5.10 Emergencies	
9.2.5.11 List of Trees, Shrubs & Ground Covering Materials	323
9.2.5.12 Landscape Permit Required	
9.2.5.13 Clearing Permit Required	
9.2.5.14 Tree & Shrub Pruning or Removal Permit Required	
9.2.5.15 Inspection of Landscape Installations	324
9.2.5.16 Violations	
9.3 PARKING IMPACT FEE SECTION	325
9.3.1 Intent & Purpose of Section	325
9.3.2 Grant of Parking Variances in the B-3 Old Mandeville Business	District
	326
9.3.3 Parking Mitigation Fee; Use of Funds	326
9.3.4 Required Connection between Parking Deficiency	
& Existing Buildings or Structures	
9.3.5 Other Solutions to Parking Deficiencies not Prohibited	328
9.3.6 Return of Funds	
9.3.7 Change of Use; Credit for Prior Mitigation	330
9.3.8 Penalty for Violation	330
ARTICLE 10 - SIGN CODE	
10.1 <u>TITLE</u>	331
10.2 STATEMENT OF PURPOSE AND NEED FOR REGULATION	331
10.3 <u>DEFINITIONS</u>	332
10.4 PROHIBITED SIGNS	341
10.5 REGULATION OF ON-PREMISE SIGNS	
10.5.1 On-Premise Signs Allowable without a Permit	342
10.5.2 On-Premise Signs Allowable by Temporary Permit Only	347
10.5.3 On-Premise Signs Allowed in all Residential Zoning Districts	351
10.5.4 On-Premise Signs Allowed in Non-Residential &	
Combined Use Districts	352
10.5.5 Comprehensive Signage Plans	359

	10.6 <u>REGULA</u>	<u>TIONS OF OFF-PREMISE SIGNS</u>	361
	10.6.1	Off-Premise Signs Allowable without a Permit	361
	10.6.2	Off-Premise Signs Allowable by Temporary Permit Only	361
	10.6.3	Regulations of all Off-Premise Signs; Height	
		Clearance & Setback Requirements	361
	10.6.4	Regulation of Off-Premise Signs; Maximum Sign Area	
		Regulation of Off-Premise Signs; Permitted & Prohibite	
	10.6.6	Limitation of Number of Off-Premise Sign Faces per Sig	n Structure
	10.6.7	Regulation of Public Information Signs	363
	10.7 <u>DESIGN</u>	AND CONSTRUCTION STANDARDS FOR SIGNS	363
	10.7.1	Compliance with Building Code	363
	10.7.2	Wind Pressure; Design Requirements & Working Stresses	363
	10.7.3	Material Specifications	364
		Auxiliary Specifications	
	10.8 <u>ELECTR</u>	ICAL SIGNS REGULATIONS	365
	10.8.1	Electrical Code	365
	10.8.2	Licensing	366
		Electrical Sign Permit	
	10.8.4	Electrical Inspection Fees	366
	10.9 <u>PROCED</u>	URES AND FEES FOR SIGN PERMITS; SIGN INSPECTIONS	
		AND CERTIFICATES OF COMPLETION	366
	10.9.1	Requirement to Obtain a Sign Permit	366
	10.9.2	Application Requirements for Issuance of Sign Permits	366
	10.9.3	Procedure for Issuance of Sign Permit; Required Review of	Application
by			
		Community Appearance Commission; Appeals;	Display of
		Proposed	
		Signage pending Community Appearance Commiss	sion Review
			368
	10.9.4	Landscaping Requirements in Conjunction with the	
		Issuance of a Sign Permit	369
	10.9.5	Encroachment on Utilities in Conjunction with the	
		Issuance of a Sign Permit	369
	10.9.6	Issuance of Permit not a Waiver	

10.9.7 Commencement of Work under Sign Permit; Automatic Expi	ration
	370
10.9.8 Suspension or Revocation of Sign Permit	
10.9.9 Required Fees for Sign Permits	370
10.9.10 Regulations for Inspection of Signs & Issuance of	
Certificate of Completion	
10.10 <u>LICENSING OF SIGN CONTRACTORS</u>	
10.11 <u>COMMUNITY APPEARANCE COMMISSION</u>	372
10.12 ENFORCEMENT OF THE PROVISIONS OF THE SIGN CODE & PENALTY	
FOR VIOLATION	
10.12.1 Maintenance of Signs & Premises	
10.12.2 Regulation of Legally Non-Conforming Signs	
10.12.3 Amortization of Legally Non-Conforming Signs	374
10.12.4 Permitting of Conforming Signs Erected without Permit	
prior to this Sign Code	
10.12.5 Unsafe & Unlawful Signs	
10.12.6 Interpretation & Application	
10.12.7 Enforcement & Administration	
10.12.8 Violation	
10.12.9 Penalty	376
DIVISION III - SUBDIVISION AND PUBLIC IMPROVEMENTS REGULATIONS ARTICLE 11 - GENERAL SUBDIVISION AND PUBLIC IMPROVEMENTS PROVISIONS	
11.1 TITLE, PURPOSE AND POLICY	
11.1.1 Title	
11.1.2 Purpose	
11.1.3 Policy	378
11.2 <u>AUTHORITY OF PLANNING COMMISSION AND JURISDICTION</u>	200
OF REGULATIONS	
11.2.1 Authority	
11.2.2 Jurisdiction	
11.3 <u>DEFINITIONS</u>	
11.4 ADMINISTRATION AND ENFORCEMENT	
11.4.1 Responsibilities of City Officials & Personnel	
11.4.2 Classification of Subdivisions & Approvals Required	
11.4.3 Official Submission Dates	
11.4.5 Method of Payment of Fees	
11.4.6 Enforcement	
11.4.7 Violations	
11.4.8 Reconsideration	
11.4.0 1/50011510011	500

11.5 GENERAL PROVISIONS	386
11.5.1 Interpretation & Conflicting Provisions	386
11.5.2 Private Provisions	
11.5.3 Saving Provisions	386
11.5.4 Appeals	386
11.5.5 Waivers	
11.5.6 Amendment of Regulations	387
11.5.7 Advertisement of Requests	387
11.5.8 Vacation of Plats	
11.5.9 Responsibilities of Developers/Applicants	388
11.5.10 Inspection & Approval of Improvements	389
ARTICLE 12 - SUBDIVISION AND PUBLIC IMPROVEMENTS APPLICATIONS & PROCEDURES	
12.1 SUBDIVISION & RESUBDIVISION APPLICATIONS AND	
PROCEDURES	393
12.1.1 Pre-Application Procedures for all Subdivisions/Resubdivisions	
12.1.2 General Procedure	
12.1.3 General Content	
12.1.4 Deadlines for Submission of Applications	
12.1.5 Major Subdivision/Resubdivision Applications & Procedures	
12.1.6 Minor Subdivision/Resubdivision Applications & Procedures	
12.1.7 Resubdivision Approval Procedures	407
12.2 <u>APPLICATION & PROCEDURES FOR APPROVAL OF PUBLIC</u>	
IMPROVEMENTS ONLY	408
12.3 <u>ASSURANCE OF COMPLETION & MAINTENANCE OF</u>	
PUBLIC IMPROVEMENTS	
12.3.1 Assurance of Completion	
12.3.2 Inspection, Acceptance & Maintenance of Improvements	
12.3.3 Acceptance of Dedication Offers	
12.3.4 Maintenance of Improvements	
12.3.5 Issuance of Building Permits & Certificates of Occupancy	
12.4 SUBDIVISION APPLICATION AND ASSOCIATED FEES	
12.4.1 Reserved	
12.4.2 Reserved	
12.4.3 Fees for Approval of Public Improvements Only	414
	414
12.5.1 Information Required on the Conceptual Sketch plat	414
12.5.2 Information Required on the Site Features Map	416
12.5.3 Information Required on the Preliminary PlatPlat	
12.5.4 Information Required on Subdivision & Public Improvements	

Construction Plans & Specifications	418
12.5.5 Information Required on Final Subdivision Plat	
for Major Subdivisions/Resubdivisions	421
12.5.6 Additional Information Required for Resubdivision Plats	
12.5.7 Basis for Rejection of Documents Submitted	
12.3.7 Dasis for rejection of Documents submitted	12 1
ARTICLE 13 - SUBDIVISION AND PUBLIC IMPROVEMENTS DEVELOPMENT	
STANDARDS	
13.1 GENERAL PROVISIONS	425
13.1.1 Monuments	
13.1.2 Alteration of the Topographic Character of the Land	
13.1.3 Improvements in Excess of the Requirements of these Reg	
zorze improvemente in ziroso er une rieden entente er uneste ried	
13.1.4 Off-site Extensions of Utilities	
13.1.5 Minimum Servitudes Required & Minimum Size Allowed	
13.1.6 Size of Drainage Servitude Determines Dedication or Ea	
Status	
13.1.7 Public Use Sites & Open Space	
13.1.8 Control of Erosion & Sedimentation	
13.2 SUBDIVISION LAYOUT DESIGN	
13.2.1 Lot Layout & Improvements	
13.2.2 Block Layout	
13.2.3 Stormwater Drainage Requirements	436
13.2.4 Street Improvement Requirements	
13.2.5 Pedestrian & Bicycle Facilities Requirements	
13.2.6 Sanitary Sewer Facilities Requirements	
13.2.7 Water Facilities Requirements	
13.2.8 General Utility Requirements	
13.2.9 Preservation of Natural Features & Amenities	
13.2.10 Non-Residential Subdivision Layout	450
13.2.11 Other Subdivisions	
13.3 CONSTRUCTION STANDARDS FOR SUBDIVISIONS &	
PUBLIC IMPROVEMENTS	452
13.3.1 General Requirements	
13.3.2 Lot & Block Standards	
13.3.3 Storm Drainage Specifications & Standards	453
13.3.4 Street Specifications & Standards	458
13.3.5 Sidewalk, Pedestrianway, & Bicycle Path Specificat	
Standards	
13.3.6 Sanitary Sewers Design Standards	
13.3.7 Water Facility Design Standards	
13.3.8 Other Utilities Standards	

13.3.9	Standard	s for Prese	ervation o	f Natural	Features	& Amenities.	470
13.3.10) Standar	ds for Nor	n-Resident	ial & Tov	vnhouse S	Subdivisions	470

DIVISION I

GENERAL & ADMINISTRATIVE PROVISIONS ARTICLE 1 - TITLE AND BASIC PROVISIONS

1.1 TITLE

This Document in its entirety, to be codified as Appendix A of Code of Ordinances of the City of Mandeville, and including Divisions I, II and III hereof shall be known as the "Comprehensive Land Use Regulation Ordinance of the City of Mandeville, Louisiana". It may be referred to in short titles as the "Land Use Regulations", or the "CLURO".

1.2 <u>SCOPE AND PURPOSE</u>

1.2.1 Scope of Enactment

This "Land Use Regulations" ordinance adopted in its entirety, including Divisions I, II and III and including the "Official Zoning Map" referred to in Article 6, supersedes and replaces the previously adopted "Appendix A, Zoning", of the City of Mandeville, Code of Ordinances entitled the "Zoning and Restrictions Ordinance of the City of Mandeville, Louisiana" in its entirety; replaces the City of Mandeville Subdivision Regulations, adopted by the Planning Commission, in its entirety; supersedes and replaces the following sections of the City of Mandeville Code of Ordinances: Article II Planning Commission of Chapter 2 - Administration, comprising Sections 2-16 through 2-26; Chapter 5 - Buildings and Structural Appurtenances; Chapter 7 - Electricity; Chapter 8 1/4 - Flood Damage Prevention; Chapter 13 - Streets, Sidewalks and Public Places, Sections 13-6 through 13-9 and 13-14 and 13-15; and Chapter 15 - Trailers; amends Chapter 17, Section 17-79 to move the Sewer Impact Fees to this document; and supersedes and replaces any zoning districts as depicted on any previously prepared or recorded maps or surveys.

1.2.2 Content Description

This "Land Use Regulations" ordinance includes three major divisions of regulation. The first, Division I - General and Administrative Provisions sets forth the basic provisions relevant to the entire "Land Use Regulations" ordinance and establishes, describes and defines the authority of each administrative body having jurisdiction over the various sections of this ordinance. Division II - Building and Zoning Regulations sets forth the regulations governing the building of all structures and use of all land within the incorporated area of the City of Mandeville. Division III - Subdivision and Public

Improvements Regulations sets forth the regulations which apply to the subdivision or resubdivision of land and the design and installation of public improvements both in conjunction with newly created subdivisions and when new improvements are installed in areas previously subdivided.

1.2.3 Purpose

These Land Use Regulations have been developed in accordance with the City of Mandeville's adopted Comprehensive Land Use Plan for the implementation of the goals and policies of the community stated therein and for the purpose of promoting the health, safety, welfare and aesthetics of the community. The regulations have been established to lessen congestion in the streets, secure safety from fire, panic and other dangers; to provide adequate light and air; to prevent the overcrowding of land to avoid undue concentration of population; and to facilitate the adequate provision of vehicular and pedestrian circulation, water, sewerage, storm drainage, schools, parks, open space and other public requirements. regulations have been established with consideration for the character of the City of Mandeville as a whole as well as the character of each zoning district and its peculiar suitability for particular uses to preserve property values and promote the economic well-being of the community as a whole by encouraging the most appropriate land use throughout the City while acknowledging the importance of maintaining the health of the environment for the health of the public. All references to these regulations shall be based upon the regulations as adopted or to any subsequent amendments thereto.

1.2.4 Character of Community

Mandeville has a rich, cultural and historical background. Many structures remain in testimony to this heritage. Since its incorporation in 1840, Mandeville has been a place to live and work in retreat from the stress, congestion and pollution of the metropolitan city. It is important that the land use regulations of the community create a framework for residential and commercial development that is compatible with this heritage.

To that end residential development should be as nearly compatible with that which presently exists, giving consideration to preserve ample lot sizes, adequate setbacks, appropriate height restrictions and preservation of existing vegetation to the greatest degree practicable. In addition, efforts should be made to preserve those structures and settings which exist in Old Mandeville which have historic, cultural and architectural significance.

It is also important to note that commercial enterprises are necessary for the servicing of the needs of the community and for its fiscal stability. To that end commercial enterprises should be encouraged which best fit the needs and character of the city, and enhance the desirability of Mandeville as a place in which to both live and work.

1.3 BASIC INTENT PROVISIONS

1.3.1 Consistency with the Comprehensive Plan

These Land Use Regulations shall be interpreted and applied in a manner consistent with the Mandeville Comprehensive Land Use Plan, also referred to herein as the "Comprehensive Plan", and with any supplemental land use and community development policies which may be adopted by the City Council.

The regulatory framework established by these regulations requires that all amendments to these regulations maintain and enhance, to the extent deemed possible by the City Council, the consistency between these regulations and the provisions of the Mandeville Comprehensive Land Use Plan.

1.3.2 Applicability

These Land Use Regulations shall apply to all of the incorporated areas of the City of Mandeville. The use of or alteration of all land and any buildings or structures located upon the land, and the construction, reconstruction, alteration, expansion, or relocation of any building or structure upon the land shall conform to all applicable provisions of these Land Use Regulations. No land, building, structure or premises shall be used for any purpose or in any manner other than that which is permitted in the zoning district in which such land, building, structure or premise is located except as provided herein.

1.3.3 Conflicting Provisions

These Land Use Regulations shall be held to be the minimum requirements for the promotion of the public health, safety, welfare and aesthetics of the community. The provisions of these regulations are intended to supplement and to be read, construed and applied in with all existing laws, ordinances and regulations of this City so that whenever possible full and due effect is given to all such enactments. The provisions of these regulations shall not be deemed to have repealed or suspended any such existing law, ordinance or regulation of this City unless such result shall have been expressly stated or be clearly intended by the context and language of the provision in question. In the event of a conflict in any particular circumstances between the provisions or requirements of these regulations and the provisions or requirement of any other law, ordinance or regulation of this City the more restrictive provision or requirement shall apply unless a contrary application thereof is expressly directed or clearly intended by the context and language of the laws, ordinances and regulations in question.

1.3.4 Relief from other Provisions

Except as otherwise specifically provided, no provision of these regulations shall be construed as relieving any party to whom a site plan approval, permit or variance is issued from any other provision of state or federal law or from any provision, ordinance, rule or regulation of the City requiring a license, franchise, or permit to accomplish, engage in, carry on or maintain a particular business, enterprise, occupation, transaction or use.

1.3.5 Captions and Headings

Captions and headings to specific parts, sections, and provisions of these regulations are not a part of the provisions or requirements of these regulations or of any particular parts or sections hereof, and are provided merely as a convenience to persons reading or studying this ordinance.

1.3.6 Severability of Provisions

If any section, subsection, sentence, clause or phrase of these regulations is for any reason, held to be invalid or unconstitutional, such invalidity or unconstitutionality shall not affect the validity or constitutionality of the remaining portions of these regulations, it being hereby expressly declared that this ordinance and each section, subsection, sentence, clause and phrase hereof would have been prepared, proposed, adopted, approved and ratified separately irrespective of the fact that any one or more other sections, subsections, sentences, clauses or other phrases be declared invalid or unconstitutional.

1.3.7 Reference to Statutes, Ordinances and Regulations

Any reference or citation made in this Comprehensive Land Use Regulations Ordinance to any duly enacted statute of the United States of America or of the State of Louisiana; or to any duly enacted ordinance of the Parish of St. Tammany or of the City of Mandeville; or to any regulation enacted or promulgated by any department, agency, board, bureau or other institution or subdivision of the United States of America, the State of Louisiana, the Parish of St. Tammany or the City of Mandeville shall refer to the provisions of the cited or referenced statute, ordinance or regulation as of the date of the enactment of this Comprehensive Land Use Regulations Ordinance or as the cited or referenced statute, ordinance or regulation may thereafter from time to time be amended, reenacted or redesignated.

1.4 AMENDMENTS TO THE LAND USE REGULATIONS

The City Council may, from time to time, amend, supplement, or change the provisions of these regulations or the boundaries of the Official Zoning Map as established by these regulations. Such amendments may be initiated and effected according to the manner prescribed in Article 4 under Procedures for Zoning Amendments and Amendments to the Land Use Regulations.

1.5	RESERVED
1.6	RESERVED
1.7	RESERVED
1.8	RESERVED
1.9	ENFORCEMENT, VIOLATIONS AND PENALTY PROVISIONS
1.9.1	Administration of Enforcement

The provisions of these Land Use Regulations shall be administered by the City Planner, the Building Inspector and the Director of the Department of Public Works, as designated herein, who shall have the power to make inspections of buildings or premises necessary to carry out their administrative duties in the enforcement of these regulations. The provisions of these regulations shall be enforced by the City's police; the City Planner, the Building Inspector or the Director of Public Works may call upon the Chief of Police to furnish the necessary police personnel to carry out enforcement.

1.9.2 Violations and Enforcement Proceedings

In case any structure is erected or structurally altered or maintained, or any structure or land is used or altered or is not maintained, in violation of these regulations, any proper City official or his or her duly authorized deputies or representatives may institute any appropriate action or proceedings to prevent such unlawful construction or alteration or use or other violations, to restrain, to correct or to prevent any illegal act, conduct any business or maintain any use in or about such premises. Furthermore, any resident of the community who believes that a violation of any of the provisions of these regulations is occurring may file a written complaint with the Building Inspector. Such complaint shall fully set forth the acts or omissions constituting the alleged violation and the site or sites at which such violation or violations are alleged to be occurring. The Building Inspector shall record properly such complaint, promptly investigate the allegations underlying said complaint, and take action on such complaints as provided by these regulations.

1.9.3 Responsibility for Violations

The owner of any structure or land, or part thereof, and the user of any structure or land or part thereof, where anything in violation of these regulations shall be placed or shall be used, and any architect, builder, contractor, agent, or any other person employed in connection therewith who may have assisted or contributed to the commission of any such violation, may each be deemed responsible for a violation of these regulations.

1.9.4 Compliance Information Required

Whenever the Building Inspector, on the basis of a written complaint from the City Planner, the Director of Public Works, any other City official or any credible person or on the basis of his own available information, has reason to believe that a violation of these regulations may exist, he may require any person owning the structure or land or operating a use thereon to provide, within 30 days of notification, information as may be necessary, in his judgement, to determine the existence or extent of any violation.

1.9.5 Penalty

Any person violating any provision of these regulations shall be guilty of a misdemeanor, and deemed a public nuisance and upon conviction shall be punished for each separate offense by a fine not exceeding five hundred dollars (\$500.00) or imprisonment for a term not exceeding sixty (60) days, or by both such fine and imprisonment, or as provided in Section 1.9 of the Code of Ordinances of the City of Mandeville, whichever is greater. Each day any violation of any provision of these regulations shall continue shall constitute a separate offense.

ARTICLE 2 - GENERAL ADMINISTRATIVE PROVISIONS

2.1 MANDEVILLE PLANNING COMMISSION PROVISIONS

2.1.1 Mandeville Planning Commission Preauthorized

There is hereby preauthorized and continued a municipal planning commission, to be known as the "Mandeville Planning Commission" also designated "the Planning Commission" herein, previously established by City ordinances adopted on May 11, 1954 and February 24, 1977 under the authority of the Louisiana R.S. 33:102.

2.1.2 Number and Terms of Members, Appointments, Qualifications and Removal

The Planning Commission shall consist of seven (7) members who shall serve staggered terms of seven (7) years. Members to serve on this board shall be residents and qualified voters in the City of Mandeville. They shall be appointed and confirmed by a vote of the Council. Should any member of the Commission fail to meet any of the above qualifications, that position shall be declared vacant and another appointment shall be made for the unexpired term. All members presently serving and the expiration dates of their terms will remain the same at the time this is enacted. They are as follows:

PRESENT MEMBERS TERM EXPIRATION

Ron Green	8-31-02
Leonard Rohrbough	8-31-03
R. Dennis Ford	8-31-04
Jean Champagne	8-31-05
Nixon Adams	8-31-06
A. J. Marciante	8-31-07
John Moore	8-31-08

The remainder of each member's present term as listed above shall constitute that member's first term of office under the provisions of this section. If a vacancy in the Planning Commission occurs otherwise than by expiration of term it shall be filled by appointment by the original appointing authority for the duration of the unexpired term. Vacancies created by completion of a term shall be filled for a new seven (7) year term. Members shall be eligible for appointment to an unlimited number of successive terms. No member of the Planning Commission shall also be an elected official of this state or any political subdivision thereof. In addition, no member of the Planning Commission shall be an employee of the City of Mandeville. All successive appointments to the Planning Commission shall be appointed by a majority vote of the City Council. No person shall be appointed to the Planning Commission until a public hearing before the City Council attended by the nominee has been held. The outgoing member of the Planning Commission shall, absent some disqualifying condition, continue to serve until such time as his successor is appointed.

The City Council may, by a vote of a majority of the City Council members, remove any member of the Planning Commission, after notice and public hearing, for inefficiency, neglect of duty or malfeasance in office.

2.1.3 Election and Term of Chairman

The Mandeville Planning Commission shall elect a chairman from its members and create and fill such other of its offices as it may determine necessary. The term of the chairman shall be one year, with eligibility for reelection. The chairman of the Zoning Board shall also serve as vice-chairman of the Planning Commission and shall, in the absence of the chairman, assume all power, duties and responsibilities of the chairman.

2.1.4 Rules and Records of Proceedings, Meetings and Quorum

The Mandeville Planning Commission shall adopt rules of procedure for the transaction of its business, not in conflict with any other city or state laws, and shall keep a record of its resolutions, motions, transactions, findings and determinations, which record shall be a separate record from the record of the Mandeville Zoning Commission and shall be a public record. The Mandeville Planning Commission shall hold at least one regular meeting in each month and all meetings shall be open to the public. Meetings at other than regularly scheduled times may be announced at a prior meeting and thereby made a part of the record.

The Chairman, may, or upon the request of three (3) members of the Planning Commission, shall, by giving notice to the members of the Planning Commission, call a previously unannounced special meeting of the Planning Commission for a time no earlier than 24 hours after the notice is given. Notice of a previously unannounced meeting shall be posted at City Hall and, to the extent feasible, provided to interested persons at least 24 hours prior to the

meeting. Four (4) members of the Planning Commission constitutes a quorum necessary to hold a meeting. Furthermore, no motion, resolution or other official action shall be passed upon except by a majority vote of the authorized membership of the Planning Commission.

2.1.5 Compensation

The members of the Mandeville Planning Commission shall serve without compensation, and shall hold no other public office, except they shall also serve as members of the Mandeville Zoning Commission and may also serve as members of any duly constituted commission of the region or parish of which the municipality forms a part.

2.1.6 Expenditures

The expenditures of the Mandeville Planning Commission, exclusive of those made from funds received by gift, shall be within the amounts appropriated for that purpose by the City Council. Members of the Planning Commission may, when authorized by the Planning Commission, attend planning and zoning conferences, meetings of planning and zoning institutions and hearings upon pending planning and zoning legislation and the Planning Commission may, by resolution, pay the

reasonable traveling expenses incidental to such attendance out of any funds appropriated by the City Council for that purpose.

2.1.7 Personnel and Duties of Secretary

The City shall provide the Planning Commission with the services of a secretary and a planning consultant. The City Planner shall serve as planning consultant. The Mandeville Planning Commission may request of the City such other employees as it may deem necessary for its work, and with the advance consent of the City Council may also contract with planning experts, engineers, attorneys, architects, landscape architects, arborists, horticulturists and other consultants for such services as they may require within the scope of budgetary authority.

The secretary shall be responsible to the Commission and for the issuance of prior notice of term expirations, for the advertisement and posting, as required by law, of all matters to be heard by the Planning Commission, for the notification of applicants regarding meetings and for all other correspondence and duties as required by these regulations or the adopted rules of procedure of the Commission. The secretary shall keep a true and correct record of all proceedings which shall set forth the reasons for its decisions, the vote of each member participating therein, the absence of any member and any failure of a member to vote at both general and special meetings in a book or books to be kept specially for that purpose and separate from the records of the Zoning Commission which record shall be a public record. Certified copies of the adopted minutes of all such proceedings may be obtained from the

secretary upon request and in accordance with established fees. The secretary shall prepare and submit reports of all actions and recommendations of the Planning Commission to the City Council, case applicants, or other parties as required by these regulations or other state or local laws.

2.1.8 Powers and Duties of the Planning Commission

The Mandeville Planning Commission shall exercise all of the powers and duties conferred by Louisiana R.S. 33:101 through R.S. 33:119, inclusive, and shall exercise all powers and duties which are now or may hereafter be assigned to it by City charter, these Land Use Regulations or any other of the City Council, such duties to include but not limited to the following:

- 1. Adoption of the Comprehensive Land Use Plan of the City Mandeville, including a Major Streets Plan, Future Land Use Plan and other special plans as called for in the Comprehensive Land Use Plan and any subsequent amendments;
- 2. Hear and decide the approval or denial of all subdivision or resubdivision applications in accordance with the Subdivision and Public Improvements Regulations division of these Land Use Regulations and in connection therewith to waive certain provisions of the subdivision regulations as specified therein;
- 3. Hear and make recommendations to the City Council in matters regarding the acceptance of public improvements and/or the posting of bonds for the installation or maintenance of public improvements in accordance with the Subdivision and Public Improvements Regulations division of these Land Use Regulations as specified therein;
- 4. Hear and make recommendations to the City Council regarding the approval or denial or amendment of conceptual development plans submitted in conjunction with Conditional Use Permits and Planned District development s and the conditions associated with such recommendations;
- 5. Hear and make recommendations to the City Council regarding areas proposed for annexation to the City of Mandeville;
- 6. Hear and make recommendations to the City Council regarding the dedication or revocation of public streets, the dedication or abandonment of public servitudes, easements or other public land, or any other actions which affect the Official Base Map of the City; and

7. Serve as the Mandeville Zoning Commission.

2.2 MANDEVILLE ZONING COMMISSION PROVISIONS

2.2.1 Mandeville Zoning Commission Preauthorized

There is hereby preauthorized and continued a municipal zoning commission to be known as the "Mandeville Zoning Commission" also designated "the Zoning Commission", "the Zoning Board" or "the Board" herein. Under the authority of Louisiana R.S. 33:4726 the members of the Mandeville Planning Commission herein reestablished shall serve as the Mandeville Zoning Commission, and when acting as such, shall hold separate meetings with separate minutes and records. The Mandeville Zoning Commission, shall exercise all of the powers and duties conferred by Louisiana R.S. 33:4721 through R.S. 33:4729 inclusive.

2.2.2 Number and Terms of Members, Appointments, Qualifications and Removal

The number and terms of members of the Zoning Commission and the appointments, qualifications and removal of members shall be in accordance with those specified for the Planning Commission in Section 2.1.2.

2.2.3 Election and Term of Chairman

The Zoning Commission shall elect a chairman from its members and create and fill such other of its offices as it may determine necessary. The term of the Chairman shall be one year, with eligibility for reelection. the chairman of the Planning Commission shall also serve as vice-chairman of the Zoning Board and shall, in the absence of the chairman, assume all of the power, duties and responsibilities of the chairman.

2.2.4 Rules and Records of Proceedings, Meetings and Quorum

The Mandeville Zoning Commission shall adopt rules of procedure for the transaction of its business, not in conflict with any other city or state laws, and shall keep a record of its resolutions, motions, transactions, findings and determinations, which record shall be a separate record from the record of the Planning Commission and shall be a public record. The Zoning Commission shall hold at least one regular meeting in each month and all meetings shall be open to the public. Meetings at other than regularly scheduled times may be announced at a prior meeting and thereby made a part of the record.

The Chairman, may, or upon the request of three (3) members of the Commission, shall, by giving notice to the members of the Commission, call a previously unannounced special meeting of the Commission for a time no earlier than 24 hours after the notice is given.

Notice of a previously unannounced meeting shall be posted at City Hall, and, to the extent feasible, provided to interested persons at least 24 hours prior to the meeting. Four (4) members of the Zoning Commission shall constitute a quorum necessary to hold a meeting. Furthermore, no motion, resolution or other official action shall be passed upon except by a majority vote of the authorized membership of the Zoning Commission.

2.2.5 Compensation and Expenditures

The members of the Mandeville Zoning Commission may receive such compensation as may be fixed by the City Council. Expenditures of the Zoning Commission shall be governed by the provisions set forth in Section 2.1.6 of these regulations relative to expenditures by the Mandeville Planning Commission.

2.2.6 Personnel and Duties of Secretary

The City shall provide the Zoning Commission with the services of a secretary and a zoning consultant. The City Planner shall serve as zoning consultant. The Mandeville Zoning Commission may request of the City such other employees as it may deem necessary for its work, and with the advance consent of the City Council may also contract with planning and zoning experts, engineers, attorneys, architects, landscape architects, arborists, horticulturists and other consultants for such services as they may require within the scope of budgetary authority.

The secretary shall be responsible to the Zoning Commission and for the issuance of prior notice of term expirations, the advertisement and posting, as required by law, of all matters to be heard by the Zoning Commission, the notification of applicants regarding meetings and for all other correspondence and duties as required by these regulations or the adopted rules of procedure of the Zoning Commission. The secretary shall keep a true and correct record of all proceedings which shall set forth the reasons for its decisions, the vote of each member participating therein, the absence of any member and any failure of a member to vote at both general and special meetings in a book or books to be kept specially for that purpose and separate from the records of the Planning Commission which record shall be a public record. Certified copies of the adopted minutes of all such proceedings may be obtained from the secretary upon request and in accordance with established fees. The secretary shall prepare and submit reports of all actions and recommendations of the Zoning Commission to the City Council, case applicants, or other parties as required by these regulations or other state or local laws.

2.2.7 Powers and Duties of the Zoning Commission

The Zoning Commission shall exercise all of the powers and duties conferred by Louisiana R.S. 33:4721 through R.S. 33:4729 inclusive and shall exercise all powers and duties which are now

or may hereafter be assigned to it by City Charter, these Land Use Regulations or any other ordinance of the City Council, including but not limited to the following:

- 1. Recommend to the City Council, after public hearing before the Zoning Commission, the boundaries of the various zoning districts as well as the restrictions and regulations to be enforced therein, and any supplements, changes or modifications thereof.
- 2. Make recommendations to the City Council regarding the zoning of parcels of land upon annexation into the City.

2.2.8 Additional Powers and Duties of the Zoning Commission

Under the authority of Louisiana R.S. 33:4727.1 and Ordinance 85-34 of the City of Mandeville in accordance with the general or specific rules provided in these Land Use Regulations and in harmony with the general purpose and intent of these regulations and the goals and policies of the Comprehensive Plan, the Zoning Commission of the City of Mandeville shall continue to exercise the powers, duties and responsibilities which may be exercised by a Board of Adjustments and Appeals under the provisions of Louisiana R.S. 33:4727. The Zoning Commission in its capacity as the Board of Adjustments and Appeals shall be known as the Zoning Board, also designated "the Board" herein.

The members of the Zoning Commission, when serving as the Zoning Board, shall exercise all of the powers and duties conferred by Louisiana R.S. 33:4727 and in accordance may determine and vary the application of these Land Use Regulations in harmony with the general purpose and intent of the regulations and the goals and policies of the Comprehensive Plan and in accordance with general and specific rules provided herein, including but not necessarily limited to the following:

- To hear and decide appeals where it is alleged there is error in any order, requirement, decision, or determination made by an administrative official in the enforcement of any of the provisions of this Land Use Regulations Ordinance including the Flood Damage Prevention Regulations and any supplemental regulations including the codes of the Southern Building Code Congress adopted by reference in these regulations.
- 2. To hear and decide all matters referred to it or upon which it is required to pass under the Land Use Regulations Ordinance including but not necessarily limited to the following:
 - a. applications for zoning permits,

- b. applications for development in Drainage Overlay Districts,
- c. requests of in-lieu contributions of dollars or land dedicated for special purpose in accordance with provisions for in-lieu contributions,
- d. applications for non-conforming development site variances as provided.
- 3. In ruling upon appeals where there are practical difficulties or unusual hardships in the way of carrying out the strict letter of these Land Use Regulations, the Board may vary or modify the application of any of these regulations or provisions of the regulations relating to the use, construction, or alteration of buildings or structures or the use of land so that the spirit of the regulations shall be observed, public safety and welfare secured, and substantial justice done. In hearing and deciding appeals, the Board shall have the power to grant an exception or vary the provisions of these regulations as follows:
 - a. Where the boundary line of a district divides a lot held in single ownership at the time of the passage of this ordinance, to permit extension of the district to include the entire lot.
 - b. Determine the boundaries of the zoning districts accompanying and made a part of this ordinance where the actual street layout on the ground varies from the street layout as shown on the Official Zoning Map by interpreting the provisions of this ordinance in such a way as to carry out the intent and purposes of these Land Use Regulations as they relate to the Official Zoning Map.
 - c. Authorize a variance in the yard requirements, height restrictions, lot areas or density requirements of any district, but only where there are unusual and practical difficulties or unnecessary hardships in carrying out of those provisions due to an irregular shape of the lot, topographical or other physical conditions, providing such variance will not seriously affect any adjoining property or the general welfare.
 - d. Waive or reduce the landscaping, parking and loading requirements in any district whenever the use of a building or land is so extraordinary as to make unnecessary the full provision of parking or loading facilities, or whenever it can be shown that provision of required off-street parking spaces within three hundred (300) feet of the main building is not feasible and would impose an

unreasonable hardship as contrasted with merely granting an advantage or a convenience.

- e. Authorize a variance in sign setback requirements, sign height or sign area limitations, or limitations on the numbers or types of signs allowed on any premise, but only where there are unusual or practical difficulties or unusual hardships in the strict application of the provisions of Article 10 of these regulations due to an irregular shape of the premises involved, or topographical or other physical conditions so great as to warrant a deviation from the signage plan established by Article 10 of this ordinance with the following limitations:
- I. No condition resulting from previous decisions regarding the use or development of the premises involved in the request may be considered as a practical or unusual difficulty or unnecessary hardship; and
- ii. No variance may be allowed which would permit the erection or display of any of the prohibited signs set forth in Article 10, infra, of this ordinance.
- f. Authorize a variance to the Flood Hazard Prevention Regulations in accordance with procedures and guidelines established therein and described in Article 8, Section 8.3, infra.
- 4. In connection with the authority set forth in this section, the Chairman or acting Chairman may administer oaths and compel the attendance of witnesses.
- 2.3 Reserved

2.4 <u>COMMUNITY APPEARANCE COMMISSION PROVISIONS</u>

2.4.1 Community Appearance Commission Preauthorized

There is hereby preauthorized and continued a Community Appearance Commission, also known as the CAC, previously established within the City of Mandeville under the provisions of Ordinance 86-21 of March 12, 1987, which shall continue to exercise the duties and authority granted to it.

2.4.2 Number and Terms of Members, Appointments, Qualifications and Removal

The CAC shall consist of seven (7) members who shall serve staggered terms of seven (7) years. The terms of persons holding membership on the CAC at the time this ordinance is enacted shall not be affected by its enactment and shall serve the remainder of their present terms which shall constitute the initial term of the CAC under the provisions of these regulations. Initial and successive members of the CAC shall be appointed by a majority vote of the total membership of the City Council. If a vacancy in the CAC occurs other than by expiration of term it shall be filled by appointment for the remainder of the unexpired term. Vacancies created by the completion of a term shall be filled for a new seven (7) year term. The outgoing member of the CAC shall, absent some disqualifying condition, continue to serve until such time as his successor is appointed. Members shall be eligible for appointment to an unlimited number of successive terms. The City Council may, at its discretion and by a vote of not less than a majority of its membership and subject to the veto authority of the Mayor, alter the number of members of the CAC or the terms of such members provided that no such alteration shall in any way alter or affect the term or membership of any sitting member of the CAC at the time of such amendment.

All members presently serving and their expiration dates will remain the same at the time this ordinance is enacted. They are as follows:

PRESENT MEMBERS TERM EXPIRATION

Nancy Clark	4/03
Dennis Thomas	4/04
Michael Holland	4/05
Carla Buchholz	4/06
	4/07
Tom Klekamp	4/08
Peg Usner	4/09

Persons qualified to serve on the CAC shall be a resident of and a registered voter in the City of Mandeville in the incorporated area of the municipality. No member of the CAC shall also be an elected official of this state or any political subdivision thereof. In addition, no member of the CAC shall also be an employee of the City of Mandeville or a member of the Planning Commission or the Zoning Commission. The City Council may, by a vote of a majority of its members, remove any member of the CAC, after notice and public hearing, for inefficiency, neglect of duty or malfeasance in office.

2.4.3 Election and Term of Chairman

The CAC shall elect a Chairman and vice-chairman from its members and create and fill such other of its officers as it may determine necessary. The term of the Chairman shall be one year, with eligibility for reelection.

2.4.4 Rules and Records of Proceedings, Meetings and Quorum

The CAC shall adopt rules of procedure not in conflict with any applicable laws of this state or ordinances of this City. However, in no case shall a quorum be authorized which consists of less than a majority of the membership of the CAC. Meetings of the CAC shall be held at the call of its Chairman, and at such other times as the CAC may determine. The chairman or, in his absence, the acting Chairman, may administer oaths and compel the attendance of witnesses. All meetings of the CAC shall be open to the public except for closed or executive sessions convened in accordance with law.

The CAC shall keep minutes of its proceedings and shall keep records of its examinations and other official actions, all of which shall be filed and maintained in the office of the CAC and shall be public records. All testimony, objections thereto, and rulings thereon shall be electronically or stenographically recorded and such recordation shall be further maintained in the offices of the CAC for a period of not less than three (3) years.

2.4.5 Compensation

Members of the CAC may receive a reasonable per diem for attendance at meetings in accordance with standards set by the City Council.

2.4.6 Expenditures

The expenditures of the CAC shall be within the amounts appropriated for that purpose by the City Council.

2.4.7 Appointment of a Secretary

The City shall provide the CAC with a secretary. It shall be the duty of the secretary to keep a true and correct record of all proceedings at both general and special meetings of the CAC in a book or books to be kept specially for that purpose.

2.4.8 Powers and Duties of the CAC

The CAC shall interpret the provisions of Article 10 - Sign Code of these regulations and shall review and approve or disapprove all applications for sign permits except permits for temporary signs as set forth in the provisions of Section 10.5.2.

The CAC shall also serve as the Official Tree Board of the City of Mandeville. The authority and duties of the CAC, when acting as the Tree Board, shall be as follows:

- a. To study the urban forest including problems and opportunities involving the city tree population, determine tree related needs of the community, compose and annually review the Community Tree Plan and seek ways to implement needed work by formulating and adopting an Annual Work Plan and making recommendations to the City Council regarding public funding of the Annual Work Plan.
- b. To advise and assist the officials of the town, as well as citizens and community groups, in the dissemination of news and information regarding the preservation, selection, planting and maintenance of the urban forest.
- c. To provide regular and special meetings at which the subject of the urban forest may be discussed by the citizens of the City.
- d. To make recommendations to the City Council regarding the preservation, planting, maintenance and removal of tree or limbs on public lands in the City and recommend streetscape/landscape designs illustrating the type and kind of trees and other vegetation to be planted on public lands.
- e. The CAC may engage in any other lawful activity in pursuit of the mission of the Tree Board which may benefit the City's urban forest, including such activities as:
 - 1. Apply for Tree City U.S.A. status and Growth Awards from the National Arbor Day Foundation.
 - 2. Conduct seminars and public education programs.
 - 3. Plan and coordinate an annual Arbor Day or Week observance.
 - 4. Develop a community forest preserve.
 - 5. Organize and plan community tree planting and public open space projects.
 - 6. Seek grant money, public funding and private contributions to implement the Annual Work Plan and further the work of the Commission.
 - 7. Formulate Community Tree Plan for the City of Mandeville to be presented to the Mandeville Planning Commission to become a part of the Comprehensive Land Use Plan upon the acceptance and approval of the Planning Commission.

The Community Tree Plan may include, but not be limited to, the mission and goals of the Tree Board, an inventory of tree resources, policies regarding the preservation, planting and maintenance of public trees, the identification of specific sites for public tree preservation, planting and open space projects and the strategy and recommendations for achieving the identified goals. The Community Tree Plan may also include the Master Tree Plan and Tree Management Plan for the City. The Master Tree Plan may include a map of public trees inventoried and tree species proposed to be planted on public streets and lands; plans for identified tree preservation, planting and open space projects; the identification of appropriate tree species for public tree planting projects and standard planting specifications. The Tree Management Plan may include standard tree maintenance specifications and establish policies and recommendations for annual tree maintenance programs, including associated costs for maintenance work.

2.4.9 Appeals of CAC Decisions

Appeals may be taken to the Zoning Commission by any person aggrieved or by any officer, department, commission, board, bureau or any other agency affected by any decision of the CAC in the application or interpretation of the provisions of Article 10 of this Land Use Regulations Ordinance. Such appeal shall be governed by the applicable provisions of Article 10 of this ordinance.

2.5 RESPONSIBILITIES OF CITY DEPARTMENTS AND STAFF

2.5.1 Responsibilities of Department of Planning and Development in Administering CLURO

The provisions of this Comprehensive Land Use Regulations Ordinance (CLURO) shall be administered through the Department of Planning and Development in accordance with the provisions of this ordinance, except for the interpretation and administration of provisions concerning the construction of public improvements which shall be administered by the Department of Public Works with the assistance of the City Engineer as deemed necessary by the Director of Public Works.

2.5.2 Qualifications and Duties of the City Planner

1. <u>Qualifications of the City Planner</u> - The City Planner shall have (a) a college degree in Urban Planning, Landscape Architecture, Architecture or other related field and a minimum of two (2) years supervisory experience in a

- planning department or (b) five (5) years experience in a planning department plus a minimum of two (2) years in a supervisory position.
- 2. <u>Duties of the City Planner</u> The City Planner shall perform or cause to be performed all duties as assigned by the Mayor, and in the discharge of those duties shall:
 - a. Interpret and administer the provisions of Division II of this CLURO,
 - b. Interpret the administrative provisions and administer the Subdivision and Public Improvements Regulations of Division III of this CLURO,
 - c. Coordinate with other departments of the City in the implementation of the Comprehensive Land Use Plan and the administration of the CLURO, the Capital Improvements Program and other related planning programs,
 - d. Review and approve or deny all administrative permits and building permit applications in interpreting and administering the provisions of Division II of this CLURO and in accordance with adopted guidelines,
 - e. Serve as advisor to the City administration, City Council, Planning and Zoning Commission, Zoning Board and Community Appearance Commission in all planning and zoning matters,
 - f. Assist members of the public in planning and zoning matters.

2.5.3 Qualifications and Duties of the Building Inspector

- 1. Qualifications of the Building Inspector The qualifications of the Building Inspector shall be defined by the civil service codes of the City of Mandeville. In addition, the Building Inspector shall be minimally required to have Southern Building Code Congress International, Inc. (SBCCI) Building Inspector, Plan Review and Coastal Construction Inspection certifications. The Building Inspector, or his designee(s), being the party responsible for plumbing, electrical and mechanical inspections, shall be required to have Plumbing, Electrical and Mechanical Inspection certifications by SBCCI, respectively.
- 2. <u>Duties of the Building Inspector</u> The Building Inspector or his duly authorized representative shall perform the following duties:

a. Interpret the provisions of the adopted codes of the Southern Building Code Congress International, Inc. as identified in Article 5 and receive applications required by these regulations, issue permits and furnish the prescribed certificates. He shall examine premises for which permits have been issued and shall make necessary inspections to see that the provisions of law are complied with and that construction is prosecuted safely. He shall assure that all permits issued are in conformance with all adopted codes of the City by enlisting the approval of the City Planner and the Director of Public Works (or by the City Engineer when the services of the City Engineer are determined necessary by the Public Works Director) for all permits involving work falling within the code provisions administered by these individuals.

The Building Inspector shall enforce all laws relating to the construction, alteration, removal, demolition, raising or lowering equipment, use and occupancy, location and maintenance of buildings and structures, including electrical, plumbing, air-conditioning, heat and appurtenances thereto, together with elevators, signs, billboards, marquees and awnings and any and all other separate ordinances wherein he is charged with the enforcement authority and responsibility. He shall, when requested by proper authority or when the interests of the municipality so require, make investigations in connection with matters referred to in these codes and render written reports on same. To enforce compliance with law, to remove illegal or unsafe conditions, to secure the necessary safeguards during construction, or to require adequate exit facilities in buildings and structures, he shall issue notices or orders as may be necessary; and

- b. Administer adopted codes of Southern Building Code Congress International, Inc. and all other pertinent codes in the issuance of permits; and
- c. Enforce provisions of Division II of these regulations, including Building Codes; and
- d. Keep comprehensive records of applications for permits and of permits issued, of certificates issued, of inspections made, of reports rendered and of notices or orders issued. He shall retain on file copies of required plans and all documents relating to building work. All such records shall be open to the public for inspection and copying at the office of the Building Inspector during office hours observed by the municipality but shall not be removed from the office of the Building Inspector; and

e. Make or cause to be made the inspections called for by these regulations. He may accept reports of inspections by inspectors of recognized inspection services provided that, after investigation, he is satisfied as to the qualifications and reliability of such inspectors. No certificate called for by any provision of these regulations shall be based on such reports unless the same are in writing and certified by a responsible officer of such service. Before issuing a permit, the Building Inspector may examine or cause to be examined any building for which an application has been received for a permit to enlarge, alter, repair, move, demolish or change the occupancy thereof. He shall inspect all buildings and structures from time to time during and upon completion of the work for which a permit was issued. He shall make a record of every such examination and inspection and of all violations of these regulations. No permit shall be issued (except for one- and two-family residences, signs and minor alterations) until plans and specifications receive approval of the State Fire Marshal. When deemed necessary by the Building Inspector, he shall make an inspection of material or assemblies at the point of manufacture or fabrication. He shall make a record of every such examination and inspection and of all violations of these regulations.

f. <u>Floodplain Administrator</u> - The Building Inspector shall function as the Floodplain Administrator as specified in the FEMA Flood Damage Protection Regulations of Section 8.3 herein.

2.5.4 Responsibilities of Public Works Director

The Public Works Director shall be responsible for administering those provisions of Division III of this CLURO which address the construction of public improvements, and shall review and approve or disapprove utility and drainage plans in conjunction with building permit applications calling upon the assistance of the City Engineer when determined to be necessary.

ARTICLE 3 - GENERAL DEFINITIONS

3.1 GENERAL PROVISIONS

The following definitions are presented to clarify the meaning of terms as they apply to specific sections of the Land Use Regulations. Unless specifically defined below, words or phrases shall be interpreted to give them the meaning they have in common usage and to give these regulations the most reasonable application. Words in the present tense shall include the future; the singular number shall include the plural and the plural the singular;

the word "structure" shall include, unless the context clearly indicates otherwise, the word "building"; and, the word "shall" is mandatory and not discretionary.

3.2 GENERAL INTENT

Any word, term or phrase defined herein shall reflect the context in which the word, term or phrase is used. All terms not specifically defined shall carry their usual and customary meanings. Terms indigenous to a trade, industry or profession shall be defined when used in such context in accordance with their usual and customary understanding in the trade, industry or profession to which they apply.

3.3 GENERAL DEFINITIONS OF THE LAND USE REGULATIONS

3.3.1 Abandonment

To cease or discontinue a use or activity without intent to resume, but excluding temporary or short term interruptions to a use or activity during periods of remodeling, maintaining, or otherwise improving or rearranging a facility, or during normal periods of vacation or seasonal closure. The date of disconnection of any permanent utilities shall be sufficient to commence the period of abandonment and be interpreted as intent to discontinue a use.

3.3.2 Abutting

To touch along a common border such as "adjoining" lots.

3.3.3 Accessory (Building or Use)

A building, attached to or detached from the principal building, the use of which a) is clearly incidental to and customarily found in connection with a principal building or use; b) is subordinate to and serves a principal building or principal use; c) is subordinate in area, extent, or purpose to the principal building or principal use served; d) contributes to the comfort, convenience, or necessity of occupants, business, or industry in the principal building or principal use served; and, e) is located on the same lot as the principal building or use served.

3.3.4 Addition

Any construction which increases the size of a building or structure in terms of site coverage, height, length, width, or gross floor area.

3.3.5 Agent of Owner

Any person showing written verification that he is acting for, and with the knowledge and consent of a property owner.

3.3.5(A) All Weather Materials

Materials which consist of gravel, limestone, or other loose aggregate.

3.3.6 Alley

Any public way set aside for public travel, which provides a secondary means of access to property abutting thereon.

3.3.7 Alteration, Structural

Any change in the supporting members of a building or structure, such as bearing walls, columns, beams, or girders, or any substantial change to the roof or exterior walls; provided, however, that the application of any exterior siding to an existing building for the purpose of beautifying and modernizing shall not be considered a structural alteration.

3.3.8 Antenna (see also Satellite Dish Antenna)

Any system of wires, poles, rods, reflecting discs, or similar devices used for the transmission or reception of electromagnetic waves external to or attached to the exterior of any building.

3.3.9 Applicant

The record owner(s) of land proposed to be rezoned or subdivided or otherwise acted upon by the City or the owner's authorized representative with confirmed written consent of the owner.

3.3.10 Area of Periodic Inundation

All land areas that are determined to be below the 5' Mean Sea Level (MSL) contour, or lands determined to be jurisdictional wetlands by the U.S. Army Corps of Engineers pursuant to the Clean Water Act.

3.3.11 As-built Drawings

Construction drawings which have been noted or amended to show all changes during the construction process. (Sometimes referred to as record drawings). 3.3.12 Assembly Area

The area of a facility used to assemble members or users of the facility in a group including church sanctuaries, bingo halls, dance or reception halls.

3.3.13 Attached

Having one or more walls common with a principal building, or joined to a principal building by a covered porch, loggia or passageway, the roof of which is a part or extension of a principal building.

3.3.14 Bicycle Path

A right-of-way set aside from vehicular traffic for travel by bicycle.

3.3.15 Block

A tract of land bounded by streets or by a combination of streets and public land, railroad rights-of-way, waterways or any other barrier to the continuity of development.

3.3.16 Block Length

The distance between intersections of through streets, such distance being measured parallel to the longest street bounding the block and from right-of-way to right-of-way line of the two

intersecting streets.

3.3.17 Board

The Zoning Board of the City of Mandeville which consists of the members of the Zoning Commission acting in their capacity as the Board of Adjustments and Appeals under the authority of Louisiana R.S. 33:4727.1 and Ord. 85-34 of the City of Mandeville.

3.3.18 Boat Slip

That area of wet storage bounded by four (4) tie pilings, including one-half the adjacent catwalk or pier.

3.3.19 Bond

Any form of security including a cash deposit, surety bond, letter of credit, collateral or property in an amount and form suitable to the City Council. Any surety bond offered to satisfy any requirement imposed by these regulations shall be issued by a surety company

or bond issuer licensed to do business in this state which has attained a rating of B+ or better in the then latest publication by the A. M. Best Company. The Bond shall also be countersigned by a person who is contracted with the surety company or bond issuer as an agent of that company or issuer and who is licensed as an insurance agent in this state and resides in this state.

3.3.20 Boulevard

A double street or roadway separated by a median or neutral ground.

3.3.21 Buffer zone

A strip of land identified on a site plan and required by this ordinance to protect one type of land use from another land use which is incompatible.

3.3.22 Buildable Area

The area of a lot remaining to be used for the construction of a principal structure bounded by the minimum required front, rear and side yards, landscape, buffer and open space areas, and excluding any servitudes or easements and any areas subject to periodic inundation as defined.

3.3.23 Building

Any structure designed or built or used for the support, enclosure, shelter or protection of persons, animals, chattels, or property of any kind. The definition of the word "building" includes the word "structure", but any use of the term "building" shall not include the term "mobile building" unless specifically provided in the context of the discussion of the word "building."

3.3.24 Building Height

See Height of Structure

3.3.25 Building, Principal

A non-accessory building in which the primary use of the site is conducted. In residential districts a dwelling shall be deemed to be the principal building.

3.3.26 Building Setback Line

That line which is the required minimum distance from the lot line at the street right-of-way line or any other lot line that establishes the buildable area within which the principal structure must be erected or placed.

3.3.27 Capital Improvements Program

A prioritized schedule of capital improvement projects and cost estimates and the anticipated means of financing each project.

3.3.28 Central Sewerage System

A city-wide sewer system, including collection and treatment facilities, with the capability to serve outlying areas.

3.3.28.1 Cellular Transmission & Relay Equipment (see also Satellite Dish Antenna)

Any system of wires, poles, rods, reflecting discs, or similar devices used for the transmission or reception of telecommunication signals external to or attached to the exterior f any building or other structures.

3.3.29 Central Water System

A city-wide distribution system for potable water, including storage and distribution facilities, with the capability to serve outlying areas.

3.3.30 Change of Use

The replacement of an existing use by a new use, or a change in the nature of an existing use to a different specific use classification, as defined by these regulations, but not including a change of ownership, tenancy, name, or management, or change in product or service within the same specific use classification where the previous nature of the use, line of business or other function is substantially unchanged.

3.3.31 City Engineer

A registered professional engineer on staff or a consulting engineering firm under contract to the City.

3.3.32 City Planner

The city staff member appointed to assist the City Planning and Zoning Commission in planning and zoning matters.

3.3.33 Classification of Streets

Ranking the street network by type of service, including local, collector and arterial streets (see specific definitions under each heading).

3.3.34 Clerk of Court

The St. Tammany Parish Clerk of Court.

3.3.35 Cluster Development

A development design technique that concentrates buildings in specific areas on a site to allow the remaining land to be used for recreation, common open space, and preservation of environmentally sensitive areas yet does not exceed the allowable density for the district in which it is located.

3.3.36 Commission

The Planning Commission or Zoning Commission of Mandeville, Louisiana.

3.3.36.1 Communication Tower

A tower, pole, or similar structure which supports a telecommunications antenna operated for commercial purpose above ground in a fixed location, freestanding, guyed, or on a building or other structures.

3.3.37 Comprehensive Land Use Plan

The text, maps, charts and other descriptive material that is part of the comprehensive plan for the city, including the master streets plan and future land use plan and any subsequent supplemental plans adopted by the Planning Commission as part of the Comprehensive Land Use Plan.

3.3.38 Conceptual Plan

A set of plans sufficient to fully illustrate the proposed development concept, drawn to scale and sufficiently dimensioned to define the proposed setbacks and location of proposed and existing structures, sizes of parking and other use areas, sizes of landscaped and open space areas by identifying and describing all proposed uses on the site and providing the proposed ratios of land to site coverage and density, parking for proposed uses, and supported with the necessary details, illustrative elevations of buildings, cross sections and planting concepts.

3.3.39 Condominium

An estate in real property consisting of an undivided interest in common with other purchasers in a portion of a parcel of real property, together with a separate interest in space in a building, such as a residential apartment or retail unit. A condominium may include, in addition, a separate or limited common interest in other portions of such real property.

3.3.40 Construction Plans

A set of plans signed and sealed by a licensed architect or engineer fully documenting all aspects of the proposed construction including but not limited to site plans, foundations plans, pavement and drainage plans, building elevations, utility plans, electrical, mechanical and plumbing plans and accompanied by all details and written specifications needed to fully describe the proposed construction. When accompanying a subdivision plat, such construction plans shall be scaled by a licensed engineer and shall show the specific location and design of all improvements to be installed in the subdivision in accordance with the requirements of the Planning Commission as a condition of the approval of the plat.

3.3.41 Contiguous

See Abutting

3.3.42 Corner Lot

See Lot, Corner 3.3.43 Council

The City Council of the City of Mandeville.

3.3.44 Criteria

Adopted standards to measure the quality of components of land development.

3.3.45 Crosswalk

A designated pedestrian path permitting access across street rights-of-way to or through blocks or squares separated by streets.

3.3.46 Cul-de-sac

See Street, Cul-de-sac

3.3.47 DBH

See Diameter Breast Height

3.3.48 Dead End

A street that does not continue to connect with another perpendicular street at its end or continue into the next subdivision or development.

3.3.49 Detached

Fully separated from any other building, or joined to another building by structural members not constituting an enclosed or covered space.

3.3.50 Developed Lot

Any lot or parcel of land upon which a structure as defined herein has been constructed whether or not such structure is presently habitable or in use; any lot or parcel of land which serves as a yard for such a structure; or, any lot or parcel of land which has been wholly or partially cleared of its naturally growing vegetation or which is the subject of a building or clearing permit issued by the City of Mandeville.

3.3.51 Developer

The legal or beneficial owner or owners of a parcel of land including the holder of an option or contract to purchase or other persons having enforceable proprietary interests in such land with the intent of preparing the land for or achieving the result of the development of said parcel of land.

3.3.52 Development

Any man-made change to improved or unimproved land including but not limited to the construction of buildings or other structures or the mining, dredging, filling, grading, paving, excavation of or performance of drilling operations on the land.

3.3.53 Development Criteria

A set of criteria for site development applied in the review of requests for administrative permits, zoning permits and conditional use applications and in other cases as specified in these regulations or determined by the Zoning Commission or City Council.

3.3.53 Development Site

A contiguous tract or parcel of land, subdivided lot or contiguous lots or parts thereof in the same or multiple ownership intended and suitable for development which is treated as one cohesive development site devoted to a unity of use in a permitting, subdivision or plan review procedure. A site shall not extend across a public street or right-of-way except in the case of a subdivision application in which the development site includes all the land within the boundaries of the proposed subdivision plat and may also include proposed roadways.

3.3.54 Diameter Breast Height

A term used in measuring the size of a tree which refers to the diameter of the tree trunk at four (4) feet from ground level which is approximately breast height, and is abbreviated as "dbh".

3.3.55 Dock

A place for mooring which accommodates a craft lying along side a wharf, pier or bulkhead having the benefit of only single side ties to piling or dock cleats.

3.3.56 Donation Receptacles

A box, building or structure for the collection of food, clothes or other items for the express purpose of distributing these items, or proceeds from the sale of these items, to the poor, needy or disadvantaged.

3.3.57 Drainage Ditch

A man-made drainage facility, preferably with an engineered cross-section, to conduct the flow of storm water.

3.3.58 Drainage Plan

See Storm Drainage Report

3.3.59 Drainage, Subsurface

An underground system of inlets, closed conduits and other structures designed to collect and convey storm water runoff through or to an area.

3.3.60 Drainage System, Storm

A system of inlets, closed conduits, open channels and other structures that are designed to collect and convey storm water runoff from, through or to an area.

3.3.61 Drainageway

Any natural or man-made watercourse, trench or swale or similar depression into which surface water flows.

3.3.62 Driveway

That space specifically designated and reserved on a site for the movement of vehicles from one site to another, or from a site to a public street.

3.3.63 Dwelling Unit

One or more rooms physically arranged so as to create an independent housekeeping establishment for occupancy by one family with sleeping facilities, a separate toilet and a single facility for cooking for the exclusive use of the occupying family.

3.3.64 Easement

See Servitude

3.3.65 Elevation, Building

The side view of a building as depicted in a set of construction plans.

3.3.66 Elevation, Finish Floor

See Finish Floor

3.3.67 Elevation, Grade

The height of a surface relative to a fixed height such as Mean Sea Level (MSL)

3.3.68 Enclosed

A roofed or covered space fully surrounded by walls, including windows, doors, and similar openings or architectural features, or an open space of less than 100 square feet fully surrounded by a building or walls exceeding 8 feet in height.

3.3.69 Engineer

A professional engineer registered in the State of Louisiana, or a professional engineer in the employ of a state or federal agency acting in the area of his registered specialty or area of expertise.

3.3.70 Escrow

A deposit of cash in lieu of an amount due for improvements and still in force on a performance or maintenance bond.

3.3.71 Expressway

A divided arterial highway for through traffic with full or partial control of access and generally with grade separation at major intersections.

3.3.72 Family

One or more persons related by blood or marriage, a group of not more that six (6) persons living together by joint agreement, or a group home for handicapped occupying a premise and living as a single housekeeping unit with a single culinary facility, on a non-profit cost-sharing basis.

3.3.73 Fence

Any artificially constructed barrier of any material or combination of materials erected to enclose or screen areas of land.

3.3.74 Finish Floor Elevation

The elevation, based on Mean Sea Level (MSL), of the surface of the lowest floor of a structure.

3.3.75 Fixed Seat

A seat permanently affixed to a floor or bleachers structure. Two feet of permanently affixed bench shall be considered a fixed seat.

3.3.76 Floodplain

Any area subject to being inundated from any source and particularly any area subject to periodic inundation adjacent to a natural drainageway.

3.3.77 Floodway

The channel of a natural stream or river and portions of the floodplain adjoining the channel, which are reasonably required to carry and discharge flood water or flood flow on any natural stream or river.

3.3.78 Floor Area, Gross

The sum of the total horizontal areas of the several floors of all buildings on a lot, measured from the interior faces of exterior walls. The term gross floor area shall include: basements; elevator shafts; stairwells at each story; floor space used for mechanical equipment with structural head room of six feet, six inches (6'6") or more; penthouses; attic space, whether or not a floor has actually been laid, providing structural headroom of six feet, six inches (6'6") or more; interior balconies; and, mezzanines.

3.3.79 Frontage

That portion of a lot abutting on a street right-of-way measured along the property line of the public right-of-way and the private property.

3.3.80 Future Land Use Plan

A land use plan adopted by the Planning Commission which illustrates the community's goals for future land use and which is intended to serve as the basis for the creation of zoning districts.

3.3.81 Garage, Private

An accessory building for the private use of the owner or occupant of a principal building situated on the same lot as the principal building for the storage of motor vehicles with no facilities for mechanical service or repair of a commercial or public nature.

3.3.82 Grade

The degree of rise or descent of a sloping surface.

3.3.83 Grade, Average

The mean of the highest and lowest point of elevation of the finished surface of the ground, paving, or sidewalk within the area between the building and the property line, or when the property line is more than five (5) feet from the building, between the building and line drawn parallel to the front property line, five (5) feet from the building. The mean of the finished ground level at the center of all walls of a building drawn parallel to the front property line. In case walls are parallel to and within five (5) feet of a sidewalk, the ground level shall be measured at the sidewalk.

3.3.84 Grade, Existing

The natural elevation of land prior to any man-made changes or prior to a proposed change in grade.

3.3.85 Grade, Finished

The final grade of streets, drives, lawns or paved areas, etc. when all site development is complete, designated with contours and spot elevations.

3.3.86 Grade, Proposed

The elevation of the ground surface proposed in conjunction with any development activity on the site.

3.3.87 Gross Leasable Area

The total floor area designed for tenant occupancy and exclusive use, including basements, mezzanines, and upper floors, if any, expressed in square feet and measured from the center line of joint partitions and from outside wall faces.

3.3.88 Gross Site Area

An area defined as the total site area including required yard setbacks, easements, floodplains, waterways, ponds and any other area set aside for preservation.

3.3.89 Guest House

An accessory building containing a lodging unit without kitchen facilities, and used to house occasional visitors or non-paying guests of the occupants of a dwelling unit on the same site.

3.3.90 Hazardous Substances

Any substances or materials that, by reason of their toxic, caustic, corrosive, abrasive, or otherwise injurious properties, may be detrimental or deleterious to the health of any person handling or otherwise coming into contact with such material or substance.

3.3.91 Health Department and Health Officer

The Louisiana Department of Health and Hospitals and its Department Head.

3.3.92 Height of Structure

The vertical distance from "average grade" to the highest point of the coping of a flat roof or to the deck line of a mansard roof or to the mean height between the lowest eave and the highest gable on a pitched, hipped, gable or gambrel roof, or if none of the preceding apply, then to the highest point

of a structure. The height shall be measured from an elevation derived from the average of the highest and lowest grade adjacent to the building as defined herein.

3.3.93 Highway, Limited Access

See Limited Access Highway

3.3.93.1 Height of Tower

The distance from the crown of the nearest street elevation to the top of the structure of any attached cellular transmission or Relay Equipment.

3.3.94 Home Occupation

See Article 6, Definitions, for Home Occupation definition

3.3.95 Homeowners Association

A private, nonprofit corporation, partnership, unincorporated association or other entity comprised of homeowners for the purpose of owning, operating, and maintaining various common properties or organized for the pursuit of common goals.

3.3.96 Illustrative Site Plan

See Conceptual Site Plan

3.3.97 Impervious Cover

Impervious coverage of a site shall include the total horizontal area of all buildings, roofed or covered spaces, paved surface areas, walkways and driveways. Pools of water including but not limited to swimming pools, reflecting ponds and fountains are excluded from this calculation.

3.3.98 Impervious Surfaces

Any material that substantially reduces or prevents the infiltration of storm water into previously undeveloped land. Impervious surfaces shall include graveled driveways and parking areas.

3.3.99 Improvements

See Lot Improvements, Public Improvements or Temporary Improvements

3.3.100 Joint Ownership

Common and undivided ownership of a single item of property by two or more persons. For the purposes of enforcement of these regulations, but specifically not for land use related applications,

any one of the joint owners of property may be treated as the sole owner of the jointly owned property.

3.3.101 Junk

Old, dilapidated, scrap or abandoned metal, paper, building material and equipment, bottles, glass, appliances, furniture, beds and bedding, rags, rubber, motor vehicles, and parts thereof.

Used machinery, scrap, iron, steel, other ferrous and nonferrous metals, tools, implements or portions thereof, plastic, cordage, or other waste that has been abandoned from its original use but may be used again in its present or in a new form.

3.3.102 Landscaped Area

An area which is devoted to and consists of plant material, including but not limited to grass, trees, shrubs, flowers, vines and other ground cover, native plant materials, planters, brick, stone, natural forms, water forms, aggregate and other landscape features, but not including the use of smooth concrete or asphalt; provided, however, that the use of brick, stone,

aggregate or other inorganic materials shall not predominate over the use of organic plant material.

3.3.103 Land Use Plan

A plan showing the existing and proposed location, extent and intensity (density) of development of land to be used in the future for varying types of residential, commercial, industrial, agricultural, recreational and other public and private purposes or combination of purposes.

3.3.104 Lease

A contract by which one party gives to another the enjoyment of a thing for a fixed or determinable term at a fixed or determinable price.

3.3.105 Letter of Credit

An engagement by a bank or other person made at the request of a customer and of a kind that the issuer will honor drafts or other demands for payment upon compliance with the conditions specified in the credit. All letters of credit shall be satisfactory to the City Council and shall be approved by the City Attorney whenever a letter of credit is submitted in accordance with these regulations. Any letter of credit submitted in accordance with these regulations that is issued by a bank shall be issued by a bank doing business in this state that is chartered under the banking laws of the United States of America or the banking laws of the State of Louisiana. A letter of credit submitted in accordance with these regulations must be irrevocable and shall state clearly that it is irrevocable. The engagement may be either an agreement to honor or a statement that the bank or other person is authorized to honor.

3.3.106 Limited Access Highway

A main arterial street providing a traffic way for traffic passing through the area in respect to which owners or occupants of abutting property have no legal right to direct access to or from the same, except at such points and in such a manner as may be determined by the public authority with jurisdiction over the roadway.

3.3.107 Lot

A parcel of land legally platted and recorded or otherwise required to be treated as one development site and having its principal frontage on an officially approved street right-of-way.

3.3.108 Lot Area

The net horizontal area within bounding lot lines, but excluding any portion of a flag (panhandle) lot providing access to a street and excluding any public or private easement or right-of-way providing access to another lot.

3.3.109 Lot, Corner

A lot or parcel of land abutting upon two or more streets at their intersection, or upon two parts of the same street forming an interior angle of less than 135 degrees.

3.3.110 Lot, Coverage

The area of a lot covered by buildings or roofed areas, but excluding incidental projecting eaves, balconies, and similar features and excluding ground level, landscaping, and open recreational facilities.

3.3.111 Lot Depth

The distance measured from the front lot line to the rear lot line. For lots where the front and rear lot lines are not parallel, the lot depth should be measured by drawing lines from the front to rear lot lines, at right angles to the front lot line, every ten (10) feet and averaging the mean length of these lines.

3.3.112 Lot, Double Frontage (or "through")

A lot which fronts on two parallel streets, or which fronts upon two streets that do not intersect along the boundaries of the lot.

3.3.113 Lot, Flag

A lot having access to a street by means of a private driveway, access easement, or parcel of land not meeting the requirements of this Land Use Regulations ordinance for lot width, but having a dimension of at least fifteen (15) feet at its narrowest point.

3.3.114 Lot Improvements

Any building, structure, paving, grading, connection to utilities or other development of the land constituting physical or economic betterment of real property.

3.3.115 Lot, Interior

A lot abutting adjacent lots on at least two sides and other than a corner lot.

3.3.116 Lot Line

A line or series of connected line segments bounding a lot as herein defined.

3.3.117 Lot Line, Front

On an interior lot, the lot line abutting the street. On a corner lot, the shorter lot line abutting a street or the line designated as the front lot line by subdivision or parcel map. On a through lot, the lot line abutting the street providing the primary access to the lot. On a flag (panhandle) lot, the interior lot line designated as a front lot line by a subdivision or parcel map, or the line determined by the Building Inspector to be the front lot line.

3.3.118 Lot Line, Interior

A lot line not abutting a street right-of-way and common between two or more lots.

3.3.119 Lot Line, Rear

The lot line opposite and most distant from the front lot line. In the case of a triangular or irregularly shaped lot, a line ten (10) feet long lying entirely within the lot, parallel to and at a maximum distance from the front lot line.

3.3.120 Lot Line, Side

A lot line intersecting the front lot line and extending therefrom a minimum distance within the lot of 75 feet.

3.3.121 Lot Line. Zero

A common lot line on which a wall of a structure may be constructed.

3.3.122 Lot of Record

A lot which is a part of an approved subdivision, the plat or survey which has been recorded in the office of the parish Clerk of Court; or a parcel of land which became legally established and defined by deed or act of sale on or before May 3, 1964.

3.3.123 Lot, Reverse Frontage

A double frontage or through lot which is not accessible from one of the parallel or non-intersecting streets upon which it fronts.

3.3.124 Lot, Reverse Corner

A corner lot, having a side lot line which is substantially a continuation of the front line of a lot to its rear.

3.3.125 Lot Width

The horizontal distance between the side lines of a lot measured at right angles to the depth along a straight line parallel to the front lot line at the minimum required front setback line.

3.3.126 Manufactured Modular Building and Manufactured Housing

A structure transportable in one or more sections, which is designed for use only with a permanent foundation and which uses standard sheathing, roofing, siding, and electrical, plumbing, and heating systems which comply with the City's adopted building codes.

3.3.127 Master Development Plan

A plan of the proposed phases of a major subdivision.

3.3.128 Mobile Building

A movable or portable building which is constructed on a chassis, and/or which is designed to be towed over Louisiana roads and highways under special permit, designed for year-round occupancy, and designed primarily to be used without a permanent foundation, but which may sit on a permanent foundation, and designed to be connected to utilities. It may consist of one or more sections that can be telescoped when transported and expanded later for additional capacity, or of two or more sections, separately transportable, but designed to be joined together into one integral unit. Building onto or around a mobile home will not change its identification as a mobile home. The following shall not be included in this definition:

- 1. Travel trailers, pickup coaches, motor homes, camping trailers or other recreational vehicles.
- 2. Manufactured modular building as defined.

3.3.129 Mobile Home

A mobile building designed for use as a residential dwelling.

3.3.130 Mobile Home Park

A unified development of 20 or more mobile home spaces for rent or lease which includes common areas and facilities for management, recreation, laundry and utility services, storage, and similar services for the convenience of residents of the mobile home park.

3.3.131 Model Home

A dwelling unit used initially for display purposes which typifies the type of units that will be constructed in a subdivision.

3.3.132 Municipal Utility

A utility service that is provided by the municipality.

3.3.133 Natural Waterbody

An ocean, lake, lagoon, river, stream or bayou shown on the current USGS quadrangle map, or on a recorded survey or map.

3.3.134 Natural Water Course

A river, stream or bayou shown on the current USGS quadrangle map or survey or map recorded with the parish Clerk of Court in which water flows in a definite direction, either continuously or intermittently, having a definite channel and including its floodplains.

3.3.135 Non-Conforming Use

A use of any land, building or structure which does not conform with currently applicable use regulations for the district in which it is located, but which complied with use regulations in effect at the time the use was established; however, uses specified on Planned Development sites approved by ordinance shall not be deemed non-conforming uses when approved under the provisions of these Land Use Regulations or under any regulations previously in effect.

3.3.136 Non-Complying Structure or Site

A building, structure or site area, including off-street parking or loading areas, and landscape areas which does not comply with currently applicable site development regulations for the

district in which it is located, but which complied with applicable regulations at the time of construction; however, Planned Development sites approved by ordinance shall not be deemed non-complying when approved under these Land Use Regulations or under any regulations previously in effect.

3.3.137 Non-Residential Subdivision

A subdivision whose intended and permitted use is other than residential, for example, commercial, industrial or institutional.

3.3.138 Official Base Map

A map legally adopted by the City Council that shows the official name and location of all public streets in the City.

3.3.139 Official Zoning Map

A map legally adopted by the City Council that conclusively shows the location of the official zoning districts in the City.

3.3.140 Off-site

Beyond the boundaries of the site which is the subject of a specific plan of development or subdivision plat.

3.3.141 Off-site Improvements

Improvements other than site-related improvements as defined herein beyond the boundaries of the site which is the subject of a specific development plan. However, in the case of a subdivision application the street and utility improvements installed by the developer in rights-of-way set aside for future dedication to the City as "public improvements" may also be referred to as "off-site" improvements in acknowledgment of their eventual public ownership beyond the boundaries of any of the individual lots being created as future development sites. (See Site-related Improvement)

3.3.142 Off-street Parking Facility

An area on a lot or site or within a building, or both, which is elevated or below grade including one or more parking spaces together with driveways, aisles, turning and maneuvering areas, clearances, and similar features, and meeting the requirements established by these regulations. The term "parking facility" shall also include accessory parking lots, parking garages, and parking structures, elevated parking structures, and

parking structures for accessory parking or parking spaces exceeding the requirements of these regulations.

3.3.143 Open Space

An area that is intended to provide light and air, and is designed for either environmental, scenic, or recreational purposes. Open space may include, but is not limited to, lawns, decorative planting, walkways, active and passive recreation areas, playgrounds, fountains, swimming pools, wooded areas, and water courses. Open space shall not be deemed to include driveways, parking lots, or other surfaces designed or intended for vehicular travel.

3.3.144 Open Space Contribution

Land used for recreation, resources protection, pedestrian or bicycle amenities, and/or buffers. In no event shall any area of a lot constituting the minimum lot area or a required yard setback, nor any part

of an existing or future road or vehicular right-of-way be counted as constituting open space for purposes of open space contribution in conjunction with the subdivision or development of land.

3.3.145 Open Space, Common

Land within or related to a development, not individually owned or dedicated for public use, which is designed and intended for the common use or enjoyment of and is owned in common by the owners of the units of the development in which the common open space is located for outdoor living, recreation, pedestrian access or landscaping but excluding parking, driveways, utility and service areas. Common Open Space may include commonly owned recreation associated structures.

3.3.146 Open Space Lot

Any subdivided parcel of land set aside as a separate lot to be used for open space with pedestrian or bicycle amenities but without any buildings.

3.3.147 Ordinance

A municipally adopted law or regulation.

3.3.148 Outdoor Storage

The keeping, in an unroofed area, of any goods, junk, material, merchandise, or vehicles in the same place for more than 24 hours.

3.3.149 Park

A tract of land designated for and used by the public for active and/or passive recreation.

3.3.150 Parking Area

Any public or private land area not a part of a street right-of-way which is used for temporary parking of automobiles and other vehicles, including driveways and access ways. Also referred to as a vehicular use area.

3.3.151 Parking Lot

An area within a building, or on a lot or site, or both, which is not elevated and which includes one or more parking spaces together with driveways, aisles, turning and maneuvering areas, clearances and similar features and meets the requirements of these regulations.

3.3.152 Parking Space

An impervious surface area accessible to vehicles and of sufficient size to meet the minimum requirements of these regulations. An area on a lot and/or within a building intended for the use of temporary parking of a personal vehicle. This term is used interchangeably with parking stall. Each parking space must have a means of access to a public street. Tandem parking stalls in single-family detached, single-family attached, and townhome residential uses shall be considered to have a means of access to a public street.

3.3.153 Parking, Remote

Off-street parking at another location not on the development site for which its use is intended.

3.3.154 Parkway

A public route intended to be used primarily by passenger vehicles which may have varying width or right-of-way and which right-of-way is or is intended to be developed with a park-like character.

3.3.155 Performance Standards

A list of criteria to establish control of noise, odor, smoke, toxic or noxious matter, vibration, heat, glare or explosive potential generated by or inherent in the use of land or buildings.

3.3.156 Permit

A written authorization to commence an activity allowed by city codes on a form approved by the appropriate city administrator, and signed by the person or persons having the jurisdiction over the approval or denial of the authorized activity.

3.3.157 Personal Use Items

Those items sold to individuals for routine household use.

3.3.158 Pervious Surfaces

Surface materials that allow for the infiltration of surface runoff into the earth and not otherwise classified herein as impervious surfaces.

3.3.159 Plans, Conceptual

See Conceptional Plan

3.3.160 Plans, Construction

See Construction Plan

3.3.161 Plans, Set Of

All plans, elevations, details, and specifications to fully describe construction of a proposed development project.

3.3.162 Plan, Site

A plan view or bird's eye view of site drawn to scale, dimensioned and providing all of the information required to fully describe the proposed construction showing the boundaries of the site and all of the buildings, structures and principal site development features, including parking, access, landscaping and screening, and the use(s) proposed.

3.3.163 Planned Development

Land under unified control to be planned and developed as a whole in a single development operation or a definitely programmed series of development operations or phases. A

Planned Development includes principal and accessory structures and uses substantially related to the character and purposes of the planned development. A Planned Development is built according to general and detailed plans that include not only streets, utilities, lots and building location, but also site plans for all buildings as are intended to be located, constructed, used, and related to each other, and plans for other uses and improvements on the land as related to the buildings. A Planned Development includes a program for the provisions, operations, and maintenance of such areas, facilities, and improvements as will be for common use by some or all of the occupants of the Planned Development District, but which will not be provided, operated, or maintained at general public expense.

3.3.164 Plat

A survey of a tract of land showing the boundaries, dimensions and location of individual lots and streets, survey monuments, topographic data, easements, servitudes, rights-of-way, existing structures, proposed utilities and significant natural features. For purposes of these regulations the term plat is not to be construed as a site plan.

3.3.165 Plat, Conceptual Sketch

A conceptual representation of a proposed subdivision survey preparatory to the preparation of the preliminary plat (or final plat in the case of minor subdivision) sufficient for a tentative subdivision/resubdivision application which may be drawn to scale in architectural freehand style to enable the subdivider to save time and expense in reaching general agreement with the Planning Commission regarding the objectives of these regulations but accurately representing the size and proposed dimensions of lots and the extent of any public facilities that are proposed for dedication and which is accompanied by a site features map and conceptual drainage study. Also referred to herein as a "sketch plat".

3.3.166 Plat, Final

A subdivision or resubdivision survey in substantial conformance with any preceding preliminary plat in accordance with the provisions of these regulations submitted to and approved by the Planning Commission to be signed by the required City officials placed on file with the Clerk of Court of the Parish.

3.3.167 Plat, Preliminary

A subdivision survey plat preparatory to the preparation of a final plat, accompanied by engineering construction plans and specifications for the construction of any and all public and private improvements shown or required to be shown on the preliminary plat.

3.3.168 Portable Building

A transportable accessory building other than a mobile building as defined with or without a permanent foundation.

3.3.169 Practical

Shown in practice.

3.3.170 Practicable

Capable of being put into practice.

3.3.171 Previously Developed Site

See Developed Site

3.3.172 Private Street

A private vehicular accessway not owned and maintained by a public agency which affords the principal means of access to individual occupants or a residential or community development, including Planned Developments, or similar development areas or mobile home spaces or auxiliary buildings in a mobile home park.

3.3.173 Private Utility

Any person, firm or corporation duly authorized and licensed by the appropriate public agencies to furnish electricity, gas, steam, telephone, telegraph or other utility service for a fee or other form of compensation. A utility not maintained or owned by the City.

3.3.174 Professional Land Surveyor

A person qualified to accurately draft surveys and subdivision plats who is properly licensed and registered in the State of Louisiana.

3.3.175 Property Owner

An individual, firm, association, syndicate, partnership, or corporation having sufficient proprietary interest to seek development of land.

3.3.176 Public Improvements

Any capital improvement such as streets, public utilities, drainage ditches or structures, sidewalks, bicycle paths, landscaping or open space whether within publicly owned property or in a dedicated or prescribed servitude or right-of-way on privately owned land, dedicated or intended to be dedicated to the public, which the City currently maintains or will maintain after construction is satisfactorily completed and accepted by the Council.

3.3.177 Public Land Use

A use of the land by a public not-for-profit entity.

3.3.178 Queue Line

An area for temporary parking of motor vehicles in a line while awaiting service or other activity.

3.3.179 Record Drawings

Construction drawings which have been noted or amended to show all changes made during the construction process. (Sometimes referred to as "as-built-drawings")

3.3.180 Recreational Vehicle

A vehicle towed or self-propelled on its own chassis or attached to the chassis of another vehicle and designed or used for temporary dwelling, recreational or sporting purposes. The term recreational vehicle shall include but not be limited to the following: travel trailers, pickup truck campers, camping trailers and self-propelled motor homes, covered trucks and buses, and boats and boat trailers.

3.3.181 Reservation

A legal obligation to keep property free from development for a stated period of time, but not involving any transfer of property rights.

3.3.182 Resubdivision

The reconfiguration of lot lines within a previously platted and recorded subdivision.

3.3.183 Right-of-way

A strip of land occupied or intended to be occupied by a street, crosswalk, railroad, road, electric transmission line, oil or gas pipeline, water main, sanitary or storm sewer main, shade trees, or for another special use. The use of the term "right-of-way" for land-platting

purposes shall mean that every right-of-way hereafter established and shown on a final plat is to be separate and distinct from the lots or parcels adjoining such right-of-way and not included within the dimensions or areas of such lots or parcels. Rights-of-way intended for street crosswalks, watermains, sanitary sewers,

storm drains, shade trees, or any other use involving maintenance by a public agency shall be dedicated to public use by the maker of the plat on which such right-of-way is established.

3.3.184 Runoff

The portion of rainfall, melted snow or irrigation water that flows across the ground surface and eventually returns to streams or waterbodies.

3.3.185 Sale

The non-gratuitous exchange of goods or ownership interest in real property.

3.3.186 Same or Common Ownership

Ownership by the same individual, corporate entity or legally recognized association, ownership by more than one corporate entity in which a principal has an interest.

3.3.187 Satellite Dish Antenna

A device incorporating a reflective surface that is solid, open mesh, or bar configured and is in the shape of a shallow dish, cone horn, or cornucopia. Such device shall be used to transmit and/or receive radio or electromagnetic waves between terrestrially and/or orbitally based uses. This definition is meant to include but not be limited to what are commonly referred to as satellite earth stations, TVROs (television reception only satellite dish antennas), and satellite microwave antennas.

3.3.188 Screening

A method of visually shielding or obscuring a land use by fencing, walls, berms or densely planted vegetation.

3.3.189 Servitude

The right granted by the property owner to use a parcel of land for specified purposes, such as public utilities, drainage and other public purposes, the title of which shall remain with the property owner, subject to the right of use designated in the reservation of the easement.

Easement and servitude are the same and the term utility servitude shall mean a servitude for any utility including stormwater drainage.

3.3.190 Site

See Development Site

3.3.191 Site Plan

See Plan, Site

3.3.192 Site Related Improvements

Capital improvements and right-of-way dedications for direct access improvements to and/or within the development in question. Direct access improvements include but are not limited to the following:

- 1. Access roads leading to the development;
- 2. Driveways and roads within the development;
- 3. Acceleration and deceleration lanes, and right and left turn lanes leading to those roads and driveways;
- 4. Traffic control measures for those roads and driveways; and
- 5. Extensions of utility lines and public drainage facilities to serve the development.

3.3.193 Sketch, Plat

See Plat, Conceptual Sketch

3.3.194 Specifications

A written statement containing a description or enumeration of particulars, as of the terms of a contract or details of construction or land development criteria not shown in architects or engineers drawings but essential to the communication of the architect or engineers intentions for the work.

3.3.195 Standards

The minimum requirements established by these regulations or accepted industry standards as a rule for the measure of quality, particularly as it relates to the application of components of land development.

3.3.196 Storm Drainage Report

A drainage plan and calculations as described in Article 13, Sect. 13.3.3.3.1

3.3.197 Story

That portion of a building, other than a cellar, included between the surface of any floor and the surface of the next floor above it, or if there is no floor above it, the space between the floor and ceiling.

3.3.198 Street, Boulevard

A street that is divided by a median or neutral ground.

3.3.199 Street, Collector

Streets that provide access to abutting property and also serve to connect local streets with major arterial streets.

3.3.200 Streets, Commercial or Industrial

Collector or arterial streets that serve commercial or industrial development.

3.3.201 Streets, Cul-de-sac

A local street with only one outlet and having an appropriate termination based on adopted development standards for the safe reversal of traffic movement.

3.3.202 Street, Local

A street intended to provide access to abutting properties and connecting to collector or major arterial streets.

3.3.203 Street, Major Arterial

Major street in the city's major streets plan that serves traffic moving into, out of and through the city carrying high volumes of traffic.

3.3.204.1 Street, Marginal Special Access

Minor streets which are parallel to or adjacent to arterial streets and highways; which provide access to a butting properties and protection from through traffic. The term is also referred to as a frontage street or road.

3.3.204 Street, Private

See Private Street

3.3.205 Street, Public

A public thoroughfare which affords a primary means of vehicular access to abutting property, and including all land within the right-of-way thereof. The word "street" shall include the words, avenue, road, highway, and thoroughfare, or any other similar terms and include all land within the right-of-way of the street.

3.3.206 Street, Service

An auxiliary street located parallel to a limited access highway for service to abutting properties and adjacent areas to control access and protect adjacent property from the impact of through traffic.

3.3.207 Street, Standard

A street without a median or neutral ground.

3.3.208 Structure

A combination of materials forming a construction which requires a permanent foundation on the ground and includes, among other things, buildings, stadiums, platforms, radio towers, sheds,

storage bins, fences, freestanding signs, air-conditioning compressors, satellite receiving stations and antennas.

3.3.209 Structural Alteration

Any change in the supporting members of a building or structure, such as bearing walls, columns, beams, or girders; or any substantial change to the roof or exterior walls provided, however, that the application of any exterior siding to a previous existing wall shall not constitute a structural alteration.

3.3.210 Subdivider

Any person who has an interest in land and causes it to be divided into a subdivision or who intends to subdivide land in which there is an interest, or who engages an agent to offer a subdivided tract of land for sale or any person in direct or common control of property on which the above action is taken or contemplated.

3.3.211 Subdivision

Any land, vacant or improved, which is divided or proposed to be divided into two (2) or more lots, parcels, sites, units, or plots for the purpose whether immediate or future of offer, sale, lease, or development, either on the installment plan or upon any and all other plans, terms and conditions, including resubdivision. Subdivision includes the division or development of residential and non-residential zoned land, whether by deed, metes and bounds description, devise, intestacy, lease, map, plat or other recorded instrument. For the purpose of these Regulations, subdivision includes the dedication, vacation or reservation of any public or private road, highway, street, alley, servitude or easement through a tract of land regardless of the area involved. The word subdivision includes resubdivisions and, when appropriate to the context, relates to the process of subdividing or to the land or territory subdivided.

3.3.212 Subdivision/Resubdivision, Major

Any subdivision or resubdivision not classified as a minor subdivision/resubdivisions which includes the construction or extension of either new public utilities, roadways or storm drainage improvements or changes in existing public utilities, roadways or storm drainage to serve the lots created by the subdivision/resubdivision.

3.3.213 Subdivision/Resubdivision, Minor

Any subdivision or resubdivision which involves four or fewer lots and will not require either new public utilities, roadways or storm drainage improvements or changes in existing public utilities, roadways or storm drainage, will not adversely affect the remainder of the subdivision or adjoining

property and which is not in conflict with any provision of the Comprehensive Land Use Plan, Official Zoning Map or Comprehensive Land Use Regulations.

3.3.214 Subdivision Agent

A person who represents or acts for a subdivider or developer in selling, leasing or developing or making an offer of same in any site or plat in a subdivision, except an attorney retained as such, whose representation of another person consists solely of rendering legal service.

3.3.215 Subdivision Plat

See definition of Plat

3.3.216 Subdivision Restrictions

See definition of Restrictive Covenants

3.3.217 Substandard Lot

A lot or tract of record by deed or plat that does not comply with minimum area, width, or depth requirements currently applicable to the district in which it is located, but which complied with applicable requirements when it became a lot of record.

3.3.218 Surface, All-weather

Refers to a surface treatment other than concrete or asphalt surfacing such as shells or limestone gravel, that may be used to surface parking lots.

3.3.219 Surface, Pervious

See Pervious Surface

3.3.220 Surveyor

See Professional Surveyor

3.3.221 Temporary Improvement

Improvement built and maintained by a subdivider or developer during construction of a subdivision and prior to release of the performance bond.

3.3.221.1 Telecommunications

As defined in the federal Telecommunications Act of 1996, means the transmission, between or among points specified by the user, of information for the user's choosing, without change in the form or content of the information as sent and received.

3.3.222 Townhouse

A single-family attached dwelling forming part of a series of attached dwellings with property lines and the required fire walls separating each dwelling.

3.3.223 Townhouse Sub-lot

A lot approved by the Planning Commission consisting of land fronting on an improved street or having access to an improved street by way of a commonly owned lot held in common ownership by the owners of the townhouse sub-lots which fronts on an improved street and land containing not less than the minimum required lot area for a single-family townhouse dwelling except when a portion of the required minimum lot area per unit has been set aside for commonly used and commonly owned open space and the overall density of the development site does not exceed the allowable density based on the minimum required lot area per dwelling for the zoning district in which the land is located.

3.3.224 Tract

A parcel of land identified by metes and bounds the boundaries of which are not shown on a recorded subdivision development plat.

3.3.225 "U" Shaped Streets

Streets that form an access from a collector street that project perpendicularly from the local street at two points and have a straight tangent between these two projections.

3.3.226 Utility, Public or Private

Any agency which, under public franchise or ownership, or under certificate of convenience and necessity, provides the public with electricity, gas, heat, steam, communication, rail transportation, water, sewerage collection, stormwater drainage or other similar service.

3.3.227 Variance

A dispensation permitted on individual parcels of property as a method of alleviating unnecessary hardship by allowing a reasonable use of the building, structure, or property, which, because of unusual or unique circumstances, is denied by the terms of the zoning code.

3.3.228 Vehicular Use Area

See definition of Parking Lot

3.3.229 Wetlands

An area that is inundated or saturated by surface water or groundwater at a frequency and duration sufficient to support, and that, under normal circumstances, does support, a prevalence of vegetation typically adapted for life in saturated soil conditions, commonly known as hydrophytic vegetation or an area determined by the U.S. Army Corps of Engineers or any other wetlands jurisdictional agency to be a wetlands.

3.3.230 Yard

A required open space on a lot adjoining a lot line unobstructed by a principle structure from the ground upward, except for accessory buildings, swimming pools, parking and such uses as otherwise provided by these regulations.

3.3.231 Yard, Front

An open space extending across the front of the lot between the side lot lines, and being the required minimum horizontal distance between the street and the nearest part of the principal building, including covered or uncovered porches. On corner lots, the front yard shall be

provided facing the street upon which the lot has its lesser dimensions. In the case of a double

frontage lot or through lot, the setback on the second frontage shall be consistent with the setbacks required on lots fronting on the same street as the second frontage.

3.3.232 Yard, Perimeter

The required open space between the perimeter property lines of a multi-family development site, including common open space and having or not having sublots for units within the development, and the wall of the structure or structures nearest to perimeter property lines.

3.3.233 Yard. Rear

A required open space extending across the rear of a lot between the side lot lines, and being the required minimum horizontal distance between the rear lot line and the nearest part of the principal building, including covered porches or raised decks exceeding three (3) feet in height from grade, but excluding any area located within the street side yard of a corner lot. On both corner lots and interior lots, the rear yard shall in all cases be at the opposite end of the lot from the front yard.

3.3.234 Yard, Side

A required yard extending the depth of a lot from the front yard to the rear yard between the side lot line and the side setback line. In the case of a corner lot, the street side yard shall extend from the front yard to the rear lot line.

3.3.235 Zoning District Map

See definition of Official Zoning Map.

DIVISION II

BUILDING & ZONING REGULATIONS

DIVISION II BUILDING AND ZONING REGULATIONS

ARTICLE 4 - GENERAL AND NON-CONFORMING PROVISIONS AND ADMINISTRATIVE PROCEDURES

4.1 GENERAL BUILDING AND ZONING PROVISIONS

4.1.1 Zoning Affects Every Structure and Use

No structure or land shall hereafter be used and no structure or part thereof shall be erected, reconstructed, converted, moved, or structurally altered unless in conformity with the regulations as set forth in this Ordinance, except in the case of legally non-conforming building sites as provided in this Article.

4.1.2 Reduction in Lot Area

No portion of the required area of a lot shall be used or considered as part of the required area for any other lot. No lot shall be reduced in area, width, or depth to less than the minimum requirements of this Ordinance.

4.1.3 Required Yard Areas

No part of a yard, parking space, or other open space required for any building or use for the purpose of complying with the provisions of this Ordinance shall be included as part of a yard, parking space or open space required for another building under the provisions of this Ordinance.

4.1.4 Change in Zoning District Boundary

Whenever the boundaries of a district shall be changed to transfer an area from one district to another district of a different classification, the following provisions for non-conforming lots, buildings, uses and signs shall also apply to any new non-conforming situations existing after the district modification.

4.2 NON-CONFORMING PROVISIONS

4.2.1 General Non-Conforming Provisions

4.2.1.1 Purpose of Non-Conforming Provisions

The purpose of these provisions are:

- 1. To assure reasonable opportunity for use of legally created lots which do not meet current minimum requirements for the district in which they are located, subject to the provisions for contiguous substandard lots in single ownership.
- 2. To assure reasonable opportunity for use, maintenance and improvement of legally constructed buildings, structures and site development features which do not comply with current minimum requirements for the district in which they are located.
- 3. To assure reasonable opportunity for continuation of legally established uses which do not conform to current use regulations for the district in which they are located.
- 4. To limit continuation and expansion and encourage eventual replacement of non-conforming uses having potentially undesirable impacts on surrounding conforming uses.

4.2.1.2 Restrictions Additive

The regulations applicable to a non-conforming use are in addition to regulations applicable to a non-complying structure and in the event of any conflict, the most restrictive provision shall apply.

4.2.1.3 Construction Establishes Use

To avoid undue hardship, nothing in this Ordinance shall be deemed to require a change in the plans, construction or designated use of any building on which actual construction was lawfully begun prior to the effective date of adoption or amendment to this Ordinance and upon which actual building construction has been diligently carried on. Actual construction is hereby defined to include the placing of construction materials in permanent position and fastened in a permanent manner, and demolition, elimination, and removal of an existing structure in connection with such construction; provided, that actual construction shall be diligently carried on until such building is completed.

4.2.1.4 Types of Non-Conforming Situations

For purposes of these regulations non-conforming situations shall be grouped into four categories:

- 1. Legally Non-Conforming Uses
- 2. Legally Non-Conforming Sites
- 3. Legally Non-Conforming Structures
- 4. Legally Non-Conforming Lots

4.2.2 Provisions for Legally Non-Conforming Uses

4.2.2.1 Purpose and Intent

- 1. It is the intent of this Ordinance to permit legally non-conforming uses to continue, but not to encourage their survival. Such uses are declared by this Ordinance to be incompatible with permitted uses in the districts involved.
- 2. The purpose of this Section 4.2, et seq. is to establish regulations and limitations on the continued existence of uses established prior to the effective date of this Section which do not conform to the provisions of Article 7 Table of Permitted Uses. Many such non-conformities may continue, but the provisions of this Section are designated to curtail substantial investment in such non-conformities and to bring about their eventual elimination, where appropriate, in order to preserve the integrity of the zoning districts and the regulations established by this Ordinance.

4.2.2.2 Continuation and Termination of Legally Non-Conforming Uses

- 1. <u>Continuation</u> An existing non-conforming use may be continued, and structures associated with the use may be maintained, provided no non-conforming use shall be enlarged or expanded in terms of floor space utilized or site area occupied nor may any legally non-conforming use be changed to another non-conforming use of a different specific use classification as defined under the provisions of Article 6. Enlargement or expansion shall include:
 - a. extension of such use to any structure or land area other than that occupied by such non-conforming use on the effective date of this Ordinance, or any amendment hereto which causes such use to become non-conforming; or

b. extension of such use within a building or other structure to any portion of the floor area that was not occupied by such non-conforming use on the effective date of this Ordinance, or any amendment hereto which causes such use to become non-conforming.

- c. attachments of signs to the building, placement of signs, or display material or goods or equipment on land outside of the building or the attachment of racks, balconies, or other projections from the building.
- 2. Reestablishment - An existing non-conforming use may not be reestablished after the non-conforming use of the building or land has ceased for a continuous period of six (6) calendar months. If the lessee of any building or place used or occupied for non-conforming purposes under a bona fide lease shall at any time before the expiration of said lease cease to occupy or use said building or land, it shall not be considered vacant until the owner of said building or place shall again obtain legal control of its occupancy and use. This extension of the vacancy provision shall not apply if the lessor for any reason is legally entitled to regain possession and does not by legal or other effective means take prompt action to do so. Once changed to a conforming use, no building or use shall be permitted to revert to a non-conforming use. If a nonconforming use is changed to a conforming use, the non-conforming use shall not be resumed. However, the Zoning Board may grant a one-time six (6) month extension to this period provided that the extension application is filed not later than six (6) weeks prior to the end of the initial six month period. Extension applications shall constitute a variance and be filed in accordance with the procedures for filing a variance as included herein.
- 3. Damage or Destruction - Except as otherwise provided under the Provision of the Flood Damage Prevention Regulations of this Code, in the event that any structure that is devoted in whole or in part to a non-conforming use is damaged or destroyed, by any means, to the extent of more than seventy-five (75%) percent of the fair market value of such structure then, except in otherwise provided herein, that structure shall not be restored unless such structure and the use thereof shall thereafter conform to all regulations of the zoning district in which such structure and use are located. When such damage or destruction is seventy-five (75%) percent or less of the fair market value of the structure as it existed immediately prior to such damage, such structure may be repaired and reconstructed and used for the same purposes as it was before the damage or destruction, provided that such repair or reconstruction is commenced and completed within twelve (12) months of the date of such damage or destruction. For purposes of these regulations the Building Inspector will accept the assessed value of the improvements as the fair market value or the value as determined by the Zoning Board on appeal of the Building Inspector's determination. In the event that the damaged

structure which is devoted to a non-conforming use is more than fifty (50) years old, that structure can be restored, at the owner's option, regardless of the extent of the damage sustained, without the loss of the non-conforming use status. Such restoration must reasonably resemble the original structure and must be commenced within twelve (12) months of the date of such damage or destruction.

- 4. Relocation No structure that is devoted in whole or in part to a non-conforming use shall be relocated in whole or in part to any other location on the same or any other lot, unless the entire structure and the use thereof shall hereafter conform to all the regulations of the zoning district in which such structure and use are located after being so relocated. No non-conforming use of land shall be relocated in whole or in part to any other location on the same or any other lot, unless such use shall thereafter conform to all the regulations of the zoning district in which such use of land is located after being so relocated.
- 5. <u>Change in Use</u> A non-conforming use of land or of a structure shall not be changed to any use other than a use permitted in the zoning district in which such land or structure is located. When such non-conforming use has been changed to a permitted use, it shall only be used thereafter for a use permitted in the zoning district in which it is located. For purposes of this Section, a use shall be deemed to have been so changed when an existing non-conforming use shall have been terminated and the permitted use shall have commenced and continued for a period of seven (7) days.
- 6. <u>Non-Conforming Signs</u> Provisions regarding the continuation and termination of legally non-conforming signs are included under Article 10 of these regulations.

4.2.2.3 Criteria for the Repair of Legally Non-Conforming Uses

- 1. <u>Ordinary Repair Maintenance</u> Normal maintenance and incidental repair or replacement, and installation or relocation of non-bearing walls, non-bearing partitions, fixtures, wiring or plumbing, may be performed on any structure that is devoted in whole or in part to a non-conforming use; provided, that this Section shall not be deemed to authorize any violation of this Section 4.2, et seq.
- 2. <u>Exception for Repairs by Public Order</u> Nothing in this Ordinance shall be deemed to prevent the strengthening or restoring to a safe condition of any building or part thereof declared to be unsafe by any city official charged with

protecting the public safety, upon order of such official. Repairs and alterations that restore a building to the same condition that existed prior to damage or deterioration, altering the building only in conformity with the provisions of this Ordinance in such a manner that does not extend or increase an existing nonconformity may be made with the same kind of materials as those of which the building is constructed.

4.2.3 Provisions for Legally Non-Conforming Development Sites

- 1. <u>Background</u> This Code has modified previous zoning requirements and established specific site development standards. Consequently, many development sites do not meet current requirements for such items as parking lot standards, landscaping, and other open space specifications.
- 2. <u>Purpose</u> The intent of this Section is to insure that such non-conforming development sites are brought into conformance with the site development standards prescribed by this Ordinance.
- 3. <u>Authority to Continue</u> Any lawfully existing non-conforming development site may be continued so long as it remains otherwise lawful subject to Section 4.2.3 of this Code.
- 4. <u>Non-Conforming Shopping Centers</u> Non-conforming shopping centers shall have nine (9) years from the effective date of this Ordinance either to bring the site into conformance with the provisions of the Ordinance or have a non-conforming development site variance approved for the site. All owners of record of commercially zoned properties shall be notified by the Building Inspector by first class mail of this provision prior to the end of the nine (9) year period.
- 5. <u>Annexation of Non-Conforming Development Sites</u> As a condition of annexation approval, the City Council shall required the owner of the subject property to provide a plan for bringing the annexed development site into conformance with the provisions of these regulations. The plan shall provide a schedule which outlines a timetable for bringing the non-conforming site into maximum conformance to the provisions of this Ordinance within five (5) years of the date of annexation approval.
- 6. <u>Extension</u> A conforming use located on a non-conforming development site shall not be expanded until the site is brought into conformance with the provisions of this Ordinance. However, single-family residential structures which are located on a legally non-conforming site with respect to required yard areas or height may be structurally altered or enlarged providing that

- portion of the building which is altered or enlarged conforms with the provisions of this Ordinance.
- 7. <u>Relocations</u> No structure shall be relocated to a non-conforming development site until the site is brought into conformance with the provisions of this Ordinance.
- 8. <u>Change in Use</u> No existing structure located on a non-conforming development site shall be changed from one use classification to another use classification as listed in Article 6, until the site is brought into conformance with the provisions of this Ordinance or a non-conforming development site variance has been approved by the Zoning Board.
- 9. <u>Abandonment or Discontinuance</u> When the use of a non-conforming development site has been abandoned for a period of six (6) months, regardless of any reservation of intent not to abandon use of the site, such site shall not thereafter be used, developed or improved until it is brought into conformance with the provisions of this Ordinance. However, the Zoning Board may grant a one-time six (6) month extension period for the purpose of bringing the site into conformance with the provisions of these regulations. Extension applications shall be filed in accordance with the procedures for site plan review for zoning permits included herein. For purposes of this Ordinance, rental payments or lease payments and taxes shall not be considered as a continued use and the disconnection of utilities shall constitute a means of establishing the commencement of the abandonment of the use of the development site.
- 10. <u>Inability to Meet Current Site Development Standards</u> Practical difficulties may exist which prevent the upgrading of certain non-conforming development sites to the standards imposed by this Ordinance. Consequently, a variance procedure has been established in this Article 4 both to allow a viable economic use of an existing structure and insure that the non-conforming development sites are brought into compliance with the requirements of these regulations to the maximum extent deemed feasible by the Zoning Board.
- 11. Exception for Repairs Pursuant to Public Order Nothing in this Section shall be deemed to prevent the strengthening or restoration to a safe condition of a building or structure in accordance with an order of a public official who is charged with protecting the public safety and who declares such structure to be unsafe and orders it to restoration to a safe condition, provided such restoration is not otherwise in violation of the various provisions of this

Section prohibiting the repair or restoration of partially damaged or destroyed buildings or structures.

4.2.4 Provisions for Non-Conforming Structures/Buildings

4.2.4.1 Continuation of Use of Non-Conforming Structures

The use of a non-conforming structure may be continued and the structure may be enlarged, maintained, repaired or altered in accordance with the requirements of this Ordinance. Except as provided herein, no such enlargement, maintenance, repair or alteration shall either create an additional non-compliance or increase the degree of existing non-compliance of all or part of such structure. No building or structure shall be deemed to be non-conforming solely as a result of a change in the use, zoning or development of adjacent property.

4.2.4.2 Repair of Non-Conforming Structures

1. Percent of Repair Allowable under Normal Conditions - If, within any period of twelve (12) months, alterations or repairs are proposed to be made to a non-conforming building, and the aggregate cost of such alterations or repairs is in excess of fifty (50) percent of the assessed value of the building at the time the alteration or repair is proposed, the building shall be made to conform to the requirements of this code for new buildings in the district in which it is located. Buildings that are located in fire districts shall conform to the provisions of Chapter III of the Standard Building Code of the SBCCI.

2. Percent of Repair Allowable under Emergency Conditions

- a. If an existing non-conforming building is damaged by fire or other Act of God and the building is thereafter proposed to be restored, altered or remodeled at cost in excess of fifty (50) percent of the assessed value of the building before the damage was incurred, the building shall conform to the requirements for new buildings in the district in which it is located.
- b. If the cost of such alterations or repairs, or the amount of such damage is more than twenty-five (25) but less than fifty (50) percent of the assessed value of the building prior to damage of the building, nothing herein shall

prevent the restoration of the building within a six (6) month period from the date the damage was sustained, and the portions of the building to be altered or repaired shall be made to conform to the requirements of these regulations for the district in which the building is located to the greatest extent possible.

c. The preceding provisions of this section notwithstanding, in the event an existing non-conforming structure fifty (50) years old or older is damaged by fire or other Act of God, it can be restored regardless of the extent of the damage sustained, at the owner's option, if it is restored to reasonably resemble the original structure. Restoration must be commenced within twelve (12) months of the date of such damage or destruction and so long as such restoration does not conflict with the Flood Damage Prevention Regulations of this code.

4.2.4.3 Loss of Non-Conforming Status with Change in Use

If the use of an existing non-conforming building is partially or entirely changed to a use of a different classification as established in Article 6, the building shall be made to conform to the requirements of the district in which it is located. Any change in use that requires an increase in parking will terminate the legally non-conforming status of the property and require conformity with the requirements of the district in which it is located.

4.2.4.4 Criteria for the Repair of Non-Complying Structures

The Building Inspector may order an unsafe, non-conforming structure to be restored to a safe condition. Any such order is subject to the requirements of the preceding provisions regarding the repair or restoration of partially damaged or destroyed non-conforming structures.

4.2.5 Provisions for Legally Non-Conforming Lots-of-Record

- 1. Where a lot has less area than the minimum required for the district within which the lot is located, was a lot-of-record in separate ownership from adjacent property at the time of passage of this Ordinance and is currently a lot-of-record under separate ownership, such lot may be used as a building site for a use permitted in the district within which the lot is located; provided, however, that the proposed development of the building site conforms with the required yard areas and other requirements of this Ordinance for the district in which the site is located.
- 2. When a substandard lot is used together with one or more contiguous lots for a single use or unified development, all of the lots so used, including any lots

- used for off-street parking, shall be considered a single lot for the purposes of these land use regulations.
- 3. If two (2) or more lots-of-record or parts thereof or combination of lots and portions of contiguous lots are in single ownership and if all or part of the lots do not meet the requirements for lot width, area, or buildable area lying outside of areas of periodic inundation (defined in Article 3) and as established by this Ordinance the lands involved shall be considered to be an undivided parcel for the purposes of this Ordinance, and no portion of said parcel shall be used or sold which does not meet the minimum lot width, depth and area requirements established by this Ordinance, nor shall any division of the parcel be made which leaves remaining any lot with width, depth or area below the requirements stated in this Ordinance.

4.3 ADMINISTRATIVE PROCEDURES

4.3.1 Procedure and Fees for Zoning Amendments and Amendments to the Land Use Regulations

4.3.1.1 Methods of Initiation of Amendment

The City Council may, from time to time, amend, supplement, or change the regulations and restrictions of these regulations or district boundaries of the Official Zoning Map as adopted herein or subsequently established. Such amendments may be initiated according to the then current rules of procedure of the Planning and Zoning Commission and in the following manner:

- 1. By action of the City Council by introduction of an ordinance.
- 2. On petition by property owners, by filing with the secretary of the Planning and Zoning Commission a petition in writing which conforms with the standards and requirements of the Planning and Zoning Commission, provided that such petition is duly signed and acknowledged by the owners or authorized agents of not less than fifty (50) per cent of the area of land in

- which a change of classification is requested or within a radius of five hundred (500) feet of such area of land.
- 3. By initiative, subject to the procedures and limitations of the City Charter, and further provided that amendments initiated through the initiative shall not involve or seek the re-classification of property to a different zoning district or seek to alter or re-locate the boundaries of an established zoning district.
- 4. Upon resolution of the Zoning Commission duly adopted by the affirmative vote of not less than two-thirds (2/3) of its members at a duly called and convened meeting of its membership.
- 5. Upon resolution of the Planning Commission, duly adopted by the affirmative vote of not less than two-thirds (2/3) of its members at a duly called and convened meeting of its membership, provided that amendments initiated by the Planning Commission be limited to matters involving matters administered by the Planning Commission in these regulations such as subdivisions and Planned District zoning.
- 6. Upon resolution of the Community Appearance Commission duly adopted by the affirmative vote of not less than two-thirds (2/3) of its members at a duly called and convened meeting of its membership, provided that amendments initiated by the Community Appearance Commission be limited to matters involving signage regulation.

4.3.1.2 Procedure

No amendment shall become effective until:

1. There shall have been held a public hearing in relation thereto before the City Planning or Zoning Commission, as applicable, at which time interested citizens and parties shall have had an opportunity to be fully heard. In addition, in the event that the proposed amendment involves any changes, additions or deletions to or from the provisions of Article 10 of these regulations or otherwise involves or affects the regulation or control of signage, there shall also be held a similar public hearing before the Community Appearance Commission. The public hearing provided for herein may, with the consent of both the Chairman of the Zoning Board and the Chairman of the

Community Appearance Commission be held conjointly and coincidentally at a joint meeting of the Zoning Board and the Community Appearance Commission.

2. Notice of the proposed change and of the time and place of the public hearing or hearings thereon shall have been published once a week for three (3) weeks in the City's official journal. At least fifteen (15) days shall elapse between the first publication and the date of the hearing or hearings to which said publication relates.

In addition to notice by publication, and at least ten (10) days prior to the hearing, a good faith attempt to notify the owner(s) of record of the properties to be zoned or rezoned shall be made by the sending of an official notice by regular mail of the time and place of the hearing and subject matter of the regulations and restrictions. When more than 10 parcels are to be zoned or rezoned by enactment of a zoning ordinance, the advertisement in the official journal shall be adequate notice to the property owners.

In addition, a printed notice in bold type shall be posted for not less than ten (10) consecutive days prior to the public hearing on signs prepared, furnished, and placed by the Planning Department staff upon the principal and accessible rights-of-way adjoining the area proposed for rezoning. However, for comprehensive rezoning proposals initiated by the City, signs need only be placed in the general geographic area(s) affected by the proposed change and need not list all specific rezoning proposals. For all text changes amending or supplementing the regulations or restrictions of this Ordinance itself, no signs need be posted.

3. After the public hearing or hearings provided above, the Planning or Zoning Commission and, when applicable, the Community Appearance Commission, shall have submitted its report and recommendation on the proposed amendment to the Clerk of the City Council. The Zoning Commission and Community Appearance Commission shall in all instances in which a public hearing is conducted by the Community Appearance Commission make separate and independent reports and recommendations on the proposed amendment to the City Council. In the event that the report of the Planning or Zoning Commission or Community Appearance Commission is not filed with the Clerk of the City Council within thirty (30) days after the date of the public hearing conducted by said Commission the recommendation of the Commission regarding the proposed change shall be deemed favorable.

- 4. A final yea or nay vote on the proposed amendment shall have been taken by the City Council within one hundred twenty (120) days, dated from the introduction of an ordinance in correct form by the City Council, the submittal to the City Council of the recommendation of the Planning Commission, Zoning Commission or the Community Appearance Commission, or from the final filing of the petition of the property owner or owners in correct form, whichever event is first to occur.
- 5. The provisions of this Section shall not apply in cases where there is a proposal to enact an entire land use ordinance, to change the text of the land use regulations ordinance as a whole, or to change all of the official zoning maps, or both, in which event the procedures set forth in Act 240 of 1926, as may be amended from time to time (Louisiana Revised Statutes of 1950 Title 33: Sections 4721, et seq.) shall be followed.

4.3.1.3 One Year Limitation

Whenever a petition is filed requesting or proposing a change in or amendment to these regulations or Official Zoning Map and said petition has been finally acted on by the Council in accordance with the above outlined procedure, then the Council shall not consider any further petition requesting or proposing the same change or amendment for the same property within a period of one (1) calendar year from the date of the Council's final action on said petition.

4.3.1.4 Fees for Requests to Amend the Official Zoning Map

Before any action shall be taken as provided in this Article, the party or parties (other than the City Council, CAC or Zoning Commission) proposing or recommending a change in the official zoning map shall deposit with the City of Mandeville the following fees:

Request	Amount
R-1, R-1X or R-2 Zoning	\$75.00 per acre or fraction thereof, up to a maximum of \$1,500.00
R-3, MH, O/R or B-3 Zoning	\$150.00 per acre or fraction thereof, up to a total maximum of \$2,100.00

I, O, B-1, B-2, B-4, PM-1, PM-2, Planned District, M-1, M-2, or Industrial Zoning

\$250.00 initial processing fee, plus \$150.00 per acre up to ten (10) acres. For each acre or fraction thereof over 10 acres, \$10.00 per acre shall be charged.

In addition to the fees, the party or parties must reimburse the City for the cost of the legal advertisements necessary for legal notice of the request. Under no condition shall the fee or advertising cost reimbursement be refunded for the failure of such request to be granted or for the withdrawal of the request.

4.3.2 Procedures and Fees for Site Plan Review for Administrative and Zoning Permits

4.3.2.1 Title and Purpose

The procedures set forth herein for approval of Administrative and Zoning Permits shall be known as the Site Plan Review Procedure. The purpose of this procedure is to provide for review and evaluation of site development and design features of selected uses, and to afford a procedure for mitigation of potentially unfavorable effects on adjacent land uses.

4.3.2.2 Jurisdiction

The City Planner shall be responsible for the administration of the Administrative Permits and Zoning Permits procedure. Except as may otherwise be provided by this Section, the City Planner shall review, evaluate and act on all site plans submitted pursuant to the Administrative Permit procedure. The Zoning Board shall be responsible for review, evaluation and action on all site plans (1) submitted as an appeal of an Administrative Permit denial pursuant to this procedure, and (2) submitted as required for Zoning Permits. Provided, however, that site plans required to be reviewed in conjunction with conditional use approvals and Planned District Zoning approvals shall be reviewed by the Planning Commission and shall require City Council approval in accordance with the provisions of Section 4.3.3, et seq.

4.3.2.3 Uses Requiring Site Plan Review

All uses as noted in the Table of Permitted Uses by Zoning Districts requiring Administrative Permits or Zoning Permits shall follow the Site Plan Review procedure in this Section. Uses requiring Zoning Permits shall automatically be forwarded by the City Planner to the Zoning Board for review at the first meeting following the required public notice as specified herein.

4.3.2.4 Application and Fee

Applications for Site Plan Review shall be filed with the City Planner. The application shall include the following unless material is determined to be unnecessary by the City Planner.

- 1. Completed application form provided to applicant by Department of Planning & Development.
- 2. Name, signature and address of the owner and applicant, if agent of owner, on the application clearly stating the requested action.
- 3. Address and legal description or boundary survey of proposed development site with bearing and distances of the property.
- 4. If the applicant is not the legal owner of the property, a sworn statement of the owner that the applicant is the authorized agent of the owner.
- 5. The municipal address or lot, square and subdivision, and the name and mailing address of the owner of each lot abutting or opposite the subject property.
- 6. A brief description of the proposed use, including information pertinent to the review criteria and findings provisions of this Section.
- 7. A site plan and the number of copies required by the Zoning Board's Rules of Procedure a minimum of 8.5" x 11" inches and a maximum of 24" x 36" inches, drawn to scale and sufficiently dimensioned as required to show the following:
 - a. The date, scale, north point, title, name of owner, and name of person preparing the site plan.
 - b. The location and dimensions of boundary lines, easements, and required yards and setbacks of all existing and proposed buildings and land development improvements.

- c. The location, height, and intended use of existing and proposed buildings on the site, and the approximate location of existing buildings on abutting sites within fifty (50) feet of the proposed development site.
- d. The location and dimensions of existing and proposed site improvements including parking and loading areas, pedestrian and vehicular access, utility or service areas, fencing and screening, and lighting.
- e. The center line of existing water course, drainage features and location and size of existing and proposed streets and alleys, the 100-year floodplain, and any areas of periodic inundation.
- f. The number of existing and proposed off-street parking and loading spaces, and a calculation of applicable minimum requirements.
- g. A conceptual drainage plan showing existing and proposed topography and grading and proposed subsurface drainage structures and retention and water quality enhancement facilities.
- h. The approximate location and size of proposed signs, if known; subject to Article 10 Sign Code.
- i. A conceptual landscape plan showing the location and size of the existing and proposed landscaped areas and the number and location of Class A & B trees proposed or required to be preserved.
- j. Application fee of fifty (\$50) dollars per acre site or fraction thereof shall be submitted in conjunction with an application for a Zoning Permit.

4.3.2.5 Public Notice

- 1. <u>Administrative and Zoning Permits</u> Not more than 10 days after the date of filing of an application for an Administrative Permit or a Zoning Permit the City Planner shall post the site as required for zoning amendments including the following information:
 - a. A brief description of the nature of the application.
 - b. A statement of how and where information regarding the application may be obtained.
- 2. <u>Zoning Permits</u> In addition to the notice requirements of Item 1 above, one notice regarding the application at which action on the application is to be

taken shall be published in the official journal of the City at least seven (7) days prior to the meeting of the Zoning Board.

4.3.2.6 Administrative Permit Approval and Appeal Procedure

Not less than ten (10) days after posting nor more than twenty (20) days after official acceptance of a complete application by City Planner, the City Planner shall prepare a report to the Building Inspector of those administrative permit site plans approved, approved subject to conformance or disapproved. Copies of this report shall be forwarded to applicant and Building Inspector, with the signature of approval of the City Planner or marked disapproved. A decision of the City Planner may be appealed to the Zoning Board within 10 days after the date on which the decision was forwarded to the applicant. Notice of appeal shall be filed in writing with the City Planner stating the specific reasons for and basis of the appeal. If the tenth day falls on a weekend or holiday the appellant shall have until the next business day to file.

4.3.2.7 Appeal of Administrative Permit to Zoning Board

- 1. Action by the Zoning Board On the first available meeting date after the required advertising period for appeal as established in the current adopted Rules of Procedure of the Zoning Board, the Zoning Board shall consider disapproved site plans, and site plans appealed by an applicant or other interested party aggrieved by a decision of the City Planner made pursuant to this Site Plan Review Procedure. The appeal will identify in writing the specific reasons for and basis of the appeal.
- 2. Any person or persons, or any officer, department, board, bureau or any other agency of the community jointly or severally aggrieved by any decision of the Zoning Board may present to the Civil District Court of the parish, within thirty (30) days after filing of the decision in the office of the Board, a writ of certiorari asking for such relief and under such rules and regulations as are provided for such matters in appropriate legislation of the state.

4.3.2.8 Zoning Permit Approval and Appeal

1. Not more than thirty (30) days after official acceptance of a complete application by the City Planner, the Zoning Board shall consider the application for a Zoning Permit at a regularly scheduled meeting. Within ten (10) days of the decision of the Zoning Board, the City Planner shall prepare a report to the Building Inspector and the applicant regarding the approval,

- approval with modifications, or disapproval of the Zoning Permit and site plans by the Zoning Board.
- 2. Any person or persons, or any officer, department, board, bureau or any other agency of the community jointly or severally aggrieved by any decision of the Zoning Board may present to the Civil District Court of the parish, within thirty (30) days after filing of the decision in the office of the Board, a writ of certiorari asking for such relief and under such rules and regulations as are provided for such matters in appropriate legislation of the State.

4.3.2.9 Review and Evaluation Criteria

- 1. Site plans for uses subject to the Special Use Criteria as provided in Article 8 shall be reviewed and evaluated for consistency with such standards.
- 2. Site plans shall be reviewed and evaluated for consistency with all applicable regulations of this Comprehensive Land Use Regulations Ordinance.
- 3. In the event that a proposed site plan does not satisfy the applicable criteria established for review by this Section, modifications to the site plan by the applicant which would result in increased compatibility or would mitigate unfavorable impacts or would cause the site plan to conform to applicable requirements may be considered.

4.3.2.10 Findings

The City Planner or Zoning Board shall make the following findings before site plan approval:

- 1. That the proposed use and site development, together with any modifications applicable thereto, will be compatible with existing conforming or permitted uses on adjacent sites or sites across from the proposed development site.
- 2. That any required modifications to the site plan are reasonable and are the minimum necessary to minimize potentially unfavorable impacts.
- 3. That the site plan complies with these Comprehensive Land Use Regulations.

4.3.2.11 Modification of Site Plan

The City Planner or the Zoning Board may require modification of a site plan as a prerequisite for approval when required by the Special Use Criteria of Article 8 or Special District Criteria for the district in which the use is proposed, or other provisions of these regulations or other

City, state or federal regulations; or, when the site plan is reviewed in connection with a rezoning, they may recommend such modifications as may be reasonably necessary to achieve the purposes of these regulations. Such modifications may include, but shall not be limited to: Provision for special yards, open spaces, buffers, fences, walls, and screening; for installation and maintenance of landscaping and drainage control measures; improvements of access and circulations; rearrangements of structures or activities within the site; location and character of signs; and such other site plan features as necessary to ensure compatibility with surrounding uses and to support the findings required by this Section.

4.3.2.12 Effective Date

The decision of the City Planner shall be effective fifteen (15) days after a decision rendered by the Planner on the application, unless appealed. The decision of the Zoning Board shall take effect ten (10) days after the date a decision is rendered, unless appealed. The decision of the City Council shall be effective immediately subject to modification provisions of the site plan.

4.3.2.13 Lapse of Approval

- 1. Unless a longer time shall be specifically established as a condition of approval, a Site Plan approval shall lapse and become void two (2) years following the date on which such approval became effective, unless prior to the expiration of two (2) years a building permit is issued and construction is commenced and diligently pursued toward completion, or a Certificate of Occupancy is issued for the use, or the site is occupied if no building permit or Certificate of Occupancy is required.
- 2. A Site Plan approval subject to lapse may be renewed by the Zoning Board for an additional period of one (1) year, provided that prior to the expiration date, a written request for renewal is filed with the City Planner.

4.3.2.14 Modification of Site Plan Review Approval

The procedural requirements for Site Plan Review Approval as specified in this Section 4.3 et seq. shall apply to an application for modification, expansion, or other change in an approved Site Plan, provided that minor revisions or modifications may be approved by the City Planner if he determines that the circumstances or conditions applicable at the time of original approval remain valid, and that changes would not affect the findings prescribed in this Section. The City Planner shall report to the Zoning Board on a quarterly basis the number and kinds of modifications being approved.

4.3.2.15 Suspension and Revocation

- 1. Upon violation of any applicable provision of these regulations, or, if granted subject to conditions, upon failure to comply with conditions, a Site Plan approval shall be suspended upon notification by the City Planner to the owner of a use or property subject to the Site Plan.
- 2. The Zoning Board shall give notice as required for zoning permits and hold a public hearing within forty (40) days of such notification, and upon a finding that the regulation, general provision, or condition is not being complied with, may revoke the Site Plan approval or take such action as the Zoning Board deems necessary to ensure compliance with the regulation, general provision, or condition.
- 3. The decision of the Zoning Board to revoke a Site Plan approval shall be effective immediately.

4.3.2.16 New Applications

Following the denial of a site plan or revocation of a site plan by the Council no application for Site Plan Review for the same or substantially the same Site Plan on the same or substantially the same site shall be filed within one year from the date of denial or revocation.

4.3.2.17 Approval to Run with the Land

The approved site plan shall be signed by the approving official and recorded with the Clerk of Court of the Parish. A Site Plan approval pursuant to these provisions shall run with the land and shall continue to be valid upon a change of ownership of the site or structure which was the subject of the application, subject to the lapse of approval provisions regarding lapse of approval provided herein. Cost of recordation shall be born by the applicant.

4.3.2.18 Site Plans Approved under Prior Regulations

A Site Plan approved pursuant to regulations in effect prior to the effective date of these provisions shall be deemed a pre-existing approved plan, and the use for which such site plan was approved shall be permitted to continue subject to any conditions prescribed at the time of site plan approval. A use subject to a pre-existing approved site plan shall be subject to these provisions with respect to lapse, modification, suspension or revocation.

4.3.3 Procedures and Fees for Conditional Use Permits and Planned District Zoning

4.3.3.1 Title and Purpose

This Section shall be known as the Conditional Use and Planned District Zoning Procedure. The purpose of this procedure is to provide for review and discretionary approval of uses or development sites typically having unusual site development features or unique operating characteristics requiring special consideration so that they may be located, designed, and operated compatibly with uses on surrounding properties and within the City at large. The Conditional Use Procedure and Planned District Zoning process is intended to encourage broad public review and evaluation of site development features and operating characteristics, and to ensure adequate mitigation of potentially unfavorable impacts.

4.3.3.2 Jurisdiction

The City Planner shall be responsible for administration of the Conditional Use Procedure, and the Planning Commission shall be responsible for review, evaluation, and action on all applications for a Conditional Use Permit.

4.3.3.3 Concurrent Applications

Application for a Conditional Use Permit and for Rezoning or Planned District zoning for the same property may be made concurrently, subject to the fees applicable to a rezoning or Planned District zoning only. The Planning Commission and Zoning Commission may hold the public hearing on the Rezoning and the Conditional Use Permit at the same meeting and may combine the two hearings. The City Council likewise may hold the two public hearings in combination and may approve both the Conditional Use and rezoning or Planned District zoning by one ordinance.

4.3.3.4 Application and Fee

Applications for Conditional Use Permits and Planned District zoning approvals shall be filed with the City Planner. The application shall include the following:

- 1. Name and address of the owner and applicant and sworn affidavit of ownership.
- 2. Address, legal description and boundary survey of the property, including any existing structures.
- 3. If the applicant is not the legal owner of the property, a sworn statement by the owner that the applicant is the authorized agent of the owner of the property.

- 4. A statement describing the nature and operating characteristics of the proposed use, including any data pertinent to the findings required for approval of the application. For uses involving public assembly or industrial processing, or uses potentially generating high volumes of vehicular traffic, the City Planner may require specific information relative to the anticipated peak loads and peak use periods, relative to industrial processes and the ability of the use to meet performance standards, or substantiating the adequacy of proposed parking, loading, and circulation facilities.
- 5. Site plans, conceptual building elevations, conceptual improvement plans, and such additional maps and drawings, all sufficiently dimensioned as required to illustrate the following:
 - a. The date, scale, north point, title, name of owner, and name of person preparing the site plan.
 - b. The location and dimensions of boundary lines, with distances and bearings, easements, and required yards and setbacks, water courses, drainage features and location and size of existing and proposed streets and alleys, 100-year floodplains, as well as areas of periodic inundation.
 - c. The location, height, bulk, percent of impervious site surface, general appearance, and intended use of existing and proposed buildings on the site, and the approximate location of existing buildings and their existing uses on abutting sites within fifty (50) feet.
 - d. The location of existing and proposed site improvements including parking and loading areas, pedestrian and vehicular access, landscaped areas, utility or service areas, fencing and screening, signs, and lighting.
 - e. A conceptual landscape plan showing the location and size of the existing and proposed landscaped areas and the number and location of Class A and B trees proposed or required to be preserved.
 - f. The number of existing and proposed off-street parking and loading spaces, and a calculation of applicable minimum requirements.
 - g. A conceptual drainage plan showing existing and proposed topography and grading and proposed drainage structures, retention ponds or water quality enhancement facilities.

- h. The relationship of the site and the proposed use to surrounding uses, including pedestrian and vehicular circulation, current use of nearby parcels, and any proposed off-site improvements to be made.
- 6. In addition, Site Plans submitted in conjunction with Planned District zoning, shall include:
 - a. The dwelling intensity of any residential areas and the lot sizes and locations of any other uses within the Planned Development.
 - b. Areas proposed to be conveyed, dedicated or reserved for parks, parkways, playgrounds, school sites, public buildings and similar public and semipublic uses.
 - c. A plan for the location of all public utilities.
 - d. A copy of all agreements, provisions or covenants which are proposed to govern the use, maintenance and continued protection of the Planned Development and any of its common open space.
 - e. A representation of the general use and character of land adjacent to the Planned Development area within two hundred (200) feet.
 - f. A landscape plan along the boundary of a Planned Development to a depth of one hundred (100) horizontal feet.

However, exact building locations need not be dimensioned on the site plan for a Planned Development so long as all areas within which buildings may be constructed or maintained are specifically delineated by building setback lines.

7. <u>Fees for Conditional Use Requests</u> - To initiate any Conditional Use Permit request, the party or parties requesting the Conditional Use approval shall deposit a sum based on the fee established by the City Council for Planned District zoning approvals for each request to the City. Under no condition shall the fee be refunded for the failure of the requested Conditional Use Permit to be granted or for the withdrawal of the request.

4.3.3.5 Public Hearing and Notice

The Planning Commission shall hold a public hearing on each application for a Planned District zoning or amendment or for a Conditional Use Permit. Public notice shall be given

as required for zoning amendments. At the public hearing, the Commission shall review the application and shall receive pertinent evidence concerning the proposed use and the proposed conditions under which it would be operated or maintained, with respect to the findings prescribed herein.

4.3.3.6 Report of the City Planner

The City Planner or Designee shall review the application or proposal and shall prepare a report thereon which shall be filed with the Planning Commission and available to the applicant at least ten (10) days prior to the public hearing.

4.3.3.7 Action by the Planning Commission

The Planning Commission shall act on the application not more than twenty (20) days following the closing of the public hearing on a Planned District zoning or Conditional Use Permit application. The Commission may recommend to grant a Conditional Use Permit or approve a Planned District zoning or amendment as applied for or in a modified form or subject to conditions, or may recommend denial of the application to the City Council. The Commission shall notify the applicant of its recommendation by mail.

4.3.3.8 Review and Evaluation Criteria

The City Planner, the Planning Commission and the City Council shall review and evaluate and make the following findings before granting a Conditional Use Permit or Planned District zoning using the following criteria:

- 1. Comparison with applicable regulations and standards established by the Comprehensive Land Use Regulations applicable to the proposed use and site.
- 2. Compatibility with existing or permitted uses on abutting sites, in terms of building height, bulk and scale, setbacks and open spaces, landscaping and site development, and access and circulation features.
- 3. Potentially unfavorable effects or impacts on other existing conforming or permitted uses on abutting sites, to the extent such impacts exceed these which reasonably may result from use of the site by a permitted use.
- 4. Modifications to the site plan which would result in increased compatibility, or would mitigate potentially unfavorable impacts, or would be necessary to conform to applicable regulations and standards and to protect the public health, safety, morals, and general welfare.

- 5. Safety and convenience of vehicular and pedestrian circulation in the vicinity, including traffic reasonably expected to be generated by the proposed use and other uses reasonable and anticipated in the area considering existing zoning and land uses in the area.
- 6. Protection of persons and property from erosion, flood or water damage, fire, noise, glare, and similar hazards or impacts.
- 7. Location, lighting, and type of signs; and relation of signs to traffic control and adverse effect on adjacent properties.
- 8. Adequacy and convenience of off-street parking and loading facilities and protection of adjacent property from glare of site lighting.
- 9. Conformity with the objectives of these regulations and the purposes of the zone in which the site is located.
- 10. Compatibility of the proposed use and site development, together with any modifications applicable thereto, with existing or permitted uses in the vicinity.
- 11. That any conditions applicable to approval are the minimum necessary to minimize potentially unfavorable impacts on nearby uses and to ensure compatibility of the proposed use with existing or permitted uses in the same district and the surrounding area.
- 12. That the proposed use, together with the conditions applicable thereto, will not be detrimental to the public health, safety, or welfare, or community aesthetics, or materially injurious to properties or improvements in the vicinity.

4.3.3.9 Conditions of Approval

The Planning Commission may recommend and the City Council may establish conditions of approval. Conditions may include, but shall not be limited to, requirements for special yards, open spaces, buffers, fences, walls, and screening; requirements for installation and maintenance of landscaping and erosion control measures; requirements for street improvements and dedications, regulation of vehicular ingress and egress, and traffic circulation; regulation of signs; regulation of hours or other characteristics of operation; requirements for maintenance of landscaping and other improvements; establishment of development schedules or time limits for performance or completion; and such other

conditions as the Commission or City Council may deem necessary to insure compatibility with surrounding uses to preserve the public health, safety, and welfare, and to enable the Commission to make the findings required by the preceding Section.

4.3.3.10 Planning Commission Recommendations to the City Council

The recommendations of the Planning Commission shall be forwarded to the City Council within 10 days after the date on which action is announced.

4.3.3.11 Enactment by Ordinance

The decision of the Planning Commission on an application for a Conditional Use Permit or Planned District shall be forwarded to the City Council as a recommendation to grant or deny after the Planning Commission holds a public hearing thereon. The City Council shall hold a public hearing on said application. Notice of the public hearings before the Planning Commission and City Council shall be given in the same manner as the notice required for zoning amendments. In the event the decision of the Planning Commission on the application for a Planned District or Conditional Use Permit is adverse thereto, or in the event a protest against the proposed Planned District or Conditional Use Permit is presented, in writing, to the City Council, duly signed and acknowledged by the owners of at least twenty (20) percent of the property situated in the area bounded by lines two hundred (200) feet in each direction and one each side of the area included in such proposed Planned District, such ordinance approving the Planned District shall not be passed except by an affirmative vote of a two-thirds majority of the members of the City Council.

4.3.3.12 Lapse of Conditional Use Permit or Uses Approved in Conjunction with Planned District Zoning

- 1. For the purposes of this subsection, a Conditional Use Permit shall be deemed to have been granted upon the final affirmative action on the matter by the City Council, notwithstanding the fact that the effectiveness of the Conditional Use Permit for any other purposes may have been stayed due to the necessity of meeting conditions lawfully placed on the granting of the Conditional Use Permit.
- 2. Except as otherwise stated in these regulations, authority to issue construction or occupancy permits pursuant to an approved Conditional Use Permit shall expire two years following City Council approval unless during the two year period, a construction permit is obtained. If a construction permit is obtained, the Conditional Use Permit shall continue in force and effect until (1) two years following the issuance of such construction permit, or (2) two years following the issuance of a Certificate of Occupancy,

whichever is later, providing that construction is not stopped for a period of six months or more. If construction is stopped, the Conditional Use Permit shall expire (1) at the end of such six month period, or (2) two years following the issuance of such construction permit, whichever is later.

- 3. Where more than one building permit is covered by a Conditional Use Permit and when the Certificate of Occupancy has been obtained on the last building as set forth above (or when the time has been extended as set forth below), the developer shall have an additional two years during which to secure a second construction permit and so on until the project is completed. Otherwise, the Conditional Use Permit shall expire as set forth in these regulations.
- 4. Construction permits obtained more than five years following the conditional use approval date shall comply with the rules, regulations and ordinances that have become effective since the approval date of the Conditional Use Permit. For the purposes of this subsection, a Conditional Use Permit shall be deemed approved upon the affirmative action of the City Council, notwithstanding the necessity of meeting conditions lawfully placed on the Conditional Use Permit approval.
- 5. The City Council may, upon application of the developer, grant a one-year extension to any two-year time period during which a construction permit or Certificate of Occupancy may be issued when one or more of the following conditions have been met:
 - a. Construction permits have been issued, materials have been acquired and the foundation of at least one building has been placed on the site.
 - b. Where no construction is required, an occupancy permit has been issued and actual operation of the use has begun.
 - c. The developer has made application to the City Council stating reasons, prior to the expiration date of the Conditional Use Permit.
- 6. The Building Official shall report to the City Council the actual development accomplished as it relates to the approved Conditional Use Permit. The Council, before acting, shall hold a public hearing on the application.
- 7. After the authority for the issuance of construction permits or certificates of occupancy has expired by default pursuant to an approved Conditional Use

Permit, no construction permit or Certificate of Occupancy shall be issued except under a Conditional Use Permit approved upon a new application.

8. Planned District sites having site plan approval or building permits issued prior to the effective date of this Ordinance shall be subject to the provisions of this Section.

4.3.3.13 Modification of Planned District or Conditional Use Permit

An application for modification, expansion, or other change in a Conditional Use Permit or Planned District approved shall be in accordance with the procedures of Section 4.3.3, provided that minor revisions or modifications may be approved by the City Planner upon determination that the circumstances or conditions applicable at the time of original approval remain valid, and that changes would not affect the findings prescribed in Section 4.3.3.8.

4.3.3.14 Suspension and Revocation

- 1. Upon violation of any applicable provision of this Section, or, if granted subject to conditions, upon failure to comply with conditions, a Conditional Use Permit may be suspended upon notification by the City Clerk to the owner of the use or property subject to a Conditional Use Permit.
- 2. The City Council shall hold a public hearing within forty (40) days of such notification, and upon a finding that the regulation, general provision, or condition is not being complied with, may revoke the Conditional Use Permit or take such action as may be necessary to ensure compliance with the regulation, general provision, or condition.
- 3. The decision of the Council to revoke a Conditional Use Permit shall be effective immediately.

4.3.3.15 New Applications

Following the denial or revocation of a Conditional Use Permit or Planned District zoning, no application for a Conditional Use Permit or Planned District zoning for the same or substantially the same use or conceptual plan, on the same or substantially the same site shall be filed within one (1) year from the date of denial or revocation.

4.3.3.16 Approval to Run with the Land

The ordinance approving the Conditional Use or Planned District zoning and the plans approved in conjunction with the ordinance shall be recorded with the Clerk of Court in the

Parish courthouse. A Conditional Use Permit or Planned District zoning granted pursuant to these provisions shall run with the land and shall continue to be valid upon a change of ownership of the site or structure which was the subject of the Conditional Use Permit application subject to the provisions regarding expiration of approval. Cost for recordation shall be born by the applicant.

4.3.3.17 Prior Special Permits and Covenants

- 1. A use legally established pursuant to a Conditional Use Permit or a Planned District zoning prior to the date of adoption of these regulations shall be deemed a pre-existing conditional use or planned development and shall be permitted to continue,
 - provided that it is operated and maintained in accord with any conditions prescribed at the time of its establishment.
- 2. Alteration or expansion of a pre-existing conditional use or planned development shall be permitted only upon the granting of a Conditional Use Permit or an amendment to the Planned District Ordinance as prescribed in these regulations, except for alterations not exceeding \$2,500 in value as determined by the Building Official.
- 3. A Conditional Use Permit or Planned District amendment ordinance shall be required for the reconstruction of a structure primarily approved as a Conditional Use or planned development, if the structure is destroyed by fire or other calamity, by Act of God, or by the public enemy to a greater extent than fifty (50) percent of the repair to replacement cost. The extent of damage or partial destruction shall be based upon the ratio of the estimated cost of restoring the structure to its condition prior to such damage or partial destruction to the estimated cost of duplicating the entire structure as it existed prior thereto. Estimates for this purpose shall be made by or shall be reviewed and approved by the Building Inspector and shall be based on the minimum cost of construction in compliance with the Building Code.
- 4.3.4 Procedures and Fees for Filing Appeals and Requests for Variances to the Zoning Board

4.3.4.1 Procedures for Filing Appeals

1. Any person aggrieved by a decision of any of the officers, departments, commissions, boards or bureaus that administer the provisions of these land use regulations may appeal to the Zoning Board within thirty (30) days after

- the decision has been rendered. If a building or structure is believed by the Building Inspector to be unsafe or dangerous, the City Planner may limit the time for such appeal to a shorter period with the consent of the Mayor.
- 2. City officials or employees shall produce all papers, correspondence, and records requested by the Zoning Board for any hearing or meeting held by the Board. An appeal stays all proceedings and furtherance of the action appealed from, unless the Building Inspector certifies to the Board, after a notice of appeal shall have been filed with the City Planner, by reason of facts stated in the certificate, that a stay would, in his opinion, cause imminent peril of life or property. In such cases, proceedings shall not be stayed other than by a restraining order which may be granted to the Zoning Commission by the District Court of the parish, and notice to the City Clerk and on due cause shown. The Zoning Board shall adopt rules of procedure to fix a reasonable time for hearing appeals, give notice thereof, as well as due notice to the interested parties, and shall decide the appeal within a reasonable time and in accordance with the Adopted Rules and Procedures of the Zoning Board. At the hearing, any party may appear in person or by an attorney.

4.3.4.2 Fees for Filing Appeals and Variance Requests

- 1. <u>Fees for Appeals to the Zoning Board</u> A fee of twenty-five dollars (\$25.00) shall be paid to the Secretary of the Zoning Board at the time the notice of appeal is filed, which the Secretary shall transmit to the Director of Finance. Fees related to appeals shall be credited of the general revenue fund of the community.
- 2. <u>Fees for Variance Requests</u> To initiate any variance request, as permitted by this code or in conjunction with an appeal, the party or parties requesting the variance shall deposit with the Secretary of the Zoning Board the sum of seventy-five dollars (\$75.00) for each variance requested. Under no circumstances shall the fee be refunded for the failure of such variance to be granted or for the withdrawal of the request by the applicant.

4.3.4.3 Conformity with Purpose of Land Use Regulations in Evaluation of Appeals

In consideration of all appeals and all proposed exceptions or variances under the terms of this Ordinance the Zoning Board shall, before making any exception or variance from the ordinance in a specific case, give due consideration to the general purposes of these regulations as stated in Article 1. In consideration of all appeals and all proposed exceptions or variances under the terms of this Ordinance the Board shall, before making any exception

or variance from the ordinance in a specific case, first determine that it will not impair an adequate supply of air or light to adjacent property, or unreasonably increase the congestion in public streets, or danger of fire, or endanger the public safety, or unreasonably diminish or impair established property values within the surrounding area or in any other respect impair the public health, safety, morals, comfort, welfare or aesthetics of the inhabitants of the community.

4.3.4.4 Scope of Action by the Zoning Board

In exercising its powers, the Board may reverse or affirm, wholly or partly, or may modify the order, requirement, decision or determination appealed from, and may make such order, requirement, decision or determination as ought to be made, and to that end shall have all powers of the officers, departments, commissions, boards or bureaus that administer the provisions of these Land Use Regulations. In granting a variance, the Board may attach thereto such conditions as it may deem advisable to further the purposes of this Ordinance.

The concurring vote of four members of the Board shall be necessary to reverse any order, requirement, decision or determination of the officers, departments, commissions, boards or bureaus that administer the provisions of these regulations, or to decide in favor of the applicant on any matter upon which it is required to pass under this or any ordinance, or to effect any variance of this Ordinance under which it has the power to grant a variance.

4.3.4.5 Variance Procedures

- 1. Authority The Zoning Board may authorize a variance upon appeal of a decision of an official administering the provisions of this Ordinance when a property owner can show that a strict application of the terms of this Ordinance relating to the use, construction, or alteration of buildings or structures, or the use of land will impose upon him unusual and practical difficulties or particular hardship; but only when the Board is satisfied that granting such variance will not merely serve as a convenience to the applicant, but will alleviate some demonstrable and unusual hardship or difficulty so great as to warrant a deviation from provisions established by this Ordinance and at the same time the surrounding property will be properly protected.
- 2. <u>Procedure</u> Variance requests shall accompany the application for appeal, shall clearly identify the Section of the ordinance from which the variance is requested and shall be filed in accordance with the Rules of Procedure of the Zoning Board. Variance requests shall not be considered by the Zoning Board until:

- a. There shall have been held a public hearing in relation thereto before the Zoning Board at which time interested citizens and parties shall have had an opportunity to be fully heard. In addition, in the event that the requested variance involves any exception to the provisions of Article 10 of this Ordinance or otherwise involves or affects the regulation or control of signage, there shall also be held a hearing before the Community Appearance Commission. The public hearing provided for herein may, with the consent of both the Chairman of the Zoning Board and the Chairman of the Community Appearance Commission be held conjointly and coincidentally at a joint meeting of the Zoning Board and the Community Appearance Commission.
- b. Notice of the requested variance and of the time and place of the public hearing or hearings thereon shall have been published once a week for three (3) weeks in the City's official journal. At least fifteen (15) days shall elapse between the first publication and the date of the hearing or hearings to which said publication relates.

In addition, a printed notice in bold type shall be posted for not less than ten (10) consecutive days prior to the public hearings on signs prepared, furnished, and placed by the Department of Planning and Development staff upon the principal and accessible rights-of-way adjoining the site where the variance is requested.

4.3.4.6 Procedures and Fees for Non-Conforming Development Site Variances

- 1. <u>Background</u> The non-conforming provisions of Section 4.2 of these regulations establishes specific conditions under which non-conforming development sites must be upgraded to meet the standards established by this Code. Practical difficulties may exist which prevent the upgrading of certain non-conforming development sites to standards imposed by this Code.
- 2. <u>Purpose</u> This variance procedure has been established in order to provide a viable economic use of existing structures and to insure that non-conforming development sites are upgraded to the extent deemed possible by the Zoning Board. Provision of adequate parking and access shall take precedent over other applicable sections of

these regulations in evaluating applications for non-conforming development site variances.

3. Occupancy of Non-Conforming Development Site

- a. Occupancy of a non-conforming site in conjunction with a change of use:
- i. <u>Comparable or less parking demand</u>: If a proposed change of use does not increase the number of required off-street parking spaces by more than five (5) percent over the number of spaces required for the then present use, an applicant for a non-conforming development site variance may, at his own risk, occupy the site prior to approval of non-conforming development site variance.
- ii. <u>Increase of parking demand</u>: If the proposed change of use increases the number of required off-street parking spaces by more than five (5) percent over the number of spaces required for the then present use, alteration of the site to conform to this Ordinance or approval of a non-conforming development site variance is required prior to issuance of permits or occupancy of the site.
- b. <u>Non-Conforming Shopping Center</u> Subsequent to the loss of legally non-conforming status for reasons defined by this Ordinance alteration of the site to conform to this Ordinance or approval of a non-conforming development site variance is required prior to issuance of a development permit or occupancy of the site.
- 4. <u>Time Variance</u> The Zoning Board may provide a one-time temporary variance for up to 365 days. In reviewing other than time variance requests, the Board shall not approve variance requests without first finding that practical difficulties do exist.
- 5. <u>Practical Difficulties</u> For non-conforming development sites, practical difficulties shall be limited to those situations:
 - a. In which existing buildings would have to be relocated in order to meet the setbacks or landscaping requirements or other requirements prescribed by this Ordinance, or
 - b. In which inadequate parking or maneuvering areas would be created or rendered less in conformance if the landscaping prescribed by this Ordinance were implemented, or
 - c. In which other site conditions that would make conformance impossible.

The cost incurred to remove concrete or other impervious surfaces to conform to the provisions of this Ordinance shall not constitute practical difficulties.

- 6. <u>Initiation</u> The owner or person having a contractual interest, other than a lessee, in the property for which a variation is sought may initiate a request for a non-conforming development site variance.
- 7. <u>Application</u> The application and the number of copies required by the Rules of Procedure of the Zoning Board shall be submitted on forms provided by the City Planner. The application shall include the following information:
 - a. Name, address, and telephone number of applicant.
 - b. Nature and extent of the applicant's interest in the property for which the variance is requested.
 - c. A survey drawn to scale indicating the existing dimensions and improvements of the property. Such plan shall be no smaller than $8.5" \times 11"$ inches and no larger than $24" \times 36"$ inches.
 - d. A site plan drawn to scale indicating the proposed site plan of the property. Such plan shall be no smaller than $8.5" \times 11"$ inches and no larger than $24" \times 36"$ inches.
 - e. A statement identifying the practical difficulties applicable to the variance request and, if applicable, the nature of the time variance requested.

8. City Planner Responsibilities

- a. The City Planner shall have five (5) working days to determine if the application, as submitted, meets the requirements established herein. If an application is found to be incomplete, the City Planner shall notify the applicant within five (5) working days of the reasons therefore and advise the applicant of the requirements for an acceptable application.
- b. Upon receipt of an acceptable application the City Planner shall place the application on the agenda of the next regularly scheduled Zoning Board meeting for which the required advertising procedures can be met.
- c. The City Planner shall prepare a staff report for the Board and shall provide it to the Board and the applicant at least ten (10) working days prior to the meeting.

- 9. <u>Public Notice</u> The following actions shall be taken to provide public notice of the variance request:
 - a. The property will be posted with a sign no smaller than $11" \times 17"$ at least fifteen (15) days prior to the Public Hearing.
 - b. The Public Notice shall be advertised in the City's official journal once a week for three (3) consecutive weeks, and no less than fifteen (15) days shall elapse between the first publication and the date of the Public Hearing.
- 10. <u>Zoning Board Action</u> Within forty (40) days of the meeting at which the variance request is first eligible for decision, the Zoning Board shall take one of the following actions or the request shall be deemed approved:
 - a. Approve the variance as requested,
 - b. Approve the variance subject to specific conditions,
 - c. Deny the variance request, or
 - d. Table at written request of the applicant.
- 11. <u>Conditions on Variance</u> The Board may set forth conditions in the motion granting a non-conforming development site variance. Such conditions may include, but are not limited to the following:
 - a. Limit the manner in which the use is conducted, including restrictions on the time an activity may take place and restraints to minimize such environmental effects as noise, vibration, discharge of pollutants, glare and odor.
 - b. Establish a special yard or other open space or lot area or dimension.
 - c. Limit the height, size or location of a building or other structure.
 - d. Designate the size, number, location or nature of vehicle access points.
 - e. Increase the amount of street dedication, roadway width, or improvements within the street right-of-way.
 - f. Designate the size, location, screening, drainage, surfacing or other improvements of a parking or truck loading area.

- g. Limit or otherwise designate the number, size, location, height or lighting of signs.
- h. Limit the location and intensity of outdoor lighting or require its shielding.
- i. Require diking, screening, landscaping or another facility to protect adjacent or nearby property and designate standards for installation or maintenance of the facility.
- j. Designate the size, height, location or materials for a fence.
- k. Protect existing trees, vegetation, water resources, wildlife habitat or other significant natural resources.
- l. Specify other conditions to permit development of the site in conformity with the intent and purpose of the code.
- 12. <u>Tenure of Variance</u> Except for time variances as provided for above, variances are valid for an unlimited time, irrespective of ownership.
- 13. <u>Fee</u> An application fee of \$75.00 as for other variances is hereby established.

4.3.4.7 Scope of Civil Court Recourse

Any person or persons, or any officer, department, board, bureau or any other agency of the community jointly or severally aggrieved by any decision of the Zoning Board may present to the District Court of the parish, within thirty (30) days after filing of the decision in the office of the Board, a writ of certiorari asking for such relief and under such rules and regulations as are provided for such matters in appropriate legislation of the state.

4.3.4.8 Notification of Decisions of the Zoning Board

- 1. Every appeal or variance decision of the Zoning Board shall be recorded in the Board's official record and shall indicate the vote upon the decision. Every decision shall be promptly filed in the case file in the Office of the Department of Planning and Development and shall be open to public inspection. A copy shall be sent by mail or otherwise to the appellant and a copy shall be kept in the case file of the Office of the Department of Planning and Development for two (2) years after filing.
- 2. The Zoning Board shall in every case reach a decision without unreasonable or unnecessary delay.

3. If a decision of the Zoning Board reverses or modifies a decision of the City Planner, Building Inspector, or Director of Public Works or varies the application of any provision of this Land Use Regulation Ordinance action shall be taken by the City Planner or Building Inspector or Director of Public Works in accordance with such decision within a reasonable period of time.

4.3.5 Procedure and Fees for Issuance of a Home Occupation Permit

An application for a home occupation permit shall be filed with the Building Inspector by any resident who wishes to establish a home occupation in a private residence. Said permit shall establish the extent and duration of the home occupation as well as grant permission for the proposed use. In addition, the home occupation permit will establish conditions of approval and procedures for revocation and renewal as follows:

1. <u>Application</u> - Application for a home occupation permit shall be made to the City Planner on a form provided by the City Planner and shall be accompanied by a filing fee of fifteen dollars (\$15.00). A decision on the application and notification to the applicant of that decision shall be made within fifteen (15) calendar days of the date a completed application is received.

- 2. <u>Scope</u> In cases where the application is deemed not to be within the scope of a home occupation as defined herein, the application will be denied.
- 3. <u>Time Limit</u> All home occupation permits shall be valid for a period of one year from initial date of approval.
- 4. <u>Voiding of Permit</u> The City Planner may void any home occupation permit for noncompliance with the criteria set forth in this Section. Revocation may take place at any time prior to the expiration date of the permit. If the permit is revoked or is not renewed, it becomes null and void, and said use shall be terminated.
- 5. <u>Appeal to Zoning Board</u> The decision of the City Planner concerning approval or revocation shall be final unless a written appeal is filed with the Zoning Board within thirty (30) calendar days of the decision.
- 6. <u>Inspection</u> Home occupation applicants shall permit a reasonable inspection of the premises by the Planning Department to determine compliance with this Section.
- 7. Renewal Home occupation permits shall be renewed annually provided there has not been any violation of the provisions of this Chapter. Requests for renewals shall be submitted to the Planning Department in writing, accompanied by a ten (\$10) dollar renewal fee, no later than one (1) month prior to expiration of the permit.

ARTICLE 5 - BUILDING CODES AND PERMITTING REQUIREMENTS

5.1. GENERAL PROVISIONS

5.1.1 Title

The provisions of Article 5 shall be known and may be cited as "The Building Codes".

5.1.2 Building Inspector as Administrator

The Building Inspector shall administer the provisions of Article 5 where needed in conjunction with other City personnel in accordance with the responsibilities assigned by the provisions of Article 2 of these regulations. In administering the provisions of Article 5 the Building Inspector shall interpret and enforce the provisions of the adopted codes of the Southern Building Code Congress, International (SBCCI), issue all permits and perform or cause to be performed all necessary inspections as provided.

5.1.3 Code Remedial

This Article is hereby declared to be remedial and minimum and shall be construed to secure the beneficial interests and purposes thereof, which are public safety, health and general welfare, through structural strength, stability, sanitation, adequate light and ventilation, and safety to life and property from fire and other hazards incident to the construction, alteration, repair, removal, demolition, use and occupancy of buildings, structures, signs or premises.

5.1.4 Scope

The provisions of this Article shall apply to the use, occupancy or development of land, including but not necessarily limited to the construction, alteration, repair, equipment, plumbing, gas, piping and appliances, heating, air conditioning, excavation, grading, swimming pools, electrical installations, use and occupancy, location, maintenance, removal and demolition of every building or structure or any appurtenances connected or attached to such buildings or structures, including signs.

5.1.5 Rules and Procedures of Building Inspector

The Building Inspector, through the Director of Planning, shall promulgate rules and procedures for the issuance of permits and certificates as prescribed in this Article and consistent therewith, it being the intent of this requirement that the standards of federal or state bureaus, national technical organizations or fire underwriters, as the same may be amended from time to time, shall serve as a guide in fixing the minimum rules of practice under this code. Rules promulgated as herein provided shall have the same force and effect as provisions of this Article.

5.1.6 Permit Required

Permits and inspections shall be required for all work within the scope of this Article to assure compliance with the provisions of this Land Use Regulations Ordinance and all other applicable codes of the City of Mandeville.

5.1.7 Permit Fees, General

No permit shall be issued until the fees prescribed in this Article shall have been paid. Nor shall any amendment to the permit be approved until additional fees associated with an increase in the estimated cost of the building or structure have been paid.

5.1.8 Failure to Obtain a Permit

If any person commences any work on a building or structure before obtaining the necessary permit from the City, such person shall be subject to the penalties prescribed by this Ordinance.

5.1.9 Double Permit Fees as Penalty

On all work commenced without first obtaining a permit as required by the provisions of this Article, the fees for the required permits shall be doubled. However, the payment of such double fee shall not relieve any person from fully complying with the requirements of this Article and all applicable provisions of this Land Use Regulations Ordinance in the execution of the work permitted.

5.1.10 Right of Entry

The Building Inspector shall enforce the provisions of this code and he, or his duly authorized representative, may enter any building, structure or premises in the City to perform any duty imposed upon him by this Article.

5.1.11 Posting Permit and Inspection Cards

Work requiring a building permit shall not be commenced until a building permit card has been posted in a conspicuous place on the front of the premises. The permit and inspection card shall be protected from the weather and be posted at such location as to permit the Building Inspector to conveniently make the required entries thereon. This permit and inspection card shall be maintained in such location by the permit holder until the Certificate of Occupancy has been issued by the Building Inspector.

5.1.12 Stop Work Orders

Upon notice to the owner or owner's agent from the Building Inspector that work on any building or structure is being done contrary to the provisions of this code or otherwise required by law or development agreement or is determined to be in a dangerous or unsafe manner, such work shall be immediately stopped by the owner or owner's agent. Notice to stop other than unsafe work shall be in writing and shall be given to the owner of the property or to his agent, or to the person doing the work to be stopped, and shall state the conditions under which work may be resumed. Where an

emergency exists, verbal notice may be given by the Building Inspector to the owner and subsequently confirmed in writing.

5.1.13 Revocation of Permits

The Building Inspector shall revoke a permit or approval issued under the provisions of this act, whenever he discovers there has been any false statement or misrepresentation as to a material fact in the application or plans on which the permit or approval was based.

5.1.14 Issuance of Permit Not a Waiver

The issuance of a permit shall in no case be construed as a waiver of any of the provisions of this Ordinance or of any other ordinances or regulations of the City. No permit issued shall be deemed to constitute permission or authorization to perform unlawful work nor shall any permit issued constitute a defense in an action to abate unlawful work.

5.1.15 Unsafe Buildings

All buildings or structures which are unsafe, unsanitary or not provided with adequate egress or which constitute a fire hazard or are otherwise dangerous to human life, or which in relation to existing use constitute a hazard to safety or health by reason of inadequate maintenance, dilapidation, obsolescence or abandonment, or not are severally in contemplation of this section, unsafe buildings. All such unsafe buildings are hereby declared illegal and shall be abated by repair or rehabilitation or by demolition in accordance with the following procedure:

- 1. Whenever the Building Inspector shall find any building or structure or portion thereof to be unsafe as defined in this section, he shall give the owner, agent or person in control of such building or structure notice by certified mail stating the defects thereof. This notice shall require the owners within a stated time period either to make repair or improvement of the building or structure or to demolish and remove the building or structure or unsafe portion thereof.
- 2. If necessary, such notice shall also require the building, structure or portion thereof to be vacated forthwith and not reoccupied until the specified repair or improvement is completed, inspected and approved by the Building Inspector. The Building Inspector shall cause to be posted at each entrance to such building a notice: "THIS BUILDING IS UNSAFE AND ITS USE OR OCCUPANCY HAS BEEN PROHIBITED BY THE BUILDING INSPECTOR." Such notice shall remain posted until the needed repairs are made or demolition is completed. It shall be unlawful for any person, firm or corporation or their agents or other servants, to remove such notice without written permission of the Building Inspector, or for any person to enter the building except for the purpose of making the required repairs or for demolishing same.

- 3. The owner, agent or person in control of the building or structure shall have the right, except in cases of emergency, to appeal the decision of the Building Inspector, as provided hereinafter, and to appear before the Zoning Board at a specified time and
 - place to show cause why the building or structure in question should not be declared unsafe.
- 4. In case the owner, agent or person in control cannot be found within the stated time limit, or if such owner, agent or person in control shall fail, neglect or refuse to comply with notice to repair, rehabilitate or to demolish and remove said building or structure or portion thereof, the Building Inspector after having ascertained the cost, shall with the consent of the Mayor cause said building or structure or portion thereof to be demolished, secured or required to remain vacant.
- 5. The decision of the Building Inspector shall be final in cases of emergency which, in his opinion, involve imminent danger to human life or health. He shall promptly cause such building, structure or portion thereof to be made safe or removed. For this purpose he may at once enter such structure or land on which it stands, or abutting land or structures, with such assistance and at such cost as he may deem necessary. He may vacate adjacent structures and protect the public by an appropriate fence or such other means as may be necessary and for this purpose may close a public or private way.
- 6. Costs incurred under Paragraphs 4 and 5 above shall be charged to the owner of the premises involved and shall be collected in the manner provided by law.

5.1.16 Floor Loads

- 1. No existing or new building shall be occupied for any purpose which will cause the floors thereof to be loaded beyond their safe capacity. The Building Inspector shall permit occupancy of a building when he is satisfied that such capacity will not thereby be exceeded.
- 2. It shall be the responsibility of the owner, agent, proprietor or occupant of Group F and G occupancies (as defined by the SBCCI), or any occupancy where excessive floor loading is likely to occur, to employ a competent architect or engineer in computing the safe load capacity. All such computations shall be certified by the architect or engineer and state the safe allowable floor load on each floor in pounds per square foot uniformly distributed; which statement shall thereupon be filed as a permanent record of the Division of Permits and Inspections.

5.1.17 Floor Load Signs Required

In every building or part of a building used for business storage, industrial or hazardous purposes, the safe floor loads shall be marked on plates of approved design which shall be supplied and securely affixed by the owner of the building in a conspicuous place in each story to which they relate. Such plates shall not be removed or defaced, and if lost, removed or defaced, shall be replaced by the owner of the building.

5.1.18 Loads in Excess of Posted Capacity

No such owner shall place or permit to be placed on any floor of a building a greater load than the safe load so determined and posted.

- 5.1.19 Establishing Guidelines For Potential Damage to City Streets, Rights-Of-Way, and Drainage During New Construction Or Major Renovation
 - 1. Prior to the clearing of any property for construction, the responsible individual (usually the builder) must post a damage deposit of \$500.00. The maximum deposit for an individual with multiple construction sites within a city block or the same subdivision is \$1,500.00.
 - 2. Prior to the beginning of clearing for construction, an individual from the Permits Division and the responsible individual will inspect the adjoining City-Owner property to establish its preconstruction or preclearing condition. The City employee will schedule this inspection. The property to be inspected will normally be limited to a 100 foot perimeter around any portion of the construction site.
 - 3. A representative of the Permits Division and the responsible individual will perform a post-construction inspection of City streets, roadside or dedicated drainage ways and rights-of-way for damage, and inspect the new construction for adherence to the approved plans and specifications prior to the issuance of an occupancy permit. Any damage or non-compliance found to be caused by the permittee or his subcontractors will be recorded and the permittee notified by certified mail.
 - 4. Damage deemed significant or minor by the Permits Division will result in denial of an occupancy permit until such time and corrective action has been taken by the permittee or his subcontractors.

5.2. CODES ADOPTED

5.2.1 Codes of Southern Building Code Congress International, Inc.

This Article adopts and incorporates by reference into these regulations the following building and construction codes of the Southern Building Code Congress International, Inc., namely:

- 1. Standard Building Code, as amended;
- 2. Standard Plumbing Code, as amended;
- 3. Standard Gas Code, as amended;
- 4. Standard Mechanical Code, as amended;
- 5. Standard Excavation and Grading Code, as amended;
- 6. Southern Standard Swimming Pool Code, as amended;

5.2.1.1 Codes for Buildings over 35' in Height

All standards in the codes of the Southern Building Code Congress International, Inc., (SBCCI) which apply to "high-rise buildings" seventy-five (75) feet or above shall apply to all buildings over thirty-five (35) feet in height within the City of Mandeville, in addition to the normal requirements of the codes.

5.2.1.2 Requirements Not Covered By Code

Any requirement necessary for the strength or stability of an existing or proposed building or structure or for the safety of the public or occupants, not specifically covered by this code shall be determined by the Building Inspector subject to appeal to the Zoning Board.

5.2.1.3 Alternate Materials and Alternate Methods of Construction

The provisions of this code are not intended to prevent the use of any material or method of construction not specifically prescribed by this code, provided any such alternate has been approved and its use authorized by the Building Inspector. The Building Inspector shall approve any such alternate provided he finds that the proposed design is satisfactory and complies with the provisions of Chapter 12 "Minimum Design Loads" of the Standard Building Code of SBCCI, and that the material, method or work offered is, for the purpose intended, at least the equivalent of that prescribed in the code in quality, strength, effectiveness, fire-resistance, durability and safety. The Building Inspector shall require that sufficient evidence or proof be submitted to substantiate any claims that may be made regarding its use. If, in the opinion of the Building Inspector, the evidence and proof are not

sufficient to justify approval, the applicant may appeal the decision of the Building Inspector to the Zoning Board. For the purpose of securing for the public the benefits of new developments in the building industry and yet insuring public safety, the Building Inspector shall make or cause to be made investigations or may accept duly authenticated reports from recognized sources of new materials or modes of construction, intended for use in the construction of buildings or structures in the municipality or the territory covered by this code which are not provided for in this code and shall promulgate rules setting forth the conditions under which such materials or modes of construction may be used.

5.2.1.4 Tests

The Building Inspector may require tests or test reports as proof of compliance. Tests, if required, are to be made at the expense of the owner or his agent by an approved testing laboratory or other approved agency. Copies of such test reports or the results of all such tests shall be kept on file in the office of the Building Inspector.

5.2.2 National Electric Code

- 1. All electrical construction, materials and appliances used in connection with electrical work, and the operation of all electrical apparatus within the City shall conform to the rules and requirements of the National Fire Protection Association as set forth in the National Electrical Code, 1984 Edition, or as amended.
- 2. All wiring done by any person on or outside of any building or other structure within the corporate limits of the City shall be in full compliance with the provisions and requirements of the National Electrical Code, as adopted and amended.

5.2.3 Grading Supplement

The following grading criteria will be required for all structures and where there is a conflict between any requirements set forth herein and the Southern Building Code Congress International Incorporated Standard Excavation and Grading Code, as amended, the most restrictive shall govern:

1. <u>Slab Foundation</u> - The top of the slab shall be not less than twelve (12) inches above the crown of a paved street, where the crown or center line of street is elevation plus eleven (+11) feet or higher. This requirement may be waived for good cause by the Building Inspector; provided that no waiver granted herein may lessen any applicable elevation requirements of the FEMA.

- 2. Where the crown of the street is below elevation plus eleven (+11) feet, the top of the finished floor shall be not less than elevation plus twelve (+12) feet or current applicable FEMA requirements.
- 3. Lot Grade All filling required to meet the above shall be approved material. Lots shall be graded as follows: from the curb grade to the property line, slope upwards 1/2 inch per foot of width of sidewalk to a maximum of six (6) inches. No more than twenty-four (24) inches of fill material shall be used to raise the existing ground elevation under the roof shed area of the structure. When the fill required under a structure is greater than twenty-four (24) inches, the developer shall use retainer or pier methods of construction. At completion, a final certificate from a Louisiana Licensed Civil Engineer shall be furnished by the owner stating that the lot was graded according to the CLURO, before occupancy is granted.
- 4. Beginning at the property line the lot shall slope upwards not less than one (1) inch in twenty (20) feet away from the receiving drainage facility unless an alternate slope is approved in writing by the Building Inspector. Drainage plans are required. Such drainage to be submitted by a Louisiana Licensed Land Surveyor or Civil Engineer that states the drainage meets all state and local codes not to inundate adjacent property.
- 5. In filling lots, sufficient additional material should be furnished to allow for settlement dependent on the character of material used for fill.
- 6. In case a lot or part of a lot is higher than the adjoining lot, a drainage plan shall be submitted to and approved by the Director of Public Works for the purpose of compliance with overall area drainage.
- 7. No grading shall be done nor fill placed or removed in areas under the jurisdiction of any state or federal authority, such as but not limited to the state or local coastal management authority or U. S. Army Corps of Engineers pursuant to Section 10 of the Rivers and Harbors Act or Section 404 of the Clean Waters Act, or any other provisions under the jurisdiction of these or other state or federal authority, except in accordance with the requirements of all regulations of the City and only after the required permits have been issued by all the appropriate jurisdictional authorities.
- 8. The owner shall be required to furnish a drainage plan showing the location of the proposed building, existing and proposed ground elevations and contours of not greater than one foot, property line, street name, lot number and setback of structure from all property lines.

9. A final Certificate of Occupancy shall not be issued until the provisions of this section have been met.

5.2.4 Plumbing Supplement

The following plumbing criteria will be required by one and two-family dwellings and where there is a conflict between any requirement set forth herein and the Southern Building Code Congress International Standard Plumbing Code. The use or installation of leaded pipe of any kind is strictly prohibited.

- 1. All water closets shall be on a four-inch waste in the slab.
- 2. All sewer lines shall be four-inch ABS, PVC, or ductile iron.
- 3. All water closets shall have a three-inch vent through the roof.
- 4. No fixtures shall be vented in front of any windows.
- 5. All revents shall be increased through the roof.
- 6. No 1/2-inch vents or wastes shall have more than 1/2-inch P trap.
- 7. All fixtures shall be vented through the roof.
- 8. No fixtures on the second floor shall be wasted over any fixtures except water closets.
- 9. All revents shall be carried one foot above the fixtures.
- 10. All arms for sink, lavatory, laundry tray, bar sink, etc., shall not exceed thirty (30") inches.
- 11. One four-inch clean out shall be required on a four-inch trunk line extended to the outside of the slab.
- 12. No sanitary tee shall be used on the flat in the slab.
- 13. No more than one fixture shall be on any arm, unless back to back with a twin ninety-degree ell.
- 14. No sewage shall be discharged in any place except into the City's sewerage lines.
- 15. All air conditioning P traps shall have tight connections and also be vented.

- 16. All restaurants, diners or any other place where food is served shall have the required grease traps.
- 17. Water service from the meter to the house shall be of copper or polyethylene tubing.
- 18. All water supply lines to house shall have a stop and waste.
- 19. All hot water heaters shall have a stop on the cold water line not more than three (3) feet from the heater.
- 20. All hot water relief valves must have a drain pipe to the outside of the building.
- 21. All copper tubing shall be covered when it comes in contact with steel mesh or steel rods.
- 22. All fixtures shall have air chambers.
- 23. No slab shall be poured up until the plumbing inspector inspects under ground following written request for said inspection.

5.2.5 Sewer and Water Supplement

5.2.5.1 Required Connection to City Water Service

The drilling, maintenance and use of private water wells for domestic use and human consumption within the City of Mandeville is hereby prohibited, except as provided in Chapter 17 of the Code of Ordinances.

5.2.5.2 Required Use of Public Sewers

It is unlawful for any person to cause to be deposited in an unsanitary manner, upon public or private property, any human or animal excrement, garbage, or other objectionable waste matter.

All buildings permitted under the provisions of Article 5 shall be required to install building sewers in accordance with the provisions of Chapter 17 of the Code of Ordinances.

5.2.5.3 Application for Building Sewer Permits and Connections

It is unlawful to uncover, make any connection with or opening into, use, alter, or disturb any public sewer or appurtenance thereof without first obtaining a written permit from the sewer superintendent, as provided in Chapter 17 of the Code of Ordinances of the City.

5.2.5.4 Sewer and Water Impact Fees

A. Sewer Impact Fees

Sewer impact fees shall be paid prior to the issuance of a building permit, in accordance with the following schedule:

Residential Impact Fee \$900.00 per residential unit

Non-residential Impact Fee \$.02 per square foot of lot or parcel of

Property affected including all areas for setbacks, green space,

parking lots

All additions to existing properties will be reviewed for sewer impact and will be assessed fees according to the above schedule.

B. Water Impact Fees

Water impact fees shall be paid prior to the issuance of a building permit, in accordance with the following schedule:

Residential Impact Fee \$1,100.00 per residential unit

Non-residential Impact Fee \$.02 per square foot of lot or parcel of

Property affected including all areas for setbacks, green space,

parking lots

The City may waive the requirement for collection of sewer and water impact fees for all existing residences or residences permitted for construction as of August 11, 1988 connecting to the newly extended sewer lines constructed under Phase I of the Sewer Improvement Program begun in 1987,

5.2.5.5 Water and Sewer Connections Fees

The following fees for connection to City water and sewer services shall be paid at the time of application for a building permit associated with such connections. These fees shall be paid in accordance with the following schedule.

1. Water Connections Fees

a.	3/4" tap	\$315.00
b.	1" tap	420.00
	1.5" tap	
	2" tap	
	Contractors deposit	
	(Maximum \$90.00 refundable based on use)	
f.	Water connection inspection fee	25.00

2. <u>Sewer Connection Fees</u>

a.	4" tap	\$300.00
	6" tap	
c.	Sewer connection inspection fee	25.00

The City may reduce the normal connection fee charged by the City from \$300 to \$125 for residences mentioned in Section 17-79(a). Owner must bear the cost of extending the sewer line to connect to the City's clean-out at the property line.

5.2.6 Storm Drainage Supplement

No person, firm, corporation or other legal entity shall place or cause to be placed in any public subsurface drain, swale drainage ditch, or other drainage way, any material of any form, type or nature the placement of which would alter, impede, block or otherwise detrimentally affect the rate of flow of water through said drainage way except in conjunction with the issuance of a permit by the City in conjunction with the submission by the person or entity of a drainage plan, as defined in Article 3, which meets or exceeds the minimum requirements of all provisions of this Land Use Regulations Ordinance, and all requirements of any state or federal agencies having jurisdiction over such drainageway. Whoever is found guilty of obstructing a drainageway or of placement of non-approved culverts as described herein shall be subject to the enforcement provisions of Article 1 Section 1.9 of this Land Use Regulations Ordinance.

5.2.6.1 Installation of Culverts

- 1. Any person who desires to install or have installed culverts in any public drainageways or servitudes of drain shall make application to the City of Mandeville for permission to install such culverts.
- 2. The Public Works Director shall determine the appropriate culvert size which may be installed by the applicant. The culvert shall be installed in accordance with the material specifications, depth and grade requirements established by the Public Works Director.

- 3. Culvert installation for both residential and commercial construction shall be the responsibility of the owner and performed by his/her representative. At completion of the permitted project, a final certificate from a Louisiana Licensed Civil Engineer is required stating the installation of the culvert(s) was performed not to impede upstream or downstream drainage before occupancy is granted. The certificate shall indicate msl inverts for the installed culvert(s) and its location on the property.
- 4. Installation shall be completed within sixty (60) days of City's approval of the culvert installation or the approval shall be null and void.
- 5. No person shall construct any improvement on, across, over or through a public drainage way or public servitude of drain, or any part or parts thereof, unless and until the proposed construction is approved by the Building Inspector or his designee after review by the Public Works Director or City Engineer.
- 6. In all instances in which concrete parking or access facilities (such as, by way of illustration only, parking spaces, walkways, or driveways) are constructed over, across, on or through a public drainage way or public servitude of drain, said construction shall include the provision of construction or expansion joints two and one-half (2.5) feet on each side of the center line of any culverts underlying said construction. In all cases in which there are no culverts either underlying the proposed construction or to be installed in connection therewith, said construction or expansion joints shall be located within one (1) foot of the boundaries of said public drainage way or servitude of drain. The provisions of this subsection (6) shall not apply to constructions comprised entirely of earthen, gravel or shell materials.
- 7. The owner of the property accessed across culverts installed in public rights-of-way shall be required to maintain said culverts in good working order, clear of obstruction and shall be required to replace culverts damaged or deteriorated to the point of requiring replacement as determined by the Director of Public Works.
- 8. Culverts shall be limited in subdivisions designed with "open ditches" to ensure proper surface drainage. No more than forty (40) feet of culverting shall be allowed for each lot of record without the express written authorization of the Public Works Director of his designee.

5.3 <u>GENERAL PERMITTING AND INSPECTION REQUIREMENTS</u>

5.3.1 When a Building Permit is Required

Any owner, authorized agent or contractor who desires to develop land, construct, enlarge, alter, repair, move, or demolish a structure must apply for a development permit. A permit is also necessary to install, construct, enlarge, alter, repair or move parking or driveway areas, air conditioning and ventilating systems, electrical systems, plumbing and gas piping systems, sprinkler and fire extinguishing systems, elevators, signs, incinerators, furnaces and boilers or any and all appurtenances which are regulated by this Ordinance. Any person who causes any such work or installation to be done shall first make application to the Building Inspector and obtain the required permit for the project prior to beginning of any work other than preliminary site investigation that does not disturb regulated trees.

5.3.2 Special Approvals Required Prior to Permits

Permit applications for uses requiring special approvals in conjunction with Administrative Permits, Zoning Permits, Conditional Use Permits shall not be approved until the required special approvals have been obtained.

5.3.3 General Procedure for All Permit Applications

Each application for a permit along with the required fees shall be filed with the Building Inspector on a form furnished by him, and shall contain a description of the proposed work and its location. The application shall be signed by the owner of the property on which the work is proposed or his authorized agent. Each application for a permit shall indicate the proposed occupancy of all parts of the building and of that portion of the site or lot, if any, not covered by the building or structure and shall contain such other information as may be required by the Building Inspector.

5.3.4 General Requirements for Development Permit Applications

Plans and specifications sufficient to fully describe the work proposed in conjunction with an application for a development or other permit shall be required to be submitted for approval prior to the issuance of a development or other permit. A building permit shall not be issued unless these plan requirements are met.

- 1. The required plans and specifications shall include the location of the proposed structure and any existing structure on the property.
- 2. The site plan shall show the parking layout, including driveways and other vehicular use areas, subsurface and surface drainage facilities, finished floor grades and existing sanitary sewer and water and gas mains and all other information required by the provisions of Article 9.

- 3. A boundary survey, prepared by a licensed surveyor, may also be required, as well as street center elevation and existing and proposed elevations of the perimeter of the structure.
- 4. If planting is required as provided in Article 9 of these Regulations, the survey or site plan shall show the general location of all existing vegetation and the location of existing trees to be preserved and their size in diameter breast height (dbh). If planting is required, a detailed landscape plan as required by the provisions of Article 9 shall be submitted with specifications designating by name, size and location the plant material to be installed and/or the existing trees and plant material to be preserved in accordance with these regulations.
- 5. Computations, strain and stress diagrams and other data prepared by a registered architect or a professional engineer necessary to show the correctness of the plans shall accompany the plans and specifications when required by the Building Inspector.
- 6. Every reference to these regulations in specifications or on drawings shall be to the Article or sub-article applicable to the material to be used or to the method of construction proposed.
- 7. Upon receipt of the required permit fees and a complete application, plans and specifications filed by an applicant for a permit, the Building Inspector shall review and call upon the required individuals to review the application as soon as possible. When such plans and specifications are found to be in conformity with the requirements of this Article and all other applicable laws or ordinances, the Building Inspector shall issue a permit for the specified construction.
- 8. Design Civil Engineer/Architect/Landscape Architect who seals the plans for the site design (including drainage) shall certify at the end of construction that the project was built in conformance with the approved site plan, and complies with the provisions of the CLURO. A Certificate of Occupancy shall not be issued prior to the plan being certified.
- 9. Design Guidelines Applicability: Design Guidelines shall apply to the new building construction, exterior renovations and modifications which require a building permit in B-1, B-2, B-3, B-4, I, O/R, PM-1, and PM-2 zoning districts.

5.3.5 Waiving of Requirement for Submittal of Plans

If, in the opinion of the Building Inspector, the character of the work is sufficiently described in the application, he may waive the filing of plans provided the cost of such work does not exceed five thousand (\$5,000) dollars. The whole building or group comprising a unified

development plan shall be considered one project and the work cannot be broken into units under five thousand (\$5,000) dollars to have each considered as a separate project to evade the provisions of this code.

5.3.6 Amendment of Permit Documents

It shall be unlawful to erase, materially alter or modify any lines, figures, letters, works or coloring contained upon any such drawings, specifications or computations filed with or stamped by the Building Inspector. If during the process of the execution of such work it is desired to deviate in any manner from the terms of the applications, plans or specifications or statement of cost of work, which deviation would affect the construction or other essentials of the buildings, notice of such intention to alter or deviate shall be given in writing to the Building Inspector, and his written assent shall be obtained before such alterations or deviations may be made. If such change or deviation affects the construction of structural parts of the building or structure or its classification or its grade of occupancy, new plans of the structure shall be submitted to the Building Inspector for approval and, if necessary, an additional permit shall be secured.

5.3.7 Conditions Attached to a Building Permit

- 1. No permit of any kind shall be issued for any other related work involving new construction, additions or relocated buildings or structures until the original building permit is issued.
- 2. The Building Inspector shall act upon an application for a permit with plans as filed, or as amended, without unreasonable or unnecessary delay. A permit issued shall be construed to be a license to proceed with the work and shall not be construed as authority to violate, cancel, alter or set aside any of the provisions of these regulations, nor shall a permit prevent the Building Inspector from requiring correction of errors in plans or in construction or violations of this code. The issuance of a permit should in no way be construed as a guarantee, warranty or otherwise that said construction, alterations, etc., will be or have been performed in conformance with the ordinances and regulations of the City of Mandeville, and no liability shall rest with the City to any applicant for said permit, owner, contractor or otherwise.

5.3.8 Abandonment and Extension of Building Permits

If construction has not commenced, an application for a permit for any proposed work shall be deemed to have been abandoned ninety (90) days after issuance of the permit unless work under the permit has commenced or a written extension of thirty (30) days is obtained from the Building Inspector before the expiration of the 90 days following issuance of the

permit. Extensions shall only be granted for good cause shown and no more than two (2) extensions shall be permitted. A permit shall become null and void twelve (12) months from the date of issuance unless substantial progress, as defined by this Article, has been made on the permitted work. Building permits can be canceled and refunds of application fees will be made if no work has been done on the property for which a permit has been requested, subject to a penalty and inspection fee of twenty-five (\$25) dollars. For long term projects, the Building Inspector shall define what progress should be made within a twelve month period to constitute "substantial progress", to be attached as a condition of permit approval. For short term projects expected to take less than twelve months to complete, permit expiration periods less than twelve (12) months in duration shall be defined according to and agreement with the applicant. If the drawings and specifications submitted with a permit application do not conform to the requirement of these regulations, the Building Inspector shall not issue a permit but shall return the drawings to the applicant with his refusal to issue such permit. If requested within fifteen (15) days of mailing of notice of refusal, the refusal shall be in writing, containing the reasons for refusal of the permit.

5.3.9 Adequate Waste Disposal for each Building Site During Construction

Adequate waste disposal must be provided for each building site starting with the commencement of construction, as determined by the Building Inspector.

It shall be unlawful for any person to load or unload, pile or stack, any wood, logs and any building material upon the shoulder of any part of a City street or Right-of-Way, thereby

obstructing in any manner the normal drainage or use of that street, without first receiving permission from the City's Planning Department.

5.3.10 Action on Application for a Building Permit

A permit shall be issued if the Building Inspector is satisfied that the work described in the permit application and the drawings and specifications filed with the application conform to the requirements of these regulations and other pertinent laws and ordinances. When the Building Inspector issues a permit, all parties required to approve the permit shall endorse in writing or stamp the set of plans "Approved."

5.3.11 Inspections - General

Inspections required under the provisions of this Article shall be made by the Building Inspector or a duly appointed assistant. The Building Inspector may accept reports of inspectors of recognized inspection services, after investigation of their qualifications and reliability. No certificate called for by any provision of this Article shall be issued on such reports unless the same are in writing and certified to by an authorized representative of such service.

5.3.12 Inspections Required

The Building Inspector shall inspect or cause to be inspected at various intervals all construction or work for which a permit is required and a final inspection shall be made of every building or structure upon completion, prior to the issuance of the Certificate of Occupancy, as required in this Article.

5.3.13 Minimum Inspections Required

The permit holder shall be required to notify the Building Inspector or request the required inspection at least twenty-four (24) hours in advance. (Holidays and weekends shall not be included in the 24 hour period.) The Building Inspector upon notification from the permit holder or his agent shall make the following inspections when applicable to the type of permit issued. and shall either approve that portion of the construction as completed or shall notify the permit holder or his agent wherein the same fails to comply with the law:

- 1. <u>Site Clearing Inspection</u> Prior to issuance of a Clearing Permit or Development Permit to inspect trees required to be preserved and required tree barriers.
- 2. <u>Foundation Inspections</u> When the foundation and basic plumbing is in place, and before a slab type foundation may be poured.
- 3. <u>Rough-In Inspection</u> When framing is complete and prior to closing walls.
- 4. <u>Sub Inspections</u> When plumbing extensions, electrical wiring, heating and cooling and other mechanical systems are roughed in.
- 5. <u>Paving Inspection</u> When forms are in place and prior to pouring vehicular use areas and flatwork (not required for single and two family residential).
- 6. <u>Landscaping Inspection</u> When required landscaping is installed (not required for single and two family residential).
- 7. <u>Final Inspection by State Fire Marshall</u> as required.
- 8. <u>Final Inspection</u> Completion of the structure and all work including electrical, plumbing, air-conditioning and heating is required prior to final inspection.

Final Inspection and approval of all work by the Building Inspector is required prior to the issuance of a Certificate of Occupancy or occupancy of the structure inspected.

5.3.14 Inspection and Approval

No work shall be done on any part of a building or structure beyond the point indicated in each successive inspection without first obtaining the approval of the Building Inspector. Such approval shall be recorded on the inspection card only after an inspection shall have been made of each successive step in the construction as indicated by each of the foregoing inspections.

5.3.15 Prohibition of Concealing in any Manner

No reinforcing steel or structural frame work of any part of any building or structure shall be covered or concealed in any manner whatsoever without first obtaining the approval of the Building Inspector.

5.3.16 Approval to Apply Plaster

In all buildings where plaster is used for fire protection purposes, the permit holder or his agent shall notify the Building Inspector after all lathing and backing is in place. No plaster shall be applied until the approval of the Building Inspector has been received.

5.4 <u>PERMITTING REQUIREMENTS BY PERMIT TYPE</u>

There are nine (9) types of permits required to be issued for work covered by the provisions of this Land Use Regulations Ordinance as follows:

- 1. Development Permit
- 2. Electrical Permit
- 3. Plumbing Permit
- 4. Mechanical Permit
- 5. Public Improvement Permit (Culverts, sidewalks, sewer and water

extension)

- 6. Sign Permit
- 7. Demolition Permit
- 8. Moving Permit
- 9. Special Permit

5.4.1 Development Permits

A development permit covers any man-made alterations to real estate or land and includes the following elements which are required to be addressed in the permit documents if applicable to the proposed development: (a) Clearing, (b) Grading and Paving, (c) Landscaping, (d) Building (and Building Renovations), and (e) Accessory Structures. One development permit may be issued which includes all elements of development required to be permitted under the category of development permit, except for electrical, plumbing and mechanical work which requires a separate permit, or a separate permit may be obtained

for each element of development requiring permitting. When the application for a development permit includes one or more of the types of development required to be separately permitted as listed below, all of the submittal documents and fees required for each of the types of development included and all inspections required for each of the individual development types will be included under the one development permit except for plumbing, electrical and mechanical work which shall be issued a separate permit.

If work is proposed to be done which was not included in an overall development permit or if, subsequent to the development of the site, additional work is proposed, a separate permit shall be required for each type of development work proposed. Clearing, Grading and Paving, Landscaping and Residential Accessory and Renovation permits shall not be issued except in conjunction with the issuance of a Building Permit for a principal structure or subsequent to the issuance of a permit for construction of the principal structure on the site, except in the case of a permit to construct a commercial or public parking lot on a site when such parking lot is to be the principal use of the site. Development permits for major renovations and non-residential accessory structures are required to meet all the applicable requirements of a building permit for new construction. Major renovations are renovations where the construction costs exceed five thousand (\$5,000) dollars.

Development permits for the construction of buildings fall within several Building Permit categories as follows:

Building Permits

- 1. Single-Family and Two-family Residential
- 2. Multi-Family Residential
- 3. Commercial and Industrial (including structures accessory to principal commercial and industrial buildings)
- 4. Combined Uses
- 5. Major Renovations and Non-Residential Accessory Structures

Development permits for the construction of residential accessory structures and minor renovation are issued as a separate permit type.

5.4.2 Development Permit Application Requirements for Single Family Residences and Two-Family Residences

The following items must accompany any application for a permit for the construction of single family or two family residences or major renovations of such developments:

- 1. Completed Permit Form
- 2. <u>Survey</u> one (1) certified copy signed by a Registered Land Surveyor which depicts the dimensions of the site and identifies any utility servitudes or

- rights-of-way that may effect the building setback requirement. Any existing structures must be shown on the survey.
- 3. <u>Tree Preservation Plan</u> a depiction of the site identifying the location of all trees proposed and/or required to be preserved under the provisions of Article 9, infra.
- 4. <u>Site Plan</u> Fully dimensioned including all building setbacks, required parking spaces, and existing structures.
- 5. Complete set of building plans. This includes, but is not limited to:
 - a. Floor plans (square footage must be noted)
 - b. Foundation plans
 - c. Roofing plan
 - d. Elevations of all sides
 - e. All details required to fully describe construction
- 6. Drainage Plan
- 7. Any other information determined to be required by the Building Inspector, Public Works Director, or Planning Director in order to document compliance with the requirements of these regulations for the issuance of the requested permit.
- 8. For construction in FEMA Flood Zones A or V, a flood elevation certificate is required before authorization for the provision of permanent electrical service will be issued for building.
- 9. Permit and inspection fees as required.
- 10. Plumbing, electrical and mechanical plans shall be submitted and shall be sufficient to describe all work. A separate permit shall be required for plumbing, electrical and mechanical work.
- 5.4.3 Development Permit Application Requirements for Non-Residential and Multi-Family Development

The following items shall accompany any application for a permit for multi-family or non-residential construction such as commercial or industrial development,, combined use developments and any major renovation to these facilities.

- 1. Completed Permit Form
- 2. If the application requests a clearing permit a plan showing all trees to be preserved on the site (See Article 9).
- 3. <u>Landscape Plan</u> A landscape plan is required if existing trees to be preserved are not sufficient to meet the requirements of these regulations or if any additional planting is required (see Article 9). Tree protection barriers protecting existing trees to be preserved must be in place prior to the issuance of the clearing or building permit, if clearing is done in conjunction with the construction of a building.
- 4. Fire Marshal Approval (of building plans).
- 5. Grading and Paving Plan
- 6. Boundary Survey of Site
- 7. <u>Site Plan</u> Site plan shall be fully dimensioned and drawn to scale showing all applicable items as required (see Article 9 for parking and landscape requirements, Article 7 for district yard requirements and Article 8 for special use requirements when applicable).
- 8. Complete set of Building Plans (approved by Fire Marshall) including:
 - a. Foundation plan
 - b. Floor plans (square footage of area by use must be noted)
 - c. Elevations of each side
 - d. Details and sections to fully describe construction of building.
 - e. Specifications for construction
- 9. <u>Drainage Plan</u> Drainage plans shall provide finished floor elevations, existing and proposed site elevations, defining direction of surface and sub-surface run-off, fully describing all proposed sub-surface drainage features and including any additional information required by the Public Works Director or the City Engineer. The drainage plan shall be approved by the Director of Public Works or a duly authorized designee. All development(s) over one (1)

- acre shall require a drainage plan and hydrologic report showing predevelopment and post-development watershed calculations.
- 10. <u>Utilities Plan</u> Utility plans shall fully describe water and sewer services to the building site and connections to City and other utility services. The utility plan shall be submitted for the approval of the Director of Public Works or an authorized designee.
- 11. <u>Fees</u> All applicable fees shall be paid prior to the issuance of the permit as provided in this Article.
- 12. Any other information determined to be required by the Building Inspector, Director of Public Works, City Planner or the City Engineer in order to document compliance with the requirements of these regulations for the issuance of a permit.
- 13. If structure is located in a FEMA flood zone A or V, a flood certificate shall be submitted to insure proper elevation.
- 14. <u>Plumbing, Mechanical and Electrical Plans</u> Electrical, plumbing and mechanical plans shall be sufficient to describe all work and a separate permit shall be required to be obtained for plumbing, electrical and mechanical work.
- 15. <u>Sign Plan</u> A completed sign application with a signage plan designating the type and color of signage must be submitted for the entire site of all commercial developments. Individual tenant sign applications shall not be accepted until this requirement is met.
- 5.4.4 Application Requirements for Development Permits for Residential Renovations and Accessory Uses
 - 1. An application for a permit for minor residential renovations and residential accessory use on single and two family residential developments is required for the following:
 - a. Construction of accessory buildings/structures including but not limited to:
 - i. Garages and Carports
 - ii. Greenhouses
 - iii. Pools and Patios
 - iv. Tennis courts
 - v. Play Structure
 - vi. Fences (residential and non-residential)

- b. Remodeling or renovating when electrical wiring, plumbing or structural change of the building is effected. A structural change includes extending a foundation slab to build increasing square footage of living area or extending living area into previously open areas including, but not limited to:
- i. Garage enclosures
- ii. Patio or Porch enclosures
- iii. Addition to house or accessory buildings
- iv. Bay window, fireplaces or any extensions or protrusions of exterior walls
- v. Decks, wooden or other
- 2. Required permit application items for the referenced construction:
 - i. Boundary survey of the property with existing structures.
 - ii. Site plan depicting the existing and proposed structures and other site features in relation to property lines.
 - iii. Set of building plans showing proposed changes and additions to the extent necessary to fully describe the work being done and how it connects with any existing structures.
 - iv. Any other information determined to be required by the Building Inspector, Public Works Director or City Planner in order to satisfy the requirements of these regulations for the issuance of a permit.
 - v. Permit and inspection fees as required.

5.4.5 Requirements for Public Improvement Permits

A public improvements permit shall be required, except when installed by the City, prior to the installation of public improvements such as culverts, sidewalks, and extensions of water and sewer lines. Plans adequate to fully describe the public improvements work shall be submitted to the Building Inspector and reviewed by the Director of Public Works prior to the issuance of a public improvements permit.

5.5 FEES FOR PERMIT APPLICATIONS

- 5.5.1 General Structure of Fees for Development Permits for Buildings
 - 1. Building permit fees shall be based upon the actual contract price of all construction, additions, repairs and alterations or upon based on the valuation of average current building cost per square foot based on building construction type in accordance with the classification of building by construction type provisions of the SBBCI, whichever is greater.

- 2. It shall be the duty and responsibility of the Building Inspector, with the approval of the Mayor, to establish and maintain a cost formula to determine the actual, current cost of all construction work which, for simplicity and practicality shall be on a square foot basis by type of construction.
- 3. Notice of any revision of the cost formula shall be given by publication of the revision in the official journal of the City ten (10) days prior to the effective date of such revision.

5.5.2 Fees for Development Permits

Fees for all development permits shall be based on the actual construction cost or as stated above for building construction costs.

- 1. A minimum permit fee of twenty-five (\$25) dollars shall be charged for any work governed by the requirements of these regulations when the construction cost is less than five thousand (\$5,000) dollars.
- 2. For all such work costing in excess of five thousand (\$5,000) dollars, an additional fee of three (\$3) dollars per one thousand (\$1,000) dollars or fraction thereof of the construction cost over five thousand (\$5,000) dollars shall be charged.
- 3. <u>Demolition</u> The permit fee for demolition of any building or structure shall be twenty-five (\$25) dollars when such demolition cost is less than five thousand (\$5,000) dollars; for any portion of such work costing in excess of five thousand (\$5,000) dollars an additional fee of three (\$3) dollars per one thousand (\$1,000) dollars or fraction thereof shall be charged.
- 4. <u>Moving Building</u> the permit fee for moving a building or structure across or along any street or streets, or alley or alleys, shall be:

For a one-story building	\$25.00
For a 1.5 story building	
For all other buildings 2 stories or over	

- 5. In addition to the above fee for moving, a penalty of one hundred (\$100) dollars shall be charged for each day the building being moved obstructs a street or alley.
- 6. Before moving any building, the permit shall be presented to the police department for approval of the route for travel and making arrangements for a travel escort as required.

- 7. The permit fee collected will be turned over by the Building Inspector to the Finance Director.
- 5.5.3 Required Building Permit Fees for Minor Remodeling and Renovation

A permit is required for remodeling or renovating when electrical wiring, plumbing or structural change of the building is effected. A minimum fee of twenty-five (\$25) dollars shall be charged when the value of such renovation is less than five thousand (\$5000) dollars in cost of construction. The Building Inspector shall perform such inspections as he may deem necessary to assure that work under the permit is constructed in accordance with the requirements of these regulations.

5.5.4 Required Fees for Sign Permits

Required fees for sign permits shall be paid in accordance with the requirements of Article 10 Sign Code of this Ordinance.

5.5.5 Required Fees for Tree & Shrub Removal Permits For Other Than R-1, R-1X and R-2 Districts

Permit fees for landscape installations and tree and shrub removal shall be based on the actual cost of work at a ratio of three (\$3) dollars per one thousand (\$1000) dollars of work. A minimum twenty-five (\$25) dollar permit fee will be required to process applications for landscape installation, construction or tree and shrub removal. Landscaping inspections and inspection fees shall be in accordance with the provisions of Article 9.

5.5.6 Required Fee for Clearing Permit

A permit fee of fifty (\$50) dollars shall be charged for a separate clearing permit or for the inspection of site clearing work when done in conjunction with a building permit, which may only be issued in conjunction with a development permit for construction of a building.

5.6.7 Required Fees for Single and Two-Family Residential Accessory Use Permits

The permit fee for single and two-family residential accessory structure for work listed above shall be:

- 1. The minimum fee for any accessory permit shall be twenty-five (\$25) dollars for up to two thousand (\$2000) dollars in value of construction costs.
- 2. Fees for permits for accessory structures such as, but not limited to pools, fences, or masonry walls; storage buildings and detached structures; painting;

sandblasting; steam or chemical cleaning; fixed or movable marquees and awnings; costing in excess of two thousand (\$2,000) dollars, the fee shall be three dollars (\$3.00) per one thousand (\$1,000) dollars or fraction thereof.

5.5.8 Required Fees for Plumbing Permits and Inspections

Permit Fee:	
For issuing each permit	\$15.00
Plus the following when provided:	
For each plumbing fixture, floor drain or trap	
including water and drainage piping	\$ 2.50
For each house sewer	5.00
For each house sewer having to be replaced or repaired	5.00
*For each cesspool	5.00
*For each septic tank and seepage pit or drainfield	10.00
For each water heater and/or vent	2.50
For installation, alteration or repair of water piping	
and/or water treating equipment	5.00
For repair or alteration of drainage or vent piping	5.00
For vacuum breakers or backflow protective devices	
installed subsequent to the installation	
of the piping or equipment served	
One to five	2.50
Over Five, each	1.50
For installation and inspection of automatic sprinkler systems	
Each sprinkler head	50
Lacii spi ilikici ileau	

^{*}Not allowed within three hundred (300) feet of City sewer services.

5.5.9 Required Fees for Electrical Permits and Inspections

The following permit fees are required for the issuance of permits for electrical work, other than for signs as specified in Article 10, for the following:

Residential Additions and Renovations Permit	\$35.00
New Construction - Residential Permit	55.00
Commercial, New Construction or Renovation	75.00
Plus the following:	

Service:

0	-	99 amp	3.00
100	-	199 amp	4.00
200	-	399 amp	6.00

400	-	799 amp	10.00
800	-	1199 amp	15.00
1200	-	1999 amp	25.00
2000	-	4999 amp	50.00
		5000 - over 5,000 amps	100.00
Circuit	S:		
1 circuit\$3.00			
2 circuits4.5			4.50
3 circuits			
3 circu	iits		6.00

5.5.10 Required Fees for Mechanical Equipment Permits and Inspections

The following permit fees are established for permits for mechanical equipment elements:

a.	Gas or electric heat to five tons	\$14.00
b.	Gas or electric heat greater than five tons	28.00
c.	Air conditioning units to five tons	14.00
d.	Air conditioning units over 5 tons	28.00

5.5.11 Reinspection Fees

The required inspections shall be included as a part of the permit fees listed herein unless specified differently. However, if a permit holder calls for an inspection and the work inspected does not meet code or requires a second or subsequent inspection, a reinspection fee of twenty- five (\$25) dollars shall be submitted prior to reinspection.

5.5.12 Fees for Public Improvement Permits

Fees for all public improvement permits shall be based on the actual construction cost or as stated above for building construction costs.

- 1. A minimum permit fee of twenty-five (\$25) dollars shall be charged for any work governed by the requirements of these regulations when the construction cost is less than five thousand (\$5,000) dollars.
- 2. For all such work costing in excess of five thousand (\$5,000) dollars, an additional fee of three (\$3) dollars per one thousand (\$1,000) dollars or fraction thereof of the construction cost over five thousand (\$5,000) dollars shall be charged.

5.6 <u>CERTIFICATES OF OCCUPANCY PROVISIONS</u>

5.6.1 Requirement for Certificate of Occupancy

No new building shall be occupied and no change in the individual or entity occupying a building or part of a building shall occur until the Building Inspector and any other official required to inspect the building shall have conducted any required inspections and the Building Inspector has issued a Certificate of Occupancy approving occupancy of the structure.

5.6.2 Certificate of Occupancy for New Construction

Upon completion of a building hereafter erected in accordance with approved plans, and after the final inspection herein referred to and upon application therefor, the Building Inspector shall issue a Certificate of Occupancy stating the nature of the occupancy permitted, and that the proposed use is found to be in conformity with the provisions of these regulations. Within three (3) days after notification that a building or premises or part thereof is ready for occupancy or use, it shall be the duty of the Building Inspector to make a final inspection thereof and to issue a Certificate of Occupancy if the building or part thereof, development site and the proposed use thereof are found to conform with the provisions of this Ordinance; or, if such certificate is refused, to state refusal in writing with the cause. No permanent utilities may be connected until a Certificate of Occupancy is issued.

5.6.3 Temporary Certificate of Occupancy

A temporary Certificate of Occupancy may be issued for that portion of a building which may safely be occupied prior to final completion of the building or for the testing of mechanical and electrical equipment prior to occupancy. A temporary Certificate of Occupancy may be issued for a specific period of time not exceeding six (6) months and the expiration date of the certificate shall be noted clearly on the certificate.

5.6.4 Certificate of Occupancy for Existing Building

A certificate for the occupancy of an existing building shall be obtained by applying to the Building Inspector and supplying the information and data necessary to determine compliance with these regulations and with the provisions of the SBCCI for the occupancy intended. Where necessary, in the opinion of the Building Inspector, one (1) set of detailed drawings or general or specific inspections or both may be required. When upon examination and inspection and receipt of payment for cost of inspection as specified herein, it is found that the building conforms to the provisions of these regulations for such occupancy, a Certificate of Occupancy shall be issued by the Building Inspector.

5.6.5 Records to be Maintained by Building Inspector

A record of all Certificates of Occupancy, building permits, applications, sketches and plans shall be maintained in the office of the Building Inspector for a minimum of seven (7) years and maintained beyond that time for as long as space allows.

5.7 <u>CERTIFICATES OF COMPLETION</u>

Certificates of completion for the construction of signs shall be required in accordance with the provisions of Article 10.

5.8 REQUIRED LICENSING AND BONDING OF CONTRACTORS

5.8.1 Bonding of Contractors Required

Any person, firm or corporation, including subcontractors, specialty contractors or landscape contractors, who for a fixed price, fee or wage shall undertake, direct or assume charge of any work covered by these regulations when the cost exceeds five thousand (\$5,000) dollars shall first file, and annually renew with the City a certificate evidencing the existence of public liability and property damage insurance issued to such person by an insurance or surety company authorized to do business in this state in a sum of not less than three hundred thousand (\$300,000) dollars for bodily injury and not less than fifty thousand (\$50,000) dollars for damage to property in any one occurrence.

5.8.2 Licensing of Contractors

To the extent permitted by law, the Mayor and City Council shall have power and authority to require the issuance of licenses or permits to persons who desire to engage in building or construction work, including the construction, alteration, repair of demolition of buildings or other structures, and the construction of landscape improvements upon such terms and conditions and upon payment of such fee therefor as the Mayor and City Council shall deem appropriate; and it shall be unlawful for any person to engage in such work until he has obtained the license or permit as required herein.

5.8.2.1 General Contractor/Subcontractor License Requirement

Where such licensing is applicable, all subcontractors shall be licensed by the governing authority of St. Tammany Parish before engaging in work as a subcontractor within the City of Mandeville.

5.8.2.2 Sign Contractors License Requirement

1. No person shall engage in any business or activity described in Article 10 without complying with the terms of the following requirements.

- 2. Every person commercially engaged in constructing, erecting, installing, maintaining or operating outdoor advertising, advertising structures, billboards, advertising signs, painted signs on structures, signboards or similar devices, whether as a primary or incidental activity, and whether or not such person is otherwise licensed by the City, shall obtain a sign contractor's license and pay a fee of one hundred fifty (\$150) dollars for the first year and fifty (\$50) dollars annually thereafter.
- 3. In addition to the general sign contractors licensing requirement, electrical signs may only be installed by an electrician licensed by the City.
- 4. <u>Application and Issuance</u> Applications for licenses shall be made to the City Clerk, on forms to be provided by the clerk. If the application is accompanied by the fee provided in this sign code and if there is no violation of any state law or City Ordinance in the application, the license shall be issued.
- 5. Public Liability Insurance Required It shall be unlawful for any person to engage in the business of constructing, erecting, installing, maintaining or operating signs within the City, unless and until such person shall have filed with the City a certificate evidencing the existence of public liability and property damage insurance issued to such person by an insurance or surety company authorized to do business in this state in a sum of not less than three hundred thousand (\$300,000) dollars for bodily injury and not less than fifty thousand (\$50,000) dollars for damage to property in any one occurrence.

5.8.2.3 Landscape Contractors License

Landscape contractors and landscape maintenance firms shall be required to be licensed in accordance with state laws.

RESERVED PAGE

RESERVED PAGE

ARTICLE 6 - LAND USE CLASSIFICATIONS

6.1 <u>PURPOSE AND GENERAL PROVISIONS OF USE CLASSIFICATIONS</u>

The purpose of this provision is to group land uses into a limited number of use types based

on similarities and compatibility in functional characteristics and in impact on the surrounding properties, the environment and the community as a whole, thereby providing a basis for the regulation of uses in accordance with criteria which are directly relevant to the preservation of the health, safety and welfare of the public and the aesthetics of the community. All land uses in the City of Mandeville shall be classified into one of the following list of use classifications. These classifications shall apply throughout these Land Use Regulations.

6.1.1 Determination of Use Classifications

In the event of any question as to the appropriate use classification of any existing or proposed use or activity, the City Planner or Designee shall have the authority to determine the appropriate classification, subject to the right of appeal as provided elsewhere in these regulations. In making such determinations, the Zoning Board shall consider the characteristics of the particular use in question, and shall consider any functional, product, service, or physical facility requirements common with or similar to uses cited as examples of use classifications.

6.1.2 List of Determinations

The Zoning Board shall maintain a list of all determinations made pursuant to the above section and shall annually recommend to the City Council additions, deletions, or revisions

to the use classifications to reflect contemporary usage and terminology, and to minimize ambiguity.

6.1.3 Broad and Specific Use Classifications

The list of use classifications is divided into seven broad use classification categories as follows: 6.2 Residential Use Classifications, 6.3 Civic Use Classification, 6.4 Commercial Use Classifications, 6.5 Industrial Use Classifications, 6.6 Agricultural Use Classifications, 6.7 Combined Use Classifications, and 6.8 Accessory Use Classifications. Within each broad use classification category are listed the specific use classifications within the broader categories.

6.1.4 Classifications of Use Combinations - Same Site

When two or more individual establishments on the same site fall within the descriptions of different use classifications, each establishment shall be classified separately into its respective use type.

6.1.5 Classification of Use Combinations - Same Establishment

When several principal uses falling within the descriptions of different use classifications are conducted by one individual establishment, management or institution on one development site all

such principal uses shall be classified in the use type whose description most closely portrays the overall nature of the establishment.

6.1.6 Change of Use

A change of use shall occur when the nature of the use conducted on the site changes from a use described within one specific use classification to a use described within another specific use classification whether or not each of the two use classifications fall within the same or different broader use classification categories. A change of use shall occur whether or not the name or ownership of the establishment changes if the nature of the use conducted by the establishment changes from one specific use classification as described to another specific use classification as described. Furthermore, a change of use may not occur even when the occupancy of a structure changes from one establishment to another establishment if the description of the uses of both establishments falls within the description of the same specific use classification.

6.1.7 Definitions of Terms in Use Classifications

The following is a list of words and terms utilized in the classification of uses provided herein:

6.1.7.1 Adult Arcade

An establishment in which, for any form of consideration, adult material is presented to or is available for exclusive viewing by five or fewer persons.

6.1.7.2 Adult Bookstore

An establishment that possesses adult material as more than five (5%) percent of its inventory or stock-in-trade and that offers such material for sale or rental for any form of consideration.

6.1.7.3 Adult Cabaret

A nightclub, bar, restaurant, or similar establishment that regularly features, shows, presents or displays adult material on its premises.

6.1.7.4 Adult Encounter Establishment

An establishment in which for any form of consideration two or more persons may congregate, associate or consort for the purpose of engaging in an adult encounter service. This section shall not apply to any of the following:

- 1. a hotel, motel or similar establishment offering public accommodations; or
- 2. any church or any recognized and established school, as defined under the provisions of La. R.S. 14:106; or
- 3. any medical clinic, hospital, or consulting or treatment offices of any physician, psychologist, psychiatrist or similar medical practitioner licensed by the State of Louisiana; or
- 4. any athletic club, health club, holistic health care center, gymnasium, reduction salon, spa or similar establishment in which massage or similar manipulation of the human body is offered as an incidental or accessory service.

6.1.7.5 Adult Encounter Service

Any practice or activity involving the exposure, touching or stimulation of the genitals, buttocks, or female breast of any person for any form of consideration.

As used in these regulations, the term "adult encounter services" shall be applied to any activity that meets the definitions set forth in this section whether or not, as a matter of law, such activity is or could be classified as pornographic, obscene or as any other violation of law.

6.1.7.6 Adult Material

As used in these land use regulations, adult material shall consist of movies, films, motion pictures, video tapes, video discs, slides, photographs, or other medium of visual representation; or live performances, exhibitions or presentations; or books, papers, pamphlets, magazines, periodicals or publications which are characterized by an emphasis upon the depiction or description of any of the conduct or activities set forth and described in the provisions of Louisiana Revised Statutes Title 14, Chapter 1, part VI, Sub-part C, and in that portion thereof designated as sub-paragraphs (2)(b) and (3) of paragraph A of Section 106 [La. R. S. 14:106 A (2)(b), (3)], or as such provisions may hereafter be amended, reenacted or re-designated from time to time, or shall consist of any instruments, devices or paraphernalia that are designed or marketed for use in connection with any such conduct or activities.

As used in these regulations, the term "adult material" shall be applied to any material that meets the definition set forth in this section whether or not, as a matter of law, such material is or could be classified as pornographic or obscene.

6.1.7.7 Adult Theater

A theater, concert hall, auditorium, or similar establishment characterized by activities featuring adult material available for viewing by six or more persons.

6.1.7.8 Amusement Arcade

A building or part of a building in which five or more pinball machines, video games, or other similar player-operated amusement devices are maintained.

6.1.7.9 Auto Salvage Yard

The dismantling or wrecking of used motor vehicles or trailers, or the storage, sale, or dumping of dismantled or wrecked vehicles or their parts. The presence on any lot or parcel of land of two or more motor vehicles, which, for a period exceeding 30 days, have not been capable of operating

under their own power and from which parts have been or are to be removed for reuse or sale, shall constitute prima-facie evidence of an auto salvage yard.

6.1.7.10 Boarding House

Any dwelling in which more than three (3) persons but less than fifteen (15) persons are provided with sleeping facilities and where meals, served family-style without service or ordering of individual portions from a menu, or lodging and meals, are provided for compensation by previous arrangement.

6.1.7.11 Boatel

A building or group of buildings which contains living or sleeping accommodations used primarily for transient occupancy and which is immediately accessible by boat.

6.1.7.12 Car Wash

An area of land and/or a structure with machine or hand operated facilities used principally for the cleaning, washing, polishing, or waxing of light motor vehicles.

6.1.7.13 Child Care Center

A building or structure where care, protection and supervision are provided, on a regular schedule, at least twice a week to more than six children.

6.1.7.14 Clinic

A building designed and used for the medical, psychological or chiropractic diagnosis and treatment of human patients that does not include overnight care facilities.

6.1.7.15 Club or Lodge (Private)

The buildings and facilities owned and operated by a group of people organized for a common purpose to pursue common goals, interest or activities, usually characterized by certain membership qualifications, payment of fees and dues, regular meetings and a constitution and by-laws.

6.1.7.16 Conditional Use

A land use that requires Council approval by ordinance in designated zoning districts and which is subject to the requirements of all applicable City and state regulations and any conditional requirements as deemed appropriate by the City Council in approval of the use.

6.1.7.17 Congregate Housing

A building or buildings designed or used in whole or in part to provide, for compensation, the housing of persons desiring or in need of special services such as 24 hour emergency assistance. Such facilities may furnish services to their permanent residents similar to those services furnished by hotels, including accessory uses such as home health services, meals, maid and linen services, grocery and drug stores and banking services, provided such uses are located in and accessed from entirely within the facility with no direct entrance from the street nor visibility from the outside of the facility indicating the existence of these services. Congregate housing includes elderly housing.

6.1.7.18 Dockmaster Facility

An accessory structure for a marina, which may be a caretaker facility for room and board or a day use facility to monitor the use of a marina.

6.1.7.19 Dormitory

A building used as group living quarters for a student body or religious order as an accessory use for a college, university, boarding school, orphanage, convent, monastery or other similar institutional use.

6.1.7.20 Dry Cleaning Plant

An industrial establishment providing dry cleaning services in volume for commercial establishments with personal service pick-up stations or other commercial establishments requiring the dry cleaning services, such as uniform services.

6.1.7.21 Dwelling Unit

One or more rooms physically arranged so as to create an independent and separate housekeeping establishment for occupancy by one family with separate sleeping and toilet facilities and a single facility for cooking.

6.1.7.22 Elderly Housing

Congregate housing facilities occupied by persons 62 years or older or in the case of multiple occupancy by spouses, where at least one of the spouses is 62 years of age or older with the exception of handicapped adults over 18 years of age whose disability requires the amenities and services of an elderly housing facility. Nursing, rest and convalescent homes providing

personal care facilities to non-ambulatory patients are not considered congregate housing facilities.

6.1.7.23 Event Center

A public facility for performances, displays, meetings, receptions and convocations.

6.1.7.24 Funeral Home

A structure used and occupied by a professional licensed mortician for burial preparation and similar services.

6.1.7.25 Garden Center

An establishment for the retail sale of plants, plant care and maintenance products and tools.

6.1.7.26 Group Home for the Handicapped

A dwelling unit shared by four or more handicapped persons, including resident staff, who live together as a single housekeeping unit and in a long-term, family-like environment in which staff persons provide care, education, and participation in community activities for the residents with the primary goal of enabling the residents to live as independently as possible in order to reach their maximum potential. As used herein, the term "handicapped" shall mean having: 1) a physical or mental impairment that substantially limits one or more of such person's major life activities so that such person is incapable of living independently; 2) a record of having such an impairment; or 3) being regarded as having such an impairment. However, handicapped shall not include current illegal use of or addiction to a controlled substance, nor shall it include any person whose residency in the home would constitute a direct threat to the health and safety of other individuals or whose tenancy would result in physical damage to the property of others. The term "group home for the handicapped" shall not include an alcoholism or drug treatment center, work release facilities for convicts or ex-convicts, or other housing facilities serving as an alternative to incarceration.

6.1.7.27 Home Occupation

Any business or commercial activity carried out for financial gain within a dwelling unit by the occupant of the dwelling unit, provided the activity (a) is clearly incidental and subordinate to the use of the dwelling unit as a residence; (b) is carried on solely within the main dwelling and does not alter or change the exterior character or appearances of the dwelling; (c) is located in a residential district; and (d) is in conformance with the regulations for home occupations as provided.

6.1.7.28 Home Occupation Permit

A written agreement between the City and a resident who wishes to establish a home occupation at a private residence in accordance with the Special Use Criteria for Home Occupations as provided in Article 8. Said permit shall establish the extent and duration of the home occupation as well as grant permission for the proposed use.

6.1.7.29 Hospital or Sanitarium

A building for the care of mentally or physically ill or infirmed individuals in which lodging or board and lodging are provided and offered to the public for compensation and in which ingress and egress to and from all rooms is made through an inside lobby or office supervised by a person in charge at all hours.

6.1.7.30 Hotel

A building or group of buildings designed for guest rooms or dwelling units intended primarily for automobile transients, each unit having a separate entrance opening out-of-doors or into a foyer, with parking space provided on the lot for use by guests of the hotel, operation of such hotels to be supervised by a person in charge at all hours. The definition of hotel includes auto-courts, motels, motor courts, motor hotels, and motor-inns.

6.1.7.31 Institution

A building or group of buildings designed or used for non-profit, charitable or public service purposes. Institutions may provide board, lodging or health care for aged, indigent or infirmed persons, or may provide educational or religious services to the general public.

6.1.7.32 Junk

Old, dilapidated, scrap or abandoned metal, paper, building materials and equipment, bottles, glass, appliances, furniture, beds and bedding, rags, rubber, motor vehicles and parts thereof or other waste that has been abandoned from its original use and may be used again in its present or in a new form.

6.1.7.33 Junkyard

A parcel of land on which junk is collected, stored, salvaged, or sold.

6.1.7.34 Land Use, Commercial

An occupation, employment, or enterprise that is carried on for profit by the owner, lessee, or licensee as opposed to the public land use and defined in Article 3 herein.

6.1.7.35 Light Vehicles

Vehicles weighing no more than a standard pick-up truck.

6.1.7.36 Lodge, Fraternal Order or Club

A hall or meeting place of a local branch or the members composing such a branch of a fraternal order or society, such as the Lions, Masons, Knights of Columbus, Moose, American Legion or other similar organization. Serving food is permitted, and the sale of alcoholic beverages is subject to review by the Zoning Board and to all applicable federal, state and City regulations.

6.1.7.37 Manufacturing

Establishments engaged in the mechanical or chemical transformation of materials or substances into new products, including the assembling of component parts, the manufacturing of products and the blending of materials such as oils, plastics, resins or liquors.

6.1.7.38 Nursing Home (see also Institutional Housing)

A home for the aged, chronically ill or incurable persons in which three or more persons not of the same immediate family are received, kept or provided with food and shelter for compensation, not including hospitals, clinics or similar institutions that are devoted to the diagnosis and treatment of the sick or injured.

6.1.7.39 Office

A room or group of rooms used for conducting the day-to-day affairs of a business, profession, service, industry or government.

6.1.7.40 Permitted Use

A land use that is classified in a specific zoning district without conditions for review by the zoning commission or by the City Council.

6.1.7.41 Plant Nursery

Land or structures used to store flowers, shrubs and plants primarily in containers, and other gardening associated products, for sale in retail or wholesale trade.

6.1.7.42 Repair, Major

Any alterations to the primary mechanical components or body parts of automotive or mechanical equipment.

6.1.7.43 Repair, Minor

Those activities necessary for the routine maintenance of automotive or mechanical equipment.

6.1.7.44 School, Vocational-Technical, Trade or Industrial

A public or private establishment offering training to students in the skills required for the practice of trades, technical enterprises and industrial occupations.

6.1.7.45 Ship's Store

An accessory use in a marina, for the sale of articles for the operation and routine maintenance of private recreation boats.

6.1.7.46 Sorority/Fraternity Residence

A dwelling or dwelling unit maintained exclusively for sorority or fraternity members and their guests or visitors, affiliated with an academic or professional college, university or other educational institution.

6.1.7.47 Blank

6.1.7.48 Blank

6.1.7.49 Theater

A building or part of a building devoted to showing motion pictures or for dramatic, musical or other live performances before an audience.

6.1.7.50 Transit Depot

Areas designated as primary transfer points in public transit routes or reserved for hired transportation.

6.1.7.51 Wholesale Trade

Establishments that are primarily engaged in selling merchandise to retailers, to industrial, commercial, institutional or professional business users, or to other wholesalers, or acting as agents or brokers, buying merchandise for or selling merchandise to such individuals or companies.

6.2 RESIDENTIAL USE CLASSIFICATIONS

Residential use types include the occupancy of living accommodations on a wholly or primarily non-transient basis but exclude institutional living arrangements providing 24-hour skilled nursing or medical care and those providing forced residence, such as asylums and prisons.

6.2.1 Single-Family Residential

The use of a site for only one (1) dwelling unit, other than a mobile or manufactured home.

6.2.2 Cluster Residential

A residential development where a reduction in individual residential lot area and dimensional requirements does not increase the density or total number of units permitted for the tract as a whole and where the remaining land area that results from grouping the dwellings is dedicated to common or public open space or the preservation of environmentally sensitive areas.

6.2.3 Duplex Residential

The use of a site for two dwelling units, other than a mobile homes, within a single building.

6.2.4 Two-Family Residential

The use of a site for two dwelling units, other than a mobile home, each in a separate building.

6.2.5 Townhouse Residential

The use of a site for two or more townhouse dwelling units, constructed with common or abutting walls and with each unit located on a separate subdivided parcel of ground within the total development site, providing for the individual ownership of each unit and the parcel of ground upon which it rests, as well as the direct or indirect ownership by all the unit owners on a proportional, undivided basis the common areas and facilities serving all dwelling units within the townhouse group.

6.2.6 Condominium Residential

The use of a site for a building or group of buildings forming a series of attached dwelling units constructed with common or abutting walls and located on a commonly owned site, where the units are owned individually and the land, structure(s), common areas and facilities are owned directly or indirectly by all the unit owners on a proportional, undivided basis.

6.2.7 Multiple-Family Residential

The use of a site for three or more dwelling units, within one or more buildings including condominium and cluster residential.

6.2.8 Congregate Living Residential

The use of a site for three or more dwelling units within one or more buildings in conjunction with a communal kitchen/dining facility and other personal services provided for the convenience of the residents of the site only. Typical uses include congregate housing facilities such as elderly housing and group homes for handicapped and retirement communities with a communal kitchen/dining facility.

Congregate living developments may furnish services to their permanent residents similar to those services furnished by hotels, including accessory uses such as home health services, meals, maid and linen services, grocery and drug stores and banking services, provided such uses are located in and accessed from entirely within the facility with no direct entrance from the street nor visibility from the outside of the facility indicating the existence of these services.

6.2.9 Community Residential

The use of a site for the providing of residential living accommodations on a weekly or longer basis for individuals not in need of personal care facilities within a structure with one (1) communal kitchen/dining facility for groups of more than six (6) persons not defined as a family. Typical uses include fraternity or sorority houses, dormitories, residence halls, retirement homes, cooperatives, halfway houses or boarding houses.

Community Residential developments may furnish services to its permanent residents similar to those services furnished by hotels, including accessory uses such as home health services, meals, maid and linen services, grocery and drug stores and banking services, provided such uses are located in and accessed from entirely within the facility with no direct entrance from the street nor visibility from the outside of the facility indicating the existence of these services.

6.2.10 Mobile Home Residential

The use of a site for the residential occupancy of mobile home units. Typical uses include mobile home parks, mobile home subdivisions, or mobile home condominiums.

6.2.11 Accessory Residential

A residential structure, other than a mobile home or a portable building, accessory to the principal commercial, institutional or industrial use on the same site. Typical uses include living quarters for the residential manager of a warehousing facility or the caretaker responsible for 24-hour supervision of a building or principal use.

6.2.12 Boathouse Residential

Residential living unit containing a minimum of 600 square feet of living area constructed partially over the edge of land or over water within a boat slip or dock.

6.3 <u>CIVIC USE CLASSIFICATIONS</u>

Civic use types include the performance of utility, educational, recreational, cultural, protective, governmental, and other services which are strongly vested with public or social importance.

6.3.1 Administrative Services

The use of a site for offices, administrative, clerical or public contact services, together with incidental storage and maintenance of necessary vehicles. Typical uses include federal, state, parish and City offices.

6.3.2 Cemetery

Land used or intended to be used for the burial of the dead and dedicated for cemetery purposes including columbariums, crematoriums, mausoleums and mortuaries when operated in conjunction with and within the boundaries of such cemetery.

6.3.3 Club or Lodge

A use providing meeting, recreational, or social facilities for a private or non-profit association, primarily for use by members and guests. Typical uses include private social clubs and fraternal organizations but exclude any residential uses or sale of goods or services for profit.

6.3.4 College and University Facilities

The use of land for an educational facility which offers a course of study associated with the issuance of a degree and typically including classroom and lab facilities, research services, housing facilities, parking facilities and recreational amenities.

6.3.5 Community Parking Facilities

The use of a site for publicly owned or privately owned parking services including parking garages and parking lots which serve the public at large.

6.3.6 Community Recreation

The use of a site for public or private recreational, social, or multi-purpose uses typically associated with parks, playfields, golf courses, swimming and tennis facilities and community recreation buildings or public open space.

6.3.7 Convalescent Services

A use providing bed care and in-patient services for persons requiring regular medical attention, but excluding a facility providing surgical or emergency medical services, and excluding a facility providing care for alcoholism, drug addiction, mental disease, or communicable diseases.

6.3.8 Cultural Services

A library, museum, performing arts center or similar registered non-profit organizational use performing or displaying, preserving and exhibiting objects of community and cultural interest in one or more of the arts and sciences.

6.3.9 Day Care Centers, Preschools, Nursery Schools (Public)

A state licensed and non-profit organization operating a facility providing supervisory or day care services to children or adults, excluding overnight care and public or private primary and/or secondary educational facilities. A day care center at a religious institution providing services at times other than in conjunction with the primary use (for religious purposes) of the site shall be considered a public day care center.

6.3.10 Essential Services

Uses which are necessary to support principal development and involve only minor structures such as lines and poles, phone booths, fire hydrants, bus stops, benches, mailboxes and other similar essential facilities.

6.3.11 Guidance Services

A use providing counseling, guidance, recuperative, or similar services to persons requiring rehabilitation assistance as a result of mental illness, alcoholism, detention, drug addiction, or similar condition, on a daytime care basis.

6.3.12 Group Care Facilities

The use of a site to provide living accommodations in facilities authorized, certified, or licensed by the state to provide room, board and personal care in a facility with one (1) communal kitchen/dining facility for more than six individuals not defined as a family who are unable to care for themselves due to physical disability, mental disorder or for dependent or neglected children, excluding those uses classified as Major Impact Services and Utilities. Typical uses include nursing homes, alcohol and drug rehabilitation facilities and intermediate care facilities.

6.3.13 Hospital Services (General)

A facility providing medical, psychiatric, or surgical services for sick or injured persons primarily on an in-patient basis, and including ancillary facilities for out-patient and emergency treatment, diagnostic services, training, research, administration, and services to patients, employees, or visitors.

6.3.14 Hospital Services (Limited)

A facility providing medical, psychiatric, or surgical services for sick or injured persons exclusively on an out-patient basis, including emergency treatment, diagnostic services, training, administration and services to out-patients, employees or visitors.

6.3.15 Major Impact Services and Utilities

Municipal or private services and utilities which have substantial impact. Typical uses are sanitary landfills, airports, railroad lines, detention and correctional institutes, mass transit waiting stations or turnarounds, and included spectator sports and entertainment with a capacity for 300 or more, such as large exhibition halls or sports stadiums.

6.3.16 Minor Impact Utilities

Municipal or private utilities which have a local impact on surrounding properties and are necessary to provide essential services. Typical uses are sewerage lift stations, electrical and gas distribution substations, and radio, microwave, and telephone transmitters.

6.3.17 Postal and Parcel Delivery Services

Mailing services and processing as traditionally operated or leased by the United States Postal Service and includes United Parcel Service and other similar facilities.

6.3.18 Public and Private Primary Educational Facilities Including Kindergarten, Elementary, Middle and Junior High School

A public, private or parochial school offering instruction at the kindergarten, elementary, middle, and junior high school level in the branches of learning and study required to be taught in the public schools of the state of Louisiana.

6.3.19 Public and Private Secondary Educational Facilities Including Senior High School

A public, private or parochial school offering instructions at the senior high school levels in the branches of learning and study required to be taught in the public schools of the state of Louisiana.

6.3.20 Public Recreation and Park Services

Publicly owned and operated parks, playgrounds, recreation facilities and open space.

6.3.21 Public Safety Services

The providing of public protection by a district or entity pursuant to fire, life, and safety code sections together with the incidental storage and maintenance of necessary vehicles,. typical uses include fire stations, police stations, and ambulance services.

6.3.22 Religious Assembly

Religious services involving public assembly such as customarily occur in synagogues, temples, and churches, but excluding day care facilities except when provided in conjunction with the primary use of the site.

6.4 COMMERCIAL USE CLASSIFICATIONS

Commercial use types include the sale, rental, servicing, and distribution of goods; and the provision of services other than those classified as Industrial or Civic Uses.

6.4.1 Administrative and Business Offices

Offices or private firms or organizations which are primarily used for the provision of executive, management, or administrative services. Typical uses include administrative offices, and services including real estate, insurance, property management, investment, personnel, travel, secretarial services, telephone answering, photocopy and reproduction, and business offices of public utilities, organizations and associations, or other use classifications when the service rendered is that customarily associated with administrative office services.

6.4.2 Adult Uses

Establishments in which adult materials constitute a substantial portion of the stock-intrade, entertainment or performances or adult encounter establishments.

Inclusion in these land use regulations of a classification of adult uses is intended for the purpose of reasonable and proper regulation of land uses within the City of Mandeville. The inclusion of adult uses in this classification of land uses is not intended in any way to announce, establish or describe the contemporary standards of this community regarding any specific use which is defined herein as an adult use. Nothing herein shall in any way limit or prevent the City of Mandeville or any other governmental agency or unit with appropriate jurisdiction from regulating or prohibiting any adult uses to the extent permitted by law or from enforcing any laws or ordinances relative to any conduct or activity that may be defined by these regulations as an adult use.

6.4.3 Agricultural Sales and Services

Establishments or places of business engaged in sale from the premises of feed, grain, fertilizers, pesticides and similar goods or in the provision of agriculturally related services with incidental storage on lots other than where the service is rendered. Typical uses include nurseries, hay, feed and grain stores, and tree service firms.

6.4.4 Art and Craft Studio (Limited)

A use involving the production of works of art by individuals and their assistants, not to exceed one (1) assistant per individual, and the incidental sale to consumers of these works

produced on site. Such use shall be placed and operated in such a manner that the adjoining property owners or tenants are not inconvenienced by external noise, vibration, smoke, dust, odor, heat, glare, fumes, electrical interference or waste runoff.

6.4.5 Art and Craft Studio (General)

A use involving the production of works of art by individuals and the incidental sale to consumers of those works produced, limited to the use of hand tools or domestic mechanical equipment not exceeding two horsepower or a single kiln not exceeding 8 kilowatts.

6.4.6 Animal Kennels

Boarding and care services for dogs, cats and similar small animals. Typical uses include boarding kennels, pet motels and dog training centers.

6.4.7 Animal Sales and Services (Limited)

Retail sales, veterinary services, outdoor kennels, grooming, and boarding when totally within a building, of dogs, cats, birds, fish, and similar small animals customarily used as household pets. Typical uses include pet stores, small animal clinics, dog bathing and clipping salons, and pet grooming shops but excluding outdoor kennels and uses for livestock and large animals.

6.4.8 Animal Sales and Services (General)

Veterinary services and boarding for livestock, horses or large animals. Typical uses include boarding stables for horses not owned by the occupants of the premises, and veterinary hospitals for livestock and large animals.

6.4.9 Automotive and Equipment (A&E) Services - Washing

Washing and cleaning of automobiles and related light equipment. Typical uses include auto laundries, car washes or auto detail services.

6.4.10 A & E Services - Fuel Station

Fuel dispensing stations, primarily self-serve, with lubricants and associated automotive products for self dispensing. Typical uses include self-serve stations and fuel islands in conjunction with convenience stores.

6.4.11 A & E Services - Auto Service Station

Any premises where fuel and other petroleum products are sold to light vehicles and/or light maintenance activities such as engine tuneups, lubrication, motor repairs and carburetor cleaning are conducted. Service stations shall not include premises where heavy automobile maintenance activities such as engine overhauls, automobile painting, and body/fender work are conducted.

6.4.12 A & E Services - Truck Service Station

Stations for the provision of fuel, lubricants and associated products, servicing and minor repairs, truck washing and driver related services such as food and lodging. Typical uses are truck stops.

6.4.13 A & E Services - Commercial Parking

Parking of motor vehicles on a temporary basis within a privately owned off-street parking facility, other then accessory to a principal use. Typical uses include commercial parking lots or parking garages for automobiles.

6.4.14 A & E Services -Sales and Rentals

Sale or rental of automobiles, non-commercial trucks or trailers, capable of being pulled by automobiles, motorcycles, recreational vehicles or boats, including incidental storage, maintenance and servicing. Typical uses include new and used car dealerships, motorcycle dealerships, boat, trailer and recreational vehicle dealerships, auto and trailer rental agencies and taxicab parking, dispatching and fleet storage.

6.4.15 A & E Services - Equipment Sales

Sale or rental of trucks of one (1) ton or greater capacity, tractors, construction equipment, agricultural implements, mobile homes, and similar heavy equipment, including incidental storage, maintenance, and servicing. Typical uses include truck dealerships, construction equipment dealerships, and mobile home sales establishments.

6.4.16 A & E Services - Auto and Equipment Repair (Enclosed)

Repair of automobiles, trucks, motorcycles, motor homes, recreational vehicles, or boats, including the sale, installation, and servicing of equipment and parts in an enclosed area screened from view of any adjacent streets or property. Typical uses include muffler shops, auto repair garages, tire sales and installation, wheel and brake shops, body and fender shops, and similar repair and service activities, but excluding dismantling or salvage.

6.4.17 A & E Services - Repair

Repair and storage of automobiles, motorcycles, recreational vehicles, boats, trucks, tractors, construction equipment, agricultural implements, and similar heavy equipment. Typical uses include auto repair facilities in unenclosed areas, truck repair garages, tractor and farm implement repair services, and machine shops, but exclude dismantling and salvage.

6.4.18 A & E Services - Vehicle Storage (Enclosed)

Storage of operating or non-operating vehicles, recreation vehicles and boats in an enclosed area screened from view of any adjacent streets or property. Typical uses include storage of private parking tow-away or impound yards, short term storage of towed vehicles awaiting insurance claims adjustment, storage of recreation vehicles and boats, excluding dismantling or salvage.

6.4.19 Building Maintenance Services

Establishments primarily engaged in the provision of maintenance and custodial services to firms rather than individuals. Typical uses include janitorial, landscape maintenance, or window cleaning services.

6.4.20 Business Support Services - General

Establishments or places of business primarily engaged in the sale, rental or repair of equipment and supplies used by office, professional and service establishments to business firms (not individuals), but excludes automotive, construction and farm equipment. Typical uses include office equipment and supply firms, small business machine repair shops or hotel equipment and supply firms.

6.4.21 Business Support Services - Limited

Establishments primarily engaged in the provision of services of a clerical, employment, protective, or minor processing nature to business firms rather than individuals and where the storage of goods other than samples is prohibited. Typical uses include secretarial services, telephone answering services, or blueprint services.

6.4.22 Business or Trade School

A use providing education or training in business, commerce, language, or other similar activity or occupational pursuit, and not otherwise defined as a home occupation, college or university, or public or private educational facility.

6.4.23 Commercial Recreation - Indoor Sports

Establishments or places of business primarily engaged in the provision of sports, entertainment, or recreation for participants or spectators within an enclosed building. Typical uses include athletic clubs, bowling alleys, billiard parlors, ice and roller skating rinks, amusement arcades, electronic video games and indoor racquetball courts.

6.4.24 Commercial Recreation - Outdoor Sports

Establishments or places of business primarily engaged in the provision of sports or recreation facilities in open, partially enclosed or screened facilities. Typical uses include driving ranges, miniature golf courses, golf courses, swimming pools, tennis courts, and outdoor racquetball courts.

6.4.25 Commercial Recreation - Indoor Entertainment

Predominantly spectator and participant uses conducted within an enclosed building. Typical uses include motion picture theaters, meeting halls, bingo halls, event centers and dance and reception halls.

6.4.26 Commercial Recreation - Outdoor Entertainment

Predominantly spectator uses conducted in open or partially enclosed or screened facilities. Typical uses include sports arenas, racing facilities, and amusement parks.

6.4.27 Communications Services

Establishments primarily engaged in the provision of broadcasting and other information relay services accomplished through the use of electronic and telephonic mechanisms but excludes those classified as Major Impact Utilities. Typical uses include television studios, telecommunication service centers or telegraph service offices, and film and sound recording studios.

6.4.28 Construction Sales and Services

Establishments or places of business primarily engaged in construction activities and incidental storage on lots other than construction sites as well as the retail or wholesale sale, from the premises, of materials used in the construction of buildings or other structures other than retail sale of paint, fixtures and hardware; but excludes those classified as one of the Automotive and Equipment Services use types. Typical uses include building materials stores, tool and equipment rental or sales, or building contractors.

6.4.29 Construction Sales and Services (Enclosed)

Construction sales and services uses as described herein conducted entirely within an enclosed area screened from view of any adjacent streets or property.

6.4.30 Consumer Repair Services

Establishments primarily engaged in the provision of repair services, conducted entirely within an enclosed area, for individuals and households rather than firms, but excluding Automotive and Equipment Services use types. This use classification does not include any outdoor storage of appliances or equipment. Typical uses include small appliance and washer repair shops, watch or jewelry repair shops, or musical instrument repair shops.

6.4.31 Convenience Storage

A building or group of buildings in a controlled access compound consisting of individual, small, self-contained units that are leased or owned primarily for the storage of the personal effects and household goods of individuals and for storage of materials for the operation of businesses located elsewhere, excluding materials that are inflammable or explosive or that create hazardous or commonly recognized offensive conditions. Incidental uses in convenience storage facilities may include the repair and maintenance of stored materials, excluding automobiles or other vehicles, by the tenant but in no case may storage spaces function as an independent retail, wholesale, business or service use. In addition, spaces shall not be used for workshops, hobby shops, manufacturing, or similar uses and human occupancy of said space shall be limited to that required to transport, arrange, and maintain stored material. Typical uses include mini-warehousing.

6.4.32 Day Care Centers - Commercial

Private for-profit businesses, whether licensed by the state or not to provide daytime care of children or adults, excluding overnight care and public or private primary and/or secondary educational facilities. Typical uses include child care centers.

6.4.33 Exterminating Services

Services related to the eradication and control of rodents, insects, and other pests with incidental storage on lots other than where the service is rendered.

6.4.34 Financial Services

Institutions licensed, and regulated by state or federal regulations primarily engaged in providing financial and banking services. Typical uses include banks, homesteads, savings

and loan institutions, stock and bond brokers, loan and lending activities, and similar services.

6.4.35 Food Sales

Establishments primarily engaged in the retail sale of food or household products for home consumption. Typical uses include grocery stores (including the sale of alcohol in containers for off-premise consumption) where (a) revenue from the sale of groceries comprises at least 51% of the gross income of the establishment, and (b) at least 51% of the total display or shelf space is devoted to products other than alcohol. Typical uses include grocery stores, delicatessens, meat markets, retail bakeries, and candy stores.

6.4.36 Funeral Services

Establishments engaged in undertaking services such as preparing the human dead for burial and arranging and managing funerals. Typical uses include funeral homes or mortuaries.

6.4.37 General Retail Sales (Convenience)

An establishment for the sale or rental of commonly used goods and merchandise for personal or household use, but excludes those classified more specifically herein. Typical uses include apparel stores, or establishments providing the following products or services: household cleaning and maintenance products; drugs, cards, stationery, notions, books, tobacco products, cosmetics, and specialty items; apparel, jewelry, fabrics, and like items; cameras, photography services; household electronic equipment, records, sporting equipment, kitchen utensils, small home appliances, art supplies and framing, arts, antiques, interior decorating services, office supplies; and bicycles.

6.4.38 General Retail Sales (General)

An establishment for the sale or rental of commonly used goods and merchandise for personal or household use excluding those uses specifically classified herein. Typical uses include department stores, discount stores, or establishments providing the following products or services: paint, wallpaper, carpeting and floor covering; and automotive parts and accessories, excluding service and installation.

6.4.39 General Retail Sales (Bulk)

A general retail establishment engaging primarily in the sale or rental of large and/or bulky items such as household appliances or home furnishings which require a greater square footage of retail area for display of merchandise than general retail establishment.

6.4.40 Lodging (Transient) - Bed and Breakfast Inn

A dwelling unit having no more than one (1) culinary facility where no more than six guest rooms for short-term lodging and at least one (1) meal per day are provided for compensation and where the operator of the inn is a resident on the premises.

6.4.41 Lodging (Transient) - Bed and Breakfast Residence

An owner-occupied dwelling unit having no more than one (1) culinary facility and no more than two guest rooms where short-term lodging with continental breakfast only is provided for compensation by the owner/operator of the residence.

6.4.42 Lodging (Transient) - Campground

Camping facilities providing camping or parking areas and incidental services for travelers in recreational vehicles or tents. Typical uses include recreation vehicle parks.

6.4.43 Lodging (Transient) - Hotel/Motel

A facility offering transient lodging accommodations on a daily or weekly rate to the general public with or without providing additional services, such as restaurants, meeting rooms, and recreational

facilities available to guests of the facility or the general public. Typical uses include hotels, motels, and transient boarding houses.

6.4.44 Laundry Services - Coin-Operated

Establishments primarily engaged in the provision of coin-operated laundry machines to the general public with or without the incidental provision of laundry processing services for individuals by an attendant in the facility.

6.4.45 Laundry Services - Neighborhood

Establishments primarily engaged in the provision of laundering and pressing services for individuals at the site of a laundry and dry cleaning pick-up station where dry cleaning services are provided at an off-premise dry cleaning plant.

6.4.46 Laundry Services - Commercial

Establishments primarily engaged in the provision of laundering services for commercial establishments and for off-premise laundry pick-up stations. Typical uses include bulk laundry, diaper services, or linen supply services.

6.4.47 Marine Services - Boat Fuel Area

An area for the storage and dispensing of oil and fuel for the servicing of recreational boats, which does not exceed 20,000 gallons in storage capacity and meets all state and federal safety guidelines.

6.4.48 Marine Services - Boat Sales/Service

Establishments for the sale, rental, servicing and routine maintenance, including cleaning and minor topside painting only, of recreational boats and accessories for recreational purposes. Typical uses include boat sales and service facilities.

6.4.49 Marine Services - Boat Repairs

Establishments for the repair of recreational boats including the painting of boat bottoms, entirely within an enclosed facility and including a boat lift not to exceed twenty-five (25) tons in capacity.

6.4.50 Marine Services - Commercial and Charter Fishing

Private establishments providing facilities for the docking and departure of commercial and charter fishing, shrimping or crab boats including accessory parking facilities.

6.4.51 Marine Services - Incidental Storage

Facilities, located on the same premises with marina facilities and utilizing 20% or less of the total premises, for the dry storage of recreational boats and the storage of personal effects of individuals when such facilities are provided as an incidental service to the tenants of marina boat slips.

6.4.52 Marine Services - Marinas

A boat basin or harbor for renting to the general public of recreational boat moorings and the mooring of official craft, including incidental uses such as marine sanitation devices, sanitary restrooms and dock master facilities.

6.4.53 Marine Services - Retail

Establishments for the retail sales of marine associated items including fresh seafood, bait and tackle, boat hardware and equipment, ice, soft drinks, foodstuffs and alcoholic beverages, subject to licensing requirements. Typical uses include ship's stores, chandlery, sail lofts, dockside convenience stores and retail seafood stores.

6.4.54 Marine Services - Restaurants/Transient Lodging

Establishments providing for the preparation and service of food for on-premise consumption at tables and/or transient lodging, located adjacent to, and accessible to patrons by, a navigable waterway and including the provision of temporary mooring facilities for patrons arriving by boat as well as parking facilities for patrons arriving by vehicle and including the delivery of seafood for consumption on the site by means of the navigable waterway. Typical uses include boatels and waterfront restaurants.

6.4.55 Marine Services - Yacht Clubs

A private community recreation facility accessible by water or associated with a marina facility either on the premises or in close proximity.

6.4.56 Medical Services

Establishments primarily engaged in the provision of personal health services ranging from prevention, diagnosis and treatment, or rehabilitation services provided by physicians, dentists, nurses and other health personnel, as well as the provision of medical testing and analysis services, but excluding those medical services classified as any civic or residential use. Typical uses include medical offices, dental laboratories, or health maintenance organizations.

6.4.57 Personal Improvement Services

Establishments primarily engaged in the provision of informational, instructional, personal improvements and similar services of a nonprofessional nature. Typical uses include secretarial schools, driving schools, health or physical fitness studios, reducing salons, dance studios, personal computer training, handicraft and hobby instruction.

6.4.58 Personal Services

Establishments or places of business primarily engaged in the provision of frequently or recurrently needed services of a personal nature. Typical uses include beauty and barber shops, seamstress, tailor, shoe repair shops, or dry cleaning and laundry pick-up stations.

6.4.59 Research Services

Establishments primarily engaged in research of an industrial or scientific nature which is generally provided as a service or which is conducted by and for a private firm, but excludes medical testing and analysis and product testing. Typical uses include electronics research laboratories, environmental research and development firms, or pharmaceutical research labs.

6.4.60 Restaurants - Drive-In

A retail outlet where food or beverages are sold and served, to a substantial extent, for consumption by customers in parked motor vehicles and where the food order is placed from an individual station at the parked vehicle.

6.4.61 Restaurants - Fast-Food

An establishment which offers quick food service, accomplished through a limited menu of items already prepared and held for service, or prepared, fried or griddled quickly, or heated in a device such as a microwave oven. Orders are not taken at a customers's table and food is generally served in disposable wrapping or containers. Fast food establishments may or may not deliver food or beverages to customers in motor vehicles at drive-up windows.

6.4.62 Restaurants - Outdoor Fast-Food

A fast-food restaurant establishment without indoor seating where prepared food is sold for consumption either off the premises or on the premises in outdoor seating areas provided by the establishment for the use of patrons. Typical uses include snow ball stands and sandwich shops.

6.4.63 Restaurants - Sit-Down

A business establishment whose principal business is the selling of unpackaged food to the customer in a ready-to-consume state, in individual servings, or in non-disposable containers, where the customer consumes these foods while seated at tables or counters located within the building, where alcoholic beverages may be served to dining patrons from a service bar (not accessible to patrons) and where there typically is not a rapid turn over of clientele.

6.4.64 Restaurants - Sit-Down with Lounge

A sit-down restaurant establishment utilizing up to 25% of area accessible to patrons as a lounge for the serving and consumption of alcoholic beverages.

6.4.65 Swap Meets - Enclosed

The display, exchange, barter, or sale of new or used common household items or office equipment and furnishings, entirely within an enclosed structure or screened from view of any adjacent streets or property, on a regular basis. Typical uses include flea markets where clothing, personal effects, household furnishings, and household appliances are sold or otherwise exchanged.

6.4.66 Swap Meets - Unenclosed

A swap meet conducted in an unenclosed area visible from the street right of way or adjacent property.

6.4.67 Shopping Center - Neighborhood

A multi-tenant commercial development consisting of uses permitted within the district in which it is located and a maximum of 25,000 square feet of gross leasable area overall and a maximum of 3,000 square feet of gross leasable area per individual establishment.

6.4.68 Shopping Center - Minor

A multi-tenant commercial development consisting of uses permitted within the district in which it is located and between 25,000 - 100,000 square feet of gross leasable area.

6.4.69 Shopping Center - Major

A multi-tenant commercial development consisting of uses permitted within the district in which it is located and greater than 100,000 square feet of gross leasable area.

6.4.70 Tavern - Bar or Lounge

Any premises where the principle business is the sale of alcoholic beverages at retail for consumption on the premises, where minors are excluded therefrom by law, and where incidental service of food may or may not occur, provided an excess of 25% of the patron area is used for the consumption of alcoholic beverages. Typical uses include cocktail lounges and piano bars.

6.4.71 Tavern - Night Club

A tavern in which a dance floor or dance area is provided for patron use.

6.5 INDUSTRIAL USE CLASSIFICATIONS

6.5.1 Basic Industrial

A use engaged in the basic processing and manufacturing of materials or products predominately from extracted or raw materials, or scrap and salvage operations engaged in the storage, sale, dismantling or other processing of used, source separated, or waste materials which are not intended for reuse in their original form, or a use engaged in storage of, or manufacturing processes utilizing flammable or explosive materials, or storage or manufacturing processes which potentially involve hazardous or commonly recognized offensive conditions. Typical uses include dry cleaning plants, auto salvage and junk yards.

6.5.2 Custom Manufacturing

Establishments, in an enclosed area without outside open storage of materials or equipment, primarily engaged in the on-site production of goods by hand manufacturing which involves only the use of hand tools or domestic mechanical equipment not exceeding two horsepower or a single kiln not exceeding 8 kilowatts and the incidental direct sale to consumers of only those goods produced on-site. Typical uses include ceramic studios, woodworking and candle-making shops or custom jewelry manufacturing and instructional studios for similar arts and crafts.

6.5.3 Light Manufacturing (Enclosed)

A use, conducted entirely within an enclosed area and with any outside storage of materials or equipment screened entirely from view of any adjacent streets or property, engaged in the manufacture, predominantly from previously prepared materials, of finished products or parts, including processing, fabrication, assembly, treatment, and packaging of such products, and incidental storage, sales, and distribution of such products, but excluding basic industrial uses. Typical uses include millworks and automotive body shops.

6.5.4 Light Manufacturing (Open)

Light manufacturing establishments as described under Light Manufacturing (Enclosed) which are conducted outside of fully enclosed facilities.

6.5.4(A) Old Town Manufacturing

Those light manufacturing uses similar to those listed below which do not create any danger to health and safety in surrounding areas and which do not create any offensive noise, vibration, smoke, dust, lint, odor, heat or glare, than that which is generally associated with light industries of the types specifically permitted below and which are engaged in the

manufacture, predominantly from previous prepared materials, of finished projects or parts, including processing, fabrication, assembly, treatment, and packaging of such projects, and incidental storage, sales, and distribution of such projects, but excluding basic industrial uses and automotive body shops or automotive uses. Uses shall be conducted entirely within an enclosed area. Any outside storage of materials or equipment, including vehicles shall be screened from view of any adjacent streets or property,

- 1. Manufacture of assembly and sale of medical and dental equipment, draft, optical, and musical instruments, watches, clocks, toys, games, and electrical or electronic apparatus.
- 2. Beverage blending or bottling, bakery products, candy manufacture, dairy products, and ice cream, distilling of beverages but no fruit and vegetable processing and canning, packing and processing of fish, meat, and poultry products, or slaughtering of poultry or animals.
- 3. Manufacture of furniture, boxes, cabinets, doors, windows, baskets, and other wood products of similar nature. Typical uses include millworks.
- 4. Compounding of cosmetics, toiletries, drugs and pharmaceutical products.

6.5.5 Research Services - Hazardous

Establishments engaged in research of an industrial or scientific nature which is generally provided as a service or which is conducted by and for a private firm, including medical testing and analysis, and product testing. Typical uses include bio-medical research and testing of materials that are hazardous or produce hazardous by-products.

6.5.6 Resource Extraction

A use involving the on-site extraction of surface or subsurface mineral products or natural resources. Typical extractive uses are quarries, borrow pits, sand and gravel operations, oil and gas extraction, and mining operations.

6.5.7 Wholesale, Storage and Distribution - Light

Wholesaling, storage and warehousing services within enclosed structures. Typical uses include wholesale distributors of manufactured products and food products, including seafood; business and industrial storage warehouses or moving and storage firms.

6.5.8 Wholesale, Storage and Distribution - Heavy

Open air storage, distribution and handling of materials and equipment. Typical uses include stone or prefabricated concrete yards, grain elevators, off-shore equipment storage related to the oil industry.

6.6 AGRICULTURAL USE CLASSIFICATIONS

6.6.1 Animal Production

The raising of animals or production of animal products such as eggs or dairy products, on an agricultural or commercial basis. Typical uses include raising and breeding of livestock, grazing, ranching, dairy farming and poultry farming for other than personal use.

6.6.2 Animal Waste Processing

A use primarily involved in the processing of animal waste by-products, including but not limited to animal manure, animal bedding waste, and similar by-products of an animal raising agricultural operation, for use as a commercial fertilizer or soil amendment and including composting for commercial purposes.

6.6.3 Aquaculture

Premises primarily devoted to aquacultural research and specialties and also including the raising of seafood including catfish and crawfish farms.

6.6.4 Horticulture - Cultivation

Premises primarily devoted to the cultivation of plants such as flowers, shrubs and trees intended for ornamental or landscaping purposes and sold wholesale.

6.6.5 Horticulture - Storage

Premises primarily devoted to the storage of plants such as flowers, shrubs, and trees intended for ornamental or landscaping purposes, primarily in containers and sold wholesale only.

6.6.6 Horticulture-Storage (Enclosed)

Premises primarily devoted to the storage of plants such as flowers, shrubs, and trees intended for ornamental or landscaping purposes, primarily in containers for wholesale or retail trade, with all materials and equipment, other than the plants themselves, stored in areas fully enclosed or screened from view of any adjacent street or property.

6.6.7 Packing and Processing - Limited

The packing or processing of agricultural crops, animals, and their by-products which were produced or grown on the premises and which entails more than picking, cutting, sorting, and boxing or rating, but does not include canning, rendering, tanning, or reduction of meat.

6.6.8 Packing and Processing - General

The packing or processing of agricultural crops, animals, and their by-products regardless of where they were produced or grown and which entails more than picking, cutting, sorting, and boxing or rating, but does not include canning, rendering, tanning, or reduction of meat.

6.6.9 Row and Field Crops

Premises primarily devoted to the cultivation of agricultural products grown in regular or scattered patterns such as vines, field, forage, and other plant crops intended to provide food or fibers.

6.6.10 Tree Crops

Premises primarily devoted to the cultivation for other than personal use of tree grown agricultural products such as orchards for peaches and oranges.

6.7 COMBINED USE CLASSIFICATIONS

6.7.1 Combined Uses - Residential/Office

A combination of those use classifications separately listed as permitted by right in the district and in accordance with the Special Use Criteria for Combined Uses as provided in Article 8.

6.7.2 Combined Uses - Residential/Commercial

A combination of those use classifications separately listed as permitted by right in the district and in accordance with the Special Use Criteria for Combined Uses as provided in Article 8.

6.7.3 Planned Combined Use

A combination of principal uses from the same or different broad use classifications categories located on the same site in a district planned for a combination of uses subject to a site plan review process. A typical planned combined use may be a combination of multi-

family and single-family residential uses on a single development site in the Planned Combined-Use District or a combination of several different use classification types such as a marina and a boatel or a marina and residential duplexes on the same site in a Planned Marina District.

6.8 ACCESSORY USE CLASSIFICATIONS

6.8.1 Accessory Uses (General)

Accessory uses shall be those uses incidental to, and on the same lot as, a principal use as defined in Article 3 herein under Accessory Building or Use, and subject to all regulations for accessory uses provided in these regulations or any other regulations of the City or state. Accessory uses may only occur subsequent to the commencement of occupancy of the principal use on the same site.

6.8.2 Accessory Uses (Home Occupation)

The use of a residence for a home occupation as defined, within the guidelines of the requirements for the permitting of Special Use Criteria.

6.8.3 Accessory Uses - Residential

The following activities are specifically regarded as accessory to residential principal uses and may only occur subsequent to the occupancy of the principal structure:

1. Incidental storage of household items or yard maintenance equipment owned by the occupant of the principal structure.

- 2. Offices or studios for personal use within an enclosed building and used by an occupant of a residence located on the same lot as such building to carry on personal business or artistic activities of a non-commercial nature, so long as such activities do not fall within the definition of a home occupation.
- 3. Hobbies or recreational activities of a noncommercial nature.
- 4. The renting out, on a monthly or longer basis, of one (1) room within a dwelling unit to not more than two persons who are not part of the family that resides in the dwelling unit, provided the room is not equipped with cooking facilities.
- 5. Yard sales or garage sales, so long as such sales are not conducted on the same lot for more than three (3) days (whether consecutive or not) during any 90 day period.

6.8.4 Signs Accessory Uses - On-Premise

The use of a site for the erection of any on-premise sign as defined and in accordance with the regulations for sign uses provided in Article 10 or elsewhere herein.

6.8.5 Signs Accessory Uses - Off-Premise Signs

The use of a site for the erection of any off-premise sign as defined and in accordance with the regulations for off-premise sign uses provided in Article 10 or elsewhere herein.

ARTICLE 7 - ZONING DISTRICT REGULATIONS

7.1. <u>TITLE AND PURPOSE</u>

This Article shall be known as the District Regulations. The zoning districts are created in order to promote compatible patterns of land use in pursuit of the stated purposes of this Comprehensive Land Use Regulations Ordinance and to regulate the use and development of land within each district in a manner appropriate to the purposes of each district.

7.2 <u>ADOPTION, AMENDMENT AND INTERPRETATION OF ZONING DISTRICTS</u>

7.2.1 Creation of Zoning Districts

All land within the corporate limits of the City of Mandeville is hereby divided into the following Base Zoning Districts, and portions of land within the corporate limits of the City of Mandeville which are included in the following Overlay Districts:

1. Base Zoning Districts

Map Code	<u>District Name</u>
R-1	Single-Family Residential District
R-1X	Single-Family Residential-Existing Small Lots District
R-2	Two-Family Residential District
R-3	Multi-Family Residential District
МН	Mobile Homes District
I	Institutional District
0	Open Space/Recreation District
B-1	Neighborhood Business District
B-2	Highway Business District
B-3	Old Mandeville Business District
B-4	Major Crossroads Business District

O/R Office/Residential District

PM-1 Planned Marina District 1 - Waterfront Lots

PM-2 Planned Marina District 2 - Non-waterfront Lots

PD Planned District (PRD, PCD, PID, PCUD)

M-1 Light Manufacturing District

M-2 General Manufacturing District

TC Town Center District

2. Overlay Zoning Districts

Map Code	<u>District Name</u>
D-0	Drainage Overlay District
G-0	Gateway Overlay District
L-O	Lakefront Overlay District
TCOD	Town Center Overlay District

7.2.2 Application of Base and Overlay Districts

- 1. A base district designation shall apply to each lot, site or parcel within the City. Portions of a site may be classified within different base districts, provided only one base district designation shall apply to the same portion of any site.
- 2. An overlay district designation may be applied to any lot or site or to any portion of a lot or site, in addition to its base district designation.

7.2.3 Ordinance Linkage with Official Zoning Map

The boundaries of the created zoning districts shall be established in accordance with their designated location as shown on the map entitled "Official Zoning Map of the City of Mandeville, Louisiana" dated June 10, 1993 and subsequent revisions. The official zoning map shall be identified by the signature of the Mayor attested by the City Clerk and bearing the words, "This is to certify that this is the official zoning map referenced in Article 7, Section

7.3 of the Comprehensive Land Use Regulations Ordinance of the City of Mandeville, Louisiana." This Official Zoning Map together with all legends, symbols and other information thereon shall be a part of this Comprehensive Land Use Regulations Ordinance and shall be maintained in a file in the Department of Planning and Development at City Hall and shall be in reproducible form from which prints or reproductions can be made.

7.2.4 Amendments to Official Zoning Map

Amendments to the Official Zoning Map shall be made in accordance with the procedures and requirements for amendments to the Comprehensive Land Use Regulations Ordinance as provided in these regulations. It will be the responsibility of the City Planner or his designee to update the Official Zoning Map with the adopted amendment within ten (10) working days of the effective date of the zoning district amendment and to record the number of the amending ordinance and the date of its passage on the amended Official Zoning Map. No unauthorized person may alter or modify the Official Zoning Map. The Department of Planning and Development shall keep a file of copies of all superseded versions of the official zoning map for historical reference.

7.2.5 Determination of District Boundaries

The following rules shall apply in the determination of the boundaries of the districts shown on the Official Zoning Map:

- 1. Unless otherwise indicated, the district boundaries are lot lines, the center lines of streets or roads, the rear lot lines of parcels of land fronting a street right-of-way of the same zoning district classification, or the corporate limit lines as they exist at the time of the enactment of this Ordinance or subsequent amendments.
- 2. Where the district boundaries approximately coincide with lot lines, the lot lines shall be construed to be the district boundary lines unless otherwise indicated. Where district boundaries as shown on the zoning map do not coincide with street lines, alley lines or lot lines, and no dimensions are shown, the location of said district boundaries shall be determined by use of the scale of the zoning map.
- 3. When a lot or site is divided by a district boundary, the regulations applicable within each district shall apply to each portion of the site situated in a separate district.
- 4. Map codes indicating the district shall apply to the whole of the area within the district boundaries.

- 5. Any areas under water within the corporate limits of the City which are not included within a zoning district on the Official Zoning Map shall be subject to all the regulations applying to the district adjacent to the water area. If the water area adjoins two (2) or more districts, the boundaries of each district shall be construed to extend into the water area to the centerline of the water area which will then divide the two districts.
- 6. Revocation of Public Rights-of-Way Whenever any street, alley or other public right-of-way is revoked by ordinance, the zoning district adjoining each side of such street, alley or other public way shall automatically extend to the center of same, and the land area shall become subject to the regulations of the extended districts.

7.3 DEFINITIONS APPLICABLE TO ZONING DISTRICT REGULATIONS

General definitions applicable to this Article and the articles of Division II - Building and Zoning Regulations of this Comprehensive Land Use Regulations Ordinance shall be found in Division I - Article 3, General Definitions. Definitions specific to separate articles or subsections within articles shall be found in the separate article or subsection to which they apply, including the definitions for use classifications which are to be found in Article 6, Land Use Classifications.

7.4 GENERAL PROVISIONS OF THE ZONING DISTRICT REGULATIONS

- 1. Only structures and buildings of permanent materials, finished inside and out in a good and workmanship like manner, meeting all applicable building codes, qualifying under regular construction loan financing standards, and meeting all other applicable provisions of this Land Use Regulations Ordinance shall be allowed as both principle and accessory buildings within all districts, except as otherwise provided in the MH Mobile Homes District.
- 2. No living vegetative matter shall be removed from any undeveloped site prior to, and in accordance with the provisions for, the issuance of the required clearing permit in conjunction with the issuance of a development permit, except as needed for the purposes of surveying or testing or to remove specific dead or diseased trees to protect the public health, safety or welfare. Areas cleared for surveying or testing shall be no greater than fifteen (15) feet wide and no tree three (3) inches dbh or greater shall be removed.
- 3. Areas subject to periodic inundation shall not be included in the calculations of required lot area or required buildable area for purposes of subdivision of land or the issuance of a development permit, except in the case of legally non-

conforming lots of record as provided in Article 4 or as provided in Section 7.6.1.4. In addition, when determined by the Planning Commission or Zoning Board in conjunction with a Subdivision or Resubdivision request, Zoning Permit or Conditional Use application areas identified as wetlands subject to Corps of Engineers determination may be treated as areas of periodic inundation for purposes of protecting the general health, safety and welfare of the public and adjacent property values.

- 4. For purposes of these regulations, service areas shall include areas provided for the storage and removal of solid waste.
- 2. Residential security lighting shall be designed to minimize light spill over into adjoining streets and nearby residential areas and shall be hooded or shielded so that the light shines downward and within the boundary of the premises being lighted and away from adjoining property and abutting streets in such a way as not to create a nuisance.

7.5 BASE DISTRICT REGULATIONS BY ZONING DISTRICTS

The regulations of this Article identify the purpose of each of the created zoning districts and shall be considered the minimum requirements, in addition to all other applicable regulations of this Comprehensive Land Use Regulations Ordinance or other laws of the City or state as provided, for the use and development of all land within the separate zoning districts in conjunction with the Table of Permitted Uses By Zoning District included in this Article.

7.5.1 R-1 - Single-Family Residential District

7.5.1.1 Purpose of the R-1 District

The R-1 Single-Family Residential Zoning District shall be for the purpose of providing low-density residential neighborhoods where single families occupy single-family detached dwelling units on individual lots in a healthy, safe and peaceful environment in combination with accompanying accessory uses and community oriented recreation and service facilities while being protected from the adverse impacts of incompatible land uses which belong in non-residential areas.

7.5.1.2 R-1 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.1.3 R-1 Site Development Regulations

Each development site in the R-1 Single-Family Residential Zoning District shall be subject to the following site development regulations in addition to any regulations applicable under the provisions of Article 8.

 Minimum lot area (except for legal non-conforming lots as provided) 	10,800 SF
2. Minimum building area (SF per unit)	1,200 SF
3. Minimum lot width	90'
4. Minimum lot depth	120'
5. Minimum Yard Setback Requirements	
a. Front Yardb. Both Side Yards - Totalc. Interior Side Yardd. Street Side Yarde. Rear Yard	25' 15' 5' 15' 30'
6. Maximum Height of Structures	35'
7. Maximum Impervious Site Coverage	45%

7.5.1.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance.

7.5.2 R-1X - Single-Family Existing Residential District

7.5.2.1 Purpose of the R-1X District

The R-1X Single-Family Existing Residential district shall be for the purposes of bringing predominantly single-family neighborhoods existing at the time of adoption of these regulations, which are (a) zoned for multi-family uses, or (b) zoned R-1 but in which the majority of the home sites are on legally non-conforming sub-standard lots of record, into legally conforming status while protecting the character of the neighborhood as a single family residential area and providing low to medium density neighborhoods for single-family detached dwelling units on individual lots where the predominant existing development consists of single-family residences on parcels of land less than ninety (90) feet in width in separate ownership from adjacent parcels, and which were subdivided into lots smaller than the minimum required R-1 lot size prior to the adoption of this Land Use Regulations Ordinance. In general the maximum density in R-1X districts shall be in accordance with the predominant density of the existing development in the area.

7.5.2.2 R-1X Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.2.3 R-1X Site Development Regulations

Each development site in the R-1X Single-Family Existing Residential Zoning District shall be subject to the following site development regulations in addition to any regulations applicable under the provisions of Article 8. The minimum lot area and width requirement within each specific R-1X district shall be noted on the Official Zoning District map within the specific R-1X district and determined as provided herein but under no circumstances shall the minimum lot width requirement be less than fifty (50) feet or the minimum lot area requirement be less than

5,000 square feet. For purposes of the creation of R-1X districts, the minimum lot area and widths shall be determined by the size of the predominant number of parcels of land under separate ownership from adjacent property within the area defined as the R-1X zoning district. A parcel of land under the same ownership, whether consisting of one or more lots of record, shall be considered a parcel of land in separate ownership from adjacent property owned by another individual or entity. Ownership shall be determined by tax rolls unless more recent information, recorded in the St. Tammany Parish Courthouse, is provided.

1. <u>Minimum Lot Area</u> - 5,000 SF or as noted on the Official Zoning Map for the district where the lot is located.

- 2. Minimum Building Area (SF per unit) 1,000 SF
- 3. <u>Minimum Lot Width</u> (Variable) 50' or the width of the predominant number of parcels of land under separate ownership from adjacent property within the zoning district (usually derived from the width of the existing subdivided parcels or a multiple thereof), whichever is the greater.

4. <u>Minimum Yard Setback Requirements</u>

a.	Front Yard	25'
b.	Both Side Yards - Total	15'
c.	Interior Side Yard	5'
d.	Street Side Yard	10'
e.	Rear Yard	30'
Max	ximum Height of Structures	35'
<u>IVI az</u>	Aimum Height of Structures	33

6. <u>Maximum Impervious Site Coverage</u> 50%

7.5.2.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance.

7.5.3 R-2 - Two-Family Residential District

7.5.3.1 Purpose of the R-2 Two-Family Residential District

The purpose of the R-2 Two-Family Residential District is to provide an area for moderate density single-family residential uses and duplex uses with one structure on one lot while maintaining a single-family neighborhood character.

7.5.3.2 R-2 Permitted Uses

5.

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end

of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.3.3 R-2 Site Development Regulations

Each development site in the R-2 Residential District shall be subject to the following site development regulations in addition to applicable regulations under the provisions of Article 8.

1. Minimum Lot Area	9,000 SF
2. Minimum Lot Area Per Unit	5,000 SF
3. Minimum building area (SF per unit)	1,000 SF
4. Minimum lot width	75'
5. Minimum lot depth	120'
6. Minimum Yard Setback Requirements	
a. Front Yardb. Both Side Yards - Totalc. Interior Side Yardd. Street Side Yarde. Rear Yard	25' 15' 5' 10' 30'
7. Maximum Height of Structure	35'
8. Maximum Impervious Site Coverage	50%

7.5.3.4 Parking and Landscape Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance.

7.5.4 R-3 - Multi-Family Residential District

7.5.4.1 Purpose of the R-3 Multi-Family Residential District

The purpose of the R-3 Multi-Family Residential district shall be to provide moderate to high density residential neighborhoods for individual buildings on individual lots or for more than one building on one lot. The R-3 district shall accommodate single family attached dwellings that have common walls, including town houses, condominiums, congregate and cluster developments as well as multi-family structures ranging in type from tri-plexes to apartment buildings.

7.5.4.2 R-3 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.4.3 R-3 Site Development Regulations

1. Minimum lot area

Each development site in the R-3 Multi-Family Residential District shall be subject to the following site development regulations in addition to any regulations applicable under the provisions of Article 8.

1. Filling Total ou	3,000 81
2. Minimum lot area per unit	3,000
3. Minimum building area (SF per unit)	700 square feet unless otherwise provided for in Article 8
4. Minimum lot width	75'
5. Minimum Yard Setback Requirements	
a. Front Yard	20' or required depth of greenbelt, whichever is greater
b. Combined Side Yards - Total	20'
c. Interior Side Yard	10'
d. Street Side Yard	10'
e. Rear yard	20'
f. Between Structures	20'
6. Maximum Height of Structure	35'

9,000 SF

60%

7.5.4.4 Parking and Landscaping Requirements

7. Maximum Impervious Site Coverage

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in special district criteria below or Special Use Criteria as provided in Article 8.

7.5.4.5 Special R-3 District Criteria

Special Use Criteria for individual uses permitted within this district shall be in accordance with the regulations provided in Article 8.

Parking lots shall be located in the rear of structures accessed by no more than one two-way drive or two one-way drives for every one hundred fifty (150) feet of greenbelt and out of view of the fronting street. When parking spaces are provided in conjunction with each unit, such as when garages or carports are provided for each unit instead of shared parking lots, the driveway access shall be from the rear of the units on a service access drive to minimize the number of paved accessways through the greenbelt along the street front and to screen the view of parking areas from the fronting street.

7.5.5 <u>MH - Mobile Homes District</u>

7.5.5.1 Purpose of the MH District

The purpose of the Mobile Homes District is to provide locations for development of mobile home residential parks and mobile home subdivisions, with standards that ensure a residential environment and compatibility with adjoining residential neighborhoods.

For purposes of regulations within this district the following definitions shall apply:

- 1. <u>Mobile Home Park</u> A unified development of twenty or more mobile home spaces for rent or lease, and which may include common areas and facilities for management, recreation, laundry and utility services, storage, and similar services for the convenience of residents of the mobile home park.
- 2. <u>Mobile Home Space</u> An area within a mobile home park which is designed for and designated as the location for a single mobile home and the exclusive use of its occupants.
- 3. <u>Mobile Home Stand</u> That portion of a mobile home space upon which the mobile home is placed.

7.5.5.2 MH Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.5.3 Site Development Regulations For Mobile Home Parks

Mobile home parks shall be subject to the following site development regulations:

- 1. The minimum number of mobile home spaces contained in any mobile home park shall be twenty (20) spaces, and the minimum site area shall be ninety thousand (90,000) square feet.
- 2. The minimum mobile home park site area per dwelling unit shall be forty-five hundred (4,500) square feet.
- 3. Each mobile home shall have a minimum front yard setback of twenty-five (25) feet and minimum interior side yard and rear yard of twenty (20) feet, respectively.
- 4. The mobile home park shall have direct access to a street having a dedicated and accepted right-of-way of not less than sixty (60) feet.
- 5. Interior vehicular circulation shall be provided by private internal streets paved to a width of not less than thirty (30) feet. Internal streets shall be continuous and connect with other internal streets or with public streets, or shall be provided with a paved cul-de-sac having a diameter of eighty (80) feet. No internal street ending in a cul-de-sac shall exceed four hundred (400) feet in length.
- 6. Each mobile home space shall contain minimum area of twenty-five hundred (2,500) square feet, adjacent to an internal street designed to permit movement of mobile homes to and from each space.
- 7. Each mobile home park shall have a minimum of two (2) off-street parking spaces per dwelling unit, at least one of which shall be provided on each mobile home space. The balance of the required parking may be located in

- common parking areas distributed throughout the mobile home park in a manner that provides reasonable and convenient access to all mobile home spaces.
- 8. Each mobile home and any attached accessory structures shall be separated from every other mobile home, building or structure by at least ten (10) feet.
- 9. The minimum distance between a mobile home stand and the pavement of an internal street, common parking area, or other common areas shall be ten (10) feet.
- 10. A landscaped buffer in accordance with the provisions of Article 9 including a solid wall or fence at least six (6) feet high shall be erected and thereafter properly maintained along all boundaries of the mobile home park; except,
 - a. Where the boundary of the mobile home park abuts a public right-of-way, in which case the provisions of Article 9 regarding greenbelts shall apply.
 - b. Where the boundary of the mobile home park abuts another mobile home development.
- 11. The height of the mobile home chassis above the ground elevation, measured at 90 degrees to the frame, shall not exceed three feet at the low end, and the provisions of Article 8 Flood Damage Prevention Regulations shall apply.
- 12. Required front yards shall be landscaped in accordance with the greenbelt requirements as provided in Article 9, excluding the necessary driveways and walkways providing access to the mobile home park.
- 13. Each mobile home park shall provide for reasonable and safe pedestrian access to and among each mobile home space and all common facilities. Walks not designed in common with internal streets or parking areas shall have a minimum paved width of two (2) feet.
- 14. Each mobile home park shall have a minimum of three hundred (300) square feet of open space per dwelling unit, with at least one hundred fifty (150) square feet being located on each mobile home space. The balance of the required open space may be located in common open space areas distributed throughout the mobile home park in a manner that provides reasonable and convenient access to all mobile home spaces.
- 15. Maximum height shall be thirty-five (35) feet.

7.5.5.4 Site Development Regulations for Mobile Home Subdivisions

Mobile home subdivisions designed for the placement of mobile home dwellings on individual subdivided lots with frontage on a public street shall be subject to the regulations of the R-2 Two-Family Residential District and the provisions of Division III Subdivision and Public Improvements Regulations of this Land Use Regulations Ordinance and any other applicable laws of the City, state or federal government, including but not limited to the Flood Damage Prevention Regulations of Article 8 herein.

7.5.5.5 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the special district criteria above or Special Use Criteria as provided in Article 8.

7.5.6 <u>I - Institutional District</u>

7.5.6.1 Purpose of the Institutional District

The purpose of the institutional district shall be to accommodate uses of a civic, religious, educational, institutional or public nature in areas which provide maximum accessibility for the public to utilize the facilities provided in the institutional district.

7.5.6.2 Institutional District Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.6.3 Institutional Site Development Regulations

Each development site in the Institutional District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

1. Minimum lot area 30,000 S.F.

2. Minimum lot width and depth 150'

3. Minimum Yard Setback Requirements

a. Front YardRequired depth of greenbelt,whichever is greater	25' or
b. Street Side or Rear YardRequired depth of greenbelt,whichever is greater	15' or
c. Interior Side or Rear Yard- Adjacent to Residential Districts Adjacent to Other Districts With firewall at property line	20' 5' or 0'
4. Maximum Height of Structures	35'
5. Maximum Impervious Site Coverage	75%

7.5.6.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the special district criteria below or Special Use Criteria as provided in Article 8.

7.5.6.5 Special Institutional District Criteria

- 1. <u>Access</u> Institutional districts shall be located with frontage on major arterial or collector streets or on local streets in locations which do not require travel through existing or proposed residential districts.
- 2. Special Requirements Adjacent to Residential Districts
 - a. <u>Landscape Buffer Requirements</u> A vegetative buffer, as provided in Article 9, shall be required in the twenty (20) foot required setback adjacent to all residential districts. If parking, service facilities and/or outside use areas, such as but not limited to play fields and assembly areas, for institutional uses

abut R-1, R-1X or R-2 districts, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent if reasonably necessary to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed toward the institutional property, and be shielded from and shall not adversely impact any adjacent use or traffic.

- b. <u>Hours of Operation and Noise</u> If night activities are conducted by the onsite use, such activities shall not interfere with the peace of any adjacent residential district and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.
- 3. Storage of Equipment and materials To the extent possible, materials and equipment used or employed in connection with an institutional use shall not be stored in any open area of an institutionally zoned site. In no event shall any equipment or materials be stored in any open area on any institutionally zoned site for a period of time in excess of 72 hours unless such material or equipment cannot be readily stored in an appropriate enclosure and is essential to the institutional use being conducted on the site.

7.5.7 <u>O - Open Space/Recreational District</u>

7.5.7.1 Purpose of the Open Space/Recreational District

The purpose of the open space/recreational district is to provide for open space, parks and recreational areas; for uses that are accessory thereto, as well as certain facilities that are

generally associated with recreational uses; and for such other uses as are specifically permitted in this district under the provisions of these regulations.

7.5.7.2 Open Space/Recreational District Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10. Any other proposed use of property located in this district shall be reviewed and approved in accordance with the procedures set forth in these regulations for the review and approval of site plans for the proposed use or development of property located in a Planned District Zoning District. In addition to all factors and criteria required to be considered in connection with the review of a Planned District site plan under

the provisions of these regulations, no site plan proposing additional use or uses of property located in an Open Space/Recreational District shall be approved unless the proposed use or uses as determined by the City Council to be in accordance with and to serve the stated purposes for which the Open Space/Recreational District has been established.

7.5.7.3 Open Space/Recreational Site Development Regulations

1. Minimum Open Space Lot Area

Each development site in the Open Space/Recreational District shall be subject to the following site development regulations in addition to any regulations applicable under the provisions of Article 8. For purposes of these district regulations an open space lot shall be a parcel of land devoted exclusively to providing open space areas with possible pedestrian or bicycle amenities and without any buildings. A buildable lot shall be a lot established for one of the uses permitted within the Open Space/Recreational District.

2.000 SF

75%

2. Minimum Buildable Lot Area	10,000 SF
3. Minimum Buildable Lot Width and Depth	100'
4. Minimum Yard Setback Requirements (For	Buildings)
a. Front Yard Required depth of greenbelt, whichever is greater	25' or
b. Street Side or Rear Yard Required depth of greenbelt, whichever is greater	15' or
c. Interior Side or Rear YardAdjacent to Residential DistrictsAdjacent to Other DistrictsWith firewall at property line	20' 5' or 0'
5. Maximum Height of Structures	35'

Maximum Impervious Site Coverage

7.5.7.4 Parking and Landscaping Requirements

6.

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the supplemental regulations or special use criteria as provided in Article 8.

7.5.8 <u>B-1 - Neighborhood Business District</u>

7.5.8.1 Purpose of the B-1 Neighborhood Business District

The purpose of the B-1 neighborhood business district shall be to provide sites for small scale service and retail establishments to support adjacent residential neighborhoods. This district includes personal service and retail or office establishments, which are 3,000 square feet or less, which conduct all business operations within an enclosed facility, and which do not present any adverse impact on the peace, appearance or value of adjacent residential areas.

7.5.8.2 B-1 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.8.3 B-1 Site Development Regulations

1 Minimum lot area

Each development site in the B-1 Neighborhood Business District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

1. Millimum lot area	10,000 3.1.
2. Maximum unit size	3,000 S.F.
3. Minimum building size	800 S.F.
4. Minimum lot width and depth	100'
5. Minimum Yard Setback Requirements	
a. Front Yard	25' or

10 000 S F

Required depth of greenbelt, whichever greater

b. Street Side or Rear Yard15' or Required depth of greenbelt, whichever greater

c. Interior Side or Rear Yard
Adjacent to Residential Districts
Adjacent to Other Districts
5' or
With firewall at property line
0'

6. Maximum Height of Structures 35'

7. Maximum Impervious Site Coverage 75%

7.5.8.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the special district criteria below or Special Use Criteria as provided in Article 8.

7.5.8.5 Special B-1 Neighborhood Business District Criteria

- 1. <u>Access</u> B-1 districts shall be located with frontage on major arterial or collector streets or on local streets in locations which do not require travel through existing or proposed residential districts to access the B-1 district.
- 2. Special Requirements Adjacent to Residential Districts
 - a. <u>Landscape Buffer Requirements</u> A vegetative buffer, as provided in Article 9, shall be required in the twenty (20) foot required setback adjacent to all residential districts. If parking or service facilities abut R-1, R-1X or R-2 districts, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed toward the B-1 property, and be shielded from and shall not adversely impact any adjacent use or traffic.

b. <u>Hours of Operation and Noise</u> - If night activities are conducted by the onsite use, such activities shall not interfere with the peace of any adjacent residential district or on-site residential use and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.

c. <u>Outside Storage or Display</u> - There shall be no display or storage of goods outside of the principal structure or any permitted accessory structures on the site.

7.5.9 <u>B-2 - Highway Business District</u>

7.5.9.1 Purpose of the B-2 Highway Business District

The purpose of the B-2 Highway Business District shall be to provide sites for office, retail and service establishments to serve the needs of the community as a whole. This district includes

both multi-tenant shopping centers and individual development sites located typically on major arterial and collector streets.

7.5.9.2 B-2 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.9.3 B-2 Site Development Regulations

Each development site in the B-2 Highway Business District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

1. Minimum lot area 15,000 S.F.

2. Minimum unit size 800 S.F.

3. Minimum lot width	150'		
4. Minimum lot depth	100'		
5. Minimum Yard Setback Requirements			
a. Front YardRequired depth of greenbelt,whichever is greater	25' or		
b. Street Side or Rear Yard Required depth of greenbelt, whichever is greater	15' or		
c. Interior Side or Rear Yard Adjacent to Residential Districts Adjacent to Other Districts With firewall at property line	20' 5' or 0'		
6. Maximum Height of Structures	35'		
7. Maximum Impervious Site Coverage	75%		

7.5.9.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the

provisi ons of Article 9 of this Land Use Regula tions Ordina nce and any additi onal requir ement

S as specifi ed in the special distric criteri below or Specia 1 Use Criteri as provid ed in Article 8.

7.5.9.5 Special B-2 Highway Business District Criteria

- 1. Access B-2 districts shall be located on lots with street frontage on major arterial or collector streets only and shall not require travel through existing or proposed residential districts to access the B-2 district. Because of the potential for traffic congestion created by uses within B-2 districts at major intersections Traffic Impact Analysis may be required by the City Planner prior to the issuance of permits for major commercial developments in accordance with the provisions for Traffic Impact Analysis provided in Article 8.
- 2. Special Requirements Adjacent to Residential Districts
 - a. <u>Landscape Buffer Requirements</u> A vegetative buffer, as provided in Article 9, shall be required in the twenty (20) foot required setback adjacent to all residential districts. If parking or service facilities abut R-1, R-1X or R-2 districts, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed

toward the B-2 property, be shielded from and shall not adversely impact any adjacent use or traffic.

- b. <u>Hours of Operation and Noise</u> If night activities are conducted by the onsite use, such activities shall not interfere with the peace of any adjacent residential district or on-site residential use and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.
- 3. <u>Outside Storage or Display</u> There shall be no display or storage of goods outside of the principal structure or any permitted accessory structures on the site except as specifically provided by these regulations for such uses as auto trailer and boat sales or storage.
- 4. <u>Storage of Waste Materials</u> No waste materials that are the product of any research or testing activity may be stored on-site.

7.5.10 <u>B-3 - Old Mandeville Business District</u>

7.5.10.1 Purpose of the B-3 Old Mandeville Business District

The purpose of the B-3 Old Mandeville Business District shall be to provide a district which acknowledges the historic character of the area and the pedestrian orientation of the neighborhood by continuing to combine residential uses with small scale commercial, service and office establishments which are relatively compatible with residential uses. Lot sizes, setbacks, parking and landscaping requirements shall be more flexible to address the unique characteristics of an area substantially developed as a commercial district with smaller lots and greater development densities than newer areas of the City, prior to the regulation of such elements of site development by local codes.

7.5.10.2 B-3 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.10.3 B-3 Site Development Regulations

Each development site in the B-3 Old Mandeville Business District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

1. Minimum lot area	8,000 S.F.
2. Minimum unit size	800 S.F.
3. Minimum lot width	60'
4. Minimum lot depth	120'
5. Minimum Yard Setback Requirements	
 a. Front Yard Average of adjacent existing setbacks, whichever is less, but under no 	25' or
circumstances less than	10'
b. Street Side or Rear Yard Average of existing adjacent setbacks, whichever is greater	10' or
c. Interior Rear Yard	20'
d. Interior Side Yard - Non-Residential Adjacent to Residential Districts Adjacent to Existing Residential Adjacent to Other Districts With firewall at property line	20' 10' 5' or 0'
6. Interior Side Yard-Residential Uses	5'
7. Maximum Height of Structures	35'
7. Maximum Impervious Site Coverage	75%

7.5.10.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and as specified in the special district criteria below and Special Use Criteria as provided in Article 8.

7.5.10.5 Special B-3 - Old Mandeville Business District Criteria

1. <u>Access</u> - No use which requires regular deliveries by tractor/trailer trucks or vehicles of a load or size greater than the capacity of the streets or bridges or existing clearances of utilities or trees in the area shall be allowed in the B-3 district except on lots fronting on collector streets and which do not require access through residential districts.

2. Special Requirements Adjacent to Residential Districts and Uses

a. <u>Landscape Buffer Requirements</u> - A vegetative buffer, as provided in Article 9, shall be provided in the required setback adjacent to all residential districts and existing residential. If parking or service facilities abut R-1, R-1X or R-2 districts, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed toward the B-3 property, be shielded from and shall not adversely impact any adjacent use or traffic.

b. <u>Hours of Operation and Noise</u> - If night activities are conducted by the onsite use, such activities shall not interfere with the peace of any adjacent residential district or on-site residential use and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.

3. Reductions in Required Parking

- a. <u>By Right</u> When on-street parallel parking is available in areas where ditches have been culverted and shoulders are adequate for parking or when public on-street parking bays are available, the required number of off-street parking spaces may be reduced by up to a number equal to fifty (50) percent of the number of on-street spaces immediately adjacent to or, on Lakeshore Drive only, across the street from the proposed B-3 site in public parking bays on the right-of-way.
- b. <u>By Parking Variance</u> In the B-3 District parking requirements may be reduced or waived by the Zoning Board in conjunction with a variance request and based on the findings of the Zoning Board of the existence of public parking within the area sufficient to accommodate the proposed use, provided

such a reduction or waiver does not adversely affect surrounding commercial or residential uses. The Zoning Board may vary up to fifty (50) percent the number of required parking spaces or may vary the size of parking spaces and parking accessways to provide for an increase in the amount of green space or may accept a contribution to the Optional Parking/Open Space Mitigation Fund established for the purpose of providing public parking and pedestrian/open space amenities in accordance with a Master Plan of the B-3 District. Public improvements such as the culverting of roadside ditches to provide on-street parking within three hundred (300) feet of where the proposed use is located and the installation of sidewalks or pedestrianways between the on-street parking and the site, if done in a manner approved in advance in accordance with the provisions of Section 5.2.6.1, may be accepted as a contribution to the Optional Parking/Open Space Mitigation Fund in lieu of the required parking for the proposed use.

- 4. Reductions in Required Landscaping Because of the inconsistency of building setbacks and the frequency of existing structures which were constructed on the property line at the street frontage or closer than fifteen (15) feet to the street right-of-way, landscaping requirements in the B-3 district may be reduced in conjunction with an application for approval of such reduction in accordance with the procedures and requirements for an Administrative permit when the placement of the existing building or the need for additional on site parking, based on a determination by the City Planner or Designee, make landscaping to the full extent of the requirements impractical or a hardship. A determination by the City Planner or Designee shall be based on the following guidelines:
 - a. No greenbelt in the B-3 district shall be required to be greater than the required building setback as indicated in the B-3 Site Development Regulations above.
 - b. When an existing building(s) or parking lot is less than fifteen (15) feet from a street right-of-way, the required greenbelt shall be the width of the existing setback in the area between the existing building or parking lot and the street. In areas where the existing building(s) or parking lot do not encroach into the required greenbelt the full provisions of the greenbelt requirements shall apply.
 - c. When an existing building is within five (5) feet of a street right-of-way, class B trees may be substituted for class A trees and planter boxes may be utilized instead of in-the-ground installations so long as such planter boxes do not impede pedestrian circulation.

- d. When an existing building or a required setback is within ten (10) feet of a street right-of-way, the required greenbelt may be reduced as provided in (a) and (b) above provided that any open ditches in front of the site are culverted and planted and sidewalks or pedestrian ways are installed, in accordance with plans and requirements approved by the Director of Public Works, if not currently existing, to compensate for the diminished green space.
- 5. Outside Storage or Display There shall be no display or storage of goods outside of the principal structure or any permitted accessory structures on the site except in conjunction with special sidewalk sales or other promotional activities permitted in accordance with the approval of the Community Appearance Commission in accordance with regulations for Special Events in the B-3 district as provided in Article 10 Sign Code.
- 6. <u>Combined Uses</u> The combining of commercial and residential uses on one site shall be in accordance with the Special Use Criteria set forth in Article 8.
- 7. <u>Conversion of Residential Structures to Non-Residential Uses</u> The conversion of residential structures to non-residential uses shall be allowed in the B-3 District subject to the requirements for the establishment of non-residential uses in the district as provided herein and the provisions of Article 8 for the Conversion of Residential Structures to Non-Residential Uses.

7.5.11 <u>B-4 - Major Crossroads Business District</u>

7.5.11.1 Purpose of the B-4 Major Crossroads Business District

The purpose of the B-4 district shall be to provide sites for multi-purpose commercial centers to serve the community at large at the intersections of major arterial streets and to provide for additional height of buildings in such intensely commercial areas located an adequate distance from residential districts to protect residential uses from the impact of multi-story buildings.

7.5.11.2 B-4 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.11.3 B-4 Site Development Regulations

Each development site in the B-4 Major Crossroads Business District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

1. Site Development Regulations for Structures with 35' Height or Less

a. Minimum lot area	30,000 S.F.	
b. Minimum unit size	800 S.F.	
c. Minimum lot width	150'	
d. Minimum lot depth	200'	
e. Minimum yard setback requirements		
i. Front YardRequired depth of greenbelt,whichever greater	25' or	
ii. Street Side or Rear Yard Required depth of greenbelt, whichever greater	15' or	
iii. Interior Side or Rear YardAdjacent to Residential DistrictsAdjacent to Other DistrictsWith firewall as required at property line	20' 5' or 0'	
f. Maximum Height of Structures	35'	
g. Maximum Impervious Site Coverage	75%	

- 2. Site Development Regulations for Structures Greater than 35'
 - a. No structure other than buildings shall exceed thirty-five (35) feet in height under any conditions except as provided in Article 8, Supplemental Regulations.

- b. Buildings greater than thirty-five (35) feet in height and no greater than forty-nine (49) feet in height shall be permitted in B-4 districts but only under the following conditions:
- i. The building is on a lot or parcel which is at least one hundred thousand (100,000) square feet in size.
- ii. The property on which the building is located is not within seven hundred fifty (750) feet or less of a residential zoning district within the City of Mandeville.
- iii. The building is set back at least one hundred (100) feet from every property line.
- iv. At least fifty (50) percent of the additional required yard area, over and above the yard requirements for sites with structures thirty-five (35) feet in height or less, shall be devoted to landscaping. This landscaped area shall be in addition to the normal landscaping requirements of the city and shall be landscaped with grass or other appropriate living ground cover, shrubs, and/or trees. This additional yard area may also be used to provide retention ponds or water quality enhancement features which utilize hydric vegetation to filter storm water.
- v. Maximum impervious site coverage shall be seventy-five (75) percent.

7.5.11.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the special district criteria below or Special Use Criteria as provided in Article 8.

7.5.11.5 Special B-4 Major Crossroads Business District Criteria

1. <u>Location</u> - In order to preserve the character of the city in particular as it relates to the view of the city from the lake or the gateway corridors which provide a first impression of the character of the north shore as a whole to visitors arriving via the Causeway, B-4 districts shall be located a minimum of 10,000 feet from the shoreline of the lake and shall extend to a distance no greater than 2500 feet in any direction from the intersection of the centerlines of two major arterial streets.

2. <u>Access</u> - B-4 districts shall be located on lots with street frontage on major arterial streets only and shall not require travel through existing or proposed residential districts to access the B-4 district. Because of the potential for traffic congestion created by uses within B-4 districts at major intersections Traffic Impact Analysis may be required by the City Planner prior to the issuance of permits for major commercial developments in accordance with the provisions for Traffic Impact Analysis provided in Article 8.

3. Special Requirements Adjacent to Residential Districts

a. <u>Landscape Buffer Requirements</u> - A vegetative buffer, as provided in Article 9, shall be required in the twenty (20) foot required setback adjacent to all residential districts. If parking or service facilities abut R-1, R-1X or R-2 districts, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed toward the B-4 property, be shielded from and shall not adversely impact any adjacent use or traffic.

b. <u>Hours of Operation and Noise</u> - If night activities are conducted by the onsite use, such activities shall not interfere with the peace of any adjacent residential district or on-site residential use and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.

- 4. <u>Outside Storage or Display</u> There shall be no display or storage of goods outside of the principal structure or any permitted accessory structures on the site.
- 5. <u>Storage of Waste Materials</u> No waste materials that are the product of any research or testing activity may be stored on-site.

7.5.12 <u>O/R - Office/Residential District</u>

7.5.12.1 Purpose of the O/R Office/Residential District

The purpose of the Office/Residential District shall be to provide sites for administrative, executive, professional and general offices which are compatible with medium to high density residential uses, but excluding all retail or service oriented commercial uses. This district provides a buffer between commercial and low density residential areas

7.5.12.2 O/R Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.12.3 O/R Site Development Regulations

Each development site in the O/R Office/Residential District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

1. Minimum lot area	15,000 S.F.
2. Minimum lot width and depth	150'
3. Minimum lot area per unit	3,000 S.F.
4. Minimum yard setback requirements	
a. Front YardRequired depth of greenbelt,whichever greater	25' or
b. Street Side or Rear Yard Required depth of greenbelt, whichever greater	15' or
c. Interior Side or Rear Yard- Adjacent to Residential Districts	20'
Adjacent to Other Districts With firewall as required at property line	5' or 0'
5. Maximum Height of Structures	35'
6. Maximum Impervious Site Coverage	75%

7.5.12.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the special district criteria below or Special Use Criteria as provided in Article 8.

7.5.12.5 Special O/R Office/Residential District Criteria

- 1. <u>Access</u> O/R districts shall be located with frontage on major arterial or collector streets or on local streets in locations which do not require travel through existing or proposed residential districts to access the O/R district.
- 2. Special Requirements Adjacent to Residential Districts
 - a. <u>Landscape Buffer Requirements</u> A vegetative buffer, as provided in Article 9, shall be required in the twenty (20) foot required setback adjacent to all residential districts. If parking or service facilities abut R-1, R-1X or R-2 districts, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed toward the O/R property, be shielded from and shall not adversely impact any adjacent use or traffic.
 - b. <u>Hours of Operation and Noise</u> If night activities are conducted by the onsite use, such activities shall not interfere with the peace of any adjacent residential district or on-site residential use and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.
- 3. <u>Outside Storage or Display</u> There shall be no display or storage of goods outside of the principal structure or any permitted accessory structures on the site.

4. Additional Parking Provisions

- a. Parking provided in the O/R office residential district shall be sufficient to accommodate the peak season or peak hours of the business use and shall not adversely impact any residential use within or abutting the O/R district.
- b. When offices with hours of operation between 8:00 a.m. and 5:00 p.m. on Monday through Friday only occupy the same site as residential uses the

required parking may be adjusted to provide for up to fifty (50) percent of the required residential parking to be accommodated on the required parking spaces of the office use under a shared parking agreement signed by the parties entering into such arrangement when a recorded copy of the agreement is submitted to the Planning Department for filling with the case file for the approval of the combined office/residential use by the Zoning Board.

7.5.13 PM-1 Marina District - Waterfront Lots

7.5.13.1 Purpose of the PM-1 Marina District

This district includes all the land within previously subdivided lots, lots of record or parcels of land under the same ownership not previously subdivided into lots which front on Bayou Castain or on navigable bodies of water subject to tidal waters of Bayou Castain on the date of the adoption of this Comprehensive Land Use Regulations Ordinance. The purpose of this district is to provide for the utilization of waterfront lots in association with recreational boating uses and the preservation of critical wetland vegetation areas which provide for the filtering of stormwater runoff prior to its discharge into Lake Pontchartrain as well as floodplain areas during periods of high water to reduce the impact of flood waters on adjacent developed areas.

7.5.13.2 PM-1 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.13.3 PM-1 Site Development Regulations

Each development site in the PM-1 Marina District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

1. Non-Residential Uses Site Development Regulations

a. Minimum lot area 10,000 S.F.

b. Minimum lot width and depth 75' x 133.34'

c. Minimum yard setback requirements

i. Front YardRequired depth of greenbelt,whichever greater	25'	or
ii. Street Side or Rear Yard Required depth of greenbelt, whichever greater	15'	or
iii. Interior Side or Rear Yard Adjacent to Residential Uses Adjacent to Other Uses	20'	5'
d. Maximum Height of Structures	35'	

2. Residential Site Development Regulations

e. Maximum Impervious Site Coverage

Residential site development regulations shall be in accordance with the site development regulations of the R-2 - Two-Family Residential District as provided under Section 7.5.3.

60%

7.5.13.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the special district criteria below or Special Use Criteria as provided in Article 8.

7.5.13.5 Special PM-1 Marina District Criteria

- 1. <u>Standards for Marina and Marina Associated Uses</u> Standards for marina and marina associated uses shall be in accordance with the Special Marina Use Criteria provisions of Article 8.
- 2. <u>Standards for All Non-Residential Uses</u> Non-residential uses permitted in the PM-1 District in accordance with the Table of Permitted Uses by Zoning District shall be constructed in accordance with the PM-1 site development regulations for non-residential uses and any other provisions applicable to the specific use and in addition:

- a. <u>Landscape Buffer Requirements</u> A vegetative buffer, as provided in Article 9, shall be provided in the required setback for non-residential uses adjacent to all residential districts and existing residential uses. If parking or service facilities of a non-residential use abut R-1, R-1X or R-2 districts or adjacent one or two family residential uses, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed toward the non-residential use property, be shielded from and shall not adversely impact any adjacent use or traffic.
- b. <u>Hours of Operation and Noise</u> If night activities are conducted by the onsite use, such activities shall not interfere with the peace of any adjacent residential district or on-site residential uses and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.
- c. <u>Outside Storage or Display</u> There shall be no display or storage of goods outside of the principal structure or outside any permitted accessory structures in
- conjunction with a non-residential use except as specifically provided by these regulations for such uses as auto trailer and boat sales or storage.
- d. <u>Fuel Service Facilities</u> Fuel service facilities on Bayou Castain shall be located in areas south of the Madison Street right-of-way in order to prevent boat traffic congestion in the sharper turns of Bayou Castain north of Madison Street. In addition, access for fuel supply trucks shall be by way of Jackson Avenue to Madison Street to prevent the fuel trucks from traveling through the predominantly single-family residential streets of the area.
- 3. <u>Combined Uses</u> The combining of non-residential and residential uses on one site shall be in accordance with the applicable Special Use Criteria in Article 8 and shall be subject to the requirements for the approval of a Zoning Permit as described in Article 4.

7.5.14 PM-2 Marina District - Non-Waterfront Lots

7.5.14.1 Purpose of the PM-2 Marina District

The purpose of the PM-2 Marina District for non-waterfront lots shall be to provide the opportunity for site plan review of developments, including a mixture of residential and non-residential uses, in close proximity to marina and other water oriented uses. Should an area

be in an environmentally sensitive area which serves as the floodplain of Bayou Castain, site plan reviews are required in order to provide for the appropriate use of the lands within the district while protecting the existing residential character and important floodplain and water quality enhancement aspects of the district.

7.5.14.2 PM-2 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.14.3 PM-2 Site Development Regulations

a. Minimum lot area

Each development site in the PM-2 Marina District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

10,000 S.F.

1. Non-Residential Uses Site Development Regulations

b. Minimum lot width and depth	100'	
c. Minimum Yard Setback Requirements		
i. Front YardRequired depth of greenbelt,whichever greater	25' or	
ii. Street Side or Rear Yard Required depth of greenbelt, whichever greater	15' or	
iii. Interior Side or Rear Yard Adjacent to Residential Uses Adjacent to Other Uses	20'	5'
d. Maximum Height of Structures	35'	

e. Maximum Impervious Site Coverage

60%

2. Residential Site Development Regulations - Residential site development regulations shall be in accordance with the site development regulations of the R-2 - Two-Family Residential District as provided under Section 7.5.3.

7.5.14.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the special district criteria below or Special Use Criteria as provided in Article 8.

7.5.14.5 Special PM-2 Marina - Non-Waterfront Lots District Criteria

- 1. <u>Access Non-Residential Uses</u> No non-residential use which requires a Zoning Permit or requires regular deliveries by tractor/trailer trucks or vehicles which are of a load or size greater than the capacity of the streets or bridges or existing clearances of utilities or trees in the area shall be allowed in the PM-2 district.
- 2. Non-Residential Use Requirements Adjacent to Residential
 - a. <u>Landscape Buffer Requirements</u> A vegetative buffer, as provided in Article 9, shall be provided in the required setback for non-residential uses adjacent to all residential districts and existing residential uses. If parking or service facilities of a non-residential use abut R-1, R-1X or R-2 districts or adjacent one or two family residential uses, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed toward the non-residential use property, be shielded from and shall not adversely impact any adjacent use or traffic.
 - b. <u>Hours of Operation and Noise</u> If night activities are conducted by the non-residential use, such activities shall not interfere with the peace of any adjacent residential district or on-site residential use and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.

- 3. <u>Outside Storage or Display</u> There shall be no display or storage of goods outside of the principal structure or any permitted accessory structures in conjunction with a non-residential use except as specifically provided by these regulations for such uses as auto trailer and boat sales or storage.
- 4. <u>Combined Uses</u> The combining of non-residential and residential uses on one site shall be in accordance with Special Use Criteria in Article 8.
- 5. <u>Marina Associated Non-Residential Use Standards</u> Any non-residential use usually associated with marina uses located on a site in the PM-2 district shall be subject to the standards for such uses as provided under the Special District Criteria for the PM-1 District.

7.5.15 <u>PD - Planned District</u>

7.5.15.1 Purpose of the Planned District

The purpose of the Planned District shall be to provide for an improved level of aesthetics, safety and environmental sensitivity and design flexibility in conjunction with a site plan review procedure for the approval of residential, commercial, industrial or a combination of these uses on one unified development site by ordinance of the City Council subsequent to the recommendation of the Planning Commission. Development sites approved by ordinance under the site plan review procedures of a Planned District shall be approved as a Planned Residential District (PRD), a Planned Commercial District (PCD), a Planned Industrial District (PID) or a Planned Combined Use District (PCUD) in accordance with the classification of use or uses proposed and/or existing. Planned District applications shall contain a statement by the developer as to how the submitted plan departs from the existing requirements of this Land Use Regulations Ordinance and any other regulations applicable to the proposed use or uses for the district in which the proposed use could be established of right and how each departure improves what otherwise would be required under these regulations.

7.5.15.2 Planned District Permitted Uses

All uses permitted in the Planned District are conditional uses and shall be subject to the procedural requirements for Conditional Use Permits and Planned District Zoning as provided in Article 4. The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.15.3 Enactment By Ordinance

The City Council approval of a Planned District development shall be by amendment to these regulations and the Official Zoning Map. Said amendment shall designate and define the boundaries of the Planned District and include such conditions as the City Council finds are necessary to secure and protect the public health, safety, and general welfare. The procedure for the approval or denial of a Planned District application shall be in accordance with the procedural guidelines for Planned District and Conditional Use Applications as provided in Article 4.

7.5.15.4 Flexible Site Planning

When considering a Planned District application, the unique nature of each proposal may require, under proper circumstances, the departure from the strict enforcement of certain present codes and ordinances, included but not limited to the width and surfacing of streets and highways, alleyways and street lights, public parks and playgrounds, school sites, storm drainage, water supply and distribution, sanitary sewers, sewage collection and treatment, lot and area regulations, landscaping and parking requirements. Final approval of a Planned District Development by the City Council shall constitute authority for such flexible planning to the extent that the Planned District as approved, departs from existing codes and ordinances.

7.5.15.5 Review of Plan Based on Existing Regulations

Review of the conceptual site plan submitted in conjunction with a request for Planned District zoning or amendment shall be based upon the requirements of this Land Use Regulations Ordinance for the proposed use, including any special use criteria provided in Article 8, and the requirements of the zoning district in which the use would be permitted by right. Unless specific regulations regarding lot area, lot area per unit, or building area are set forth in Article 8 of these regulations, requirements for lot area, lot area per unit, and building area for any proposed use shall be based upon any applicable site development regulations of the following designated zoning districts.

Single Family Residential or Community Residential

R-1 Single Family Residential District, provided, however, that for good cause demonstrated by the applicant, the Zoning Board and City Council may apply the applicable requirements of any R-1X Single Family Existing Residential District which is adjacent to the single family development proposed on the site plan.

Duplex Residential, Cluster

R-2 Two Family Residential District

Residential, or Two Family Residential

Condominium Residential or Multiple Family Residential R-3 Multifamily Residential District

Boathouse Residential R-2 Two Family Residential District

Mobile Home Residential MH Mobile Home District

Civic Uses B-1 Neighborhood Business District

Commercial Uses B-2 Highway Business District, provided,

however, that commercial uses designated on a site plan providing for a mix of commercial uses and residential uses shall comply with the requirements of the B-1 Neighborhood Business

District

Custom Manufacturing and

Light Manufacturing

M-1 Industrial District

Other Manufacturing Uses M-2 Industrial District

Agricultural Uses M-2 Industrial District

Combined Uses: In applying the provisions of this section to site plans proposing a maxed or combined use of property, separate consideration shall be given to the area of the site plan designated for each use.

7.5.15.6 Preliminary Subdivision Plat

A preliminary subdivision plat shall be submitted, in accordance with the requirements as specified in Division III of these Land Use Regulations, along with the site plan submitted for Planned District approval if any subdivision of land is proposed or public improvements are proposed to be installed to service the planned development.

7.5.15.7 Existing Planned Developments

Planned District developments approved under procedures applicable prior to the date of adoption of these Land Use Regulations shall be deemed to be Planned Districts and shall

continue to be governed by regulations and requirements previously applicable. Previously existing Planned Districts shall be shown on the Zoning Map as Planned Districts.

7.5.15.8 Existing Developments Zoned as Planned Districts by this Ordinance

Planned Districts zoning enacted in conjunction with the adoption of this Land Use Regulations Ordinance and the Official Zoning Map shall be subject to the requirements of this Ordinance. Previously developed lots shall be approved as a Planned District under the provisions of this ordinance, in accordance with the site plan approved in conjunction with the issuance of the permit for the construction of the existing structures on the individual lot for which the permit was issued. However, previously issued permits shall be subject to the provisions regarding the expiration of conditional use permits as provided in Article 4 under Procedures for Conditional Use Permits and Planned District Zoning and any proposed changes to the previously permitted plans shall be subject to the procedures for the amendment of a Planned District site plan.

7.5.15.9 Undeveloped Sites Zoned as Planned District by this Ordinance

Previously undeveloped lots zoned Planned District by the adoption of this Comprehensive Land Use Regulations Ordinance and its accompanying Official Zoning Map shall be subject to site plan review in accordance with the Procedures for Conditional Use Permits and Planned District Zoning, as provided in Article 4, prior to the issuance of any development permit for any development or use of the previously undeveloped lot.

7.5.15.10 Change of Zoning Classification

In the event any Planned District is changed by ordinance to another zoning district, the Planned District Site Plan and Preliminary Subdivision Plat, if applicable, shall become null and void on all portions thereof affected by such change.

7.5.15.11 Parking and Landscaping Requirements

Parking and landscaping requirements for the Planned District shall be in accordance with the provisions of Article 9 and/or Special Use Criteria as provided in Article 8 of this Land Use Regulations Ordinance and any additional requirements or special exceptions as specified in the provisions for Planned Districts.

7.5.16 <u>M-1 Light Manufacturing District</u>

7.5.16.1 Purpose of the M-1 Light Manufacturing District

The purpose of the M-1 district shall be to accommodate enterprises engaged in the manufacturing, processing, creating, repairing, renovating, painting, cleaning or assembling of goods, merchandise or equipment, all within an enclosed structure and subject to the performance standards set forth in Article 8 of these regulations.

7.5.16.2 M-1 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.16.3 M-1 Site Development Regulations

1. Minimum lot area

Each development site in the M-1 Light Manufacturing District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Ordinance or any other laws of the City, parish, state or federal government.

15,000 S.F.

2. Minimum unit size	800 S.F.
3. Minimum lot width	150'
4. Minimum lot depth	100'
5. Minimum Yard Setback Requirements	
a. Front Yard Required depth of greenbelt, whichever is greater	25' or
b. Street Side or Rear Yard Required depth of greenbelt, whichever is greater	15' or

c. Interior Side or Rear Yard

Adjacent to Residential Districts 20'
Adjacent to Other Districts 5' or
With firewall as required at property line 0'

6. Maximum Height of Structures 35'

7. Maximum Impervious Site Coverage 75%

7.5.16.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the special district criteria below or Special Use Criteria as provided in Article 8.

7.5.16.5 Special M-1 Light Manufacturing District Criteria

1. Access - M-1 districts shall be located on lots with street frontage on major arterial or collector streets only and shall not require travel through existing or proposed residential districts to access the M-1 district. Because of the potential for traffic congestion created by uses within M-1 districts, Traffic Impact Analysis may be required by the City Planner prior to the issuance of permits for major commercial developments in accordance with the provisions for Traffic Impact Analysis provided in Article 8.

2. Special Requirements Adjacent to Residential Districts

- a. <u>Landscape Buffer Requirements</u> A vegetative buffer, as provided in Article 9, shall be required in the twenty (20) foot required setback adjacent to all residential districts. If parking or service facilities abut R-1, R-1X or R-2 districts, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed toward the M-1 property, be shielded from and shall not adversely impact any adjacent use or traffic.
- b. <u>Hours of Operation and Noise</u> If night activities are conducted by the onsite use, such activities shall not interfere with the peace of any adjacent

residential district and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.

- 3. <u>Outside Storage or Display</u> There shall be no display or storage of goods outside of the principal structure or any permitted accessory structures on the site except as specifically provided by these regulations for such uses as auto trailer and boat sales or storage.
- 4. <u>Storage of Waste Materials</u> No waste materials that are the product of any research, testing or manufacturing activity may be stored on-site.

7.5.17 <u>M-2 General Manufacturing District</u>

7.5.17.1 Purpose of the M-2 General Manufacturing District

The purpose of the M-2 district shall be to provide an area for industrial development that is compatible with the residential emphasis of the community. Industrial uses that will be permitted by right shall be safe and non-offensive to protect the natural character of the community and the integrity of surrounding residential districts. An industrial use that may adversely impact surrounding districts will be subject to the Zoning Permit or conditional use permitting procedure as outlined in Article 4. Industrial uses, whether permitted conditionally or by right, will be subject to the performance standards of Article 8 of these regulations.

7.5.17.2 M-2 Permitted Uses

The uses permitted in this zoning district, including signage, shall be in accordance with those uses listed under this district in the Table of Permitted Uses By Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance

including any supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.17.3 M-2 Site Development Regulations

Each development site in the M-2 General Manufacturing District shall be subject to the following site development regulations in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

1. Minimum lot area	30,000 S.F.
2. Minimum unit size	800 S.F.
3. Minimum lot width	150'
4. Minimum lot depth	200'
5. Minimum Yard Setback Requirements	
a. Front Yard Required depth of greenbelt, whichever is greater	25' or
b. Street Side or Rear Yard Required depth of greenbelt, whichever is greater	15' or
c. Interior Side or Rear Yard Adjacent to Residential Districts Adjacent to Other Districts With firewall as required at property line	20' 5' or 0'
6. Maximum Height of Structures	35'
7. Maximum Impervious Site Coverage	75%

7.5.17.4 Parking and Landscaping Requirements

Parking and landscaping requirements for this district shall be in accordance with the provisions of Article 9 of this Land Use Regulations Ordinance and any additional requirements as specified in the special district criteria below or Special Use Criteria as provided in Article 8.

7.5.17.5 Special M-2 General Manufacturing District Criteria

1. Access - M-2 districts shall be located on lots with street frontage on major arterial or collector streets only and shall not require travel through existing or proposed residential districts to access the M-1 district. Because of the potential for traffic congestion created by uses within M-2 districts Traffic Impact Analysis may be required by the City Planner prior to the issuance of

permits for major commercial developments in accordance with the provisions for Traffic Impact Analysis provided in Article 8.

2. Special Requirements Adjacent to Residential Districts

- a. <u>Landscape Buffer Requirements</u> A vegetative buffer, as provided in Article 9, shall be required in the twenty (20) foot required setback adjacent to all residential districts. If parking or service facilities abut R-1, R-1X or R-2 districts, the landscape buffer requirement may be required by the City Planner or Designee to be increased by up to fifty (50) percent to diminish the effects of the impact of lighting, noise, odors or other negative effects on adjacent residential developments. All outdoor lighting must be directed toward the M-2 property, be shielded from and shall not adversely impact any adjacent use or traffic.
- b. <u>Hours of Operation and Noise</u> If night activities are conducted by the onsite use, such activities shall not interfere with the peace of any adjacent residential
- district and shall conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances.
- 3. <u>Outside Storage or Display</u> There shall be no display or storage of goods outside of the principal structure or any permitted accessory structures on the site except when screened from view of any adjacent property or street rights-of-way and as specifically provided by these regulations for such uses as auto trailer and boat sales or storage.
- 4. <u>Storage of Waste Materials</u> No waste materials that are the product of any research, testing or manufacturing activity may be stored on-site.

7.5.18 <u>TOWN CENTER DISTRICT</u>

7.5.18.1 Purpose of the Town Center District

The purpose of the Town Center District (TC) is to develop an identifiable center of the City of Mandeville with the Trailhead as its nucleus. Its intent is to further define a sense of community and to promote and encourage the development of residential, business and related distribution and service facilities which will reflect the indigenous eclectic architecture, people oriented scale and environmentally friendly character indicative of Old Mandeville.

7.5.18.2 Town Center Permitted Uses

The uses permitted in this zoning district, shall be in accordance with the uses listed under this district in the Table of Permitted Uses by Zoning District found at the end of this Article and shall be subject to all applicable provisions of this Land Use Regulations Ordinance including all supplemental or special use criteria provided in Article 8 and the Sign Code in Article 10.

7.5.18.3 Town Center Site Development Regulations

Each development site in the Town Center District shall be subject to the site development regulations as outlined in the B-3 Zoning District, Overlay Districts as may be applicable in addition to any other applicable regulations under the provisions of this Land Use Regulations Ordinance or any other laws of the City, state or federal government.

7.5.18.4 Parking Requirements

Each development site in the Town Center District shall be subject to the site development regulations as outlined in the B-3 Zoning District, if the property is not located in an Overlay District, or the Overlay Districts in which the property is located, in addition to any other laws of the City, state or federal government.

7.6 OVERLAY ZONING DISTRICT REGULATIONS

7.6.1 <u>D-O Drainage Overlay District</u>

7.6.1.1 Purpose of the Drainage Overlay District

The purpose of the drainage overlay district is to provide for the maintenance of existing natural drainage areas in a naturalistic state while protecting the public health, safety and welfare in order to provide for areas (1) to accommodate the spread of stormwaters during high water conditions, (2) to preserve the natural beauty and character of the community, (3) to provide natural habitat for native vegetation and for the preservation of wildlife in linear corridors which provide linkage between open space habitat parcels, and (4) to retain and filter storm water as it flows to the lake in order to reduce the effects of run-off pollutants on the lake.

7.6.1.2 D-O Permitted Uses

The permitted uses in the D-O District shall be in accordance with the uses permitted in the base zoning districts as listed in the Table of Permitted Uses By Zoning District found at the end of this Article, except in the case of proposed construction in areas of periodic inundation

as provided below which require a Zoning Permit issued in conjunction with the approval of the Zoning Board.

7.6.1.3 Area Within D-O Drainage Overlay Districts

Any lot, lot of record, or parcel of land made up of lots, lots of record or combination of lots and portions of lots in single ownership, which are considered to be a single parcel under the provisions of these regulations as provided in Article 4, which is adjacent to or includes areas of periodic inundation from the flood waters of a natural drainageway through the City of Mandeville including those areas adjacent to natural drainageways subject to the regulations of the State Coastal Management Division (currently areas below the five (5) foot MSL contour), subject to Section 10 of the Rivers and Harbors Act and subject to Section 404 of the Clean Water Act, shall be considered to be a parcel within the D-O Drainage Overlay District and shall be subject to the regulations of this district.

7.6.1.4 D-O Site Development Regulations

For the purposes stated in the creation of this overlay district, no fill shall be placed and the vegetative matter shall be preserved undisturbed in areas of periodic inundation, and any development of land shall only occur in the buildable area of lots outside of areas of periodic inundation except in cases where the remaining buildable area of the site is insufficient to meet the requirements of the district in which it is located and to construct a proposed building in which case the following provisions shall apply.

- 1. Construction in Buildable Area of Lot When a previously subdivided or legally created lot of record or a single undivided parcel of land under the provisions of Article 4, Non-Conforming Lots, is deficient in required area or required buildable area due to areas of periodic inundation, as defined, or areas subject to state and federal jurisdiction as listed above, the owner may construct a single-family home in base residential districts or a non-residential building as permitted in the base zoning district on the non-conforming lot provided that the building is placed in such a way so as not to disturb the vegetation or require the placement of fill in the areas of periodic inundation; provided, however, that the proposed building site falls within the buildable area of the lot or lots under single ownership and meets all other requirements of the base district in which it is located.
- 2. <u>Construction in Areas of Periodic Inundation</u> In those situations in which the buildable area is insufficient for the construction of a proposed building or structure which in all other respects meets the requirements of the base zoning district in which it is to be located, exclusive of any areas of periodic inundation, the building or structure may be constructed within an area of periodic inundation, provided that:

- a. the proposed building or structure is to the greatest extent possible located out of any area of periodic inundation; and
- b. existing vegetation within areas of periodic inundation is disturbed by construction to the minimum extent possible and is restored to the extent possible to its condition prior to the commencement of construction as soon as is reasonably possible after construction is completed or abandoned; and
- c. no fill material shall be used to satisfy the floor elevation requirements of the National Flood Insurance Program for construction proposed for any lot or parcel of land located in an area designated as an "A" or "V" zone on the latest flood insurance rate map issued by or under the auspices of the Federal Emergency Management Agency.
- 3. Permits Required Prior to Construction No development within an area of periodic inundation shall occur until a development permit for the proposed activity has been approved for issuance by the Zoning Board and actually issued to the applicant by the Building Inspector, in accordance with the procedures for the issuance of Zoning Permits set forth in Article 4 of these regulations. Nothing herein shall excuse an applicant who has obtained a development permit from the City for work or construction in an area of periodic inundation that is subject to the jurisdiction of a state or federal permitting agency from any requirement to obtain all licenses or permits required by such agencies for the proposed work or construction. In those instances in which proposed development activity will be subject to the permitting authority of both the City of Mandeville and of one or more state or federal agencies, applicants are urged to first obtain all permits required by the City under the terms of these regulations.
- 4. <u>Construction of Parking and Accessory Structures</u> Any required parking spaces shall be located in areas outside of any areas of periodic inundation except in cases where such area is insufficient to provide the required parking spaces, in which case the minimum number of required parking spaces and an access drive may be provided in the areas of periodic inundation, provided they are located outside of any areas of periodic inundation to the greatest extent possible and no more than the minimum required area per parking space and access drive is constructed within any areas of periodic inundation.
- 5. <u>Culverting of Natural Drainageways</u> It shall be prohibited to culvert an area identified by the city as a natural drainageway unless determined by the City Council to be in the interest of the health, safety and welfare of the general public after holding a public hearing on the matter.

6. Greenbelts Adjacent to Natural Drainageways

- a. Areas of Periodic Inundation as Greenbelt On parcels of land containing areas of periodic inundation adjacent to natural drainageways, the areas of periodic inundation, or an area a minimum depth of fifteen feet, whichever is greater, shall be maintained in its natural vegetative state except for one fifteen (15) foot accessway to the drainageway in every one hundred fifty (150) feet or major fraction thereof.
- b. <u>Greenbelts Adjacent to Ridges, Bulkheads and/or Manmade Edges</u> When the edge of an area adjacent to a natural drainageway or navigable waterway is not subject to periodic inundation and does not fall within the jurisdiction of the state or federal agencies noted above due to a ridgeline, bulkheading, filling or other manmade alteration of the edge of the drainageway, or is the edge of a marina basin, such edge shall be required to preserve any existing vegetation within an area fifteen (15) feet in depth from the waters edge or any built edge adjacent to the water except for one fifteen (15) foot accessway to the waters edge every 150' or major fraction thereof. In addition, if such a greenbelt does not currently exist on a previously developed site, a fifteen (15) foot greenbelt meeting the requirements of the greenbelt provisions of Article 9 Landscaping Provisions shall be required to be installed in conjunction with the issuance of any development permits on the development site.

7.6.2 <u>G-O Gateway Overlay District</u>

7.6.2.1 Purpose of the G-O Gateway Overlay District

The purpose of the G-O Gateway Overlay District shall be to preserve the identified character of the City in the corridors identified as the gateways to the north shore.

7.6.2.2 Finding of Facts, Statement of Purpose and Need for Regulation

- 1. The Gateway Overlay District is identified as those corridors listed in Section 7.6.2.3 of this Ordinance, and that all commercially zoned property within the City Limits adjacent to the above Gateway Corridors shall be subject to Design Guidelines.
- 2. The need for the Gateway Overlay District including design guidelines is to create a sense of place for Mandeville and Western St. Tammany Parish has been identified by the citizenry of Mandeville.
- 3. The City of Mandeville has a strong historic context with seven principle architectural

styles, six of which were used to establish the design guidelines. Public comment indicated that the modern style, although identified was not desirable and does not fit into the image of historic context. Those styles are as follows:

- Creole Colonial: the earliest buildings in town, which generally can be linked to deMarigny. These houses have a central hall, porches front and back, and symmetrical plan. This style influenced the next two styles:
- Northshore: a cottage with porches on two or more sides, with rooms opening onto outside galleries. Has its roots in the need for rooms for tourists (boarding houses). Has a very vertical style with simple columns and railings (if any). Ornamentation was added to some for Victorian or Carpenter hybrids.
- Coastal Classical: a humble hybrid of the neo-classical with a center hall plan, and larger heavier columns with capital and base, and heavier railings on the front porch. Some more classical elements and details, such as a cornice were added. Symmetrical plan and front elevation was usually a single family dwelling.
- Victorian (high, middle, and low): Characterized by vertical proportions, front facing gables, ornamentation, and a departure from symmetry. Frequently served as the basis for "modernizing" the earlier Creole and Northshore cottages.
- Carpenter Style: Reflects a period in Mandeville history when a large segment of the population was employed by the sawmill. Based on several other styles including the Coastal, Victorian, and Bungalow. Characterized by the use of jigsaw-cut boards for ornamentation, including railing pickets, rake trim, trellis, columns, etc.. In the twenties and thirties, this style became the "modern" bungalow style, with tapered square columns supported by brick railing high piers. Several Creole cottages were "modernized" into this style. First Mandeville style to utilize longer spans and more horizontal proportions.
- Modern: Mechanical cooling and automobile accommodation mark the basis for this style. Usually stripped clean of detail, introverted, and very industrial in style, and simplistic use of material, and almost total rejection of classical discipline. This style was not recommended by the citizenry as a style to be used to establish historic context.
- Post-Modern: An attempt to make Modernism more humane with the incorporation of classical elements. Most better-designed shopping centers in the area are in this style.
- 4. The use of the Gateway Corridors is one of local and commuter traffic at lower speed levels, without need to service large numbers of transients or interstate commerce, and therefore the current design vocabulary used by some national chains and franchises is typically not valid within these corridors.
- 5. The proximity of residential zoning and uses to the Gateway Overlay District

- Corridors demands respectful design standards to be compatible with the character of residential development.
- 6. Energy conservation will be enhanced by buildings being designed with respect for our environment.
- 7. Gateway Overlay District should be more pedestrian friendly and integrate alternative transportation elements such as sidewalks, and bike paths.
- 8. A format for businesses to be put on an even footing of cooperation rather than competition will be provided.
- 9. The Gateway Overlay District has been historically developed and zoned "business" and not industrial and should reflect that image.
- 10. The Design Guidelines complement the existing Landscape, Sign, and Lighting Ordinances.
- 11. Agreements must be developed and enforced between St. Tammany Parish and the City of Mandeville to require that all Commercial properties within the Gateway Overlay District Corridors be treated according to the Guidelines applicable to both Parish and City Properties.

7.6.2.3 G-O Permitted Uses

The permitted uses in the G-O District shall be in accordance with the uses permitted in the base zoning districts as listed in the Table of Permitted Uses By Zoning District found at the end of this Article, except in the case of proposed construction in areas of periodic inundation as provided below which require a Zoning Permit issued in conjunction with the approval of the Zoning Board.

7.6.2.4 Pre-application Procedures for all properties within the Gateway Overlay District:

Prior to preparing a conceptual site plan for any development and before any construction, clearing or other ground wok is undertaken, the applicant shall schedule a pre-application meeting with the City Planner or Designee to discuss the procedure for approval of site plans and for the development and construction of properties in the Gateway Overlay District.

7.6.2.5 Area Within G-O Gateway Overlay District

Any lot, lot of record, or parcel of land made up of lots, in single ownership, which are considered to be a single parcel under the provisions of these regulations including the provisions of Article 4, which has frontage on the following:

- 1. LA Highway 22 from the western city limits eastward to its intersection with US Highway 190;
- 2. US Highway 190, from its intersection with LA Highway 22 eastward to the eastern city limits at Bayou Castain;
- 3. West Causeway Approach, and adjacent service roads, identified as Weldon Park Drive and Beau Rivage Village service road;
- 4. East Causeway Approach
- 5. North Causeway Approach from the foot of the Lake Pontchartrain Causeway to its intersection with LA Highway 22 and adjacent service roads, including Lovers Lane and Elmwood Place.

7.6.2.6 G-O Site Development Regulations

- 10. <u>Control of Access</u> Because of the high volume of traffic on the gateway corridors and their associated service roads, existing non-conforming development sites which do not conform to the controlled access provisions of Article 9 associated with parking and landscaping, shall be required to conform to the provisions of Article 9 at the end of a three year amortization period which begins at the time of the adoption of this Land Use Regulations Ordinance, or three years from the time they were included within the Gateway Overlay District.
- 2. <u>Landscaping</u> The appearance of the sites which front on the gateway corridors is critical to the image that the City of Mandeville and the north shore as a whole presents to visitors arriving by way of the Causeway. For this reason the landscaping requirements of Article 9 shall apply to all lots within the G-O District and existing non-conforming development sites which do not conform to the full provisions of the landscape requirements of Article 9, shall be required to conform at the end of a three year amortization period which begins at the time of the adoption of this Comprehensive Land Use Regulations Ordinance, or three years from the time they were included within the Gateway Overlay District, with the following additional criteria:
- 3. <u>Height</u> The height limitations of structures in the base zoning districts shall be strictly adhered to in the G-O District to preserve the identified character of the City which states that trees are and shall remain the dominant vertical element of the community. This provision applies to all structures including signs.

- 4. <u>Signs</u> Signs in the G-O District shall be in conformance with CLURO Section 10.5.4.6.2 Monument Signs
- 11. Periphery Greenbelt A twenty -five foot (25') periphery landscape area, also known as the greenbelt area, shall be required to be located adjacent to the property line of the right of way of any road or street within the G-O District upon which the site fronts. That if undue hardship is created on any specific lot due to the increased depth of the greenbelt, then variance procedures as outline din Section 4.3.4.5 of the CLURO may be followed.
 - The greenbelt throughout the Highway 190 Widening Project Corridor, (Highway 190 right-of-way from its intersection with Highway 22 to its intersection with the Mandeville City Limits at Bayou Castine), shall be defined as the area from the existing DOTD right-of-way line a distance of twenty-five (25') feet or from the existing DOTD right-of-way line to the new DOTD right-of-way line, whichever is greater; and
 - All non-conforming development sites along the Highway 190 Widening Project Corridor, shall follow Section 4.2.3 provisions for legally non-conforming development sites as outlined under the CLURO.
- 12. Minimum off-street parking requirements may be reduced by right in order to allow for the same density of development utilizing the following formula: 10 x frontage of property / 162.
- 13. <u>Shared Vehicular Access</u> Interconnection of parking areas via access drives within and between adjacent lots is required. Written assurance and/or deed restriction, satisfactory to the City Attorney, binding the owner and his/her successors and assignees to permit and maintain such internal access and circulation and interconnection of parking facilities.
- 14. <u>Shared Pedestrian Access:</u> Interconnection of pedestrian accesses within and between adjacent lots and connection to existing pedestrian infrastructure is required. Written assurance and/or deed restriction, satisfactory to the City Attorney, binding the owner and his/her successors and assignees to permit and maintain such internal access and circulation and interconnection of parking facilities.
- Communication towers located within the Gateway Overlay District shall constructed in the form of a monopine (tree like structure). All other Communication Tower Application Requirements as outlined under Section 8.1.2.2.1 of the CLURO shall apply.

7.6.2.7 Applicability

The Gateway Overlay District Design Guidelines shall apply to the new building construction, exterior renovations and modifications which require a building permit and which are located in the designated gateway overlay district.

Provisions for Non-Conforming buildings are in accordance with Section 4.2.4 of the CLURO.

Prior to preparing design plans for any development, the applicant shall schedule a pre-application meeting with the City Planner or designee to discuss the procedure for approval of design drawings and the development of properties within the Gateway Overlay District

7.6.2.8 Architectural Review

Architectural Review shall be performed by Tulane University, Regional Urban Design Center or alternative successor review body designated by the Mayor, which shall make recommendations to the Building Inspector prior to permits being issued.

7.6.2.9 Building Elements

Seven historic architectural styles for the City of Mandeville have been identified. These styles are: Creole Colonial, Northshore, Coastal Classical, Victorian, Carpenter Style, Modern, and Post-Modern. It is recognized that six of the above styles are hybrids of the pure historic Architectural styles, and the intrinsic common elements are used to form the basis for the following guidelines:

Building Design elements:

- 1. Compatibility with the environment. Buildings shall exhibit the ability to provide protection from rain, sun, and high humidity.
- 2. Entrances each principle building shall have a clearly defined, inviting, highly visible customer entrance enhanced with distinguishing features such as canopies, galleries, and porticos.
- 3. Facades of buildings visible to the public shall maintain the same standard of design as the front facade, including:
- 1. Screening of utilities, equipment, a building services.
 - ii Continuation of building design elements such as the quality of materials, galleries, cornices, and treatment of openings.

- Disruption of horizontal planes with vertical elements are required. This may include significant interruption by change in plane, material, opening, or design element, such as a tower or gable.
- Disciplined visible structural vocabulary must be maintained. Arcades, galleries, and roofs shall not appear to levitate in space, but shall have a visible means of support with columns and/or brackets. No overhangs in excess of 3' allowed without a visible means of support. Rafter tails are encouraged on smaller overhangs.
- Consistent design vocabulary for multiple structures on one property will be employed. A unifying design element such as material, color, or form should be used for all structures.
- No building with an industrial appearance is allowed, such as a pre-engineered metal building with metal siding and devoid of historic context.
- Service bays shall be oriented away from the principal street or screened.
- Smaller buildings should reflect the design of the historic styles, and larger buildings should be divided into smaller elements in order to incorporate historic design context.
- Buildings should maintain classic proportions. For example, smaller columns should be placed closer together for a more vertical proportion, and as the structure becomes more horizontal in scale the supports (columns) should have additional mass.
- Fascias of buildings not to exceed 16" in depth, including gutters; except for fascias used as a unifying element for multi-tenant buildings and for placement of signage for tenants. No backlit fascias

For purposes of this ordinance, fascia is defined as the horizontal plane just below the roof or coping and above the wall and/or supports.

- Mansard roofs used in conjunction with canopies, covered walkways and entries shall have a roof-like slope not greater than 12:12 or less than 4:12.
- Building should have substance design should include base, intermediate and cap. Changes in materials should have a clear line of demarcation either by offset, reveal, or border.

Shadows shall be considered as a design element.

MATERIALS: Materials shall be reviewed for compliance with historic context. The following materials have historic context:

Walls, Wood, Brick, Cement Plaster(stucco).

Roofing: Wood shakes, Slate/tile, Rigid Shingles with Ridge Tiles, Metal (Corrugated, V-crimp, and Standing Seam).

- 3. COLOR: Colors shall be reviewed for compliance with historic context.
 - 4. Facade colors shall be low reflective and subtle. The use of primary, high intensity or metallic color is prohibited outside of the sign face
 - 5. Any activity that involves changing color or refreshing color shall require a permit and shall be reviewed by the Design Review Board.

Accurate color drawings with a list of paint numbers and elevations of every building will be required to be submitted prior to any modification.

CANOPIES:

- Free (or semi-free) standing canopies, such as those used as shelters for pump islands in gas stations and porte cocheres, shall be of similar style and materials as the building. Canopies are not considered the principle structure.
- b. Unless site conditions preclude, canopies shall be attached to and made an integral part of the main building.
- Canopies shall have columns, beams, or brackets of sufficient scale to give a visible means of support.
- Clearance under canopies shall not exceed 16', and cantilevered overhangs shall not exceed 15 feet.
- Task lighting shall be utilized to reduce light spillage. Intense general lighting under canopies in not allowed.

Fascias - refer to building design elements

Disrupt long horizontal plains with vertical elements.

5. SITE FEATURES:

Fence or hedge of not less than 30" or more than 48" in height along property

lines of each public street (not to interfere with site triangles.) Fence shall be wood picket, or wrought iron pickets with masonry columns. Vehicular screening as required under Section 9.2.5.5.2 of the CLURO may be waived if a fence is used.

Sidewalks of not less than four feet in width connecting sidewalks in public right of way to sidewalks with the building entry.

Sidewalks along the facade with a customer entrance and connecting parking areas and adjacent buildings in order to minimize pedestrian traffic within vehicular drives and parking areas.

Distinguish internal pedestrian walkways from driving surfaces through the use of special materials.

Screen Mechanical equipment, electrical service entries, dumpsters, and equipment (not used by the customer) from the public view. Screening may be by fence, landscaping, or building element, such as a parapet.

Automobile and marine sales and displays shall follow the restrictions of parking areas in the CLURO and the landscaping, signage, and lighting requirements associated with parking lots and circulation. Areas used for storage or display of vehicles to be serviced shall be screened with opaque fencing and/or landscaping.

Signage

No off-premise signs as defined under Section 10.3.28 are permitted in the Gateway Overlay District.

7.6.2.10 Procedure for Filing Appeals

Appeals shall be in accordance with Section 4.3.4 of the CLURO.

7.6.3 <u>L-O Lakefront Overlay District</u>

7.6.3.1 Purpose of the L-O Lakefront Overlay District

The purpose of the L-O Lakefront Overlay District shall be to preserve the historic character of the area and the appearance that the lakefront presents to travelers of the Causeway at night.

7.6.3.2 L-O Permitted Uses

The permitted uses in the L-O District shall be in accordance with the uses permitted in the base zoning districts as listed in the Table of Permitted Uses By Zoning District at the end of this Article, except where proposed construction is in areas of periodic inundation provided below which require a Zoning Permit issued in conjunction with the approval of the Zoning Board.

7.6.3.3 Area Within L-O Lakefront Overlay Districts

Any lot, lot of record, or parcel of land made up of lots, lots of record or combination of lots and portions of lots in single ownership, which are considered to be a single parcel under the provisions of these regulations including the provisions of Article 4, which is within or adjacent to any portion of the Open Space/Recreation district which follows the Lake Pontchartrain shoreline and includes the public harbor at Bayou Castain shall be considered to be a parcel within the L-O Lakefront Overlay District and shall be subject to the regulations of this district.

7.6.3.4 L-O Site Development Regulations

- 1. <u>Front Yard Setback</u> The front yard setback requirement of sites within the L-O District shall be required to be the average of the existing front yard setbacks of the adjacent lots or a minimum of twenty-five (25) feet except in the B-3 base district, in which case the requirements of the base district shall apply.
- 2. <u>Lighting</u> In order to maintain the historic character of the night time view of the lakefront from Lake Pontchartrain and in order to preserve the view of the nighttime sky for evening strollers along the lakefront area, both high and low pressure sodium lights or any lights which produce a color spectrum, other than that produced by the existing incandescent lighting in the area, shall be prohibited.

7.6.4 <u>TOWN CENTER OVERLAY ZONING DISTRICT</u>

STATEMENT OF NEED AND PURPOSE:

A. The purpose of the Town Center Overlay District (TCOD) is to develop an identifiable center of the City of Mandeville with the Trailhead as its nucleus. Its intent is to further define a sense of community and to promote and develop the culture, history, and environment of Mandeville for the betterment of the City. This fully realized Town Center will incorporate a planned and architecturally enhanced area, including, but not limited to, building orientation, scale and human relationship, Street scape, vehicular and

pedestrian movement, services and utilities, and uses necessary to develop the overall fabric of a Town Center.

The area encompassed by this district shall include not only that area adjacent to the Trailhead, but may include additional properties designated as critical to the integration of the Town Center into the community.

B. The Town Center is implemented by use of an overlay district which imposes additional criteria on the underlying zoning district or districts. The boundaries of the Town Center Overlay District shall be shown on the Zoning Map of the City of Mandeville.

C. Applicability.

The Town Center Overlay District shall apply to the new building construction, renovations and modifications which require a building permit and which are located within the designated boundaries of said district.

7.6.4.1 OBJECTIVES:

In order to achieve the Town Center, the following objectives shall be addressed:

- A. Human scale, realized by building orientation, setback, height and articulation.
- B. Streetscape, including parking, sidewalks, lighting, signs, landscaping, utilities and amenities.
- C. Applicable elements of Old Mandeville and St. Tammany Parish will serve as a model for implementation to the district requirements and restrictions, as identified with the assistance of the Tulane Regional Urban Design Center.

7.6.4.2 PERMITTED USES - AS PERMITTED IN THE TOWN CENTER ZONING DISTRICT

7.6.4.3 PEDESTRIAN AND STREETSCAPE AMENITIES:

A. Purpose.

The purpose of these standards is to promote and improve the pedestrian environment in the Town Center Overlay District through the provision of appropriate amenities.

B. Applicability.

The standards in this section are applicable to all actions proposed within the TCOD which are subject to site plan review. In addition to the materials

regularly submitted for site plan review, the following items shall be incorporated into plans and specifications for a project located in the TCOD.

C. Amenities.

- 1. Amenities shall include but not be limited to bike racks, seating, trash receptacles, lighting, landscaping, signage, sidewalks and fences.
- 2. Amenities shall be required as part of the streetscape, shown on the site plan, and subject to review.

D. Minimum Site Development Criteria.

The following minimum site development criteria shall be utilized for Site designs within the Town Center Overlay District.

- 1. Sidewalks. Minimum six (6') Sidewalks shall be made part of the Streetscape.
 - a. Surfaces for sidewalks shall be consistent with the criteria established by the Trailhead.
 - b. Owner shall provide the city with an adequate servitude to provide for pedestrian passage adjacent to the Street Right of Way of at least twelve feet (12') at ground level and not less than 12' structural height.

2. Landscaping

Landscaping shall be an architectural element and subject to review.

- b. Landscaping will be utilized to the fullest extent possible as part of the streetscape elements
- 15. Landscaping shall be placed in planter sections between the edge of street (curb) and the sidewalk. This landscape area shall be incorporated into the servitude of passage.
- 16. Required trees shall be planted in a minimum 25 sf, 5' minimum on one side planter sections. The location of these sections will be subject to site plan review.

E. Signs:

- 1. Signs shall be considered an architectural element and subject to review..
- 2. Signs shall be a maximum of 15 sf; no interior lit signs; bottom of sign over sidewalk must have a minimum nine (9') foot clearance above the sidewalk.

F. Lighting.

- 1. Lighting shall be considered an architectural element and subject to review.
- 2. Lighting shall comply with the style and specifications of the Trailhead, Gerard Street Corridor and comply with the CLURO.

7.6.4.4 Parking.

- A. All onsite parking shall be located to the rear of the structure, and should be designed to be shared with adjacent property with shared access.
- B. Access drives and alleyways shall be accessible to adjacent properties.
- C. 50% of available on-street parking (parallel or angle) immediately adjacent to the site can be counted by right. The remaining 50% of the available on-street parking immediately adjacent to the site can be counted with a contribution to the parking mitigation fund in the amount of \$1200.00 per space.
- D. The owner shall provide the city with a servitude to provide for parking and pedestrian passage adjacent to any Street right-of-way of at least twelve feet (12') at ground level and not less than 12' structural height.
- Should the owner desire to provide the City with additional servitude to accommodate 60 degree angle parking, and the Architectural Review Committee agrees that this is compatible with parking on adjacent parcels, then the additional parking spaces can be counted towards the minimum parking requirement.
- 6. The minimum required number of Parking spaces shall be in accordance with Article 9.1.4 of the CLURO. Section 6.4.67 Shopping Center, Neighborhood, for commercial uses and Section 6.2 for Residential uses. For each on-street parking space provided, over the 50% that can be counted by right the owner shall contribute to the City, a sum of \$1200.00 per pace

7.6.4.4 5 Building Setbacks Requirements-

A. Building setback requirements shall be in accordance with Overlay District Guidelines. (Visual Performance Standards) See attached.

7.6.4.5 6 Permitted Uses:

a. Permitted Uses shall be in accordance with CLURO Section 7.7 Table of Permitted Uses for the T-C, Town Center District.

7.6.4.6 7 Architectural Review Standards.

- A. Purpose. The purpose of these standards is to achieve an integrated design that provides an architectural and visual environment consistent with the town center concept.
- B. Applicability. This section is applicable to new building construction and building exterior renovations/modifications which require a building permit.

- C. Minimum performance criteria. In order to determine that new building construction or building exterior modifications contribute to a harmonious effect in the Town Center Overlay District and promote a cohesive architectural appearance, the following minimum performance criteria shall apply:
 - 1. Materials. Traditional materials are generally required in the Town Center; however, contemporary materials may be considered if they are treated in a manner complementary to the concept of the Town Center.
 - 2. Mechanical and electrical equipment. Mechanical equipment shall be screened, subject to review.
 - 3. Architectural features shall be in accordance with the Visual Performance Standards, which are attached hereto and are made a form of this ordinance.
 - 4. Visual Performance Standards shall include but not be limited

to:

- a. Building setbacks, including porches and balconies
- b. Building height (not to exceed 35' as per the CLURO)
- c. Materials
- d. Amenities

See attached Visual Performance Standards

- 7.6.4.8 Architectural Review shall be performed by Tulane University, Regional Urban Design Center or alternative successor review body designated by the City of Mandeville.
- 7.6.4. 9 Application procedures shall comply with Article 5 of the CLURO.
- 7.6.4.10 Procedure for Filing Appeals
 - 1. Appeals shall be in accordance with Section 4.3.4 of the CLURO
- 7.7 TABLE OF PERMITTED USES BY ZONING DISTRICT
- 7.7.1 Table of Permitted Uses By Zoning District

The Table of Permitted Uses By Zoning District provides a listing of the specific use classifications and an indication of the zoning district in which each use is permitted by-right,

permitted with administrative approval, permitted in conjunction with the approval of the Zoning Board or in which the use requires a conditional use permit approved by the City Council by ordinance.

7.7.2 Use of Symbols in Table of Permitted Uses

The following is a list of the symbols used in the Table of Permitted Uses to represent the procedure required for the placement of the use within the designated zoning district:

- P Permitted A use as defined in Article 6 permitted by-right in the designated zoning district.
- A Administrative Use A use permitted in the designated zoning district with the issuance of an Administrative Permit by the City Planner or Designee in accordance with the procedure provided in Article 4, except in cases when the Zoning Board has jurisdiction over the issuance of an Administrative Permit as provided below.
- Z Zoning Permit Use A use permitted-conditionally in the designated zoning district with issuance of a Zoning Permit by the Zoning Board in accordance with procedures as provided in Article 4, based on standards applicable to the use.
- C Conditional Use A use permitted in the designated zoning district with the approval of the City Council by ordinance in accordance with the procedure for Conditional Use Permits and Planned Districts as provided in Article 4.
- * Special Development Criteria Uses shown with an asterisk are uses that are permitted in accordance with the guidelines of the Special Use Criteria provided in Article 8 of this Land Use Regulations Ordinance for the regulation of the specific use.
- ** Sign Code Accessory uses shown with two asterisks and identified as onpremise and off-premise signs shall be regulated in accordance with the provisions of Article 10 - Sign Code of this Land Use Regulations Ordinance.

7.7.3 Zoning Board Jurisdiction Over Administrative Use Permits

Notwithstanding any other provisions of these regulations, whenever the Table of Permitted Uses provides that a use is permissible with an Administrative Permit, Zoning Board approval shall nevertheless be required if the City Planner or Designee finds that the proposed use would have an extraordinarily adverse impact on neighboring properties or the general public. In making this determination, the Administrator shall consider, among other factors, whether the use is proposed for an undeveloped or previously developed lot, whether the proposed use constitutes a change from one principal use classification to another, whether the use is proposed for a site that poses peculiar traffic or other hazards or difficulties, and whether the proposed use is substantially unique or is likely to have impacts that differ substantially from those presented by other uses that are permissible in the zoning district in question.

ARTICLE 8 - SUPPLEMENTAL DEVELOPMENT AND USE REGULATIONS

8.1. <u>SUPPLEMENTAL REGULATIONS - GENERAL</u>

These provisions shall be considered as supplemental to all other provisions of the regulations established by this Land Use Regulations Ordinance. In event of any conflict in provisions, the more restrictive provision shall apply, unless specifically indicated to the contrary.

8.1.1 Supplemental Lot, Yard and Open Space Regulations

- 1. More than one (1) main institutional, public or semi-public or commercial or industrial building may be located upon a lot or tract in districts which permit such uses provided that no such building or portion thereof shall be located outside of the buildable area of the lot and no building shall be closer than ten (10) feet to any other building unless approved by the State Fire Marshall in conformance with all fire codes applicable to an addition to the building within the ten (10) foot distance.
- 2. Every part of a required yard shall be open to the sky and unobstructed by accessory structures except where accessory buildings are permitted in a rear or side yard and except for the ordinary projection of sills, belt courses, cornices, and ornamental features projecting into the required yard no more than eighteen (18) inches, provided however, that a roof, gutter, eaves and raised landings, such landings not exceeding four (4) feet by six (6) feet in area, may project to the extent of four (4) feet into a required side yard if a minimum distance of three (3) feet remains open to the sky between the farthest projection of such roof, gutter, eaves or landing and any interior property line.

- 3. An unenclosed carport or canopy which is attached to the primary building and has no other support may project into a required side yard provided that every part of the projection of such carport is removed at least three (3) feet from the nearest side lot line and does not extend more than twenty-five (25) feet in length or more than thirteen (13) feet in height; provided further that canopies can be located in required front or side yards adjacent to streets on lots occupied by churches, schools, day care centers, hospitals, clinics, funeral parlors, hotels, public buildings and institutions of a philanthropic, educational, religious or eleemosynary nature and such canopies can be supported by other means than attached to the buildings.
- 4. An open, unenclosed, uncovered porch or terrace not exceeding the ground elevation by more than six (6) inches, may project into a required front yard a distance not more than ten (10) feet, but in no case more than half the distance from the required building line to the front property line. This shall not be interpreted to include or permit fixed marquees or canopies except where otherwise provided herein.
- 5. In R-1, R-1X and R-2 districts, if significant and otherwise healthy trees, unusual topography or other desirable physical features of the site would be damaged or required to be removed by building within the buildable area of a site, the buildable area of the site may be shifted into the front or rear yard by up to five (5) feet so long as the area lost in one required yard is provided in the opposite yard and the total square footage of required yard and open space area of the site remains the same.
- 6. No structure shall be constructed or placed within a utility servitude without prior written approval of the affected utilities. Approval may be withheld by any utility upon its determination that the proper size or location of the structure would adversely affect the operation, maintenance or function of the servitude. Approval of the construction by a utility shall create no obligation to repair or replace a structure damaged or removed by the utility in the course of its lawful use of the servitude.
- 7. Obstruction to Vision at Street Intersections Prohibited On corner lots in all districts, within the area formed by the center line of the intersecting streets and a line connecting points on such center lines at a distance of eighty (80) feet from their intersection, there shall be no obstruction to vision between the height of three (3) feet and a height of seven (7) feet above the average grade of each Street at the center line thereof. The requirements of this section shall not be deemed to prohibit the preservation of any tree or the

construction of retaining walls that may be required for tree preservation unless said retaining wall is more than three (3) feet in height.

8.1.2 Supplemental Use Regulations

8.1.2.1 Exception to Height Limitations

8.1.2.1.1 Non-Occupancy Structures

When approved in conjunction with an application for a zoning permit from the Zoning Board after a determination that the extension of such structures beyond the height elevations of the district will not create an adverse effect on the surrounding community or significantly affect the stated character of the city, and when FAA height regulations and requirements for warning signals have been complied with, the height limitations of this Ordinance shall not apply to:

- 1. Chimneys, church spires, cupolas, elevator shafts and similar ancillary structures that are not intended for occupancy or storage,
- 2. Flagpoles, firetowers, monuments, observation towers, radio and television towers, smoke stacks, transmissions towers or water towers or other similar devices,
- 3. Heating and air conditioning equipment, solar collectors and similar equipment, fixtures and devices, when located on the roof of a building and not more than one-third of the roof area is consumed by such features and if the features are set back from the edge of the roof at least one foot for every foot of projection above the roofline.

8.1.2.1.2 Buildings Exceeding Thirty-five (35) Feet in Height

No residential or non-residential building or structure shall exceed thirty-five (35) feet in height except as provided by this Land Use Regulations Ordinance and only in conjunction with the fire department of the district in which the structure is located certifying to the city that the building is designed or equipped to provide adequate protection against fire. Construction of any building or structure which exceeds forty (40) feet in height at the highest point of the structure, using the center line of the Street on which the premises fronts as the base level for measurement, shall require zoning permit approval.

8.1.2.2 Supplemental Regulations for Antennas, Aerials and Satellite Receiving Stations

- 1. No antennas, aerials or satellite receiving stations shall be constructed except as an accessory to an existing or concurrently constructed building.
- 2. Antennas, aerials and satellite receiving structures shall be situated to the rear of the front building facade and shall be situated in such portion of the yard as to be screened from view of the Street right-of-way or shall be situated on such a portion of a roof as not to be visible from the Street.
- 3. No antenna, aerial or satellite receiving station or any of its anchoring or supporting appurtenances, including guy wires, shall be constructed or located within ten (10) feet of any lot line.
- 4. No more than one (1) satellite receiving station shall be constructed on any lot development site as defined in Article 3.
- 5. Antennas, aerials or satellite receiving stations shall be constructed of nonreflective and noncombustible materials and shall be free of any lettering, slogans, logos or similar marks or messages.
- 6. All antennas, aerials or satellite receiving stations shall be wired and grounded in conformity with the requirements of the National Electric Code.
- 7. No antennas, aerials or satellite receiving station shall contain more than one (1) circular or parabolic collecting structure which shall be no greater than twelve feet in diameter. No antennas, aerials or satellite receiving station, including its base, shall be constructed to a height in excess of the height limitations of the zoning district in which it is located.
- 8. Any antennas, aerials or satellite receiving station constructed on any lot or parcel of ground which abuts residentially zoned property shall be screened along those sides visible from any such abutting residential property with living materials of such type and nature as shall grow to be a seventy-five (75) percent opaque barrier of a height not less than the actual height of the satellite receiving station being screened within two (2) years of the planting of such living screen except to the extent that screening would interfere with reception.
- 9. All antennas, aerials or satellite receiving stations shall be filtered and/or shielded so as to prevent the emission from such stations of radio frequency

energy that would cause any noticeable interference with the radio and/or television broadcasting or reception capabilities of adjacent properties. In the event that such interference is caused by any of the above subsequent to the granting of a building permit, the owner of such apparatus shall take prompt action to eliminate the interference. In addition to all other penalties which may be applicable to a violation of this section, any owner or user of such apparatus who fails to so filter or shield his apparatus may be required by order of any competent court, to disconnect and dismantle same.

- 10. No antennas, aerials or satellite receiving station shall be constructed nearer than five (5) feet horizontal distance and ten (10) feet vertical distance from any overhead utility line.
- 11. All antennas, aerials or satellite receiving stations shall be constructed in such a manner and of such materials as to withstand a ninety-mile per hour wind and the plans for any proposed apparatus shall be so certified by a civil engineer licensed to practice his profession in the state of Louisiana prior to the issuance of any building permit.
- 12. No mobile or portable antennas, aerials or satellite receiving station shall be used or allowed in any district except on a temporary basis in connection with special events and civil emergencies.
- 13. In the proposed placement of antennas, aerials or satellite receiving station, if it is required that a tree be removed as a result of the proposed placement, the applicant for the permit shall be required to show that there is no reasonable alternative or placement in another location that would allow preservation of the tree.

8.1.2.2.1 Communication Tower Application Requirements

- 1. Application Requirements. The applicant for an Administrative Permit for construction of a Communications Tower or placement of a Commercial Telecommunication Antenna on an existing structure other than a tower previously permitted must file an application accompanied by a fee as set by the fee schedule for Administrative and Zoning Permits under the CLURO and the following documents, if applicable:
 - 1. One copy of typical specifications for proposed structures and antennae, including description of design characteristics and material.

- 2. A site plan drawn to scale showing property boundaries, tower location, tower height, guy wires and anchors, existing structures, photographs or elevation drawings depicting typical design of proposed structures, parking, fences, landscape plan, and existing land uses on adjacent property. A site plan is not required if antenna is to be mounted on an approved existing structure.
- 3. A current map, or update for an existing map on file, showing locations of applicant's antennae, facilities, existing towers, and proposed towers serving any property within the City.
- 4. A report from a structural engineer showing the tower antenna capacity by type and number, and a certification that the tower is designed to withstand winds in accordance with ANS/EIA/TIA 222, latest revision, standards.
- 5. Identification of the owners of all antennae and equipment to be located on the site.
- 6. Written authorization from the site owner allowing applicant to submit the application.
- 7. Evidence that a valid FCC license for the proposed activity and site has been issued.
- 8. A line of sight analysis, i.e. photo enhancement via graphic design, showing the potential visual and aesthetic impacts on adjacent residential districts or combined use districts.
- 9. A written agreement to remove the tower and/or antenna within 180 days after cessation of use.
- 10. Additional information as required to determine that all applicable zoning regulations are met.
- 11. A tower is not allowed if technically suitable space can be found on an existing communications tower or other structure within the search area that the new site is to serve.
- 12. Existing on-site vegetation shall be preserved to the maximum extent practicable. All towers shall be landscaped with at least one row of evergreen shrubs or trees and shrubs within 20 feet of the tower's base. Such plantings shall be in accordance with the minimum requirements of Section 9.2.5.5.3 Buffer Zone Requirements of the CLURO.

- 13. The entire facility must be aesthetically and architecturally compatible with its environment. The use of residentially compatible materials such as wood, brick, or stucco is required for associated support buildings, which shall be designed to architecturally match the exterior of residential structures within the neighborhood. In no case will metal buildings be allowed for accessory buildings.
- 14. Walls or fences constructed of wood, brick, or masonry shall be used to secure the site and provide an opaque barrier. Such walls or fences may be used in combination with landscaping to provide security or increase the buffer to other land uses. Other types of fences shall be allowed only if used in conjunction with evergreen shrubs or hedges when upon maturity are equal to or greater than the height of the fence for the purposes of providing an opaque barrier.
- 15. The cellular transmission tower, antennas, and support structures shall be constructed of a material with a neutral color and shall be designed to blend in with the surrounding landscape, buildings and uses.
- 2. Conditions. Applicant must show that all applicable conditions are met.
 - 1. The proposed communications tower, antenna or accessory structure will be placed in a reasonably available location which will minimize the visual impact on the surrounding area and allow the facility to function in accordance with minimum standards imposed by applicable communications regulations and applicant's technical design requirements. Priority shall be given applications which desire locations on existing structure, property zoned for industrial us and public property
 - 2. Applicant must show that a proposed antenna and equipment cannot be accommodated and function as required by applicable regulations and applicant's technical design requirement without unreasonable modifications on any existing structure or tower under the control of or reasonably available to the applicant.
 - 3. Applicant for a permit in a residential district must show that the area cannot be adequately served by a facility placed in a non-residential district for valid technical reasons.
 - 4. Prior to consideration of a permit for location on private property which must be acquired, applicant must show that available City owned sites, and available

privately owned sites occupied by a compatible use, are unsuitable for operation of the facility under applicable communications regulations and applicant's technical design requirements.

- 5. Applicant must show that a new tower is designed to accommodate additional antennae equal in number to applicant's present and future requirements. Applicant is encouraged to anticipate any co-location agreements that can be made with other users and design the tower to accommodate said users.
- 6. Applicant must show that all applicable health, nuisance, noise, fire, building and safety code requirements are met.
- 7. A permit for a proposed tower site within 1,000' of an existing tower shall not be issued unless the applicant certifies that the existing tower does not meet applicant's structural specifications and applicant's technical design requirements, or that a collocation agreement could not be obtained.
- 8. Applicant must show by certificate from a registered engineer that the proposed facility will contain only equipment meeting FCC rules, and must file with the Department of Planning & Development a written indemnification of the municipality and proof of liability insurance or financial ability to respond to claims up to \$1,000,000 in the aggregate which may arise from operation of the facility during its life, at no cost to the municipality, in form approved by the City Attorney.
- 9. Site development regulations, visibility, fencing, screening, landscaping, parking access, lot size, exterior illumination, sign, storage and all other general zoning district regulations except setback and height, shall apply to the use. Setback and height conditions in this section apply.

3. Abandonment

In the event the use of any communication tower has been discontinued for a period of 180 consecutive days, the tower shall be deemed to be abandoned. Determination of the date of abandonment shall be made by Director of Planning who shall have the right to request documentation and /or affidavits from the communication tower owner/operator regarding the issue of tower usage. Upon such abandonment, the owner/operator of the tower shall have an additional 180 days within which to: (1) reactivate the use of the tower or transfer the tower to another owner/operator who makes actual use of the tower, or (2) dismantle and remove the tower. At the earlier of 181 days from the date of abandonment without reactivation or upon completion of dismantling and removal, any special exception and/or variance approval or the tower shall automatically expire.

4. Denial of permit.

The Telecommunications Act of 1996 requires that a denial of a permit be supported by substantial evidence. The denial of any permit may be appealed under Section 4.3.4.1 of the CLURO.

5. Severability.

If any of the terms or conditions of this ordinance shall for any reason be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect the other provisions, and the provisions hereof shall thereafter be construed as if such invalid, illegal or unenforceable terms or conditions had never been contained herein.

8.1.3 Supplemental Fence and Wall Regulations

- 1. In all zoning districts, no fence or wall shall be erected to exceed the height of four (4) feet in the required front yard area as defined and, for purposes of this provision, in the area from the front of the structure to the front property line. Rear and side fences or walls from the front line of the structure to the rear property line shall not exceed seven (7) feet in height from average between the primary structure and the fence grade. No fence or wall shall be erected or constructed along a side Street property line or within a required Street side yard exceeding four (4) feet in height. No fence or wall shall obstruct the view of traffic at an intersection between a height of three (3) and seven (7) feet.
- 2. Electrical fences in any form are prohibited.
- 3. Barb wire for fences shall be prohibited in all districts except B-2, B-4, M-1 or M-2 and when allowed shall only be used on fences in an area higher than six (6) feet above ground.
- 4. No fence or wall shall be constructed within a utility servitude without prior written approval of the affected utilities. Approval may be withheld by any utility upon its determination that the proper size or location of the fence or wall would adversely affect the operation, maintenance or function of the servitude. Approval of the construction by a utility shall create no obligation to repair or replace a fence or wall damaged or removed by the utility in the course of its lawful use of the servitude.

8.1.4 Supplemental Regulations for Temporary Buildings

Mobile buildings and portable buildings shall not be used as principal buildings except in accordance with these regulations for temporary structures:

- 1. <u>Temporary Construction Buildings</u> Temporary buildings used in connection with construction work and only in conjunction with a valid development permit may be located in any district during the period of construction, but such temporary building shall be removed on completion of the construction work and prior to issuance of the permanent certificate of occupancy for the structure for which the permit was issued.
- 2. <u>Temporary Seasonal Buildings</u> Temporary buildings used in connection with the sale of Christmas trees may be placed on a site in conjunction with the issuance of a temporary development permit during the months of November and December only and shall be removed from the site during the remaining months of the year.
- 3. <u>Temporary Structures in Parking Lots</u> Temporary structures, such as those used for tent sales, may be placed no more frequently than twice annually for a maximum of seven (7) days on any development site in conjunction with the issuance of a temporary permit so long as not more than five (5) percent of the required parking spaces are affected by the placement of the structure.

8.1.5 Supplemental Regulation of Accessory Buildings and Structures

For purposes of these regulations accessory buildings include but are not necessarily limited to accessory storage buildings, pool cabanas, detached covered porches or decks, playhouses, private studios or craft buildings, and greenhouses and shall be regulated as follows:

- 1. Except on corner lots, any accessory building that is not a part of the main building may be built in a required side yard, provided such accessory building is not less than sixty (60) feet from the front lot line nor less than three (3) feet from the nearest interior side lot line and provided not more than one (1) accessory building covers any part of the required side yard.
- 2. On corner lots, accessory buildings are not permitted in required side yards on the side Street side or within any portion of the rear yard area which lies between the side Street and the prolongation of the required side yard line into the rear yard area.

- 3. Detached accessory buildings not exceeding one (1) story nor fourteen (14) feet in height may be built in required rear yards; provided, however, such accessory buildings shall not be located less than three (3) feet from either side or rear lot lines.
- 4. The combined gross area of all accessory buildings or portions thereof located in required side and rear yards shall not exceed thirty percent of the required rear yard area.
- 5. None of the provisions of this sub-section shall apply to the use, construction or location of antennas, aerials or satellite receiving stations.
- 6. Accessory recreational structures such as pools and tennis courts shall be subject to the locational requirements of this sub-section.
- 7. No accessory building or structure shall be constructed within a utility servitude without prior written approval of the affected utilities. Approval may be withheld by any utility upon its determination that the proper size or location of the accessory building or structure would adversely affect the operation, maintenance or function of the servitude. Approval of the construction by a utility shall create no obligation to repair or replace an accessory building or structure damaged or removed by the utility in the course of its lawful use of the servitude.

8.1.6 Supplemental Regulations for Manufactured Modular Buildings

Manufactured modular buildings as defined in Article 3 may be used as principal buildings and shall be permitted and regulated in accordance with all provisions applicable to structures other than manufactured modular buildings.

8.1.7 Supplemental Regulations for Recreational Vehicles

Recreational vehicles, as defined in Article 3 and specifically excluding mobile homes, mobile buildings or manufactured housing as defined, may be parked in any district in accordance with the following provisions:

- 1. The site on which a recreational vehicle is parked shall be a previously developed site; recreational vehicles shall be prohibited from parking on vacant and undeveloped sites.
- 2. Recreational vehicles parked on a developed site shall be for personal and private use of the occupants of the site and shall not be occupied as living

quarters or for business purposes on a permanent or part-time basis except in the case of a temporary visitor to the permanent occupants of the site on which the vehicle is parked. In the case of a temporary visitor a recreational vehicle may be parked on a site and occupied by a visitor for a maximum of seventy-two (72) hours without an occupancy permit or for a maximum of two (2) weeks upon issuance of a temporary certificate of occupancy by the Building Inspector. Temporary occupants of recreational vehicles as provided above, however, when located in an R-1, R-1X or R-2 district shall be prohibited from operating generators or other noise producing mechanisms which may disturb the peace of the residential area.

- 3. Recreational vehicles shall not be procured and parked on the site for sales purposes. However, personal recreational vehicles utilized by the occupants of the site on which the vehicle is parked may sell the vehicle from the site so long as no more than one sale of such a personally owned vehicle occurs in any six (6) month period.
- 4. Recreational vehicles parked on a site shall be situated on the site outside of the required front yard area when the placement of structures on the site permits, however, under no circumstances shall a recreational vehicle be parked so that any part of the vehicle encroaches onto a Street right-of-way.

8.1.8 Standards For Temporary Residences on Construction Sites

- 1. A temporary permit for a mobile home, mobile building or recreational vehicle to be used as a temporary residence to be occupied during the construction, repair or renovation of a permanent non-residential building on a site may be approved as a conditional use by the City Council in accordance with procedures for Conditional Use Permits as provided in Article 4 for a period not to exceed six months from the date the temporary permit is issued.
- 2. The temporary conditional use may be granted one extension of an additional period of not more than three months, by the building inspector if such extension is necessary to complete the construction, repair or renovation work.
- 3. Temporary residences used on construction sites shall be removed immediately upon completion of the project and prior to the issuance of a certificate of occupancy for the building constructed on the site.

8.1.9 Construction of Buildings on Improved Streets Only

Every building hereafter erected, reconstructed, converted, moved, or structurally altered shall be located on a lot of record which abuts at least one improved Street.

8.1.10 SUPPLEMENTAL REGULATIONS FOR OUTDOOR LIGHTING

8.1.10.1 STATEMENT OF NEED AND PURPOSE:

Good outdoor lighting at night benefits everyone. It increases safety, enhances the City's night time character, and helps provide security. New lighting technologies have produced lights that are extremely powerful, and these types of lights may be improperly installed so that they create problems of excessive glare, light trespass, and higher energy use. Excessive glare can be annoying and may cause safety problems. Light trespass reduces everyone's privacy, and higher energy use results in increased costs for everyone. Appropriately regulated, and properly installed, outdoor lighting will contribute to the safety and welfare of the residents of the City. It is intended to eliminate problems of glare, minimize light trespass, and help reduce the energy and financial costs of outdoor lighting, by establishing regulations which limit the area that certain kinds of outdoor-lighting fixtures can illuminate, and by limiting the total allowable illumination of lots located in the City of Mandeville. Luminaries on all properties, in all zoning districts, shall be installed with the idea of being a "good neighbor", with attempts to keep unnecessary direct light from shining onto abutting properties or streets.

8.1.10.2 DEFINITIONS:

For the purposes of this Ordinance, terms used shall be defined as follows:

8.1.10.2.1

<u>Direct Light</u>: Light emitted directly from the lamp, off of the reflector or reflector diffuser, or through the refractor or diffuser lens, of a luminaire.

8.1.10.2.2

<u>Disability Glare</u>: Glare resulting in reduced visual performance and visibility. It is often accompanied by discomfort.

8.1.10.2.3

<u>Fixture</u>: The assembly that houses the lamp or lamps and can include all or some of the following parts: a housing, a mounting bracket or pole socket, a lamp holder, a ballast, a reflector or mirror, and/or a refractor or lens.

8.1.10.2.3(a)

<u>Flat, Single-Plan Lens:</u> A refractor of lens, mounted in the horizontal plane, which by design, allows direct light to be emitted only through the horizontal plane.

8.1.10.2.4

<u>Flood or Spot light</u>: Any light fixture or lamp that incorporates a reflector or a refractor to concentrate the light output into a directed beam in a particular direction.

8.1.10.2.5

<u>Footcandle</u>: A unit of measure for illuminance. A unit of illuminance on a surface that is everywhere one foot from a uniform point source of light of one candle and equal to one lumen per square foot.

8.1.10.2.6

<u>Full Cut-Off Type Fixture</u>: A luminaire or light fixture that; by design of the housing, does not allow any light dispersion or direct glare to shine above a 90 degree, horizontal plane from the base of the fixture. Full cut-off fixtures must be installed in a horizontal position as designed, or the purpose of the design is defeated, and disability glare will result.

8.1.10.2.6(a)

<u>Fully Shielded Fixture</u>: A luminaire or fixture constructed in such a manner that an opaque shield extends, on all sides, below the lowest direct-light-emitting part (LDLEP) of the luminaire. The lowest edge of such a shield shall surround the LDLEP and be level with the horizontal plane, regardless of the orientation of the luminaire or fixture.

8.1.10.2.7

<u>Glare</u>: Light emitting from a luminaire with an intensity great enough to reduce a viewer's ability to see, and in extreme cases causing momentary blindness.

8.1.10.2.8

<u>Grandfathered luminaries</u>: Luminaries not conforming to this code that were in place at the time this code was adopted into effect. When an ordinance "grandfathers" a luminaire, it means that such already-existing outdoor lighting does not need to be changed unless a specified period is specified for adherence to the code.

8.1.10.2.9

<u>Height of Luminaire</u>: The height of a luminaire shall be the vertical distance from the ground directly below the centerline of the luminaire to the lowest direct-light-emitting part of the luminaire.

8.1.10.10

<u>Horizontal Illuminance</u>: The measurement of brightness from a light source, usually measured in footcandle or lumens, which is taken through a light meter's sensor at a horizontal position.

8.1.10.2.11

<u>Indirect Light</u>: Direct light that has been reflected or has scattered off of other surfaces. 8.1.10.2.12

<u>Lamp</u>: The component of a luminaire that produces the actual light.

8.1.10.2.12(a)

<u>Lowest Direct-Light Emitting Part (LDLEP)</u>: The lowest part of either the lamp or lamps, the reflector or mirror, and/or refractor of lens.

8.1.10.2.13

<u>Light Trespass</u>: The shining of light produced by a luminaire beyond the boundaries of the property on which it is located.

8.1.10.2.14

<u>Lumen</u>: A unit of luminous flux. One footcandle is one lumen per square foot. For the purposes of this Ordinance, the lumen-output values shall be the INITIAL lumen output ratings of a lamp.

8.1.10.2.15

<u>Luminaire</u>: This is a complete lighting system, and includes a lamp or lamps and a fixture.

8.1.10.2.16

<u>Outdoor Lighting</u>: The night-time illumination of an outside area or object by any man-made device located outdoors that produces light by any means.

8.1.10.2.17

<u>Temporary outdoor lighting</u>: The specific illumination of an outside area of object by any man-made device located outdoors that produces light by any means for a period of less than 7 days, with at least 180 days passing before being used again.

8.1.10.2.18

<u>Uplighting</u>: Any light source that distributes illumination above a 90 degree horizontal plane.

8.1.10.3 REGULATIONS:

All public and private outdoor lighting installed in the City of Mandeville shall be in conformance with the requirements established by this Ordinance. All previous language in Mandeville bylaws and ordinances regarding outdoor lighting is replaced with this ordinance.

8.1.10.3.1 CONTROL OF GLARE -- LUMINAIRE DESIGN FACTORS:

a. <u>Luminaire Design</u>

Any luminaire with a lamp or lamps rated at an average of MORE than 3000 lumens shall be either:

Full Cutoff Type Fixture with a flat, single plane lens, or Fully-Shielded Fixture

b. Luminaire Height

- 1. Any luminaire with a lamp or lamps rated at an average of LESS than 3000 lumens may be mounted at any height.
- 2. Any luminaire with a lamp or lamps rated an average of MORE than 3000 lumens may be mounted up to a height of 35 feet in accordance with the following:

HEIGHT	FULL CUTOFFS WITH SINGLE-PLANE LENS	FULLY SHIELD FIXTURE
0-25 feet	Acceptable	Shield EVEN with LDLEP
>25-30 feet	Acceptable	Shield 1 INCH below LDLEP
>30-35 feet	Acceptable	Shield 2 INCHES below LDLEP

c. Total footcandles shall be in accordance with the following standards:

Building exteriors		
Entrances Active (pedestrian and/or conveyance)	5	
Inactive (pedestrial and/of conveyance) Inactive (normally locked, infrequently used)	1	
Vital locations or structures	5	
Building surrounds	1	
Floodlit Buildings and Monuments	10	
S .	20	
Loading and Unloading Platforms	20	
Service Stations		
Approach	1.5	
Driveway	1.5	
Pump Island		20
Service Areas	3	
Storage Yards		
Active	20	
Inactive	1	
mactive	1	
Retail Outdoor Lighting	10	
8.1.10.3.2 RESIDENTIAL APPLICATIONS	-	

- a. Spotlights and floodlights shall be angled so that the center of the beam will strike the ground within the said property line.
- 2. Any exterior lighting that can be reasonably expected to create a nuisance to the adjacent neighboring properties is prohibited.

8.1.10.3.3 EXCEPTIONS TO CONTROL OF GLARE:

a. Luminaire Redirection

Any luminaire with a lamp or lamps rated at an average of 3000 lumens or LESS may be used without restriction to light distribution or mounting height, except that if any spot of flood luminaire rated 3000 lumens or LESS is aimed, directed, or focused such as to cause direct light from the luminaire to be directed toward residential buildings on adjacent or nearby land, or to create glare perceptible to persons operating motor vehicles on public ways, the luminaire shall be redirected or its light output controlled as necessary to eliminate such conditions.

b. Public-Roadway Illumination

Luminaries used for public-roadway illumination shall be installed at a maximum height of 25 feet and shall be positioned at that height up to the edge of any bordering property.

c. Police or Fire Departments or other Emergency Services

All temporary emergency lighting need by the Police or Fire Departments or other emergency services, as well as all vehicular luminaries, shall be exempt from the requirements of this article.

d. Federal Regulatory Agencies

All hazard warning luminaries required by Federal regulatory agencies are exempt from the requirements of this article, except that all luminaries used must be red and must be shown to be as close as possible to the Federally required minimum lumen output requirement for the specific task.

e. Law Governing Conflicts

Law Governing Conflicts. Where any provision of federal, state, parish, or City statutes, codes, or laws conflicts with any provision of this code, the most restrictive shall govern unless otherwise regulated by law.

f. Flags

Up-Lighting for flags are exempt from the requirements of this article.

8.1.10.3.4 OUTDOOR ADVERTISING SIGNS.

a. <u>Top Mounted Fixtures</u>

Top Mounted Fixtures Required. Lighting fixtures used to illuminate an outdoor advertising sign shall be mounted on the top of the sign structure. All such fixtures shall comply with the shielding requirements of Section 8.1.10.3.1 with the exception of the portion of the luminaire parallel with the sign, provided this portion does not allow any light dispersion or direct glare to shine above a 90 degree horizontal plane.

b. <u>Translucent Outdoor Advertising Signs</u>

Outdoor advertising signs of the type constructed of translucent materials and wholly illuminated from within do not require shielding. Dark backgrounds with light lettering or symbols are preferred, to minimize detrimental effects.

c. Compliance Limit

Existing outdoor advertising structures shall be brought into conformance with this Code within five (5) years from the date of adoption of this provision.

d. Prohibitions

Electrical illumination of outdoor advertising off-site signs between the hours of 11:00 p.m. and sunrise is prohibited.

8.1.10.3.5 RECREATIONAL FACILITIES (PUBLIC & PRIVATE)

- 1. Recreational field lighting, public or private, such as, football fields, soccer fields, baseball fields, and softball fields, shall be exempt to the height requirement of 25' and total lumen output provided all of the following conditions are met:
 - 1. Lighting for parking lots and other areas surrounding the playing field, court, or track shall comply with this Code for lighting in the specific Area as defined in Sections 8.1.10.3.1 of this Code.
 - b. All fixtures used for event lighting shall be fully shielded as defined in Section 8.1.10.3.1 of this Code, or be designed or provided with sharp cut-off capability, so as to minimize up-light, spill-light, and glare.
 - 17. All events shall be scheduled so as to complete all activity before or as near to 10:30 p.m. as practical, but under no circumstances shall any illumination of the playing field, court, or track be permitted after 11:00 p.m. except to conclude a scheduled event that was in progress before 11:00 p.m. and circumstances prevented concluding before 11:00 p.m.

8.1.10.3.5 6 PROHIBITIONS.

a. Laser Source Light.

The use of laser source light or any similar high intensity light for outdoor advertising or entertainment, when projected above the horizontal is prohibited.

b. Searchlights.

The operation of searchlights for advertising purposes is prohibited.

c. Outdoor Advertising Off-Site Signs.

Electrical illumination of outdoor advertising off-site signs is prohibited between the hours of $11:00\,$ p.m. and sunrise.

d. Commercial landscape lighting may not be above the 90 degree plane to emit excessive light glare.

8.1.10.3.6 7 TEMPORARY OUTDOOR LIGHTING.

a. Any temporary outdoor lighting that conforms to the requirements of this Ordinance shall be allowed. Nonconforming temporary outdoor lighting may be permitted by the Zoning Commission after considering: (1) the public and/or private benefits that will result from the temporary lighting; (2) any annoyance or safety problems that may result from the use of the temporary lighting; and (3) the duration of the temporary nonconforming lighting. The applicant shall submit a detailed description of the proposed temporary nonconforming lighting to the Community Appearance Commission, who shall consider the request at a duly called meeting of the Community Appearance Commission. Appeal of the Community Appearance Commission's decision shall be heard at the next regularly scheduled Mandeville Zoning Board meeting.

8.1.10.4 EFFECTIVE DATE AND GRANDFATHERING OF NONCONFORMING LUMINARIES:

a. Effective Date

This ordinance shall take effect immediately upon approval by the City Council of the City of Mandeville and shall supersede and replace all previous ordinances pertaining to outdoor lighting.

b. **Grandfathering Luminaries**

All luminaires at a height of 35' and under shall be grandfathered for five years from date of Ordinance 00-13. All luminaires over 35' in height shall be grandfathered for five years from date of Ordinance 00-13, after which they shall be fully shielded 3" below LDLEP.

c. <u>Grandfathering Luminaries Causing Disability Glare</u>

Grandfathered luminaries that direct light toward streets or parking lots that cause disability glare to motorists or cyclists should be either shielded or re-directed within 90 days of notification, so that the luminaries do not cause a potential hazard to motorists or cyclists.

8.1.10.5 NEW CONSTRUCTION.

a. Submission Contents.

The applicant for any permit required by any provision of the laws of this jurisdiction in connection with proposed work involving outdoor lighting fixtures shall submit (as part of the application for permit) evidence that the proposed work will comply with this Code. The submission shall contain but shall not necessarily be limited to the following, all or part of which may be part or in addition to the information required elsewhere in the laws of this jurisdiction upon application for the required permit:

- (1) plans indicating the location on the premises, and the type of illuminating devices, fixtures, lamps, supports, reflectors, and other devices;
- (2) description of the illuminating devices, fixtures, lamps, supports, reflectors, and other devices and the description may include, but is not limited to, catalog cuts by manufacturers and drawings (including sections where required);
- (3) photometric data, such as that furnished by manufacturers, or similar showing the angle of cut off or light emissions.
- b. <u>Additional Submission</u>. The above required plans, descriptions and data shall be sufficiently complete to enable the reviewer to readily determine whether compliance with the requirements of this Code will be secured. If such plans, descriptions and data cannot enable this ready determination, by reason of the nature or configuration of the devices, fixtures, or lamps proposed, the applicant shall additionally submit as evidence of compliance to enable such determination such certified reports of tests as will do so provided that these tests shall have been performed and certified by a recognized testing laboratory.

- c. <u>Subdivision Plat Certification</u>. If any subdivision proposes to have installed Street or other common or public area outdoor lighting, the preliminary and final plat shall contain a statement certifying that the applicable provisions of the City of Mandeville Outdoor Lighting Code will be adhered to.
- d. <u>Lamp or Fixture Substitution</u>. Should any outdoor light fixture, or the type of light source therein, be changed after the permit has been issued, a change request must be submitted to the building official for his approval, together with adequate information to assure compliance with this code, which must be received prior to substitution.

8.1.10.6 NOTIFICATION REQUIREMENTS:

- a. The City of Mandeville building permit shall include a statement asking whether the planned project will include any outdoor lighting.
- b. Within 30 days of the enactment of this ordinance, the Building Inspector or his designee shall send a copy of the Outdoor Lighting Ordinance, with cover letter to all local electricians and local electric utility (including at least those in the City of Mandeville, as listed in the Yellow Pages).

8.1.10.7 VIOLATIONS, LEGAL ACTIONS, AND PENALTIES:

a. Violation.

It shall be a civil infraction for any person to violate any of the provisions of this Code. Each and every day during which the violation continues shall constitute a separate offense.

b. Violations and Legal Actions:

If, after investigation, the Building Inspector or his designee finds that any provision of the Ordinance is being violated, he shall give notice by hand delivery or by certified mail, return-receipt requested, of such violation to the owner and/or to the occupant of such premises, demanding that violation be abated within thirty (30) days of the date of hand delivery or of the date of mailing of the notice. If the violation is not abated within the thirty-day period, the Building Inspector or his designee may institute actions and proceedings, either legal or equitable, to enjoin, restrain, or abate any violations of this Ordinance and to collect the penalties for such violations.

c. Penalties:

Any person violating any provision of these regulations shall be guilty of a misdemeanor, and deemed a public nuisance and upon conviction shall be punished for each separate offense by a fine not exceeding five hundred dollars (\$500.00) or imprisonment for a term not exceeding sixty (60) days, or by both such fine and imprisonment, or as provided in Section 1.9 of the Code of Ordinances of

the City of Mandeville, whichever is greater. Each day any violation of any provision of these regulations shall continue shall constitute a separate offense.

8.2 SPECIAL USES CRITERIA

This section shall be known as Special Uses Criteria. These provisions shall apply to the indicated specific use classifications in Article 6. The Planning Commission may, by resolution, adopt guidelines or establish standard practice for appropriate and practical interpretation of special use criteria whenever a literal interpretation of such provisions is impractical.

The provisions of this Section shall be in addition to all other applicable regulations established by this Code. In event of any conflict in provisions, the more restrictive provision shall apply unless specifically indicated to the contrary.

8.2.1 Special Residential Uses Criteria

8.2.1.1 Cluster Residential Criteria

8.2.1.1 Purpose

It is the express intent of the scale and clustering requirements of this section to create a human scale development appropriate to low-density residential uses. These regulations are intended to prevent structures significantly more massive than those structures usually constructed in single family districts and to allow for the creation of multi-family or other structures exhibiting a human scale and massing appropriate to low-density residential uses.

8.2.1.1.2 Scale Limitations and Clustering Standards

1. The following requirements shall apply, and the appropriateness of individual projects in fulfilling these requirements shall be determined by the City Planner or Designee, Planning Commission, Zoning Board, or City Council, as the case may be according to approval being sought.

The massing of the building(s) and appropriate scale relationships to each other can be accomplished in any of the following ways:

- a. Avoiding unreasonably continuous and unbroken wall planes, or
- b. The introduction of architectural elements or features that create a variety of scale relationships, or
- c. The use of materials that are consistent throughout the project, and that are human in scale, or
- d. The use of architectural elements that create the appearance or feeling of a more residential scale, or
- e. The use of architectural elements or details that are sympathetic to structures on adjoining properties, or
- f. Other design techniques or elements of design that will reasonably comply with the purpose stated in 8.2.1.1.1.
- 2. Absent good reason, which may be evidenced by meritorious compliance with the above criteria, the first tier of structures in a multiple family or mixed use project subject to these Standards shall be clustered in buildings no more than fifty (50) feet in width, measured on the side most parallel to the property line. The first tier of buildings shall be no more than two (2) units deep or sixty (60) feet deep. The structures shall be not less than twenty (20) feet apart measured from face of wall to face of wall.
- 3. The overall density of a clustered residential development shall not exceed the maximum allowable density for the zoning district in which it is located or a minimum land area of three thousand (3,000) square feet per residential unit within a Planned District development.

8.2.1.1.3 Buffering Requirements

- 1. Front building line setbacks shall be a minimum of twenty-five (25) feet from the right-of-way, provided that this requirement shall not apply when located in an B-3 district in which case the yard requirements of the district shall apply.
- 2. Landscaping shall be provided according to the provisions of Article 9.
- 3. Yards, fences, vegetative screening or berms shall be provided to screen adjacent more restrictive residential districts from views of off-Street parking areas, mechanical equipment, storage areas and areas for refuse collection. If

fences are selected for screening, the height shall be restricted to seven (7) feet.

8.2.1.1.4 Regulations to Protect Natural Features of the Environment

- 1. Residential density shall be based on the gross site area and shall be in accordance with the density requirements of the district in which the residential development is located or a minimum of 3,000 square feet per residential unit in Planned Districts.
- 2. Passive uses (i.e. parks, hike and bike trails, etc.), shall be allowed in areas of periodic inundation provided that the requirements of the Drainage Overlay Districts are met and provided that any required state or federal permits and license agreements as may be necessary to place such uses in a dedicated drainage easement are obtained.

8.2.1.1.5 Miscellaneous Regulations

1. Each project shall be permitted a maximum of two identification signs. The sign or signs must be located at the principal entrance or entrances and shall be

allowed in accordance with Article 10 regulations for signs at subdivision entrances.

- 2. No sign shall be self-lit, but a sign may have an indirect lighting source.
- 3. The noise level of mechanical equipment shall conform to the requirements regarding noise and sound as set for in the City's Code of Ordinances.
- 4. All dumpsters and any permanently placed refuse receptacles shall be located a minimum of twenty (20) feet from any adjacent property zoned or used as R-1, R-1X, R-2 or used for a purpose permitted in such a district. The location of and access to dumpsters or any other refuse receptacles shall be reviewed and approved by the Zoning Board.
- 5. The use of highly reflective surfaces, such as reflective glass and reflective metal roofs, whose pitch is more than a run of seven (7) to a rise of twelve (12) shall be prohibited. This requirement shall not apply to solar panels and to copper, galvanized or painted metal roofs.
- 6. No intensive recreational use (e.g. swimming pool, tennis court, ball court, or playground) shall be permitted within fifty (50) feet of an adjoining R-1X, R-1, or R-2 district.

7. No parking or driveways shall be permitted within twenty-five (25) feet of a lot zoned or used as R-1X, R-1 or R-2.

8.2.1.2 Two-Family Residential Criteria

Two-family residential developments shall not be eligible for a subdivision of the two residences into two separate development parcels unless each individual parcel and the building situated thereon meets all the site development regulations including lot dimensions and required yards of the district in which the development is located.

8.2.1.3 Townhouse Residential Criteria

Townhouses, as defined in Article 3, shall be used for purposes of single-family residences only and shall be developed in accordance with the following regulations.

8.2.1.3.1 Townhouse Lot and Area Regulations

- 1. A townhouse development parcel shall consist of land suitable to be subdivided into a minimum of two (2) townhouse sublots capable of meeting the minimum requirements of the district in which the development is located.
- 2. <u>Maximum Density</u> The minimum area of land per unit in a townhouse development shall be three thousand (3,000) square feet provided all other requirements of this Ordinance are met.
- 3. <u>Minimum Square Feet Per Unit</u> The minimum living space within a townhouse unit shall be nine hundred (900) square feet per unit, excluding utility spaces, heating rooms, porches, garages or carports.
- 4. <u>Minimum Sublot Width</u> The minimum width of a townhouse sublot shall be twenty-five (25) feet.
- 5. <u>Private Open Space</u> Each townhouse sublot shall provide at least nine hundred (900) square feet of private outdoor open space and no portion of this minimum open space shall be used for off-Street parking, common recreation areas or accessory buildings.
- 6. <u>Perimeter Yards</u> The minimum perimeter yard requirements of the R-3 Residential district shall apply to all townhouse developments except for the provisions of the required distance between structures listed below.

7. <u>Distance Between Structures</u> - No portion of a townhouse building or accessory structure in or related to one townhouse building shall be closer than twenty-five (25) feet to any portion of another townhouse building or accessory structure related to another townhouse building or to any building outside the townhouse site.

8.2.1.3.2 Townhouse Design Standards

- 1. A townhouse building shall not contain more than six (6) townhouse units built in a row.
- 2. The facade of each unit shall be different from its adjacent facade with not more than two like facades appearing in any one townhouse building.
- 3. No more than three (3) units in a series of six shall have the same front building setback line.

8.2.1.3.3 Townhouse Fire Walls Required

- 1. <u>Fire Walls Between Units</u> Fire walls between each townhouse unit shall be not less than two-hour rating and shall be constructed in conformance to the building codes of the City or the State Fire Marshal for structures built on property lines, whichever is more restrictive.
- 2. <u>Fire Walls Between Groups of Units</u> Each townhouse development shall be divided into groups of not more than four (4) units each with each group separated by a fire wall of not less than four-hour rating, constructed in conformance with the building codes of the city, extending three (3) feet above the roof and eighteen (18) inches beyond the outside face of the building or five (5) feet along the face of the building each side of the fire wall, except that roof and wall penetration by the fire wall may be omitted when units are off-set by not less than five (5) feet in each direction and facing materials are noncombustible.

8.2.1.3.4 Townhouse Parking Regulations

- 1. The number of required parking spaces per unit shall be in accordance with the provisions of Article 9.
- 2. A front yard which contains off-Street parking space shall be a minimum of forty (40) feet in depth from the front property line, shall conform to the

provisions of Article 9 for parking in front yards and not more than one parking space shall be provided in the required front yard setback of a townhouse lot.

3. Access to parking shall be provided in accordance with the requirements of Article 9 and shall be grouped in bays or, if attached to each separate unit, shall be located to the rear of units and in the interior of the development site. All parking facilities shall be adequately and properly drained and constructed to the standards for parking provided in Article 9.

8.2.1.3.5 Townhouse Utility Services

Utility services shall be made available to each townhouse and shall be installed underground. Water, electrical, sewerage, storm drainage and telephone shall be considered utility services. In addition, cable TV and gas may also be provided and shall be required to be installed underground.

8.2.1.3.6 Conversion of Existing Developments

Non-townhouse developments existing at the time of the passage of these regulations shall not qualify for conversion to townhouse conversion and subdivision of townhouse sublots unless constructed to meet the requirements of these regulations as they apply to required fire walls.

8.2.1.4 Congregate Living and Community Residential Criteria

8.2.1.4.1 Requirements for Approval

Congregate housing and community residential uses as defined in Article 6 shall be allowed in the Districts as noted in the Table of Permitted Uses by Zoning District in Article 7.

- 1. <u>Conditions of Approval</u> In reviewing a proposed application for congregate housing or community residential housing, the Zoning Board shall determine from a review of a conceptual site plan and the projects program, the following:
 - a. That the subject property is suitable for the type of development proposed and the proposed development will not be detrimental to the use, peaceful enjoyment, economic value or development of surrounding properties and the general neighborhood; and that

b. If title restrictions exist that they are recorded in the Conveyance Records of the St. Tammany Parish restricting the use of the property to that approved by the Zoning Board in accordance with the provisions of this section and such restrictions shall not be amended except with the approval of the Zoning Board.

8.2.1.4.2 Design Criteria for Congregate Living and Community Residential

1. <u>Area Regulations</u> - Congregate living residential facilities shall be required to be developed in accordance with the following minimum area requirements.

a. Minimum Lot Size - 30,000 sq. ft.

b. Minimum Land Area Per Unit - 1,500 sq. ft.

c. Minimum Unit Area - 325 sq. ft.

d. Minimum Building Common Space- 20%

e. Maximum Impervious Lot Coverage 35%

f. Required Setbacks - Shall be in accordance with the site development standards of the R-3 Residential District.

2. <u>Parking Regulations</u> -

- a. The development must provide the following minimum parking spaces: 1.5 spaces per unit for the development, .75 spaces per unit shall be paved and .75 spaces per unit shall be shown as set aside in a landscaped parking bank..
- b. Any documentation indicating an insufficient number of parking spaces, based on the above formula, shall be justification for the Building Inspector to give notice to the Owner and/or Operator of the facility to appear before the Zoning Board at a hearing to show cause why the parking facilities provided should not be increased by a number to be determined by the Board, if at all.
- c. In the event the Zoning Board requires additional spaces be removed from the previously approved landscaped parking bank and paved, the Owner and/or Operator shall submit documents showing in detail the construction drainage and landscaping of the new spaces for approval.
- d. The Zoning Board will set a reasonable time schedule for compliance.

- 3. <u>Affidavit of Conformity</u> A notarized affidavit of the Owner or Operator of the congregate or community housing facility stating that the facility meets the requirements and restrictions placed on the property by the Zoning Board. A like affidavit shall be submitted annually thereafter to the Mayor or his designee.
- 4. <u>Conversion to Multi-family</u> Any proposed conversion to multi-family use shall be undertaken only in conjunction with the issuance of a zoning permit by the Zoning Board subsequent to a determination by the Zoning Board that the conversion will be

in reasonable conformance with the multi-family requirements of the R-3 residential zoning district.

8.2.1.4.3 Submittal Requirements for Building Permit

Following approval by the Zoning Board of a congregate housing or community residential development as stated in Section 8.2.1.4.1 et seq., the applicant shall submit to the Planning Department all documentation listed in Article 4 necessary to obtain a building permit.

8.2.2 Special Civic Uses Criteria

8.2.2.1 Donation Receptacles

Donation receptacles as defined in Section 3.3.57, including those existing at the time this Ordinance becomes effective, shall be located in accordance with the following requirements.

- 1. The receptacle shall not be placed without a permit from the Building Inspector.
- 2. To apply for a permit, the applicant must supply the following information:
 - a. Articles of Incorporation or other verification that applicant is a non-profit and/or charitable organization.
 - b. A drawing or rendering of proposed receptacle with dimensions.
 - c. A plot plan showing where on a particular piece of property the receptacle is proposed to be located.

- d. Written permission from the property owner allowing receptacles to be located on the property.
- e. What kinds of items are expected to be donated.
- f. Schedule of collections by applicant.
- 3. No receptacle shall be located on City property without the approval of the City Council.
- 4. No receptacle shall be located on any public Right-of-Way.

8.2.2.2 Group Care Facilities

Group care facilities as defined in Article 6 shall be constructed in accordance with the following requirements.

- 1. Site development requirements shall be in accordance with the area regulations of the R-3 Residential District.
- 2. Parking facilities and landscaping shall be provided in accordance with the provisions of Article 9.
- 3. Cooking and other meal preparation activities shall be restricted to common food service areas only with no kitchen facilities provided in individual resident rooms.
- 4. Deliveries of foodstuffs and supplies and pick up of solid waste shall occur only between the hours of 6:00 a.m. and 6:00 p.m. on week-days only.
- 5. No solid waste shall be incinerated on-site and all solid waste containers and access drives shall be located on the site to minimize the impact of such facilities on adjacent sites.

8.2.2.3 Major Impact Utilities

Major impact utilities shall be allowed when approved in conjunction with the Conditional Use Permit application and shall be subject to the Industrial Use Performance Standards of this Article. Existing railroads may continue to be operated and maintained in residence and non-residence districts.

8.2.2.4 Minor Impact Utilities

Public Utilities including electric substations, sewer and water pumping stations, drainage pumping stations, water towers, and buildings and structures of a similar nature may be located in any district when authorized by the Zoning Board and provided that such uses be placed and operated as to minimize any inconvenience to owners or tenants of adjoining premises by reason of the emission of odors, fumes, gases, dust, smoke, noise, vibration, light or glare or other nuisances.

8.2.2.5 Public and Private Primary and Secondary Schools

Public and Private Primary and Secondary Schools shall be subject to the following limitations:

- 1. <u>Access</u> The site shall be located with frontage on an arterial or collector or on a local Street in a location which does not require travel through existing or proposed residential districts and the access Street shall have a paved width of not less than forty (40) feet at all points until it connects with another Street with a paving width of at least forty (40) feet.
- 2. Parking and landscaping requirements shall be in accordance with the requirements of Article 9.
- 3. Each public and private elementary or secondary school shall provide off-Street loading and unloading facilities for buses and students to be located wholly on the
 - premises and to be of adequate size to prevent stacking of buses and cars on Street rights-of-way. A covered transfer station on a circular drive shall be provided.
- 4. Accessory residential units on a school site, in excess of one unit, shall be subject to the residential use classification requirements of the district in which such residential use would be permitted by right.
- 5. <u>Schools in Open Space/Recreation Districts</u> When located in Open Space/Recreation districts in conjunction with a zoning permit, primary and secondary schools shall provide public access to recreational facilities such as basketball or tennis courts, play fields or playgrounds during non-school hours and summer months.

8.2.3 Special Commercial Uses Criteria

8.2.3.1 Adult Uses Criteria

8.2.3.1.1 Findings and Objectives

This section is intended to regulate the noted uses based on the findings of the City Council:

- 1. That the business establishments herein regulated including but not necessarily limited to Adult Arcades, Adult Book Stores, Adult Cabarets, Adult Theaters and Adult Encounter Establishments because of their very nature have serious objectionable operational characteristics and when several of them are concentrated under certain circumstances the result is a deleterious effect on both the area in which they are located and adjacent areas. The special regulations contained herein pertaining to these business establishments are necessary to ensure that these adverse effects will not contribute to the blighting or downgrading of the neighborhood in which these businesses are located or of adjacent neighborhoods.
- 2. That the regulations contained herein neither have the purpose nor effect of imposing any content limitation on those who produce adult books, films or photographs or their ability to make them available to whom they desire and that these regulations neither have the purpose nor effect of restricting in any way the purpose or viewing of these materials by those who desire to view them.

All businesses must comply with Mandeville City Code of Ordinances, Chapter 11, Article II, Section 11-70, entitled "Obscenity".

8.2.3.1.2 Location of Adult Use

No adult use shall be located in any district on any lot within one thousand (1,000) feet of any lot on which there is located another adult use; or within one thousand (1,000) feet of any residential district; or on any lot on which there is located a church, public or private school, public park, playground, residential unit, recreational center, public library, public educational facility, or on any lot dedicated to a church, public or private school, public park, playground, residential unit, recreational unit, recreational center, public library, or educational facility.

For purposes of this section, the following words have the meanings set forth below.

1. <u>Church</u> - Any structure used principally as a place wherein persons regularly assemble for religious worship, including sanctuaries, chapels and cathedrals and on-site buildings adjacent thereto, such as parsonages, friaries, convents, fellowship halls, Sunday Schools and rectories.

2. <u>School</u> - Any structure wherein systematic instruction in useful branches of knowledge is given to children, the majority of whom are between the ages of five years and eighteen years old, by methods common to institutions of learning.

8.2.3.1.3 Display of Material

No display of adult materials shall be visible from any Street right-of-way or adjacent property.

8.2.3.1.4 Adult Uses - Applicability of Other Laws

All other provisions of these Zoning Regulations and all other city ordinances pertaining to any Adult Use remain in full force and effect and the provisions of this Section are in addition to said provisions and ordinances.

8.2.3.2 Automotive and Equipment - Fuel Service Criteria

Fuel dispensing devices and fuel storage areas shall be located a minimum of one hundred (100) feet distance from any adjacent residential district or use.

8.2.3.3 Lodging - Bed and Breakfast Inn Criteria

Bed and breakfast inns, as defined in Article 6, shall be subject to the following general requirements in addition to the parking and landscaping requirements as provided in Article 9 and the district regulations for the district in which the facility is located.

- 1. All of the required approvals shall be obtained prior to establishment of the use including an Administrative or Zoning Permit and Conditional Use approval depending upon the district in which the use is proposed to be located. In addition, an occupational license and a certificate of occupancy for the proposed use shall be obtained from the City. Any additional requirements of the state shall also be required to be satisfied.
- 2. Common bathroom facilities may be provided rather than private baths for each guest room.
- 3. No cooking facilities shall be permitted in the individual guest rooms.
- 4. A common dining area may be provided but cannot be leased for social events except when located in non-residential districts or the B-3 district and only

- where off-Street parking is provided in accordance with the requirements for meeting and reception facilities as set forth in Article 9.
- 5. No exterior signage shall be permitted except in accordance with the requirements set forth in Article 10 which are applicable to the district in which the bed and breakfast facility is located.

8.2.3.4 Lodging - Bed and Breakfast Residence Criteria

Bed and breakfast residences, as defined in Article 6, shall be subject to the following general requirements in addition to the parking and landscaping requirements as provided in Article 9 and the district regulations for the district in which the facility is located:

- 1. All of the required approvals shall be obtained prior to establishment of the use, including an occupational license and certificate of occupancy for the proposed use from the City.
- 2. Common bathroom facilities may be provided rather than private baths for each guest room.
- 3. Residence kitchens shall not be refitted to meet health department requirements for food preparation. Only continental breakfast food service, with foods purchased from a licensed food seller and served "as is" or only warmed at the bed and breakfast facility shall be allowed.
- 4. No cooking facilities are permitted in the individual guest rooms.
- 5. Parking spaces provided for guests, in accordance with the requirements of Article 9, shall be provided in side or rear yards and shall not be located in required front yards.
- 6. No exterior signage shall be permitted except in accordance with the regulations of Article 10 for the district in which the facility is located.

8.2.3.5 Convenience Storage (Mini-Warehouses) Criteria

Convenience storage or mini-warehouse facilities shall be constructed in accordance with the following criteria and in accordance with all other applicable regulations for the district in which the facility is located, including landscaping and signage.

1. <u>Maximum Impervious Surface</u> - Maximum amount of impervious surface shall be no greater than one (1) acre.

- 2. Driveways, Parking and Outdoor Storage of Vehicles
 - a. <u>Driveways</u> Because interior driveways in mini-warehouse facilities serve as loading, parking and circulation spaces the following standards shall apply in the design of interior driveways:
 - i. All one-way driveways shall provide for one ten (10) foot parking lane and one fifteen (15) foot travel lane. Traffic direction and parking shall be designated by signing or painting.
 - ii. All two-way driveways shall provide for one ten (10) foot parking lane and two twelve (12) foot travel lanes.
 - iii. The parking lanes may be eliminated when the driveway does not serve storage cubicles.

b. <u>Parking</u> - Parking spaces shall be provided in accordance with the following standards:

- i. One space for each ten (10) storage cubicles, equally distributed throughout the storage area which may be provided in the interior driveways as provided above.
- ii. Any accessory residence provided for the manager of the facility shall require parking in accordance with Article 9.
- iii. One space for every 25 storage cubicles to be located at the facility office for prospective customers.
- iv. If office space for lease is constructed in accordance with city building codes as a feature of the facility, parking shall be provided in the area of the leased office space in accordance with the requirements of Article 9.

c. Outdoor Storage

- i. No outdoor storage of goods shall be allowed except in M-2 districts and provided such areas are screened from view of all adjacent roadways or sites.
- ii. Outdoor storage of vehicles, recreation vehicles, trailers, campers and boats shall be in areas specifically provided for such storage and shall not be located on required parking spaces for the facility including those provided on interior driveways.
- iii. Outdoor storage areas for vehicles as described above shall be separated from storage buildings by at least twenty-five (25) feet, shall be located in the rear of the buildings and shall be screened from view of adjacent property or roadways with an

opaque fence and, when adjacent to residential districts, with a vegetative buffer as provided in the regulations of Article 9.

3. <u>Lighting</u> - Lighting shall be in accordance with the standards for parking lot lighting as provided in Article 9.

4. Accessory Uses

- a. <u>Accessory Residential</u> One accessory residence may be provided on the site for a resident manager.
- b. <u>Accessory Office</u> Leasable office space constructed in accordance with city building codes may be permitted in conjunction with a mini-warehouse facility.
- c. Other Uses Except for the uses allowed in accordance with the above provisions, mini-warehouse developments shall be limited to dead storage use only and no business activities other than the rental of storage units shall be conducted on the premises. No activities such as miscellaneous or garage sales shall be conducted on the premises. The servicing or repair of motor vehicles, boats, trailers, lawn mowers, and other similar equipment shall not be conducted on the premises. The operation of such a facility shall in no way be deemed to include a transfer and storage business where the use of vehicles for purposes of deliveries is part of such business.
- 5. <u>Storage</u> No hazardous materials or substances and no explosive or radioactive materials shall be stored on the premises.

8.2.3.6 Day Care Centers Criteria

Day Care centers as defined in Article 6 shall be required to be established in accordance with the following criteria and in accordance with all other applicable regulations as provided in these Land Use Regulations and all other codes of the City or state. For purposes of these provisions activity area shall be defined as indoor space utilized for the activities of the participants in the day care program and shall not include any area utilized for administrative offices, kitchens, hallways, bathrooms or storage areas.

- 1. Applicants for proposed Day care centers shall be required to submit a floor plan and site plan which conform to the following standards:
 - a. A minimum of 35 square feet of indoor activity area for each participant enrolled in the day care facility during any one time period or the minimum required for Class A state licensing, whichever is greater.

- b. In the case of day care centers for children, a minimum of seventy-five (75) square feet of outdoor play area for each participant enrolled in the day care facility during any one time period.
- c. In the case of adult day care centers, an outdoor area for strolling and sitting of a minimum of five hundred (500) square feet for each ten (10) participants enrolled in the facility during any one time period.
- 2. An area for the transfer of participants from vehicles to the facility shall be provided with a queue area for at least four vehicles in an area on-site and out of the right-of-way of any Street.
- 3. In accordance with the buffer requirements of Article 9, day care centers in or adjacent to B-2, B-4, M-1 or M-2 districts shall provide a buffer of ten (10) feet adjacent to the property line of such district, or on the side and rear property lines if in the listed district.
- 4. Parking shall be in accordance with the requirements of Article 9. 8.2.3.7 Special Marina Use Criteria
 - 1. Definitions Applicable to the Marina Development
 - a. <u>Marina Site</u> A marina site shall consist of a development site which includes a boat basin or harbor for private pleasure craft where boat moorings are available for rental to the general public. Official craft, i.e. USCG, harbor master, etc. may be permitted to moor in a marina site.
 - b. <u>Boat Slip</u> That area of wet storage bounded by four (4) pilings, including one-half of the adjacent catwalk/pier.
 - c. <u>Dock</u> A mooring which accommodates a craft lying along side a wharf, pier or bulkhead having the benefit of only single side ties to piling or dock cleats.
 - d. <u>Suitable Size Boat Slip</u> A slip, with reference to a specific type or design of boat, which can physically and safely accommodate such a boat as designated by the marina owner and/or supervisor, considering the safety of the vessel and of the property owned by the marina owner.
 - e. <u>Catwalk</u> The walkway constructed on pilings to provide pedestrian access to boat docks.
 - 2. Standards for Marina Design

- a. Access to Boat Slips Boat slips shall be accessible by at least one (1) catwalk.
- b. <u>Repair of Non-Conforming Docking Structures</u> Provisions of Section 4.2 Non-Conforming Provisions do not apply.
- c. <u>Off-Street Parking</u> Each marina development site shall be required to provide a minimum of one off-Street vehicular parking space for each four (4) boat slips contained within the marina facility as shown on the development plan for the marina.

d. Sanitation Requirements

- i. Marina development sites shall provide one on-site comfort station, which shall include minimally a lavatory and toilet facility connected to city utilities, for each fifty (50) boat slips or docks.
- ii. All boats associated with marina development sites shall comply with all USCG and EPA regulations regarding marine sanitation.
- iii. Marina owners shall be required to provide a marine sanitation device designed and tied into the City of Mandeville sewerage system which complies with all sanitation and health requirements of state and federal law for the use of boat tenants.
- iv. Regardless of the number of boat slips, no raw sewage, garbage, junk, waste, oil or any other substance whatsoever which might pollute or damage the aesthetics or the water quality shall be deposited or permitted to be deposited in the waters of Bayou Castain and/or Lake Pontchartrain or any harbor, canal or canals leading into same.
- e. <u>Refuse Collection</u> Suitable trash containers shall be located on all marine development sites and provisions for the collection and removal of garbage and solid waste shall be provided by the marina owner.
- f. Residential Occupancy of Boats Residential occupancy of boats within marina development sites shall be permitted provided not more than five (5) percent of the total number of boat slips or a total of five watercraft, whichever is greater, are utilized for residential occupancy and provided residential occupants shall use the marine sanitation device monthly or more frequently.

g. <u>Non-residential Commercial Occupancy of Boats</u> - The use of boats or watercraft for non-residential occupancy in any marina shall be prohibited.

3. Permits and Approvals for Marina Developments

Marina development sites proposed to be developed in areas subject to the jurisdiction of the U.S. Army Corps of Engineers and/or the Louisiana Coastal Zone Management authority shall be required to obtain all the necessary permits or approvals and to comply with all city, parish, state and federal regulatory requirements prior to the issuance of a permit from the City for construction of the marina.

4. Standards for Marina Associated Uses

a. <u>Dwellings</u> - Dwellings constructed in conjunction with a marina site shall be constructed in accordance with the Site Development Regulations of the zoning

district in which the marina site is located or, in the case of Planned Districts, in accordance with the density requirements for the type of residential usage proposed, if the use were located in the district where the proposed use would be permitted by right. Additionally, dwelling uses shall comply with any special use criteria established in Article 8 for the type of dwelling use proposed.

- b. <u>Accessory Structures</u> Accessory structures shall be constructed in accordance with the requirements of Article 8, shall not be constructed in any required landscaped areas and, when constructed on catwalks shall be located a minimum distance of 30' between structures. When located on catwalks at the edge of navigable waterways no greater than 50% of the water frontage of the development lot shall be obstructed from view by accessory structures.
- c. <u>Minor Boat Repairs</u> Repair, servicing, and routine maintenance of boats, including cleaning, in conjunction with a marina site or on a separate site, shall be in an area screened from view of any adjacent roadways or waterways. However, any sand blasting or spray painting shall be conducted in compliance with OSHA and EPA Regulations.
- d. <u>Major Boat Repairs & Construction</u> Repairs and construction of boats for recreational and personal use, especially in conjunction with a boat lift shall be in an area surrounded by a twenty (20) foot buffer area with an opaque fence. In addition, any sand blasting or spray painting shall be conducted in compliance with OSHA and EPA Regulations.

- e. <u>Dry Boat Storage Areas</u> Dry boat storage areas shall not exceed 20% of the land of a marina site, exclusive of front, side and rear yard setback.
- f. <u>Fuel Dispensing Areas</u> Fuel dispensing areas for the storage of oil and fuel for the servicing of all craft may be located on marina development sites provided access to the site by fuel supply trucks shall be required to be by way of non-residential streets only. There shall be no permanent docking within thirty (30) feet of fuel pumps or other fuel equipment. Fuel storage tanks shall not exceed 10,000 gallons in capacity, shall meet all state and federal safety guidelines and shall not be located within 100' of any residential use or district.
- g. <u>Non-Residential Uses</u> Non-residential uses permitted in conjunction with a marina development site in the PM-1 District in accordance with the Table of Permitted Uses by Zoning District shall be constructed in accordance with the PM-1 site development regulations for non-residential uses and any other provisions applicable to the specific use.
- h. In addition to the required off-Street parking spaces for the marina use based on the number of boat slips or docks, off-Street parking for any residential or non-residential use located on a marina development site shall be required to be provided in accordance with the provisions of Article 9 for the specific use.
- I. <u>Temporary Boat Docking</u> Temporary boat docking facilities shall be required to be provided in conjunction with all commercial or transient lodging facilities which are accessible to and provide services to customers arriving by boat at a ratio of one boat slip for each three hundred (300) square feet of retail or restaurant patron area or one boat slip for each three (3) transient lodging units, up to a maximum of five (5).
- j. <u>Laundry Service</u> Establishments owned and operated by the marina primarily engaged as an accessory use in the provision of coin-operated laundry machines with or without the incidental provision of laundry processing services for individuals by an attendant in the facility are permitted.

5. Special Criteria for Boathouse Development

a. Applicants for Boathouse Developments shall be required to apply for Major Subdivision/Resubdivision and comply with all application and procedure requirements as outlined in Article 12 of this Ordinance.

- b. The height of a Boathouse shall not exceed 35' as required in this Ordinance.
- c. No Boathouse shall be built beyond the established bulkhead line or waterway line or former property line before excavation for boat slips, whichever is greater. This line shall establish a lot depth line for the purposes of this Code.
- d. Detached boathouse developments shall comply with restrictions of PM-1 zoning district.
- e. Attached boathouse developments shall comply with restrictions of PM-1 zoning district and in addition to the following restrictions:
- 18. Each lot shall provide a minimum lot width of 25', a minimum lot depth of 75' and a minimum lot area of 1,875 square feet.
 - ii Each Boathouse shall not have less than 600 square feet of living area and shall not exceed 2,000 square feet under beam.
 - iii. Front yard setback 25'.
 - iv. Side yard setback 20'.
 - v. Minimum of 25' between structures.
 - vi. There shall not be more than six (6) attached units in any one building.
 - f. If deemed necessary by the Planning Commission, a servitude of sufficient width providing the common use of the adjacent waterway area(s) shall be provided to accommodate navigation and shall be indicated on the plan for subdivision.
 - g. Individual utility systems shall be provided for each boathouse.
 - h. Municipal sewerage services must be provided for each boathouse. (No septic tanks or other waste disposal systems may be installed.)
 - i. Developer must comply with all applicable regulations of the U.S. Army Corps of Engineers, the Louisiana Coastal Zone Management and other governmental agencies with jurisdiction.

8.2.3.8 Tavern and Night Club Criteria

In addition to all of the applicable regulations of the district in which the establishment is located, the requirements of Article 9 regarding landscaping and parking, all licensing requirements of the city and state, and any other applicable regulations of the city, taverns and night club uses shall be required to conform to the following standards:

- 1. Taverns and night clubs shall not be located on lots which front on the principal access Street into a single-family residential district or within three hundred (300) feet of an R-1 or R-1X residential district.
- 2. Taverns or nightclubs shall be located on lots with Street frontage 1) on arterial streets, 2) on collector streets or 3) on local streets which do not require exclusive travel through existing or proposed residential districts to access the site.
- 3. Sound levels will conform to the regulations of the City of Mandeville's Sound Ordinance.

8.2.4 Special Industrial Uses Criteria

8.2.4.1 Industrial and Commercial Uses Performance Standards

The following performance standards shall apply to all land uses in the City of Mandeville and particularly to those industrial uses allowed in the M-1 and M-2 districts, the Planned Industrial Districts and any other district where industrial uses may be allowed in accordance with the Table of Permitted Uses By Zoning District in Article 7 and are provided to insure protection of the environment and adjacent property by regulating direct impacts to air and water resources and by regulating indirect impacts from ambient pollution, radiation hazards, noise, fire and explosive hazards.

- 1. <u>Exhaust Emission</u> No use shall emit from any exhaust pipe or fire chimney any emission that shall be deemed harmful by the State Department of Environmental Quality. Every use shall be operated to prevent the emission of dust or other solid matter into the air that may cause damage to property or endanger the health of residents of the City.
- 2. Odor The emission of obnoxious odors of any kind beyond the property boundaries shall not be permitted, and particular industries may be required to present comprehensive statements of measures to be taken for elimination of obnoxious odors to the planning commission before the required building permits are granted. Odorous matter released from any operation or activity in an industrial district shall

not exceed the odor threshold concentration established by applicable state agencies beyond lot lines, measured at ground level or habitable level.

- 3. <u>Water Quality</u> In compliance with the Federal Water Pollution Control Act, Louisiana Water Control Law, and the City of Mandeville sewer use ordinance no industry shall discharge harmful substances into a waterway or waste water collection or disposal system.
- 4. Noise Industrial uses in the M-1 or Planned Industrial Districts shall not emit noise at levels greater than the requirements regarding noise and sound as set forth in the City's Code of Ordinances and industrial uses in M-2 Districts shall not emit noise levels greater than the requirements regarding noise and sound as set forth in the City's Code of Ordinances at the lot boundary line measured at ground level or habitable elevation between the hours of seven in the morning and seven in the evening. During evening hours, the noise level for both districts must conform to the requirements regarding noise and sound as set forth in the City's Code of Ordinances. Applicable measurement standards shall be taken by an independent lab institute at the expense of the applicant or legal property owner. The council shall be the discretionary governing body to determine the frequency of decibel measurements taken annually.
- 5. <u>Radiation</u> No operation involving radiation hazards which violate the rules or regulations of Nuclear Energy and Radiation Control Law regulated by the Department of Environmental Quality shall be conducted in the City of Mandeville.
- 6. <u>Fire & Explosive Hazards</u> All industrial uses shall comply with applicable standards set forth in the rules and regulations of the State Fire Marshall.
- 7. <u>Electrical Disturbances</u> No use, activity or process shall be conducted which produces electromagnetic interference with normal radio or television reception in the city.
- 8. <u>Disposal of Liquid Wastes</u> No use may discharge any waste into the municipal sewerage system except in accordance with all city codes.
- 9. Administration & Enforcement
 - a. <u>State Jurisdiction</u> To the extent required by state Law, the Department of Environmental Quality will administer, monitor and enforce the requirements

of this Section which fall within its jurisdiction. Under the provisions of Article 7, special

zoning district criteria may require mitigation measures such as larger buffer areas to protect adjacent property owners from adverse impacts.

b. Requirement for Conditional Use Procedures - Any use not in full compliance with these standards shall automatically be classified as a Conditional Use and shall be subject to City Council approval prior to permitting. The council may require of the applicant any additional information as is reasonably necessary to protect the health, safety or welfare of the public and the preservation of the environmental quality and adjacent property values in conjunction with a conditional use permit application for the proposed use of the property. Studies by independent experts may be required by the City Council to be conducted at the expense of the applicant or legal property owner requesting the use.

8.2.5 Special Combined Uses Criteria

In considering an application for a combined use development site, all applicable provisions of this Ordinance and the district in which the site is located shall apply except as follows:

- 1. When a residential and non-residential use are combined on one site the density requirements for the residential use for the district in which the site is located shall prevail, however, the yard setbacks applicable to the non-residential use may be applied so long as a minimum of four hundred (400) square feet of landscaped area per residential unit, in addition to the otherwise required landscape areas, shall be provided for the use of the residential occupants of the site.
- 2. The parking required for each of the proposed uses on the development site may be adjusted by the approving authority in accordance with the Parking for Mixed Use Developments provisions of Article 9.

8.2.6 Special Accessory Uses Criteria

8.2.6.1 Home Occupations Criteria

Home occupations as defined in Article 6 shall be required to conform to the following standards:

- 1. No person shall be employed on the premises who is not a bona fide resident of the dwelling and the individual primarily responsible for the home occupation shall live in the dwelling.
- 2. The use of the dwelling unit for home occupation shall be clearly incidental and secondary to its use for residential purposes and not more than twenty (20) percent of the living area of the dwelling unit or four hundred (400) square feet, whichever is the lesser, shall be used in the conduct of home occupations; and no outdoor display or storage of equipment or supplies associated with the home occupation is permitted.
- 3. There shall be no change in the exterior appearance of the building or premises as a result of such occupation or occupations, with the exception of a sign as provided in the sign code.
- 4. No home occupation shall be conducted in any accessory building or attached garage exceeding four hundred (400) square feet.
- 5. No mechanical equipment shall be used or stored on the premises except such that is normally used for purely domestic or household purposes, nor shall the home occupation create noise, vibration, glare, fumes, odors, dust, smoke, or heat detectable to the normal senses outside the dwelling unit. No equipment or process shall be used which creates visual or audible interference in any radio or television sets off the premises, or causes fluctuations in line voltage. There shall be no illegal discharge of any materials, fluids or gases into the sewer or drainage system or any other manner of discharging such items in violation of any applicable government code.
- 6. No stock-in-trade shall be sold on the premises nor displayed or warehoused on the premises for sale or use elsewhere, provided that orders previously made by telephone, mail or at a sales party conducted off-premises may be filled on the premise and delivered.
- 7. No traffic shall be generated by such home occupation in greater volume than three vehicles per 24 hour day in the residential neighborhood and any need for parking generated by the conduct of such home occupation shall be met off the Street and other than in the required front yard. Deliveries from commercial suppliers shall be made during daylight hours and shall not restrict circulation in the neighborhood.
- 8. Personal services such as cosmetology, barber shops, beauty parlors, kennels, dog grooming; real estate and insurance offices; radio, television and

appliance repair, cabinet making, boat building for others, auto servicing or rebuilding and repair for others; metal fabrications or cutting, employing welding or cutting torches, ambulance service, helium balloons, house painters and other uses similar in nature or in effect on the surrounding neighborhood shall not be allowed to be conducted as a home occupation.

- 9. No more than one home occupation related vehicle, regardless of the number of home occupations, is permitted at any one premise, any such vehicle must be twenty (20) feet or less in overall length and not more than seven (7) feet in overall height and must be parked off any public right-of-way. All exterior storage of cargo, equipment or other material on the vehicle shall be shielded from view at all times when such vehicle is located on a residential lot.
- 10. A home occupation that consists solely of the receiving of phone calls, mail, and keeping business records in connection with any profession or occupation, shall be known as an "address of convenience" and shall not require a home occupation permit. Any home occupation which receives clients or customers shall not be classified as an "address of convenience."
- 11. When in compliance with the above requirements, a home occupation includes but is not limited to the following: (a) art studio; (b) child care for not more than six (6) children, including any children of the adult provider; (c) dressmaking and tailoring; (d) professional office of a lawyer, engineer, architect, accountant, salesman, or other similar occupation; (e) teaching or tutoring, including musical instruction and dance instruction, limited to not more than two (2) pupils at a time; (f) typing/word-processing service; (g) small scale seafood harvesting with no more than one recreational type boat stored on the premises and without outside storage of equipment unless screened from view of the Street and adjacent property.
- 12. Home occupations which are existing as legal uses shall not be allowed to continue once the occupants who have established the legal use status no longer occupy the premises.
- 13. The City Planner, or his designee, shall determine whether the home occupation meets the established criteria and shall issue a home occupation permit when such application is in compliance with the established criteria. Any person aggrieved by a decision of the City Planner, or his designee, may appeal that decision to the Zoning Board in accordance with the procedure for filing appeals to the Zoning Board as defined in Article 4.

8.3 FLOOD DAMAGE PREVENTION REGULATIONS

8.3.1 Authorization, Findings of Fact, Purpose and Methods

8.3.1.1 Statutory Authorization

The Legislature of the state of Louisiana has in R.S. 38:84 delegated the responsibility to local governmental units to adopt regulations designed to minimize flood losses. Therefore, the Council of the City of Mandeville, has enacted the flood control ordinance set forth below.

8.3.1.2 Findings of Fact

- 1. The designated flood hazard areas of the City of Mandeville are subject to periodic inundation which can result in loss of life and property, health and safety hazards, disruption of commerce and governmental services, and extraordinary public expenditures for flood protection and relief, all of which adversely affect the public health, safety and general welfare.
- 2. These flood losses are created by the cumulative effect of obstructions in floodplains which cause an increase in flood heights and velocities, and by the occupancy of flood hazard areas by uses vulnerable to floods and hazardous to other lands because they are inadequately elevated, floodproofed or otherwise protected from flood damage.

8.3.1.3 Statement of Purpose

It is the purpose of this Section 8.3, et seq. to promote the health, safety and general welfare of the citizens of Mandeville and to minimize public and private losses due to flood conditions in specific areas by provisions designed to:

- 1. Protect human life and health;
- 2. Minimize expenditure of public money for costly flood control projects;
- 3. Minimize the need for rescue and relief efforts associated with flooding which are and generally undertaken at the expense of the general public;
- 4. Minimize prolonged business interruptions;
- 5. Minimize damage to public facilities and utilities such as water and gas mains, electric, telephone and sewerlines, streets and bridges located in floodplains;

- 6. Help maintain a stable tax base by providing for the sound use and development of flood-prone areas which would then minimize future flood blight areas; and,
- 7. Provide to potential buyers upon request information regarding whether the property is located in a flood area.

8.3.1.4 Methods of Reducing Flood Losses

In order to accomplish its purposes, this section uses the following methods:

- 1. Restrict or prohibit uses that are dangerous to health, safety or property in times of flood, or cause excessive increases in flood heights or velocities;
- 2. Require that uses vulnerable to floods, including facilities which serve such uses, be protected against flood damage at the time of initial construction;
- 3. Control the alteration of natural floodplains, stream channels, and natural protective barriers, which are involved in the accommodation of flood waters;
- 4. Control filling, grading, dredging, and other development which may increase flood damage;
- 5. Prevent or regulate the construction of flood barriers which will unnaturally divert flood waters or which may increase flood hazards to other lands.

8.3.2 Definitions of Flood Damage Prevention Regulations

Unless specifically defined below, words or phrases used in this Section 8.3 et seq. shall be interpreted to give them the meaning they have in common usage and to give this section its most reasonable application.

8.3.2.1 Appeal

A request for a review of the floodplain administrator's interpretation of any provision of this Section 8.3, et seq. or a request for a variance.

8.3.2.2 Area of Shallow Flooding

A designated AO, AH, or VO zone on a community's Federal Flood Insurance Rate Map (FIRM) with a one percent or greater annual chance of flooding to an average depth of one to three feet where a clearly defined channel does not exist, where the path of flooding is

unpredictable and where velocity flow may be evident. Such flooding is characterized by ponding or sheet flow.

8.3.2.3 Area of Special Flood Hazard

The land in the floodplain within a community subject to a one percent or greater chance of flooding in any given year. The area may be designated as Zone A on the Flood Hazard Boundary Map (FHBM). After detailed rate making has been completed in preparation for publication of the FIRM, Zone A may be refined into Zones A, AE, AH, AO, A1-99, VO, V1-30, VE or V.

8.3.2.4 Base Flood

The flood having a one percent chance of being equaled or exceeded in any given year.

8.3.2.5 Breakaway Walls

A wall that is not part of the structural support of the building and is intended through its design and construction to collapse under specific lateral loading forces, without causing damage to the elevated portion of the building or supporting foundation system.

8.3.2.6 Coastal High Hazard Area

The area subject to high velocity waters, including but not limited to hurricane wave wash or tsunamis (tidal waves caused by seismic or volcanic activity). The area is designated on a FIRM as Zone V1-30, VE or V.

8.3.2.7 Critical Feature

An integral and readily identifiable part of a flood protection system, without which the flood protection provided by the entire system would be compromised.

8.3.2.8 Elevated Building

A non-basement building (I) built, in the case of a building in Zones A1-30, AE, A, A99, AO, AH, B, C, X, and D, to have the top of the elevated floor, or in the case of a building in Zones V1-30, VE, or V, to have the bottom of the lowest horizontal structure member of the elevated floor elevated above the ground level by means of pilings, columns (posts and piers), or shear walls parallel to the flow of the water and (ii) adequately anchored so as not to impair the structural integrity of the building during a flood of up to the magnitude of the base flood. In the case of Zones A1-30, AE, A, A99, AO, AH, B, C, S, D, "elevated building" also includes a building elevated by means of fill or solid foundation perimeter walls with openings sufficient to facilitate the unimpeded movement of flood waters. In the case of Zones V1-30,

VE, or V, "elevated building" also includes a building otherwise meeting the definition of "elevated building," even though the lower area is enclosed by means of breakaway walls if the breakaway walls meet the standards of the National Flood Insurance Program regulations.

8.3.2.9 Existing Construction

For purposes of determining rates, structures for which the "start of construction" commenced before the effective date of the FIRM or before January 1, 1975, for FIRMs effective before that date. "Existing construction" may also be referred to as "existing structures".

8.3.2.10 Flood or Flooding

A general and temporary condition of partial or complete inundation of normally dry land areas from:

- 1. The overflow of inland or tidal waters;
- 2. The unusual and rapid accumulation or runoff of surface waters from any source.

8.3.2.11 Flood Hazard Boundary Map (FHBM)

Flood Hazard Boundary Map (FHBM) means an official map of a community, issued by the Administrator, where the boundaries of the flood, mudslide (i.e., mudflor) related erosion areas having special hazards have been designated as Zone A, M, and/or E.

8.3.2.12 Flood Insurance Rate Map (FIRM)

An official map of a community, on which the Federal Emergency Management Agency (FEMA) has delineated both the areas of special flood hazards and the risk premium zones applicable to the community.

8.3.2.13 Flood Insurance Study

The official report provided by the Federal Emergency Management Agency. The report contains flood profiles, water surface elevation of the base flood, as well as the flood boundary-floodway map.

8.3.2.14 Floodplain or Flood-Prone Area

Any land area susceptible to being inundated by water from any source (see definition of flood or flooding).

8.3.2.15 Floodplain Management

Floodplain management means the operation of an overall program of corrective and preventive measures for reducing flood damage, including but not limited to emergency preparedness plans, flood control works and flood plain management regulations.

8.3.2.16 Floodplain Management Regulations

Floodplain management regulations means zoning ordinances, subdivision regulations, building codes, health regulations, special purpose ordinances (such as a floodplain ordinance, grading ordinance and erosion control ordinance) and other applications of police power. The term describes such state or local regulations, in any combination thereof, which provide standards for the purpose of flood damage prevention and reduction.

8.3.2.17 Floodplain Protection System

Floodplain protection system means those physical structural works for which funds have been authorized, appropriated, and expended and which have been constructed specifically to modify flooding in order to reduce the extent of the area within a community subject to a "special flood hazard" and the extent of the depths of associated flooding. Such a system typically includes hurricane tidal barriers, dams, reservoirs, levees or dikes. These specialized flood modifying works are those constructed in conformance with sound engineering standards.

8.3.2.18 Flood-Proofing

Any combination of structural and non-structural additions, changes or adjustments to structures which reduce or eliminate flood damage to properties, water and sanitary facilities and other utilities, structures and their contents.

8.3.2.19 Flood Protection System

Those physical structural works for which funds have been authorized, appropriated, and expended and which have been constructed specifically to modify flooding in order to reduce the extent of the areas within a community subject to a special flood hazard" and the extent of the depths of associated flooding. Such a system typically includes hurricane tidal

barriers, dams, reservoirs, levees or dikes. These specialized flood modifying works are those constructed in conformance with sound engineering standards.

8.3.2.20 Functionally Dependent Use

A use which cannot perform its intended purpose unless it is located or carried out in close proximity to water. The term includes only docking facilities, port facilities that are necessary for the loading and unloading of cargo or passengers, and ship building and ship repair facilities, but does not include long-term storage or related manufacturing facilities.

8.3.2.21 Habitable Floor

Any floor usable for the following purposes; working, sleeping, eating, cooking or recreation, or a combination thereof. A floor used for storage purposes only is not a "habitable floor".

8.3.2.22 Highest Adjacent Grade

The highest natural elevation of the ground surface prior to construction next to the proposed walls of a structure.

8.3.2.23 Levee

A manmade structure, usually an earthen embankment, designed and constructed in accordance with sound engineering practices to contain, control, or divert the flow of water so as to provide protection from temporary flooding.

8.3.2.24 Levee System

A flood protection system which consists of a levee, or levees, and associated structures, such as closure and drainage devices, which are constructed and operated in accordance with sound engineering practices.

8.3.2.25 Lowest Floor

The lowest floor of the lowest enclosed area (including basement). An unfinished or flood resistant enclosure, usable solely for parking of vehicles, building access or storage in an area other than a basement area is not considered a building's lowest floor, provided that such enclosure is not built so as to render the structure in violation of the applicable non-elevation design requirement of federal flood insurance regulations.

8.3.2.26 Manufactured Home

Manufactured home, for the purpose of this Section 8.3 only, means a structure transportable in one or more sections, which is built on a permanent chassis and is designed for use with or without a permanent foundation when connected to the required utilities. For floodplain management purposes the term "manufactured home" also includes park trailers, travel trailers, and other similar vehicles placed on a site for greater than one hundred eighty (180) consecutive days. For insurance purposes the term "manufactured home" does not include park trailers, travel trailers, and other similar vehicles.

8.3.2.27 Mean Sea Level

For purposes of the National Flood Insurance program, the National Geodetic Vertical Datum (NGVD) of 1929 or other datum to which base flood elevations shown on a community's federal flood insurance rate map are referenced.

8.3.2.28 New Construction

For floodplain management purposes, structures for which the "start of construction" commenced on or after the effective date of a floodplain management regulation adopted by a community.

8.3.2.29 Start of Construction

For other than new construction or substantial improvements under the Coastal Barrier Resources Act (Pub. L. 97-348), includes substantial improvement and means the date the building permit was issued, provided the actual start of construction, repair, reconstruction, placement, or other improvement was within one hundred eighty (180) days of the permit date. The actual start means either the first placement of permanent construction of a structure on a site, such as the pouring of slab or footings, the installation of piles, the construction of columns, or any work beyond the stage of excavation; or the placement of a manufactured home on a foundation. Permanent construction does not include land preparation, such as clearing, grading and filling; nor does it include the installation of streets and/or walkways; nor does it include excavation for basement, footings, piers or foundations or the erection of temporary forms; nor does it include the installation on the property of accessory buildings, such as garages or sheds not occupied as dwelling units or not part of the main structure.

8.3.2.30 Structure

For purposes of floodplain management, a walled and roofed building, including a gas or liquid storage tank, that is principally above ground, as well as a manufactured home.

8.3.2.31 Substantial Improvement

For the purposes of administration of federal flood management regulations, substantial improvement shall mean any repair, reconstruction, or improvement of a structure, the cost of which equals or exceeds fifty (50) percent of the market value of the structure either, (1) before the improvement or repair is started, or (2) if the structure has been damaged and is being restored, before the damage occurred. The total cost of any and all repairs, reconstructions, or improvements shall be cumulative, beginning from the date that the first alteration commenced. For the purpose of this definition "start of construction" is considered to occur when the first alteration of any wall, ceiling, floor, or other structural part of the building commences, whether or not that alteration affects the external dimensions of the structure. The term does not, however, include either (1) any project for improvement of a structure to comply with existing state or local health, sanitary, or safety code specifications which are solely necessary to assure safe living conditions, or (2) any alteration of a structure listed on the National Register of Historic Places or a State Inventory of Historic Places.

8.3.2.32 Variance

A grant of relief to a person from the requirements of this Ordinance when specific enforcement would result in unnecessary hardship. A variance, therefore, permits construction or development

in a manner otherwise prohibited by this Section 8.3, et seq. (For full requirements, see Section 60.6 of the National Flood Insurance Program regulations.)

8.3.2.33 Violation

The failure of a structure or other development to be fully compliant with the community's floodplain management regulations. A structure or other development without the elevation certificate, other certifications, or other evidence of compliance required in Section 60.3(b)(5), (c)(4), (C)(10), (d)(3), (e)(2), (e)(4) or (e)(5) or other datums where specified of the National Flood Insurance Program regulations is presumed to be in violation until such time as that documentation is provided.

9.3.2.34 Water Surface Elevation

The height, in relation to the National Geodetic Vertical Datum (NGVD) of 1929 of floods of various magnitudes and frequencies in the floodplains of coastal or riverine areas.

8.3.3 General Provisions

8.3.3.1 Lands to Which This Section Applies

This section shall apply to all areas of special flood hazard within the jurisdiction of the city.

8.3.3.2 Basis for Establishing the Areas of Special Flood Hazard

The areas of special flood hazard (100 year flood zones A and V classification) identified by the Federal Emergency Management Agency in a scientific and engineering report entitled "The Flood Insurance Study for the City of Mandeville", dated April 4, 1983, with accompanying flood insurance rate maps and flood boundary floodway maps (FIRM and FBFM) and any revisions thereto are hereby adopted by reference and declared to be a part of this section.

8.3.3.3 Compliance

No structure or land shall hereafter be located, altered, or have its use changed without full compliance with the terms of this section 8.3, et seq. and other applicable regulations.

8.3.3.4 Abrogation and Greater Restrictions

This article is not intended to repeal, abrogate, or impair any existing easements, covenants, or deed restrictions. However, where this section 8.3, et seq. and another ordinance conflict or overlap, whichever imposes the more stringent restrictions shall prevail.

8.3.3.5 Interpretation

In the interpretation and application of this section, all provisions shall be:

- 1. Considered as minimum requirements,
- 2. Liberally construed in favor of the governing body; and
- 3. Deemed neither to limit nor repeal any other powers granted under state statutes.

8.3.3.6 Warning and Disclaimer of Liability

The degree of flood protection required by this section 8.3, et seq. is considered reasonable for regulatory purposes and is based on scientific and engineering considerations. On rare occasions greater floods can and will occur and flood heights may be increased by manmade or natural causes. This Ordinance does not imply that land outside the areas of special flood hazards or uses permitted within such areas will be free from flooding or flood damages. This section shall not create liability on the part of the community or any official or employee

thereof for any flood damages that result from reliance on this section or any administrative decision lawfully made thereunder.

8.3.4 Administration

8.3.4.1 Designation of Floodplain Administrator

The building inspector is hereby appointed the floodplain administrator to administer and implement the provisions of this section 8.3, et seq. and other appropriate sections of 44 CFR (National Flood Insurance Program Regulations) pertaining to floodplain management.

8.3.4.2 Duties and Responsibilities of the Floodplain Administrator

Duties and responsibilities of the floodplain administrator shall include, but not be limited to, the following:

- 1. Maintain and hold open for public inspection all records pertaining to the provisions of this article.
- 2. Review permit applications to determine whether a proposed building site will be reasonably safe from flooding.
- 3. Review, approve or deny all applications for development permits required by adoption of this article.
- 4. Review permits for proposed development to assure that all necessary permits have been obtained from those federal, state or local governmental agencies (including Section 404 of the Federal Water Pollution Control Act Amendments of 1972, 33 U.S.C. 1334) from which prior approval is required.
- 5. Where interpretation is needed as to the exact location of the boundaries of the areas of special flood hazards (for example, where there appears to be a conflict between a mapped boundary and actual field conditions), the floodplain administrator shall make the necessary interpretation.
- 6. Notify, in riverine situations, adjacent communities and the state coordinating agency, which is the Louisiana Department of Urban and Community Affairs, as well as the U. S. Army Corps of Engineers and the State of Louisiana Coastal Area Management Agency, prior to any alteration or relocation of a watercourse, and submit evidence of such notification to the Federal Emergency Management Agency.

- 7. Assure that the flood carrying capacity within the altered or relocated portion of any watercourse is maintained.
- 8. When base flood elevation data has not been provided in accordance with subsection 8.3.3.2, the floodplain administrator shall obtain, review and reasonably utilize any base flood elevation data and floodway data available from a federal, state or other source in order to administer the provisions for flood hazard reduction.
- 9. When a regulatory floodway has not been designated, the floodplain administrator must require that no new construction, substantial improvements, or other development (including fill) shall be permitted within Zones A1-30 and AE designated on the community's flood insurance rate map (FIRM), unless it is demonstrated that the cumulative effect of the proposed development, when combined with all other existing and anticipated development, will not increase the water surface elevation of the base flood more than one foot at any point within the community.

8.3.4.3 Establishment of Development Permit

A development permit shall be required to ensure conformance with the provisions of Section 8.3.

8.3.4.4 Permit Procedures

- 1. Application for a development permit shall be presented to the floodplain administrator on forms furnished by him and may include, but shall not be limited to, plans in duplicate drawn to scale showing the location, dimensions, and elevation of proposed landscape alterations, existing and proposed structures, and the location of the foregoing in relation to areas of special flood hazard. Additionally, the following information is required.
 - a. Elevation (in relation to mean sea level), of the lowest floor (including basement) of all new and substantially improved structures;
 - b. Elevation in relation to mean sea level to which any non-residential structure shall be flood-proofed;
 - c. A certificate from a registered professional engineer or architect that the non-residential flood proofed structure shall meet the flood proofing criteria of Article 8 of these regulations;

- d. Description of the extent to which any watercourse or natural drainage will be altered or relocated as a result of proposed development;
- e. Maintain a record of all such information in accordance with 8.3.4.2, paragraph 1 of these regulations;
- 2. Approval or denial of a development permit by the floodplain administrator shall be based on all of the provisions of this section and the following relevant factors:
 - a. The danger to life and property due to flooding or erosion damage;
 - b. The susceptibility of the proposed facility and its contents to flood damage and the effect of such damage on the individual owner;
 - c. The danger that materials may be swept onto other lands to the injury of others;
 - d. The compatibility of the proposed use with existing and anticipated development;
 - e. The safety of access to the property in times of flood for ordinary and emergency vehicles;
 - f. The costs of providing governmental services during and after flood conditions including maintenance and repair of streets and bridges, and public utilities and facilities such as sewer, gas, electrical and water systems;
 - g. The expected heights, velocity, duration, rate of rise and sediment transport of the flood waters and the effects of wave action, if applicable, expected at the site;
 - h. The necessity of a waterfront location for the facility, where applicable;
 - I. The availability of alternative locations, not subject to flooding or erosion damage, for the proposed use;
 - j. The relationship of the proposed use to the comprehensive plan and other provisions of this Comprehensive Land Use Regulations Ordinance for that area.

8.3.4.5 Variance Procedures

- 1. The Zoning Board shall hear and render judgment on requests for variances from the requirements of this section.
- 2. The Zoning Board shall hear and render judgement on an appeal only when it is alleged there is an error in any requirement, decision, or determination made by the Floodplain Administrator (Building Inspector) in the enforcement or administration of Section 8.3.
- 3. Any person or persons aggrieved by the decision of the Zoning Board may appeal such decision in a court of competent jurisdiction.
- 4. The floodplain administrator shall maintain a record of all actions involving an appeal and shall report variances to the Federal Emergency Management Agency upon request.
- 5. Variances may be issued by the Zoning Board for the reconstruction, rehabilitation or restoration of structures listed on the National Register of Historic Places or the State Inventory of Historic Places, without regard to the procedures set forth in this Section 8.3.
- 6. Variances may be issued for new construction and substantial improvements to be erected on a lot of one-half (½) acre or less in size contiguous to and surrounded by lots with existing structures constructed below the base flood level, providing the relevant factors in Section 8.3.4.4, item 2 of this article have been fully considered. As the lot size increases beyond the one-half (½) acre, the technical justification required for issuing the variance increases.
- 7. Upon consideration of the factors noted above and the intent of this section, the board may attach such conditions to the granting of variances as it deems necessary to further the purpose and objectives of this section.
- 8. Variances shall not be issued by the Zoning Board within any designated floodway if any increase in flood levels during the base flood discharge would result.
- 9. Requirements for granting variances to the provisions of this Section 8.3.
 - a. Variances shall only be issued upon a determination that the variance is the minimum necessary, considering the flood hazard, to afford relief.
 - b. Variances shall only be issued upon, (I) showing a good and sufficient cause; (ii) a determination that failure to grant the variance would result in

exceptional hardship to the applicant, and (iii) a determination that the granting of a variance will not result in increased flood heights, additional threats to public safety, extraordinary public expense, create nuisances, cause fraud on or victimization of the public, or conflict with existing federal, state or local laws or ordinances.

- c. Variances shall only be issued to construct the lowest flood elevation a maximum of two (2) feet below the base flood elevation.
- d. Any applicant to whom a variance is granted shall be given written notice that the structure will be permitted to be built below the base flood elevation, and that the cost of flood insurance will be commensurate with the increased risk resulting from the reduced lowest elevation.
- 10. Variances may be issued by a community for new construction and substantial improvements and for other development necessary for the conduct of a functionally dependent use provided that (I) the criteria outlined in this article are met, and (ii) the structure or other development is protected by methods that minimize flood damages during the base flood and creates no additional threats to public safety.

8.3.5 Provisions for Flood Hazard Reduction

8.3.5.1 General Standards

In all areas of special flood hazards the following provisions are required for all new construction and substantial improvements:

- 1. All new construction or substantial improvements shall be designed (or modified) and adequately anchored to prevent flotation, collapse or lateral movement of the structure resulting from hydrodynamic and hydrostatic loads, including the effects of buoyancy;
- 2. All new construction or substantial improvements shall be constructed by methods and practices that minimize flood damage;
- 3. All new construction or substantial improvements shall be constructed with materials resistant to flood damage;
- 4. All new construction or substantial improvements shall be constructed with electrical, heating, ventilation, plumbing, and air conditioning equipment and other service facilities that are designed and/or located so as to prevent water

from entering or accumulating within the components during conditions of flooding;

- 5. All new and replacement water supply systems shall be designed to minimize or eliminate infiltration of flood waters into the system;
- 6. New and replacement sanitary sewage systems shall be designed to minimize or eliminate infiltration of flood waters into the system and discharge from the systems into flood waters; and
- 7. On-site waste disposal systems shall be located to avoid impairment to them or contamination from them during flooding.

8.3.5.2 Specific Standards

In all areas of special flood hazards where base flood elevation data has been provided as set forth in Sect. 8.3.4.2, paragraph 8 or Sect. 8.3.5.3, the following provisions are required:

- 1. Residential construction New construction and substantial improvement of any residential structure shall have the lowest floor (including basement), elevated to 12" above the base flood elevation. A registered professional engineer, architect, or land surveyor shall submit a certification to the Floodplain Administrator that the standard of this subsection as proposed in Section 8.3.4.4, paragraph 1.a. is satisfied.
- 2. Non-residential construction - New construction and substantial improvements of any commercial, industrial or other non-residential structure shall either have the lowest floor (including basement) elevated to 12" above the base flood level or, together with attendant utility and sanitary facilities, be designed so that below the base flood level the structure is watertight with walls substantially impermeable to the passage of water and with structural components having the capability of resisting hydrostatic and hydrodynamic loads and effects of buoyancy. A registered professional engineer or architect shall develop and/or review structural design, specifications, and plans for the construction, and shall certify that the design and methods of construction are in accordance with accepted standards of practice as outlined in this subsection. A record of such certification which includes the specific elevation (in relation to mean sea level) to which such structures are floodproofed shall be maintained by the Floodplain Non-residential or otherwise previously non-inhabited Administrator. structures such as but not limited to accessory buildings, which do not meet the minimum requirements of the provisions of Section 8.3, shall not be

converted to habitable structures except in accordance with the regulations of Section 8.3.

- 3. <u>Enclosures</u> New construction and substantial improvements, with fully enclosed areas below the lowest floor that are subject to flooding shall be designed to automatically equalize hydrostatic flood forces on exterior walls by allowing for the entry and exit of floodwaters. Designs for meeting this requirement must either be certified by a registered professional engineer or architect and meet or exceed the following minimum criteria:
 - a. A minimum of two (2) openings having a total net area of not less than one square inch for every square foot of enclosed area subject to flooding shall be provided.
 - b. The bottom of all openings shall be no higher than one foot above grade.
 - c. Openings may be equipped with screens, louvers, valves, or other coverings or devices provided that they permit the automatic entry and exit of flood waters.

4. <u>Manufactured/Mobile Homes</u>

Manufactured/mobile homes shall not be placed in A or V Flood Zones.

- a. Any manufactured/mobile homes that are placed within Flood Zone B or C shall be installed using methods and practices which minimize flood damage. For the purpose of this requirement, manufactured/mobile homes must be elevated and anchored to resist flotation, collapse, or lateral movement. Methods of anchoring may include, but are not limited to, use of over-the-top or frame ties to ground
- anchors. This requirement is in addition to applicable state and local anchoring requirements for resisting wind forces.
- b. All manufactured homes used for residential purposes shall be in compliance with the provisions of this article and any other provisions of the Comprehensive Land Use Regulations which apply, including but not limited to Article 7, Zoning District Regulations.
- c. All manufactured homes to be placed or substantially improved within Zones A, A1-30, AH and AE on the community's FIRM shall be elevated on a permanent foundation such that the lowest floor of the manufactured home is at or above the base flood elevation; and shall be securely anchored to an

adequately anchored foundation system in accordance with the provisions of sub-section a, above.

8.3.5.3 Standards for Subdivision Proposals

- 1. All subdivision proposals including manufactured home parks and subdivisions shall be consistent with 8.3.1.2, 8.3.1.3 and 8.3.1.4 of this Section 8.3.
- 2. All proposals for the development of subdivisions including manufactured home parks and subdivisions shall meet development permit requirements of 8.3.4.3 & 8.3.4.4.
- 3. Base flood elevation shall be calculated for subdivision proposals and other proposed development including manufactured home parks and subdivisions which are greater than fifty (50) lots or five (5) acres, whichever is lesser, if not otherwise provided pursuant to section 8.3.3.2 or section 8.3.4.2, paragraph 8 of this Section 8.3.
- 4. All subdivision proposals including manufactured home parks and subdivisions shall provide adequate drainage to reduce exposure to flood hazards.
- 5. All subdivision proposals including manufactured home parks and subdivisions shall have public utilities and facilities such as sewer, gas, electrical and water systems located and constructed in a manner to minimize or eliminate flood damage.

8.3.5.4 Standards for Areas of Shallow Flooding (AO/AH Zones)

Located within the areas of special flood hazard established in section 8.3.3.2 of this Section 8.3, are areas designated as shallow flooding (AO/AH zones). These areas have special flood hazards associated with base flood depths of one to three (3) feet where a clearly defined channel does not exist and where the path of flooding is unpredictable and where velocity flow may be evident. Such flooding is characterized by ponding or sheet flow; therefore, the following provisions apply:

- 1. All new construction and substantial improvements of residential structures shall have the lowest floor (including basement) elevated above the highest adjacent
 - grade at least as high as the depth number specified in feet on the community's FIRM (a minimum of two (2) feet if no depth number is specified).

- 2. All new construction and substantial improvements of non-residential structures;
 - a. Shall have the lowest floor (including basement) elevated above the highest adjacent grade at least as high as the depth number specified in feet on the community's FIRM (a minimum of two (2) feet if no depth number is specified), or;
 - b. Together with attendant utility and sanitary facilities shall be designed so that below the base flood level the structure is watertight with walls substantially impermeable to the passage of water and with structural components having the capability of resisting hydrostatic and hydrodynamic loads of effects or buoyancy.
- 3. A registered professional engineer or architect shall submit a certification that the standards of this section, as proposed in Section 8.3.4.4, paragraph 1.a. are satisfied.
- 4. Drainage paths adequate to guide flood waters around and away from proposed or improved structures shall be required within Zones AH or AO. 8.3.5.5 Standards for Coastal High Hazard Areas (V flood zones).

Located within the areas of special flood hazard established in Section 8.3.3.2 of this article are areas designated as coastal high hazard areas (Zones V1-30, VE and/or V). These areas have special flood hazards associated with high velocity waters from tidal surges and hurricane wave wash; therefore, in addition to meeting all provisions outlined in this Section 8.3 the following shall also apply:

- 1. Applicant shall provide the elevation, in relation to Mean Sea Level, of the bottom of the lowest structural member of the lowest floor (excluding pilings and columns) of all new and substantially improved structures, and indicate whether or not such structures contain a basement. The Floodplain Administrator shall maintain a record of all such information.
- 2. All new construction shall be located landward of the reach of mean high tide.
- 3. All new construction and substantial improvements shall be elevated on pilings and columns so that:
 - a. The bottom of the lowest horizontal structural member of the lowest floor (excluding the pilings or columns) is elevated to or above the base flood level;

- b. The pile or column foundation and structure attached thereto is anchored to resist flotation, collapse and lateral movement due to the effects of wind and water loads acting simultaneously on all building components. Wind and water loading values shall each have a one percent chance of being equaled or exceeded in any given year (one hundred year mean recurrence interval);
- c. A registered professional engineer or architect shall develop or review the structural design, specifications and plans for the construction and shall certify that the design and methods of construction to be used comply with accepted standards of practice for meeting the provisions of (3) a, b and c of this section.
- 4. All new construction and substantial improvements shall have the space below the lowest floor either free of obstruction or constructed with non-supporting breakaway walls, open wood lattice work, or insect screening intended to collapse under wind and water loads without causing collapse, displacement, or other structural damage to the elevated portion of the building or supporting foundation system.

For the purpose of this section, a breakaway wall shall have a design safe loading resistance of not less than ten (10) and not more than twenty (20) pounds per square foot. Use of breakaway walls which exceed a design safe loading resistance of twenty (20) pounds per square foot (either by design or when so required by local or state codes) may be permitted only if a registered professional engineer or architect certifies that the designs proposed meet the following conditions:

- a. Breakaway wall collapse shall result from a water load less than that which would occur during the base flood; and
- b. The elevated portion of the building and supporting foundation system shall not be subject to collapse, displacement, or other structural damage due to the effects of wind and water loads acting simultaneously on all building components (structural and nonstructural). Maximum wind and water loading values to be used in this determination shall each have one percent chance of being equaled or exceeded in any given year (one hundred-year mean recurrence interval).
- 5. If breakaway walls are utilized, such enclosed space shall be useable solely for parking of vehicles, building access, or storage. Such space shall not be used for human habitation.

- 6. The use of fill for structural support of buildings is hereby prohibited.
- 7. Manmade alteration of sand dunes and mangrove stands which would increase potential flood damage is hereby prohibited.

8.4 TRAFFIC IMPACT ANALYSIS PROVISIONS

8.4.1 Title and Purpose

This section shall be known as the Traffic Impact Analysis Requirements.

The Traffic Impact Analysis Requirements acknowledge and respond to the relationships between land uses and the vehicular traffic generated by those uses on the public roadways. The purpose of these requirements is to adopt methods to assess and address the traffic related impacts at the time significant land use decisions are deliberated and made.

8.4.2 Traffic Impact Analysis

A Traffic Impact Analysis (TIA) is a study that provides information on the projected traffic likely to be generated by a proposed development and assesses its impact on the roadways in the immediate proximity of a proposed development. The TIA shall be designed to identify any potential traffic operational problems or concerns and recommend appropriate actions to address such problems or concerns.

A Traffic Impact Analysis required by these regulations shall be made in accordance with the provisions of this Section 8.4, based on site plans developed by applicants. The geographical area to be considered in the TIA shall be established by the Planning Commission. The TIA shall consider and account for the potential traffic to be generated by any other undeveloped sites within the established study boundaries.

8.4.3 Threshold Conditions Requiring A Traffic Impact Analysis

The purpose of this section is to describe the conditions under which a Traffic Impact Analysis (TIA) shall be required. It shall be the responsibility of the applicant to submit the data needed to determine whether or not a TIA is required under the provisions of this Section.

8.4.3.1 Definitions of Roadway Types

- 1. Arterial Streets shall be those roadways designated and classified as major arterials in the Mandeville Master Streets Plan.
- 2. <u>Local and Collector Streets</u> Any roadway not defined as an arterial Street shall be considered a local or collector Street. Local and collector streets shall be further differentiated by actual pavement widths and the predominant type of land use served.

8.4.3.2 Conditions Requiring a Traffic Impact Analysis Threshold and Operating Standards

1. Arterial Streets

- a. Non-Residential Arterials The following requirements shall apply to projects abutting a major arterial Street, along which less than seventy-five (75) percent of the frontage on the arterial Street is used or zoned as R-2 or more restrictive within five hundred (500) feet of the proposed project's property lines. A TIA shall be required if any of the following conditions exists:
- i. The existing pavement width of the arterial Street is forty-four (44) feet or wider and the expected number of trips generated by the project exceeds two thousand (2000) vehicle trips per day;
- ii. The existing pavement width of the arterial Street is forty (40) to less than forty-four (44) feet and the expected number of trips generated by the project exceed one thousand (1000) vehicle trips per day; or
- iii. The existing pavement width of the arterial Street is less than 40 feet and the expected number of trips generated by the project exceeds 650 vehicle trips per day.
- b. <u>Residential Arterials</u> The following requirements apply to projects abutting a minor arterial Street along which seventy-five (75) percent or more of the frontage of the arterial is used or zoned as R-2 or more restrictive within five hundred (500) feet of the project. A TIA shall be required if the following condition exists.

Regardless of pavement width, the expected number of trips generated by the project exceeds three hundred (300) vehicle trips per day.

2. Local and Collector Streets Serving Predominantly Single-Family Residential Land Uses.

The following applies to projects abutting a local or collector Street along which fifty (50) percent or more of the frontage is used or zoned for R-2 or more restrictive within fifteen hundred (1500) feet of the project (or to the nearest arterial whichever is less). The following does not apply to projects abutting an arterial Street and which would use one or more driveways on the arterial Street for access and egress.

a. For streets with a pavement width of less than thirty (30) feet.

The desirable operating level is six hundred (600) vehicles per day. Traffic volume in excess of twelve hundred (1200) vehicles per day is considered undesirable. A TIA is required if either of the following conditions exists:

- i. The expected number of vehicle trips generated by the project exceeds 75 vehicle trips per day over the existing use; or
- ii. With the addition of the traffic generated by the project, the traffic volume on the Street would be expected to exceed nine hundred (900) vehicles per day.
- b. For streets with a pavement of thirty (30) to less than forty (40) feet:

The desirable operating level is nine hundred (900) vehicles per day. Traffic volumes in excess of eighteen hundred (1800) vehicles per day is considered undesirable. A TIA is required if either of the following conditions exists:

- i. The expected number of vehicle trips generated by the project exceeds one hundred ten (110) vehicle trips per day over the existing use: or
- ii. With the addition of the traffic generated by the project, the traffic volume on the Street would be expected to exceed fourteen hundred (1400) vehicles per day.
- c. For streets with a pavement width of forty (40) feet or more:

The desirable operating level is two thousand (2000) vehicles per day. Traffic volume in excess of four thousand (4000) vehicles per day is considered undesirable. A TIA is required if either of the following conditions exists:

- i. The expected number of vehicle trips generated by the project exceeds four hundred (400) vehicle trips per day over the existing uses; or
- ii. With the addition of the traffic generated by the project, the traffic volume on the Street would be expected to exceed three thousand (3000) vehicles per day.
- 3. Local and Collector Street Predominantly Serving Land Uses Other than Single-Family Residential

The following applies to projects abutting a local or collector Street along which less than fifty (50) percent of the frontage is used or zoned for R-2 or more restrictive within fifteen hundred (1500) feet of the project (or to the nearest arterial, whichever is less). The following does not apply to projects abutting an arterial Street and which would use one or more driveways on the arterial Street for access and egress.

a. For streets with a pavement width of less than forty (40) feet:

A TIA is required if the expected number of vehicle trips generated by the project exceeds six hundred fifty (650) vehicle trips per day.

b. For streets with a pavement width of forty (40) feet or more:

A TIA is required if the expected number of vehicle trips generated by the project exceeds one thousand (1000) vehicle trips per day.

8.4.3.2 Data Requirements of Applicants

It shall be the responsibility of the applicant to submit at the time of application all data needed to determine whether or not a TIA will be required under the provisions of this Section. This data shall be certified by a Registered Professional Engineer or other qualified individual. The requirement that this data be provided at the time of application may be waived by the City Planner if the applicant agrees to provide a TIA for the proposed project that complies with the provisions of the section.

Estimates of the average number of vehicle trips per day expected to be generated by the project shall be based on the appropriate trip generation rate data provided in the latest edition of the <u>Trip Generation Informational Report</u> published by the Institute of Transportation Engineers (ITE). The data submitted by the applicant shall document the

specific trip generation rate (or rates) used and the specific land use assumptions made in applying the trip generation rate (or rates) in developing the estimate of average number of vehicle trips per day expected to be gene-rated by the project. If

specific information is not available on the proposed land use, the trip generation estimate shall be based on the maximum allowable density for the most intensive use.

If the provisions of conditions requiring a TIA for Local and Collector Streets Serving Predominantly Single-Family Residential Land Uses is applicable to a project, then it is the responsibility of the applicant to submit a 24-hour vehicle traffic count for the local or collector Street (or streets) on which the project abuts. This traffic count shall be collected as determined by the Director of Public Works. The 24-hour traffic count data shall be collected under the supervision of a Registered Traffic Engineer or other qualified individual.

8.4.4 Responsibility for Required Traffic Impact Analysis

If a Traffic Impact Analysis (TIA) is required for a project, the performance of the required TIA shall be the responsibility of the applicant. The TIA must be performed under the supervision of a qualified Professional Registered Civil Engineer or other qualified individual such as a transportation planner or traffic engineer. A TIA report must be prepared documenting the study, the data used, the findings and the recommendations of the study consistent with Section 8.4.5 and 8.4.6. The TIA Report shall be signed by the Registered Professional Engineer or other qualified individual responsible for the supervision of the study and the preparation of the TIA report. The applicant shall submit twenty (20) copies of the TIA report at least two weeks prior to the date on which the project is scheduled for consideration by the Planning Commission.

8.4.5 Scope of Traffic Impact Analysis and Study Area

The geographical area to be considered in the Traffic Impact Analysis (TIA) shall be established in an initial meeting with the Director of Planning. The TIA shall consider and account for the potential traffic to be generated by other undeveloped sites within the established study boundaries.

The scope of the Traffic Impact Analysis and the Traffic Impact Analysis Report shall be determined by the Planning Department and may include, but is not limited to:

- 1. Current average daily traffic counts.
- 2. Morning and evening peak hour traffic counts.

- 3. Estimated peak hour vehicle movements from the proposed project, or from the study area, whichever is applicable.
- 4. Estimated total daily trips generated by the proposed project, or from the study area, whichever is applicable.
- 5. Anticipated traffic impacts on the nearest intersections and on traffic control devices in the study area.
- 6. The impact of on-Street parking on the capacity of streets in the study area during the peak hours.
- 7. The location of churches, schools, parks and similar public or civic uses along routes serving the proposed project or within the study area where increased traffic may endanger pedestrians.
- 8. A cross section or cross sections of the Street or roadways on which access to or egress from a project is proposed showing pavement dimensions.
- 9. A plan view of the streets in the study area from which site distances can be calculated, the location of surrounding driveways examined, and adequate to assist in locating and explaining any features that may pose unusual traffic circulation or pedestrian problems.
- 10. Conclusions and recommended action.

8.4.6 Traffic Data Requirements

Traffic count data used in the TIA or submitted as part of the application shall be collected using one or more of the following methods:

- 1. Automatic twenty-four (24) hour counters.
- 2. Traffic counts conducted manually.

Vehicle trip generation estimates used in the study shall be based on the appropriate trip generation rate data provided in the latest edition of the <u>Trip Generation Information Report</u> published by the Institute of Transportation Engineers or using other trip generation rate data acceptable to the Urban Transportation Department. The TIA Report shall document the specific trip generation rates used and the specific land use assumptions used therein.

8.4.7 Application of Requirements

- 1. These TIA requirements shall apply to all land located in the City of Mandeville. Such TIA requirements shall become applicable as to each
 - individual lot at such time an application on such lot is made for a zoning change, conditional use permit, site plan approval, or building permit.
- 2. These requirements shall not apply to the following:
 - a. Building permits for single-family or duplex residences where only one such structure is constructed per lot.
 - b. Building permits for substantial restoration within a period of twelve (12) months of a building which has been damaged by fire, explosion, flood, tornado, riot, act of the public enemy, or accident of any kind.
 - c. Building permits for restoration of buildings with a historic designation.
 - d. Building permits for remodeling as long as all exterior walls of the building remain in the same location.
- 8.4.8 Actions Based on the Results of a Traffic Analysis

The Planning Commission and/or the City Council may disapprove a project subject to these Traffic Impact Analysis Requirements when the results of a Traffic Impact Analysis demonstrate that such a proposed project is likely to overburden the City's Street system.

It is the intention, however, that an applicant be allowed to modify a proposed site plan to minimize the traffic-related impacts identified as part of a required Traffic Impact Analysis.

Modifications to site plans for non-residential streets may include:

- 1. A reduction in the projected number of vehicles per day.
- 2. Dedication of additional right-of-way.
- 3. Rerouting of traffic and of proposed access and egress points serving the proposed project.

The Director of Public Works or City Engineer shall review modifications and make recommendations to the Planning Commission and/or City Council at the time of site plan reviews.

ARTICLE 9 - PARKING, LANDSCAPING AND OPEN SPACE REGULATIONS

9.1 PARKING & LOADING REGULATIONS

9.1.1 General Provisions

9.1.1.1 Off-Street Parking Required

An area suitable for parking or storing automobiles in off-street locations in accordance with the regulations provided herein shall be required in all zoning districts at the time of the initial construction of any principal building or when a structural alteration or change in use produces an increase in the parking required to serve that use. Such required parking shall be required to be installed prior to the issuance of a Certificate of Occupancy for the use which it serves.

9.1.1.2 Applicability of this Article

No land shall be used or occupied and no structure erected or used unless the off-street parking spaces required herein are provided. These regulations do not replace but act in concert with any other parking requirements under state or federal law, such as laws pertaining to providing parking for the handicapped.

1. These parking regulations do not apply to any structure or use existing at the time of enactment of these regulations. However, existing off-street parking spaces and loading spaces shall not be reduced in number or encroached upon so that the minimum requirements of this Article would not be met. A

reduction below the existing parking provisions may occur only when a change in use allows a corresponding reduction in the associated parking.

- 2. If a building, structure, or use in existence at the time of enactment of these regulations is damaged or destroyed, and the building, structure, or use can otherwise be reestablished under the provisions of these regulations then any associated off-street parking or loading spaces which existed must be retained. If an enlargement or other change in use is proposed for an existing conforming use, which increases the number of parking, loading or unloading spaces required for the use under the provisions of this , all parking and loading spaces required for the enlargement or change shall be provided in accordance with this Article. If the use is non-conforming at the time of the proposed enlargement, it shall be subject to the provisions for non-conforming uses as provided in Article 4.
- 3. Any time a use classification is changed or when a building or use is enlarged or increased, the parking and loading requirements for this Article shall apply to the enlargement or increase. A Certificate of Occupancy for the new use or area of enlargement shall not be issued until all required parking and loading spaces have been provided.

9.1.1.3 Use of Residential Parking Facilities

No parking facilities accessory to residential structures shall be used for the storage of commercial vehicles prohibited by the regulations of the residential district in which the structure is located or for the parking of automobiles belonging to employees, owners, tenants, or customers of nearby non-residential uses.

9.1.1.4 Joint Parking Facilities

Off-street parking facilities for combinations of mixed buildings, structures, or uses may be provided collectively in any district, except residential districts, in which separate parking facilities for each separate building, structure, or use will be required. The total number of spaces provided shall equal the sum of the separate requirements for each non-residential use if computed separately.

9.1.1.5 Control of Off-street Parking Facilities

When required accessory off-street parking facilities are provided elsewhere than on the lot in which the principal use served is located, the properties shall be in the same ownership or control as the property occupied by such principal use either by deed or lease the term of which approximates the expected life of the use to which the parking facilities are accessory.

The owner of the principal use shall file with the Building Inspector a written act in a form suitable for recordation in the conveyance records of St. Tammany Parish the terms of which require the owner and his or her heirs and assigns to maintain the required number of off-street spaces throughout the existence of said principal use.

9.1.1.6 Use of Parking Area

Any vehicle parking space shall be used for parking only. Any other use of such space, including use for open storage of goods, for the storage of vehicles for sale or rent, for storage of inoperable vehicles, for repair work or servicing of any kind other than in an emergency, or the requirement of any payment for the use of such space, shall be deemed to constitute a separate commercial use in violation of provisions of this Article.

9.1.1.7 No Building shall be Erected in Off-Street Parking Areas

No building of any kind shall be erected in any off-street parking area, except a parking garage containing parking spaces conforming to the requirements set forth in this Article, or a shelter house for a parking attendant, providing the number of spaces required is not reduced.

9.1.1.8 Off-street Parking Area shall not be Reduced

The required parking area on any lot, as set forth and designated in this Article, shall not be reduced or encroached upon in any manner.

9.1.1.9 Compilation of Total Employment

The number of employees shall be compiled on the basis of the number of persons employed on the premises at one time on an average day or average night, whichever is greater. Seasonal variations in employment may be recognized as determining an average day.

9.1.1.10 Parking Requirements for Uses not Specifically Listed

For purposes of determining the required number of parking spaces associated with a use, the parking space requirements for a use not specifically listed in this Article shall be the same as for a listed use of similar characteristics of parking demand generation, as determined by the City Planner.

9.1.1.11 Provisions for "Parking Bank"

Parking spaces otherwise required by this ordinance when varied by the Zoning Board or Planning Commission in accordance with the provisions of this ordinance to allow for such parking spaces to be held in reserve as landscaped open space shall be known as a "parking bank". A parking bank shall not be used for the construction of any structure which would interfere with its intended use as future parking spaces.

9.1.1.12 Truck Parking in Residential Areas

No motor vehicle over two-ton capacity or bearing a commercial license, and no commercially licensed trailer or commercial type truck trailer shall be parked or stored in a residential district except when loading, unloading, or rendering a service. This provision shall not apply to vehicles operated, parked or stored on the site of a legally non-conforming use, to recreation vehicles and pickups, or to temporary use of rental vehicles for non-commercial purposes.

9.1.1.13 Parking for Mixed Use Developments

- 1. The Planning Commission or Zoning Board, in conjunction with any approval or variance procedure applied for, may authorize an adjustment in the total parking requirement for separate uses located on the same site, or for separate uses located on adjoining sites and served by a common parking facility, pursuant to this Section. A request for such adjustment shall require submission of a site plan and transportation engineering report addressing the following relevant factors:
 - a. All parking spaces subject to adjustment under this Section shall be located in a common, contiguous parking facility providing reasonably equivalent accessibility and usability to all uses which the parking is intended to serve.
 - b. In determining whether to approve an adjustment for mixed use developments, the Planning Commission or Zoning Board shall consider all relevant factors, including:
 - i. The characteristics of each use and the differences in projected peak parking demand, including days or hours of operation.
 - ii. Potential reduction in vehicle movements afforded by multipurpose use of the parking facility by employees, customers, or residents of the uses served.
 - iii. Potential improvements in parking facility design, circulation, and access afforded by a joint parking facility.
 - iv. The report and recommendation of the City Planner.

9.1.1.14 Handicapped Facilities

In each parking facility of 20 or more spaces, a portion of the total parking spaces should be specifically designed, located, and reserved for vehicles licensed by the State for use by the handicapped, according to the following schedule or the current state or federal requirements, whichever is greater:

<u>Total Spaces</u>	Minimum Number of Handicapped Spaces Required
19 or less 20 - 50 51 - 100 101 - 150 151 or greater	0 1 4% of total in excess of 50 3% of total in excess of 50 2% of total in excess of 50

9.1.2 Construction Design Standards for Parking and Loading

- 1. <u>Standards</u> Design standards are established by this Section to set basic minimum dimensions and guidelines for design, construction, and maintenance of parking and loading facilities.
- 2. <u>Parking and Loading Space Dimensions</u> The following basic dimensions shall be observed for parking spaces and loading spaces.
 - a. Each standard parking space shall consist of a rectangular area not less than 8.5 feet wide by 18.5 feet long. Each space shall have a vertical clearance of not less than 7.5 feet. Each space shall be independently accessible.
 - b. Each parking space designated for use by the handicapped shall consist of a rectangular area not less than 12.5 feet wide by 18.5 feet long, with a vertical clearance of 7.5 feet, shall be located in an area not exceeding a 2 percent slope, and shall be located near and convenient to a level or ramped entrance accessible to handicapped persons. Parking spaces for the handicapped shall be signed and restricted for use by the handicapped only.
 - c. Each off-street loading space shall consist of a rectangular area not less than 12 feet wide and 55 feet long, with a vertical clearance of not less than 15 feet exclusive of aisles, accessways or maneuvering space.
 - d. Off-street parking spaces shall have access from driveways on the development site and not directly from the public streets. Each parking and

loading space shall have adequate drives, aisles, and turning and maneuvering areas for access and usability, and shall at all times have adequate access to a public street. Such access to a public street shall not be less than 20 feet nor exceed 35 feet in width for two-way access or be less than 12 feet not exceed 15 feet in width for a one-way access.

3. <u>Parking Facility Design</u> - Minimum parking facility design standards are illustrated in the following table. Additional supplemental guidelines and standards for parking facility design, internal layout, acceptable turning radii and pavement slope, vehicular and pedestrian circulation, and other design features may be adopted by resolution of the Planning Commission.

Parking lots developed on contiguous parcels of property shall be designed to accomplish circulation between and among the parking lots without the use of public streets.

- 4. <u>Paving and Drainage</u> The following basic standards shall be observed:
 - a. In all districts, all loading facilities shall be surfaced and maintained with asphaltic, concrete, or other permanent hard surfacing material sufficient to prevent mud, dust, loose material, and other nuisances. Materials may be pervious such as grass pavers.
 - b. Loose aggregate surfacing may be permitted for up to eight (8) spaces.
 - 3. Loose aggregate surfacing in excess of eight (8) spaces will be permitted provided the driveways are surfaced with asphalt, concrete, or other permanent surfacing material.
 - 1. No base preparation will be allowed within the dripline of a tree
 - 2. Loose aggregate surfacing shall be contained by cubing. Curbing shall not sever roots 2" diameter or greater or penetrate natural grade greater than 3" in depth within the drip line of a tree.
- 19. ii Drip line of trees encroached upon by parking shall be depicted on the site plan.
 - d. All parking and loading facilities shall be graded and provided with permanent storm drainage facilities, according to construction specifications approved by the Public Works Director. Surfacing, curbing, and drainage improvements shall be sufficient to preclude free flow of water onto adjacent properties or public streets or alleys, and to provide adequate drainage.

	A	В	С	D	E Width	F
	Angle o	fWidth	Depth of	Width	of Stall	
	Parking		Stall 90°	of	Parallel	Module
	(Degree		to Aisle	<u>Aisle</u>	to Aisle	<u>Width</u>
		,				
Standard Parking Spaces						
	30	8.5'	16.9'	12.5'	17.0'	47'
	30	9.0	17.3	12.5	18.0	47
	30	9.5	17.8	12.5	19.0	48
	30	10.0	18.3	12.5	20.0	49
	45	8.5	17.5	13.0	12.0	48
	45	9.0	17.5	12.5	12.7	48
	45	9.5	17.5	12.5	13.4	48
	45	10.0	17.5	12.5	14.1	48
	60	8.5	19.0	18.0	9.8	56
	60	9.0	19.0	16.0	10.4	54
	60	9.5	19.0	15.0	11.0	53
	60	10.0	19.0	15.0	11.6	53
	75	8.5	19.5	25.0	8.8	64
	75	9.0	19.5	23.0	9.3	62
	75	9.5	19.5	22.0	9.8	61
	75	10.0	19.5	22.0	10.3	61
	90	8.5	18.5	28.0	8.5	65
	90	9.0	18.5	26.0	9.0	63
	90	9.5	18.5	25.0	9.5	62
	90	10.0	18.5	25.0	10.0	62
Parallel Parking Spaces						
	0	8.5	8.5	12.5	22.0	21
	(width)			(length)	

- 5. <u>Safety Features</u> Parking and loading facilities shall meet the following standards:
 - a. Safety barriers, protective bumpers or curbing, and directional markers shall be provided as needed to assure safety, prevent encroachment onto adjoining public or private property.
 - b. Visibility of and between pedestrians, bicyclists, and motorists shall be assured when entering individual parking spaces, when circulating within a parking facility, and when entering and exiting a parking facility.
 - i. <u>Sight Triangle at Street and Accessway Intersection</u> Where an accessway intersects a street there shall remain clear of obstruction to vision between the height of 3 feet and 7 feet above the centerline grade of the street a sight triangle measuring 25 feet on the side adjacent to the street and 15 feet on the side adjacent to the accessway.
 - ii. <u>Sight Triangle at Accessway Intersection</u> Where an accessway intersects another accessway, there shall remain clear of obstruction to vision between the height of 3 feet and 7 feet from the grade of the accessways a sight triangle measuring 15 feet along both of its sides adjacent to the intersecting accessways.
 - c. Internal circulation patterns, and the location and traffic direction of all access drives, shall be designed and maintained in accordance with accepted principles of traffic engineering and traffic safety.
- 6. Parking Lot Lighting Parking lot lighting shall meet the following standards:
 - a. Parking lot lighting design shall provide for the reasonable safety, comfort, and convenience of the parking of patrons and use of pedestrians.
 - b. Parking lot lighting illumination design levels and visibility glare shall in general comply with the latest issue of IES Lighting Handbook Section on Parking Facilities Lighting.
 - c. Parking lot and loading space lighting shall be designed to minimize light spill over into adjoining streets and nearby residential areas and shall be directed downward and away from adjoining property and abutting streets by shielding the light source from visibility from adjoining properties or streets in such a way as not to create a nuisance. All exterior lighting shall be hooded

or shielded so that the light source is not visible from adjacent more restrictive residential districts.

d. Parking lots developed on contiguous parcels of property shall be designed to accomplish circulation between and among the parking lots without the use of public streets

7. <u>Fencing and Screening</u>

- a. A parking facility in any nonresidential district which adjoins or abuts property in a residential district shall have a wall or fence not less than 4 feet in height located for the length of the common boundary.
- b. A parking facility in any residential district which has more than 10 spaces and which adjoins or abuts other property in a residential district shall have a wall or fence or landscape screen not less than 4 feet in height located for the length of the common boundary.
- 8. <u>Noise</u> Areas used for primary circulation, for frequent idling of vehicle engines, or for loading activities shall be designed and located to minimize impacts on adjoining properties, including provisions for screening or sound baffling.
- 9. <u>Maintenance</u> All parking and loading facilities shall be maintained to assure desirability and usefulness of the facility. Such facilities shall be maintained free of refuse, debris, or other accumulated matter and shall at all times be available for the off-street parking or loading use for which they are required or intended.
- 10. <u>Adjustments</u> For a use of a site subject to Administrative or Zoning Permit Review or a Conditional Use Permit, the minimum requirements of this Section may be adjusted in their application, provided such change is determined by the permitting authority, whichever is applicable, to provide improved design, usability, attractiveness, and protection to adjoining uses, in a manner equal to or greater than the specific requirements of this Section.

9.1.3 Location and Maintenance

9.1.3.1 General Location

Off-street parking facilities required herein shall be located as follows:

- 1. Required off-street parking spaces for mobile homes, and for one- and two-family dwellings, shall be located on the same lot(s) as or lot(s) that abut the lot on which the principal use is located.
- 2. Required parking spaces for multiple-family dwellings shall be located within two hundred (200) feet of the main entrance to the principal building served. Such spaces may be located on another lot(s) not contiguous to the lot on which the principal use is located as long as the spaces are within two hundred (200) feet of the main entrance to the principal building served.
- 3. Required parking spaces for non-residential uses shall be located within four hundred (400) feet of the main entrance to the principal use served. Such spaces may be located on another lot(s) not contiguous to the lot on which the principal use is located if the spaces are within four hundred (400) feet of a main entrance to the principal building served.
- 4. Where required off-street parking is located on a lot other than the lot occupied by the use requiring it, site plan approval for both lots shall be required.
- 5. There shall be no off-street parking space within fifteen (15) feet of any street right-of-way except as specifically provided by this.
- 6. Access drives shall be no closer than five (5) feet to interior property lines and fifteen (15) feet to side street property lines on local streets and shall minimally meet the requirements of the State Department of Transportation and Development for collector and arterial streets.

9.1.3.2 Maintenance

It shall be the joint and solidum responsibility of the operator and owner of the principal use, uses, and/or buildings to maintain, in a neat and serviceable manner, the parking spaces, access-ways, landscaping, fences and buffering materials serving such use or building.

9.1.4 Minimum Off-Street Parking Requirements by Use

The use classifications of Article 6 are referenced to the parking requirements set forth in this Article. The uses listed are illustrative only; the provisions of the use classification system shall prevail for any use not specifically referenced.

USE CLASSIFICATION

6.2	RESIDENTIAL	
6.2.1	Single-Family Residential	2 per dwelling unit
6.2.2	Cluster Residential	2 per dwelling unit
6.2.3	Duplex Residential	2 per dwelling unit
6.2.4	Two-Family Residential	2 per dwelling unit
6.2.5	Townhouse Residential	2 per dwelling unit
6.2.6 6.2.7 6.2.8	Condominium Residential Multiple-Family Residential Congregate Living Residential	1.5 per efficiency unit; 2 per 1-3 bedroom dwelling unit and 1 per additional bedroom
6.2.9	Community Residential	2 per dwelling unit
6.2.10	Mobile Home Residential	2 per dwelling unit
6.2.11	Accessory Residential	2 per dwelling unit
6.2.12	Boathouse Residential	2 per dwelling unit

6.3	CIVIC		
6.3.1	Administrative Services	1 per 200 s.f. of gross floor area or 1 per 50 s.f. of public assembly area,	
whichever		50 s.f. of public assembly area, Is greater	
6.3.2	Cemetery Without mortuary	1 per employee or a minimum of three spaces	
	With mortuary	See 6.4.36 Funeral Services of this Table	
6.3.3	Club or Lodge	1 per 50 s.f. of assembly area or 1 per 150 s.f. of gross floor area of when community recreation facilities are provided, the parking required for such community recreation facility as provided herein, whichever is greater. See 6.3.6, Community Recreation.	
6.3.4	College & University Facilities	8 per classroom, lab, studio or other instructional facility plus 1 per faculty and administrative staff member or the required parking spaces for any auditorium, gymnasium or stadium facility on the site, whichever is greater.	
6.3.5	Community Parking Facilities	A minimum of 3 spaces	
6.3.6	Community Recreation Administrative Offices	1 per 200 s.f. of gross floor area	
	Spectator Sports & Entertainment Facilities such as gymnasiums and auditoriums	1 per 4 fixed seats (as defined), or 1 per each 50 s.f. of assembly area where there are no fixed seats	
	Athletic Fields and Stadiums	20 per play field or diamond or 1 per 4 fixed sets, whichever is greater	
	Tennis, Racquetball &	2 per court	

Other Courts

1 per 50 s.f. per room
1 per 100 s.f. of water surface
6 per golf hole plus 50 percent of spaces otherwise required for accessory uses such as taverns, restaurants, clubhouses & retail facilities
1 per 8 beds and 1 space per employee at largest shift
10 spaces minimum plus 1 additional space for each 400 s.f. of gross floor area in excess of 2,500 s.f
1 per each 350 s.f. of care, instructional or Play area
1 per anticipated essential service vehicle regularly parking at site

6.3.12 Group Care Facilities	1 per 8 beds and 1 space per employee at largest shift
6.3.13 Hospital Services (General)	1 per 2 beds plus 2 per 200 s.f. of gross floor area
6.3.14 Hospital Services (Limited) (out-patient only)	1 per 200 s.f. of gross floor area plus 12 per employee at largest shift
6.3.15 Major Impact Services & Utilities	As approved by the City Council based on the type of facility, the anticipated number of employees working at the facility and the peak operational parking demand
6.3.16 Minor Impact Utilities	1 per service vehicle anticipated to regularly park at site
6.3.17 Postal & Parcel Delivery Services	1 per delivery vehicle maintained or stored on the site plus 1 per 200 s.f. of gross floor area
6.3.18 Public and Private Primary Educational Facilities including	2 per classroom plus 1 per faculty and administrative staff member, or the required
kindergarten, elementary, middle & junior high schools	parking spaces for any auditorium or gymnasium facility on the site, whichever is greater
6.3.19 Public and Private Secondary Educational Facilities including Senior High	8 space per classroom, lab, or other instructional facility plus 1 per faculty and administrative staff member or the required Parking spaces for any auditorium, or gymnasium or stadium facility on the site, whichever is greater
6.3.20 Public Recreation & Park Services	Same as 6.3.6, Community Recreation
6.3.21 Public Safety Services	1 per public safety vehicle maintained or stored on the site plus 1 per 200 s.f. of gross floor area

6.3.22	Religious Assembly	1 per each 4 fixed seats in primary assembly area or 1 space per 50 s.f. of assembly area where there are no fixed seats not less than 20 spaces minimum
6.4	COMMERCIAL	
6.4.1	Administrative & Business Offices	1 per 200 s.f. of gross floor area
6.4.2	Adult Uses Adult Arcade	1 per 50 s.f. of rooms with billiard or arcade Facilities
	Adult Bookstore	1 per 250 s.f. of gross leasable area
	Adult Cabaret	1 per 100 sf of gross floor area
	Adult Theatre	1 per 4 fixed seats or 1 per 50 s.f. of Assembly area where there are no fixed
seats		
	Adult Encounter Establishment	1 per guest rom plus 1 per 200 s.f. of gross floor area excluding guest rooms
6.4.3	Agricultural Sales and Services	1 per 400 s.f. of gross floor area plus 1 per delivery vehicle or customer loading station
6.4.4	Art and Craft Studio (Limited)	1 per 400 s.f. of gross floor area
6.4.5	Art and Craft Studio (General)	1 per 400 s.f. of gross floor area
6.4.6	Animal Kennels	1 per 400 s.f. of gross floor area
6.4.7	Animal Sales & Services (Limited)	1 per 200 s.f. of gross floor area
6.4.8	Animal Sales & Services (General)	1 per 200 s.f. of gross floor area plus 2 per Large animal stall

6.4.9	Automotive & Equipment Services-Washing	4 per washing bay and 1 per 200 s.f. of retail floor area
6.4.10	Automotive & Equipment Services-Fuel Station	4 per service bay and 1 per 200 s.f. of retail floor area
6.4.11	A & E Services-Auto Service Station	4 per service bay and 1 per 200 s.f. of Retail floor area
6.4.12	A & E Services-Truck Service Station	Parking to be determined in accordance with Parking uses included at the station such as Restaurants, lodging or retail areas in addition to spaces adequate to serve the anticipated volume of truck traffic; to be sized in accordance with loading space requirements
6.4.13	A & E Services-Commercial Parking	No minimum
6.4.14	A & E Service-Sales & Rentals	1 per 400 s.f. of gross retail/office sales area Plus 4 per service bay and 1 per maximum anticipated number of vehicles for sale to be located on the site
	A & E Service-Sales & Rentals A & E Services-Equipment Sales	area Plus 4 per service bay and 1 per maximum anticipated number of vehicles for sale to
6.4.15		area Plus 4 per service bay and 1 per maximum anticipated number of vehicles for sale to be located on the site 1 per 500 s.f. of gross leasable area plus
6.4.15 6.4.16	A & E Services-Equipment Sales A & E Services-Auto and	area Plus 4 per service bay and 1 per maximum anticipated number of vehicles for sale to be located on the site 1 per 500 s.f. of gross leasable area plus adequate space to store equipment for sale 1 per 400 s.f. of gross office/retail area plus
6.4.15 6.4.16 6.4.17	A & E Services-Equipment Sales A & E Services-Auto and Equipment Repair (Enclosed)	area Plus 4 per service bay and 1 per maximum anticipated number of vehicles for sale to be located on the site 1 per 500 s.f. of gross leasable area plus adequate space to store equipment for sale 1 per 400 s.f. of gross office/retail area plus 4 per service bay 1 per 400 s.f. of gross office/retail area plus

6.4.20 Business Support Services-General 1 per 400 s.f. of gross floor area 6.4.21 Business Support Services-Limited 1 per 400 s.f. of gross floor area 6.4.22 Business or Trade School 10 per classroom, shop or lab used for Instruction 6.4.23 Commercial Recreation -**Indoor Sports:** Billiards & Amusement Arcades 1 per 50 s.f. of rooms with billiard or Arcade facilities **Bowling Alleys & Skating Rinks** 4 per lane or per 500 s.f. of rink area plus 50 percent of spaces otherwise required for accessory uses such as taverns or restaurants as provided herein Athletic Clubs Parking spaces in accordance with the required parking for community recreation facilities as provided herein 6.4.24 Commercial Recreation -**Outdoor Sports** Parking spaces in accordance with the required parking for Community Recreation facilities as provided herein 6.4.25 Commercial Recreation -**Indoor Entertainment:** 1 per 4 fixed seats or 1 per 50 s.f. of assembly area where there are no fixed Theaters, Bingo, Convention seats Reception & Dance Halls 6.4.26 Commercial Recreation -Outdoor Entertainment Parking spaces in accordance with the parking required for Community Recreation facilities as provided herein **Amusement Parks** 4 per amusement ride or activity area plus 50 percent of spaces otherwise required for accessory uses as restaurants or retail spaces as provided herein

6.4.27	Communications Services	1 per 400 s.f. of gross floor area	
6.4.28	Construction Sales and Services	1 per 400 s.f. of gross floor area	
6.4.29	Construction Sales and Services (Enclosed)	1 per 400 s.f. of gross floor area	
6.4.30	Consumer Repair Services	1 per 400 s.f. of gross floor area	
6.4.31	Convenience Storage (Mini-Warehouse)	Per Special Use Criteria of Article 8	
6.4.32	Day Care Centers - Commercial	1 per 350 s.f. of care, instructional or play Area plus 2 covered spaces at entrance for Transfer of participants	
6.4.33	Exterminating Services	1 per 400 s.f. of gross floor area	
6.4.34	Financial Services	1 per 200 s.f. of gross floor area	
6.4.35	Food Sales	1 per 200 s.f. of gross floor area	
6.4.36	Funeral Services	1 per 50 s.f. of public assembly area including lobbies, parlors or chapels plus 1 per vehicle maintained or stored on the premises	
6.4.37	General Retail Sales (Convenience)	1 per 200 s.f. of gross floor area	
6.4.38	General Retail Sales (General)	1 per 250 s.f. of gross floor area	
6.4.39	General Retail Sales (Bulk)	1 per 400 s.f. of gross floor area	
6.4.40	Lodging (Transient) - Bed & Breakfast Inn	1 per guest room plus 2 spaces for resident occupants	
6.4.41	Lodging (Transient) - Bed & Breakfast Residence	1 per guest room plus 2 spaces for resident occupants	
6.4.42	Lodging (Transient) - Campground	1 per designated lodging space plus 1 per 200 s.f. of gross floor area	

6.4.43	Lodging (Transient) - Hotel/Motel	1 per each lodging units plus 1 per 200 s.f. Of gross floor area including restaurant, Lobby & meeting rooms excluding guest rooms and access halls
6.4.44	Laundry Services - Coin-Operated	1 per 400 s.f. of gross floor area
6.4.45	Laundry Services - Neighborhood	1 per 400 s.f. of gross floor area
6.4.46	Laundry Services - Commercial	1 per 600 s.f. of gross floor area
6.4.47	Marine Services - Boat Fuel Area	1 per 200 s.f. of gross floor area
6.4.48	Marine Services - Boat Sales/Service	1 per 200 s.f. of gross floor area of building Area plus 4 spaces for each service bay plus 1 per maximum anticipated number of boats for sale to be located on the site
6.4.49	Marine Services - Boat Repairs	1 per 200 s.f. gross floor area of building Area plus 4 spaces for each service bay plus spaces for boat sales as provided herein if boats are sold from the site
6.4.50	Marine Services - Commercial and Charter Fishing	1 per 200 s.f. of gross building area or 3 minimum
6.4.51	Marine Services - Incidental Storage	1 per 200 s.f. of gross building area or 3 Minimum
6.4.52	Marine Services - Marinas	1 per every 4 boat slips
6.4.53	Marine Services - Retail	1 per 400 s.f. of gross floor area
6.4.54	Marine Services - Restaurant	1 per lodging unit plus 1 per 200 s.f. of gross

	/Transient Lodging	Floor area including offices, restaurants, lobbies, and other use areas, excluding s.f. area of lodging units
6.4.55	Marine Services - Yacht Clubs	1 space per 3 boats slips plus 1 space per 2 200 s.f. of gross floor area or per parking required herein for Community Recreation facilities associated with the yacht club, whichever is greater
6.4.56	Medical Services	1 per 200 s.f. of gross building area
6.4.57	Personal Improvement Services	1 per 100 s.f. of gross building area
6.4.58	Personal Services	1 per 200 s.f. of gross building area
6.4.59	Research Services	1 per 300 s.f. of gross building area
6.4.60	Restaurants - Drive-In	1 per 200 s.f. of gross building area
6.4.61	Restaurants - Fast-Food	1 per 150 s.f. of gross building area
6.4.62	Restaurants - Outdoor Fast-Food	1 per table or 8' of bench
6.4.63	Restaurants - Sit-Down	1 per 200 s.f. of gross building area
6.4.64	Restaurants - Sit-Down/Lounge	1 per 150 s.f. of gross building area
6.4.65	Swap Meets - Enclosed	2 spaces per booth or participant
6.4.66	Swap Meets - Unenclosed	2 spaces per booth or participant
6.4.67	Shopping Center - Neighborhood	4 spaces per 1,000 s.f. of gross floor area
6.4.68	Shopping Center - Minor	4.5 spaces per 1,000 s.f. of gross floor area
	Shopping Center - Major Tavern - Bar or Lounge	4.5 spaces per 1,000 s.f. of gross floor area 1 per 150 s.f. of gross floor area
6.4.71	Night Club	1 per 100 s.f. of gross floor area

6.5	<u>INDUSTRIAL</u>		
6.5.1	Basic Industrial	1 per each 1,000 s.f. of gross floor area or 1 per employee, whichever is greater	
6.5.2	Custom Manufacturing	1 per each 1,000 s.f. of gross floor area or 1 per employee, whichever is greater	
6.5.3	Light Manufacturing (Enclosed)	1 per each 400 s.f. of gross floor area or 1 per 2 employees, whichever is greater	
6.5.4	Light Manufacturing (Open)	1 per each 400 s.f. of gross floor area or 1 per 2 employees, whichever is greater	
6.5.5	Research Services - Hazardous	1 per each 400 s.f. of gross floor area or 1 per 2 employees, whichever is greater	
6.5.6	Resource Extraction	1 per each 400 s.f. of gross floor area or 1 per 2 employees, whichever is greater	
6.5.7	Wholesale, Storage, and Distribution - Light	1 space per each 5,000 s.f. of gross floor Area dedicated to storage	
6.5.8	Wholesale, Storage, and Distribution - Heavy	1 space per each 5,000 s.f. of gross floor area dedicated to storage, and adequate Space for parking and storage of all vehicles associated with the use	
6.6	AGRICULTURAL		
6.6.1	Animal Production	1 per each 400 s.f. of gross floor area	
6.6.2	Animal Waste Processing	1 per each 400 s.f. of gross floor area	
6.6.3	Aquaculture	1 per each 400 s.f. of gross floor area	
6.6.4	Horticulture - Cultivation	1 per 2 employees	
6.6.5	Horticulture - Storage	1 per 2 employees	
6.6.6 6.6.7	Horticulture - Storage (Enclosed) Packing and Processing - Limited	1 per 2 employees 1 per 2 employees	

6.6.8	Packing and Processing - General	1 per 2 employees
6.6.9	Row and Field Crops	1 per 2 employees
6.6.10	Tree Crops	1 per 2 employees
6.7	COMBINED	
6.7.1	Combined Uses -	On sites with combined uses parking
	Residential/Office	spaces For residential use shall be provided
6.7.2	Combined Residential/	according to the type of residential use classification and the parking
	Commercial Uses	requirements Provided herein; al other use types shall be In accordance with the required parking
for 6.7.3	Planned Combined Uses	the specific use proposed except as provided For in Section 9.1.1.13 Parking for Mixed Use Developments
6.8	ACCESSORY USES	
6.8.1	Accessory Uses - General	In accordance with the required parking applicable to the specific use as provided herein

9.1.5 Off-street Loading and Unloading Areas

Areas suitable for loading and unloading motor vehicles in off-street locations and specifically designated for this purpose shall hereafter be required at the time of initial construction, converor alteration of any structure used or arranged to be used for commercial, industrial, institutional or multi-family residential purposes. Such off-street loading areas shall have access to a public street and shall be provided and maintained in accordance with the following requirements, the area of which shall not be included in the off-street parking requirements.

9.1.5.1 Location of Loading Areas

All required loading berths shall be off-street and shall be located on the same lot as the building or use to be served except as otherwise provided. Loading areas shall not occupy the required front yard space and shall not be located in such a manner that parking spaces cannot be accessed when the loading space is occupied.

9.1.5.2 Access to Loading Areas

Each required area shall be located with appropriate means of vehicular access to a street or public alley in a manner which will minimize interference with street and on-site traffic.

9.1.5.3 Size of Loading Areas

A required loading area shall be not less than twelve (12) feet in width, forty (40) feet in length, and fourteen (14) feet in height, exclusive of aisles, accessways or maneuvering space.

9.1.5.4 Surface of Loading Areas

All loading areas shall be constructed of asphalt, concrete, or other dust free surface as approved by the Building Inspector.

9.1.5.5 Alternate Use of Loading Areas Prohibited

Any place allocated as a loading area or maneuvering space so as to comply with this Article

shall be clean and free of obstructions at all times and shall not be used for the placement of accessory buildings, the storage of goods or inoperable vehicles, or be included as part of the space requirements necessary to meet the off-street parking area requirements.

9.1.5.6 Required Loading Areas

- 1. Any commercial or industrial building, structure, or use requiring on a regular basis the receipt or distribution of materials or merchandise by trucks or similar vehicles shall be provided off-street loading area(s).
- 2. For those buildings, structures, or uses of ten thousand (10,000) square feet or less, the required loading area may be shared with an adjacent establishment, if the parking areas are also shared, or may be waived, subject to the approval of the Planning or Zoning Commission.
- 3. The loading and unloading area must be of sufficient size and number to accommodate the numbers and types of vehicles that are likely to use this area, given the nature of the subject development. The Planning or Zoning Commission may require more or less area to safely accommodate all types of vehicular traffic using the following guidelines:

Gross Leasab	ole Area		Number of Spaces
1,000	-	19,999	1
20,000	-	79,999	2
80,000	-	127,999	3
128,000	-	191,000	4
192,000	-	255,999	5
256,000	-	319,999	6
320,000	-	391,999	7

Plus one (1) space for each additional 72,000 square feet or fraction thereof.

- 4. All loading areas shall be such that all maneuvering takes place within the property line of the premises and does not hinder the free movement of the public over sidewalks, streets, roads, highways or public servitudes.
- 5. Screening of off-street loading spaces from adjacent street Rights-of-Way and residentially zoned property is required and shall be the same as screening for parking lots.

9.1.6 Parking Plan Approval

Plans, fully dimensioned and drawn-to-scale for all parking, loading, and unloading facilities shall be submitted to the City Planner for review and compliance with these requirements as part of the building permit review process or as otherwise required by the building code and these regulations. Such plans will clearly show all necessary information sufficient to demonstrate compliance with these regulations including the location and dimensions of all required parking spaces, loading spaces, and driveways accessing such spaces. In addition, the total square footage of vehicular use areas, which includes parking and loading spaces and driveways accessing such spaces shall be noted on the plan. All landscaped areas shall also be dimensioned and shown on the required parking plan. The Planning Commission and Zoning Board may require changes in the design of the proposed parking plan as are necessary to meet all requirements and to assure the protection of public health, safety, and welfare by providing safe and adequate parking. Any amendment to a parking plan approved in conjunction with the issuance of a permit shall be submitted in writing and approved by the City Planner prior to implementing such amendments.

9.1.7 Inspection of Parking Prior to Construction

An inspection of areas to be paved for parking and loading shall be requested a minimum of 24 hours prior to paving. Approval of the Building Inspector shall precede paving of such areas.

9.2 LANDSCAPE REGULATIONS

9.2.1 Short Title

Section 9.2 shall be known as the Landscape of the City of Mandeville.

9.2.2 Purpose and Intent

It is the intent of the City of Mandeville to promote the health, safety and welfare of existing and future residents by establishing minimum standards for the protection of natural plant communities, natural features and the installation and continued maintenance of landscaping within the City of Mandeville for the following purposes:

1. <u>Water Conservation</u> - To promote the conservation of potable and non-potable water by encouraging the preservation of existing plant communities, encouraging the planting of natural or uncultivated areas, encouraging the use of site specific plant materials, providing for natural water recharge, preventing excess off-site runoff, mitigating flood impacts down stream and in down pipe, establishing techniques for the installation and maintenance of landscape materials and irrigation systems.

- 2. <u>Aesthetics</u> To improve the aesthetic appearance of commercial, industrial, and residential areas through landscape design which incorporates living plant materials, appropriate non-living landscape materials and other site elements in open space development in ways that harmonize and enhance the natural and built environment in a way that is conducive to economic development.
- 3. Environmental Quality To improve environmental quality by recognizing the numerous beneficial effects of landscaping upon the environment including (1) improving air and water quality through such natural processes as photosynthesis, mineral uptake and chemical conversions that will promote oxygen production, carbon dioxide reduction and greenhouse effect mitigation; (2) maintaining permeable land areas essential to surface water management, aquifer recharge and the conservation of fresh water resources; (3) reducing and reversing air, noise, heat, and chemical pollution through the biological filtering capacities of trees and other vegetation; (4) promoting energy conservation through the creation of shade, reducing heat gain in or on buildings or paved areas; (5) providing habitat for urban wildlife, (6) reducing the temperature of the microclimate through the process of evaportranspiration; and (7) encouraging the conservation of topsoil resources through the use of site specific plants and various planting and maintenance techniques to prevent erosion and farm lands wastage.
- 4. <u>Land Values</u> To maintain and increase the value of land by requiring a minimum amount of landscaping to be incorporated into development, thus becoming by itself a valuable capital asset.
- 5. <u>Human Values</u> To provide direct and important physical benefits to human beings through the use of landscaping to reduce noise and glare, and to break up the monotony and soften the harsher aspects of urban development. To provide a sense of the countryside and nature in the City thereby promoting a psychological sense of place for citizens and visitors alike.
- 6. <u>Preservation and Addition of Vegetation</u> To preserve existing natural vegetation and encourage the incorporation of plant materials, especially native plants, plant communities and ecosystems into landscape design, where possible.
- 7. <u>Improved Community Design</u> To promote innovative and cost-conscious approaches to the design, installation and maintenance of landscaping.
- 9.2.3 Definitions of Landscape Regulation Terminology
- 9.2.3.1 Accessways

For purposes of landscape regulations, an accessway is an all weather surface for vehicular access to parking and loading spaces traversing a greenbelt area on a development site.

9.2.3.2 Buffer Zone

A landscaped area between any building or vehicular use area and providing a visual screen of vegetation only or vegetation in conjunction with a non-living screening material for the purpose of providing a buffer between non-compatible land uses.

9.2.3.3 Drip Line

The outer edge of the foliage of a tree extending in all directions parallel to the ground.

9.2.3.4 Ground Cover, Decorative

Any mulch material (vegetative or mineral) that is used to cover the surface of the ground to prevent erosion or retain moisture.

9.2.3.5 Ground Cover, Vegetative

Plant material which reaches a maximum height of not more than twelve (12) inches at maturity, including turf.

9.2.3.6 Interior Landscape Area

Any landscaped area within the interior of a development site and beyond the required periphery landscape area that is planted with trees, shrubs and ground covering material to provide for infiltration of runoff, shade of parking areas or aesthetic enhancement of the site.

9.2.3.7 Landscaping Material

Material such as, but not limited to, living trees, shrubs, vines, turf, ground cover, landscape water features, and non-living, durable materials commonly used in landscaping including, but not limited to, rocks, pebbles, sands, decorative walls and fences, brick pavers, and earthen mounds, but excluding paving for vehicular use. Any chain link fence and opaque screening fences greater than four (4) feet in height shall not be considered as decorative walls and fences.

9.2.3.8 Mulch

Any material that is used to cover the ground surface to prevent erosion, retain moisture and protect plant material.

9.2.3.9 Periphery Landscape Area

An area of land between the property line adjacent to any street right-of-way and any vehicular use areas or building that is intended for the placement or preservation of landscape materials, which may also be referred to as the "greenbelt area".

9.2.3.10 Plant Material

Any plant including trees, vines, shrubs, ground covers and annuals or vegetation of any size, species or description.

9.2.3.11 Planting Area

An area suitable for the installation and maintenance of plant materials. Although small planting areas may be provided, any required planting areas shall be a minimum of one hundred (100) square feet with no side less than five (5) feet. Planting areas surrounding existing trees proposed or required to be preserved shall be a minimum of two-thirds (2/3) of the tree canopy.

9.2.3.12 Tree Canopy

The area within the circumference of the drip line of a tree. For purposes of these regulations, the average canopy of a mature Class A tree, except for live oaks and pines, shall be 700 square feet and the average canopy of a mature Class B tree shall be 125 square feet, the average canopy of a live oak shall be calculated as 1,500 square feet and the average canopy of a pine shall be 200 square feet.

9.2.3.13 Tree - Class A

Any self-supporting woody plant of a species which normally grows to an overall height of approximately fifty (50) feet, usually with one main stem or trunk although some species may have multiple trunks, and with many branches. A list of species considered to be Class A trees may be obtained in the office of the Department of Planning and Development.

9.2.3.14 Tree - Class B

Any self-supporting woody plant of a species which normally grows to an overall height of approximately twenty-five (25) feet, with one or more main stem(s) or trunk(s) and many branches. A list of species considered to be Class B trees may be obtained in the office of the Department of Planning and Development.

9.2.3.15 Under Story Plants

All shrubs and trees twenty-five (25) feet or less in height suitable for growth under the canopy of Class A trees.

9.2.3.16 Vegetation Protection Zone

The area within the drip line of a tree required by this to be preserved.

9.2.3.17 Visual Screen

An opaque barrier of living or non-living landscape material put in place for the purpose of separating and obscuring from view those areas so screened.

9.2.4 Provisions for Landscaping on Public Property

9.2.4.1 Pruning and Removal of Public Trees

It shall be unlawful for any person to prune, destroy, deface or cut down any public trees located in public rights-of-way or on other publicly owned land within the corporate limits of the City except in accordance with the Guidelines for the Pruning and Removal of Public Trees and Shrubs provided herein and the policies and standards of the Community Tree Plan adopted by the CAC and the Planning Commission and with the express written approval of the Landscape Inspector.

9.2.4.2 Guidelines for the Pruning or Removal of Public Trees and Shrubs

- 1. Whenever a person or a utility provider shall present a request in writing to the City for the removal or pruning of primary or secondary limbs on a public tree over 1" dbh, or removal or pruning of any tree, limb or shrub planted by the City or in conjunction with a City project on public property, the Landscape Inspector shall investigate such request. If it is found as a fact by the Landscape Inspector that such trees, limbs or shrubs interfere with the safe and proper maintenance of public utilities or substantially interfere with the lawful use of private property so as to cause loss, damage or deprivation of the lawful use of such property to the owner or tenant thereof, then the City may permit the cutting, pruning or removal of such tree(s), limb(s) or shrub(s). The following conditions must be present but may not be sufficient grounds supporting approval of a request for tree or limb removal, upon investigation and findings by the Landscape Inspector:
 - a. the tree or limb is a hazard to traffic, public utilities, buildings or structures; or,

- b. the tree or limb, as determined by a licensed arborist in writing, is injured, diseased or insect-infested such that it is a hazard to people, structures or other trees; or,
- c. the tree or limb prevents any and all access to a lot or parcel; or,
- d. the proposed pruning is for the nurturing and health of the tree; or
- e. the tree will be properly transplanted to another public location in the City by the applicant with the consent of the CAC; and,
- f. any tree removed will be replaced with an equivalent shrub or tree which is:
 - i. donated to the public, and
 - ii. planted by the applicant on public property in conformance with the provisions of Section 9.2.4.6; and
 - iii. guaranteed by the applicant for one (1) year after planting.
- 2. The following shall require disapproval of an application:
 - a. the species, size, historical importance and/or condition of the tree make it a unique or rare specimen; or
 - b. the size or location of the public tree, limb or shrub make it easily accessible to public view and none of the reasons for removal listed above are present.
- 3. Any person requesting the cutting, pruning or removal of a tree or shrub must first obtain written authorization from the Landscape Inspector, except for emergency situations as defined herein. All such requests for cutting, pruning or removal of a public tree or limb shall be accomplished by or under the supervision of a qualified arborist, as per the following provisions:
 - a. Such requests shall give the City at least five (5) working days advance notice of the date and time contemplated for such work. No such cutting, pruning or removal of public trees or limbs shall be done without prior specific written permission of the City by the Landscape Inspector. Such notification and permission is to be returned to the applicant following inspection by agents of the City.
 - b. All work shall be accomplished by qualified persons licensed under applicable state law. In no instance shall trees, limbs or shrubs be cut, pruned or removed

contrary to the expressed stipulations of the City's permit nor shall work commence prior to the receipt of the permit by the applicant or his agent, even should the aforementioned five (5) working day period expire. The City further reserves the right and authority to inspect the work in progress and demand said work comply with standards for arboricultural work as well as all applicable regulations and policy decisions of the City regulating such work. All qualified persons, firms or corporations engaged to cut, prune or remove public trees, limbs or shrubs shall be lawfully licensed and bonded under applicable state law under the jurisdiction of the Louisiana Horticulture Commission. Furthermore, all licensed arborists cutting, pruning or removing public trees or limbs shall furnish proof of a valid occupational license as issued by the City in addition to compliance with provisions of the state law regulating such arboricultural work.

- c. No person, firm or corporation shall be permitted to personally cut, trim or remove a public tree, limb or shrub adjoining, adjacent to or abutting their real property without such license or bond unless specifically authorized by the Landscape Inspector in the permit.
- d. Any applicant who is aggrieved by the refusal of the Landscape Inspector to issue a permit to cut, prune or remove a public tree, limb or shrub may appeal within thirty (30) days from such decision to the Community Appearance Commission. The Community Appearance Commission shall review the decision of the Landscape Inspector and either approve, disapprove or modify it.
- e. In emergencies (i.e., hurricanes, thunderstorms, ice storms or tornadoes) presenting apparent imminent threat to person or property, any person may cause removal of, or major maintenance to, a tree, limb or shrub provided that this action is reasonably calculated to dissipate the threat. Within five (5) days of such action, said person shall file a request for an after-the-fact approval by the Landscape Inspector. In such emergency situations, work on trees, limbs or shrubs endangering electric lines shall not require prior approval.
- f. Major Public Works projects and/or contracts which are subject to State Licensing Board for Contractors procedures (generally projects of \$50,000 or more) shall be reviewed by the CAC prior to commencing work to ensure that said work minimally impacts public tree or limbs and adjacent privately owned trees and conforms to the provisions of Section 9.2.4 at seq. of this article and the Community Tree Plan. Daily maintenance or minor projects performed by City maintenance or Public Works crews that may affect major public tree or limbs shall be reviewed by the Landscape Inspector to ensure that said work minimally impacts public tree or limbs and adjacent privately owned trees and conforms to the provisions of Section 9.2.4 at seq. of this article and the Community Tree Plan.

In emergency situations, prior approval is not required though every effort will be made to minimize the impact to adjacent trees.

9.2.4.3 Signs on Public Trees Prohibited

It shall be unlawful for any person to nail, tack or otherwise affix signs or advertisements on any of the trees located on the public rights-of-way or other public places within the corporate limits of the City of Mandeville. Any violation of the provisions of this Section is hereby declared to constitute a public nuisance and such nuisance may be abated by the City in accordance with the procedures set forth in Section 1.9 of the City's Code of Ordinances. The abatement of any nuisance under the provisions of this Article shall in no way alter or affect the institution of any proceedings available to the City under this or other City codes and regulations.

9.2.4.4 Protection of Public Trees

- 1. On public rights-of-way and all public property no person shall:
 - a. Cut, disturb or interfere in any way with the roots of any public tree or limb or adjacently privately owned tree, except in the area of approved accessways;
 - b. Spray with chemicals, insecticides or other oils or whitewash or allow any gaseous liquid or solid substance harmful to tree or limbs to come in contact with any public tree or its root system;
 - c. Place any wire, rope, sign, poster, barricade or other fixture on a public tree or limb or tree guard;
 - d. Injure, misuse or remove any device placed to protect any public tree; or
 - e. Deposit or place any material which may impede the free passage of water, air, or fertilizer to the roots of any public tree, except by written approval of the CAC.
- 2. No person shall use the neutral grounds, parks, sidewalks or public places to dump grass clippings, tree or limb trimmings, rocks or refuse of any nature.
- 3. Unless approved for removal by the CAC or Landscape Inspector, all public trees near any excavation or construction activity shall be protected with a vegetation protection zone barrier prior to commencement of the excavation or construction activity. The barrier shall be constructed to protect the area within the dripline. All construction materials, equipment, dirt of other debris shall be kept outside the protective tree barrier.

9.2.4.5 Planting on Beach, Near or on Seawall

It shall be unlawful for any person to plant any trees, flowers, hedges, turf or plants on the beach or neutral ground between Lakeshore Drive and the seawall, or on or attached to the seawall itself,

without first having obtained permission therefor from the CAC, acting in accordance with the Master Tree Plan.

9.2.4.6 Plantings on Public Property

No shade or ornamental trees and shrubs shall be planted on any public rights-of-ways, or other public place of the City by any person without the approval of the Community Appearance Commission, in accordance with the Master Tree Plan, and the written permission of the Landscape Inspector. Any tree, plant or shrub planted on public property by any person shall become the property of the City and the City may order such tree, shrub or other plant to be removed at the expense of the person who planted such tree or shrub. Any unauthorized private

plantings on public property are subject to removal, relocation or pruning by the City if they interfere with utility maintenance.

9.2.4.7 Planting in Utility Servitudes

No planting shall be installed within a utility servitude without prior written approval of the affected utility. Approval may be withheld by any utility upon its determination that the proposed type or location of vegetation would adversely affect the operation, maintenance or function of the utility. Approval of the installation of plant materials by a utility shall create no obligation to replace vegetation removed or damaged by the utility in the course of its lawful use of the servitude.

9.2.5 Provisions for Landscaping on Private Property

9.2.5.1 Obstructions to Vision Prohibited

No vegetation shall be allowed to obstruct vision at vehicle intersections in accordance with the requirements of Section 8.1 Supplemental Regulations and Section 9.1.2 Construction Design Standards for Parking and Loading.

9.5.5.2 Vegetation Protection Zones

An area extending at least fifteen (15) feet in all directions from the trunk of any tree required or proposed to be preserved to meet the requirements of this or encompassing a minimum of two-thirds (2/3) of the entire canopy area of the tree, whichever is greater, shall

be required to be maintained undisturbed under the provisions of this Article. This area is defined as the Vegetation Protection Zone. Exception: The Vegetation Protection Zone for live oaks will be a circle with a radius which is eighty-two (82) percent of the canopy of the tree, measured from the trunk to the drip line. A barrier shall be erected and maintained around this area at all times during construction. No soil deposits, construction materials, equipment, or other materials shall be temporarily or permanently stored in locations within or immediately adjacent to the Vegetation Protection Zone which would cause suffocation of root systems of trees required or proposed to be preserved. No paving with concrete, asphalt, or other impervious material shall be allowed within the Vegetation Protection Zone. No structure shall be placed or constructed at any time within the Vegetation Protection Zone.

9.2.5.3 Vegetation Protection Zone Barriers

- 1. The Vegetation Protection Zone barrier shall be continuous and at least two (2) feet above the ground. The material used to construct the barrier can be either rigid and semi-permanent (such as lumber) or orange "safety mesh" and must be specified on the landscape plan.
- 2. The required tree barriers shall be properly installed and verification of such installation shall be made by the landscape inspector prior to the issuance of a development or clearing permit.

9.2.5.4 Landscape Requirements in Low-Density Residential Districts

In the R-1, R-1X and R-2 districts, a minimum of 50 percent of all existing trees larger than six (6) inches dbh in the required yard setback areas shall be required to be preserved. The landscape inspector shall verify the preservation of all required trees before a Certificate of Occupancy will be issued for the structure. Trees required to be preserved shall be shown on the residential site plan approved in conjunction with the development permit. In addition, the provisions of the Live Oak Protection Section shall also apply in R-1, R-1X and R-2.

9.2.5.5 Landscape Requirements in Districts Other than Low-Density Residential

The requirements of this Article shall apply to all zoning districts other than R-1, R-1X and R-2 residential districts, with the exception of the Live Oak Protection requirements, which apply in all zoning districts. In all zoning districts other than R-1, R-1X and R-2, development sites shall be required to meet the minimum requirements as specified by this Article for Landscaping within the periphery landscape areas, interior planting areas and buffer areas. All required plant materials shall be installed or preserved in accordance with this Article and the landscape inspector shall inspect the required landscaping to verify adherence to

code and the landscape plan approved in conjunction with the permit prior to the issuance of a Certificate of Occupancy.

9.2.5.5.1 Periphery Landscape (Greenbelt) Requirements

- 1. Required Area of Greenbelt In all zoning districts other than the R-1, R-1X and R-2 districts, a periphery landscape area, also known as the greenbelt area, shall be required to be located adjacent to the property line of the right-of-way of any public street, road, lane, or other public accessway (excluding an alley) upon which the site fronts. In calculating the required greenbelt area the area of any utility servitude, either existing or proposed as part of the development permit, shall not be included as a part of the greenbelt. The required area of the greenbelt shall be calculated as an area fifteen (15) feet in depth measured at right angles from the property line edge of the street right-of-way or from the interior edge of any utility servitude which is adjacent to and parallel with the street right-of-way less the maximum allowable accessways through the greenbelt. Except in accessways and as prohibited by the utility provider's use of the utility servitude. the servitude shall also be landscaped minimally with a vegetative or decorative ground cover. On corner or through lots with more than one street frontage, the greenbelt shall be required adjacent to each street frontage. The periphery area shall contain trees and vegetative or decorative ground covering material, as specified herein.
- 2. Flexibility of Greenbelt Depth The required depth of the greenbelt may be articulated to provide for a depth of greater than or less than the minimum fifteen (15) foot depth so long as the required area of greenbelt on that street frontage is maintained. The depth may be reduced for a portion of the length of the greenbelt to a minimum of ten (10) feet provided that a depth greater than fifteen (15) feet is added to other areas of the greenbelt to maintain the overall required greenbelt area. In addition, the depth of the greenbelt may be reduced to five (5) feet for up to a maximum of twenty (20) percent of the length of the greenbelt so long as the overall required area of the greenbelt on that street frontage is maintained.
- 3. Reduction in Greenbelt Area When a utility servitude which occurs between the street right-of-way and the required greenbelt exceeds ten (10) feet in depth measured from the street right-of-way, the required depth of the adjacent greenbelt may be reduced by one (1) foot for every additional five (5) feet of servitude in excess of ten (10) feet.
- 4. Access Through Greenbelts For street frontage up to one hundred fifty (150) linear feet, no more than two (2) one-way accessways a maximum of 17.5 feet in width or one (1) two-way accessway a maximum of 35 feet in width shall be permitted through the greenbelt. For more than one hundred fifty (150) feet of

street frontage, one (1) additional two-way accessway or two (2) additional one-way accessways of the maximum width specified may be permitted for each additional one hundred fifty (150) feet of frontage or major fraction (seventy-six [76] feet or greater) thereof.

- 5. Preservation of Trees in Greenbelts Except in accessways as described above, all trees which have a trunk six (6) inches dbh and which are in the area of a required greenbelt shall be preserved or replaced if diseased or dead. In addition, if the number of trees six (6) inches or more dbh which are in the front periphery do not equal the required number of Class A and Class B trees (one (1) per twenty-five (25) linear feet), then Class A and Class B trees must be planted to the extent necessary to comply with the requirements of this Article.
- 6. <u>Planting in Greenbelts</u> Each required greenbelt shall contain a minimum of one (1) Class A tree (see definitions) and one (1) understory Class B tree for every twenty-five (25) linear feet of lot frontage or fraction thereof. In addition a ground covering material shall be established in the required greenbelt area. Vegetative ground covering material may include turf or other material that forms a consistent vegetative cover. Ground covering material may include pine straw or other mulches, including those of mineral composition.
- 7. Applicability of Greenbelt Requirements The periphery landscape requirements shall apply as a condition for the issuance of all new construction building permits in all zoning districts except the R-1, R-1X, and R-2 low density residential districts. These provisions also apply for existing structures or uses when there is a change in use classification which requires an increase in the number of off-street parking spaces from the number of such spaces required in connection with the preceding use of the development site, or when a new building permit is required for new or additional construction on the development site.

9.2.5.5.2 Site Interior Planting Regulations

Site interior planting is required in order to provide for groundwater recharge, to mitigate the effects of storm water runoff over impervious surfaces in on-site vehicular use areas, to provide shade and reduce heat and glare reflected from paved areas, to purify the air in intensely developed areas, and to screen visibility of vehicular use areas from adjacent street corridors.

1. <u>Screening of Vehicular Use Areas</u> - When a vehicular use area is visible from a public street right-of-way, the vehicular use area shall be screened from view from the adjacent street with an opaque vegetative screen as part of the interior planting requirements. The screen shall be of living material that is opaque from ground height to a height of three feet, with intermittent visual obstruction from

above the opaque portion to a height of at least twenty (20) feet. This screen shall be planted in a prepared planting area no less than twenty-four inches wide immediately adjacent to the vehicular use area or may be located within the required greenbelt area. This requirement applies to all street frontages of lots if the vehicular use area is visible from the adjacent street.

- 2. Site interior landscaped area shall be provided in the interior of vehicular use areas larger than eight (8) parking spaces or 3,000 square feet. The total of all interior landscaped areas shall occupy a minimum of eight (8) percent of the vehicular use areas, including associated service drives and loading areas.
- 3. Although smaller areas may be provided, interior planting areas shall be a minimum of 100 square feet in size with a minimum side dimension of five (5) feet to count towards the eight (8) percent total. The minimum planting area for an interior tree is twenty-five (25) square feet per tree.
- 4. The interior landscaped areas shall be raised and curbed with permanently anchored material at least six (6) inches in height. Curb material may be concrete, natural stone, asphalt, railroad ties or landscape timbers.
- 5. A required site interior landscaped area may be connected with a required greenbelt or buffer area so long as the area of the interior planting area is in addition to the area of the required greenbelt or buffer.
- 6. All interior landscaped areas must be planted with a vegetative ground covering material.
- 7. Interior landscaped areas must be planted with trees according to the following ratios:
 - a. A minimum of one tree per 2,000 square feet (approximately 5.5 parking spaces) of vehicular use area shall be required to be preserved or planted within a site interior landscaped area provided the distance and shading requirements specified below are complied with by the selection of tree species.
 - b. The required trees shall be evenly distributed throughout the vehicular use area to maximize infiltration of stormwater and the beneficial effects of the shade provided. No parking space shall be located more than forty (40) feet from any tree within the periphery greenbelt or the interior landscape areas.
 - c. A minimum of 50 percent of the trees provided within the interior planting areas shall be "Class A" trees.

d. A minimum of thirty (30) percent of the vehicular use area must be shaded by tree canopies of trees located within the required interior landscape areas.

9.2.5.5.3 Buffer Zone Requirements

1. Requirements Within Required Buffer Zones

- a. The buffer zone shall contain one (1) Class A tree for each twenty-five (25) linear feet and one (1) Class B tree for each ten (10) linear feet of buffer zone.
- b. The buffer zone shall contain a minimum six (6) foot high visual screen (see Section 9.2.3.17) of living and/or non-living landscape material. If only living material is used, plantings shall be of a form, size, and type which will provide a seventy (70%) percent or more opaque screen within no longer than twelve (12) months of the date planted. Plantings shall be a minimum of four (4) feet in height from the ground immediately after planting.
- c. If a six (6) foot high non-living opaque screen is used it shall be placed a minimum of two (2) feet from the property line and a minimum of one (1) shrub or vine for each ten (10) linear feet of screen shall be planted abutting the non-living screen on the side adjacent to the more restrictive zoning district. These need not be evenly spaced at ten (10) feet apart but may be grouped. They shall be planted along the outside of the non-living barrier unless they are of sufficient height at the time of planting to be readily visible over the top of such barrier. The remainder of the buffer zone shall be landscaped with turf and vegetative or decorative ground covering materials.
- d. No vehicular parking, utility servitude or structure of any kind shall be allowed in the required buffer zone.
- e. The buffer zone shall consist of an area not less than the required depth measured at right angles to the property line(s) along the entire length of and contiguous to the property line adjacent to the more restrictive zoning district.
- f. The required depth of a buffer zone on a development site 200 feet in depth or greater, measured at right angles from the property line along which the buffer is required to be located, shall be expanded by an additional one (1) foot for each additional twenty (20) feet of site depth up to a maximum additional buffer of ten (10) feet.
- g. The landscape buffer zone shall be required to be provided in conjunction with the issuance of a building permit for new construction on a development site or

when a change in use classification from the preceding use of the site renders the development site subject to the provisions of the buffer zone requirements of this Section.

2. When Buffer Zones are Required

When a use occurs in an R-3, I, O, B-1, B-2, B-3, B-4, O/R, PM-1, PM-2, PD, M-1, or M-2 zoning district which is not allowed in an adjacent and more restrictive zoning district, the less restrictive use shall provide a landscape buffer zone to mitigate the impact of the less restrictive use on the adjacent property. The width of the required buffer zone shall be as follows:

- a. <u>Buffer Requirement for Multi-Family Adjacent to Low-Density Residential</u> Residential Use Classifications 6.2.5 through 6.2.10 adjacent to R-1, R-1X, or R-2 zoning districts shall provide a buffer zone not less than ten (10) feet in depth.
- b. <u>Buffer Requirement for Commercial Adjacent to Residential</u> All 6.3 Civic, 6.4 Commercial and 6.7 Combined Use Classifications adjacent to land in R-1, R-1X, R-2, R-3, or MH zoning districts, or O/R, PM-1, PM-2 and PD zoning districts with existing or proposed residential uses or with vacant parcels adjacent to the development site shall provide a buffer zone not less than fifteen (15) feet in depth.
- c. <u>Buffer Requirement for Industrial Adjacent to Residential</u> All 6.5 Industrial and 6.6 Agricultural Use Classifications adjacent to land in R-1, R-1X, R-2, R-3 or MH zoning districts, or O/R, PM-1, PM-2 and PD zoning districts with existing or proposed residential uses or with vacant parcels adjacent to the development site shall provide a buffer zone not less than twenty (20) feet in depth.
- d. <u>Buffer Requirement for Industrial Adjacent to Commercial</u> All 6.5 Industrial and 6.6 Agricultural Use Classifications adjacent to land in B-1, B-2, B-3, B-4, I and 0 zoning districts or O/R, PM-1, PM-2 and PD zoning districts with existing or proposed uses other than residential adjacent to the development site shall provide a buffer zone not less than ten (10) feet in depth.

9.2.5.6 Landscape Maintenance and Replacement Provisions

1. <u>Maintenance of Vegetation</u> - All trees, required in R-1 and all trees, vegetation, and screening required for other than R-1, R-1X, and R-2 residential developments or buffer zones, periphery and interior landscaped areas shall be preserved and maintained. Any cutting, clearing, removal or land filling which does not preserve and maintain the required trees and vegetation is prohibited. Any required trees

cut to meet flooding or drainage requirements shall be replaced to meet the provisions of Item 2 of this subsection.

- a. The owner shall be responsible for the maintenance of required landscaping in good condition so as to present a healthy, neat and orderly appearance; and said landscaped areas shall be kept free from refuse and debris.
- b. Plant beds shall be mulched to prevent weed growth and maintain soil moisture.
- c. Plant materials shall be pruned as required to maintain good health and character.
- d. Turf areas shall be mowed periodically.
- e. All roadways, curbs and sidewalks shall be edged when necessary in order to prevent encroachment from the adjacent grassed areas.

f. Watering

- i. <u>Operation of Automatic Irrigation Systems</u> Whenever possible, automatic irrigation systems should be operated between the hours of midnight and 6 A.M. Irrigating during these hours reduces fungus growth and loss of water due to evaporation.
- ii. <u>Maintenance of Irrigation Systems</u> Irrigation systems shall be constantly maintained to eliminate waste of water due to seasonal change, loss of heads, broken pipes or mis-adjusted nozzles.
- iii. Manual Watering When no irrigation system is provided, the owner of the property shall be responsible for the provision of adequate water to required plant materials.
- g. Fertilizers shall be used as necessary to maintain good plant health.
- h. <u>Maintenance of Natural Plant Communities</u> Natural plant communities left intact on all site developments shall be maintained to promote good ecology.
- 2. Replacement of Vegetation Should any required tree, shrub or other landscape vegetation die or be removed, or a non-living screen need replacement, the tree, shrub, vegetation, or screen shall be replaced by a similar tree, type of vegetation, or screen meeting the requirements of this Article. Class A trees less than six (6)

inches in diameter shall be required to be replaced with one (1) two (2) inch dbh Class A tree, a minimum of ten (10) feet in height per Class A tree removed. Class A trees six (6) inches dbh or greater which are required to be replaced shall be replaced with a two (2) inch dbh replacement tree a minimum of ten (10) feet in height for each six (6) inches dbh of tree removed. Replacement vegetation shall be required to be installed within twenty (20) days of written notice by the landscape inspector.

9.2.5.7 Live Oak Protection Requirements

In all zoning districts, including the R-1, R-1X and R-2 districts, all live oak trees 6" dbh shall be protected as follows:

- 1. A tree removal permit shall be obtained from the Building Inspector prior to cutting, clearing or removing any live oak tree.
- 2. The applicant wishing to remove a live oak tree must state in writing that such activity will enhance the health, safety and welfare of the public, or otherwise benefit the public interest and the applicant must offer evidence to that effect. The Building Inspector is empowered to issue or deny the permit based on the application and the evidence. Prior to the issuance of a tree removal permit the applicant must submit a plan or written statement offering evidence of compliance with the tree replacement provisions of this Article.
- 3. It shall be unlawful for any person to place soil in such a way that would cause live oaks to become diseased or die. If filling with soil is necessary to properly drain the land, all efforts should be made to protect the area within the drip line of a live oak from the impact of such activity. Should all efforts fail and a tree removal permit be issued for the removal of the live oak the provisions of these regulations regarding replacement of trees shall be required to be met.
- 4. A tree removal permit will be required to prune the primary and secondary branches of any live oak tree 12" dbh or greater. Such pruning shall be required to be recommended in writing and supervised by a licensed arborist or a state forester.

9.2.5.8 Screening of Trash and Garbage Cans

In all zoning districts, storage areas related to uses other than those allowed in the R-1 and R-2 districts, and containing three (3) or more refuse, garbage, or rubbish containers or one (1) or more dumpsters, shall be screened with a minimum six (6) foot high screen of living or non-living landscape material, or of a height sufficient to screen the objects from view.

Living vegetation shall be evergreen, a minimum of four (4) feet at planting and 70% opaque within twelve (12) months.

9.2.5.9 Supplemental Planting and Plan Preparation Requirements

1. <u>Minimum Planting Requirements</u> - The following standards shall be considered the minimum requirements for the installation of all required landscaping within the City.

Tree planting methods and overall care and maintenance shall conform to the standards contained in the Forestry Handbook, 2nd Edition, Karl Werger, Editor for the Society of American Foresters Sections 8 and 16, 1984, John Wiley & Sons, N.Y., N.Y.

- a. <u>Installation Standards</u> All landscaping shall be installed in a sound workmanlike manner by a person knowledgeable of proper horticultural practices (such as a licensed landscape contractor, horticulturalist or other persons having similar training and experience), and according to accepted and proper planting procedures with the quality of plant materials as hereinafter described.
- b. <u>Plant Quality Standards</u> Plants installed pursuant to this shall conform to or exceed the minimum standards as promulgated by the American Association of Nurserymen or the Louisiana Horticulture Commission.
- c. <u>Plant Ball Sizes</u> Ball sizes on all transplanted plant materials shall conform to or exceed the minimum standards as noted in the most current edition of "Grades and Standards for Nursery Plants," prepared by the American Association of Nurserymen.
- d. <u>Use of Planting Soil</u> All required landscape materials shall be installed using planting soil of a type appropriate to the individual plant material and the soil conditions in which the planting is occurring.
- e. <u>Use of Organic Mulches</u>: The use of organic mulches reduces the growth of weeds and adds nutrients to the soil as well as retains moisture over the root zones of plant materials.
- I. <u>Application specifications</u>: When appropriate, a minimum of 2" (3" preferred), of organic mulch shall be placed over all newly installed tree, shrub and ground cover planting areas.

- ii. <u>Types of Mulch</u>: The use of pine bark, rather than cypress (or other valuable species) mulch is encouraged. Pine straw from tree farming operations is acceptable.
- f. <u>Size Standards</u> All trees required to be planted shall be a minimum of ten (10) feet in height and two inches dbh immediately after planting.
- g. <u>Proximity to Accessways</u> Landscape material, except turf or ground cover, shall not be located closer than three (3) feet from the edge of any accessway pavement.

2. Plan Preparation Requirements

- a. Prior to the issuance of a landscape permit or a development permit for a site on which landscaping or tree preservation is required, a landscape plan shall be required to be submitted to, reviewed by, and approved by the Landscape Inspector. The landscape plan shall be accompanied by the required permit fee in accordance with the provisions in Article 5. If, in addition to the preservation of trees, plant materials are required to be installed the landscape plan shall include or be accompanied by a detailed planting plan.
- i. <u>Plans for Single Family or Duplex Residence</u> The tree plan submitted for an individual single family or duplex residence on its own lot may be a plot plan or drawing prepared by the owner or his agent.
- ii. <u>Plans for all Other Development</u> The landscape plan and any additional material required for the development shall be prepared by and bear the seal of a registered landscape architect or otherwise be prepared by persons authorized to prepare landscape plans or drawings in Louisiana according to Revised Statutes Acts 1950, No. 224 and amended and specifically SS 3808-E-(1) & (2). Plans may be prepared by other legally qualified persons such as architects or engineers, where applicable. Owners of the property may prepare landscape plans for the property which they own.
- B. <u>Contents of Landscape Plans</u> The landscape plan shall be drawn to scale and fully dimensioned and include the following:
 - i. The location and dimensions of all structures and vehicular use areas and accessways and relative topographic elevations.
 - ii. The dimensions of all planting areas and the location, type and size of all existing vegetation required or proposed to be preserved and all new vegetation proposed to be planted.

- iii. A legend clearly describing the required landscaping in comparison to the proposed landscaping including the number and type of vegetation and square foot area of landscaping area.
- iv. A detailed planting plan, if additional planting is required or proposed, including the size and type of plant materials proposed to be planted and including specifications and cross sections describing proper planting techniques in accordance with these regulations.
- v. The name, address and telephone number of the person who drew the plan and the date, including any revision dates, that the plan was drawn.
- vi. The landscape and/or planting plan shall be sufficient to illustrate compliance with this; the Landscape Inspector may require additional information if needed to document compliance.

9.2.5.10 Emergencies

In the case of emergencies, such as wind storms, ice storms, hurricanes, general pestilence or disease, or other disasters, the requirements of this Section may be waived by the Building Inspector during the period of such emergencies so as not to hamper private or public work to restore order to the City. In the event of emergency situations such as hurricanes, thunderstorms, ice storms, tornadoes, work by public utilities on trees endangering electric lines shall not require prior approval.

9.2.5.11 List of Trees, Shrubs and Ground Covering Materials

A list of examples of the acceptable types of Class A and Class B trees and shrubs approved by the Zoning Board is available from the City of Mandeville Department of Planning and Development.

9.2.5.12 Landscape Permit Required

A landscape permit shall be required to be issued by the Building Inspector for any proposed landscape construction in all zoning districts, except R-1, R-1X, and R-2, prior to commencement of landscape construction. Requirements and fees for the issuance of landscape permits are included in the provisions of Article 5.

9.2.5.13 Clearing Permit Required

1. <u>General</u> - Unless otherwise provided in this Section, no person, corporation, association, public agency, or agent or employee thereof, shall effectively destroy

- or remove vegetation from any property within the City of Mandeville without first obtaining a
- clearing or tree and shrub removal permit from the Building Inspector.
- 2. <u>Objectives of Clearing Permit</u> The objectives of the requirement for the issuance of a clearing permit for the monitoring of land clearings are:
 - a. to limit the removal of valuable existing vegetation in advance of the planning and approval of land development plans; and
 - b. to limit the destruction of natural stormwater retention basins and water recharge zones by promoting the preservation of existing plant communities and natural areas on site.
- 3. Requirements of Issuance of Clearing Permit Prior to the cutting, clearing, or removal of any tree greater than two (2) inches dbh on any lot(s) or parcel(s) of land on which there is no existing building, or which has an existing building and additional undeveloped portions of the lot not required to be preserved as landscaped area, a clearing permit for such activity shall be obtained from the Building Inspector. No clearing permit shall be issued except in conjunction with a duly approved development plan for the site proposed for clearing and the issuance of a landscape permit and development permit or the authorization to proceed with the construction of public improvements in conjunction with the development of a subdivision. Prior to the issuance of a clearing permit the barriers for the protection of vegetation required to be preserved shall be erected and the landscape inspector shall inspect the site to determine compliance with the provisions of this Article.

9.2.5.14 Tree and Shrub Pruning or Removal Permit Required

A tree and shrub pruning or removal permit shall be obtained from the Building Inspector when cutting, clearing, or removing any required Class A or Class B tree, or for pruning live oak trees as provided herein. A tree and shrub removal permit shall only be granted on previously developed sites or for surveying, testing or removal of diseased or dead trees on undeveloped sites as provided in Article 7, in conjunction with the issuance of a new building permit or if the tree or shrub proposed to be removed would be required if a new building permit were being applied for on the site. This regulation applies to any lot or parcel on which there is a building that was constructed prior to the effective date of these regulations, and any lot for which a building permit was issued prior to the effective date of these regulations. The applicant for the permit must state in writing or submit a planting plan for the replacement of vegetation as required by the provisions of this Article.

9.2.5.15 Inspection of Landscape Installations

Inspection of required landscape installation by a qualified landscape inspector employed by the City shall be required prior to the issuance of a Clearing Permit, a Tree or Shrub Removal Permit, prior to the pouring of pavement for vehicular use areas and prior to the issuance of a Certificate of Occupancy for a structure associated with the required landscaping to be inspected.

- 1. Landscape Inspection Fee A twenty (\$20) dollar fee shall be paid for each required landscape inspection.
- 2. Landscape Reinspection Fee An additional twenty (\$20) dollar fee will be paid for reinspection of any facility required to be reinspected after failure to pass an inspection.

9.2.5.16 Violations

- 1. Each required tree, shrub, non-living screen, or other plant matter cut, cleared, removed, caused to become diseased or die, or otherwise acted upon in violation of the provision of this Article shall constitute a separate offense subject to the provisions of Section 1.9 of this. Each separate day on which a violation occurs or continues shall be considered a separate violation of this Article.
- 2. Should any tree(s), shrub(s), non-living screen(s) or other plant matter be cut, cleared, removed, caused to become diseased or die, or otherwise acted upon in violation of this section prior to the issuance of a building permit, no such permit shall be issued until the proper replacements are provided and all fines resulting from the violation are paid.
- 3. Should any tree(s), shrub(s), non-living screen(s) or other plant matter be cut, cleared, removed, caused to become diseased or die, or otherwise acted upon in violation of this section after the issuance of a building permit, the permit shall automatically be suspended until the appropriate replacements are provided and all fines resulting from the violation are paid.
- 4. Failure to maintain the required vegetation protection zone barrier during the construction process shall constitute a violation, shall automatically suspend the development permit for which the tree barrier was required to be erected and shall be subject to the maximum penalty of Section 1.9 of this.

9.3 OPTIONAL PARKING MITIGATION FEE SECTION

This Section shall be known and may be cited as the Optional Parking Mitigation Fee Section.

9.3.1 Intent and Purpose of Section

- 1. This Section is intended to be consistent with the requirements of Article 7 and Article 9 of this, to ensure that adequate off-street parking is provided to support commercial property uses within the B-3 Old Mandeville Business District; that the cost of providing such parking is borne by the owners or occupants of commercial property whose use of such property necessitates the provision of off-street parking; and that provision of such parking not lead to the damage or destruction of the character of the Old Mandeville Business District.
- 2. The purpose of this Section is the implementation of a regulatory program that requires the owners or occupants of commercial property situated within the Old Mandeville Business District to pay a parking impact fee that does not exceed the cost of acquiring and constructing public parking places of an equal number as that which are required under the provisions of Article 9 of this for the use for which the property will be used but which in the opinion of the Zoning Board cannot reasonably be provided on premise. The opinion of the Zoning Board required by this Section shall be evidenced by the grant by the Zoning Board of a variance requested by an owner or occupant of property from the provision of the otherwise required number of on-premise parking spaces.

9.3.2 Grant of Parking Variances in the B-3 Old Mandeville Business District

- 1. No building permits for property situated in the B-3 Old Mandeville Business District or, as a result of its proposed use, considered as a part of the Old Mandeville Business District, shall be issued by the Building Inspector unless the applicant for the permit can demonstrate to the satisfaction of the Building Inspector the existence of all off-street parking spaces required under the terms of this Section to support the proposed use or the applicant has received from the Zoning Board a variance excepting the applicant from the necessity of providing the number of off-street parking spaces which is deficient.
- 2. In considering any request for a variance from the otherwise required number of off-street parking spaces submitted in connection with the use or proposed use of property located in or considered a part of the Old Mandeville Business District, if the Zoning Board should determine that a substantial cause of the applicant's request for the variance is the presence on the property in question of an existing building or structure that the applicant intends to maintain in connection with his proposed use of the property, the Zoning Board shall attach to any variance that it may grant to the applicant the condition that the applicant mitigate in a manner prescribed by this Section all deficient parking spaces that are determined by the Zoning Board to be attributable in whole or in part to the existence of the buildings or structures to be maintained on the site.

- 3. The existence of structures or buildings on property situated in or considered a part of the Old Mandeville Business District which an applicant for a building permit is proposing to maintain in connection with the use of the property may constitute sufficient hardship or practical difficulty to support the grant of a variance to the provision of the otherwise required off-street parking spaces when the deficiency in such parking spaces can be attributed in whole or in part to the existence on the site of that building or structure.
- 4. Regarding of the forgoing provisions of this Section, the Zoning Board may refuse to grant a variance from the provision of required off-street parking spaces, even if it is established that the deficiency in parking spaces is attributable to buildings or structures existing on the site, for any of the reasons set forth in this Section.

9.3.3 Optional Parking Mitigation Fee; Use of Funds

- 1. Any person seeking a parking variance in connection with an existing or proposed use of property located in the Old Mandeville Business District on which there is located one or more existing building(s) or structure(s) the presence of which prevents the provision of the otherwise required number of on-premise parking spaces, shall pay to the City of Mandeville a sum of money equal to the cost to the City of providing an equal number of off-site parking spaces as cannot be located on site due to the presence of the existing building(s) or structure(s).
- 2. The cost to the City for acquiring and improving public parking on or in public rights-of-way is hereby set and established at the sum of one thousand two hundred (\$1,200.00) dollars per parking space. The cost to the City for acquiring and improving public parking on property acquired by the City for that purpose shall be set at the City's actual cost per parking space for the most recent such project completed or planned for which costs are available.
- 3. The funds realized under the provisions of this Section shall be collected by the Building Inspector at the time of issuance of a building permit for any property affected by this Section. The Building Inspector shall give due receipt for all such funds as are received by him. All such funds shall thereafter be paid over to the treasury of the City and there maintained in a special designated account. Funds deposited into or otherwise credited to that account and any interest accruing thereon shall thereafter be utilized solely for the purpose of acquiring public parking areas in the Old Mandeville Business District, whether by lease or ownership or other real right; of improving or constructing on-street or off-street public parking spaces in the Old Mandeville Business District; or of maintaining any parking spaces thus acquired, constructed or improved.

- 4. Alternatively, but only in accordance with all applicable rules and regulations of the City and with the advance written consent of the City's Director of Public Works, the Zoning Board may instead condition any parking variance that would otherwise be subject to the provisions of paragraph 1 of this Section upon the applicant's construction, at his cost, of a like number of public on-street parking spaces in such portion or portions of the Old Mandeville Business District as may be proposed by the applicant.
- 5. In the further alternative, the Zoning Board may instead condition any parking variance that would otherwise be subject to the provisions of paragraph 1 of this Section upon the dedication by the applicant to the City, in full ownership, of a parcel of land of sufficient size to accommodate the number of parking spaces for which the variance is granted, provided that all of the following conditions are satisfied:
 - a. the offer to dedicate the property has first been proposed to the Zoning Board by the applicant;
 - b. the parcel of land in question is dedicated to the City in a form acceptable to the City Attorney and in such a manner as to convey to the City full, unencumbered and merchantable title to the property in question;
 - c. the parcel to be dedicated is located in the Old Mandeville Business District;
 - d. the parcel of land to be dedicated is located in a zoning district that includes the use of the parcel as a public parking area as a permitted use;
 - e. the parcel of land to be dedicated fronts on an open public street;
 - f. the parcel of land to be dedicated is presently available for use or improvement as a public parking area without further need of review or permitting by any agency, department or political subdivision of the Parish of St. Tammany, the State of Louisiana or the United States of America;
 - g. the Zoning Board finds that the dedication of the parcel to the City as a public parking area would serve to alleviate parking deficiencies in the Old Mandeville Business District; and
 - h. the City Council finds the dedication to be in the best interests of the City and, in accordance with all applicable requirements of the City Charter, affirmatively accepts the proffered dedication.
- 6. No building permit for the proposed use shall be issued by the Building Inspector until the applicant presents to him (1) a copy of the act effecting the dedication of

the parcel of land to the City which has been certified as a correct copy of the duly recorded original act by the St. Tammany Parish Clerk of Court, and (2) mortgage and conveyance certificates issued or updated by the St. Tammany Parish Clerk of Court to a day and time subsequent to the recordation of the act of dedication in the St. Tammany Parish conveyance records which certificates reflect no evidence that the City has not received full, unencumbered and merchantable title to the property in question. All property thus acquired by the City shall be used solely as a public parking area; such property may be sold or exchanged by the City provided that the net proceeds or property received as a result of that transaction be used to provide public parking spaces in the Old Mandeville Business District.

- 7. The Zoning Board shall at least annually review the mitigation fee set pursuant to this Section and make written recommendation to the City Council of any adjustments to that fee as are felt to be needed in order to carry out the intent of this Section. After receipt of the recommendation of the Zoning Board, the City Council may, by , make such adjustments to the fee as the Council deems necessary.
- 9.3.4 Required Connection between Parking Deficiency and Existing Buildings or Structures

Mitigation of parking deficiencies by compliance with the provisions of this Section 9.3, et seq. shall be permitted only for deficiencies in required on-site parking spaces resulting in whole or in part from the presence of existing buildings or structures intended for use in connection with the proposed use of the site. On-site parking deficiencies resulting from other causes shall not be excused or permitted, or a variance granted therefor, upon the condition of mitigation in the manner set forth under the terms of this Section.

9.3.5 Other Solutions to Parking Deficiencies not Prohibited

Nothing contained herein shall be interpreted or applied in such a manner as would prevent the Zoning Board from considering and granting parking variances for properties and uses otherwise applicable to the provisions of this Section 9.3, et seq. without the otherwise required mitigation by the applicant upon the applicant's demonstration that the on-site parking deficiency has been addressed by the applicant by some other means acceptable to the Zoning Board.

9.3.6 Return of Funds

1. If it is determined by the City Council that fee assessments collected pursuant to Section 9.3, et seq. have not been spent or encumbered for expenditure by the end of the tenth (10th) fiscal year of the City after the end of the fiscal year in which

the fee was received, the funds thus received shall be eligible for refund to the present owner of the property in accordance with the following procedures:

- a. The then present land owner must petition the City Council for the refund on or after the date from which the claimed refund is due.
- b. The petition for refund must contain or be accompanied by the following:
 - i. a notarized, sworn statement by the applicant that the applicant is the current owner of the property. If applicant is a corporation this statement shall be given by its President or by a duly authorized representative of the corporation and there shall be attached to the statement due proof of the appointment of the representative of the corporation. If the applicant is a joint venture, partnership, trust or other unincorporated entity or association this statement shall be given by the managing partner, trustee or other authorized representative of the entity or association. Evidence establishing such person's authority to act on behalf of the entity or association shall be attached to the statement;
 - ii. a copy of the receipt issued by the City at the time the fee was paid;
 - iii. a certified copy of the act establishing applicant's title to the property;
 - iv. a copy of applicant's most recent statements for parish and municipal ad valorem taxes.

The City may thereafter require of applicant such other information as is reasonably necessary to ascertain applicant's current ownership of the property.

- 2. Within sixty (60) days of the receipt by the City Council of the applicant's petition and supporting documents, the Mayor or his designee shall report to the City Council regarding the status of the fee for which a refund is requested. A copy of this report shall be forwarded to applicant at the same time that it is forwarded to the City Council. For the purpose of determining whether fees have been spent or encumbered, the City shall presume, in the absence of other compelling evidence to the contrary, that the first monies placed in the parking fund account were the first monies taken out of that account when withdrawals or encumbrances have been made.
- 3. When the money for which a refund is requested is still in the parking fund account and has not been spent or encumbered as of the date that the petition for refund and all supporting documents were received by the City Council, the money shall be returned to the applicant with legal interest thereon from the date of payment of the fee. Such payment shall be withheld by the City, however, until such time as applicant has submitted and the City has reviewed and found acceptable any other information that is reasonably necessary to ascertain applicant's entitlement to the requested refund. The City's liability for interest on refunded sums shall be abated during such time as should elapse between the City's request for the additional information and 5:00 p.m. of the second business day following applicant's submittal of such information.

9.3.7 Change of Use; Credit for Prior Mitigation

- 1. Should property for which a parking variance is granted under the provisions of Section 9.3, et seq. thereafter discontinue being used for the purpose for which the parking variance was granted, such discontinuance shall not entitle the owner of the property to a refund of any fees paid as mitigation for the parking deficiencies associated with that variance.
- 2. Any subsequent use of the property for which there is granted a variance to the number of required on-site parking spaces due to the presence of existing buildings or structures, shall be entitled to credit toward satisfaction of any mitigation required under the provisions of this Section as a condition of that variance, the number of parking spaces for which mitigation was provided in connection with any prior use of the property.

9.3.8 Penalty for Violation

1. Violators of the provisions of this Section shall be guilty of a misdemeanor punishable in accordance with the provisions of Sections 1-9 of this. Each day of a continuing violation shall constitute and may be penalized as a separate offense.

2. In addition to the penalty provided under 9.3.8(1), the City may seek to enforce the provisions of this Section by all applicable civil remedies provided by law including but not limited to recovery of monetary damages and injunctive relief.

ARTICLE 10 - SIGN CODE

10.1 TITLE

This Article shall be known as the "Sign Regulation Ordinance of the City of Mandeville, Louisiana" and for ease of reference, may also be referred to as the "Sign Code."

10.2 STATEMENT OF PURPOSE AND NEED FOR REGULATION

The Mayor and City Council of the City of Mandeville, Louisiana, after due and careful study and deliberation, and in full consideration of comments received from interested members of the general public, hereby find and declare:

- 1. That the people of the City have a primary interest in controlling the erection, location and maintenance of signs in a manner designed to protect the public health, safety and morals and to promote the public welfare.
- 2. That the rapid economic development of the City has resulted in a great increase in the number of businesses located in the City, with a marked increase in the number and size of signs advertising such business activities, creating conflicts between advertising signs themselves and between traffic regulating devices and advertising signs, which by their primary purpose draw mental attention to them potentially to the detriment of sound driving practices.
- 3. That it is necessary to the public safety that these official traffic regulating devices be easily visible and free from such nearby visual obstructions as blinking signs, distracting signs, as excessive number of signs, or signs in any way resembling official signs.
- 4. That it is necessary to provide equity and equality in displaying identification signs by establishing regulations on size and location of such signs to afford local businesses equal and fair opportunity to advertise and promote their products and services.

- 5. That, in addition thereto, the construction, erection and maintenance of large outdoor signs, suspended from or placed on top of buildings, walls or other structures constitutes a direct danger to pedestrian traffic below such signs, especially during periods when winds of high velocity are prevalent.
- 6. That the uncontrolled erection and maintenance of large or distracting signs seriously detracts from the enjoyment and pleasure of the natural scenic beauty characteristic of the Mandeville area, and the fact that such signs are intended to command visual contact grants them a proportionately greater role than other structures in determining the overall aesthetic and visual quality of the community.
- 7. That this code is enacted to provide for fair and equal treatment of all sign users and for a reasonable period of time for the elimination of non-conforming signs, to assure that sign users who erected signs prior to this code shall not have an unfair advantage over sign users who conform to this code.
- 8. That this code shall apply to the design, quality of materials, construction, location, electrification, illumination and maintenance of all signs and sign structures to be located within the City.
- 9. That the purpose of the adoption of the Sign Code is hereby declared to be:
 - a. The protection of the health, safety and welfare of the citizens of Mandeville;
 - b. The protection and preservation of property values and the promotion of economic well-being throughout the community; and
 - c. The preservation and maintenance of the visual and aesthetic quality of the community in accord with the character of the City of Mandeville and the surrounding area through the establishment and enforcement of standards for the design, quality of materials, construction, location, electrification, illumination and maintenance of all signs and sign structures within the City.

10.3 DEFINITIONS

For the purposes of this Sign Code, and unless the context indicates clearly contradictory intent, words used in the present tense include the future, the singular number includes the plural, the word "shall" is mandatory and not discretionary the word "building" includes "structures" except "sign structures."

10.3.1 Abandoned On-Premise Sign

An abandoned on-premise sign is an on-premise sign which no longer correctly directs or exhorts any person, advertises a bona fide business, lessor, owner, entity, product, or actively conducted, or product available on the premises where such sign is displayed.

10.3.2 Abandoned Off-Premise Sign

An abandoned off-premise sign is an off-premise sign which:

- 1. No longer correctly directs or exhorts any person;
- 2. Has fallen into disrepair or otherwise deteriorated as a result of a lack of maintenance, repair or upkeep; or
- 3. With regard to billboards, which carries no advertising message other than a message concerning its availability for lease or hire on its structure for any period of one hundred eighty (180) consecutive days.

10.3.3 Address Sign

A sign which only conveys the numeric address of the premises on which it is located.

10.3.4 Amplifying Sign

An amplifying sign is one which provides additional information regarding particular products or services available on a premise.

10.3.5 Attached Sign

An attached sign is any sign which is physically connected to and derives structural support from a building or building appurtenance.

10.3.6 Audible Sign

An audible sign is any sign which is designed to or which does produce sound discernable to a person of normal hearing situated off the premises on which the sign is located.

10.3.7 Bench Sign

A bench sign is an advertising message on any portion of a bench or other non-mobile structure or device intended for public seating or convenience.

10.3.8 Billboard

A billboard is an off-premise sign owned by a person, corporation, or other entity that engages in the business of selling or leasing the advertising space on that sign.

10.3.9 CAC

The CAC is the Community Appearance Commission as established by this Ordinance.

10.3.10 Changeable Message Sign

A changeable message sign is a sign on which the copy, message or sign panels may be, when specifically issued a permit as a changeable message sign, changed either electronically or manually in the field through the removal, replacement, or rearrangement of letters, symbols, blocks or panels designed for attachment to said sign.

10.3.11 Construction Sign

A construction sign is a temporary sign erected and maintained by an architect, contractor, developer, financial institution, subcontractor or materials supplier upon premises for which said person or persons is presently furnishing labor, materials, services or capital financing.

10.3.12 Directory Sign

A directory sign is an outdoor sign listing and identifying the occupants within shopping centers, industrial centers, retail centers, office centers, and other multi-use commercial or industrial sites.

10.3.13 Electrical Sign

An electrical sign is any sign containing a motor or wiring which is connected or attached or intended to be connected or attached to an electrical energy source.

10.3.14 Flags, Banners, Seals

Flags, banners and seals are mottos, emblems, designs, shapes or symbols on cloth, plastic, canvas or devices of similar type and materials intended to convey any message or to identify any person, place, idea or thing other than duly adopted flags or seals of nations, states, parishes or municipalities.

10.3.15 Freestanding Sign

A freestanding sign is a sign supported by a sign structure secured in the ground and which is wholly independent of any guy wire, support wire, building, fence, vehicle or object other than the sign structure, for support.

10.3.16 Home Occupation Sign

A home occupation sign is any on-premise sign advertising a home occupation.

10.3.17 Identification Sign

An identification sign is a sign which is limited to the name, address, and/or number of a building or institution, person, or entity which is primary to the identification of the premise and to a general statement of the activity carried on in the building or institution.

10.3.18 Illuminated Sign

An illuminated sign is any sign which has characters, letters, figures, designs or outlines illuminated by an interior or exterior light source which is primarily designed to illuminate such sign.

10.3.19 Individual Letter Sign

An individual letter sign is any sign made of self-contained letters that are mounted directly on the face of a building, a parapet, a roof edge of a building or on or below a marquee without being attached to a structure defined herein as a "sign face."

10.3.20 Inflatable Sign

An inflatable sign is any sign dependent in whole or in part for its structural integrity on the infusion into said sign of compressed air or other fluids, and specifically including balloons larger than two (2) feet in diameter or two (2) foot square in area or other gas or liquid filled figures.

10.3.21 Lights

- 1. <u>Searchlight</u> A strong or bright light with a reflector in a swivel so that its beam may be sent or directed in various directions.
- 2. <u>Beacon</u> A strong or bright light focused or directed in one or more directions.

- 3. <u>Flashing Lights</u> Any light or light source or reflection of light source which is intermittent in duration, color or intensity or which creates or is designed to create an illusion of intermittency in duration, color or intensity.
- 4. <u>String of Lights</u> A string of electrical conductors containing two (2) or more lights or light sockets.
- 5. <u>Laser</u> A device emitting a narrow, very intense beam of light waves that have been amplified and concentrated by stimulated atoms, or the light produced by such device.
- 6. <u>Tubular Signs</u> Neon will only be permitted if it is considered as an integral part of the sign being regulated. Anything within the boundary of the outline of the neon will be considered a part of the sign face.

10.3.22 Marquee Sign

A marquee sign is any sign attached to and made part of a marquee. A marquee is defined as a permanent roof-like structure projecting beyond a building wall at an entrance to a building or extending along and projecting beyond the building's wall and generally designed and constructed to provide protection against the weather. Signs painted or sewn onto awnings or canopies shall be considered marquee signs.

10.3.23 Monument Sign

A monument sign is a freestanding sign (a) the sign area of which is constructed or connected directly on or to a sign support consisting of a concrete slab base or foundation or a base or foundation of similar type of construction; or (b) which is of monolithic construction in which the sign's base or support is of uniform composition with the material comprising the sign area of said sign and the base or support of said sign is directly affixed in or to the ground. Provided, however, no sign base, foundation or support of which consists in whole or in part of above ground poles, piers, piling or similar types of supports exceeding twenty-four (24) inches in height measured above the ground shall constitute a monument sign.

10.3.24 Moving Message or Changing Image Sign

A moving message or changing image sign is any sign including public service signs designed to convey sign copy which changes in form or content with greater frequency than once an hour or which otherwise includes action or motion or the illusion of action or motion within its message or sign copy.

10.3.25 Murals

A work of art painted or otherwise applied to an exterior wall surface. Such work shall not serve to advertise or promote a business, service, product, cause, or event.

If such work does serve to advertise or promote a business, service, product, activity, cause, or event, it will be considered a "sign" and as such, will be subject to the Sign Ordinance.

10.3.26 Neon Signs

Neon will only be permitted if it is considered as an integral part of the sign being regulated. Anything within the boundary of the outline of the neon will be considered a part of the sign face.

10.3.27 Non-Conforming Sign

A non-conforming sign is any sign structure or sign which was lawfully erected and maintained prior to such time as it came within the purview of this code or any amendments thereto and which fails to conform to all applicable regulations and restrictions of this code, or a non-conforming sign for which a special permit has been issued.

10.3.28 Off-Premise Sign

An off-premise sign is a sign that directs a person to a different premise or location than that on which the sign is located; which identifies advertised goods, products, or services not available on the premises on which the sign is located; or which conveys a non-advertising idea or message; or identifies or advertises a business, person, firm or corporation not located on or occupying the premises where the sign is located; or which is not otherwise defined as an on-premise sign.

10.3.29 On-Premise Sign

An on-premise sign is a sign identifying or advertising a business, person, firm, corporation, activity, goal, product or service located or available on the premises where the sign is installed and maintained or which is displayed and maintained by the owner or occupant of the premises on which it is located and conveys a non-advertising or non-commercial idea or message.

10.3.30 Parapet Sign

A parapet sign is a sign extending above a roof line or which serves as a parapet. 10.3.31 Political Sign

A political sign is any sign urging the election or defeat of any candidate seeking any political office, or urging the passage or defeat of any ballot measure, but not including any billboard owned or maintained by a commercial firm or advertising company when leased or used as a political sign.

10.3.32 Portable Sign

A portable sign is any sign other than a trailer or vehicle sign that is not permanently affixed to a building, structure or the ground or a sign designed to be moved from place to place. These signs primarily include, but are not limited to: A frame or sandwich board signs, signs attached to wood or metal frames, and signs designed to be self-supporting and moveable.

10.3.33 Sign Premise

A Sign Premise is defined as the contiguous land in the same ownership which is not divided by any highway, street, alley or right-of-way. For purposes of this Article a single premise:

- 1. May include more than one lot of record when such lots are devoted to a single unity of use; or
- 2. May consist of a separate structure on the same lot of record when, in the opinion of the CAC, such separate structure appears to be a separate premise.

10.3.34 Premises Total

Premises total is defined as the total square footage of freestanding signage allowed on a premise under the provisions of this Sign Code.

10.3.35 Private Directional Signs

Private directional signs are on-premise signs directing vehicular or pedestrian traffic movement into a premise or within premises.

10.3.36 Project Sign

A project sign is a temporary sign announcing a proposed land development or construction project.

10.3.37 Projected Sign

A projected sign is a sign or visual image created by the projection of light onto a surface.

10.3.38 Projecting Sign

A projecting sign is any sign other than a wall sign affixed to any building or wall which sign has a leading edge extending twelve (12) inches or more beyond such building or wall. Projecting signs are of two (2) types:

- 1. <u>Fixed</u> A sign, other than a wall sign, which extends outward twelve (12) inches or more from the facade of any building and is rigidly affixed thereto.
- 2. <u>Swinging</u> A sign projecting twelve (12) inches or more from the outside wall or walls of any building which is supported by only one rigid support affixed thereto.

10.3.39 Public Directional Signs

Public directional signs are either:

- 1. Signs permanently or temporarily erected in the public right-of-way or on public property with the approval of the City Council which denote the name or route to any educational institution, public building of facility, historic place, shrine, church, synagogue, hospital, library or similar facility or institution; or
- 2. Signs permanently or temporarily erected identifying a person or entity who has undertaken to plant or maintain landscaping of that portion of the right-of-way.

10.3.40 Public Service Sign

A public service sign is a sign the primary purpose of which is to provide information as a service to the general public such as time, temperature or the promotion or announcement of public events, or other events of a civic, philanthropic, charitable or religious purpose of general interest to the public.

10.3.41 Real Estate Sign

A real estate sign is any temporary sign pertaining to the sale, lease or rental of land or buildings, which is erected or displayed on the lot or parcel to which it applies.

10.3.42 Resident Identification Sign

A resident identification sign is any on-premise sign limited in content to no more than the name of the premises, its municipal address and the names of the present occupant or occupants of the premises.

10.3.43 Revolving or Rotating Sign

A revolving or rotating sign is any sign whose sign face is designed to move or turn on any axis.

10.3.44 Roof Sign

A roof sign is any sign erected or painted upon, against or directly above a roof or on top of or above the parapet of a building.

10.3.45 Shingle Sign

A shingle sign is an identification sign customarily identified with the practice of the professions, such as law, medicine and dentistry.

10.3.46 Sign

"Sign" is defined as a medium of communication, including its structure and component parts, which is used or intended to be used to attract attention to its subject matter or location usually for advertising purposes, including paint on the surface of a building. Each distinctive message painted or placed on a building or other structure shall be considered an individual sign.

10.3.47 Sign Area

The area of a sign shall be defined as the square foot area enclosed within the perimeter of the sign face with each face contributing to the aggregate area of any sign. With respect to signs which are composed of individual symbols, letters, figures, illustrations, messages, forms, or panels, the sign area shall be defined as that area enclosed by one continuous line connecting the extreme points or edges of the advertising message. In cases where there is no definable simple geometric shape, the simplest geometric shape or rectangle enclosing the outer edges of the advertising message shall determine the sign area. In cases of backlighted awnings with advertising messages, the entire area of the awning shall be considered as the sign area.

10.3.48 Sign Face

Sign face is the part of the sign that is or can be used to identify, advertise, communicate, inform or convey a visual representation which attracts the attention of the public for any purpose. "Sign face" includes any background material, panel, trim, frame, color and direct or self-illumination that differentiates the sign from the building, structure, backdrop surface or object upon which or against which it is placed. The sign structure shall not be included as a portion of the sign face provided that no message, symbol or any of the aforementioned sign face criteria are displayed on or designed as part of the sign structure, unless it is outlined in neon.

10.3.49 Sign Height

Sign height of a sign shall be defined as the vertical distance from the adjacent street grade or upper surface of the street curb to the highest point of either the sign or sign structure. Elevated roadways shall not be used to measure height.

10.3.50 Sign Message

The words or symbols on a sign face which convey a message to those viewing the sign.

10.3.51 Sign Owner

A sign owner is that person who owns a sign and/or who is responsible for a sign. In those cases in which the sign owner cannot be determined; the owner of the premises upon which the sign is located shall be deemed the owner of the sign.

10.3.52 Sign Structure

A sign structure is the supporting structure upon which a sign or sign face is fastened, attached or displayed or is intended to be fastened, attached or displayed; provided however, this definition shall not include a building or fence.

10.3.53 Snipe Sign

A snipe sign is a sign which is tacked, nailed, posted, pasted, glued or otherwise attached to poles, stakes, fences, or to other like objects.

10.3.54 Street Banner Sign

A street banner sign is any banner sign which is stretched across and hung over a public right-of-way.

10.3.55 Subdivision Sign

A subdivision sign is a sign located within the boundaries of a subdivision identifying the subdivision and denoting the entrance or exit to the subdivision.

10.3.56 Temporary Sign

A temporary sign is any sign, the display of which is limited by law, ordinance, or regulation and which advertises a situation or event that is designed, intended, or expected to occur and be completed within a reasonably short or definite period after the erection of such sign.

10.3.57 Trailer Sign

A trailer sign is any sign or sign structure attached to or composed in whole or in part of a trailer frame or chassis or skid or skid frame or body or of any materials which have ever previously constituted in whole or in part such a trailer, skid, frame, chassis or body.

10.3.58 Vehicle Sign

A vehicle sign is any sign displayed on or from any mode of transportation, including but not limited to cars, buses, trucks/trailers, trains, boats, or airplanes.

10.3.59 Wall Sign

A wall sign is a sign other than a parapet sign which is painted on or which projects less than twelve (12) inches from the wall of a building, and is painted on, attached to or erected against any exterior wall or window of a building or structure with the exposed face of the sign being in a plane parallel to the plane of said wall or window and not extending above the building.

10.3.60 Window Sign

A window sign is any sign which is painted on, applied to, attached to or projected upon the exterior or interior of a building glass area, including doors, or located within one foot of the interior of a building glass area, including doors, whose identification, message, symbol, insignia, visual representation, logo type or any other form which communicates information, can be perceived from any off-premises contiguous property or public right-of-way.

10.3.61 Adoption of Definitions from other Sources

Any word, term or phrase used in this Sign Code and not otherwise defined herein but defined elsewhere in any other ordinance or regulation of the City of Mandeville shall be defined in accordance with the definition set forth in such other ordinance or regulation

unless the context in which such word, term or phrase is used in this Article indicates that the application of that definition would lead to a result which is inconsistent, unintended, or out of character with the purpose of this Sign Code and the plan of regulation set forth herein. All remaining terms of this Sign Code shall carry their usual and customary meanings. Terms indigenous to the industry shall be defined in accordance with their usual and customary understanding in the trade industry or profession to which they apply, unless such terms are otherwise defined herein.

10.4 PROHIBITED SIGNS

10.4.1 Enumerated

The following types of signs are prohibited within the City of Mandeville:

- 1. Abandoned signs
- 2. Audible signs
- 3. Beacons
- 4. Bench signs
- 5. Directory signs except as permitted for professional centers under the provisions of subsection 10.5.4.6(d)
- 6. Flashing signs
- 7. Inflatable signs
- 8. Lasers
- 9. Parapet signs
- 10. Portable signs
- 11. Projected signs
- 12. Revolving or rotating signs
- 13. Roof signs
- 14. Search lights
- 15. Signs attached to trees, shrubs or any living vegetative matter
- 16. Signs, other than public directional signs, public service signs, public information signs, subdivision signs or official notices which encroach into a public right-of-way
- 17. Signs resembling traffic control devices or emergency devices
- 18. Freestanding signs which restrict or impair visibility at the intersection of the right-of-way lines of two streets, or of a Street and a railroad right-of-way, or of a Street and a pedestrian or bicycle right-of-way
- 19. Snipe signs other than real estate signs, garage sale signs, political signs or signs used to convey or express a non-advertising or noncommercial idea or message

- 20. Strings of lights, except when used as holiday decorations during the period beginning the Sunday prior to Thanksgiving to the second Sunday in January of the succeeding year
- 21. Trailer signs
- 22. Any sign, other than a sign displaying or conveying a non-advertising or noncommercial idea or message, which is located in a development area (such as, for purposes of illustration only, shopping centers, industrial parks or developments, planned unit developments, subdivisions or neighborhoods of general historic or aesthetic value or character) and which, due to its proposed height, setback, location, landscaping, illumination or type of material by which it is constructed is out of character or harmony with the overall character of similar types of signage existing within the development area or is in violation of any publicly or privately imposed regulations or covenants established in order to impose an overall consistency or harmony of signage within the development area which are applicable to the sign.
- 23. Any sign not specifically defined and allowed by the provisions of this article.

10.5 REGULATIONS OF ON-PREMISE SIGNS AND MURALS

10.5.1 On-Premise Signs Allowable without a Permit

Subject to all provisions and requirements of this Article the following on-premise signs, where permissible, may be erected and displayed without the necessity of issuance of a permit by the Building Inspector:

- 1. Public Directional Signs;
- 2. Official notices duly issued by any court, public agency or officer;
- 3. Resident identification signs not exceeding three (3) square feet in sign area;
- 4. Flags and insignia of any government except when displayed in connection with a commercial promotion;
- 5. Integral decorative or architectural features of buildings, other than letters, trademarks, logos, or any feature containing moving parts or moving or flashing lights;
- 6. Indoor Signs not visible from any street right-of-way;

- 7. Private directional signs not exceeding five (5) square feet in sign area per sign;
- 8. Home occupation signs not exceeding two (2) square feet in sign area which are non-illuminated, and mounted flat against and parallel to the plane of the wall of the building to which the sign is attached;
- 9. Temporary window signs displayed on the inside of windows and intended for the purpose of disseminating information about special sales or promotional campaigns in accordance with the provision of Section 10.5.1;
- 10. Fuel service station pump signs, oil rack signs, and pricing signs in accordance with the provisions of Section 10.5.1.2;
- 11. Temporary real estate signs in accordance with the provisions of Section 10.5.1.3;
- 12. Temporary yard or garage sale signs in accordance with the provisions of Section 10.5.1.4;
- 13. Menu boards for drive-thru [drive-through] facilities providing food for offpremise consumption which signs do not exceed forty-eight (48) square feet in sign area and which are located so as not to be visible from the street rightof-way from which primary access to the premise is obtained;
- 14. Vehicle signs in accordance with the provisions of Section 10.5.1.5;
- 15. Address signs in non-residential and combined use districts in accordance with 10.5.1;
 - 16. Signs which convey a non-advertising or non-commercial idea or message, but only in accordance with the applicable provisions of Section 10.5.3.4 or Section 10.5.4.7 of this Sign Code.
 - 17. One Roadside Vendor Sign will be permitted on signs facing each side not to exceed a maximum of 16 square feet.
 - 18. Chamber of Commerce, Business of the Month, banners for the month of the award
- 10.5.1.1 Temporary Window Signs; Additional Requirements for Erection or Display without Necessity of Permit

Any temporary window sign erected or displayed without necessity of a permit pursuant to the provisions of Section 10.5.1 of this Sign Code;

- 1. Shall be constructed of such materials as to indicate the temporary nature of the sign;
- 2. Shall not, in the aggregate, cover more than fifty (50%) percent of the area of any windows in which such signs are displayed.
- 10.5.1.2 Fuel Service Station Pump Signs, Oil Rack Signs, Tire Rack Signs, and Pricing Signs; Additional Requirements for Erection or Display without Necessity of a Permit

Any fuel service station pump signs, oil rack sign, tire rack sign or pricing sign erected or displayed without necessity of a permit pursuant to the provision of Section 10.5.1 of this Sign Code shall conform to the following regulations and requirements:

- 1. <u>Fuel Service Station Pump Signs</u> Fuel service station pump signs may be displayed on fuel pumps to provide required information to the public regarding the available fuel such as "gallons," "price," "octane rating" and "type of fuel." If a trade name of the business or supplier is incorporated into the name or designation of the different types of fuels available, said trade name and any associated symbols therewith may be displayed on the pumps provided that such signs are flat signs that do not exceed three (3) square feet per sign face and an aggregate area of six (6) square feet of sign face per pump.
- 2. <u>Oil Rack Signs</u> If a fuel service establishment markets engine oil on the pump island, any identification signs on the merchandise itself visible to the public shall be permitted. Additional signs on the oil rack may be permitted provided that each such sign shall not exceed three (3) square feet per sign face with an aggregate area of six (6) square feet of sign face per rack.
- 3. <u>Tire Rack Signs</u> If a fuel service station, tire store, auto repair shop or any other business which markets tire displays the tires on racks visible to the public, additional signage on the tire rack identifying the tires displayed on that rack may be permitted, provided that such additional signage shall not exceed three (3) square feet per sign face with an aggregate area of six (6) square feet of sign face per rack.

- 4. Pricing Signs A sign advertising the price of motor fuel, other than pump signs, shall be permitted provided that such sign shall not exceed twelve (12) square feet of sign face per sign with an aggregate sign face area of twenty-four (24) square feet per sign. Such sign shall not exceed five (5) feet in height, if freestanding. One such sign per premise frontage shall be allowed with a maximum of two (2) such signs per premises. However, should such pricing sign be attached to, or be made part of, a permanent freestanding sign which identifies the premises, such sign shall not be allowed without permit and shall conform to the requirements of freestanding signs in the applicable zoning district. Pricing signs, however, when made a part of freestanding identification signs and when no additional freestanding pricing signs are placed on the premises may be allowed in excess of the allowable square footage of identification signage, provided that the maximum square footage for the premise is not exceeded in total signage.
- 10.5.1.3 Temporary Real Estate Signs; Additional Requirements for Erection or Display of Temporary Real Estate Signs without Necessity of a Permit

Any real estate sign erected or displayed without necessity of a permit pursuant to the provisions of Section 10.5.1(11) of this Sign Code shall conform to the following regulations and requirements:

- 1. In all residential zoning districts, real estate signs shall be limited to:
 - a. One non-illuminated real estate sign, not exceeding six (6) square feet in sign face area and six (6) square feet in total area and, if freestanding, not exceeding five (5) feet in height. Premises with two (2) or more frontages

shall be allowed one additional sign conforming with all of the requirements of this subsection; and

- b. Not more than two (2) additional real estate signs, each not exceeding one square foot in sign face area and two (2) square feet in total sign face area; provided however, that these additional signs may only convey information regarding the features of the property or home offered for lease or sale, state that the property or home is sold, announce that the house or property is open for inspection, or identify the agent listing the property or house.
- 2. In all other zoning districts, any such signs shall be limited to one real estate sign per listing broker on any premises offered for sale, rent or lease. Such sign shall be non-illuminated and shall not exceed thirty-two (32) square feet in sign area. A double-faced real estate sign is permitted, provided that such

sign shall not exceed sixteen (16) square feet in area per sign face and an aggregate sign face area of thirty-two (32) square feet. If freestanding, any such sign, whether single or double-faced, shall not exceed seven (7) feet in height. Property with two (2) or more premises Street frontages shall be allowed to display one additional sign conforming with all of the requirements of this subsection on a second frontage.

- 3. For purposes of the regulation under this code of real estate signs only, and irrespective of whether such sign is erected or displayed with or without necessity of a permit, each separately designated unit of a condominium development created by virtue of written instruments duly recorded in the office of the St. Tammany Parish Clerk of Court shall be considered a separate premise and real estate signs relating to any such unit shall be regulated depending upon the residential or commercial use of the unit in question, in accordance with the applicable provisions of this Sign Code.
- 10.5.1.4 Temporary Yard or Garage sale Signs, Additional Requirements for Erection or Display without Necessity of a Permit

Any garage sale sign erected or displayed without necessity of a permit pursuant to the provisions of Section 10.5.1(12) of this Sign Code shall be limited to one non-illuminated sign displayed on the residential premises at which the sale is conducted, provided that such erection and display shall be limited to three (3) days in any sixty-day period. Such sign shall not exceed six (6) square feet in sign area. Two (2) additional garage sale signs, a maximum of six (6) square feet may be placed off-premise on private property with permission of the property owner.

- 10.5.1.5 Vehicle Signs; Additional Requirements for Erection or Display without Necessity of Permit
 - 1. Vehicle signs may be displayed on any vehicle operated in the daily conduct of any business enterprise without necessity of a permit so long as such signs:
 - a. Are limited to content to a noncommercial message; or
 - b. Are required to be displayed by any federal, state or local law, rule or regulation; or
 - c. Are limited in content to an identification of the name, address and/or telephone number of the business entity operating the vehicle or the goods, products or services available from or provided by said business entity;

- d. Are not parked in front of or in line with any greenbelt areas when on the premise of the business entity operating or advertising on such vehicle;
- e. Are on a vehicle which is operable and not parked primarily for the purpose of signage.
- 2. Vehicle signs shall not be used as off-premise signs and shall not be displayed or parked on sites other than the premise of the business entity operating such vehicle other than when the vehicle is being used in connection with the business operations of the entity operating said vehicle. Such vehicles may also be parked at the residence of its operator, so long as such vehicle and vehicle sign is in conformance with the provisions and restrictions of the zoning district in which the residence is located.
- 3. Billboards may not be erected or displayed on any vehicle.
- 10.5.1.6 Address Signs in Non-Residential and Combined Use Districts Allowed Without a Permit

One address sign may be provided for each premise in addition to all other permitted signs in accordance with the following requirements:

- 1. Address sign shall not exceed two (2) feet by three (3) feet in sign area
- 2. A minimum setback from the projected line of five (5) feet shall be required
- 3. The maximum height of an address sign shall be two (2) feet from the ground.
- 10.5.2 On-Premise Signs allowable by Temporary Permit Only

Subject to all provisions and requirements of this Article, except for the requirement for the approval of the Community Appearance Commission, the following on-premise temporary signs, where permissible, may be erected and displayed upon the issuance by the Building Inspector of a temporary sign permit:

- 1. Street Banner Signs;
- 2. Flags, Streamers, Banners and Pennants;
- 3. Political Signs;
- 4. Construction Signs;

- 5. Proposed Project Signs;
- 6. Vehicle Signs;
- 7. Temporary Signs preceding permanent sign approval
- 10.5.2.1 Public Service Signs; Limitations and Requirements for Temporary Permit

Public service signs including Street banner signs are permitted only in connection with the promotion or identification of special events of a civic, philanthropic, charitable or religious purpose.

- 1. <u>General Application</u> Any person, firm, corporation or organization in charge of any festival, spectacle, play, show, or other event of such a general civic and public nature and who is in charge of placing, erecting, constructing and maintaining any public service banner, flag, emblem, bunting or freestanding public service sign upon or over any public Street, or other public place in the City shall first secure a temporary permit subject to the approval of the Community Appearance Commission and the conditions provided herein prior to the placement of the sign.
- 2. <u>Application for Permit</u> Applications for such a permit shall state the name of the person, firm, corporation or organization sponsoring the event; the location where such device or devices are to be installed and the contemplated dates during which such devices shall remain upon or over any Street or other public place in the City. Such application shall have attached to it a chart or drawing showing that the device would not interfere with traffic or the safety of persons using such public places.
- 3. Regulations for the Placement of Public Service Signs in Street Rights-of-way Public service announcements can be placed on Public Service Message Signs located on North, East, and West Causeway Approaches in accordance with the application requirements.
- 10.5.2.2 Flags, Streamers, Banners and Pennants; Limitations and Requirements for Temporary Permits
 - 1. Flags, streamers, banners or pennants may be displayed in connection with grand openings or special events no more than once semi-annually for any one business entity or applicant. Such signs may be displayed for a period not to

- exceed fourteen (14) consecutive calendar days upon the issuance of a temporary permit by the Building Inspector.
- 2. Flags, streamers, banners or pennants may be displayed in connection with events governed by, and subject to all provisions and restrictions set forth in, Section 10.5.2.1 of this Sign Code.
- 3. <u>Application for Permit</u> Applications for such a temporary permit must state the name of the person, firm, corporation or organization sponsoring the event; the locations where such device(s) are to be installed and the contemplated dates during which such devices shall remain on display.
- 20. Banners shall not exceed 32 square feet.
- 10.5.2.3 Political Signs; Limitations and Requirements for Temporary Permits
 - 1. Any person desiring to distribute or display political signs in connection with an organized campaign in support of or opposition to and candidacy, political slate or ticket, or ballot proposal shall first make application to the building inspector for the issuance of a temporary sign permit. Such application shall include the name, address and telephone number of the applicant. The applicant for the permit shall provide the building inspector with specimen copies of all signs to be distributed or displayed under the permit.
 - 2. The Building Inspector shall issue the requested permit upon receipt of the following:
 - a. An application with all requested information supplied
 - b. A specimen of the sign(s) to be used as required by paragraph 1, above
 - c. A surety bond in the amount of \$100 redeemable by the City of Mandeville if any sign permitted hereby remains on any property for more than ten (10) days following the election for which it was permitted.

Said permit shall authorize the distribution, erection and display of an unlimited number of signs of the type or types submitted as specimens by the applicant and shall allow for the placement of one sign to be erected per premise/Street frontage.

3. Any such permit shall be issued for a period of time not to exceed ninety (90) consecutive calendar days; provided, however, that in the event that signs are distributed, erected or displayed under any such permit in connection with

any candidacy or ballot proposal which involves more than one election, the permit shall be automatically extended to the tenth day following the date of the general election to which the sign pertains.

- 4. No political sign shall be erected or displayed in any public right-of-way. No solicitation of votes shall take place in the public right-of-way.
- 5. No political sign shall exceed thirty-two (32) square feet in sign face area. No political sign can be stacked one on top of the other in billboard fashion.
- 6. The applicant shall remove all political signs erected or displayed under any permit issued to him under the provisions of this section no later than ten (10) calendar days following the last election to which the sign pertains. Upon the failure to timely remove such signs the City may thereafter remove and dispose of any remaining signs.

10.5.2.4 Construction Signs; Limitations and Requirements for Temporary Permits

One non-illuminated construction sign may be permitted on the premises being developed or improved subject to the following conditions and requirements:

- 1. A building permit for the project must have been obtained prior to the issuance of the sign permit.
- 2. In residentially zoned districts such sign shall not exceed sixteen (16) square feet in area and shall be limited to the denoting of the architect, engineer, contractor, subcontractor, owner and/or financing agency providing labor, materials, services or financial capital for the proposed construction. Such sign may be displayed only during construction and for a period of not more than ten (10) days after completion of actual construction identified by the first issuance of a certificate
- 3. In any non-residential or combined use district such sign shall not exceed thirty-two (32) square feet in area and shall be limited to the name of the project and the denoting of the architect, engineer, contractor, subcontractor, owner and/or financing agency providing labor, materials, services or financial capital for the proposed construction. Such sign may be displayed only during actual construction and for a period of not more than ten (10) days after completion of the construction as identified by the first issuance of a certificate of occupancy for the project or development in question or upon expiration of the building permit for the project or development, whichever is first to occur.

4. The location of such sign shall be in strict conformance with all applicable setback requirements of this code and of other laws and ordinances of this City, provided, however, that in no instance may such sign be located within ten (10) feet of the property line of the premises on which it is displayed.

10.5.2.5 Proposed Project Signs; Limitations and Requirements for Temporary Permits

One non-illuminated temporary sign announcing a proposed land development may be erected on the premise proposed for the project provided that such sign does not exceed fifty (50) square feet in area, is set back at least ten (10) feet from any property line, and is removed within one year from the date the sign permit for its erection was issued or upon the issuance of the sign permit for a construction sign or the lapse of sixty (60) days from the issuance of a building permit for the project whichever is first to occur. The applicant for such a sign permit shall post with the Building Inspector a cash bond of one hundred dollars (\$100.00) payable to the City prior to the issuance of the required sign permit. If construction of the project is not begun within one year of the issuance of this sign permit and the sign is not timely removed, the cash bond shall be forfeited and the City shall be authorized to remove and dispose of the sign.

10.5.2.6 Vehicle Signs; Limitations and Requirements for Display by Temporary Permit

- 1. Vehicle signs not otherwise allowed under the provisions of Section 10.5.1.5 may be displayed only in connection with events governed by and subject to all restrictions and provisions set forth in Sections 10.5.2.1 and 10.5.2.2 of this Sign Code and subject to the following restrictions:
 - a. A vehicle sign shall not exceed a maximum of one hundred twenty (120) square feet visible from any one direction.
 - b. The vehicle upon which the vehicle sign is placed shall not be parked in front of or in line with any greenbelt areas.
- 2. Billboards may not be erected or displayed on any vehicle.

10.5.2.7 Temporary Signs Preceding Permanent Sign Approval

1. One temporary attached identification sign not exceeding thirty-two (32) square feet may be permitted for a period of not more than thirty (30) days for an occupant who has no other on-premise signs to identify the occupant's business subsequent to the filing of a complete application for sign approval and approval of the CAC.

- 2. An additional sixty (60) days extension may be requested in writing by the sign applicant subsequent to the filing of a complete application for a sign variance from the Zoning Board.
- 10.5.3 On-premise Signs allowed in all Residential Zoning Districts
- Signs allowed in all residential zoning districts without necessity of a permit shall be limited to those identified as numbers (1) through (6), inclusive, (8), (11), (12), (14), (15), and (16) in subsection 10.5.1 of this sign code, but in no case shall any such signs other than those identified as number (16) in subsection 10.5.1 be illuminated.
- 10.5.3.2 Signs allowed in all residential zoning districts upon issuance of a temporary permit shall be limited to those identified as numbers (3) through (6), inclusive, in Section 10.5.2 of this Sign Code.
- 10.5.3.3 Subdivision signs not greater than forty-eight (48) square feet in sign area or ninety-six (96) square feet for a two-sided sign may be located at the entrances to approved subdivisions by permit.
- 10.5.3.4 Non-advertising signs or signs conveying non-commercial ideas or messages displayed in any residential zoning district shall not exceed eight (8) square feet in total sign area and, if freestanding, shall not exceed a height of six (6) feet above the ground.
- 10.5.3.5 Additional on-premise signs allowed in R-1, R-1X and R-2 Zoning Districts. The following signs, in addition to those allowed under Section 10.5.3.1 and Section 10.5.3.2 may be erected and displayed, by permit, in the R-1 and R-2 zoning districts.
 - 1. For premises containing permitted non-residential structures or uses, other than accessory uses, one non-illuminated flat wall sign, not exceeding twenty-four (24) feet in area shall be permitted.
 - 2. Churches, schools or public buildings shall be permitted to erect on their premises in addition to the permitted wall sign as allowed in Section 10.5.3.4(1), a freestanding sign, either non-illuminated or illuminated, no closer than fifteen (15) feet to the right-of-way, not to exceed twenty-four (24) square feet per sign face an aggregate sign face area of forty-eight (48) square feet and not exceeding seven (7) feet in height, provided that such sign is so situated and shielded that its source of illumination is not visible from any abutting residence.

10.5.3.6 Additional On-Premise Signs Allowed in R-3 and MH Zoning Districts

The following signs may be erected and displayed, by permit, in the R-3 and MH zoning districts:

- 1. One (1) attached sign identifying the premises sized at a ratio of one square foot per one linear foot of building frontage but under no circumstance to exceed forty-eight (48) square feet, shall be permitted. No attached sign shall be required to be less than twenty-four (24) square feet in sign area.
- 2. One (1) freestanding sign identifying the premises may be permitted in lieu of an attached sign, provided it does not exceed thirty-two (32) square feet per sign face and aggregate area of sixty-four (64) square feet for a two-sided sign, does not exceed seven (7) feet in height, is not closer than fifteen (15) feet to the public right-of-way and is not erected in any required greenbelt. No sign shall be allowed in any area between the right-of-way and any required greenbelt.
- 3. Any sign permitted in an R-1, R-1X or R-2 district.
- 10.5.4 On-Premise Signs Allowed in Non-Residential and Combined Use Districts
- 10.5.4.1 Any sign which does not require a permit under the provisions of Section 10.5.1 of this Sign Code may be erected or displayed in any I, O, B-1, B-2, B-3, B-4, O/R, M-1, M-2, PM-1, PM-2 or Planned Zoning District.
- Any sign identified in Section 10.5.2 of this Sign Code may be erected and displayed in any I, O, B-1, B-2, B-3, B-4, O/R, M-1, M-2, PM-1, PM-2 Zoning District upon issuance of all requisite permits.
- 10.5.4.3 Regulation of On-Premise Signs in Planned Zoning District

Unless otherwise specifically provided for within the terms of the approved site plan, the regulation of signs within planned residential districts shall comply with the signage regulations applicable to an R-2 Residence District and the regulation of signs in all other types of planned districts shall comply with the signage regulations applicable to a B-1 Neighborhood Shopping District. In no event, however, shall any approval given to a site plan for a planned district allow the erection or display within the planned district of off-premise signs other than those listed in, and only in accordance with the requirements set forth in Section 10.6.1 and 10.6.2 of Article 10, infra, of this Article.

10.5.4.4 Regulation of On-Premise Signs in B-3, PM-1 and PM-2 Districts

The following signs shall be allowed in addition to those listed in Sections 10.5.4.1 and 10.5.4.2 in conjunction with the issuance of a sign permit.

10.5.4.4.1 Subdivision Signs in B-3, PM-1 and PM-2 Districts

Subdivision signs not greater than forty-eight (48) square feet in sign area or ninety-six (96) square feet for a two-sided sign may be located at the entrances to approved subdivisions by permit.

10.5.4.4.2 Regulation of On-Premise Signs in B-3, PM-1 and PM-2 Districts

- 1. As this subsection is applied to multi-occupant commercial centers, only the name and/or address of the center shall be considered its main use, and only one freestanding identification sign which identifies that main use shall be permitted in any center.
- 2. On single and multi-occupancy premise one of the following is permitted:
 - a. One freestanding identification sign not to exceed thirty-two (32) square feet in sign area per sign face and an aggregate sign area of sixty-four (64) square feet if two-sided, not exceeding ten (10) feet in height and situated no closer than five (5) feet to any public right-of-way.
 - b. Attached signage per Street frontage not to exceed a ratio of one square foot per one linear foot of store frontage but under no circumstance to exceed sixty-four (64) square feet in area. No attached sign shall be required to be less than twenty-four (24) square feet in area.
- 3. In addition to the allowable site identification sign one wall sign not to exceed maximum of eight (8) square feet shall be allowed each occupant in a multi-occupant premises.

10.5.4.4.3 Regulation of Special Signs in the B-3 District

The B-3 District, being the historic center of the City and the location and focus of special events such as festivals, parades and celebrations and being an area acknowledged as having a pedestrian orientation, provides a unique opportunity for special activities to occur. In addition, the mixture of residential and non-residential uses in the district, with many commercial establishments being located in former residential structures calls for special treatment of signage and promotional identification to distinguish the location of non-

residential uses from residential uses. In response to these unique characteristics and needs of the district, the following criteria shall be utilized by the Community Appearance Commission in evaluating applications for Special Event and Sign Permits:

- 1. Identification banners not to exceed eight (8) square feet in area may be displayed on non-residential sites to distinguish non-residential uses from residential uses.
- 2. Changeable message signs not to exceed eight (8) square feet total on one side shall be allowed during specific hours to be determined in conjunction with approval by the CAC of the issuance of a permit.
- 3. During special events and seasonal campaigns the Community Appearance Commission may permit special signage of a scale and size appropriate to the historic and human-scale quality of the B-3 area.
- 10.5.4.5 Regulation of On-Premise Signs in I, O, B-1, B-2, B-4, O/R, M-1, M-2, B-1, B-2 and B-4 Districts

10.5.4.5.1 Subdivision Signs

Subdivision signs not greater than forty-eight (48) square feet in sign area or ninety-six (96) square feet for a two-sided sign may be located at the entrances to approved subdivisions by permit.

10.5.4.6 Regulation of On-Premise Signs in I, O, B-1, B-2, B-4, O/R, M-1 and M-2 Districts.

In conjunction with the issuance of a building permit, a signage plan designating location, size, type and color of signage must be approved by the Community Appearance Commission for all commercial developments.

10.5.4.6.1 On-Premises in I, O, B-1, B-2, B-4, O/R, M-1 and M-2 Districts

One of the following two (2) alternatives shall be permitted in addition to those signs listed in Section 10.5.4.1 and Section 10.5.4.2. At the discretion of the Community Appearance Commission, alternate three shall be permitted.

a. One (1) freestanding identification sign located at least fifteen (15) feet from the Street right-of-way and to the rear of the required greenbelt. (No sign shall be allowed in any area between the right-of-way and the required greenbelt.) A freestanding sign must be set back from the public right-of-way from which

it is intended to be viewed the same distance as any setback line required on any residentially zoned property facing the same public right-of-way within one hundred (100) feet of the sign. This additional setback provision affects only signs on commercial, institutional and industrial premises on the same block and on the same side of the right-of-way as residentially zoned property.

- b. Such freestanding sign-face shall be sized at a square foot ratio equal to one-half (.5) times the Street frontage of the premise on which the sign is located. The maximum allowable sign area shall be one hundred twenty (120) square feet per sign face or a maximum of two hundred forty (240) square feet aggregate total for a two-sided back-to-back sign. A maximum of forty (40) square feet, or eighty (80) square feet aggregate total for a two-sided back-to-back sign, shall be allowed on premises with a frontage of less than eighty (80) linear feet. No freestanding sign face, shall exceed twenty (20) feet in height or except for monument signs exceed twenty (20) feet in width.
- c. On-premises with non-intersecting Street frontages a second freestanding sign on a separate Street frontage at a minimum distance of two hundred fifty (250) feet, following the Street frontage line, from the first sign may be allowed in accordance with the requirements of (b) above.
- d. For purposes of this Section as it applies to multi-occupant centers, only the name of the center, the address of the center and/or a general statement of the use, such as "Shopping Center," shall be allowed on the freestanding identification sign.
 - 1. Notwithstanding anything contained herein to the contrary, in no event shall any one occupant of a multi-occupant center be allowed more than 50% of the total signage allowable for the premises. This limitation shall be interpreted to include signage in which the name of an occupant is included within the name of the multi-occupant center.
 - 2. In the event of the conversion of a single occupant premise to a multi-occupant center, the existing signage shall be reconfigured as necessary to effect compliance with Section 10.5.4.6.1(1)(d)(I), and this requirement shall be incorporated Section 5.4.3. Development Permit **Application** into Requirements for Non-Residential and Multi-Family Developments as necessary to ensure compliance with this section.

- 2. One attached identification sign per facade not to exceed one and one-fourth (l.25) square feet per linear foot of facade frontage upon which the sign is to be placed. On building facades with linear frontage of less than twenty-six (26) feet a maximum of thirty-two (32) square feet of total signage shall be allowed. However, no attached signage may be placed on the wall of the building which faces the exterior of a premise as designed in the direction of an interior property line.
- 3. The premise total based on a ratio of one-half (.5) times the Street frontage of the premise, may be divided as follows:
 - (a) One freestanding sign not to exceed a maximum allowable sign area of one hundred twenty (120) square feet per sign face or a maximum of two hundred forty (240) square feet aggregate total for a two-sided back-to-back sign; and
 - (b) Attached signage, not to exceed a ratio of 20% (.2) of the one-half (.5) times the Street frontage of the premise. Attached signage not to exceed 24 square feet.

10.5.4.6.2 Monument Signs

A bonus of twenty (20) percent of the total square footage of freestanding signage allowed under Section 10.5.4.6.1 (1)b shall be allowed for a monument type sign in lieu of a freestanding identification sign. Any or all of this bonus may be allocated to an attached identification sign or included in the sign area of the monument sign. At the discretion of the premise owner a maximum of fifty (50) percent of the allowable square footage of monument sign may be used to identify major occupants of the premises. A monument sign may be permitted in the most rearward ten (10) feet of the front fifteen (15) feet of the premises, provided that such monument sign is landscaped in accordance with Section 10.9.3 "Landscaping Requirements In Conjunction With the Issuance of a Sign Permit." Monument signs shall not exceed a maximum of seven (7) feet in overall height from the ground. Mounds or berms on which a monument sign is located shall be no more than three (3) feet in height measured from the ground level of the lot. Changeable message signs and public service signs attached to and an integral part of the structure of a freestanding monument sign shall be allowed, however, such signs shall be considered in the calculation of the allowable square footage of the monument sign.

10.5.4.6.3 Occupant Signs

On multi-occupancy premises, in addition to the identification signs permitted under Section 10.5.4.6.1, the following additional signage shall be permitted for each occupant:

- 1. One attached store front sign per occupant to be sized at a ratio of one and one-fourth (1.25) square feet of sign area per linear foot of the facade of the store front. For store fronts of less than twenty-three (23) linear feet the maximum allowable size sign shall be thirty-two (32) square feet. However, all occupant signs shall maintain a minimum distance of one foot from the lease line of the occupant's portion of the facade and the linear footage shall be measured along the wall of the facade on which the sign will be located
- 2. On store frontages located at the corner of a building, which face two different Street frontages of a 90 degree angle, or if a business occupies an entire separate structure within a center, additional wall signs, the area of which shall be calculated as per Section 10.5.4.6.3(1) above, shall be permitted, provided that a minimum distance of thirty (30) feet, measured along the store front, is maintained between the extremities of any two signs and each sign is mounted on a separate wall facing in a separate direction. However, under no circumstances shall an attached occupant sign be allowed on a wall which is finished in a manner inferior to the quality of the facade where the main entrance is located or on a wall where the placement of the sign will call attention to building equipment or the unfinished side of a false building facade.
- 3. One rear identification wall sign must be displayed on or at a rear door of each separate business entity for purposes of emergency access and deliveries only. Such sign shall be limited in size to the minimum size required by the state fire marshal. Any door not utilized as a primary entrance-way for patrons during normal business hours or not opening directly onto the patron area of any premise shall be considered a rear door. No door located on any store frontage on which there exists a door utilized as a primary entrance-way for patrons during normal business hours or which opens directly onto the patron area of any premise shall be considered a rear door.
- 4. An occupant in a commercial or industrial center shall be permitted, as additional signage, one non-illuminated identification sign not to exceed eight (8) feet in area identifying the occupant and such additional sign shall be located on or within ten (10) feet of the primary public entrance of the occupant.
- 5. Occupants located on the interior of a building without benefit of store frontage shall be allowed one maximum twenty-four (24) square feet attached sign at the entrance nearest the occupant's space.

10.5.4.6.4 Professional Centers

- 1. For purposes of this Section, professional centers are defined as multi-use business or commercial establishments:
 - a. In which the primary business activity of occupants consists of the rendition to the general public, or a discrete segment thereof, of specialized skills or services; and
 - b. In which no tangible personal goods are sold or offered for sale or in which offers for the purchase of tangible goods are not taken or solicited.

Examples of professional occupants would include, but need not be limited to, offices maintained by physicians, surgeons, nurses, midwives, psychologists, counselors, therapists, dentists, orthodontists, lawyers, engineers, architects, accountants, chiropractors, optometrists, bookkeepers, notaries public, appraisers, realtors, insurance agents and brokers, and persons pursuing similar professions or vocations.

- 2. For purposes of this Section a professional center is defined as a commercial center in which at least seventy-five (75) percent of the occupied or occupiable units are occupied by professional occupants.
- 3. A professional center may include in or attach to its otherwise permissible identification sign information further identifying the name (including professional or educational degrees earned) of each of its professional occupants and a designation of no more than two (2) of the professional services provided by each such occupant, provided that:
 - a. No such attachment to an identification sign may exceed eight (8) square feet in sign area or sixteen (16) square feet for a two-sided sign; and
 - b. Any such attachment must conform to the character, materials and type of the identifying sign to which it is attached; and
 - c. The aggregate area of all such attachments to an identification sign may not exceed one-half (50%) of the area of any permissible identifying sign which could be displayed on the premises; and
 - d. All professional occupants desiring to display exterior signage in the center take part in such plan of on-site signage.
- 4. Any further on-site sign relating to any such professional occupant shall be limited to a non-illuminated wall or projecting sign with a total sign area of not

more than two (2) square feet per sign face or four (4) square feet aggregate total for a two-sided sign:

- a. Identifying the occupants of the premises and no more than two of the professional services provided; and
- b. Located on or within three (3) linear feet of the primary public entrance to the professional office of the occupant.
- 5. A professional occupant shall not lose his designation as a professional occupant because of the sale at his office of products ancillary to his professional; provided that the availability of such products is not advertised or announced other than by signs located within, and not visible from without, such premises.

10.5.4.7 Identification and Amplifying Signs

Unless otherwise restricted herein any person has the option of apportioning the total permissible sign area for any premise between the permitted identification signs and amplifying signs. Amplifying signs are limited to the identification, which may include the price of particular products for sale or services available on the premises or to signs conveying a non-advertising or non-commercial idea or message. The combined area of the identification sign and all amplifying signs shall not exceed the total allowable sign area for the premise or occupant as elsewhere set forth in this Sign Code, nor shall any amplifying sign be erected as an additional and separate freestanding or fixed projecting sign for any premise, unless such use is specifically allowed by express provision of this code.

10.5.4.8 Seasonal Cloth Banners

Temporary seasonal cloth banners such as but not limited to those attached to light standards within a parking area may be allowed on multi-occupant premises upon review of and approval by the CAC prior to issuance of a temporary permit for the display of such banners subject to the following requirements:

- 1. No seasonal cloth banner shall exceed eight (8) square feet in area
- 2. The total number, location and method of attachment display shall be approved by the CAC
- 3. No advertising message shall be conveyed on the banners, however, non-advertising seasonal greetings are allowed.

4. The identifying name of the multi-occupant premises may be included on the banner but such name shall not exceed fifty (50) percent of the banner area.

10.5.5 Comprehensive Signage Plans

10.5.5.1 Purpose

The Comprehensive Signage Plan procedure is intended to encourage innovative, creative and coordinated signage design for a unified development site in a flexible procedure which allows signage which is not in strict compliance with the provisions of this Sign Code, but which is appropriate to the character of the development, provides adequate identification and information, provides a good visual environment, promotes traffic safety and is regulated to the extent necessary to be consistent with the purpose and intent of this Sign Code as specified in Section 10.2.

10.5.5.2 Scope of Procedure

- 1. <u>Existing Developments</u> In conjunction with an application from the owner of the premise, the CAC may review and approve or disapprove applications for Comprehensive Signage Plans in commercial centers with gross leasable area in excess of 50,000 square feet, industrial centers, hospitals and planned district developments.
- 2. <u>Proposed Developments</u> Prior to site plan approval for new development, the CAC shall review and approve or disapprove comprehensive signage plans in commercial centers with gross leasable areas in excess of 100,000 square feet, industrial centers, hospitals and Planned District developments in conjunction with the appropriate site plan review process for the type of development proposed.
- 3. <u>Proposed Subdivision</u> In conjunction with Preliminary Subdivision approval applications the owner/developer of the proposed subdivision shall be required to submit a comprehensive signage plan for review of the CAC and obtain the approval of such signage plans prior to final approval of the subdivision.
- 4. <u>Condition of Approval</u> The approval of the CAC may contain such conditions, requirements or standards to assure that the proposed signage will not be detrimental to persons or property in the vicinity or to the public welfare in general.

10.5.5.3 Criteria for Evaluation

Comprehensive sign plans approved under this Section shall be evaluated based upon the following criteria:

- 1. <u>Placement</u> All signs shall be placed where they are sufficiently visible and readable for their function. Factors to be considered shall include the purpose of the sign, its location relative to traffic movement and access points, site features, structures, and sign orientation relative to viewing distances and viewing angles.
- 2. Quantity The number of signs that may be approved within any development shall be no greater than that required to provide project identification or entry signs, internal circulation and directional information to destinations and development sub-areas, and business identification. Factors to be considered shall include the size of the development, the number of development sub-areas, and the division or integration of sign functions.
- 3. <u>Size</u> All signs shall be no larger than necessary for visibility and readability. Factors to be considered in determining appropriate size shall include topography, volume of traffic, speed of traffic, visibility range, proximity to adjacent uses, amount of sign copy, placement of display (location and height), lettering style and the presence of distractive influences. In no event shall a plan contain a sign which exceeds by more than 25% any maximum area standard contained in this Sign Code.
- 4. <u>Materials</u> Sign materials shall be compatible with architectural and/or natural features of the project. This may be accomplished through similarity of materials for sign structures and faces, the use of complementary colors, similarity of architectural style, or the use of a consistent lettering style or copy.
- 5. Illumination shall be in conformance with Section 10.8 of this Ordinance.
- 10.5.5.4 Requests for use permits under this Section shall be accompanied by an application including, but not limited to:
 - 1. The applicant's name and address;
 - 2. A legal description of the property:
 - 3. Existing zoning on the property:
 - 4. A site plan, depicting the proposed plan of development;
 - 5. A description and/or illustration of proposed sign locations:
 - 6. Standards for size, qualities, materials, and illumination; and

- 7. A narrative description of the common theme for signage within the development, how it relates to architectural and/or landscaping elements of the development, and how the comprehensive sign plan relates to each of the criteria set forth in subsection 10.5.5.3.
- 10.5.5.5 Modifications to a use permit approved under this Section shall require the applicant to follow the same procedure utilized in obtaining the original user permit.
- 10.5.5.6 No sign permit authorized under this Section may vary from the provisions of Sections 10.4, 10.7, 10.8, 10.9 or 10.10 of this Sign Code.
- 10.5.5.7 In no event shall a plan contain a sign which exceeds by more than 25% the maximum height or size standard specified by this Sign Code.

10.6 REGULATION OF OFF-PREMISE SIGNS

10.6.1 Off-Premise Signs Allowable without a Permit

Subject to all provisions and requirements of this Sign Code, the following off-premise signs may be erected and displayed in any zoning district without the necessity of issuance of a permit by the Building Inspector:

- 1. Public Directional Signs;
- 2. Official notices duly issued by any court, public agency or officer.

10.6.2 Off-Premise Signs Allowed by Temporary Permit Only

Subject to all provisions and requirements of this Sign Code, the following off-premise signs may be erected and displayed in any zoning district upon the issuance by the Building Inspector of a temporary sign permit:

- 1. Street Banner Signs
- 2. Political Signs

Off-premise Street banner signs shall be permitted only for such purpose and subject to all restrictions and requirements as are set forth in Section 10.5.2.1 of this Sign Code. Off-premise political signs shall be regulated only in accordance with the restrictions and requirements set forth in Section 10.5.2.3 of this Sign Code. Any permit issued and bond posted pursuant to the provisions of Section 10.5.2.3 of this Sign Code shall authorize the erection of an unlimited number of off-premise political signs.

10.6.3 Regulations of all Off-Premise Signs; Height, Clearance and Setback Requirements

The following regulations shall apply to all off-premise signs:

- 1. <u>Maximum Height</u> Twenty-five (25) feet
- 2. <u>Minimum Clearance between Ground Level and Sign Face</u> Ten (10) feet on signs of one hundred (100) square feet of sign face or greater.
- 3. <u>Minimum Setback</u> Five (5) feet from any property line.
- 10.6.4 Regulation of Off-Premise Signs; Maximum Sign Area
- 10.6.4.1 RESERVED
- 10.6.4.2 RESERVED
- 10.6.4.3 Maximum Sign Area; Regulated by Width of Right-of-Way
 - 1. The maximum sign area of any off-premise sign located on a premise fronting or adjacent to any Street right of way of a width of less than two hundred (200) feet shall be one hundred fifty (150) feet.
 - 2. The maximum sign area of any off-premise sign located on a premise fronting or adjacent to any Street right of way of a width of two hundred (200) feet or more, except for those rights of way defined herein as the Gateway Corridors, shall be three hundred (300) square feet.
- 10.6.5 Regulation of Off-Premise Signs; Permitted and Prohibited Locations
 - 1. Off-premise signs other than those specifically enumerated under Section 10.6.1 and 10.6.2 may be located in B-2, B-4 and M-Industrial zoning district.
 - 2. No off-premise sign shall be located within one thousand (l,000) linear feet, measured along and on the same side of any Street or road frontage, of any other off-premise sign or, at Street intersections, within five hundred (500) feet, measured radially, of any other off-premise sign.
 - 3. Under no circumstances shall more than one off-premise sign be located on any one premise.

- 4. No off-premise sign shall be located within one thousand (1,000) feet of the shore of Lake Pontchartrain, within five hundred (500) feet of the bank of the Bayou Castain or along North Causeway Approach south of its intersection with Monroe Street.
- 5. No off-premise sign shall be located within two hundred (200) feet of any residential zoning district.
- 21. No off-premise signs shall be permitted within the Gateway Overlay District.

10.6.6 Limitation of Number of Off-Premise Sign Faces per Sign Structure

No more than one off-premise sign face visible from any one direction of travel on any roadway may be displayed on any single sign structure. A total of two signs each of which may meet the maximum sign areas permitted for off-premise signs, may be displayed on one back-to-back or V-type sign structure, provided that the angle between such signs be not more than sixty (60) degrees.

10.6.7 Regulation of Public Information Signs

The following regulations shall apply to all public information signs:

1. Signs that designate area maintained for beautification by company

1' x 1 1/2' in size No more than 2' in height

2. Historical markers

2' x 3' in size

No more than 3' in height

3. Approval of location required - No public information signs shall be constructed, erected or displayed until the location of the sign has been submitted to the CAC for its recommendation and approved by a resolution of the City Council.

10.7 DESIGN AND CONSTRUCTION STANDARDS FOR SIGNS

10.7.1 Compliance with Building Code

- 1. No sign shall be constructed, erected, installed, structurally altered, changed or relocated before first securing a permit, except those signs specifically excluded from the requirement of a permit by this Sign Code.
- 2. All new signs shall comply with the structural requirements of the Standard Building Code of the Southern Building Code Congress International, Inc.

(SBCCI) and with the provisions of this Sign Code and any other codes of the City of Mandeville, whichever is more restrictive.

- 10.7.2 Wind Pressure; Design Requirements and Working Stresses
 - 1. <u>Wind Pressure</u> In the design and erection of all signs, the effect of wind shall be carefully considered. All signs shall be constructed to withstand a wind pressure of thirty (30) pounds per square foot.
 - 2. <u>Design Requirements</u> Before any permit required by this Sign Code shall be granted the applicant shall submit to the Building Inspector a design and stress diagram or plans and elevations containing the necessary information to enable the Building Inspector to determine that such sign complies with all the regulations of this code. When necessary to make such a determination, the Building Inspector may require engineering data certified and signed by a Louisiana registered structural engineer.
 - a. <u>Strength of parapet or wall</u> A parapet wall must be designed to have sufficient strength to support any sign which is attached thereto.
 - b. <u>Supports and braces</u> Supports or braces shall be of metal and shall be adequate for wind loadings specified in Section 10.7.2.1. All metal, wire cable supports and braces and all bolts used to attach signs to brackets, or brackets and signs to the supporting building or structure, shall be of galvanized steel or of an equivalent material. All sign supports shall be an integral part of the sign design.
 - c. <u>Sign anchoring</u> Signs shall be anchored to prevent any lateral movement that would cause wear on supporting members or connections.
 - d. <u>Marquee signs</u> Marquee signs shall be constructed entirely of metal or non-combustible material and may be attached to, or hung from a marquee. Any such signs when hung from a marquee shall be at least nine (9) feet at its lowest level above the sidewalk or ground level, and further, such signs shall not extend outside the line of such marquee. Signs painted or sewn onto awnings or canopies, when considered as marquee signs, shall be exempt from the material provisions of this Section.
 - 3. <u>Working Stresses</u> In all signs, the allowable working stresses shall conform with the requirements of the Standard Building Code (SBCCI).

- a. The allowable working stresses for steel and wood shall be in accordance with the provisions of the Standard Building Code (SBCCI).
- b. The working strength of chains, cables, guys or steel rods shall not exceed one-fifth (1/5) of the ultimate strength of such chains, cables, guys or steel rods.

10.7.3 Material Specifications

Permitted signs shall be constructed only of the following materials:

- 1. <u>Incombustible Materials</u> Corrosion resistant metal or other incombustible materials;
- 2. <u>Fiberboard</u> Highly compressed fiberboard which weighs not less than sixty (60) pounds per cubic foot and is not less than one-eighth (1/8) inch in thickness;
- 3. <u>Plywood</u> Exterior grade plywood not less than three-eighths (3/8) inch in thickness and bearing the stamp of an approved testing agency;
- 4. <u>Approved Plastics</u> Of a thickness and shape necessary to withstand the loadings specified in Section 10.7.2 of this Sign Code. Proper allowance or provision shall be made in connections to provide for thermal contraction and expansion. Notwithstanding any other provisions of this code, plastic materials which burn at a rate no faster than two and one-half (2.5) inches per minute when tested in accordance with American Standard of Testing Material D 635 shall be deemed approved plastics and may be used as the display surface material and for the letters, decorations and facings on signs.
- 5. <u>Glass</u> When glass is used for sign letters or transparent or translucent panes, it shall be at least double strength thickness for sign areas up to and including three hundred (300) square inches. When glass is used for sign letters or transparent or translucent panels for sign areas in excess of three hundred (300) square inches at lease one-quarter (0.25) inch wire glass shall be used and maximum span between supports shall be four (4) feet.
- 6. <u>Wood Structure</u> The framework or standards upon which the sign rests may be of wood. Any wooden portion of such structure in contact with the ground shall be either of redwood or any other wood which is a commercially available wood treated with an approved preservative. Sign supports may be

no more than two (2) in number and shall be of sufficient strength and foundation to preclude the need for visible cross-bracing.

7. <u>Metal Structure</u> All signs of one hundred fifty (150) square feet or over shall be of metal construction and shall have no more than two (2) structural supports.

10.7.4 Auxiliary Specifications

- 1. <u>Obstruction to Exits</u> No sign shall be erected so as to obstruct any fire escape, required exit, window, or door opening intended as a means of egress.
- 2. <u>Obstruction to Ventilation</u> No sign shall be erected which interferes with any opening required for ventilation.
- 3. <u>Clearance from Electrical Power Lines and Communication Lines</u> Signs shall maintain all clearances from electrical conductors in accordance with the National Electric Code and all communications equipment or lines located within the City.
- 4. <u>Clearance from Surface and Underground Facilities</u> Signs and their supporting structures shall maintain clearance and non-interference with all surface and underground facilities and conduits for water, sewage, gas, electricity, or communications equipment or lines. Furthermore, placement shall not interfere with natural or artificial drainage of surface or underground water.
- 5. <u>Clearance of Projecting Signs</u> Signs projecting from a building or extending over public property shall maintain a clear height of nine (9) feet above the sidewalk and all such signs shall be at least eighteen (18) inches inside of the curbline as measured toward the building.
- 6. <u>Signs at Intersections</u> Freestanding signs shall not restrict or impair visibility at the intersection of the right-of-way lines of two (2) streets, or of a Street and a railroad.

10.8 ELECTRICAL SIGNS REGULATIONS

10.8.1 Electrical Code

All electrical signs shall be built and installed in compliance with the National Electric Code and the Southern Building Code. All electrical wiring for signs shall be permanently installed and placed underground in metal conduits in accordance with the National Electrical Code.

10.8.2 Licensing

Electrical signs may only be installed by an electrician licensed by the City of Mandeville. 10.8.3 Electrical Sign Permit

The following shall be required prior to the issuance of an electrical permit in conjunction with the erection of an electrical sign:

- 1. Wiring schematic or plan fully describing the electrical work to be done.
- 2. Compliance with U.L. Standards for electrical work to be done.

10.8.4 Electrical Inspection Fees

An electrical inspection shall be required in conjunction with the erection of an electrical sign. This fee in addition to the permit fee required under Section 10.9.9 "Required fees for sign permits." Such fees are hereby set as:

- 1. Twenty Dollars (\$20.00) To cover the cost of one electrical inspection.
- 2. Twenty Dollars (\$20.00) Shall be charged for each reinspection required.

10.9 PROCEDURES AND FEES FOR SIGN PERMITS; SIGN INSPECTIONS AND CERTIFICATES OF COMPLETION

10.9.1 Requirement to Obtain a Sign Permit

It shall be unlawful to construct, erect, repair, alter, relocate or display with the City of Mandeville any sign without first obtaining a sign permit from the Building Inspector and paying the fee required herein, unless specifically excluded from the requirement of a permit by Section 10.5.1 of this Sign Code. All electrical and/or illuminated signs shall, in addition, be subject to the provisions of Section 10.8 of this Sign Code and the inspection fees required therein.

If a sign permit is required for any establishment which will also require a development permit, as per Section 5.4 for renovation, remodeling or new construction, the sign permit shall be applied for in conjunction with the development permit.

10.9.2 Application Requirements for Issuance of Sign Permits

In applying to the Building Inspector for the issuance of a sign permit the following shall be required:

- 1. a completed sign permit application providing all applicable information requested by the Building Inspector;
- 2. written consent of the owner of the property or his agent granting permission for the construction, maintenance and display of the sign or sign structure;
- 3. name, address and telephone number of the premises owner, the sign owner, the sign contractor and any designated contact person;
- 4. a description of the size and location of all existing signs owned, leased or otherwise being used on the same premise by the entity making application for a sign permit; and
- 5. such additional information as may be required by the Building Inspector or the CAC in furtherance of a determination that the provisions of this Ordinance and all other applicable laws and ordinances of the City of Mandeville are being complied with. Such additional information may include, but shall not be limited to a current survey by registered land surveyor of the premises in question, detailed plans and specifications for any proposed sign or structure, or engineering certification of compliance with any technical requirements of this Sign Code, the Building Code or any other codes, ordinances or regulations of the City of Mandeville.

10.9.2.1 Additional Information which may be Required for Issuance of Sign Permits

When such information is needed to determine compliance of the sign or signs in question with the provisions of this Sign Code the following, without limitation, may be required by the Building Inspector or the CAC to be submitted with the application for a sign permit:

- 1. <u>Legal Description and/or Survey of Premise</u> The legal description or survey shall provide sufficient information to determine the allowable premise total, based on linear footage of Street frontage as required by this Sign Code.
- 2. <u>Dimensioned Plan of Premise</u> A required plan of the premise shall be drawn to scale and fully dimensioned indicating the location of all structures, including sign structures, both existing and proposed to be constructed, altered or moved on the premises. The applicant shall note in writing on the plan the existing and intended use of all buildings or structures. The applicant shall also depict on the plan the location and identity of all existing or

proposed utility poles, lines, structures, servitudes and rights-of-way. The applicant shall also depict and identify upon the plan any applicable greenbelts or vegetation protection zones and the location, size and type of all existing trees within said greenbelts or protection zones or located elsewhere on said premises if such tree is proposed to be cut, trimmed or removed in the construction or use of the proposed sign structure or any displays exhibited thereon. Tree size shall be shown both in overall height above the ground and trunk diameter at breast height (dbh).

3. Elevations and Details

- a. <u>Sign elevations and details</u> Required elevations and details, shall be drawn to scale and fully describe the dimensions, structural supports and all pertinent structural details, foundations, materials, method of attachment, conformance with wind pressure requirements and electrical wiring and components of all signs to be constructed, altered or moved sufficient to determine compliance with the provisions of this Sign Code.
- b. <u>Building elevations</u> In the case of an application for a permit for an attached sign, an elevation of the building shall provide the linear footage of the facade upon which the sign or signs are proposed to be placed as well as an accurate depiction of the location and size of the proposed sign(s) and all existing signs on the facade occupied by the applicant.
- 10.9.3 Procedure for Issuance of Sign Permit: Required Review of Application by Community Appearance Commission; Appeals; Display of Proposed Signage pending Community Appearance Commission Review

All applications for sign permits except permits for temporary signs included with the provisions of Section 10.5.2 shall be reviewed and approved by the Community Appearance Commission (CAC) and the Building Inspector prior to the issuance of any permit.

1. Required Review - If the applicant's proposed activity as set forth in his permit application is, upon review by the Building Inspector and the CAC, found to be in conformity with the provisions of the Sign Code and of all other laws ordinances and regulations of this City then in force, the chairman of the CAC shall endorse his approval on said permit application and the Building Inspector shall issue the appropriate permit for the proposed activity. If the proposed activity is found by the CAC or by the Building Inspector to violate any provisions of this Sign Code or of any other laws, ordinances or regulations of the City then in force, the Building Inspector shall so advise the applicant in writing and the request for a permit shall be denied unless plans amended to

conform to the stated codes are submitted by the applicant within thirty (30) days of the issuance by the Building Inspector of written notice of the nonconformity.

- 2. <u>Appeals</u> If the violation or nonconformity so discovered involves a matter within the jurisdictional authority of the Zoning Board, the applicant shall then have thirty (30) days, from denial of the permit, in which to submit an amended application not in violation of said laws, ordinances or regulations or to submit to the Zoning Board an appeal of the decision of the CAC or Building Inspector or a petition for other applicable relief from the provisions of the otherwise offended law, ordinance or regulation. Timely application to the Zoning Board shall stay the denial of the sign permit application for ninety (90) days. The application shall be denied after said ninety (90) days and the requested permit refused if the applicant cannot show that all necessary relief has been granted by the Zoning Board.
- 3. <u>Display of Property Signage</u> After review by the Building Inspector of a sign permit application found to be in conformity with the Sign Code; and when such conformity is so indicated by the initials of both the Building Inspector on the sign permit application; the activity proposed under the application, but specifically excluding the cutting or removal of trees or vegetation, may commence at the risk of the applicant prior to review and approval of the permit application by the CAC. If the CAC, after review of the application, should fail to approve the application, the applicant shall within seventy-two (72) hours of his receipt of notification of the CAC's failure to approve his application dismantle and remove all improvements made or begun under his sign permit application. Actual delivery to the address shown by the applicant in his permit application or physical posting on the site at which an applicant has commenced an activity proposed in his permit application of such notice shall constitute notification to the applicant for purpose of this Section.

10.9.4 Landscaping Requirements in Conjunction with the Issuance of a Sign Permit

If the application involves a freestanding sign, monument sign in the greenbelt or freestanding sign outside of the greenbelt or calls for the cutting or removal of any tree of a height in excess of twenty (20) feet or trunk diameter in excess of six (6) inches (dbh), the Building Inspector shall not approve the application or issue the requested permit until a landscaping plan for the proposed activity is submitted to and approved by the CAC. Such landscaping plan shall consist of a design to transition from the monument sign structure to a decorative ground cover and low planting. In reviewing such a plan, the CAC shall consider such factors as the location, type, number and size of the trees to be removed or cut, any other vegetation which would be damaged or destroyed by the proposed activity, the size

and nature of the proposed activity, the character of the premises on which the activity is proposed and of the area surrounding said premises, the obtrusiveness or non-obtrusiveness of the proposed activity on the surrounding area, and the avoidance of the creation or continuation of more or less denuded areas within view of adjacent properties or public ways. No permit shall in any case be granted on any application or for any activity which would call for the cutting or removal of any live oak tree or which might damage or injure any live oak tree.

10.9.5 Encroachment on Utilities in Conjunction with the Issuance of a Sign Permit

If the applicant's proposed activity as set forth in his permit application is, upon review, found to involve work or construction on, over or under any existing or proposed utility poles, lines, structures, servitudes or rights-of-way. The applicant shall then notify the affected utility or utilities in writing of his proposed activity and advise each such affected utility that any objections to the proposed activity must be submitted in writing to the Building Inspector within thirty (30) days of the receipt of such notice. The Building Inspector shall not issue any permit until such time as the applicant shall show that such notice has been received by each affected utility and the period for comment or objection to the proposed activity shall have expired. If a timely objection is submitted by an affected utility the Building Inspector shall not issue a permit until such time as the interposed objection shall be withdrawn. If the interposed objection shall not have been withdrawn after the lapse of forty-five (45) days of the last day for the timely submission of comments or objections, and if the interposition of said objection is the only impediment to the grant to the applicant of the requested permit, the Building Inspector shall submit the application to the CAC, which shall take appropriate action on the application in accordance with the full scope of its power and authority. If a requested permit under this ordinance is refused the Building Inspector shall so advise the applicant and shall provide the applicant with a written statement of the reason or reasons for such refusal.

10.9.6 Issuance of Permit not a Waiver

The issuance of a sign permit shall in no case be construed as a waiver of any of the provisions of this Article or of any other ordinances or regulations of the City. No permit issued for a sign shall be deemed to constitute permission or authorization to maintain an unlawful sign nor shall any permit issued constitute a defense in an action to abate an unlawful sign.

10.9.7 Commencement of Work under Sign Permit; Automatic Expiration

All permits issued under the terms of Article 10 Sign Codes shall expire automatically if the permitted activity or other work described in the application has not commenced within ninety (90) days from the date of the issuance of the permit and any construction or other

work required under the terms of the application shall not be substantially completed within one hundred twenty (120) days of the date of issuance of the permit. The Building Inspector may, for good cause shown, grant an applicant two (2) extensions, not to exceed a total of ninety (90) days, of such periods. Any period of time in which progress on the completion of any work authorized by the permit is stayed by operation of law shall not be considered in the accrual of the periods of time for commencement and completion of permitted work.

10.9.8 Suspension or Revocation of Sign Permit

The Building Inspector may, in writing, suspend or revoke a sign permit issued on the basis of a misstatement of material fact or fraud.

10.9.9 Required Fees for Sign Permits

- 1. At the time of submission of an application for a sign permit, a non-refundable plan review application fee of ten dollars (\$10.00) shall be paid.
- 2. When application for a permit is approved and prior to the issuance of a permit, a permit fee shall be paid based on the following schedule of rates:

a. <u>Electrical signs</u> :
Up to 32 square feet\$20.00 per face
Each square foot in excess of
32 square feet
b. Non-electrical signs:
Up to 32 square feet\$15.00 per face
Each square foot in excess of
32 square feet
c. <u>Temporary signs</u> :
Up to 32 square feet\$10.00 per face
Over 32 square feet
d. <u>Street banner sign</u> 15.00
e. Re-inspection fee (all signs)
f. <u>Billboards</u> shall pay the following fees in addition to the above fees:
Initial permit fee\$100.00 per face
Annual renewal fee50.00 per face

g.	<u>Murals</u>	. 25.	0()
ĸ.	<u>Murais</u>	23.	1	ハ

- 10.9.10 Regulations for Inspection of Signs and Issuance of Certificate of Completion
 - 1. Upon twenty-four hour advance notice by the permit holder, the following required inspections shall be made by the Building Inspector or his designated agent:
 - a. A foundation inspection prior to pouring concrete for any approved freestanding sign.
 - b. Final electrical inspection for all electrical signs.
 - c. Final inspection for completion of sign in accordance with approved plans
 - 2. No structure or sign, the construction of which necessitates the issuance of a permit under the provisions of this Sign Code, shall be used or displayed until the Building Inspector shall have issued a certificate of completion stating that the construction and proposed display or other activity has been found to be in compliance with the permit issued therefor and with the provisions of this Article. Within three (3) days after notification that a structure or sign constructed under a permit issued under the terms of this Article is ready for use or display it shall be the duty of the Building Inspector to make an inspection thereof and to issue a certificate of completion if the structure and proposed use are found to conform with the plans upon which the permit was issued and to conform with the provisions of this Article. If a requested certificate of completion is refused the Building Inspector shall state in writing the reasons for that refusal and deliver those written reasons to the applicant. No permanent utilities may be permanently connected until a certificate of completion is issued.
 - 3. A record of all certificates of completion, permits, applications sketches and plans shall be maintained in the office of the Building Inspector.

10.10 LICENSING OF SIGN CONTRACTORS

- 1. No person shall engage in any business or activity described in this Sign Code without complying with the terms of this Sign Code.
- 2. Every person commercially engaged in constructing, erecting, installing, maintaining or operating outdoor advertising, advertising structures,

billboards, advertising signs, painted signs on structures, signboards or similar devices, whether as a primary or incidental activity, and whether or not such person is otherwise licensed by the City, shall obtain a sign contractor's license and pay a fee of one hundred fifty dollars (\$150.00) for the first year and fifty dollars (\$50.00) annually thereafter.

- 3. <u>Application and Issuance</u> Applications for licenses shall be made to the City clerk, on forms to be provided by the clerk. If the application is accompanied by the fee provided in this Sign Code and if there is no violation of any state law or City Ordinance in the application, the license shall be issued.
- 4. Public Liability Insurance Required It shall be unlawful for any person to engage in the business of constructing, erecting, installing, maintaining or operating signs within the City, unless and until such person shall have filed with the City a certificate evidencing the existence of public liability and property damage insurance issued to such person by an insurance or bonding company authorized to do business in this state in a sum of not less than three hundred thousand dollars (\$300,000.00) for bodily injury and not less than fifty thousand dollars (\$50,000.00) for damage to property in any one occurrence.

10.11 COMMUNITY APPEARANCE COMMISSION

The Community Appearance Commission as established under the adoption of Ordinance 86-21 in 1987 as described in Article 2 herein shall participate in the Administration of the regulations of Article 10 as provided.

10.12 ENFORCEMENT OF THE PROVISIONS OF THE SIGN CODE AND PENALTY FOR VIOLATION

10.12.1 Maintenance of Signs and Premises

1. <u>Sign Maintenance</u> - Each sign which has been erected in accordance with the provisions of this Sign Code shall be maintained in substantially the same condition as when the final inspection was made and the sign inspection sticker was issued. Failure to maintain the sign, including exterior painting, shall constitute a violation of this chapter. The Building Inspector may after notice to the owner and hearing before the CAC order the removal of any sign that is not maintained in accordance with the provisions of this Section. Such removal shall be at the expense of the owner or lessee.

2. <u>Premises Maintenance</u> - All freestanding signs and the premises surrounding them shall be maintained by the owner thereof in a clean, sanitary, and inoffensive condition, and free and clear of all obnoxious substances, rubbish and weeds.

10.12.2 Regulation of Legally Non-conforming Signs

- 1. <u>Notification of Nonconformity</u> After April 19, 1987, the Building Inspector shall, as soon as practicable, survey the City and identify signs which do not conform to the requirements of this code. Upon determination that a sign is non-conforming, the Building Inspector shall make reasonable efforts to notify either personally or in writing the user or owner of the property on which the sign is located of the non-conforming character of the sign. If the sign owner, user, or owner of the property cannot be located, the notice may be affixed in a conspicuous place to the sign or to the business premises with which the sign is associated.
- 2. <u>Signs Eligible for Characterization as "Legally Non-conforming"</u> Any sign located within the City limits on April 19, 1987, or located in an area annexed to the City thereafter which does not conform with the provisions of this code, is eligible for characterization as a "legally non-conforming" sign, if the sign was in compliance with all applicable laws prior to the effective date of this code or the date that the provisions of this code first became applicable to the sign. Such signs shall be affixed with a legally non-conforming sign identification sticker which shall note the date of the expiration of its non-conforming status in accordance with the provisions of Section 10.12.3 of this Sign Code.
- 3. <u>Loss of Legally Non-conforming Status</u> A legally non-conforming sign shall immediately lose its legally non-conforming designation if:
 - a. The sign is altered in any way, which tends to make the sign less in compliance with the requirements of this code than it was before the alternation; or
 - b. The sign structure is relocated; or
 - c. The sign face (except for copy on a sign permitted as a changeable message sign) is replaced with a sign face which differs in sign message.

On the happening of any one of (a), (b) or (c), the sign shall be immediately brought into compliance with this code and a new permit secured thereof, or shall be removed.

- 4. <u>Maintenance and Repair of Legally Non-conforming Signs</u> Nothing in this Section shall relieve the owners or users of legally non-conforming signs or the owners of the property on which legally non-conforming signs are located from any provisions of this Sign Code regarding safety, maintenance and repair of signs provided, however, that any repainting, cleaning or other normal maintenance or repair of the sign or sign structure does not materially alter or modify the sign.
- 5. No legally non-conforming sign may be enlarged or altered in a way which would increase its nonconformity with the provisions of this Sign Code.
- 6. Should any legally non-conforming sign be damaged by any means to an extent of more than fifty (50%) percent of its replacement cost at time of damage, it shall not be reconstructed except in conformity with the provisions of this Article.
- 7. No conforming sign or sign structure shall be permitted to be erected for the same occupancy with an existing non-conforming sign until the non-conforming sign has been removed or brought into conformance with the provisions of this Article.

10.12.3 Amortization of Legally Non-conforming Signs

- 1. <u>Legally Non-conforming On-Premise Signs</u> All legally non-conforming on-premise signs not otherwise prohibited by the provisions of this Sign Code may be continued until:
 - a. The nature of the business conducted on the premises changes in such a manner as to occasion a change in the existing sign; or
 - b. The name of the business changes or the sign is changed or modified either in shape, size or legend; or
 - c. April 19, 1994 or seven (7) years from the date that this code first becomes applicable to the sign, whichever is sooner, but only if the area of the sign exceeds by greater than fifteen (15) percent the sign area which otherwise would be applicable under the provisions of this code.

2. <u>Legally Non-conforming Off-Premise Signs</u> - All legally non-conforming off-premise signs not otherwise prohibited by the provisions of this Sign Code may be continued until April 19, 1994 or until seven (7) years from the date that the code first becomes applicable to the sign, whichever is sooner.

10.12.4 Permitting of Conforming Signs Erected without Permit prior to this Sign Code

The owner of a sign that is found to be in conformance with the Sign Code but which was erected without a permit may, for ninety (90) days from the effective date of the adoption of this Article, apply for a permit and pay all required fees and, in such case, no further penalty shall be assessed.

10.12.5 Unsafe and Unlawful Signs

- 1. If the Building Inspector shall find that any sign or other advertising structure regulated herein is unsafe or insecure, is a menace to the public, is abandoned is maintained in a dilapidated condition, or has been constructed or erected or is being maintained in violation of the provisions of this Article, he shall give written notice to the permittee or owner thereof or, if he is unable to identify such persons, to the owner of the property on which the sign is located. If the person so notified fails to remove or alter the structure so as to comply with the provisions of this Sign Code, such sign may be removed or altered by the Building Inspector at the expense of the permittee, sign owner, or owner of the property upon which it is located. The Building Inspector shall refuse to issue a permit to any permittee or owner who refuses to pay costs so assessed. The Building Inspector may cause any sign which is an immediate peril to persons or property to be removed summarily without notice.
- 2. In addition to the penalties provided by Section 10.12.9 of these regulations, the provisions of this Section may be enforced and violations thereof may be abated in accordance with the provisions and procedures set forth in Sections 9-44 through 9-48 of Chapter 9 of the Code of Ordinances of the City of Mandeville.

10.12.6 Interpretation and Application

The provisions of this Article are intended to supplement and to be read and applied in pari materia with all existing laws, ordinances and regulations of this City. The provisions of this Article shall not be deemed to have repealed or suspended any such existing law, ordinance or regulation of this City unless such result shall have been expressly stated or be clearly intended by the context and language of the provision in question. In the event of a conflict in any particular circumstances between the provisions or requirements of this Article and

the provisions or requirements of any other law, ordinance or regulation of this City the more restrictive provision or requirement shall apply unless a contrary application thereof is expressly directed or clearly intended by the context and language of the laws, ordinances and regulations in question.

10.12.7 Enforcement and Administration

The provisions of this Article shall be enforced by the police. The provisions of this Article shall be administered by the City Planner or, in the absence of a City Planner, by the Building Inspector. All such officers shall have the power and authority to make inspections of signs, sign structures or premises necessary to carry out their duties in the coordination and the enforcement of the provisions of this Article.

10.12.8 Violation

1. In case any sign structure or sign is erected or structurally altered or maintained or used in violation of the provisions of this Article, any proper City official or his or her duly authorized deputies or representatives may institute any appropriate action or proceedings to prevent such unlawful act or to prevent any illegal act, conduct or use in or about or concerning any such sign, sign structure or premises. Each day any such violation continues shall constitute a separate violation of this Article. The City Planner or Building Inspector may call upon the Chief of Police to furnish necessary personnel to carry out his orders.

2. Any resident of the community who believes that a violation of any of the provisions of this Article is occurring may file a written complaint with the Building Inspector. Such complaint shall fully set forth the acts or omissions constituting the alleged violation and the site or sites at which such violation or violations are alleged to be occurring. The Building Inspector shall record properly such complaint, investigate the allegations underlying said complaint, and take action on such complaint and investigation as provided by this Article.

10.12.9 Penalty

Any person violating any provision of this Article shall be guilty of a misdemeanor, and upon conviction shall be punished as provided in Section 1.9 of these Land Use Regulations of the City of Mandeville.

DIVISION III

SUBDIVISION AND PUBLIC IMPROVEMENTS REGULATIONS

DIVISION III SUBDIVISION AND PUBLIC IMPROVEMENTS REGULATIONS

ARTICLE 11 - GENERAL SUBDIVISION AND PUBLIC IMPROVEMENTS PROVISIONS

11.1 <u>TITLE, PURPOSE AND POLICY</u>

11.1.1 Title

The official title of Division III including Articles 11, 12 and 13 of the Comprehensive Land Use Regulations shall be known as the "Subdivisions and Public Improvements Regulations of the City of Mandeville" and for ease of reference may also be referred to as the "Subdivision Regulations".

11.1.2 Purpose

It is the purpose of these regulations to promote logical and sound development in the municipality and to assist those who are developing subdivisions in the City to safeguard the interests of the property owner, the occupants of homes and other structures, the developers, and the local government. Once land has been divided into streets, lots and blocks, and these actions have been publicly recorded, the correction of defects is costly and difficult. Subdivision of land eventually becomes a public responsibility, in that roads and streets must be maintained, and various public services customary to urban areas must be provided. The welfare of the entire community is, as a result, affected in many important respects. It is also deemed to be in the interest of the general public, the developer and future home owners and other occupants, that subdivisions be conceived, designed and developed in accordance with sound rules and intelligent development standards. These regulations are adopted for the following purposes:

- 1. To protect and provide for the public health, safety and general welfare of the present, as well as future citizens of the City of Mandeville;
- 2. To provide for adequate light, air and privacy, to secure safety from fire, flood and other danger, and to prevent overcrowding of the land and undue congestion of population;
- 3. To protect the identified character and the social and economic stability of the City of Mandeville and to encourage the orderly and beneficial development of the area within its boundaries;

- 4. To protect and conserve the value of the land in the incorporated area of Mandeville and the value of buildings and improvements upon the land and to minimize the conflicts among the uses of land and buildings;
- 5. To guide public and private policy and action in order to provide adequate and efficient transportation, water, sewerage, schools, parks, playgrounds, recreation, and other public requirements and facilities;
- 6. To provide the most beneficial relationship between the uses of land and buildings and the circulation of traffic throughout the City of Mandeville, having particular regard to the avoidance of congestion in the streets and highways, and pedestrian and bicycle traffic movements appropriate to the various uses of land and buildings and to provide for the proper location and widths of streets and building lines;
- 7. To establish reasonable standards of design and procedures for subdivisions, resubdivisions and the installation of public improvements, in order to further the orderly layout and use of land, and to insure proper legal descriptions and monumenting of subdivided land;
- 8. To insure that public facilities are available and will have a sufficient capacity to serve the proposed subdivision;
- 9. To prevent the pollution of air, streams, and ponds; to assure the adequacy of drainage facilities; to safeguard the water table; to encourage the wise use and management of natural resources throughout the community in order to preserve the integrity, stability and beauty of the City of Mandeville and the value of the land;
- 10. To preserve the natural beauty and topography of the City of Mandeville and to insure appropriate development with regard to these natural features in order to provide a natural habitat for the preservation of wildlife and natural vegetation, and to maintain the natural drainageways in a naturalistic state in order to reduce the effects of run-off pollutants on the lake;
- 11. To provide open spaces to meet the needs of the community through the most efficient design and layout of the land, including the use of average density in providing for minimum width and area of lots, while preserving the density of land as established in the Mandeville Zoning Regulations; and
- 12. To guide and manage the future growth and development of the City in accordance with the City's adopted Comprehensive Land Use Plan and any

other plans which in the future may be adopted by the City such as, but not limited to, an Open Space/Drainage Plan.

11.1.3 Policy

- 1. It is hereby reaffirmed to be the policy of the City of Mandeville to consider the subdivision of land, and the subsequent development of the subdivided plat, including the construction of public improvements, as subject to the control of the City of Mandeville pursuant to objectives for the orderly, planned, efficient, economical and environmentally sound development of the City. These general improvement and lot improvement regulations are provided to establish a framework for the Planning Commission to evaluate and regulate the subdivision of land in the City, pursuant to the overall objectives of these Land Use Regulations to promote the orderly, planned, efficient, economical, and environmentally sound development of the City.
- 2. Any developer desiring to subdivide any lot, tract, or parcel of land or to change or rearrange any such lot, tract or parcel of land within the City shall comply with the plat specifications and approval procedures and the standards for the construction of public improvements established in these regulations.
- 3. Land to be subdivided shall be of such character that it can be used safely for building purposes without danger to health or peril from fire, flood, or other menace, and land shall not be subdivided until available public facilities and improvements exist and/or proper provision has been made for drainage, water, sewerage and capital improvements such as schools, parks, recreation facilities, transportation facilities and other improvements as identified in the Comprehensive Land Use Plan.
- 4. The Planning Commission shall not approve the subdivision of land if, following adequate investigation conducted by all public agencies concerned, it has been determined that the site is not suitable for platting and development purposes of the kind proposed. Land that is subject to periodic inundation or deemed to be topographically unsuitable shall not be platted for residential occupancy, or for other uses that may increase flood hazard, endanger health, life or property, or aggravate erosion. (See Article 8 for complete Flood Damage Prevention regulations.) Such land within the plat shall be set aside for conservation and passive land uses that would not be endangered by periodic or occasional inundation. In applying these provisions, land below the flood elevations as established by the State Coastal Management Division of the Department of Natural Resources (currently the

five (5) foot MSL contour) shall be considered <u>prima facie</u> as the minimum area subject to flooding, not to be included in the buildable area of any development plan and land determined by the Corp of Engineers to be subject to Section 10 of the Rivers and Harbors Act or Section 404 of the Clean Water Act may only be included in the buildable area of any development plan subject to the provisions of these Land Use Regulations and issuance of any permits required by the Corps of Engineers or state Coastal Management.

- 5. It is intended that these regulations shall supplement and facilitate the enforcement of the provisions and standards contained in the Building and Zoning Regulations, and the City's Comprehensive Land Use Plan and Capital Improvements Plan.
- 6. Pursuant to the accomplishment of these policy statements, no land shall be subdivided, sold or offered for sale within the incorporated area of the City of Mandeville until the owner or duly authorized agent for the owner of the land shall have obtained approval from the Planning Commission of a Final Subdivision Plat in accordance with the procedures and requirements established herein, and the approved Final Subdivision Plat is filed with the Parish Clerk of Court.

11.2 <u>AUTHORITY OF PLANNING COMMISSION AND JURISDICTION OF REGULATIONS</u>

11.2.1 Authority

In accordance with the provisions of Louisiana R.S. 33:101-119 the City Council of the City of Mandeville adopted an ordinance on May 11, 1954 investing the Mandeville Planning Commission with full authority and jurisdiction to exercise all the powers and duties set forth in the cited statutes. By this Ordinance that authority is reaffirmed in the Mandeville Planning Commission in accordance with the regulations and procedures contained herein for the review and approval or disapproval of subdivision/resubdivision plats and public improvements within the incorporated areas of the City of Mandeville.

By the same authority the Planning Commission exercises the power and authority to review and approve or disapprove the development of subdivisions of land where a plat of some or all of that land has already been recorded in the office of the Parish Clerk of Court, if such subdivisions are entirely or partially undeveloped. Subdivisions shall be considered to be entirely or partially undeveloped if:

- 1. The subdivision has been recorded with the Parish Clerk of Court without a prior approval by the Planning Commission; or
- 2. The subdivision has been approved by the Planning Commission, but more that two (2) years has elapsed without granting a construction permit on the partially or entirely undeveloped land and the zoning regulations for size or density or use for the district in which the subdivision is located have been changed subsequent to the original final subdivision approval; or
- 3. More than one year has elapsed since the original final subdivision approval and all streets and other required improvements on the subdivision plat have not been constructed by the developer according to the requirements in effect at the time of final approval or recording.

11.2.2 Jurisdiction

These subdivision regulations shall apply to all subdivisions of land, as defined herein, located within the incorporated limits of the City of Mandeville. Subject to the exceptions hereinafter provided, any sale or contract to sell or lease immovable property within the City of Mandeville either by lot or block description, by metes and bounds description, or to establish a condominium regime over more than one previously created lot or parcel of land shall constitute a subdivision of land and come within the provision of this subdivision ordinance.

Every subdivision of land within the City of Mandeville shall be depicted on a plat and submitted to the Planning Commission through the staff of the Department of Planning and Development for approval or disapproval.

Subdivision developers who have complied with the procedures outlined in this Ordinance do not have to resubmit plats of immovable property to be conveyed once the subdivision is approved and any public improvements depicted thereon have been accepted for maintenance in accordance with these regulations unless substantial changes in the proposed plat have been made.

When no new streets or utilities are required to be constructed and existing drainage patterns are not to be altered, Planning Commission approval of a subdivision of land will not be required in the following cases:

1. Upon the partition of an estate among the legal heirs or upon the partitioning of property required by a court judgment when the partition does not result in any parcel of land less than five (5) acres and having its smallest side dimension less than four hundred feet (400).

- 2. Land in subdivisions previously legally recorded, except in those cases listed under 11.2.1 above.
- 3. Sales of a lot, tract or parcel of land where the sale does not involve any change in the boundaries of such lot, tract, or parcel provided it can be shown to be a legal lot of record at the time of acquisition.

11.3 DEFINITIONS

The definitions applicable to these subdivision and public improvements regulations are set forth in Article 3 - General Definitions of these Land Use Regulations.

11.4 ADMINISTRATION AND ENFORCEMENT

11.4.1 Responsibilities of City Officials and Personnel

11.4.1.1 Responsibilities of the Planning Commission

The Planning Commission, or its duly authorized representative, is charged with the duty of making investigations and reports on the design and improvements of proposed subdivisions and developments and considering policies for development as enunciated by the Comprehensive Land Use Plan, and as prescribed by law and these regulations in making decisions regarding subdivisions. The Planning Commission may require such redesign of street patterns, lot layout and other such information or plans as the circumstances may warrant to carry out its duties.

The Planning Commission Chairman shall cause sufficient copies of the Subdivision and Public Improvements Regulations to be prepared for distribution to interested parties. Such copies shall be sold by the City. It shall further be the duty of the Planning Commission Chairman or authorized individual acting in the behalf of the Chairman to enforce these regulations and to bring to the attention of the City Attorney and the City Council any violations or lack of compliance herewith. Upon such written notice to the City Attorney and the City Council, the responsibility for the

prosecution of any legal action instituted as a result of such violations or noncompliance with these regulations shall be transferred to the City Attorney.

11.4.1.2 Responsibilities of the Department of Planning and Development

It shall be the responsibility of the Director of the Department of Planning and Development, or designee, to review all applications for completeness and to advise the applicant of any

deficiencies in the required submission documents, to collect all fees required to be submitted in conjunction with applications, to assist the applicant in the subdivision process, to assist and advise the Planning Commission with regard to the provisions of this Comprehensive Land Use Regulations Ordinance, and to record and distribute signed copies of final subdivision plats.

It shall be the responsibility of the secretary of the Department of Planning and Development to advertise all subdivision and public improvements cases in accordance with advertising requirements and see that written notice is sent to the applicant, by certified mail, stating the time, date and place of the public hearings on the request for whatever phase of subdivision approval has been applied for by the applicant. This notice shall be mailed not less than five (5) days before such public hearing for all proposed subdivisions.

It shall be the responsibility of the Planning Department through its staff to see that a printed notice in bold type shall be posted for not less than ten (10) consecutive days prior to the public hearing conducted by the City Planning Commission on signs prepared, furnished and placed by the Department upon the principal and accessible rights-of-way adjoining the area proposed for a subdivision. Said signs shall contain an accurate statement of the proposed changes and the time and place of the public hearing as provided above. The City will not permit the issuance of either a building permit or certificate of occupancy on any parcel or plat of land improperly created in accordance with the requirements of or considered to be in violation of these subdivision and public improvements regulations.

11.4.1.3 Responsibilities of the Director of Public Works

It shall be the responsibility of the Director of Public Works to provide input to the Planning Commission in coordination with the City Engineer regarding existing public improvements and plans for proposed public improvements.

11.4.1.4 Responsibilities of the City Council

The City Council will accept the dedication of streets and public improvements from developers but only after having received the recommendation of the Planning Commission regarding the proposed dedications.

No public funds shall be expended for the maintenance of streets, sewers, or other improvements until those improvements have been accepted by the City Council as public improvements and plat or replat of such land shall have been approved by the Planning Commission and recorded in the Office of the Clerk of Court of St. Tammany Parish.

11.4.1.5 Responsibilities of the City Engineer

Responsibilities for review of all engineering specifications and construction methods to be used by a developer/subdivider are assigned to the City Engineer or the Public Works Director. The City will be responsible for periodic observations of construction within subdivisions, including the scheduling of site visits and submittal of reports to the Planning Commission concerning the adequacy of proposed specifications and materials to be used in the construction of the subdivision development.

Resident field representatives authorized to act on behalf of and by the City shall be authorized to observe and report on all work done and all materials furnished. Such duties may extend to all or any part of the work and to the preparation, fabrication or manufacture of the materials to be used. The field representative shall not be authorized to revoke, alter or waive any requirements of the specifications or plans. He is authorized to call the attention of the developer and/or contractor to any failure of the work or materials to conform to approved specification requirements and advise the owner to suspend that portion of the work involved in such failure until any question at issue can be referred to and decided by the Planning Commission, through the City Engineer, Public Works Director or the City's duly authorized representative.

11.4.2 Classification of Subdivisions and Approvals Required

Whenever any minor or major subdivision of land (as defined in Article 3) is proposed, before any contract is made for the sale of any part thereof, and before any permit for the erection of a structure in such proposed subdivision shall be granted, the property owner, or authorized agent for the owner, shall apply for and secure the following required approvals, as described more fully elsewhere herein:

Minor Subdivision/Resubdivision

- 1. Tentative Approval
- 2. Final Approval

Major Subdivision/Resubdivision

- 1. Tentative Approval
- 2. Preliminary Approval
- 3. Final Approval

11.4.3 Official Submission Dates

For the purpose of these regulations for both major and minor subdivisions, the date of the regular meeting of the Planning Commission at which the public hearing on approval of the final subdivision plat (including any adjourned date thereof) is closed, shall constitute the

official submittal date of the plat at which the statutory period required for formal approval or disapproval of the plat shall commence to run. In accordance with provisions of Louisiana R.S. 33:113, the Planning Commission will have sixty (60) days from the official submission date to approve or disapprove the subdivision plat. If the plat is not disapproved in that period of time, it shall be deemed to have been approved, and a certificate to that effect shall be issued by the Planning Commission on demand. The owner may, however, waive this requirement and consent to an extension of such time period. Such waiver and consent must be in writing and signed by the owner or owner's authorized agent.

11.4.4 Requirement of Application and Other Fees

The Department of Planning shall collect and the applicant shall be required to submit a processing fee in conjunction with the Department's acceptance of each complete application for tentative, preliminary or final subdivision or resubdivision approval to cover the cost of processing and advertising the request. In addition to the processing fees, the applicant shall be required to pay all professional and recordation fees incurred by the City in conjunction with the approval process.

11.4.5 Method of Payment of Fees

Required fees shall be paid in cash, money orders, or checks made payable to the City of Mandeville and all fees shall be receipted on pre-numbered receipts with the original going to the payor, one copy being attached to the application and one copy remaining in the receipt book. All fees incurred by the City in association with an application for subdivision/resubdivision approval or the installation of public improvements which are reimbursable to the City by the applicant shall be required to be paid by the applicant prior to the issuance of any Certificates of Occupancy on the land associated with the application. Current hourly rates for City Engineer or Public Works Director and City attorney are available upon request in the Department of Planning and Development.

11.4.6 Enforcement

The Planning Commission or its designated agent shall inform the City of any violations of the regulations set forth in these Subdivision and Public Improvements Regulations or any subdivision plans or construction not in accordance with state law concerning subdivision, after appropriate consultation and recommendation of the City Engineer or Public Works Director and/or City Attorney.

11.4.6.1 Sale of Land in Unapproved Subdivisions

No person, or agent for such person, owning land within the City of Mandeville shall transfer or sell or agree to sell any lot or parcel of land described by reference to, or by exhibit, or by

any other use of a plat of subdivision, before such plat has received final approval of the Planning Commission and has been duly recorded. The description of such lot parcel or tract described as such by metes and bounds in any contract or instrument of transfer or other document used in the process of selling or transferring same shall not exempt the person attempting to transfer same from penalties provided, or deprive the purchaser of any rights or remedies he may otherwise have. The provisions of this section do not apply to letters of intent given to a developer to enable him to obtain construction financing, provided however, that such letters of intent may not be exercised until final subdivision approval.

11.4.6.2 Revisions of Plat After Approval

No changes, erasures, modifications or revisions shall be made on any plat of a subdivision after final approval has been given by the Planning Commission other than as may be necessary to satisfy any conditions of such approval, unless the plat is first resubmitted and the changes approved by the Planning Commission.

11.4.6.3 Improvements in Official Streets Only

No board, public officer or authority shall accept, lay out, open, improve, grade, pave, curb, or light any street or authorize utilities to be placed in any street within the territory for which the Planning Commission has adopted a major street plan, unless the street right of way has been dedicated, accepted for maintenance or has otherwise received the legal status of a public street prior to the adoption of such plan, or unless the street corresponds with a street shown on the Master Streets Plan or with a street on a subdivision plat approved by the Planning Commission or with a street plat made by and adopted by the Planning Commission and accepted by the City Council.

11.4.6.4 Dedication of Improvements Prior to Certificate of Occupancy

All improvements required or intended to be dedicated to the City, including but not limited to streets, utilities or recreational amenities, shall be required to be dedicated and accepted by city ordinance prior to the issuance of any Certificate of Occupancy for a structure on the land to be served by such public improvement.

11.4.7 Violations

11.4.7.1 Penalties for Unapproved Sales

Any person, whether owner or agent of the owner of any land located in a subdivision who transfers or sells or agrees to sell any land by reference to, or exhibit of, or by other use of a

plat of subdivision, before such plat has been approved by the Planning Commission and recorded in the office of the Clerk of Court of the Parish of St. Tammany shall forfeit and pay a penalty of \$500 to the City for each lot or parcel so transferred or sold or agreed or negotiated to be sold. The description of such lot or parcel by metes and bounds in the instrument of transfer or other document used in the process of selling or transferring shall not exempt the transaction from such penalties or from the remedies herein provided.

11.4.7.2 Civil Enforcement

Appropriate actions and proceedings may be taken by law or in equity to prevent any violation of these regulations, to prevent unlawful construction, to recover damages, to restrain, correct, or abate a violation, to prevent illegal occupancy of a building, structure, or premise, and these remedies shall be in addition to the penalties described above.

11.4.8 Reconsideration

When the Planning Commission has taken adverse final action on any subdivision proposal, no reconsideration of an application will be granted unless:

- 1. Notarized certification is furnished by the owner to the Commission that circumstances and/or conditions have been changed or altered;
- 2. Notarized certification is furnished by the owner to the Commission that data used or provided in analysis of the proposed subdivision was inaccurate; or
- 3. Additional information has become available which was not available at the time the subdivision review was made.

11.5 GENERAL PROVISIONS

11.5.1 Interpretation and Conflicting Provisions

In their interpretation and application, the provisions of these regulations shall be held to be the minimum requirements for the promotion of the public health, safety and general welfare. These regulations are not intended to interfere with, abrogate, or annul any other ordinance, rule or regulation, statute, or other provision of law. Where any provision of these regulations imposes restrictions different from those imposed by any other provision of these regulations or any other ordinance, rule or regulation, or other provisions of law, whichever provisions are more restrictive or impose higher standards shall control.

11.5.2 Private Provisions

These regulations are not intended to abrogate or assume responsibility for enforcement of any easement, covenant or any other private agreement or restriction, provided that the provisions of these other regulations are more restrictive or impose higher standards than required in the Land Use Regulations.

11.5.3 Saving Provisions

These regulations shall not be construed as abating any action now pending under, or by virtue of prior existing subdivision regulations, or as discontinuing, abating, modifying, or altering any penalty accruing or about to accrue, or as waiving any right of the City under any section or provision existing at the time of adoption of these regulations, or as vacating or annulling any rights obtained by any person, firm or corporation, by lawful action of the City except as shall be expressly provided for in these regulations.

11.5.4 Appeals

An appeal from action taken by the Planning Commission may be initiated as follows:

- 1. Any person or entity claiming to be injured or aggrieved by any final action of the Planning Commission may appeal that action to the District Court. Such appeal shall be taken within thirty (30) consecutive calendar days of the date of the meeting at which the action was taken.
- 2. All actions and decisions of the Planning Commission which have not been appealed to the District Court within the thirty (30) consecutive calendar days after the action or decision has been taken shall become final.

11.5.5 Waivers

The Planning Commission shall have the authority to waive the provisions of Division III Subdivision and Public Improvements Regulations when the waiver will not in any manner vary the provisions of Division I or II of this Comprehensive Land Use Regulations Ordinance or official zoning map and

- 1. When purposes of these regulations may be served to a greater extent by an alternative proposal, or when the waiver is of such an inconsequential nature that it will not have any substantial effect on the purpose and intent of these regulations or the authority for which is specifically granted in these regulations; or
- 2. When granting of the waiver will not be detrimental to the public safety, health, or welfare or be injurious to other property; or

3. When a particular hardship or unusual and practical difficulty, but specifically not to include financial hardship, may result from the strict application of these regulations and the conditions upon which the request for waiver is based are unique to the property for which the waiver is sought because of the physical surroundings, shape or topographical conditions of the specific property and are not applicable generally to other property.

11.5.6 Amendment of Regulations

Amendments to Division III Subdivision and Public Improvements Regulations shall be made in accordance with the procedures and requirements for amendments to the Comprehensive Land Use Regulations Ordinance as provided.

11.5.7 Advertisement of Requests

Notice of each application for matters to be decided by the Planning Commission under the provisions of Division III Subdivisions and Public Improvements Regulations and the time and place of the public hearing or hearings thereon shall have been published once a week for three (3) weeks in the City's official journal. At least fifteen (15) days shall elapse between the first publication and the date of the hearing or hearings to which said publication relates.

In addition, a printed notice in bold type shall be posted for not less than ten (10) consecutive days prior to the public hearing.

11.5.8 Vacation of Plats

- 1. <u>Before the Sale of Lots</u> Any plat or any part of any plat may be vacated by the owner of the premises at any time before the sale of any lot therein by a written instrument, to which a copy of such plat shall be attached, declaring the same to be vacated. Such an instrument shall be approved by the Planning Commission in like manner as plats of subdivisions, except in the case where public rights-of-way, public lands, or public improvements have been dedicated to the public through or in conjunction with the recordation of plat in which case Council approval shall be required to vacate the plat.
- 2. <u>After Lots Have Been Sold</u> When lots have been sold, the plat may be vacated in the manner herein stated provided all the owners of lots depicted on such plat join in the execution of such writing.

- 3. <u>Reason for Disapproval By City Council</u> The City Council may reject any such instrument which abridges or destroys any public rights in any of its public uses, improvements, streets or alleys.
- 4. Procedure Such an instrument shall be executed, acknowledged or approved by the Planning Commission and recorded or filed in like manner as plats of subdivisions; and being duly recorded or filed shall operate to destroy the force and effect of the recording of the plat so vacated. To divest all public rights in the streets, alleys and public grounds, and all dedications laid out or described in such plat in addition to the approval of the Planning Commission an ordinance of the City Council is required to authorize the action appropriate to return the public lands to private ownership.

11.5.9 Responsibilities of Developers/Applicants

The developer has the duty of compliance with reasonable conditions set by the Planning Commission for the design, dedication, improvement and restrictive use of land to conform to goals for the physical and economic development of the City and to the safety and general welfare of the future lot owner in the subdivision, as well as the community at large.

Each developer of residential land, developer of multi family, commercial or industrial complexes, or representative of such person, shall confer with the City Planner or Designee before preparing the tentative geometric layout of the sketch plan in order to become thoroughly familiar with the subdivision requirements and with the provisions of the comprehensive land use plan affecting the territory in which the proposed development lies.

The developer shall prepare plats and plans consistent with the design standards; accomplish improvements consistent with the appropriate specifications; and process said plats and plans in accordance with these regulations.

In cases where a proposed development will cause the need for off-site capital improvements, before such expenditures are provided for in the capital budget, the developer will be required to contribute a proportional share of the cost of such improvement as can be attributed to the impact of the proposed development.

1. <u>Cooperation of Developer and/or Contractor</u> - The developer and/or contractor shall have available on the project, two copies of all required plans and specifications. Such individual shall cooperate with the City representative and with other contractors in every way possible. The developer and/or contractor shall have a competent representative acting as his or her agent on the project. The representative shall be capable of reading and thoroughly understanding the plans and specifications and shall receive

and comply with instructions from the contractor. A representative shall be furnished for every project regardless of the amount of work subcontracted.

- 2. <u>Maintenance of Improvements</u> The developer and/or contractor shall be responsible for the maintenance of the improvements installed until such time as the public improvements are dedicated to the City and accepted for maintenance by the City.
- 3. Repair of Damage Any damage done to the new improvements being constructed or any streets or roadways used for access, by construction traffic, or by other means caused by the construction process shall be repaired or the damaged materials replaced by the developer prior to final acceptance of the constructed improvements.
- 4. Final Inspection Upon completion of all the improvements, the developer shall request, in writing, a final inspection by the City Engineer or Public Works Director, or his duly authorized representative, who shall make a final inspection of streets, sidewalks, curbs and gutters, sanitary and storm sewers, water mains and other improvements required in these regulations. Upon completion of the work and before acceptance, the developer shall clean up all ground occupied or affected by him in connection with the work. The entire area shall be kept in a neat and presentable condition during the entire duration of the project and left in a neat and presentable condition satisfactory to the City.

11.5.10 Inspection and Approval of Improvements

Developers will provide the City Engineer, Public Works Director and/or Engineering Consultant with sufficient information to evaluate their engineering specifications in light of the requirements of this Ordinance. The City Engineer, Public Works Director or City Engineering Consultant will review the construction plans and specifications and observe construction on the site to determine if the improvements proposed and/or constructed meet the requirements of these regulations. The City Engineer, Public Works Director and/or City Engineering Consultant will see that written notice is sent to the developer or agent of the developer, and to the City Council whenever any proposed construction or any actual construction up for approval is found to be in violation of these regulations.

Field visits shall be performed by the City representative, City Engineer and/or Department of Public Works personnel together with the developer's consultant engineer and when necessary, the testing laboratory.

Fees shall be paid by the developer to the City of Mandeville to cover the actual costs incurred for all plan review and field visits.

11.5.10.1 Responsibilities of the City Engineer

- 1. The City Engineer and Director of Public Works will be responsible for selecting a list of approved testing laboratories. The City Engineer, Public Works Director and/or City Engineering Consultant shall provide a copy of this list of approved testing laboratories to the contractor, contractor's engineer, developer, developer's engineer and other officials so designated by the City. The testing laboratory shall be hired by the developer and the fee for necessary testing shall be paid directly by the developer to the approved testing laboratory. Testing shall be done in accordance with the plans and specifications as approved by the Planning Commission.
- 2. The City Engineer, Public Works Director and/or City Engineering Consultant shall perform a sequence of site visits and shall record their findings on the report to be sent to the Planning Commission for the following items of work:
 - a. Clearing and grubbing
 - b. Excavations and embankment
 - c. Drainage channels (surface and subsurface)
 - i. Inlets, junctions and manholes
 - ii. Surface and subsurface courses
 - iii. Curbs, medians and sidewalks
 - d. Structures
 - e. Utility installations
- 3. The City Engineer, Public Works Director and/or City Engineering Consultant shall be in charge of intermediate site investigation and approval of the following:
 - a. Base Courses
 - b. Surface Courses
 - c. Structures
- 4. The City Engineer, Public Works Director and/or City Engineering Consultant shall check the visual appearance of curbs and medians, street pavement and sidewalks; notifying the developer's consulting engineer of any poor workmanship.
- 5. The City Engineer, Public Works Director and/or City Engineering Consultant shall conduct the final inspection of all improvements with representatives of the consulting engineer, the contractor, the utility company and the Testing Laboratory.

11.5.10.2 Responsibilities of Developer's Consulting Engineer

- 1. The developer's consulting engineer shall meet with the City Engineer or Public Works Director to confirm any additional plan review specifications.
- 2. The developer's consulting engineer shall provide the contractor with a benchmark and will stake out square corners and points of curvature for the contractor to use to stake out lines and grades to complete the work. Benchmarks indicating location and N.G.V.D. elevation shall be established and provided by the developer and shown on the drawings at locations specified in other sections of these regulations.
- 3. The developer's consulting engineer shall, where applicable, inspect the laying of the sanitary sewer line and storm drainage installations.
- 4. The developer's consulting engineer shall perform the inspection of the same items covered under the responsibilities of the City Engineer or Public Works Director referenced herein and described above.
- 5. The developer's engineer shall request final inspection by the City.

11.5.10.3 Responsibilities of the Testing Laboratory

The Testing Laboratory shall be responsible for the following:

- 1. The testing of the concrete used in the construction of the street pavements, curbs and sidewalks.
- 2. Providing a resident inspector to inspect the base processing.
- 3. Providing a resident inspector to inspect the street pavements including asphalt.
- 4. Seeing that a representative of the Testing Laboratory shall be present at the concrete and/or asphalt plants to inspect preparation of these materials.
- 5. Seeing that a representative of the Testing Laboratory shall attend intermediate site visits when called upon.
- 6. Seeing that a representative of the Testing Laboratory shall attend the final inspection.

- 7. Perform inspection, where applicable of sanitary sewer lines and storm drainage installations if the developer's consulting engineer does not perform this function.
- 8. Furnishing reports to the City Engineer, Public Works Director and/or City Engineering Consultant, the developer's, consulting engineers, and the contractors on Items 1-7.
- 9. Shall inspect all concrete, metal, clay, plastic or other acceptable pipe and stamp such as being approved by the testing laboratory. This is to be done prior to delivery to the job site.

If the above procedures and responsibilities are not complied with, the City has the authority, following sending written notice to the developer, contractor, and the consulting engineer, to bar further work toward completion of the project.

11.5.10.4 Acceptance of Improvements

When construction is completed and is in accordance with the approved plans and specifications and complies with the provisions of these regulations, the developer through his engineer, shall certify that all work has been completed and request final inspection by the City Engineer, Public Works Director and/or City Engineering Consultant. Following this inspection, the City Engineer, Public Works Director and/or City Engineering Consultant shall forward his report and recommendations for approval or disapproval to the City Council. Final approval of construction and acceptance of any dedications to public use and maintenanyce shall come only from the City Council.

ARTICLE 12 - SUBDIVISION AND PUBLIC IMPROVEMENTS APPLICATIONS AND PROCEDURES

12.1 SUBDIVISION AND RESUBDIVISION APPLICATIONS AND PROCEDURES

12.1.1 Pre-application Procedures for All Subdivisions/Resubdivisions

Prior to preparing a conceptual sketch plat for any subdivision and before any construction, clearing or other ground work is undertaken, the applicant shall schedule a pre-application meeting with the City Planner or Designee to discuss the procedure for approval of subdivision plats and the requirements for the development and construction of subdivisions in the City.

The applicant should acquire a copy of the Comprehensive Land Use Regulation Ordinance of the City and any other documents that the City may require so that the applicant, architect, engineer, landscape architect, land surveyor, and/or planner for the applicant can become familiar with the design and construction requirements of the City. After review of the requirements for subdivision the applicant is encouraged to discuss any questions regarding the proposed subdivision with the City Planner or Designee.

The applicant and/or the applicant's representative should discuss any questions they might have with the City Planner or Designee concerning the requirements for the general layout of streets and reservations of land, street improvements, drainage, sewerage, and similar matters. If public improvements are required to be constructed in conjunction with the proposed subdivision, the applicant shall schedule a pre-application meeting with both the City Planner or Designee and the Director of Public Works. It shall be the responsibility of the applicant to determine the status of the proposed application as a major or minor

subdivision dependent on the requirement for public improvements with the assistance of the Director of Public Works.

The City Planner or Designee should be consulted as to the present zoning of the subdivision site and abutting areas; the use, bulk and density requirements provided in Article 11 and the nature of existing public services and facilities available or planned for the subdivision site. The proposed development shall be evaluated for compatibility with the future land use plan, particularly with regard to planned capital improvements such as streets and parks.

The City Planner or Designee shall also advise the developer, where appropriate, to discuss the proposed subdivision with those City, parish, state and federal officials who may eventually approve various aspects of the subdivision plat.

The City Planner or Designee or their designee shall provide the proposed applicant with all the necessary application forms in accordance with the currently adopted Rules of Procedure of the Planning Commission and advise the applicant of deadlines for submission and required fees.

12.1.2 General Procedure

For purposes of these regulations, all proposed subdivisions and resubdivisions of land within the City of Mandeville shall be classified as either minor or major subdivisions/resubdivisions as defined in Article 3 and, based on the applicable classification, shall be required to apply for and obtain the following approvals in accordance with the procedures outlined for those approvals.

Minor Subdivision/Resubdivision

- 1. Tentative Approval of Conceptual Sketch Plat
- 2. Final Approval of Final Subdivision Plat

Major Subdivision/Resubdivision

- 1. Tentative Approval of Conceptual Sketch Plat
- 2. Preliminary Approval of Preliminary Plat and Construction Plans
- 3. Final Approval of Final Subdivision Plat by Planning Commission in conjunction with:
 - a. Acceptance of dedication of public improvements with maintenance bond by City Council; or

b. Posting of performance bond or irrevocable letter of credit for the construction of required public improvements and subsequent acceptance of dedication of improvements with maintenance bond by City Council.

12.1.3 General Content

All subdivision/resubdivision plats and associated construction plans for public improvements, when applicable, shall comply with the following laws, rules and regulations or shall be subject to disapproval:

- 1. All applicable statutory provisions;
- 2. All applicable provisions of these Comprehensive Land Use Regulations and all other applicable laws and codes of the City of Mandeville;
- 3. Any rules of the Health Department and/or appropriate state agencies, with such rules providing minimum standards to be met by all subdivision plats;
- 4. The requirements of the Louisiana Department of Transportation and Development when so noted and if the subdivision or any lot contained therein abuts a state highway or connecting street and;
- 5. All applicable standards and regulations adopted by the City and all boards, commissions, agencies and officials of the City; and
- 6. The current adopted Rules of Procedure of the Planning Commission available in the office of City Planner or Designee.

12.1.4 Deadlines for Submission of Applications

For purposes of these regulations the deadlines for submissions of all subdivision and resubdivision applications shall be in accordance with the Planning Commissions' current adopted Rules of Procedure. A list of deadlines for submission of applications for each year, based on the current Rules of Procedure shall be available in the office of the Department of Planning and Development.

- 12.1.5 Major Subdivision/Resubdivision Applications and Procedures
- 12.1.5.1 Tentative Approval Procedure for Subdivisions and Resubdivisions
- 12.1.5.1.1 Tentative Approval Application Submission Requirements

Subsequent to the required pre-application meeting the following items shall be required to be submitted by the applicant to the City Planner or Designee to initiate the Tentative Approval procedure for both subdivision and resubdivision requests. Applications received after the official deadline, as established by the Rules of Procedure of the Planning Commission, will be held for hearing at the earliest possible meeting in accordance with the established deadlines. The following items shall be submitted and distributed in accordance with the requirements of these regulations and the currently adopted Rules of Procedure of the Planning Commission on forms approved by the Planning Commission:

- 1. Application form for Tentative Approval, completed and signed, and the required number of copies for review;
- 2. Conceptual Sketch plat, drawn in accordance with the provisions of these regulations and the required number of prints for review;
- 3. Site Features Map drawn in accordance with the provisions of these regulations and the required number of prints for review;
- 4. A notarized affidavit of ownership which shall disclose all contiguous holdings of the owner including land in common ownership as defined herein, with an indication of the portion which is proposed to be subdivided, as well as dates the respective holding of land were acquired, together with the book and page of each conveyance to the present owner as recorded in the Parish Clerk of Court's office. This affidavit shall advise the planning commission as to the legal owners of the property, the Persons proposed to acquire the property, the date contract of sale was executed and, if any corporations are involved, include a listing of all directors, offices, and stockholders of each corporation owning more than five (5) percent of any class of stock.
- 5.* Environmental Impact Assessment, completed and signed, in the form required by the Planning Commission, including each phase of a Master Development Plan when applicable, and the required number of copies for review; (Note-This item may be waived by the Planning Commission in the case of a proposed resubdivision in a previously developed block where environmental concerns have already been addressed.)
- 6.* One (1) copy of any existing or proposed covenants on the property.
- 7.* Master Development Plan, with accompanying site features maps for each phase when a major subdivision/resubdivision is proposed to occur in phases, and the required number of prints for review;

- 8. All fees required to be submitted in conjunction with a Tentative Approval procedure.
- 9. A letter, sent via certified mail, return receipt requested, addressed to the St. Tammany Parish School Board notifying the School Board of the size of the new subdivision development and requesting comment on the proposed development from the School Board.
- * These items may not be required in circumstances where the specific case does not involve the issues addressed by these items. A predetermination as to need will be made by the City Planner or Designee at the pre-application conference, subject to final determination by the Planning Commission during case review.

12.1.5.1.2 Review of the Conceptual Sketch Plat

The planning commission will review the conceptual sketch plat and supporting documents for compliance with the requirements of this Ordinance. The planning commission will, where appropriate, ask for written comments or opinions from officials or agencies of the City, adjoining municipalities and the parish, school and special districts and other official bodies as they deem necessary. Such studies and evaluation will take into consideration the requirements of these land use regulations and the best use of the land being subdivided. Particular attention will be given to the arrangement, location and width of streets, their relationship to the general topography of the land, sewerage disposal, drainage, lot sizes and lot design, the future development of adjoining land that is not subdivided, and the provisions of the City's master streets plan and the comprehensive land use plan.

12.1.5.1.3 Public Hearing Required

The Planning Commission shall hold a public hearing on the request for Tentative Approval. The hearing shall be advertised as required in Article 11 herein.

Notice will be mailed to the applicant and posted on the site in accordance with the requirements of Article 11 herein.

12.1.5.1.4 Approval, Conditional Approval or Disapproval - Tentative Application

After reviewing and discussing the Conceptual Sketch Plat, other submitted materials and all other reports submitted by other agencies and/or officials, the planning commission will advise the applicant as to approval, conditional approval, or disapproval of the application and the reasons therefor. Planning commission approval or conditional approval of the conceptual sketch plat shall constitute authorization to the applicant to prepare and submit a preliminary plat (in the case of a major subdivision/resubdivision) or a final subdivision

plat (in the case of a minor subdivision/resubdivision) to the Planning Commission for approval. In the case of a conditional approval requiring amendment of the conceptual sketch plat or site features map, the applicant shall resubmit the required number of copies of the application documents with the amendments required for approval by the Planning Commission prior to submission of the preliminary or final subdivision plat, whichever the case may be. Failure to submit amended documents sufficient to meet the conditional approval requirements of the Planning Commission in the prescribed time period shall constitute a failure to meet the conditions of approval and shall cause the application as disapproved.

12.1.5.1.5 Rejection of application

If any of the items required to be submitted have not been submitted or if the Conceptual Sketch Plat submitted is incomplete or does not conform to the requirements of these regulations or if the application and/or required supplemental information is not in accordance with the provisions of these regulations, the City Planner or Designee has the authority to notify the applicant of rejection of the application. In such cases, the City Planner or Designee shall send written notice to the applicant stating the reasons for the rejection of the application.

12.1.5.1.6 Effective period of Tentative Approval

Approval or conditional approval of the Tentative Approval application and the associated Conceptual Sketch Plat by the planning commission shall be effective for a period of six (6) months from the date of the meeting at which the planning commission gives such approval. Failure to apply for Preliminary Approval within the effective time of the Tentative Approval shall render the approval of the Conceptual Sketch Plate null and void and the applicant shall be required to resubmit a new application and plat for Tentative Approval which shall be subject to all land use regulations of the City effective after the expiration of the Tentative Approval.

12.1.5.2 Preliminary Approval Procedure for Subdivisions and Resubdivisions

12.1.5.2.1 Preliminary Approval Application Submission Requirements

Subsequent to the required Tentative Approval of the proposed subdivision or resubdivision and prior to the expiration of the effective period of the Tentative approval the applicant may submit an application for Preliminary Approval. The following items shall be required to be submitted by the applicant to the City Planner or Designee to initiate the Preliminary Approval procedure for both subdivision and resubdivision requests. Applications received after the official deadline, as established by the Rules of Procedure of the Planning Commission, will be held for hearing at the earliest possible meeting in accordance with the

established deadlines. The following items shall be submitted and distributed in accordance with the requirements of these regulations and the currently adopted Rules of Procedure of the Planning Commission on forms approved by the Planning Commission:

- 1. Application form for Preliminary Approval, completed and signed, and the required number of copies for review;
- 2. A notarized affidavit of ownership if any changes in ownership have occurred since the affidavit submitted with the request for Tentative Approval;
- 3. Preliminary plat, drawn in accordance with the provisions of these regulations and in compliance with the approved or conditionally approved Conceptual Sketch Plat, and the required number of prints for review;
- 4. Copies of Construction Plans and specifications submitted in accordance with all provisions of these regulations as required;
- 5. Copies of the applicant's request for sewer and water system approval by the Parish and/or State Health Office as required;
- 6. Copies of any approval granted to the applicant from state and/or federal agencies for sewer and/or water system installations and drainage construction work as required;
- 7. Copies of any state or federal approvals or permits for construction in coastal or wetlands areas as required;
- 8. Copies of all proposed and existing restrictive covenants;
- 9. Copies of the Storm Drainage Report in accordance with the provisions of these regulations as required;
- 10. All fees required to be submitted in conjunction with a Preliminary Approval procedure, including the lump sum fee required for engineering review of the preliminary application as determined by the City Engineer or Public Works Director during the Tentative Approval procedure

12.1.5.2.2 Review of Preliminary Plat

The Planning Commission shall review the preliminary plat and its supporting documents based on the requirements of these regulations and the commission's Tentative Approval of the Conceptual Sketch Plat, the City Engineer or Public Works Director's report to the commission, the recommendations of the City Planner or Designee, any recommendation,

testimony and exhibits submitted at the public hearing and any other information submitted by the applicant, agencies and officials required or invited to comment and other interested parties.

12.1.5.2.3 Public Hearing Required

The Planning Commission shall hold a public hearing on the request for Preliminary Approval. The hearing shall be advertised and posted and a notice will be mailed to the applicant in accordance with the requirements of Article 11 of these regulations.

12.1.5.2.4 Approval, Conditional Approval or Disapproval of Preliminary Plat

After the planning commission has reviewed the preliminary plat, construction plans, and other supporting documents of the Preliminary Approval application, the recommendations of the City Planner or Designee, the City Engineer or Public Works Director, the Parish Health Officer and other such agencies having jurisdiction, and the testimony and exhibits, if any, submitted at the pubic hearing by any affected governmental unit, special districts, school boards, or private individuals, the planning commission shall approve, conditionally approve, or disapprove the application. Should the planning commission give a conditional approval or a disapproval, the applicant is to be advised by the City Planner or Designee in writing of the reason(s) for such action. In the case of a conditional approval requiring amendment of the preliminary plat or construction plans, the applicant shall resubmit the required number of copies of the documents with the amendments required for approval by the Planning Commission for signing and distribution prior to submission of the Final Subdivision application or proceeding with the preconstruction meeting with the City Engineer or Public Works Director. In the case of approval of the application, or upon receipt of the amended documents meeting the requirements of the conditional approval, the applicant shall submit the required number of copies of the approved preliminary plat and construction plans and specifications to be marked "Preliminary Approval" and signed by the City Engineer or Public Works Director, City Planner or Designee and Chairman of Planning Commission. Signed copies as required shall be retained in the Preliminary Approval case file in the Department of Planning and Development and by the City Engineer or Public Works Director and returned to the applicant. Planning Commission approval of the Preliminary Plat application shall constitute authorization for the applicant to schedule a preconstruction meeting with the City Engineer or Public Works Director to obtain a written "notice to proceed" from the City Engineer or Public Works Director which authorizes the commencement of construction of the subdivision in accordance with the approved plans.

12.1.5.2.5 Preliminary Approval Prerequisite to Construction or Final Approval

Planning Commission approval of the construction plans and specifications for required improvements shall precede the commencement of any construction of improvements or approval of the final subdivision plat. Additionally, the applicant shall either construct the required improvements or submit a performance bond or irrevocable standby letter of credit in a form approved by the City Attorney securing the construction of required improvements and irrevocable dedication of all rights-of-way or sewers depicted on the plat prior to submission of the final subdivision plat for approval. If the applicant chooses to submit a performance bond or irrevocable standby letter of credit in lieu of completing the required improvements prior to final plat approval, he may do so by addressing a letter to the Planning Commission with an estimate of the cost of all required improvements or the portion not yet constructed. The Planning Commission, upon recommendation of the City Engineer or Public Works Director regarding the estimated cost of the improvements to be included in the bond or irrevocable standby letter of credit, will advise the applicant and the City Council of the amount of such bond or letter of credit recommended by the City Engineer or Public Works Director. Final decision regarding the amount of such bond or letter of credit shall be determined by the City Council by ordinance in conjunction with the acceptance of the dedication of rights-of-way and servitudes where the proposed public improvements covered by the bond or letter of credit will be installed.

12.1.5.2.6 Rejection of Application

If any of the items required to be submitted have not been submitted or if the preliminary plat or construction plans and specifications submitted is incomplete or does not conform to the requirements of these regulations or if the application and/or required supplemental information is not in accordance with the provisions of these regulations or the Planning Commission's current adopted Rules of Procedure, the City Planner or Designee has the authority to reject the application. In such cases, the City Planner or Designee shall send written notice to the applicant stating the reasons for the rejection of application.

12.1.5.2.7 Effective period of Preliminary Approval

The approval of a preliminary plat shall be effective for a period of one (1) year from the date of the meeting at which the planning commission gives such approval. Any plan not receiving final approval within the effective time of the preliminary approval shall be null and void and the developer/subdivider shall be required to resubmit a new application and plan for preliminary approval subject to all land use regulations of the City effective after the expiration of the Preliminary Approval.

12.1.5.2.8 Public Dedications and Improvements

Before the planning commission approves a preliminary plat showing a public dedication or reservation of land for some specified use, the planning commission shall first obtain approval of such dedications or reservation from that agency or branch of government controlling the specific use activity; for example,

- 1. Proposed dedication or reservation of school sites will first be approved by the Parish School Board;
- 2. Proposed dedications or reservations of public park sites not identified in the adopted Comprehensive Land Use Plan will be required to be approved by the City Council for the area in which such dedication or reservation is to be made during the Tentative approval process;
- 3. Proposed dedications or reservations for other public facilities such as transportation facilities and utilities not identified in the adopted Comprehensive Land Use Plan must first be approved by resolution of the City Council during the Tentative approval process.

12.1.5.2.9 Vested rights

No vested rights shall accrue to any plat by reason of preliminary approval.

12.1.5.2.10 Effect of Land Use Regulations on Approval of Preliminary Application

Every subdivision plat shall conform to the requirements of the Comprehensive Land Use Regulations at the time of final approval, except any plat which has received preliminary approval in conformance with the Comprehensive Land Use Regulations shall be exempt from any subsequent amendments to the land use regulations enacted during the effective period of the Preliminary Approval rendering the plan non-conforming as to size or density or use, provided that final approval is obtained within the one-year effective period of Preliminary Approval.

12.1.5.2.11 Restriction on Clearing, Grading and Construction

No work toward construction of streets, sewerage systems, drainage canals, or any other improvements in association with a proposed subdivision shall commence until preliminary approval is granted by the Planning Commission and a work permit or notice to proceed is issued by the City Engineer or Public Works Director. Subsequent to preliminary approval, the applicant may apply for a preconstruction conference with the City Engineer or Public Works Director and request a work permit or notice to proceed to do clearing and grubbing on the proposed site in accordance with the provisions of these regulations.

12.1.5.2.12 Provision for Model Homes

For the purpose of allowing the early construction of model homes in a subdivision, the planning commission, at its discretion, may permit a portion of a major subdivision, but involving no more than two (2) lots, to be created in accordance with the procedures for a minor subdivision, provided said portion derives access on existing public streets and provided no future street or other improvement is anticipated where said lots are proposed. The model home development shall be considered as a minor subdivision and shall be subject to the minor subdivision approval process. The subdivision plat for the "minor" portion shall be submitted to the planning commission simultaneously with the preliminary plat for the entire "major" subdivision. Subsequent to preliminary approval of the preliminary subdivision plat, the models may be constructed, subject to such additional requirements as the Planning Commission may require, however, the model homes shall not be issued certificates of occupancy or sold prior to recordation of the Final Subdivision Plat.

12.1.5.3 Final Approval Procedures for Subdivisions and Resubdivisions

12.1.5.3.1 Final Approval Application Submission Requirements

Following preliminary approval of the proposed subdivision application by the Planning Commission, and prior to the expiration of the effective period of the preliminary approval, the following items shall be submitted by the applicant to the City Planner or Designee to initiate the final subdivision approval procedure for both subdivisions and resubdivisions. In the case of Major Subdivisions/Resubdivisions, prior to submission of the final subdivision plat for approval, the applicant shall either complete the construction of required improvements or submit a performance bond or irrevocable standby letter of credit in a form approved by the City Attorney securing that necessary or required improvements will be constructed within one year of the date of the approval of the final plat by the Planning Commission. Applications received after the official deadline, as established by the current Rules of Procedure of the Planning Commission, will be held for hearing at the earliest possible meeting in accordance with the established deadlines. The following items shall be submitted and distributed in accordance with

the requirements of these regulations and the currently adopted Rules of Procedure of the Planning Commission on forms approved by the Planning Commission:

- 1. Application form for Final Approval of the subdivision or resubdivision, completed and signed by the applicant or his authorized agent, and the required number of copies for review.
- 2. A notarized affidavit of ownership if changes have occurred since that affidavit submitted with the request for tentative or preliminary approval.

- 3. Copies of the Final Subdivision Plat drawn in accordance with the provisions of these regulations and in compliance with the approved or conditionally approved Preliminary Plat, in the case of Major Subdivisions or Resubdivisions, or Tentative Plat, in the case of Minor Subdivisions or Resubdivisions, as required for purposes of review and approval.
- 4. Certification by the City Clerk of the approval of any ordinance for zoning or rezoning submitted in conjunction with the proposed subdivision or resubdivision to verify that the zoning is in order prior the Final Approval, if conformance of the proposed subdivision or resubdivision with all land use regulations is dependent upon the zoning or rezoning of the land comprising the proposed subdivision or resubdivision.
- 5. Copies of any required state or federal approvals or permits for construction.
- 6. Copies of all private restrictions, restrictive covenants or trusteeships and their periods of existence as they apply to lots in the subdivision, signed by the owner or the owner's agent. (If such restrictions or covenants are not stated in full on the final plat, references to the recording of such in the Parish Clerk's Office will be stated.)
- 7.* Copies and reproducibles, as required, of the Record Improvement Plans, except when final approval is requested in conjunction with a performance bond or irrevocable standby letter of credit for the construction of public improvements, in which case these item shall be required to be submitted subsequent to completion of the improvements which they depict and prior to the acceptance of the constructed public improvements by ordinance of the Council.
- 8.* One (1) copy of official correspondence indicating approval by the parish health office of the proposed sewer and water system.
- 9.* One (1) copy of official correspondence indicating approval from all state and/or federal agencies with jurisdiction over sewer and/or water system approval, and dredging and drainage construction work.
- 10.* All formal irrevocable offers of dedication to the public of all streets/roads, local governmental uses, utilities, parks, and easements depicted on the plat, in a form approved by the City Attorney's office. In fulfillment of this requirement, the applicant shall transfer with warranty full, unencumbered and merchantable title to all such property interests to be dedicated in proper form for recording, together with a title insurance policy for the City which

shall be approved by the City attorney before signing of the final subdivision plat.

- 11.* Performance bond or irrevocable standby letter of credit, if required, in a form satisfactory to the City Attorney and in an amount established by the City Council upon recommendation of the City Engineer or Public Works Director and City Attorney and including a provision that the principal of the bond shall comply with all terms of the resolution of final subdivision plat approval as determined by the planning commission and including, but not limited to, the performance of all required subdivision and off-site improvements and stating that all improvements and property interests included in the irrevocable offer of dedication shall be dedicated to the City free and clear of all liens and encumbrances on the premises.
- 12.* One (1) copy of the traverse calculations for the subdivision. The minimum traverse calculations required shall include a closed traverse of the subdivision boundaries and a closed traverse along the centerline of each street within the development, as well as closed traverse calculations of all lots in the subdivisions and all publicly or privately reserved parcels.
- 13. All fees required to be submitted in conjunction with a Final Approval procedure.
- 14. All major subdivision data must be submitted on a computer disk of the subdivision plat, including utilities, in an Autocad compatible electronic format.
- * In the case of a Minor Subdivision Approval these items will not be required to be submitted for Final Approval.

12.1.5.3.2 Review of the Final Subdivision Plat

The planning commission shall review the final subdivision plat based on the requirements of these regulations and the commission's approval of the preliminary plat and any recommendations, testimony, and exhibits submitted at the public hearing by any public agency or official and any other interested parties.

The planning commission's review of a final subdivision plat shall not occur until there has been compliance with all requirements of the Subdivisions Applications and Procedures section of this article.

12.1.5.3.3 Public Hearing Required for Final Approval

The planning commission shall hold a public hearing on the request for Final Approval. The hearing shall be advertised and posted and a notice will be mailed to the applicant in accordance with the requirements of Article 11 of these regulations.

12.1.5.3.4 Approval, Conditional Approval or Disapproval of Final Plat

Following the public hearing, the Planning Commission shall approve, conditionally approve, or disapprove the final subdivision plat application and shall set forth in detail any conditions to which the approval is subject, or reasons for disapproval.

In the final approval the Planning Commission shall stipulate the period of time when the performance bond shall be filed or the required improvements installed, whichever is applicable. In no event shall a performance bond or irrevocable standby letter of credit be submitted later than six (6) months from the date of final approval by the Planning Commission, together with all required documents and completion of required procedures.

In no event shall the period of time stipulated by the Planning Commission for completion of required improvements exceed one (1) year from the date of final approval.

Where the Planning Commission has required the installation of improvements prior to final approval, the Planning Commission shall not unreasonably modify the conditions set forth in the preliminary approval.

12.1.5.3.5 Rejection of Application for Final Approval

If any of the items required to be submitted have not been submitted or if the final subdivision plat submitted is incomplete or does not conform to the requirements of these regulations, the preliminary plat as approved, or the application and/or required supplemental information is not in accordance with the provisions of this Ordinance, the Planning Commission may reject the application. In such cases, the City Planner or

Designee shall provide notice to the applicant stating the reasons for the rejection of the application.

12.1.5.3.6 Final Approval Accomplished

A final subdivision plat shall not be approved or signed by the Chairman of the Planning Commission until the conditions for approval of the plat and all other conditions of the action granting final subdivision plat approval have been met.

In the case of a Major Subdivision or Resubdivision involving the construction of public improvements and/or the dedication of land or improvements or other matters requiring Council action as a condition of Preliminary Approval, Final Approval shall not be accomplished and the final plat shall not be signed and recorded until all required actions of the City Council have occurred and all conditions of approval with respect to public dedications of land and improvements are met or assurance of completion and maintenance of such improvements have been provided as set forth in this Article.

The Final Subdivision Plat shall not be signed by the Chairman of the Planning Commission until all fees due the City have been paid by the applicant or have been assured of payment by the methods provided herein.

When all conditions of approval have been met, all required actions, if any, of the City Council have occurred, the required signatures have been attached to the Final Subdivision Plat, and the Final Subdivision Plat duly recorded, the Final Subdivision Plat shall become a final recorded plat, the created lots shall become eligible for sale or use and the Official Base Map and plat maps of the City shall be amended in accordance with approved Final Subdivision Plat.

12.1.5.3.7 Vested Rights

No vested rights shall accrue to any plat by reason of final approval until the actual signing of the plat by all persons required by these regulations. All requirements, conditions, or regulations adopted by the Planning Commission applicable to the subdivision, or all subdivisions generally shall be deemed a condition for any subdivision prior to the time of signing of the final plat by the Chairman of the Planning Commission. Where the Planning Commission has required the installation of improvements prior to signing the final plat, the Planning Commission shall not unreasonably modify the conditions set forth in the final approval.

12.1.5.3.8 Phased Development in Major Subdivisions

Prior to granting final approval of a major subdivision plat, the Planning Commission may permit the subdivision to be divided into two or more phases and may impose such conditions upon the approval of plats for the various phases as it may deem necessary to assure the orderly development of the subdivision. The Planning Commission may require that the performance bond be in such amount as is commensurate with the phase or phases of the subdivision then under review and may defer the posting of bond for the remaining phases of the subdivision. The developer may also only file required irrevocable offers to dedicate streets and public improvements for the phases of the subdivision then under review and defer filing offers of dedication for the remaining phases until such phases, subject to any conditions imposed by the Planning Commission, shall be proposed for final

plat approval. Each phase of a subdivision must contain at least ten percent (10%) of the total number of lots proposed for the total subdivision development.

12.1.5.3.9 Procedure for Recordation of the Final Subdivision/Resubdivision Plat

- 1. <u>Submission of Plat for Signatures</u> Following Final Subdivision Plat approval by the Planning Commission, the applicant shall submit three (3) reproducible copies (one shall be of a substantial quality, such as mylar or vellum, if the Final Subdivision Plat is larger than a size which can be reproduced by standard copy machine) and two (2) prints of the Final Subdivision Plat with original signatures of the owners of the property and surveyor, as appropriate, for the required signatures of approval.
- 2. Required Signatures of Approval It shall be the responsibility of the City Planner or Designee or their designee to see that all required signatures are affixed to the Final Subdivision Plat prior to recordation. The required signatures of approval shall be the signatures of the Mayor, the Public Works Director or the City Engineer, the City Planner or Designee and the Chairman of the Planning Commission.
 - a. When a bond or letter of credit is required in lieu of installation of all public improvements, the Chairman of the Planning Commission shall endorse approval on the plat only after the bond has been approved and the dedication of public lands, rights-of-way and servitudes for the installation of such improvements has been accepted by ordinance if the City Council and all the conditions of the Planning Commission resolution pertaining to the plat have been satisfied.
 - b. When installation of improvements is required prior to final plat approval, the Chairman of the Planning Commission shall endorse approval on the plat after all conditions of the preliminary and final approvals have been satisfied and all improvements satisfactorily completed or assured in the manner provided herein. There shall be written evidence that the required public facilities have been installed in a manner satisfactory to the City Council as shown by a certificate signed by the City Engineer or Public Works Director and the City Attorney that the procedural requirements of this section have been met.
 - c. The Planning Commission Chairman, Mayor of the City, and the Public Works Director or the City Engineer will sign and date the three (3)

reproducible originals and two (2) prints of the subdivision plat for purposes of recordation and reproduction for distribution.

- 3. <u>Responsibility for Recordation</u> It shall be the responsibility of the City Planner or Designee or their designee to file the Final Subdivision Plat, signed as required, with the office of the Clerk of Court of St. Tammany Parish within thirty (30) days of the date of signatures.
- 4. <u>Distribution of Copies of Recorded Final Subdivision Plat</u> It shall be the responsibility of the City Planner or Designee or their designee to cause to be made the required number of copies of the recorded Final Subdivision Plat for distribution to the Planning Commission files, the map files of the City, the City Engineer or Public Works Director's files, the applicant and others as specified in the current adopted Rules of Procedure of the Planning Commission or as otherwise appropriate based on the circumstance of the specific case.

12.1.5.3.10 Amendment of Official Base Map and Comprehensive Plan

1. Amendment of the Official Base Map - Upon the recordation of a Final Subdivision Plat granted by the Planning Commission and approved by City Council ordinance, the City Planner or Designee and the Department of Planning shall be responsible for the entry of any changes to the computerized Official Base Map within ten (10) working days from the date of approval by the City Council, or at such time that the amendment legally takes effect.

Amendments to the Official Base Map by ordinance of the City Council shall include any changes in the boundaries of the City, the names or spellings of names of official streets or water bodies, the boundaries or numbers of squares, the revocation or dedication of streets or other public rights-of-way or the location, shape or size of water bodies sufficient to require the amendment of the Official Base Map.

Amendment of the Comprehensive Plan - Recordation of a Final Subdivision Plat granted approval by the Planning Commission shall constitute an amendment to the Comprehensive Plan and shall authorize the City Planner or Designee to cause to be amended any maps associated with the Comprehensive Plan which are affected by the recorded plat. Such amendments to the Comprehensive Plan shall be made in accordance with the recorded plat and within thirty (30) days of the date of recordation.

12.1.6 Minor Subdivision/Resubdivision Applications and Procedures

Minor subdivisions and resubdivisions as defined in Article 3 shall be approved in accordance with the procedures for Tentative Approval and Final Approval as provided under the requirements for Major Subdivision/Resubdivision Applications and Procedures above.

12.1.7 Resubdivision Approval Procedures

- 1. <u>Procedure for Resubdivision</u> Prior Final Subdivision Approval or the establishment of a legally subdivided lot of record in accordance with a recorded plat of subdivision prior to the City's requirement for subdivision approval shall be a prerequisite to the acceptance of an application for resubdivision.
 - a. Major resubdivisions shall be required to be approved in accordance with the procedures and requirements for Major subdivisions/resubdivisions as outlined above, including the Tentative, Preliminary and Final Approval procedures.
 - b. Minor resubdivisions shall be required to be approved in accordance with the procedures and requirements for Tentative and Final Approval of Major Subdivisions as outlined above. However, in the case of minor resubdivisions which do not require any changes in existing utilities (ie. installations of additional water meters for additional lots) or other public improvements, the Tentative and Final Approval may occur simultaneously and the Final Subdivision Plat, including all information required to be provided on the Conceptual Sketch Plat, may be submitted in conjunction with the Site Features Map in lieu of the Conceptual Sketch Plat. If a Site Features Map for the affected area has already been filed with the City and there have been no significant changes to the site or surrounding areas which affect the site, no Site Features Map will be required. However, any areas of periodic inundation or live oaks required to be preserved will be required to be shown on the submittal plat in this case.

2. Guidelines for Approving Resubdivisions

- a. The size and dimensions of all newly created lots shall be in conformance with the requirements of these land use regulations.
- b. Corner lots may use either street frontage to meet the front footage requirements of these land use regulations.

- c. No lots may be resubdivided or created which front only on an existing dedicated but unimproved street unless the street is improved in accordance with these regulations.
- d. Areas of periodic inundation as defined herein shall not be included in the calculation of required lot area for purposes of resubdivision.
- e. In cases where two or more sub-standard lots of record are in single ownership and no additional adjacent land is in the same ownership as the substandard lots, the Planning Commission may approve a resubdivision of

the

- smaller lots into a larger lot even if the resultant larger lot does not meet the minimum lot width or area requirements of these land use regulations.
- 3. Procedure for Subdivisions Where Future Resubdivision is Indicated Whenever a parcel of land is subdivided and the subdivision plat shows one or more lots containing more than one acre of land and there are indications that such lots will eventually be resubdivided into small building sites, the Planning Commission may require that such parcel of land allow for the future openings of streets and the ultimate extension of adjacent streets. Easements providing for the future opening and extension of such streets may be made a requirement for approval of a plat.

12.2 <u>APPLICATION AND PROCEDURES FOR APPROVAL OF PUBLIC</u> IMPROVEMENTS ONLY

1. When public improvements (such as, but not limited to, streets, sidewalks, sewer lines, sewer lift stations or water lines) are required to be installed in order to provide access or public services to support the proposed use of a parcel of land and when, due to whatever circumstances, the parcel of land is not required to be subdivided in order to accommodate the proposed use, it shall be the responsibility of the owner to install the required public improvements in accordance with construction plans and specifications approved by the Public Works Director or the City Engineer, when called upon by the Public Works Director, and to, subsequent to installation, dedicate the improvements to the City. Construction plans and specifications submitted shall be in accordance with the City's adopted standards for public improvements except in the case of street improvements, in which case the Planning Commission, after due public hearing, may waive the required standards and allow for the construction of streets in accordance with

standards comparable to the surrounding streets in the same vicinity or subdivision.

- 2. Construction Plans and Specifications as described herein shall be submitted to the Director of Public Works for review and approval prior to the commencement of any work by the applicant. The Director of Public Works may call upon the services of the City Engineer when needed to assist in the review of plans and the inspection of work as described in these regulations. When the services of the City Engineer have been called upon by the Director of Public Works, written approval submitted to the Public Works Director by the City Engineer indicating that the construction plans and specifications meet all the requirements of the City's regulations or that applicable regulations have been waived through due process of the Planning Commission, shall be required prior to the issuance of a notice to proceed with construction.
- 3. The City Engineer when called upon by the Director of Public Works, shall perform all the required site visits and inspections in accordance with City regulations for the type of public improvements being constructed. Certification of the adherence of the construction to the required standards and approved plans shall be furnished by the inspector to the City Council prior to acceptance of the public improvements by the Council.
- 4. The dedication of public improvements shall be in accordance with the same procedures and requirements for public improvements dedicated in association with the subdivision of land, including the provision requiring a maintenance bond for public improvements.
- 5. All costs for the design and construction of public improvements, as well as the costs for any professional fees incurred by the City in the review and inspection of such public improvements shall be the responsibility of the applicant requesting to install such improvements.

12.3 <u>ASSURANCE OF COMPLETION, ACCEPTANCE AND MAINTENANCE OF PUBLIC IMPROVEMENTS</u>

12.3.1 Assurance of Completion

12.3.1.1 Completion of Improvements

Before the final plat is signed by the Chairman of the Planning Commission, the applicant shall be required to complete, in accordance with the Planning Commission approvals and

to the satisfaction of the City Engineer or Public Works Director, all streets and improvements, including lot improvements on the individual lots of the subdivision as required in these regulations, specified in the final subdivision plat and as approved by the Planning Commission and to dedicate same to the City free and clear of all liens and encumbrances on the property and public improvements thus dedicated except as provided below.

12.3.1.2 Performance Bond or Irrevocable Letter of Credit

- 1. In those instances where the Planning Commission determines that it is not necessary or not desirable that the applicant complete and dedicate all public improvements prior to the signing of the final subdivision plat, the applicant may post a performance bond or irrevocable standby letter of credit prior to final subdivision approval in an amount fixed by the City Council upon recommendation by the City Engineer or Public Works Director as sufficient to secure the satisfactory construction, installation, inspection and dedication of the uncompleted portions of all required improvements. The performance bond or irrevocable standby letter of credit shall also secure all required lot improvements on the individual lots of the subdivision and in applicable cases, the control of erosion and sedimentation as required in these regulations.
- 2. Such performance bond or irrevocable letter of credit shall comply with all statutory requirements and shall be satisfactory to the City Attorney as to form, sufficiency, and manner of execution as set forth in these regulations. The period within which required improvements must be completed shall be specified by the Planning Commission in the resolution approving the final subdivision plat and shall be the terms of the bond or letter of credit and shall not in any event exceed two (2) years from date of final approval. Such bond or letter of credit shall be approved by the City Council as to amount and surety and conditions satisfactory to the City Council. The Planning Commission may, upon proof of difficulty, recommend to the City Council that an extension of the completion date set forth in such bond or letter of credit be granted, but only for a maximum period of one (1) additional year. The City Council may at any time during the period of such bond or letter of credit accept a substitution of principal or sureties on the bond or letter of credit upon recommendation of the Planning Commission.
- 3. Governmental units to which these bond or letter of credit and contract provisions apply may file in lieu of said contract or bond or letter of credit a certified resolution or ordinance from officers or agencies authorized to act on their behalf, agreeing to comply with the provisions of this Article.

12.3.1.3 Cost of Improvements

All required improvements shall be made by the applicant at his expense without reimbursement by the City Council or any improvement district.

12.3.1.4 Liability Insurance

The applicant shall furnish such insurance as is deemed necessary by the Mayor and City Council which shall indemnify and save harmless the City of Mandeville from any and all liability arising by reason of any conditions which may arise or grow out of the construction or installation of improvements. The insurance shall be of such amount and condition as determined by the Mayor and City Council, but shall in no case be allowed to expire earlier than the date of acceptance for maintenance of the covered improvements by the City of Mandeville. A copy of the insurance policy shall remain at all times with the Clerk of Council.

12.3.1.5 Failure to Complete Improvements

For subdivisions for which no performance bond or irrevocable letter of credit has been posted, if the improvements are not completed within the period specified by the Planning Commission in the resolution approving the plat, the approval shall be deemed to have expired. In those cases where a bond or letter of credit has been posted and required improvements have not been installed within the terms of such performance bond or letter of credit, the City Council may thereupon declare the bond or letter of credit to be in default and commence all necessary collection or enforcement actions.

12.3.2 Inspection, Acceptance and Maintenance of Improvements

12.3.2.1 General Site Investigation Procedure

The City Engineer or Public Works Director shall provide for the observation of required improvements during construction to insure their satisfactory completion. If the City Engineer or Public Works Director finds that any of the required site investigation improvements have not been constructed in accordance with the appropriate construction standards and specifications, the developer shall be responsible for making any changes required to meet the specifications of the City Engineer or Public Works Director. An estimate of the anticipated fees based on the preliminary application review shall be established by the City Engineer or Public Works Director and furnished to the applicant prior to the preconstruction meeting and the commencement of construction or field inspections. These estimated fees are established for the convenience of the developer, who will be responsible for all fees incurred regardless of the fee that is estimated for inspection services. All fees shall be paid in accordance with the provisions herein as described in Application and Other Fees.

12.3.2.2 Certificate of Satisfactory Completion

The City Council will not accept dedication of required improvements, nor release or reduce a performance bond or irrevocable letter of credit, until the City Engineer or Public Works Director has submitted a certificate stating that all required improvements have been satisfactorily completed and until the applicant's engineer or surveyor has certified to the City Engineer or Public Works Director, through submission of detailed record drawings of the survey plat and construction plans of the subdivision indicating locations, dimensions, materials, and other information required by the Planning Commission, City Engineer or Public Works Director, that the layout of the line and grade of all public improvements is in accordance with construction plans for the subdivision and a letter from an attorney indicating that a title insurance policy has been furnished to and approved by the City Attorney and further indicating the improvements have been completed, are ready for dedication to the City of Mandeville and that the title to this land is marketable and that the land is free and clear of any and all liens and encumbrances. Upon such approval and recommendation, and subsequent to the filing of the required maintenance bond, the City Council shall thereafter accept the improvements for dedication in accordance with the established procedure.

12.3.2.3 Reduction of Performance Bond or Irrevocable of Letter of Credit

A performance bond or irrevocable letter of credit shall be reduced upon actual dedication of public improvements and then only to the ratio that the public improvements dedicated bears to the total public improvements for the approved plat. In no event shall a performance bond or letter of credit be reduced below twenty-five percent (25%) of its original principal amount.

12.3.3 Acceptance of Dedication Offers

Acceptance of formal offers of dedication of streets, public areas, servitudes, and parks shall be by ordinance of the City Council. A letter from the applicant's attorney indicating that a title insurance policy has been furnished to and approved by the City Attorney further indicating that the land is free of and clear of any and all liens and encumbrances shall be required to be submitted in conjunction with the ordinance. The approval by the Planning Commission of a subdivision plat shall not be deemed to constitute or imply the acceptance by the City of Mandeville of any street, servitude, or park shown on said plat. The Planning Commission shall require said plat to be endorsed with appropriate notes to this effect.

12.3.4 Maintenance of Improvements

- 1. The applicant shall be required to maintain all improvements on or serving the individual subdivided lots until acceptance of said improvements by the City Council. If there have been any certificates of occupancy issued for improvements on a lot abutting a street not yet dedicated but intended to be dedicated to the City of Mandeville, such as issued for model homes, the City Council may on twelve hours notice effect emergency repairs and charge the cost of same to the developer.
- 2. The applicant shall be required to file a maintenance bond or standby letter of credit with the City Council, prior to dedication of public improvements, in an amount considered adequate by the City Engineer or Public Works Director and in a form satisfactory to the City Attorney, in order to assure the satisfactory condition of the required improvements, including all lot improvements on the individual subdivided lots for a period of two (2) years after the date of their acceptance by the City Council and dedication of same to the City of Mandeville. Absent the adoption by the City Council of a contrary recommendation from the City Engineer or Public Works Director, the amount of the security required by this subsection shall be fifty (50) percent of the cost of constructing the improvements for the first year of the security and twenty-five (25) percent of the cost of constructing the improvements for the second year of the security.

12.3.5 Issuance of Building Permits and Certificates of Occupancy

Where a performance bond or irrevocable letter of credit has been accepted by the City in conjunction with a Final Subdivision approval in lieu of the completion of the required improvements and prior to the acceptance by the City of the dedication of such improvements, the Building Inspector may issue building permits for the construction of onsite improvements provided the owner of the lot signs a statement acknowledging that the owner shall proceed at his own risk due to the fact that no certificate of occupancy for any building in the subdivision shall be issued prior to the completion of the proposed street and utility improvements and the dedication to and acceptance of same by the City of Mandeville.

12.4 <u>SUBDIVISION APPLICATION AND ASSOCIATED FEES</u>

1. <u>Low-Density Residential Subdivision (One and Two Family Homes)</u>

a. Subdivision over 5 acres\$5,000 + \$100 per lot

b. Subdivisions less than 5 acres\$250 per lot

If the construction period specified by the developer in the construction contract is exceeded, the applicant will be charged additional fees for the City's field representatives at a rate of \$20 per hour payable prior to acceptance by the City of streets or utilities for maintenance.

2. <u>High-Density Residential and Non-Residential Subdivisions</u>

b. Non-Residential Subdivisions/Resubdivisions

\$5,000 plus \$1,000 per acres or fraction thereof for every acre over 2.5 acres

20%

3. <u>Fees for Application of High-Density Residential and Non-Residential Subdivisions</u>

The above fees will be collected in the following manner:

Payment with application for tentative approval 20%

Payment with application for preliminary approval 60%

Payment with application for final approval

If the construction period specified by the developer in the construction contract is exceeded, the applicant will be charged additional fees for the City's field representatives at a rate of \$20 per hour payable prior to acceptance by the City of streets or utilities for maintenance.

4. <u>Minor Subdivisions/Resubdivisions (No Utilities)</u>

The owner/developer will pay a onetime fee of \$300 at the time of the application. The owner/developer will be responsible for the cost of, if required by the Planning Commission or City Council, legal/engineer fees when and if it becomes necessary for the City Attorney/Engineer/Public Works Director to participate in this subdivision/resubdivision.

12.4.1 RESERVED

12.4.2 RESERVED

12.4.3 Fees For Approval of Public Improvements Only

Applicants for approval of public improvements, when not in conjunction with a subdivision approval request, shall be required to pay the actual cost of all public improvements including the actual construction costs as well as all costs incurred in the review and approval of all plans and specifications, inspections of all construction and dedication by ordinance to the City or as specifically required by ordinance of the City.

12.5 SPECIFICATIONS FOR DOCUMENTS TO BE SUBMITTED FOR APPROVALS

12.5.1 Information Required on the Conceptual Sketch Plat

The conceptual sketch plat is intended to present the elements of a proposed development without undue expense to the applicant in a manner that can be evaluated by the Planning Commission, the City Planner or Designee and the City Engineer or Public Works Director for content, composition of land use elements and conformity with the Comprehensive Land Use Regulations and Comprehensive Land Use Plan of the City. The conceptual sketch plat, entitled as such, shall be drawn to scale, but the locations of the elements shown may be approximate, and may be drawn in professional free-hand style. The conceptual sketch plat shall contain the following information:

- 1. <u>Subdivision name</u> (existing or proposed) and phase number. The proposed name of the subdivision shall be distinguishable, phonetically and in spelling, from the name of any existing subdivision located in the City except where such subdivision is a phase of a larger development. There shall not be any other designation other than "phase" for developments which are constructed in stages. The Planning Commission shall approve or disapprove the name of the subdivision and phase number.
- 2. <u>Location</u> of site by section number, township, range and council district.
- 3. <u>A vicinity map</u> drawn at a scale of 1"= 2000'. A reproduction of the most recent USGS topographic map of the area at a scale of 1:24000 with the subdivision site "drawn to scale" is recommended;
- 4. <u>North arrow, "bar" scale, and date plat</u> as well as any revision dates for amendments which occur during the review process;
- 5. The proposed number of lots, the mean square footage of lots and minimum square footage of any lot to be found on the plat, the total acreage in the

subdivision, and the approximate area of all tracts or parcels of land proposed to be set aside for street rights-of-way, parks, playgrounds, other facilities (central sewer plants, water towers, schools, etc.) and/or other public or private reservations, all of which shall be labeled according to intended purpose and conditions surrounding these dedications or reservations;

- 6. <u>Features of the Proposed Subdivision/Resubdivision</u> The Conceptual Sketch plat shall show the approximate location and dimensions and/or widths of the following:
 - a. Proposed or existing utility easements and/or servitudes within and abutting the site;
 - b. Proposed and existing lots, tracts, or parcels of land within the site;
 - c. All the names and type of surfacing existing and proposed streets and alleys including those previously platted and dedicated but not built, the general location of pedestrian and/or bicycle paths, and railroad rights-of-way within and abutting the site (concrete surfaces must be specified for all streets in or abutting the site unless waived by the Planning Commission);
 - d. Existing water bodies and drainage ways (lakes, canals, bayous, streams, marshes, etc.) including their floodplains, within and abutting the site;
 - e. The location of public water front access areas (if applicable) and the approximate location of any on-site drainage retention facilities, such as swales or ponds;
 - f. The approximate location of any lots, tracts or parcels of land proposed to be dedicated for parks, playgrounds, other public facilities (central sewer plants, water towers, schools, etc.) and/or other private reservations within the site;
- 7. The name of and type of surface material on all existing roads/streets, alleys, pedestrian, bicycle or other public paths within or abutting the site. Concrete will be specified for proposed streets unless waived by Planning Commission;
- 8. The delineation of any boundaries of incorporated areas, existing zoning and zoning overlay districts, any areas of proposed rezoning classifications and any other legally established districts the depiction of which the City Planner or Designee or Planning Commission may request;

- 9. The type of drainage, water and sewer facilities to be used. Major subdivision/resubdivision applications shall be required to include a conceptual drainage study illustrating upstream and downstream drainage patterns and watershed areas as well as approximate volumes and rates of stormwater runoff existing and proposed for each watershed area (All drainage, water and sewer facilities must conform to specifications approved by City Engineer or Public Works Director);
- 10. All existing structures and uses within and adjacent to the development;
- 11. All proposed land uses within the development, as well as land uses of adjacent property;
- 12. Statement as to the proposed dedication, if any, of streets, roads, easements, etc., and
- 13. Designation of areas in a special flood hazard zone according to the latest Federal Flood Insurance Rate Maps, areas below the five (5) foot contour, areas designated as wetlands by the U. S. Army Corp of Engineers and areas subject to periodic inundation.

12.5.2 Information Required on the Site Features Map

The following information shall be required to be included on the Site Features Map, entitled as such and with the same subdivision name as the accompanying plat, to be drawn at a scale no smaller than 1'' = 1000' and submitted in conjunction with all Tentative Approval applications for Major or Minor Subdivisions or Resubdivisions:

- 1. Contour lines at five foot intervals approximated from USGS Quadrangle maps or a more accurate source acceptable to the Planning Commission such as a City or Drainage District Comprehensive Drainage Study when available, of the area to be subdivided;
- 2. Identification and highlighting of all areas below the five (5) foot Mean Sea Level contour, all areas determined by the Corp of Engineers to be wetlands subject to permitting under Section 404 of the Clean Water Act or any other permitting procedures, and all areas subject to periodic inundation;
- 3. Direction of flow of surface water in existing and proposed drainage ways and canals;

- 4. The water elevations of adjoining lakes or streams at the date of the survey and the approximate high-and-low water elevations of such water bodies. All elevations shall be referenced to the USGS Datum Plane;
- 5. Written and/or graphically shown description of route of surface water runoff to ultimate disposal point;
- 6. Approximate location, size and type of drainage structures and drainage easements, including the identification of any dedication to be made to the City;
- 7. The location of any known or suspected historic, cultural, archaeological or architectural sites of local, state, or national importance;
- 8. The "general" location of each live oak six (6") inches dbh or greater, existing densely wooded areas plus any isolated hardwood trees outside of densely wooded areas which measures ten (10") inches dbh and/or pines measuring thirty inches (30") dbh or greater.
- 9. The location of temporary stakes or other markers which will enable official inspectors to find and appraise features of the Conceptual Sketch Plat in the field;
- 10. Any proposed rerouting of existing canals or natural drainageways or draining and/or filling of any water bodies or wetlands and date of filing and disposition and identification number of any wetland fill permit application submitted to the coastal area management agency or Corp of Engineers; and
- 11. Required floor elevations of structures under FEMA regulations.

12.5.3 Information Required on the Preliminary Plat

The preliminary plat, titled as such, shall be prepared by a professional land surveyor, and shall be certified as required by state law, at a scale of 1'' = 100' or larger. The preliminary plat shall be prepared in pen or by computer and the sheets shall be numbered in sequence if more than one sheet is used. The sheets shall be of a size acceptable for recording by the parish clerk of court, but shall not exceed thirty four by forty eight ($34'' \times 48''$) inches in size. The applicant should note that the plat prepared for the preliminary plat may also be used for the final subdivision plat and therefore should be drawn to be reproducible.

The following items shall be included on the preliminary plat:

- 1. Subdivision name and all other information required to be included on the Conceptual Sketch Plat as listed under Specifications for Conceptual Sketch Plat;
- 2. The exact number of lots, notation of the minimum and maximum square footage lot to be found in the plan, the total acreage in the subdivision, and the location and area of all lots, tracts or parcels of land proposed to be set aside for parks, playgrounds, and other public facilities (central sewer plants, water towers, schools, etc.) and/or other private reservations, all of which shall be labeled according to intended purpose and conditions, if any, surrounding these dedications or reservations;
- 3. Outer boundaries of the subdivision including bearings and dimensions, expressed in feet and decimals of a foot and the bearing and distances to a government section or quarter section corner;
- 4. Specific features of the proposed subdivision/resubdivision. The preliminary plat shall show the exact location and dimensions (expressed in feet and decimals of a foot) of the following:
 - a. All existing lots, tracts, or parcels of land within the site and their designated identification;
 - b. The exact location and rights-of-way, including any required sight flares, of all existing and proposed streets and alleys and the exact location of pedestrian and/or bicycle paths, and railroad rights-of-way within and abutting the site;
 - c. Existing and proposed water bodies and drainage ways (lakes, canals, bayous, streams, marshes, etc.) within and abutting the site, and notation of the capacity of all proposed surface or sub-surface drainage facilities, including retention ponds;
 - d. All existing and/or proposed water front access areas (if applicable) and any lots, tracts or parcels of land to be used for special purposes, such as parks, playgrounds, other public facilities (central sewer plants, water towers, schools, etc.) and/or other public or private reservations within the site;
 - e. The location of all existing and/or proposed fire hydrants, water supply improvements and boundaries of fire districts in the area.

- 5. The name of and type of surface material (existing and proposed) on all existing and proposed streets, alleys, pedestrian, bicycle or other public paths within or abutting the site;
- 6. Statement of dedication for any streets, easements, lots, or other property;
- 7. Designation of all blocks, which shall be consecutively numbered or lettered in alphabetical order, with blocks in numbered phases of a large subdivision bearing the same name to be numbered or lettered consecutively throughout the several phases.
- 8. Designation of all lots, which shall be consecutively numbered within each designated block, and designation of all out lots in alphabetical order, irregardless of whether blocks are in numbered or alphabetical order;
- 9. Typical cross section of streets and drainageways to be constructed in the subdivision;
- 10. Designation of the municipal address of each lot;
- 11. Note stating that building setback requirements shall be in accordance with the requirements of the land use regulations of the City or applicable private provisions, whichever is more restrictive; and
- 12. Required finished floor elevations based on the latest Federal Flood Insurance Rate Maps.

12.5.4 INFORMATION REQUIRED ON SUBDIVISION AND PUBLIC IMPROVEMENTS CONSTRUCTION PLANS AND SPECIFICATIONS

12.5.4.1 Information on All Construction Plans

Construction plans shall be prepared and submitted to the City for approval for all required improvements in conjunction with major subdivision/resubdivision development and for the construction of all public improvements prior to commencement of any site clearing or construction. All plan sheets shall be of the same size as the preliminary plat, when in conjunction with subdivision, and shall be no greater in size than 34" by 48", and numbered in sequence, if more than one sheet. All construction plans shall include the following:

1. <u>Title of Project or Subdivision</u> - name, address, signature and seal of the professional engineer, landscape architect, and/or surveyor submitting the plans;

and the date plans were drawn as well as any revision dates;

- 2. Existing and proposed utilities on and abutting the subdivision including location, site, grade and invert elevation of sanitary sewers and storm drains; location and size of water mains, including location of fire hydrants and valves; location of gas lines, location of underground electrical conduits; location of telephone lines; location of cable television lines; and any other utility lines;
- 3. Space for signature of owner and date of signature; and
- 4. Space for signatures of City Engineer or Public Works Director and Planning Commission Chairman (when in conjunction with subdivision proposals) and date of signatures.

12.5.4.2 Scope of Construction Plans

In general the following shall be required to be submitted, as appropriate to the specific elements of the preliminary plat, during the preliminary approval procedure (in conjunction with Major subdivision/resubdivision approvals); and when appropriate to specific public improvements proposed to be constructed for development purposes other than subdivision or resubdivision developments. The following plans, specifications and references shall be required to be submitted in accordance with the provisions of Article 13 and all other provisions of these land use regulations, all other applicable City, state or federal regulations and acceptable standards of practice in harmony with the policies of the City as stated in the Comprehensive Land Use Plan and as required by the City Engineer or Public Works Director:

- Drainage and Grading Plans and Specifications and Storm Drainage Report;
- 2. Street Alignment Plans, Profiles and Specifications
- 3. Street Construction Plans and Specifications
- 4. Sanitary Sewer Plans and Specifications
- 5. Water Facilities Plans and Specifications

12.5.4.3 Street Alignment Plans and Profiles Content

In general Street Alignment Plans, Profiles and Specifications shall include profiles showing existing and proposed elevations along the center lines of all roads/streets in the

development. Where a proposed road/street intersects an existing road or street, the elevations along the centerline of the existing road or street within one hundred feet (100) of the intersection shall be shown. Radii of all curves, lengths of tangents and central angles on all streets/roads will also be shown.

All street profile drawings shall be drawn at a scale of:

1 inch = 20 feet Horizontal 1 inch = 2 feet Vertical

All plans shall be at a scale of:

1 inch = 100 feet or 1 inch = 50 feet

Any deviation shall be subject to the prior approval of the City Engineer or Public Works Director.

12.5.4.4 Other Construction Plans Content

In general construction plans and details for streets, drainage, sanitary sewer, water facilities and other utilities shall include the following:

- 1. Street rights-of-way plans and profiles showing the proposed locations and typical cross sections of
 - a. Street pavements including curbs and gutters, sidewalks and bike paths;
 - b. Street lighting standards and street signs;
 - c. The location of proposed street trees or existing street trees proposed to be preserved, including existing live oak trees six (6) inches in diameter or greater measured four (4) feet above the ground (dbh) and other trees twenty-four (24) inches or greater in diameter or greater measured four (4) feet above the ground (dbh).
- 2. Plan and profile sheets showing all existing and proposed drainage and other utility easements and/or rights-of-way; manholes and catch basins; the locations, size and invert elevation of existing and proposed sanitary sewers, storm water drains and fire hydrants, showing connections to any existing or proposed utility systems; and the exact location and size of all water, gas or other underground utilities or structures; surface water elevations of adjoining lakes or streams and, if adjacent to a lake, river bayou or stream, the distances and bearings of the required meander line and five foot contour line.

- 3. Elevations shall be tied to a Bench Mark (USGS Bench Mark) and shall be shown on profiles.
- 4. One soil boring five (5) feet in depth for every 300 linear feet of street with a minimum of two (2) borings for each project when the project is less than 500 feet in length. The boring log and soil classification shall be shown on the plans together with appropriate Atterburg limits, and referenced to the location where they were made. Classification shall be in accordance with the A.A.S.H.T.O. Soils Classification System.

12.5.5 Information Required on Final Subdivision Plat for Major Subdivisions/Resubdivisions

The final subdivision plat, entitled as such, shall be submitted in ink or by computer plotter at the same scale as, and shall contain the same information as the approved preliminary plat. All revision dates must be shown and noted for content. In addition to all of the information shown on the approved Preliminary Plat the following shall also be shown on the final subdivision plat:

- 1. Notation of any self-imposed restrictions and locations of any building lines to be established by such restrictions, if required or approved by the planning commission in accordance with these regulations.
- 2. Notation of all monuments, erected or found with a designation as such, as provided in Article 13 of these regulations.
- 3. Engineer's, landscape architect's and surveyor's certification and seal:
 - a. When the final subdivision plat shows engineering design data, the final plat shall bear the seal of the registered professional engineer, duly licensed to practice engineering in the state, who prepared the design data or under whose direct supervision the engineering design data was developed along with the registered professional engineer's signature and the date of execution.
 - b. The final subdivision plat shall also include the seal and certification of a professional land surveyor. This certification shall be a statement by a professional land surveyor duly licensed to practice land surveying in the state, which states that the plan is based upon an actual survey made by said individual and that the distances, courses, all angles and all other required survey information are shown correctly, that monuments have been set and

the lot and block corners are staked correctly on the ground and that the professional land surveyor has fully complied with the provisions of Louisiana revised statutes 33:5051, et. seq and all City regulations governing platting.

- c. When the final subdivision plat includes a landscape plan, it shall bear the seal and certification of a landscape architect who is licensed by the state to practice the profession of landscape architecture.
- 4. <u>Certification, acknowledgments, and descriptions</u> The following certificates, acknowledgements and descriptions shall appear on the title sheet of the final plat. Such certificates may be combined when appropriate. Representative certificates, acknowledgements, approvals and descriptions that shall be used on the final plat shall be available in the office of the Department of Planning and Development and in accordance with those approved by the Planning Commission in the current adopted Rules of Procedure.
 - a. <u>Certificate by Parties Holding Title</u> A notarized certificate shall be signed and acknowledged by all parties having a title interest in the land subdivided, consenting to the preparation and recording of said plat, providing, however, that the signature of parties owning the following types of interest may be omitted if their names and nature of their interest are set forth on the plat:
 - i. Rights-of-way, easements or other interest that cannot ripen into a fee.
 - ii. Rights-of-way, easements or revisions that appear to be no longer of potential use or value, due to changed conditions, long disuse of laches.
 - iii. Any subdivision including land originally patented by the United States or Louisiana under patent, reserving interest to either or both of these entities, may be recorded under the provisions of this regulation without the consent of the United States or Louisiana thereto, or to dedication made thereon.
 - b. <u>Dedication Certificates</u> A notarized certificate shall be signed and acknowledged offering for dedication all servitude interests and parcels of land shown on the final plat and intended for public dedication, except those parcels other than streets which are intended for the exclusive use of the lot owners in the subdivision or development, their licensees, visitors, tenants and servants. This certification shall also show that there are no unpaid taxes or special assessments against the land contained in the plat.

- c. <u>Certificate and Guarantee of Clear Title</u> The final plat shall be accompanied by a statement prepared by a duly authorized title insurance company stating that when duly signed, the plat will contain the signatures of all persons whose consent is necessary to the preparation and recording of said plat and to the dedication of the streets and other public places that are clearly shown on the plat.
- d. <u>Certification of Planning Commission</u> Certification of the City Engineer or Public Works Director that required improvements have been satisfactorily installed or from the City Attorney that adequate financing guarantees have been provided.
- e. <u>Legal Description of Property</u> The legal description shall be an accurate reflection of the boundary survey. Each reference in such description to any tract, development, or subdivision shall show a complete reference to records of

the Parish of St. Tammany. The description shall also include reference to any vacated area with the book and page number of the instrument of vacation.

- f. Other affidavits, etc. The title shall contain such other affidavits, certificates, acknowledgements, endorsements, and notarial seals as are required by law and by these regulations.
- g. Places for signatures of approval and date of signing by each of the following:
 - (1) City Engineer or Public Works Director
 - (2) City Planner or Designee
 - (3) Chairman of the Planning Commission; and
 - (3) Mayor of the City of Mandeville
- h. Place for signature of acceptance for recordation by the Clerk of Court, for Map File number and for date of recordation.
- 5. All existing uses that are non-conforming uses according to the zoning district in which the development is located.
- 6. Protective covenants or deed restrictions, if any, shall be recorded and referenced by instrument number in their entirety on the subdivision plat.
- 12.5.6 Additional Information Required for Resubdivision Plats

In addition to all of the items required to be included on Conceptual Sketch Plats, Preliminary Plats and Final Subdivision Plats for subdivisions and resubdivisions as listed above, plats for resubdivision shall also include the following information:

- 1. A plat title which states the existing name of the subdivision or tract and lot designation and the proposed new lot designations and subdivision name;
- 2. All dimensions and bearings of proposed lots and monuments at all proposed lot corners noted as placed or found;
- 3. Written legal description, using metes and bounds, of each proposed lot;
- 4. Existing lot lines and lot designations shown in dash lines;
- 5. Proposed lot lines in solid lines and new lot designations for proposed lots which do not duplicate any previously created lot designation on the same block or in the same subdivision but which are in keeping with the other existing lot designations in the areas and, when possible, utilizing the existing lot designation in an altered form (ie. lots l and 2 may be redesignated as lots l-A and 2-A);
- 6. All adjoining street names and right-of-way widths;
- 7. Note as to type and size of existing sanitary sewer and water lines available to serve the proposed lots shall be shown and noted;
- 8. All existing structures, servitudes, drainage ways, ditches, canals, bayous and their floodplains, servitudes or easements and any areas of periodic inundation as well as

any additional areas of wetland determined by the Corps of Engineers, or state or local coastal management authority shall be so designated on the plat by the surveyor;

9. Any servitudes, rights-of-way or parcels of land proposed to be dedicated to the City and a statement of dedication with a place for the owner's signature and date as well as any other legal requirements of the City Attorney.

12.5.7 Basis for Rejection of Documents Submitted

The Planning Commission or the Commission's designated agent may reject any construction plans or plats that do not meet the following standards:

- 1. All prints or copies shall be clearly legible;
- 2. Sheet size of subdivision plat or construction plans does not exceed 38" X 48" and, if more than two sheets are required, an index sheet of the same dimensions shall be provided showing the entire subdivision on one sheet; and
- 3. The plans and plats shall be drawn at the following scale:

All Subdivision/Resubdivision Plats - 1" = 100' or less

All Construction Plans - 1" = 100' or less

All Site Features Maps - 1" = 1000' or less

All Master Development Plans - 1" = 200' or less

ARTICLE 13 - SUBDIVISION AND PUBLIC IMPROVEMENTS STANDARDS

DEVELOPMENT

13.1 GENERAL PROVISIONS

The development standards provided herein shall be considered to be the minimum requirements for the installation of improvements in association with the subdivision or resubdivision of land and the installation of all public improvements in the City of

Mandeville. In addition all improvements shall comply with the following laws, rules and regulations or shall be subject to disapproval:

- 1. All applicable statutory provisions;
- 2. All applicable provisions of these Comprehensive Land Use Regulations and all other applicable laws and codes of the City of Mandeville;
- 3. Any rules of the Health Department and/or appropriate state agencies, with such rules providing minimum standards to be met by all subdivision plats;
- 4. The requirements of the Louisiana Department of Transportation and Development when so noted and if the subdivision or any lot contained therein abuts a state highway or connecting street;
- 5. All applicable standards and regulations adopted by the City; and
- 6. The current adopted Rules of Procedure of the Planning Commission available in the office of the City Planner.

13.1.1 Monuments

1. Vertical Control Monuments of Record

<u>Permanent Vertical Control Monuments</u> - The subdivider or developer shall establish or confirm the prior establishment of a minimum of:

- a. One permanent vertical control monument in each subdivision of twenty lots or less, and
- b. One additional permanent control monument in each subdivision of over ten lots for every additional twenty (20) lots or fraction thereof.

Such permanent monuments shall be located along street rights-of-way for easy access and shall be concrete, not less than thirty-six (36) inches in length, by four (4) inches square or five (5) inches in diameter and marked on top with a brass plug securely imbedded in the concrete. The record elevation of the monument shall be at the top of the brass plug. The exact location and the elevation of each monument shall be clearly shown on the final plat of the subdivision. The elevation shall be in accordance with the national geodetic vertical datum (USCGS).

2. Placement of Monuments

The applicant shall place permanent reference monuments in the subdivision as required herein, as required by state law and as approved by a professional land surveyor.

- a. The external boundaries of a subdivision shall be monumented in the field by monuments of stone or concrete not less than thirty six (36) inches in length, by four (4) inches square or five (5) inches in diameter, and marked on top with a brass plug. The external boundaries shall be:
- i. not more than 1400 feet apart in any straight line;
- ii. at all corners.

b. Internal monuments shall be located on street right-of-way lines, at street intersections, angle points of curve and block corners. Internal monuments shall be at least thirty-six (36) inches long and marked by iron rods 1/2" in diameter or pipes 2" in diameter. These monuments shall be spaced so as to be within sight of each other, the sight lines being contained wholly within the street right-of-way. Additional placement guidelines are as follows:

- i. at each end of all curves;
- ii. at the point where a curve changes its radius;
- iii. at all angle points in any line; and
- iv. at all angle points along the meander line of a navigable river, bayou or stream. These points shall be located twenty-five (25) feet from the mean high water line as determined by the developer's engineer for each subdivision. If such points fall within a street, or proposed future street right-of-way, the monuments shall be located on the right-of-way line closest to the actual point.
- c. All internal boundaries and those corners and points not referred to in the preceding paragraph shall be monumented in the field by like monuments as described above. These monuments shall be placed at:
- i. all block corners:
- ii. each end of all curves:

- iii. at the point where a stream changes its radius; and
- iv. at all angle points in any line.

In such cases where the placement of a required monument at its proper location is impractical it shall be permissible to set a reference monument close to that point. The location of internal monuments shall be noted on the final subdivision plat.

- d. The lines of lots that extend to rivers or streams shall be monumented in the field by iron pipes at least thirty-six (36) inches long and seven-eights (7/8) inch in diameter or by round or square iron bars at least thirty six (36) inches long. These monuments shall be placed at the point of intersection of the river or stream lot line, with a meander line established twenty-five (25) feet landward of the mean high water line of the river, bayou or stream, as determined by the developer's engineer for each subdivision.
- e. All such monuments shall be set flush with the ground.
- f. All monuments shall be properly set in the ground and certified as to accuracy by a professional land surveyor prior to the time the Planning Commission makes a recommendation for final subdivision approval.

13.1.2 Alteration of the Topographic Character of the Land

Subdivisions shall be planned to take advantage of the topography of the land, to economize in the construction of drainage facilities, to reduce hazardous conditions, to minimize destruction of trees and topsoil and to preserve such natural features as watercourses, large trees, sites of historical significance, existing or needed recreational areas and other assets which, if preserved, will add attractiveness and value to the subdivision and the City.

13.1.3 Improvements in Excess of the Requirements of these Regulations

1. To the fullest extent permitted by law, the Planning Commission may require, as a condition of preliminary or final subdivision approval, that the applicant provide improvements in excess of what is proposed by the applicant or in excess of what would otherwise be required under the provisions of these regulations. These improvements shall be upon the recommendation of the City Engineer or Public Works Director and would be deemed necessary to ameliorate detrimental impacts to the community attributable to the development of the proposed subdivision.

- 2. a. The City Engineer or Public Works Director may also recommend to the Planning Commission that any improvements which under the provisions of these regulations are to be dedicated to public ownership or public use exceed what is being proposed by the applicant or exceed the otherwise applicable provisions of these regulations in order to secure a benefit to the community or address a community need that is not attributable to the development of the proposed subdivision. Should the applicant not agree to comply with this recommendation, the recommendation shall be referred to the City Council which shall determine whether to accept the recommendation. The City Council may, in its acceptance, modify the recommendation. The Planning Commission shall not grant any further approval to the proposed subdivision until the City Council shall have decided whether to accept the recommendation. The Clerk of the City Council shall notify the secretary of the Planning Commission of any action taken by the City Council or the recommendation. In the absence of receipt of such notice within one hundred twenty (120) days of the recommendation being referred to the City Council, it shall be deemed that the City Council has decided not to accept the recommendation. The Planning Commission shall extend this period for an additional period of time not to exceed sixty days upon the written request of the Mayor or of the presiding officer of the City Council.
 - b. Should the City Council not accept the recommendation, the Planning Commission shall not require as a condition of subdivision approval that those improvements exceed the greater of what is proposed by the applicant or applicable provisions of these regulations, except as provided in Paragraph 1 of this section.
 - c. Should the City Council accept the recommendation, those improvements shall be deleted from the subdivision proposal and be constructed by the City in accordance with law. The City Engineer or Public Works Director shall provide to the City Council an opinion of probable construction cost for constructing the improvements in accordance with the decision of the City Council and for the cost that the applicant would probably have incurred in constructing those improvements in accordance with the otherwise applicable provisions of these regulations. The City Council shall determine from this information, and from any other information provided to it on this cause, what reasonably would have been the applicant's cost to construct the improvements. The Clerk of the City Council shall report this determination to the applicant and to the secretary of the Planning Commission. At its next regular business meeting following the receipt of notice of the City Council's decision to accept the recommendation, the Planning Commission shall direct

the applicant to delete all of the improvements to be constructed by the City from the subdivision application.

- d. The Planning Commission shall not grant final approval to the proposed subdivision until the City shall have completed construction of the improvement or improvements encompassed by the recommendation accepted by the City Council, unless the City Council should, by resolution of not less than a majority of its members, authorize the Planning Commission to grant an earlier approval to the subdivision.
- e. The City shall be under no obligation to proceed with construction of any of the improvements until the applicant shall have paid to the City the sums previously determined by the City Council to be the cost that would otherwise have been incurred by the applicant to construct the improvements or improvements in question and the applicant has effected the dedication to the City of all rights as are necessary in the opinion of the City Council to complete the construction of the improvements. All such sums received by the City shall be used only for the construction of the improvements in question. Should the City fail to let a contract for the construction of those improvements within twelve months of receipt of such sums and all necessary dedications, the City shall refund such sums to the applicant, or to his heirs, successors or assigns. The Planning Commission shall thereafter review the subdivision application as though the City Council had decided not to accept the recommendation referred to it. The City Council shall revoke and abandon any portion of the dedications previously acquired by the City that the Council determined from information submitted to it is not reasonably necessary to construct and maintain the improvements to be constructed by the applicant.

13.1.4 Off-site Extensions of Utilities

When utilities are not available at the boundary of a proposed subdivision the Planning Commission, or its duly authorized representative, shall require as a prerequisite to approval of a preliminary and final plat assurances that such improvement extensions shall be provided as follows:

1. Extensions of utilities onto the property involved shall be adequate to serve the total development requirements of the service or drainage area. Utilities leaving the property shall be constructed in such a manner as to make their extension practical for servicing the adjacent areas of the service or drainage area.

2. If the Planning Commission, or its duly authorized representative, finds that the extensions across undeveloped areas would not be warranted as a special assessment to the intervening properties or as a governmental expense until some future time, the developer shall be required, if he wishes to proceed with the development, to obtain necessary easements or rights-of-way for dedication to the City and construct and pay for such extensions. Such improvements shall be available for connections by subdividers of adjoining land and the subdivider may contract with adjacent property owners and/or subdividers of adjacent land for reimbursement of the oversize and/or off-site improvements constructed.

13.1.5 Minimum Servitudes Required and Minimum Size Allowed

The minimum servitude no matter what its purpose shall be fifteen (15) feet in width and, when placed at an interior property line, the width when practical shall be equally divided between the two adjacent lots, on each side of the interior property line. Actual size of drainage servitudes shall be predicated on the width of the drainage structure and approved by the City Engineer or Public Works Director.

13.1.6 Size of Drainage Servitude Determines Dedication or Easement Status

- 1. Where the right-of-way needed to accommodate the drainage improvements required by these regulations exceeds twenty-five (25) feet in width, such right-of-way shall be dedicated to the City in fee simple.
- 2. Where the right-of-way of any drainage improvement required by these regulations is less then twenty-five (25) feet in width, such right-of-way may be designated as a dedicated easement.

13.1.7 Public Use Sites and Open Space

Where the Comprehensive Land Use Plan proposes community or public facilities or open spaces that are located in whole or in part within any proposed subdivision up for approval, the Planning Commission shall require the reservation of an area within the subdivision sufficient to accommodate such facilities in accordance with the adopted policy for maintenance of a minimum five (5) percent open space. Each new subdivision shall be required to include a minimum of five (5) percent of the total land area of the subdivision site as an open space contribution as defined in Article 3. The Planning Commission may allow the developer to provide one (1) acre of created wetlands habitat in lieu of one and one-quarter (1.25) acres of otherwise required open space.

13.1.8 Control of Erosion and Sedimentation

Where a developer/subdivider intends to make changes in the contour of any land proposed to be subdivided, developed or changed in use by grading, excavating or the removal or destruction of the natural topsoil, trees, or other vegetative covering thereon, the same shall only be accomplished after the Planning Commission shall have approved a plan for erosion and sedimentation control submitted by the owner or his agent, unless there has been a prior determination by the Planning Commission that such a plan is not necessary. Such determination shall be made in conjunction with tentative approval of the conceptual sketch plat and the site features map. Submittal of any plans for erosion and sedimentation control shall accompany and be considered a part of the drainage report submitted in conjunction with the Preliminary Subdivision Application.

1. Requirements for Submission

- a. Three (3) sets of plans for control of erosion and sedimentation shall be submitted to the Planning Commission.
- b. The estimated cost of accomplishing such erosion and sedimentation measures shall be stated in the construction agreement and shall be covered in any required performance bond and the maintenance bond.

2. Guidelines for Erosion and Sedimentation Control

- a. Lots shall be developed to maximize the amount of natural drainage which is percolated into the soil and to minimize direct runoff into adjoining streets and watercourses.
- b. Sediment basins (debris basins, desilting basins, or silt traps) should be installed to remove sediment from runoff waters from land undergoing development.
- c. The development plan should be fitted to the topography and soils so as to create the least erosion potential.
- d. Wherever feasible, and in areas where required by the provisions of this ordinance, natural vegetation should be retained and protected.
- e. Provisions should be made to effectively accommodate the increased runoff caused by changed soil and surface conditions during and after development.

- f. Where necessary, temporary vegetation and/or mulching should be used to protect areas exposed during development.
- g. The permanent final vegetation and structures should be installed as soon as practical.
- h. When land is exposed during development, only the smallest practical area should be exposed at one time and the exposure time should be kept to the shortest practical period of time.
- i. At the building permit application stage, a review will be conducted by the Inspection Department to insure conformance with the plan as approved.

13.2 <u>SUBDIVISION LAYOUT DESIGN</u>

13.2.1 Lot Layout and Improvements

13.2.1.1 Lot Arrangement

The lot arrangement shall be such that there will be no foreseeable difficulties, for reasons of topography or other conditions, in securing building permits or Corp of Engineers or coastal management authority permits to build on all lots in compliance with these regulations and parish health regulations, and in providing driveway access to buildings on such lots from an approved street. The design and layout of lots shall be such that:

- 1. No remnants of property shall be left which do not conform to lot requirements, which are not required for a private or public utility purpose, or which are not accepted by the City and/or any other appropriate public body for an appropriate use.
- 2. When land is subdivided into very large parcels they shall be of such shape and dimensions as to render possible the re-subdivision of any such parcels at some later date into lots and streets which meet the requirements of these regulations.
- 3. Lots shall be laid out so that drainageways are near the edge of lots and not near the center of a lot. Lots shall be laid out so that drainageways are located appropriately with regard to natural or man-made drainageways, including those existing and planned for the proposed development. Property lines, where feasible, shall be laid out so that the lines follow the center line of any

drainageway, except when such drainageway is greater than twenty-five (25) feet in width and required to be dedicated to the City.

- 4. Except where permitted by the Planning Commission and City Council in Planned Districts with site plan approval by ordinance, or as provided by these regulations, no lot shall be laid out so that it does not have access to and frontage on a dedicated public street according to the requirements of these regulations.
- 5. No lot may be created that is so narrow or irregularly shaped that it would be impracticable to conform to district setback regulations or to construct a building that could be used for purposes that are permissible in that zoning district.

13.2.1.2 Lot Area, Dimensions and Width-to-Length Ratio

- 1. The lot area square footage requirements of these regulations shall not include any dedicated rights-of-way, drainage easements or areas of periodic inundation.
- 2. Lot dimensions shall comply with the minimum standards of this Ordinance. In general, side lot lines shall be at a right angle to street lines (or radial to curving street lines) unless a variation from this rule will give a better street-to-lot plan in the judgement of the Planning Commission. Dimensions of corner lots shall be large enough to allow for erection of buildings, observing the minimum front yard setback from both streets. Depth and width of properties reserved or laid out for business, commercial or industrial purposes shall be adequate to provide for the off-street parking and loading facilities required for the type of use and development contemplated, as established in these regulations.
- 3. Lot width-to-depth ratio shall not exceed three and one-half to one (1:3.5) on lots with two hundred (200) foot frontage on a street, nor more than three to one (1:3) on lots with less than two hundred (200) feet of frontage on a street.

13.2.1.3 Double Frontage Lots and Access to Lots

- 1. Residential lots shall not be laid out so that they have frontage onto more than one street except:
 - a. Where the lot is adjacent to the intersection of two streets; or

- b. Where necessary to provide separation of residential development from traffic arterials, railroad rights-of-way, or to overcome specific disadvantage to topography and orientation.
- 2. Where double frontage lots cannot be avoided, as provided for above, the following requirements shall be met:
 - a. Front yard requirements shall be adjacent to the local or lesser traveled street;
 - b. An additional ten (10) foot strip of land, meeting the requirements of a vegetative buffer as described in Article 9 of these regulations, shall be added to the rear of the lot and dedicated to the City as a servitude or right-of-way for installation of protective screen planting by the developer. Such vegetative buffer strip shall not be fenced into the lots but shall remain open to view from the adjacent street to buffer the rear yard of the lot to the street; and
 - c. Lots shall have access to the lesser traveled street only.

13.2.1.4 Soil Preservation, Grading, and Seeding of Lots

- 1. Preservation and Final Grading No certificate of occupancy shall be issued until final grading has been completed in accordance with the approved final subdivision plat and the lot pre-covered with soil with an average depth of at least six (6) inches which shall contain no particles over two (2) inches in diameter over the entire area of the lot except that portion covered by buildings or included in streets, or where the grade has not been changed or natural vegetation seriously damaged. Topsoil shall be redistributed so as to provide at least six (6) inches of cover on the lots and at least four (4) inches of cover between the sidewalks and curbs, and shall be stabilized by seeding or planting.
- 2. <u>Lot Drainage</u> Lots shall be laid out so as to provide positive drainage away from all buildings, and individual lot drainage shall be coordinated with the general storm drainage pattern for the area. Drainage shall be designed so as to avoid concentration of storm drainage water to flow from any lot to any adjacent lots.
- 3. <u>Lawn-grass Seed and Sod</u> Lawn-grass seed shall be sown at not less than four (4) pounds to each one-thousand (1,000) square feet of open land area. The width of the right-of-way excluding roadway shall be seeded. Sod may be used to comply with any requirements of seeding set forth herein, if approved in advance by the City Engineer or Public Works Director.

13.2.1.5 Special Requirements

- 1. <u>Debris and Waste</u> No cut trees, timber, junk, rubbish or other waste materials of any kind shall be buried in any land, or left or deposited on any lot or street adjacent to a lot for which there is sought the issuance of a certificate of occupancy in the subdivision; nor shall any be left or deposited in any area of the subdivision at the time of expiration of the performance bond or dedication of public improvements, whichever is sooner.
- 2. <u>Fencing Under Hazardous Conditions</u> Each subdivider and/or developer shall be required to furnish and install fences wherever the Planning Commission determines that a hazardous condition may exit. The fences shall be constructed according to the Mandeville Building Code. No certificate of occupancy shall be issued until said fence improvements have been duly installed or a performance bond covering such fencing has been submitted and accepted by the City Council in an amount recommended by the City Engineer or Public Works Director.

13.2.1.6 Lots Abutting Waterbodies and Watercourses

If a tract being subdivided contains a waterbody or portion thereof, instead of the required dedication of the waterbody to the City the Planning Commission may approve an alternative plan whereby the ownership of and responsibility for safe maintenance of the waterbody is so placed that it will not become a responsibility of the City by requiring lot lines to be drawn so as to distribute the entire ownership of the waterbody among the fees of the adjacent lots. Under no circumstances, other than is provided for boat house developments, shall any of the minimum area of a lot as required under these regulations be satisfied by land which is under water or in areas subject to periodic inundation. Where a waterbody separates the buildable area of a lot from the street by which it has access, provisions for adequate access shall be designed and constructed by the developer in conjunction with the subdivision improvements and shall be of a design approved by the City Engineer or Public Works Director.

13.2.1.7 Performance Bond to Include Lot Improvements

The performance bond shall include an amount sufficient to guarantee completion of all requirements contained in these regulations under Section 13.2 including but not limited to, soil preservation, final grading, removal of debris and waste, fencing, and all other lot improvements required by the Planning Commission. Whether or not a certificate of occupancy has been issued, at the expiration of the performance bond the City Council may

enforce the provisions of the bond where the provisions of this section or any other applicable law, ordinance, or regulation have not been complied with.

13.2.2 Block Layout

- 1. <u>General standards</u> The lengths, widths, and shapes of blocks shall be designed with regard to the following considerations:
 - a. The convenient access, efficient circulation, and the control and safety of vehicular, pedestrian, and bicycle traffic.
 - b. The limitations of the topography of the land for the construction of drainage facilities and provision for open space.
 - c. The zoning requirements relative to lot size and the dimensions of the required building envelope.
 - d. Block layout shall provide adequate building sites suitable to the special needs of the type of use contemplated.
 - e. To minimize the destruction of trees and to preserve such natural features as watercourses, sites of historical or archaeological significance, and other assets which, if preserved, will add to the attractiveness and value of the subdivision and the City.
 - f. To afford good visual and physical access to and safety within existing or proposed recreational areas and parks.
- 2. Blocks designed for commercial and industrial uses shall be of such length and width as may be determined suitable by the Planning Commission for the proposed use and to accommodate anticipated development. Blocks intended to be used for commercial or industrial purposes shall be designed specifically for such uses with space set aside for buffer, off-street parking and loading and unloading facilities as required by these regulations. Specific design standards and construction standards shall be in accordance with the recommendation of the City Engineer or Public Works Director and the standards of this Article.
- 3. <u>Pedestrianways or Bike Paths</u> Pedestrianways or bike paths may be required in the design of blocks in certain areas. For reasons of safety and access the Planning Commission may require pedestrianways or bike paths in certain areas where it is necessary to provide circulation or access to schools,

playgrounds, shopping areas, community facilities or community-wide pedestrian or bike routes.

- 4. <u>Areas Where Normal Block Design is Not Possible or Undesirable</u> When a normal block arrangement is impossible or undesirable, one or more "places" may be established. Such a "place" may be in the form of a cul-de-sac street, a U shaped street or short loop street meeting the requirements of these regulations. Proper access shall be given to all lots as provided in this Article.
- 5. Whenever the necessity of the construction of sidewalks has been waived by the Planning Commission, sidewalks on or adjacent to the property for which the waiver was granted shall thereafter be required to be constructed by the Owner of the property only in any of the following instances:
 - a. Where there is a change in use, as this term is defined in subsection 6.1.6 of this Comprehensive Land Use Regulations Ordinance, from the use of the property for which the waiver was granted; or
 - b. Where a proposed expansion of diminution of the existing use would result in a change of twenty-five (25) percent or more from the floor area of the existing or proposed use for which the waiver was granted; or
 - c. For residential uses, where a proposed expansion or diminution of the existing use would result in a change of fifty (50) percent or more from the number of existing or proposed residential units for which the waiver was granted.

In any such instance the Planning Commission may, on the application of the owner of the affected property, again waive the necessity for the construction of sidewalks on or adjacent to the property in question, in accordance with the provisions of subsection 11.5.5 of this Comprehensive Land Use Regulations Ordinance.

13.2.3 Stormwater Drainage Requirements

13.2.3.1 General Requirements

A subdivision plat shall not be considered for preliminary approval until the applicant shall have submitted to the Planning Commission a storm drainage report by a civil engineer registered in Louisiana, as to the ability of existing watercourse channels, storm sewers, culverts and other improvements pertaining to drainage or flood control within the subdivision, to handle the additional run-off which would be generated by the development of the land within the subdivision. Additional information shall be submitted to adequately

indicate that provisions have been made for disposal of surface water without any damage to the developed or undeveloped land downstream, below or adjacent to the proposed subdivision. Drainage run-off shall be calculated as required by these regulations. The storm drainage report shall include: (a) estimates of the quantity of storm water entering the subdivision naturally from areas outside the subdivision; (b) quantities of flow at each pick-up point (inlet); and, (c) location, sizes and grades of required culverts, storm drainage sewers, retention ponds and other required appurtenances.

- 1. The developer shall plan all drainage facilities of the proposed subdivision or site or tract development in accordance with the City drainage improvement program, the requirements of these regulations, Louisiana highway drainage construction requirements, and/or as required by the City Engineer or Public Works Director and other applicable state and federal agency requirements.
- 2. All proposed subdivisions shall be reviewed by the Planning Commission and the City Engineer or Public Works Director in order to assure that:
 - a. All such proposals are consistent with the need to minimize flood damage;
 - b. All public utilities and facilities such as sewer, gas, electrical, and water systems are located, elevated, and constructed to minimize or eliminate flood damage; and
 - c. Adequate drainage is provided so as to reduce exposure to flood hazards.
 - d. The development plat contains a clear delineation of floodway and floodplain areas and has the notation that "land use in a floodway or floodplain is substantially restricted."
- 3. In the design of the drainage system for the proposed subdivision, right-of-way provisions shall be made to adequately take care of adjacent watershed areas. All drainage rights-of-way and structures shall be sufficient for the drainage of the adjacent watershed after complete development of the total watershed area.
 - a. Accommodation of Upstream Drainage Areas No development may be constructed or maintained so that such development unreasonably impedes the natural flow of water from higher adjacent properties across such development. All drainage rights-of-way and culverts or other drainage facilities shall be large enough to accommodate potential runoff from a subdivision's entire upstream drainage area whether inside or outside the subdivision and shall be designed minimally to handle the storm drainage runoff from a twenty-five (25) year design storm. The developer's engineer

shall determine the necessary size of the drainage facilities, assuming conditions of maximum potential watershed development permitted by these regulations. The proposed size of drainage facilities will be approved by the City Engineer or Public Works Director.

b. <u>Effect on Downstream Drainage Areas</u> - The developer's engineer shall also study the effect of each subdivision on existing downstream drainage facilities outside the area of the subdivision. Local drainage studies together with such other studies as shall be appropriate, shall serve as a guide to needed improvements. All improvements will meet the approval of the City Engineer or Public Works Director.

- i. No development may be constructed or maintained so that surface waters from such development are unreasonably collected and channeled onto lower adjacent properties at such locations or at such volumes or velocities as to cause substantial damage to such lower adjacent properties.
- ii. Where it is anticipated that the additional runoff incident to the development of the subdivision will overload an existing downstream drainage facility, the Planning Commission may withhold approval of the subdivision until provision has been made for the retention of storm water and resolution of such potential conditions in a manner satisfactory to the Planning Commission. No subdivision shall be approved unless adequate drainage will be provided to a drainage watercourse or facility adequate to receive the proposed drainage without adverse impact on downstream development.
- 4. Areas of Poor Drainage Whenever a plat is submitted for an area which is subject to flooding, the Planning Commission shall not approve such subdivision unless or until any required permits for the development of such areas have been issued by any wetlands agency with jurisdiction over the property. The Planning Commission may only approve such subdivision when the plat of such subdivision shall provide for an overflow zone along the bank of any stream or watercourse, in a width which shall be sufficient in times of high water to contain or move the water, and no fill shall be placed in the overflow zone nor shall any structure be erected or placed therein. The boundaries of the overflow zone shall be subject to approval by the City Engineer or Public Works Director. Development of areas of extremely poor drainage will be discouraged.

- 5. <u>Floodplain Areas</u> The Planning Commission may, when it deems it necessary for the health, safety, or welfare of the present and future population of the area or necessary to the conservation of water, drainage, and sanitary facilities, prohibit the subdivision or development of any portion of property which lies within the floodplain of any stream or drainage course. These floodplain areas shall be preserved from any and all destruction or damage resulting from clearing, grading, or dumping of earth, waste material, or stumps, except as otherwise expressly permitted by the Planning Commission and concurred in by appropriate state and federal agencies.
- 6. The storm or flood water drainage system shall be separate and independent of any sanitary sewer system and shall be located within the street right-of-way except where it is located in servitudes to facilitate outfall needs or for subdivision interconnections.
- 7. No individual, partnership, or corporation shall deepen, widen, fill, reroute or change the location of any existing ditch, stream, drain or drainage canal without first obtaining written permission from the City Engineer or Public Works Director. Plans for such filling, deepening, widening, rerouting, or changing the location of any existing ditch, stream, drain, or drainage canal shall comply with all design requirements and improvement standards of these regulations, as well as all applicable state and federal agency requirements. All such work shall be constructed under the review and subject to the approval of the City Engineer or Public Works Director or duly authorized representative of the City Engineer or Public Works Director. Adequate servitudes of rights-of-way must be dedicated for the construction and maintenance of any drainageways which may be relocated. No structures shall be erected or placed upon the drainage easements.
- 8. For all drainage channels originating within the subdivision, either new or existing, which are to be substantially altered by the developer, the developer shall make surface or subsurface drainage improvements according to the requirements of the drainage improvement program and these regulations. The Planning Commission and the City Engineer or Public Works Director will decide when such drainage channels are substantially altered.
- 9. The natural drainage within the subdivision shall be followed insofar as economically feasible. Streets and lots shall be arranged so as to keep artificially relocated drainage canals to a minimum.

13.2.3.2 Street Drainage Systems

All roadways shall be provided with an adequate subsurface storm drainage system. The road storm drainage system shall serve as the primary drainage system and shall be designed to carry roadway, adjacent land and building storm water drainage. No storm water shall be permitted to be run into the sanitary sewer system within the proposed subdivision. All roadways shall be provided with an adequate subsurface storm drainage system.

13.2.3.3 Off-Street Drainage Systems

The design of the drainage system and required easements shall include the watershed affecting the subdivision and shall be extended to a waterbody, natural watercourse or roadside ditch adequate to receive the storm or flood water drainage. An existing natural watercourse shall remain in its natural state. Man-made elements of this system may be designed as either open or subsurface systems.

13.2.3.4 Dedication of Drainage Easements and Rights-of-Way

1. <u>General Requirements</u> - If a subdivision is traversed by a watercourse, drainageway, channel or stream, an easement or drainage right-of-way conforming substantially to the lines of such watercourse shall be provided. The easement or right-of-way shall be of sufficient width to accommodate the watercourse and provide for maintenance of the watercourse. Whenever possible, it is desirable that the drainageway be maintained as an open channel with landscaped banks that approximates naturally occurring or predevelopment conditions and be of adequate width for maximum potential volume of flow and maintenance.

2. Drainage Easements

- a. Where topography or other conditions are such as to make impractical the inclusion of drainage facilities within road rights-of-way, perpetual unobstructed drainage easements at least fifteen (15) feet in width, depending on width of drainage facility, shall be dedicated to the City for drainageways that traverse property outside the road right-of-way lines with satisfactory access to the road. All easements shall be indicated on the plat and shall extend from the road right-of-way across the property to the easement of a natural waterbody or watercourse or to other drainage facilities as have been approved as the terminus of the easement.
- b. When a proposed drainage system will carry water across private land outside the subdivision, appropriate drainage rights and easements across abutting property must be secured prior to the final subdivision approval and such easements must be indicated on the plat.

- c. The necessary width of all drainage easements, whether supporting manmade or natural drainageways shall be determined by the applicant's engineer and approved by the City Engineer or Public Works Director. The drainage easements for natural drainageways will be identified minimally as areas less than five (5) feet in elevation (below the five foot contour), or areas that are subject to periodic inundation. When any of these drainage easements overlap, the largest area will be used to determine the minimum drainage easement.
- d. The drainage areas described above shall be included in areas for drainage easement or fee-simple dedication and shall be preserved and retained in their natural state. Such land shall not be computed in determining the number of lots to be utilized for average density procedure nor for computing the area requirement of any lot.

13.2.3.5 Nature of Storm Water Facilities

1. Responsibility for Runoff - The applicant will be responsible for the proper disposal of site runoff, whether by pipe, swale, stream or ditch. This includes any spring or surface water that may exist prior to or as a result of the subdivision. Such drainage facilities shall be located in the street right-of-way where feasible, or in perpetual unobstructed easements or dedicated rights-of-way of appropriate width as determined by the City Engineer or Public Works Director according to the requirements of this Article.

2. Accessibility to Public Storm Sewers

- a. Where a public storm sewer is accessible, the developer/subdivider shall install storm sewer facilities if so required by the City Engineer or Public Works Director. If no outlets are within a reasonable distance, adequate provision shall be made for the disposal of storm water, in accordance with the plans developed by the developer's engineer and approved by the City Engineer or Public Works Director. In subdivisions containing lots less than or equal to twenty thousand (20,000) square feet in area and in business and industrial districts, underground storm sewer systems shall be required to be constructed throughout the subdivision and be connected to an approved outfall.
- b. If connection to a public storm sewer will be provided eventually, as determined by the City Engineer or Public Works Director and the Planning Commission, the developer shall make arrangements for future storm water

disposal by a public utility system at the time the plat receives final approval. Provision for such connection shall be incorporated by inclusion in the performance bond required for the subdivision plat.

- 3. <u>Fencing, Landscaping and Maintenance Provisions for Drainage Channels</u> If a watercourse or ditch is left open the developer/subdivider shall adequately protect all such drainageways to the satisfaction of the Planning Commission, the City Engineer or Public Works Director and these regulations.
 - a. <u>Fencing</u> Where a watercourse or ditch is left open, it may be determined by the Planning Commission or City Engineer or Public Works Director that the developer will be required to protect the drainageway by the installation of fencing in accordance with the requirements of these regulations.
 - b. Landscaping The developer/subdivider shall be required to retain in its natural state or to grade and plant to adequately protect all surface drainageways from erosion to the satisfaction of the Planning Commission and the City Engineer or Public Works Director, according to the provisions of these regulations. Wherever practical, native hydric vegetation shall be maintained in place or removed and replanted after grading operations are complete. All newly constructed ditches and channels shall be seeded or sodded or replanted with native hydric vegetation depending on the slope and type of soil. The requirements for seeding or sodding improvements shall be in compliance with Louisiana highway construction standards, the requirements of this Section and the requirements of these regulations.
 - c. <u>Maintenance</u> Any watercourse or ditch easement shall be wide enough to contain the required ditch slope with ample clearance for the operation of maintenance equipment according to the provisions of this Article.

13.2.4 Street Improvement Requirements

13.2.4.1 Conformance to Comprehensive Land Use Plan

When a tract of land to be subdivided or re-subdivided includes any part of a proposed arterial or collector street as designated in the Mandeville Comprehensive Land Use Plan and its Master Streets Plan, including all subsequent amendments and additions thereto, these street rights-of-way shall be platted by the subdivider in the location so designated and at the width indicated in this Article. Provisions shall be made for the construction, extension or widening of any public street(s) or drive(s) where justified by the anticipated traffic volume and circulation needs determined by the City Planner or Designee and the Planning Commission.

13.2.4.2 Street Extensions Required

- 1. <u>Existing Streets</u> The arrangement of streets in new subdivisions shall provide for the proper continuation of existing streets into adjoining areas.
- 2. <u>To Adjacent Property</u> Arrangement of streets in new subdivisions shall make provisions for the proper projection of streets to adjoining areas that are not subdivided but may be appropriate for future development. Consideration should be given to the potential development of the adjacent property to permit a feasible extension of the street in conformity with a generalized plan for the most advantageous development of the entire neighborhood.

3. Reserve Strips

- a. Adjacent to Unincorporated Areas The creation of a reserve strip at the terminus of platted streets that end at an adjacent vacant and/or unplatted parcel of land which lies beyond the current borders of the incorporated area of the City shall be required. The purpose for creating a reserve strip is to assure the continuity of logical circulation patterns while assuring the input of the City in the approval of the connections to existing City rights-of-way when the adjacent parcel is developed. These strips shall be dedicated as rights-of-way and shall be labeled "Option to open lies with the City Council". The pavement of the platted street shall extend up to but not into the reserve strip. A required barrier rail shall be installed by the developer in the reserve strip to prohibit access into the undeveloped parcel prior to subdivision of the adjacent parcel and to discourage any illegal dumping from occurring in the area.
- b. Adjacent to Vacant Land within the City The creation of a reserve strip at the terminus of platted streets that end at an adjacent vacant and/or unplatted parcel of land which lies within the current borders of the incorporated area of the City shall be required in accordance with the same provisions that apply to streets which terminate beyond the current borders of the City as specified above.
- c. The requirement of the barricade in reserve strips shall not prohibit access by the owners to undeveloped parcels of land within the incorporated areas of the City, however, development of the vacant parcel dependent upon access to the parcel through the reserve strip shall not occur until the reserve strip has been opened by the City Council in association with the approval of a subdivision plat or development permit by the City.

- 13.2.4.3 Frontage On and Access to Improved Street Required
 - 1. Required Subdivision Access No subdivision shall be approved unless the area to be subdivided shall have frontage on and access from an existing street shown on the Master Streets Plan, or can be shown to have frontage on and access by way of a street shown upon a plat approved by the Planning Commission and recorded in the Clerk of Court's office or can be shown to be on an existing public roadway. Such streets must be suitably improved as required by the standards, criteria and specifications of this Ordinance. Wherever the area to be subdivided will utilize existing road frontage, such road shall be suitably improved as provided in these regulations.
 - 2. Required Lot Access Every lot shall have access that is sufficient to afford reasonable means of ingress and egress for emergency vehicles as well as for all those likely to need or desire access to the property for its intended use. Driveways shall enter public streets at safe locations and shall be planned for convenient circulation suitable for traffic needs and safety. All driveways shall be constructed in accordance with the following requirements:
 - a. Vehicles shall enter and exit the lot in question without posing any substantial danger to themselves, pedestrians or vehicles traveling in abutting streets.
 - b. Interference with the free and convenient flow of traffic in abutting or surrounding streets shall be minimized.
 - 3. <u>Limited Access to Arterial Streets</u> Where a subdivision borders on or contains an existing or proposed arterial street, the Planning Commission shall require that access to such street(s) be limited by one of the following means:
 - a. The subdivision of lots shall be platted to orient the back of the lots on the arterial street and the front of the lots on a parallel local street. Buffering shall be provided as described in this Article along rear property line of lots that are adjacent to arterial streets.
 - b. A series of cul-de-sacs, U-shaped streets, or short loops projecting at right angles from a street that parallels the arterial street, with the rear lines of the lots backing onto the arterial street. Provisions for buffering described in this Article shall be made for such lots.
 - c. A marginal access or service road, separated from the primary arterial by a perimeter landscaped strip with access to the arterial at appropriate locations.

- 4. <u>Criteria for Access to Arterial Streets</u> The following criteria shall be used in the design of subdivisions adjacent to arterial streets:
 - a. Street design shall have the purpose of making adjacent lots desirable by cushioning the impact of heavy traffic, and of minimizing the interference with traffic on such thoroughfares.
 - b. The number of intersections of collector streets with arterial streets shall be held to a minimum. Wherever practicable, such intersections shall be spaced not more than one-fourth mile (about 1300 feet) apart. Construction of frontage or service roads shall be encouraged.
 - c. Construction specifications for frontage roads shall conform to the standards specified in these regulations and AASHTO Geometric Design Manual. Frontage roads shall connect with arterial streets by means of a two-lane feeder capable of queuing a minimum of four cars at the arterial street intersection.
 - d. Where frontage streets are not required, residential lots adjacent to arterial streets shall be served by a local residential street paralleling this arterial street. These lots shall have an additional twenty-five (25) percent of depth to provide a buffer along the arterial street. Cul-de-sacs or loop streets may also extend toward said arterials from a local street providing a buffer equal to twenty-five (25) percent of the depth of an average lot in the subdivision.
 - e. When the rear of any lot borders any such arterial the subdivider or developer may be required to execute or deliver to the City an instrument deemed sufficient by the City Attorney, prohibiting the right of ingress and egress from the arterial street to the lot. The instrument may be an open space dedication in fee simple or a servitude granted to the City.

13.2.4.4 Street Dedications and Reservations

1. New Perimeter Streets - Street systems in new subdivisions shall be laid out to eliminate or avoid new perimeter half-streets except as provided in this Article. Where an existing half-street is adjacent to a new subdivision, the other half of the street shall be improved and dedicated by the developer/subdivider. The Planning Commission may authorize a new perimeter street when the subdivider improves and

dedicates the entire required street right-of-way width within the subdivider's own subdivision boundaries.

2. <u>Widening and Realignment</u> - Where a subdivision borders an existing narrow road or when the Master Streets Plan would require some of the land in the subdivision for road improvements, the developer/subdivider shall be required to improve and dedicate the widened or re-aligned streets. Such frontage roads and streets shall be improved and dedicated by the developer/subdivider at his own expense to the full width as required by these regulations. Land reserved for improvement purposes may not be used to satisfy yard or area requirements of Article 7 of these regulations whether dedicated in fee simple or as an easement.

13.2.4.5 Street Layout and Relationship of Topography

All streets and rights-of-way shall conform to the widths designated in this Article. Right-of-way widths in excess of the designated standards shall be required when additional width is necessary to provide adequate earth slopes due to topography.

- 1. Roads/streets shall be related appropriately to the topography. Local streets shall be arranged to maximize the number of building sites at or above the street grade and shall conform as closely as possible to the original topography. A combination of steep grades and curves shall be avoided. Criteria and specifications for street construction are contained in this Article.
- 2. Streets shall be laid out to facilitate the separation of local and through traffic, permit efficient drainage, open space and utility systems, and to require the minimum number of streets necessary to provide convenient and safe access to property.
- 3. All arterials shall be properly related to special traffic generators such as industries, business districts, schools, churches, and shopping centers; to population densities; and to the pattern of existing and proposed land uses. Pedestrian and bicycle access shall be provided where appropriate as indicated by the Planning Commission according to this Article.
- 4. The use of curvilinear streets, cul-de-sacs, U-shaped or loop streets shall be encouraged where such use will result in a more desirable layout than is possible with the standard rectangular grid. The use of curvilinear streets will also be encouraged to maintain natural drainageways. Subdivisions shall have more than one exit street, preferably using different streets and/or different directions of travel away form the subdivision.

- 5. In commercial and industrial developments, the streets and other accessways shall be planned in connection with the grouping of buildings, location of rail facilities, and the provisions of alleys, truck and loading and maneuvering areas, and walks and parking areas so as to minimize conflict of movement of the various types of traffic, including pedestrians.
- 6. Major arterial streets should not be intersected by local streets. Collector streets should not intersect major arterial streets at intervals of less than one-fourth mile.
- 7. Streets shall be graded and improved to conform to the standards provided in these regulations. Design and specifications for grading shall be approved by the City Engineer or Public Works Director in accordance with the construction plans required to be submitted prior to preliminary plat approval.
- 8. Where a cut or fill slope is outside the normal right-of-way of the street, a slope easement shall be provided of sufficient width, as determined by the Planning Commission, the City Engineer or Public Works Director, to permit maintenance of the slopes and to provide an adequate sight distance. Such slopes shall not be in excess of three to one (3:1) or thirty-three percent.

13.2.4.6 Arrangement of Streets

- 1. <u>Continuation of Principal Streets</u> The arrangement of streets shall provide for the continuation of principal streets between adjacent properties when such continuation is necessary for convenient movement of traffic, effective fire protection, efficient provision of utilities, and where such continuation is in accordance with the Mandeville Master Streets Plan.
- 3. <u>Cul-de-sac and Permanent Dead-End Streets</u> Where a street does not extend to the boundary of the subdivision and its continuation is not required by the Planning Commission for access to adjoining property, its terminus shall not be nearer to such boundary than fifty (50) feet. However, the Planning Commission may require the reservation of an easement to accommodate drainage facilities, pedestrian traffic or utilities from the terminus of the street to the boundary of the subdivision. A cul-de-sac turn-around shall be provided at the end of a permanent dead-end street in accordance with the street construction standards described in this Article. For greater traffic efficiency and effective police and fire protection, permanent dead-end streets shall be limited in length according to the design standards of this Article.

2. <u>Temporary Dead-End Streets</u> - If the adjacent property is undeveloped and the street must be a dead-end street temporarily, the right-of way shall be extended to the property line and a reserve strip provided as required. A temporary T or L shaped turnabout shall be provided on all temporary deadend streets with the notation on the subdivision plat that land outside the normal street right-of-way shall revert to abutting land whenever the street is continued. The Planning Commission may limit the length of temporary deadend streets in accordance with the design standards of these regulations.

13.2.4.7 Required Construction of Streets and Bridges

- Streets Surfacing and Improvements After sewer and water utilities have 1. been installed by the developer, curbs and gutters, where gutters are required, shall be constructed and roadways shall be surfaced to the widths prescribed in these regulations. The surfacing shall be of a character suitable for the expected traffic and in harmony with similar improvements in the surrounding areas. Types of pavement and design specifications shall be as presented in this Article or as determined by the City Engineer or Public Works Director. Adequate provisions shall be made for culverts, drains, and bridges. All street pavement, shoulders, drainage improvements and structures, curbs, turn-arounds, and sidewalks shall conform to all construction standards and specifications adopted by the City or as recommended by the City Engineer or Public Works Director and shall be incorporated into the construction plans required to be submitted by the developer for preliminary plat approval. A complete soil boring report prepared by a professional geotechnical engineer shall be required prior to the design of the cross section of any new street.
- 2. <u>Bridges</u> Bridges of primary benefit to the developer/subdivider, as determined by the Planning Commission, shall be constructed at the full expense of the developer/subdivider without reimbursement from the City. All bridges shall be constructed according to the standards, criteria and specifications recommended in the Louisiana Department of Transportation and Development Standard Specifications for Roads and Bridges and Bridge Design Manual. The sharing of any expense for the construction of bridges not of primary benefit to the developer/subdivider as determined by the Planning Commission will require City Council approval. This cost shall be charged to the developer/subdivider as a pro rata, proportionate share of the private land developed and so served.

13.2.4.8 Street Names, Street Signs, Street Lighting & Street Landscaping

1. <u>Street Names</u> - Names shall be sufficiently different in sound and in spelling from other street names in the City so as not to cause confusion. A street which exists or is planned as a continuation of an existing road shall bear the same name. All proposed street names will be checked against duplication of street names and names shall be approved by the Planning Commission.

2. Street Signs

- a. <u>Traffic Regulatory Signs</u> The applicant shall deposit with the City at the time of application for final subdivision approval the current cost for each sign required to be installed at all street intersections. The Streets Maintenance Division will install or supervise the installation of all traffic control signs on streets under the jurisdiction of the City.
- b. <u>Street Name Signs</u> Street name signs are to be placed at two locations at all intersections within or abutting the subdivision, the location of which is to approved by the City. The installation of street name signs shall be the responsibility of the developer and the installation of such shall be complete before acceptance of the dedication of the street on which the signs are required to be placed or included in the

cost of the street in conjunction with any performance bond for the construction of the street.

- 3. <u>Street Lights</u> Installation of street lights shall be required in accordance with design and specifications standards provided herein or supplied by the City Engineer or Public Works Director. The installation of street lights shall be the responsibility of the developer and the installation of such shall be complete before the issuance of a Certificate of Occupancy for any residence on the streets approved or shall be covered in any performance bond accepted by the developer.
- 4. <u>Street Landscaping</u> The medians of boulevard streets shall be required to be landscaped and City water lines shall be required to be installed in such medians for the purpose of watering landscape materials within the median.

13.2.5 Pedestrian and Bicycle Facilities Requirements

1. <u>Sidewalks</u> - Sidewalks shall be required to be located within the dedicated right-of-way of all streets and shall be improved as required by these regulations. A median strip of ground covering material or landscaped area at least four feet (4) wide shall separate all sidewalks from adjacent curbs or

- from the edge of the shoulder of a drainage channel nearest to the development lot.
- 2. Pedestrian Accessway The Planning Commission may require a perpetual unobstructed easement at least ten (10) feet in width to facilitate pedestrian access from the road to schools, parks, playground, other nearby collector roads or community-wide pedestrian routes. Such easements shall be indicated on the plat. Construction of a permanent surface for pedestrian traffic within the easement shall be required and shall be of material to be selected by the Planning Commission to coordinate with other linked pedestrian improvements. Plans and specifications for permanent surfacing shall be approved by the City Engineer or Public Works Director.
- 3. <u>Bicycle Paths</u> The Planning Commission may require a perpetual unobstructed easement at least twelve (12) feet in width to facilitate bicycle access from the road to schools, parks, playgrounds, other nearby collector roads or community-wide bicycle routes. Such easements shall be indicated on the plat. Construction of a permanent surface for bicycle traffic within the easement shall be required and shall be of materials to be selected by the Planning Commission to coordinate with other linked bicycle improvements. Plans and specifications for permanent surfacing shall be approved by the City Engineer or Public Works Director.
- 4. <u>Provisions for Handicapped Access</u> Whenever curb and gutter construction is used on public streets, wheelchair ramps for the handicapped shall be provided at intersections and other major points of pedestrian flow, such as crosswalks. Wheelchair ramps and depressed curbs shall be constructed in accordance with minimum standards required by Title III of the Americans with Disabilities Act, Public Law 101-336.

13.2.6 Sanitary Sewer Facilities Requirements

1. <u>General Requirements</u> - The applicant shall install sanitary sewer facilities in a manner approved by the City Engineer or Public Works Director according to the provisions of these regulations and applicable sections of the code of Ordinances of the City. All plans shall be designed in accordance with the rules, regulations, and standards of the City Engineer or Public Works Director, parish health department, and other appropriate local, state or federal agencies. All such plans shall be approved by the above agencies. Sanitary sewer lines may be located in the street right-of-way or side or rear easements.

- 2. a. Connection to Public Systems in Connection with Subdivision Procedures Sanitary sewer facilities shall connect with public sanitary sewerage systems. Sewers shall be installed to serve each lot and to grades and sizes required by approving officials and agencies and state and local regulations. No individual disposal system or treatment plants (private or group disposal system) shall be permitted. Sanitary sewerage facilities (including the installation of laterals in the street right-of-way) shall be subject to the specifications, rules, regulations, and guidelines of the City and of the State Board of Health, the Parish Health Office, City Engineer or Public Works Director and other jurisdictional agencies.
 - b. <u>Connection to Public Systems When Available</u> If a public sanitary sewer is or becomes accessible and a sanitary sewer line is placed within 300 feet of any habitable structure, the owner thereof shall be required to connect to said sewer for the purpose of disposing of waste, and it shall be unlawful for any such owner or occupant to maintain upon any such property an individual sewerage disposal system.

13.2.7 Water Facilities Requirements

- 1. The drilling, maintenance and use of private water wells for domestic use and human consumption within the City of Mandeville is prohibited except as provided by the regulations of the City. Necessary action shall be taken by the applicant to extend existing City water main lines or, when the nearest city water main or the capacity of the City well providing the nearest water main with water are not sufficient to adequately provide water for the proposed development, to create a new city water-supply district for the purpose of providing a water-supply system capable of providing domestic water use and fire protection for the proposed development or the proposed development shall be subject to disapproval of the City.
- 2. Where an adequate public water facility is accessible the applicant shall install adequate water facilities (including fire hydrants) in accordance with these regulations and subject to the specifications of the state and local authorities throughout the proposed development.
- 3. Fire hydrants shall be required for all subdivisions. Fire hydrants shall be located no more than 400 feet apart and within 200 feet of any structure, and shall be located at a street corner wherever possible. All underground utilities for fire hydrants and all other supply improvements shall be installed before any final paving of the street.

- 4. All water mains shall be at least eight (8) inches in diameter. Water main extensions shall be approved by the appropriate local, state and federal agencies and the City Engineer or Public Works Director.
- 5. To facilitate the above, the location of all fire hydrants, all water supply improvements, and the boundary lines of proposed districts, indicating all improvements proposed to be served, shall be shown on the preliminary plat, and the cost of installing same shall be included in the performance bond to be furnished by the developer.
- 6. Water lines shall be located in the street right-of-way and placed on the opposite side of the street from the sanitary sewer lines except when a majority of the lots to be served are on one side of the right-of-way, then the water line may be placed on the same side as the sewer line installed in accordance with the requirements of the appropriate local, state and/or federal agency and the City Engineer or Public Works Director.
- 7. Water lines for the purpose of watering landscaped areas such as boulevard medians and landscaped open space areas required in conjunction with a subdivision or public improvements application shall be required to be installed.

13.2.8 General Utility Requirements

- 1. All utility facilities, including but not limited to gas, electric power, telephone, and CATV cables, shall be located underground throughout the subdivision. All utility facilities existing and proposed throughout the subdivision shall be shown on the preliminary plat. Underground service connections to the street property line of each platted lot shall be installed at the subdivider's expense. At the discretion of the Planning Commission, the requirement for service connections to each lot may be waived in the case of adjoining lots to be retained in single ownership and intended to be developed for the same primary use.
- 2. Wherever possible, easements for public utilities shall be located in the dedicated street right-of-way. Easements for private utilities shall be provided on the front of lots, in the street right-of-way or partially in the street right-of-way and partially on the proposed lots fronting the street right-of-way whenever possible.
- 3. When for whatever reason it is necessary that utilities, including drainageways, are required to be located at the rear property line of lots, a

perpetual unobstructed easement shall be provided along the rear lot line of each lot to accommodate the utilities and any other proper public purposes and such easement shall be adequate to provide access for maintenance of the utility within the easement.

- 4. When necessary to provide utility services to a proposed development, the applicant for the proposed development shall be responsible for the coordination with the applicable utility company for the establishment of utility easements in adjoining developed or undeveloped properties.
- 5. Where topographical or other conditions are such as to make impractical the inclusion of utilities within the rear lot lines, perpetual unobstructed easements shall be provided along side lot lines of selected lots within the subdivision. Such easements shall be indicated on the preliminary and final plat.

13.2.9 Preservation of Natural Features and Amenities

- 1. General Requirements Existing features which would add value to residential development or to the City as a whole, such as trees required to be preserved by these regulations, watercourses, beaches, historic spots, and similar irreplaceable assets, shall be preserved in the design of subdivisions or other developments. No trees shall be removed from any subdivision nor the grade of the land within the subdivision be altered until approval of the preliminary plat has been granted. If certain trees on the Site Features Map or plat are required to be retained, they shall be preserved and the area of land within their drip lines shall be protected against any change of grade. The Site Features Map shall show the number and location of existing trees as required by these regulations and shall further indicate all those marked for retention.
- 2. Existing Trees to be Preserved Existing trees 3" dbh shall not be cut or otherwise damaged or destroyed within those portions of the subdivision set aside for open space, including any parcels to be dedicated as parks or open space or any required yard setback areas of developable lots. Vegetation buffer zones shall be required to be maintained in their natural state or when the natural vegetation is deficient to meet the requirements of the vegetation buffer zone as described in the Landscaping Provisions of Article 9, the minimum required planting as set forth in Article 9 shall be installed by the developer unless or until a development permit is approved by the City, in accordance with the requirements of these Land Use Regulations, for the removal of the trees.

13.2.10 Non-Residential Subdivision Layout

- 1. <u>General Requirements</u> Non-residential subdivisions shall be subject to all the requirements of these regulations as well as any additional requirements of the Planning Commission in conjunction with the policies and requirements of the Comprehensive Land Use Plan or other adopted special plans of the City.
- 2. In addition to the requirements established by these regulations, the developer shall demonstrate to the satisfaction of the Planning Commission that the street, parcel and block pattern proposed is specifically adapted to the uses anticipated and take into account other uses in the vicinity. The following principles shall be observed:
 - a. Proposed commercial and industrial parcels shall be suitable in area and dimensions to the types of development anticipated;
 - b. Street rights-of-way and pavement widths shall be adequate to accommodate the type and volume of traffic anticipated to be generated thereupon;
 - c. Special requirements may be imposed by the City with respect to the installation of public utilities and facilities, including water, sewer, and storm water drainage facilities, streets, sidewalks, pedestrianways and bicycleways and the proposed linear park system;
 - d. Special requirements may be imposed by the City with respect to street, curb, gutter, and sidewalk design and construction;
 - e. Corners of all road intersections will be of such radius as not to require trucks to use a second lane to make a turn;
 - f. Parking and loading areas for trucks will be of sufficient size that trucks will not be required to block traffic on any secondary or major traffic artery in order to maneuver into parking and loading areas;
 - g. Every effort shall be made to protect adjacent residential areas from negative impact from a proposed commercial or industrial subdivision, including the provision of extra depth in parcels adjacent to existing residential areas, areas zoned for residential development or areas identified for residential development on the adopted Future Land Use Map;

h. Streets carrying non-residential traffic, especially truck traffic, shall not be extended to the boundaries of adjacent existing or potential residential areas.

13.2.11 Other Subdivisions

- 1. Waterfront Subdivisions and Marinas Approval of the development of marinas and waterfront subdivisions shall, because of the potential flood hazards and damage to the environment, require detailed study and analysis, as well as more stringent design and improvement requirements to protect the health, safety and welfare of the public, to protect the environment and to reduce future expenditures by the City for the repair of improvements damaged by floodwaters, for providing protection from flood hazards and for correcting damage to the environment. All waterfront development shall be subject to all regulations of the City and any requirements of state or federal agencies for such development.
- 2. Townhouse and Townhouse Condominiums For purposes of these regulations, townhouse or townhouse condominium subdivisions are subdivisions in which single-family attached dwellings, each on an independent lot, are offered for sale. The attachment of these dwellings is along common or party walls that are jointly owned. Under this arrangement lots may front on driveways which have direct access to a public street or any principle access roadway exceeding three hundred (300) feet in length, provided such driveways are held in common ownership by the owners of the townhouse lots having access on such driveways. Open spaces owned in common by the owners of the individual townhouse lots may be provided in lieu of the required yard setbacks applicable to single-family development in accordance with all applicable regulations of the City, state or federal authorities.

13.3 <u>CONSTRUCTION STANDARDS FOR SUBDIVISIONS AND PUBLIC IMPROVEMENTS</u>

13.3.1 General Requirements

In consideration of the acceptance of public improvements by the City and the assumption of the responsibility for maintaining the dedicated improvements the applicant for subdivision approval shall cause to be designed and constructed, at no expense to the City, the improvements required by these regulations in accordance with the specifications and requirements of this Ordinance.

13.3.2 Lot and Block Standards

13.3.2.1 Lot Standards

Lots within a proposed subdivision or resubdivision shall be required to be sized in accordance with the minimum lot size requirements of the zoning district in which the land is located and designed in accordance with all applicable provisions of these regulations, and in particular in accordance with the requirements specified under Subdivision Layout and Design Standards of this Article as well as the following:

- 1. In no case will residential lots be faced so as to require access directly off a Class A State Highway.
- 2. Corner lots shall have an additional width of 20% over and above the minimum required to permit the additional yard setback requirement on the side of lots adjoining the side streets.

13.3.2.2 Block Standards

- 1. <u>Minimum Width</u> Blocks shall have sufficient width to provide for two (2) tiers of lots of appropriate depths for the zoning district in which the land is located. Exceptions to this prescribed block width shall be permitted in blocks adjacent to major streets, railroads or waterways.
- 2. <u>Maximum-Minimum Length</u> The lengths, widths, and shapes of blocks shall be such as are appropriate for the locality and type of development contemplated, but block lengths in residential areas shall not exceed sixteen hundred feet (1600') or twelve times the minimum lot width required in the zoning district where located, whichever is less, nor be less than four hundred (400) feet in length. Where ever practicable, blocks along major arterials and collector streets shall be not less than one thousand (1000) feet in length.
- 3. Pedestrianways In long blocks or when deemed necessary to provide connection with community-wide pedestrian routes, the Planning Commission may require the reservation of an easement or right-of-way through the block to accommodate pedestrian traffic. Pedestrianways may be required by the Planning Commission through the center of blocks more than eight hundred (800) feet long where deemed essential to provide circulation or access to schools, playgrounds, shopping centers, transportation corridors or facilities, community pedestrian routes or other community facilities. When required, pedestrianways shall be a dedicated easement or right-of-way

not less than ten (10) feet in width and shall include the construction of a minimum six (6) foot wide paved surface in accordance with the specifications for sidewalks provided herein.

4. <u>Bicycle Paths</u> - In long blocks or when deemed necessary to provide connection with community-wide bicycle routes, the Planning Commission may require the reservation of an easement or right-of-way through the block to accommodate bicycle traffic. Bicycle paths may be required by the Planning Commission through the center of blocks more than eight hundred (800) feet long where deemed essential to provide circulation or access to schools, playgrounds, shopping centers, transportation corridors or facilities, community bicycle routes or other community facilities. When required, pedestrianways shall be a dedicated easement or right-of-way not less than twelve (12) feet in width and shall include the construction of a minimum eight (8) foot wide paved surface in accordance with the specifications for bicycle paths provided herein.

13.3.3 Storm Drainage Specifications and Standards

13.3.3.1 Storm Drainage Report

A subdivision plat shall not be considered for preliminary or final approval until the applicant shall have submitted to the Planning Commission a report by a registered civil engineer in Louisiana, as to the ability of existing watercourse channels, storm sewers, culverts and other improvements pertaining to drainage for flood control within the subdivision, to handle the additional run-off which would be generated by the development of the land with the subdivision. Additional information shall be submitted to adequately indicate that provisions have been made for disposal of surface water without any damage to the developed or undeveloped land downstream, upstream or adjacent to proposed subdivision. Drainage run-off shall be calculated by the Rational method, using the appropriate runoff coefficients for the projected composition of the watershed at build-out. If the City Engineer or Public Works Director determines that the Rational method will result in an insufficient indication of runoff flow rates, then other methods of runoff calculation, including those of the Soil Conservation Service, the United States Geological Service or the State Department of

Transportation shall be required by the City. This report will serve as the basis for all decisions regarding drainage of the proposed subdivision, and shall also include:

1. Hydraulic calculations to determine the quantity of storm water entering the subdivision from areas outside the subdivision, the quantity of storm water

- crossing the subdivision and the quantity of storm water leaving the subdivision.
- 2. Hydraulic calculations to show the proposed drainage design for collecting and transporting the storm water across the subdivision, and the proposed drainage design for transporting the storm water downstream from the proposed subdivision to Lake Pontchartrain.
- 3. A contour map of the subdivision with elevation intervals of one (1) foot. This contour map shall also included contours of sufficient elevation intervals of the adjacent areas which contribute storm water to the subdivision.
- 4. Quantities of storm water flow at each pick-up point (inlet).
- 5. Location, sizes and grades of required culverts, storm drainage systems, drainage ditches and other required appurtenances.
- 6. Any required plan for erosion and sedimentation control as required by these regulations.
- 7. Drainage plans submitted with the drainage report shall include a lot drainage plan meeting the general standards of these regulations. The lot drainage plan shall show the existing surface water drainage patterns for each and every lot, proposed surface drainage patterns for each and every lot and size, shape and slope of required drainage ditches across each lot.
- 8. The lot drainage plan shall be developed to provide that the storm drainage runoff from each individual lot is carried to the curb and gutter roadway directly in front of the lot or to the open ditch drainageway in front of the lot, to an inlet and catch basin that is part of a subsurface drainage pipe at the rear or sides of each lot or to an open ditch at the rear or side of each lot.
- 9. This report shall be submitted with the request for preliminary or final subdivision approval or prior to the approval of any drainage improvements proposed within existing subdivisions.

13.3.3.2 General Design and Construction Standards of Storm Drainage

1. The minimum design of the interior drainage systems of the subdivision or site or tract proposed for development shall be based on a ten (10) year storm frequency with a twenty-four (24) hour duration and the minimum design considerations for the watershed area will be based on a twenty-five (25) year

storm frequency with a twenty-four (24) hour duration. The selection of runoff coefficients shall be based on the anticipated nature of future development in the area, and shall be subject to the approval of the City Engineer or Public Works Director.

- 2. Street drainage and grading shall be addressed on the submitted plans and shall extend the full width of the right-of-way.
- 3. a. Preservation of drainage patterns in the drainage basin in which a subdivision is located will be required by the Planning Commission.
 - b. No alteration of natural drainage channels shall be undertaken by a developer/subdivider except upon the express permission of the Planning Commission.
 - c. No increase in the rate of run-off that existed prior to development will be permitted by the Planning Commission unless the developer/subdivider, on a case by case basis, can establish to the satisfaction of the Planning Commission that the existing downstream drainage is adequate to handle the anticipated flow resulting from the proposed development of the property. Alternatively, the developer/subdivider may propose to undertake such work or improvements, at no cost to the City, to make the downstream drainage system adequate to handle the anticipated flow resulting from the development of the property. The Planning Commission may deny any such proposal to improve downstream drainage if the Commission determines that the nature or extent of the proposed work or improvements would detrimentally alter the character or condition of any downstream drainageways. The Planning Commission shall not approve any proposal to improve manmade drainageways until the proposal is first reviewed by the Public Works Director and the Planning Commission is thereafter advised by the Public Works Director, in writing, that he has no objection to the proposal.
 - d. Absent such approval by the Planning Commission runoff from the proposed development shall be retained on-site, using storage, swales, ponds, and basins, as approved by the City Engineer or Public Works Director, until the water can be released at a rate of flow that does not exceed the rate of flow from the property that existed prior to development.

13.3.3.3 General Improvement Requirements for Sub-surface Storm Drains

In consideration of the acceptance of the improvements by the City and the assumption of the responsibility for maintaining the dedicated streets constructed therein, the owners of the subdivision shall cause to be designed and constructed, at no expense to the City, the following drainage and storm sewer improvements according to the specifications set forth in this ordinance.

- 1. Drainage and Storm Sewer Construction Criteria
 - a. The design of drainage systems for the interior portions of subdivisions or site or tract development and for the watershed areas surrounding the subdivision or site or tract development shall be in accordance with the State of Louisiana Department of Transportation and Development Hydraulics Manual published in 1984, or as amended. The design of the interior drainage systems of the subdivision or site or tract development shall be based on a ten (10) year storm frequency. The selection of run-off coefficients shall be based on the anticipated nature of future development in the area and shall be subject to the approval of the City Engineer or Public Works Director.
 - b. Drainage design for new subdivisions or resubdivisions of three (3) acres or greater may be required to include retention ponds or other innovative methods of retaining storm water within the confines of the new subdivision or resubdivision. These retention ponds or other innovative method of retaining storm water within the confines of the new subdivision or resubdivision shall be designed to meet the following criteria:
 - i. Where downstream drainage systems are inadequate to accommodate the increase in the rate of runoff from the proposed development, the developer/subdivider may propose to undertake such work or improvements, at no cost to the City, to make the downstream drainage system adequate to handle the anticipated flow resulting from the development of the property. The Planning Commission may deny any such proposal to improve downstream drainage if the Commission determined that the nature or extent of the proposed work or improvements would detrimentally alter the character or condition of any downstream drainageways. The Planning Commission shall not approve any proposal to improve manmade drainageways until the proposal is first reviewed by the Public Works Director and the Planning Commission is thereafter advised by the Public Works Director, in writing, that he has no objection to the proposal. Absent such approval by the Planning Commission, the developer/subdivider shall be required to retain the increase in storm drainage runoff on the site of the development until this water can be released at a storm drainage runoff rate which does not exceed the storm drainage runoff from the site prior to development. However, when the design storm exceeds a twenty-five

(25) year storm frequency, the excess drainage runoff above the runoff created by the twenty (25) year frequency storm will be allowed to flow out of the new development.

2. Criteria for Subdivisions with Curb and Gutter Streets/Subsurface Drainage

In subdivisions with curb and gutter streets, the intervals for installation of curb drainage inlets shall be determined in accordance with the requirements of the Louisiana Department of Transportation and Development Hydraulics Manual published in November of 1984, or as amended. Where these inlets connect to storm sewers, a catch basin shall be installed with the inlet. Storm drain inlets will be placed so that surface water is not carried across intersections or crosswalks. When calculations indicate that curb capacities are exceeded at a point, no further allowance shall be made for flow beyond that point and a basin shall be used to intercept flow at that point. Drainage plans submitted with the drainage report will show surface water drainage patterns for each and every lot and block and all design specifications for land development.

3. Material and construction specifications for all subsurface drainage projects shall be in compliance with the requirements of all American Society for Testing and Materials (ASTM) and Louisiana Department of Transportation and Development Highway Construction Standards.

13.3.3.4 Open Drainage Ditch Construction and Design

- 1. Drainage ditches shall be designed, whenever possible, with sloping earthen banks and earthen bottoms. Drainage ditches shall also be designed with shoulders of not less than four feet in width. When it is determined by the Planning Commission upon the recommendation of the City Engineer or Public Works Director that a concrete lined channel is required such concrete lined channel shall have side slopes of a grade that is no steeper than one and one-half (1.5) to one (1) and shall be lined with reinforced concrete. The bottom shall be at least six (6) inches thick, sides at least four (4) inches thick, and a five (5) foot collar on each side shall be at least four (4) inches thick. Construction details for footings, joints, etc. shall be in accordance with standards provided by the City Engineer or Public Works Director.
- 2. When a proposed open ditch, whether lined or unlined, must discharge into a major unlined canal, the developer shall be required to enclose the ditch, under the access strip of the major canal, in a metal pipe or concrete culvert. The pipe or culvert shall extend at least four (4) feet into the canal beyond the

side slope, and shall discharge into a concrete flume that extends a minimum of five (5) feet into the bottom of the canal. Flume shall be constructed immediately after the pipe or culvert is installed.

13.3.3.5 Drainage Easement Criteria for Man-made Drainage Structures

The following servitude criteria shall be required for each ditch, canal, storm drainage collection line, and storm sewer:

- a. Minimum Easement Width fifteen (15) feet
- b. There will be a minimum of eight (8) feet from the top of the bank to the property line on one side of the easement for all canals with a top width of fifteen (15) feet or less.
- c. Canals with top width of greater than fifteen (15) feet but less than thirty-five (35) feet shall have a minimum of ten (10) feet on each side measured from the top of the bank to the property line.
- d. Canals with a top width of thirty-five (35) feet or greater will have a minimum of twenty-five (25) feet on both sides measured from the top of the bank to the property line.
- e. Canals with a top width in excess of forty (40) feet will have a minimum of twenty-five (25) feet from the top of the bank to the property line on both sides.

13.3.4 Street Specifications and Standards

13.3.4.1 General Regulations

The provisions of this Article are established to ensure that public street facilities are available and will have sufficient capacity to serve proposed subdivisions. This Article also provides for streets of suitable location and width with necessary pedestrian improvement to accommodate all types of traffic and afford satisfactory access for police, fire fighting, sanitation, and street-maintenance equipment. This Article regulates and coordinates existing and proposed streets to compose a convenient system and avoid undue hardships that otherwise might be imposed on adjoining properties. The following standards, criteria and specifications for streets shall be applied according to the street classification system adopted in the Comprehensive Land Use Plan. In no case shall residential lots be faced to

require access directly on an arterial street as defined by this Article. No lot area requirement shall be met by including servitudes dedicated for the improvement of natural or man-made drainage channels, or any utility servitude or other right-of-way.

13.3.4.2 Street Classification

The following definitions apply to the street classification system adopted in the City of Mandeville Master Streets Plan and required in conjunction with new subdivisions. <u>Local Streets</u> have the sole function of providing access to abutting properties. Local streets have an average daily traffic volume less than 1,000 vehicles per day. <u>Collector Streets</u> provide access to abutting properties but also serve to connect local streets with arterial streets. Collector streets have an average daily traffic volume of between 1,000 and 10,000 vehicles per day. <u>Arterial Streets</u> are major streets in the City's street network that serve traffic moving into, out of and around the City, carrying volumes of traffic greater than 10,000 vehicles per day.

1. <u>Spacing of Streets</u> - Streets shall be as close to the following spacing requirements as possible:

SPACING OF STREETS

Arterial Street	1-3 miles
Collector Street	1/2 mile
Local Streets	less than 1/2 mile

13.3.4.3 Sight Radius Standards

Lots on major street intersections and all acute angle intersections which, in the opinion of the Planning Commission or City Engineer or Public Works Director, are likely to be dangerous to the movement of traffic shall have a minimum sight radius of twenty-five (25) feet at the street corner. Where grade separation structures are proposed at the intersection of major streets, the lots and improvements in the subdivision shall be arranged so as to make adequate provision for such structures.

13.3.4.4 Street Construction Specifications

1. <u>General Street Design Criteria</u> - The general street design criteria approved by the City are shown in the following Table. Minimum and maximum grade, curvature, tangent and stop sight distances, design speed and right-of-way and

number of lanes for standard streets with central sewer systems are displayed for classified streets. Basic criteria for marginal access streets are also displayed.

General Street Design Criteria

	Min. Grade	Max. Grade	Horz. Curve	Min. Tan	Stop Sight Dist.	Des. Speed	Right of Way	# Lanes
"Marginal" Special Access	0.5%	10%	50'	50'	100'	15	40'	2
Local	0.5%	10%	100'	50'	175'	25	60'	2
Collector	0.5%	8%	350'	150'	250'	35	60'	2
Arterial	0.5%	5%	500'	300'	300'	50	80'	4

^{*} Where subsurface drainage requirements are waived, minimum Right of Way widths shall be increased to accommodate the approved drainage design, or a minimum of an additional ten feet, whichever is greater.

2. <u>Required Minimum Pavement and Lane Widths</u> - Pavement widths shall be measured from curb face or, if no curbs are required, then measurement shall include the entire paved surface. Minimum pavement and lane widths for standard streets are shown in the following Table.

Minimum Lane and Pavement Widths

Lane Width P	avement Width
--------------	---------------

Marginal Access	10'		20'
Local with curb without curb	11' 10'	22'	22'
Collector with curb without curb	11' 10'		38' 36'
Arterial with curb without curb	11' 10		48' 46'

3. Criteria for Cul-de-sacs and Dead-End Streets

- a. <u>Permanent Dead-End Streets</u> Cul-de-sac streets shall not be longer than five hundred (500) feet, serving no more than twelve (12) families, unless local topographic or other physical conditions make this provision impracticable. If a cul-de-sac street is permitted when continuation of a roadway is not required by the Planning Commission, its terminus shall not be nearer to the property boundary than fifty (50) feet. The Planning Commission may require the reservation of a ten (10) foot easement as described in this Article to accommodate pedestrian traffic. All cul-de-sacs shall be provided with:
 - i. A turning circle having a minimum right-of-way radius of fifty (50) feet and a minimum outside pavement radius of forty (40) feet. Pavement width at the turn-around shall be at least the width of the street it serves.
- ii. Interior turning circles at the end of dead-end commercial or residential streets shall be landscaped open green spaces to provide for surface

water infiltration. Such green spaces shall have a minimum sight radius of twenty (20) feet.

- b. <u>Temporary Turning Circle</u> If a dead-end street is of a temporary nature and a further extension into adjacent land is anticipated, then approval of a temporary cul-de-sac of one thousand (1,000) feet, serving no more than twenty-four (24) families may be granted. In such cases, a temporary dedication of the cul-de-sac right-of-way shall be required. Such dedications may revert to abutting property owners when the dead-end street is legally extended into adjacent land as described in this Article. A "T" or "Y" type turn-arounds may also be used to terminate temporary dead-end streets. This turn around area must be temporarily dedicated, with reversion to the abutting property as described in this Article.
- c. <u>Turning Circle Required / Not Required</u> If a dead-end street is to extend only one (1) lot depth past a street intersection and/or the street is to be less than two hundred and fifty (250) feet in length, no turning circle will be required, but, where a dead-end street is to extend two lot depths past a street intersection and/or the street is to be extended beyond two hundred and fifty (250) feet a turn-around is required.
- d. <u>Circular Offsets</u> Circular offsets that position the turning circle to one side of the center line of the street are permitted where local topography or other physical conditions are such as to render the normal cul-de-sac design impracticable.
- e. <u>Spacing of Cul-de-sacs</u> Cul-de-sacs may not be used so as to create a distance greater than one thousand (1,000) feet along any collector or arterial street for which there is no intersecting through street.

Minimum Turn-Around Dimensions for Dead-End streets

Local Roads	Residential	Commercial
Right-of-way radius	50'	60'
Pavement Radius	40'	50'
Center island radius	20'	20'

4. <u>Boulevards</u> - Right-of-way and construction width shall be commensurate with he street classification in the Master Streets Plan or as assigned by the Planning Commission or the City Engineer or Public Works Director. Design of boulevards shall meet the following minimum criteria:

Boulevard Pavement, Right-of-Way and Neutral Ground Widths*

Type of Street	Pavement Width	R/W Width**	Neutral Ground Width
Local with curb without curb	22' min. 22' min.	80' 80'	20' 20'
Collector with curb without curb	32' min.* 30' min.*	110' 110'	30' 30'
Arterial with curb without curb	36' min. 32' min.	120' 120'	30' 30'

- * All collector streets shall have space for eight (8) feet of parking or pedestrian/bicycle access. With parking or pedestrian/bicycle access the minimum pavement width shall be thirty-two (32) feet.
- ** Where subsurface drainage requirements are waived, minimum right-ofway widths shall be increased to accommodate the approved drainage design, or a minimum of an additional ten feet, whichever is greater.
- 5. <u>Alleys</u> Alleys shall not be provided in residential blocks, except under unusual conditions. Where alleys are provided in residential areas, they shall be of Portland Cement Concrete no less than six (6) inches thick and a minimum pavement width of twelve (12) feet and a minimum twenty (20) foot right-ofway. Asphaltic concrete may be used if approved by the City Engineer or Public Works Director when the main subdivision streets are also of asphaltic concrete composition.

6. Intersections

- a. Streets shall be laid out so as to intersect as nearly as possible at right angles.
- i. In no case shall the angle of the intersection be less than 75 degrees or greater than 105 degrees.

- ii. An oblique street shall be curved approaching an intersection and should be approximately at right angles for at least one hundred feet (100) therefrom.
- iii. Not more than two (2) streets shall intersect at any one point unless specifically approved by the Planning Commission.
- b. Proposed new intersections along one side of an existing street shall, whenever practicable, coincide with an existing intersection on the opposite of such street. Street jogs with centerline off-sets of less than 150 feet shall not be permitted.
- c. Alley intersections and abrupt changes in alignment within a block shall have the corners cut off in accordance with standard engineering practice to permit safe vehicular movement.
- d. The cross-slopes on all streets, including intersections, shall be three (3) percent or less.
- e. Where any street intersection will involve earth banks or existing vegetation inside any lot corner that would create a traffic hazard by limiting visibility, the developer/subdivider shall remove only shrub level vegetation between three (3) feet and seven (7) feet. No trees shall be removed except with the approval of the Planning Commission after inspection by the City Planner or Designee.
- 7. Specifications for New Street Construction
 - a. The soil boring report will be commissioned by the developer and provided to the City Engineer or Public Works Director at no charge. The report will be prepared by a licensed professional geotechnical engineer and will provide boring logs and the specifications for the cross sectional design of street pavement and required base preparation. Streets will be designed and constructed in accordance with the findings of the soil boring report, but will not be less than the minimum design criteria stated in paragraph 7(b).
 - b. Concrete Pavement & Base Requirements

Street Construction Standards by Development Density

Residential

Commercial - Industrial

Local Streets Collector Street Arterial Street 6" P.C. Concrete Pavement 7" P.C. Concrete Pavement 6" P.C. Concrete Pavement 6" P.C. Concrete Pavement 8" P.C. Concrete Pavement

- i. All pavement to be placed on a compacted base course with a minimum thickness of 6". The cross sectional requirement for the base course may be increased depending on soil conditions and the results of the soil boring report.
- ii. All concrete pavements shall have contraction, expansion and construction joints located in accordance with the latest requirements of the Portland Cement Association.
- iii. The Portland Cement concrete shall have a 28 day minimum compressive strength of 4,000 psi.
- c. <u>Asphalt Paving</u> The Planning Commission may waive the concrete pavement requirement and approve road construction of asphaltic concrete material when keeping in character with existing asphalt streets in the surrounding area. The developer's engineer must design the street based upon the results of the soil boring report. At a minimum, the street design must conform to the Louisiana Department of Transportation and Development Guidelines as defined in the then latest edition of "Louisiana Standard Specifications for Roads and Bridges". The design must be reviewed and approved by the City Engineer or Public Works Director.
- d. <u>Standard Specifications for Roads and Bridges</u> The compressive strength of all roadways and required base preparation shall be established based upon the Louisiana Department of Transportation and Development <u>Standard Specifications for Roads and Bridges.</u>
- 8. <u>Curbs and Gutters</u> When used, vertical curbs on shall be at least six inches (6) in height. Curbs may be vertical or roll top, and may be constructed of asphaltic concrete or Portland Cement concrete. The design of all curbs shall be appropriate for the proposed application, to be approved by the City Engineer or Public Works Director.

13.3.4.5 Excess Right-of-Way

Right-of-way widths in excess of the standards designated in these regulations shall be provided if required by the Planning Commission and the City Engineer or Public Works

Director to meet the standards for the preservation of trees, natural features and other amenities as referred to in Article 9 of these regulations.

13.3.4.6 Dedications along Existing Streets

- 1. <u>Half Streets</u> Dedication of one-half of the right-of-way (half streets) for streets along the boundaries of any proposed subdivision shall not be permitted except as provided in this Article or where boulevards are to be constructed along a route approved by the Planning Commission. In such instances, the adjacent land shall be platted to show the other half of the boulevard at the time such land is requested to be subdivided and a reserve strip shall be dedicated to be revoked and sold by the City at the time of request for subdivision.
- 2. Subdivisions platted along existing streets shall dedicate additional right-ofway, if necessary, to meet the minimum street width requirements as set forth in this Article. Such dedication shall be in accordance with the following:
 - a. If the subdivision is on both sides of an existing street, the minimum right-of-way width shall be dedicated;
 - b. If the subdivision is located on only one side of an existing street, one-half of the required width, measured from the centerline of the street right-of-way, shall be dedicated. In no case shall the owner or owners of such property be forced to dedicate from their land more than one-half of the required rights-of-way width.

13.3.4.7 Limited Access Highways and Railroads

Railroad rights-of-way and limited access highways that affect the subdivision of adjoining lands shall be treated as follows:

- 1. In residential districts a buffer strip that is twenty-five percent (25%) of the depth of the normal lot required in the district shall be provided in addition, adjacent to the railroad right-of-way or limited access highway. The strip shall be part of the platted lots and shall be designated on the plat: "This buffer strip is reserved for screening. The placement of any structure in the buffer is prohibited".
- 2. Streets parallel to the railroad when intersecting a street which crosses the railroad at grade shall, to the extent practicable be at a distance of at least 150 feet from the railroad right-of-way. Such distance shall be determined with

due consideration of the minimum distance required for future separation of grades by means of appropriate approach gradients.

13.3.4.8 Extension of Streets to Adjoining Properties

No "dedicated but not to be constructed" extensions shall be approved by the Planning Commission. The street pavement will extend right up to the boundary line of the one foot (1) wide reserve strip of dedicated right-of-way as provided elsewhere in this Article.

13.3.4.9 Street Signs Standards

- 1. <u>Traffic Control Signs</u> -The Streets Department shall install all traffic control signs and devices in a manner to conform with the latest Louisiana and National Standards for such signs and devices. The developer/subdivider shall install all street signs before issuance of Certificates of Occupancy for any residence on the streets approved.
- 2. <u>Street Name Signs</u> Criteria for the location of street name sign and a typical design is provided in the office of the Director of Public Works. The location of street name signs is to be approved by the City Engineer or Public Works Director. Installation of street name signs shall be covered in the required performance bond.

13.3.5 Sidewalk, Pedestrianway, and Bicycle Path Specifications and Standards

1. Sidewalks Standards and Specifications - Sidewalks may be required to be constructed in conjunction with the issuance of building permits for new construction or substantial improvements in front of lots in all new residential subdivisions of twenty-five (25) lots or more, in all new commercial subdivisions, and in front of lots in existing subdivisions where segments of existing sidewalks currently exist or where required sidewalks were previously approved by the Planning Commission to be constructed in conjunction with building permits. Plans for the installation of sidewalks meeting the minimum standards of this ordinance shall be required to be submitted to the Building Inspector, approved by the Director of Public Works and included in the permit fee. The Director of Public Works or designee shall inspect the installation of sidewalks for adherence to code. In previously approved subdivisions where the Planning Commission waived the requirement for the installation of sidewalks entirely, no sidewalks shall be required to be installed in conjunction with building permits. Sidewalks shall be placed on at least one side of local and collector streets, and on both sides of arterial streets.

Sidewalks shall be concrete, at least four (4) inches thick. An alternative surface may be selected and approved by the Planning Commission in conjunction with the approval of the subdivision in which the sidewalks are to be located for new subdivisions, however, in existing subdivisions or development areas where the Planning Commission has not previously waived the requirement for concrete sidewalks, concrete sidewalks shall be required. Standard concrete sidewalks shall be four (4) feet wide, and shall be placed within the street right-of-way, a minimum of four (4) feet from the edge of the pavement or street curb. Portland cement concrete for sidewalks shall attain 3,000 psi compressive strength within 28 days of placement. The requirements for concrete surfacing may be waived by the Planning Commission in locations where sidewalks coincide with or are linked with a linear park system or where a different surface material is selected by the developer and approved by the City Engineer or Public Works Director and the Planning Commission.

- 2. <u>Pedestrian Accessway Specifications</u> The standards for pedestrian accessways shall be determined by the City Engineer or Public Works Director on a case by case basis. However, the standards for the construction of sidewalks included herein shall be the minimum standard for the construction of pedestrian accessways.
- 3. <u>Bicycle Path Specifications</u> Bicycle paths and other bicycle facilities constructed in the City of Mandeville shall be constructed in accordance with the "Guide for the Development of Bicycle Facilities" prepared by the AASHTO Task Force on Geometric Design and published by the American Association of State Highway and Transportation Officials.

13.3.6 Sanitary Sewers Design Standards

1. <u>Alignment of Sewers</u> - All sewers shall be laid with straight alignment between manholes, unless otherwise directed or approved by the City Engineer or Public Works Director. Sanitary sewerage facilities shall connect with public sanitary sewerage systems. Sewers shall be installed to serve each lot and to grades and sizes required by approving officials and agencies and state and local regulations. No individual disposal system or treatment plants (private or group disposal system) shall be permitted. Sanitary sewerage facilities (including the installation of laterals in the right-of-way) shall be subject to the specifications, rules, regulations, and guidelines of the state board of health,

the parish health office, City Engineer or Public Works Director, and other appropriate state and federal agencies.

- 2. Sanitary Sewer Design and Improvement Standards
 - a. <u>Basis for Standards</u> Design and construction of all sanitary sewers, manholes, pumping stations, water distribution lines, water wells, water storage tanks, water pumps and all other appurtenances required in the water and sewer systems shall be in accordance with all federal, state and local rules and regulations and shall be approved by the proper federal, state and local authorities before installation and operation. All requirements of the state sanitary code shall be followed.
 - b. <u>Standards not included in Certain Situations</u> The recommended standards may not cover extraordinary situations. Deviations will be allowed and may be required in those instances where considered justified by the City Engineer or Public Works Director.
 - c. <u>Design Factors</u> Sanitary sewer systems shall be designed for their ultimate contributory population, with due consideration being given to current zoning regulations and to the comprehensive future land use plan. Sewer capacities shall be adequate to handle the projected maximum hourly flow of sewage and industrial waste, together with allowance for infiltration and other anticipated flow. Design values for average sewage flow, excluding commercial and industrial waste, shall be determined using accepted LA Dept. of Health and Hospitals standards for per capita design flow per person per day and a design population appropriate to the anticipated development of the area service area under consideration. Peak flow values shall also be determined by LA Dept. of Health and Hospitals standards. All design flows for sewerage systems are subject to the review, modification, and approval of the City Engineer or Public Works Director.
 - d. <u>Maximum/Minimum Sewer Size Permitted</u> The diameter of sewers proposed shall not exceed the diameter of the existing or proposed outlet, whichever is applicable, unless otherwise approved by the City Engineer or Public Works Director. In no case shall the proposed sewer be smaller than eight (8) inches in diameter.
 - e. <u>Minimum Sewer Slope Permitted</u> All sewers shall be designed with sufficient slope to give mean velocities when flowing full of not less than two (2) feet per second. All velocity and flow calculations shall be based on current state health department regulations.

- f. <u>Manhole Location Criteria</u> Manholes shall be installed at the end of each line at all changes in grade, size, or alignment; at all street and line intersections; and at a distance not greater than four hundred (400) feet.
- g. Manhole Design Criteria Drop manholes shall be required where the difference in elevation between any incoming sewer and the manhole invert exceeds twenty-four (24) inches. The use of drop manholes will require approval by the City Engineer or Public Works Director. The minimum inside diameter of the manhole shall conform to that specified by the City Engineer or Public Works Director. Inside drop manholes will require special considerations; however, in no case shall the minimum clear distance be less than that indicated above. When a smaller sewer joins a larger one, the insert of the larger sewer shall be lowered to maintain the same energy gradient.
- h. <u>Sewage Line Locations</u> Sanitary sewers shall be located at the front of lots where possible or within street rights-of-way unless topography dictates otherwise. When located in easements on private property on rear or side property lines, access shall be maintained to all manholes. A manhole shall be provided at each street or alley crossing. End lines shall be extended to provide access from street or alley right-of-way where possible. Imposed loading shall be considered in all locations. No less than six (6) feet of cover shall be provided over top of pipe in street and alley rights-of-way or three (3) feet in other areas.
- i. <u>Cleanouts and Lampholes</u> Cleanout and lampholes shall not be used on main line sanitary sewers, eight (8) inches in diameter and greater. Cleanouts shall be installed at the right-of-way line on all service lines.
- j. Separation of Sewer and Water Supply Systems There shall be no physical connection between public or private potable water supply system and a sewer which will permit the passage of any sewerage or polluted water into the potable supply. Sewers shall be kept removed from water supply wells or other water supply sources and structures. A minimum horizontal distance of ten (10) feet and a minimum vertical distance of eighteen (18) inches shall be maintained between parallel or crossing water and sewer lines (water above sewer). At points where sewers cross water mains, the sewer shall be constructed of a cast or ductile iron pipe or encased in concrete or other casing material approved by the City Engineer or Public Works Director for a distance of ten (10) feet in each direction from the crossing, measured perpendicular to the water line. However, all appropriate local, state and federal agency

requirements shall be followed. Distances shall be measured outside edge to outside edge of pipes.

13.3.7 Water Facility Design Standards

Required design specifications and construction standards for water facilities shall be subject to the review and approval of the City Engineer or Public Works Director.

13.3.8 Other Utilities Standards

- 1. All required design specifications and construction standards for other utilities not already described, for example telephone installations, cable television lines, pump stations, etc. will be determined by the City Engineer or Public Works Director.
- 2. Where subsurface electric lines are provided, ten (10) feet of easement shall be provided across proposed lots on one side of the street. A minimum of six (6) feet of separation shall be maintained from any electric line and another underground utility line. Where above-ground electric service is provided, a minimum of fifteen (15) feet of easement shall be granted across proposed lots on one side of the street.

13.3.9 Standards for Preservation of Natural Features and Amenities

- 1. Natural Drainageways Streams and Bayous Any subdivision site which is traversed by natural streams or bayous will have lots laid out so that the areas of periodic inundation adjacent to and including such drainageways streams or bayous are not calculated as a part of the required area of any lots in order to preserve the natural features of the these areas of periodic inundation adjacent to and including such areas and provide maximum drainage capacity while providing for the public's safety. No change to any natural drainageway stream or bayou shall be made by any person without the advance approval of the Planning Commission and all parish state or federal agencies with jurisdiction therein. The Planning Commission shall not allow change in a stream or bayou unless such change is shown to be necessary for the health, safety, or welfare of the present and future population of an area or necessary to the conservation of water, drainage, and sanitary facilities.
- 2. <u>Planting and Spacing of Trees</u> The planting or removal of both public and private trees shall be in accordance all applicable regulations of the City.

3. <u>Construction Around Trees</u> - No street paving with concrete, asphalt or other impervious material shall be placed within the drip line of any tree required to be preserved by these regulations or any other applicable city regulations. Soil and other materials shall not be temporarily or permanently within the drip line of any required tree in such a way as to cause suffocation or damage of root system.

13.3.10 Standards for Non-Residential and Townhouse Subdivisions

- 1. Non-Residential subdivisions shall be subject to all of the provisions of this requirements of the Planning Commission in conjunction with the adopted policies of the Comprehensive Land Use Plan of the City of Mandeville.
- 2. Townhouse subdivisions shall be subject to all applicable provisions of this Comprehensive Land Use Regulations Ordinance. However, townhouse lots may front on driveways held in common ownership by the owners of the townhouse lots having their access provided by such driveways, provided such driveways have direct access to a public street exceeding three hundred feet (300) in length.

In addition, when approved in conjunction with a site plan review subject to approval by ordinance in a Planned District, townhouse lots may front on open space maintained in common ownership by the owners of the townhouse lots fronting on such open space, provided adequate access under the same ownership as the owners of the townhouse lots, as described above, is also under the same ownership.

3. Subdivisions of Marine Uses

- a. Canals shall not be less than thirty-five (35) feet in width, six (6) feet in depth and have a clear navigation channel of twenty-four (24) feet.
- b. Street grades shall be of such elevations that they are twelve (12) inches above normal high tides.
- c. Additional requirements for Marine developments are provided in Division II, Article 8, Special Uses Criteria of this Land Use Regulations Ordinance.